## Seattle Municipal Coue June 2009 code update file June 2009 code update file Trable of Ordinances codified

See ordinan sections for and tables a this source

0)
8)
ment to City by grantee efray franchise
enses (21.64)
am point (1.20)
ice of street obstruction
utoff of water supply
12)
cellation of certain
rants (Repealed by
368)
ual franchise reports
64)
setting of awards
nst special assessments
ondemnation suits
76)
lth regulations;
rantine; collection of
page and swill (10.24,
26)
ranty Deposit Trust d establishment and
ration (5.76)
elty to animals pealed by 110230)
ance of bills for labor
naterials furnished by
City (5.20)
usal to return City
ks and records (3.104)
of foreclosed property
ity (20.80)
ce Department
pealed by 109757)
pital records (10.28)
tingent Fund A (5.76)
of stamps, seals,
nits and notices by
artment of Public
lth (3.30)
man's badges (3.16)
road barriers (11.94) ends § 17 of Ord.
ends § 17 of Ord. 57, dumping offal or
te (10.24)
eptance of
demnation Fund
rants (20.76)
oking prohibited in

l amending graphics, ccuracy of

For current SMC, polling places (2.20) to the Office of the City Clerk

June 2009 co

Text provide

See ordinantes sections for and tables at this source

	te file
26022	Wiping and absorbent
	materials (10.44)
26853	Amends § 24 of Ord.
A FOR III	15957, removal of garbage
C L	or refuse matter (Repealed
	by 109936)
27209	Assessments for re-laying
27209	
	and reconstruction of water
	mains (20.16)
28383	Common drinking cups
CES OF	(10.42)
28605	Cold storage eggs (10.20)
29205	Leaving drug or chemical
CULIBRIO	samples on private
	property (10.32)
22127	
32137	Designates dahlia as City
	flower (1.16)
32444	Physical examinations of
	arrested persons (10.30)
32967	Payment of local
222	improvement assessments
	against condemned
	property (20.76)
32986	Accident reports by
32980	
	physicians (10.28)
35083	Filling of private property
	(20.20)
35902	Contractors' bonds (20.48)
35932	Amends § 5 of Ord. 21593,
	railroad barriers (11.94)
38390	Appointment of deputies
30370	by City Comptroller
	(Repealed by 110938)
39034	
39034	Accounting procedure
20104	(5.12)
39104	Maintenance of healthful
	temperatures (Repealed by
	110152)
39520	Reports to City Council by
	departments regarding
	motor vehicles; repeals
	Ord. 36307 (4.68)
41908	Laundries; repeals Ord.
71700	
42.475	16003 (10.38)
43475	False and misleading
	advertising (7.08)
45312	Location of
	slaughterhouses, packing
	plants, and other offensive
45599	businesses (10.50)
45599	businesses (10.50) Contingent Fund A;
45599	businesses (10.50) Contingent Fund A; amends § 1 of Ord. 18132
	businesses (10.50) Contingent Fund A; amends § 1 of Ord. 18132 (5.76)
45599 45712	businesses (10.50)  Contingent Fund A; amends § 1 of Ord. 18132 (5.76)  Maintenance of sidewalks;
45712	businesses (10.50)  Contingent Fund A; amends § 1 of Ord. 18132 (5.76)  Maintenance of sidewalks; repeals Ord. 36160 (15.72)
	businesses (10.50)  Contingent Fund A; amends § 1 of Ord. 18132 (5.76)  Maintenance of sidewalks; repeals Ord. 36160 (15.72) Quitclaim on property
45712	businesses (10.50)  Contingent Fund A; amends § 1 of Ord. 18132 (5.76)  Maintenance of sidewalks; repeals Ord. 36160 (15.72)
45712	businesses (10.50)  Contingent Fund A; amends § 1 of Ord. 18132 (5.76)  Maintenance of sidewalks; repeals Ord. 36160 (15.72)  Quitclaim on property acquired by nonpayment of
45712	businesses (10.50)  Contingent Fund A; amends § 1 of Ord. 18132 (5.76)  Maintenance of sidewalks; repeals Ord. 36160 (15.72)  Quitclaim on property acquired by nonpayment of local improvement
45712	businesses (10.50)  Contingent Fund A; amends § 1 of Ord. 18132 (5.76)  Maintenance of sidewalks; repeals Ord. 36160 (15.72)  Quitclaim on property acquired by nonpayment of

generate only.

June 2009 co

Text provide

See ordinant sections for and tables a this source

C	ilber 4	40 Sile
ı	4-44003	re man
	46545	Contracts for public work
	47026	or improvement (20.48)
	47936	Amends §§ 2, 5, 6 and 8 of
9	C IOI	Ord. 39104, maintenance of healthful temperatures
		(Repealed by 110152)
	48022	License Code; repeals
	10022	Ords. 15855; 16299;
		20739; 20740; 28570;
	- co orea	31578 § 1050; 32034;
	ces cross	34159; 34821; 34837;
		35790; 36203; 36917;
	- ammle	37317; 37671 §§ 3, 4, 5, 30
	COIIIB	and 34; 40146; 42323;
		42775; 42949; 43645;
	- 4 to CC	43646; 44785; 44905;
A	ma to	44986; 45727; 46189 and
		47237 (6.02, 6.08, 6.14, 6.20, 6.36, 6.38, 6.42, 6.48,
	0210	6.68)
	48503	Amends § 28 of Ord.
	10303	48022, auctions and
		auctioneers (Repealed by
		109818)
	49002	Amends § 159 of Ord.
		48022, junk shops
		(Repealed by 109818)
	49946	Amends §§ 206-209 of
		Ord. 48022, side sewer
	51.622	contractors (6.52)
	51632	Adds §§ 4(a), 4(b) and 6(a) to Ord. 43475, false and
		misleading advertising
		(7.08)
	51903	Amends §§ 230 and 231 of
		Ord. 48022, temporary
		merchant (Repealed by
		109818)
	53052	Amends § 198 of Ord.
		48022, peddlers and
		solicitors (Repealed by
	52070	109271)
	53079	Amends §§ 19, 20 and 21
		of Ord. 48022, general License Code provisions
		(6.02)
	53281	Amends §§ 103, 104, 105,
	33201	108, 110, 112, 113 and 114
		of Ord. 48022, for-hire
		drivers (Repealed by
		109648)
	54547	Payment of cost and
		expense of condemnation
		of land; repeals Ords.
		34920, 37168 and 40746
	56463	Use of water by charitable
	JU <del>1</del> UJ	institutions (21.12)
	57142	Use of Auditorium Annex
	U, I 12	at Seattle Center (17.20)
,	. 01	COMPANY

eference only.

l amending
graphics,
ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

C	ilhar .	0210	
	a moda	te tille	
	58128	Amends § 10 of Ord.	
		54547, local improvement	
	a com hi	condemnation award bonds	
0	A TOL 182	(Repealed by 116368)	
	59400	Amends § 198 of Ord.	
		48022, peddlers and	
		solicitors (Repealed by	
	59597	109271) Amends § 23 of Ord.	
		48022, exhibitors or trade	
	ces crea	shows (6.20)	
<b>B</b> '	60494	Amends § 104 of Ord.	
		48022, for-hire drivers	
P	COMP	(6.22)	
	61093	Amends § 198 of Ord.	
	1 40 CC	48022, peddlers and	
	ma to	solicitors (Repealed by	
<u> </u>	61550	109271)	
	61550	Amends § 198 of Ord.	
	1116.	48022, peddlers and solicitors (Repealed by	
		109271)	
	61760	Amends § 28 of Ord.	
	31700	48022, auctions (Repealed	
		by 109818)	
	61761	Amends § 23 of Ord.	
		48022, exhibitors or trade	
		shows (Repealed by	
	61010	109818)	
	61818	Amends §§ 268, 269 and	
		270 of Ord. 48022, going out of business sales	
		(Repealed by 111441)	
	61850	Amends §§ 263, 264 and	
	01000	266 of Ord. 48022, tobacco	
		retailers (Repealed by	
		115531)	
	62364	Local Improvement	
		Guaranty Fund (5.76,	
		20.08)	
	62662	Utilities business tax (5.48)	
	63537	Amends § 183 1/2 of Ord. 48022, pawnbrokers	
		(Repealed by 109818)	
	63654	Fixes interest rate on	
		certain certificates of	
		purchase (20.76)	
	63678	Condemnation assessments	
		(20.76)	
	64185	Amends § 23 of Ord.	
		48022, exhibitors or trade	
	64567	shows (6.20)	
	64567	Amends § 58 of Ord. 48022, cabarets (Repealed	
		by 109653)	
	65088	Expiration dates of certain	
		licenses (6.14, 6.34, 6.52)	
	65313	Adds §§ 301-305 to Ord.	
		48022, locksmiths	
		(Repealed by 111441)	

eference only.

amending

applics,
accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

7 OF CHOI 48077	
of Ord. 48022,	
ense Code	27,
(6.02)	
19, 20 and 21	
2, general	
de provisions	
onstruction	
Distruction	
23 of Ord.	
ibitors or trade	
(i)	
23 of Ord.	
ibitors or trade	
0)201111	
58 of Ord.	
arets (Repealed	
arets (Repeared	
water supply	
s and	
(21.04)	
25 and 33 of	
, water supply	
04)	
2, 5, 7 and 9 of	
, false and	
advertising	
_	
103 of Ord.	
hire drivers	
y 109560)	
ed use of police	
A.58)	
ice; repeals § 6	
74 (Repealed	
of Ord. 35083,	
ivate property;	
. 18632 (20.20)	
3 of Ord. 41908,	
0.38)	
ndard Time;	
. 63635, 63714,	
64259 (1.12)	
of Ord. 4953,	
chise reports	
e with fire alarm	
signal systems	
to Ord. 66591,	
ice (Repealed	
ice (Repealed	
y seal; repeals	
(1.08)	
ice (Repealed	
	2 of Ord. 66591, ice (Repealed ontributions oyees (3.100)

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te file
	67473	Amends § 11, general
		License Code provisions;
		adds §§ 90-1-90-5 and
	A FOR III	amends §§ 89 and 90 of
3	CI ZO	Ord. 48022, detectives and
		detective agencies (6.02,
		6.14)
ľ	67585	Loan or rental of City
		equipment (3.108)
ŀ	68093	Adds § 8 to Ord. 21593,
4	SES CICE	railroad barriers (11.94)
	68362	Adds § 4C to Ord. 43474,
	100	false and misleading
	COMPLE	advertising; repeals Ords.
	6031-8	30493, 42666 and 68249
		(7.08)
_	68579	Amends § 1 of Ord. 57142,
	003/3	Auditorium Annex at
ļ	(0.49.4	Seattle Center (17.20)
	69484	Amends § 2 of Ord. 48022,
		general License Code
ļ	60502	provisions (6.02)
	69583	Adds § 7-A to Ord. 65877,
		water supply system
ļ		(21.04)
	69681	Amends §§ 236, 237, 238
		and 242 of Ord. 48022,
		for-hire vehicles and rental
		agency; repeals § 235 of
		Ord. 48022 and § 7 of Ord.
		62662 (Repealed by
L		110372)
	69823	Amends § 263 of Ord.
		48022, tobacco retailers
L		(Repealed by 115531)
	69829	Amends § 4 of Ord. 57142,
		Auditorium Annex at
		Seattle Center (17.20)
	70325	Amends § 13 of Ord.
		48022, general License
		Code provisions (6.02)
	70894	Amends §§ 3 and 4 of Ord.
		62364, Local Improvement
		Guaranty Fund (5.76,
		20.08)
Ī	70933	Amends §§ 260, 261 and
		262 of Ord. 48022,
		merchant patrolmen (6.38)
ſ	71189	Adds § 7-B to Ord. 65877,
		water supply system
		(Repealed by 108837)
ŀ	71451	Amends § 23 of Ord.
	, 1 .01	48022, exhibitors or trade
		shows (6.20)
ŀ	71460	Amends § 230 of Ord.
	/ 1400	48022, temporary
		merchants (Repealed by
- 1		109818)

reference only.

Id amending

graphics,
accuracy of

June 2009 co

Text provide

See ordinantes sections for and tables at this source

	ilhar .	C110
	a mada	te the
	71537	Adds § 7-2 to Ord. 66591,
		Special Police; repeals Ord.
	a com hi	71281 (Repealed by
	A TOK III	109759)
	71614	Registration of business
		owners for security
	71001	purposes (6.96)
	71881	Mechanical music machines; repeals Ords.
	- 0403	70621 and 70905 (6.100)
	71917	Municipal Light and Power
<b>B</b> '		Bond Reserve Fund (5.76)
	72318	Amends § 104 of Ord.
P	COINT	48022, for-hire drivers
		(Repealed by 109648)
	72495	Admission tax (5.40)
9	72529	Repeals §§ 171-177 and
		replaces with §§ 171-176
	C:10	of Ord. 48022, parlors and
	72585	bathhouses (6.36) Protection of property from
	-2303	fills and excavations
		(Repealed by 108865)
	72630	Business and occupation
		tax (5.44)
	72685	Amends § 23 of Ord.
		48022, exhibitors or trade
	72712	shows (6.20) Amends § 199 of Ord.
	72712	48022, secondhand dealers
		(Repealed by 109818)
	72834	Contracts for sale of
		foreclosed property (20.80)
	72857	Adds § 7-C to Ord. 65877,
		water supply system
		(21.04)
	72874	Amends § 43 of Ord.
		15957, air space and area in schools, hospitals,
		offices and factories
		(10.24)
	72989	Amends § 176 of Ord.
		48022, massage parlors and
		bathhouses (6.36)
	73016	Amends § 14 of Ord.
		48022, general License
	72022	Code provisions (6.02)
	73023	Amends § 6 of Ord. 72630, business and occupation
		tax (5.44)
	73223	Fumigators; repeals Ord.
		63833 (Repealed by
		109932)
	73335	Amends §§ 2, 3, 4, 10, 12
		and 13; repeals §§ 7 and 11
		of Ord. 72630, business
	72500	and occupation tax (5.44)
	73528	Cedar River Watershed; repeals Ords. 37296 and
		52259 (21.12)

eference only.

d amending

ver graphics,
50 accuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinantes sections for and tables at this source

a anda	te file
73583	Amends § 95-A of Ord.
	48022, dog and cat licenses
10-m h1	(Repealed by 110230)
73763	Amends §§ 103, 104, 105,
	106, 108 and 114; repeals
	§§ 107 and 109 of Ord.
	48022, for-hire driver's
	license (Repealed by
	109648)
74178	Renting of shoes and
ces cross	wearing apparel (10.40)
74182	Rodent control; repeals Ord. 17391 (10.34)
74223	Amends §§ 104 and 108 of
14223	Ord. 48022, for-hire
1 40 6	drivers (6.22)
75087	Amends § 13, general
	provisions; §§ 22 and 23,
	amusements; §§ 111 and
<b>\$110.</b>	112, for-hire drivers; § 131,
3330	steam engineers and boiler
	firemen; and § 164, junk
	shops and junk wagons of
	Ord. 48022, License Code
	(6.02, 6.20)
75088	Expiration dates of certain
	licenses (6.14, 6.52)
75115	Licenses for weighmasters,
	trailer camps, caravaning
	of motor vehicles, and
	master fumigators
	(Repealed by 109932)
75167	Amends § 1 of Ord. 54547,
	eminent domain
	proceedings; repeals Ord.
	59034 (20.76)
75207	Amends § 198 of Ord.
	48022, peddlers and
	solicitors (Repealed by
	109271)
75461	Amends § 176 of Ord.
	48022, massage parlors and
	bathhouses (6.36)
75725	Adds § 341 to Ord. 65877,
	water supply system
	(21.04)
75771	Amends §§ 1-9 and § 17 of
	Ord. 73223, fumigators
	(Repealed by 109932)
76148	Amends § 206 of Ord.
	48022, side sewer
	contractors (Repealed by
	111441)
77534	Amends § 7-A of Ord.
	65877, water supply
77331	
7,551	system (21.04)
77545	system (21.04) Amends § 23 of Ord.
	Amends § 23 of Ord.

eference only.

selection of a couracy of a

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	40 Sile
	ILE TITE
77699	Adds § 91-1 to and amends
	§ 89 of Ord. 48022,
a com hi	detectives and detective
A FOR IS	agencies (6.14)
77700	Amends §§ 3 and 7 of Ord.
	72495, admission tax
	(5.40)
77718	Adds §§ 308-314, 314a and
	314-b to Ord. 48022, used
- oc cred	automobile dealer
CE2 CT	(Repealed by 110136)
77754	Amends § 172 of Ord.
	48022, massage parlors and
CORRE	bathhouses (6.36)
77859	Amends § 1 of Ord. 38390,
140 0	appointment of deputies by
ing to c	City Comptroller
3350	(Repealed by 110938)
77869	Amends § 311 of Ord.
9112	48022, used-automobile
777	dealers (Repealed by
	110136)
77966	Amends § 199 of Ord.
	48022, secondhand dealers
	(Repealed by 109818)
78076	Weeds and vegetation
	nuisances (Repealed by
	114355)
78146	Amends §§ 207 and 209 of
	Ord. 48022, side sewer
	contractors (Repealed by
	111441)
78242	Amends § 95-a of Ord.
	48022, dog licenses
	(Repealed by 110230)
78444	City employees' retirement
	system; repeals Ord. 57075
	(4.36, 5.76)
79080	Amends § 262 of Ord.
	48022, merchant patrolmen
	(6.38)
79239	Amends § 309 of Ord.
	48022, used-automobile
	dealers (Repealed by
	110136)
79403	Adds § 2(bb) to and
	amends §§ 5, 14, 16, 19
	and 23 of Ord. 78444,
	retirement system (4.36)
79432	Combines City and County
	Health Departments
<b>5</b> 0.40.4	(Repealed by 109522)
79606	Amends § 107 of Ord.
	48022, for-hire drivers
	(Repealed by 109648)
79798	Amends § 6 of Ord. 78444,
	retirement system (4.36)
79849	Adds § 11-1 to Ord. 72495,
<b>50055</b>	admission tax (5.40)
79957	City office hours (3.102)

rd.

rd.

and lamending

draphics,

nd

occuracy of

For current SMC<del>, corres</del> the Office of the City Clerk

June 2009 co 80028 Preparation and sa poultry, rabbit and

Text provide

See ordinand sections for and tables a this source

	a mada	te file
	80028	Preparation and sale of
/	40	poultry, rabbit and fish
	101	(Repealed by 110052)
ŀ	80042	Amends § 7 of Ord. 65877,
ł	00042	water supply system
1		(21.04)
ŀ	80275	
l	80275	Adds § 184-1,
		pawnbrokers, and § 202-1,
ı		secondhand dealers, to
l	oos crea	Ord. 48022, License Code
Į	460	(Repealed by 109818)
l	80528	Days off for police and
ı		firemen; repeals Ord.
	COMPA	79314 (4.20)
ľ	80544	Amends § 111 of Ord.
l	1 4 0 00	48022, for-hire drivers
ı	and to co	(6.22)
ŀ	80601	Adds §§ 18-1, 18-2 and
١	00001	
١	0210	18-3 to and amends §§ 1, 3,
١	7116.	4, 8, 9, 10, 11, 12, 16 and
		18 of Ord. 80028,
		preparation and sale of
		poultry, rabbit and fish
l		(Repealed by 110052)
I	80689	Amends § 111 of Ord.
l		48022, for-hire drivers
		(Repealed by 109648)
İ	80738	Amends § 104 of Ord.
l	00730	48022, for-hire drivers
l		(Repealed by 109648)
ŀ	80818	
ŀ		Tattooing (6.102)
	80930	Amends § 161 of Ord.
l		48022, junk shops
ļ		(Repealed by 109818)
l	80950	Amends § 198 of Ord.
		48022, peddlers and
		solicitors (Repealed by
		109271)
ſ	81150	Adds § 11 to and amends
		§§ 4, 9, 12, 17, 20 and 27
		of Ord. 72630, business
		and occupations tax (5.44)
ŀ	81301	Cumulative Reserve Fund
	01301	(5.76)
ŀ	81352	Amends § 23 of Ord.
	01332	ŭ .
		48022, exhibitors or trade
ļ	21.152	shows (6.20)
	81479	Amends § 173 of Ord.
I		48022, massage parlors and
١	<u></u>	bathhouses (6.36)
f	81521	Amends §§ 2, 4, 5, 14, 16
		and 23 of Ord. 78444,
		retirement system (4.36)
ŀ	81551	Amends § 183 1/2 of Ord.
	01331	
1		48022, pawnbrokers
		(Repealed by 109818)
ŀ	81582	Rabies control (9.12)

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
81599	Impoundment of animals;
0,0	nuisances; repeals Ords.
	16004, 38751, 63755,
a for m	68044, 70026, 70408,
an real	72002, 72494, 72531,
	73122, 75476 and 75666
	(Repealed by 110230)
91709	
81698	Service award program for
	City officers and
CTEG	employees (4.88)
81727	Amends § 4 of Ord. 78444,
	retirement system (4.36)
81819	Amends § 161 of Ord.
COMPA	48022, junk shops
	(Repealed by 109818)
81830	Amends § 30 of Ord.
and to co	48022, auctions and
7.33	auctioneers (Repealed by
	109818)
82230	Amends § 2 of Ord. 66591,
1110.	Special Police (Repealed
	by 109759)
82498	Amends § 1 of Ord. 57142,
82498	Auditorium Annex at
82583	Seattle Center (17.20)
82585	Sewer construction fees;
02500	repeals Ord. 79865 (21.24)
82590	Amends § 1 of Ord. 63678,
	segregation of
	condemnation assessments
	(20.76)
82591	Amends § 14 of Ord.
	54547, cost and expense of
	condemnation of land
	(20.76)
82620	Contingent Fund B (5.76)
82624	Amends § 23 of Ord.
	48022, exhibitors or trade
	shows (6.20)
82716	Amends § 28 of Ord.
	48022, auctions (Repealed
	by 109818)
83500	Amends §§ 23, 49, 65, 147,
	183, 299, 299-A, 300A,
	300C, 300G, 328 and
	328-A of Ord. 48022,
	License Code (6.20, 6.48)
92524	
83534	Adds § 14(h) to and mends
	§§ 2(k), 9(c) and 16(d) of
	Ord. 78444, retirement
	system (4.36)
100=11	Amends § 2(1) of Ord.
83744	
83744	78444, retirement system
83744	(4.36)
83744 83825	
	(4.36)
	(4.36) Amends § 7-A of Ord. 65877, water supply
83825	(4.36) Amends § 7-A of Ord. 65877, water supply system (21.04)
	(4.36) Amends § 7-A of Ord. 65877, water supply

reference only.

It amending

graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

83854 Amends § 28 of Or 48022, auctions (Re by 109818) Vacancies in Charte offices (3.92)	
by 109818)  83897 Vacancies in Charte	
83897 Vacancies in Charte	epealed
	er
83906 Amends § 9 of Ord	48022
general License Co	
provisions (6.02)	
84040 Amends § 17 of Or	
72630, business and	
occupation tax (5.4 84041 Amends § 23 of Or	
48022, exhibitors o	
shows (6.20)	rtrade
84046 Seattle Fire District	No. 5
Fund (Repealed by	
112611)	,
84055 Septic tanks, cesspo grease traps, and se	
grease traps, and se pits (10.46)	cpage
84106 Milk regulations; re	epeals
Ord. 72174 (10.16)	
84249 Amends § 28 of Or	
48022, auctions (Re	epealed
by 109818)  84297 Refrigeration system	me:
repeals Ord. 80153	
84319 Amends § 11, gene	
provisions; adds §§	330
and 330.1-330.7, pa	
and peepshows to C 48022, License Coo	
repeals Ord. 84065	,
6.42)	(===,
84388 Amends § 9 of Ord	
refrigeration system	
84390 Sewerage rates and (5.76, 21.28)	cnarges
84414 Amends §§ 5 and 6	of Ord.
62662, utilities busi	
(5.48)	1
84510 Adds § 13(c) to Ord 78444, retirement s	
(4.36)	ystem
84566 Amends §§ 5 and 1	
Ord. 78444, retirem	ient
system (4.36)	:4
Refund of sewer pe fees (21.24)	ımıt
84706 Applications for str	eet use
and occupation per	
(15.04)	
85026 Adds § 9.1 to Ord.	
business and occup tax (5.44)	ation
85317 Sewer connections	outside
City limits (21.20)	2 2 2 2 2 2
85388 Adds § 2.1 to Ord.	
business and occup	ation
tax (5.44)	فالم

eference only.

and amending
graphics,
ccuracy of

For current SMC, tax (5.44) contact
the Office of the City Clerk

June 2009 co

Text provide

See ordinantes sections for and tables at this source

95417	ate III
85417	Sewerage rates and charges (21.28)
85544	Trespassing at garbage dumps (12A.64)
85557	Amends § 231 of Ord.
	48022, temporary
	merchants (Repealed by 109818)
85562	Amends §§ 24 and 25 of
ces cre	Ord. 78444, retirement system (4.36)
85679	Inspection of poultry (Repealed by 110052)
85744	Amends § 6 of Ord. 73223,
	fumigators (Repealed by 109932)
85746	Adds §§ 27.1-27.5 to Ord.
BRes .	84297, refrigeration
a-1 a	systems (6.82)
85773	Amends §§ 1, 4 and 11 of
	Ord. 81599, impoundment (Repealed by 110230)
85801	Adds § 2A to Ord. 71881,
	mechanical music
05005	machines (6.100)
85885	Amends § 5 of Ord. 62662, utilities business tax (5.48)
85918	Amends § 22 of Ord.
00,10	72630, business and
	occupation tax (5.44)
86133	Amends § 14 of Ord.
	78444, retirement system (4.36)
86164	Amends § 28 of Ord.
	72630, business and
86300	occupation tax (5.44)
80300	Zoning; repeals Ords. 36032 and 45382
	(Repealed by 117570)
86416	Amends § 5-A of Ord.
	48022, general License
86455	Code provisions (6.02) Amends §§ 13 and 36 of
	Ord. 65877, water supply
	system (21.04)
86654	Amends § 176.1 of Ord.
	48022, massage parlors and bathhouses (6.36)
86662	Amends § 1 of Ord. 81698,
	service award program
0.67.40	(4.88)
86749	Leash law for dogs (Repealed by 110230)
86799	Vacation leave; repeals
	Ord. 82522 (Repealed by
	110195)
86927	Suggestion awards
	program for City officers and employees; repeals
	Ords. 79944 and 81452
	(Repealed by 118966)

eference only.

I amending
graphics,
accuracy of

For current Sitter of the City Clerk the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

	a anda	te file
	87103	Amends § 2 of Ord. 72495,
	07170	admission tax (5.40)
	87178	Privately owned fire alarm
	d in in	or police signal systems
	97220	(10.08) Seattle Water Revenue
	87220	Bond Reserve Fund;
		repeals Ord. 87061 (5.76)
ŀ	87225	Amends §§ 3.03, 3.04,
		3.07, 3.13, 4.1, 5.3, 6.3,
	ces cross	9.1, 11.1, 11.2, 12.1, 12.2,
		13.1, 14.1, 15.2, 18.6, 19.6,
	- ammle	20.6, 21.5, 22.2, 22.3, 22.4,
	Colliba	22.5, 23.2, 23.6, 24.6 and
		26.2 of Ord. 86300, zoning
	1 40 66	(Repealed by 117570)
A	87546	Amends §§ 263 and 264 of
		Ord. 48022, tobacco
	C210	retailers (Repealed by 115531)
	87623	Amends subsection (d) of §
	01023	5 of Ord. 62662, utilities
		business tax (5.48)
Ì	87670	Seattle Sewerage Revenue
		Bond Reserve Fund (5.76)
	87916	Adds § 9-1 to Ord. 78444,
		retirement system (4.36)
	87983	Harbor Code; repeals Ords.
		34379, 34746, 34995,
		40555, 42337 and §§ 52-56
		of Ord. 48022 (5.76, 16.04,
		16.08, 16.12, 16.20, 16.24, 16.28, 16.32, 16.36, 16.40,
		16.44, 16.48, 16.52, 16.56,
		16.60, 16.64)
Ì	88046	General Donations of Gift
		Trust Fund (5.76)
	88208	Amends §§ 6 and 7 of Ord.
		65877, water supply
		system (21.04)
	88241	Amends §§ 1, 4 and 16 of
		Ord. 80028, preparation
		and sale of poultry, rabbit and fish (Repealed by
		110052)
ŀ	88270	Adds § 2.2 to and amends
		§§ 3, 11, 13, 17 and 18 of
		Ord. 72630, business and
		occupation tax (5.44)
	88283	Amends §§ 6.1, 13.5, 14.3,
		22.3, 22.4, 22.5, 24.2, 24.3,
		24.6, 26.3 and 27.41 of
		Ord. 86300, zoning
ļ	99410	(Repealed by 117570)
	88419	Police overtime at special events and emergencies
		(4.20)
ŀ	88479	Amends § 7 of Ord. 72495,
	'*	admission tax (5.40)
L		

eference only.

amending

applics,

classical actions

June 2009 co Text provide

See ordinant sections for and tables a this source

	igus Agairtí	te file
	88516	Amends § 3.05 "D" of Ord.
		86300, zoning (Repealed
	10-m hi	by 117570)
	88522	Sick leave (Repealed by
		117570)
	88537	Amends § 2 of Ord. 82620,
		Contingent Fund B (5.76)
	88581	Adds § 4-A to and amends
		§§ 1 and 4 of Ord. 80028,
	age crea	preparation and sale of
Ŋ		poultry, rabbit and fish
	20.504	(Repealed by 110052)
	88604	Amends § 58 of Ord.
	Com-2	48022, cabarets (Repealed
	00/14	by 109653)
	88614	Amends § 7-A of Ord. 65877, water supply
١	iller co	system (21.04)
	88650	Amends §§ 95-B and 95-C
	00000	of Ord. 48022, dog and cat
	1116.	licenses (Repealed by
		110230)
	88738	Amends § 57 of Ord.
		48022, cabarets (Repealed
		by 109653)
	88748	Amends §§ 2 and 3 of Ord.
		72495, admission tax
		(5.40)
	88780	Amends §§ 96-101; repeals
		§ 102 of Ord. 48022, dog
		kennels (Repealed by
	88789	111649)
	88789	Amends §§ 28,49, 53, 65, 90-1, 91-1, 93, 111, 118,
		131, 145-A, 153, 172, 206,
		242, 63, 267, 301 and 313
		of Ord. 48022, License
		Code (6.14, 6.36)
	88837	Amends §§ 160, 193, 195,
		300-C and 302 of Ord.
		48022, License Code
		(Repealed by 109271,
		111444)
	88838	Amends § 6-A of Ord.
		73223, fumigators and
		exterminators (Repealed by
	99940	109932)
	88840	Amends § 183H of Ord. 48022, pawnbrokers
		(Repealed by 109818)
	88841	Amends § 162 of Ord.
		48022, junk shops and junk
		wagons (Repealed by
		109818)
	88897	Amends §§ 2, 5, 16 and 19
		of Ord. 78444, retirement
		system (4.36)
	88901	Amends §§ 24 and 25 of
		Ord. 78444, retirement
		system (4.36)

eference only.

d amending
graphics,
accuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Shar	0210
	a anda	te file
	88920	Amends §§ 3.12, 6.3, 18.2
		and 28.1 of Ord. 86300,
	1 Sor 111	zoning (Repealed by
	88921	117570) Amends §§ 14.2, 15.2,
	00721	15.4, 22.2, 22.4, 23.4 and
		27.1 of Ord. 86300, zoning
		(Repealed by 117570)
	89021	Litter control (Repealed by
	000.40	113502)
Y	89048	Adds §§ 331 and 331.1-331.8 to Ord. 48022,
		bail bondsmen (Repealed
	COMPA	by 111441)
	89073	Amends § 2 of Ord. 78444,
	1400	retirement system (4.36)
4	89229	Amends §§ 10.1, 11.1,
		13.3, 15.2, 15.3, 18.2, 22.2,
	C:10	22.4, 26.2 and 26.3 of Ord. 86300; zoning (Repealed
	1116.	by 117570)
	89419	Amends § 15 of Ord.
		78444, retirement system
		(4.36)
	89420	Amends § 5 of Ord. 78444,
	89626	retirement system (4.36) Amends § 6.1 of Ord.
	89020	86300, zoning (Repealed
		by 110381)
	89634	Amends § 165 of Ord.
		48022, junk shops
		(Repealed by 109818)
	89712	Amends §§ 23.2, 26.2 and
		26.3 of Ord. 86300, zoning (Repealed by 117570)
	89715	Mobile homes and mobile
	0,7,10	home parks; repeals Ord.
		83452 (22.904)
	89733	Amends § 207 of Ord.
		48022, side sewer
		contractors (Repealed by 111441)
	89752	Adds § 17-1 to Ord. 78444,
	07,62	City employees' retirement
		system (4.36)
	89770	Amends § 104 of Ord.
		48022, for-hire drivers
	89787	(Repealed by 109648) Refund of blueprint and
	07101	photography fees (3.106)
	89796	Amends § 3.09, 6.4, 15.3,
		19.2 and 23.3 of Ord.
		86300, zoning (Repealed
	00050	by 117570)
	89859	Amends §§ 3.02 and 15.2 of Ord. 86300, zoning
		(Repealed by 117570)
	89860	Amends §§ 26.2 and 26.3
		of Ord. 86300, zoning
		(Repealed by 117570)

reference only.

amending

graphics,

curacy of

June 2009 co Text provide

See ordinantes sections for and tables at this source

	a mada	te file
	89902	Adds §§ 2-A, 2-B and 2-C
		to Ord. 82583, sewer
	- 0 - m hi	connection fees (21.24)
	89913	Engineering Department
3	CI ZOO	Operating Fund; repeals
		Ord. 45783 (Not codified)
	89939	Amends § 1 of Ord. 88522,
	0,,0,	sick leave (4.24)
	89983	Amends §§ 14.1, 14.2,
		15.2, 22.4 and 22.5 of Ord.
V	ces crea	86300, zoning (Repealed
2		by 117570)
	90012	Amends §§ 330.1 and
	0012	330.2 of Ord. 48022,
	COSSO	panorams and peepshows
		(6.42)
	90047	
٩	90047	Street use regulations;
		repeals Ords. 16081,
	0210	19346, 20152, 20536,
	711C.	21308 21389, 24836,
	880	29461, 29484, 30603,
		31148, 32515, 32578,
		34508, 36865, 38045 and
		83818 (12A.14, 15.02,
		15.04, 15.06, 15.08, 15.10,
		15.12, 15.14, 15.16, 15.18,
		15.20, 15.22, 15.24, 15.26,
		15.28, 15.32, 15.36, 15.38,
		15.40, 15.42, 15.44, 15.46,
		15.48, 15.50)
	90074	Amends § 198 of Ord.
		48022, peddlers and
		solicitors (Repealed by
		109271)
	90081	Amends § 12-a of Ord.
		48022, general License
		Code provisions (6.02)
	90010	Amends §§ 3.14, 4.1 and
		22.4 of Ord. 86300, zoning
		(Repealed by 117570)
	90138	Regulates signs adjacent to
	70130	freeways (Repealed by
	30130	freeways (Repealed by 117570)
		117570)
	90145	117570) Tobacco vending
		117570) Tobacco vending machines; repeals Ord.
		Tobacco vending machines; repeals Ord. 80009 (Repealed by
	90145	117570) Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531)
		Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork
	90145	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12)
	90145	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems;
	90145 90169 90181	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems; repeals Ord. 85895 (21.32)
	90145	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems; repeals Ord. 85895 (21.32) Adds § 2-D to and amends
	90145 90169 90181	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems; repeals Ord. 85895 (21.32) Adds § 2-D to and amends §§ 2-A and 2-B of Ord.
	90145 90169 90181	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems; repeals Ord. 85895 (21.32) Adds § 2-D to and amends §§ 2-A and 2-B of Ord. 82583, sewer connection
	90145 90169 90181 90233	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems; repeals Ord. 85895 (21.32) Adds § 2-D to and amends §§ 2-A and 2-B of Ord. 82583, sewer connection fees (21.24)
	90145 90169 90181	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems; repeals Ord. 85895 (21.32) Adds § 2-D to and amends §§ 2-A and 2-B of Ord. 82583, sewer connection fees (21.24) Adds §§ 336-346 to Ord.
	90145 90169 90181 90233	Tobacco vending machines; repeals Ord. 80009 (Repealed by 115531) Tolt River South Fork Watershed (21.12) Sewage disposal systems; repeals Ord. 85895 (21.32) Adds § 2-D to and amends §§ 2-A and 2-B of Ord. 82583, sewer connection fees (21.24)

eference only.

I amending graphics, ccuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

1-	40 Sile
4 440012	ITE Pres
90327	Amends § 198 of Ord.
	48022, peddlers and
a com hi	solicitors (Repealed by
4 TOF 15	109271)
90379	Garbage collection rates
	and charges (Repealed by
	109022, 109130)
90480	Amends § 198 of Ord.
	48022, peddlers and
0102	solicitors (Repealed by
ces crea	109271)
90497	Arterial City Street Fund
70477	(5.76)
90511	Amends subsection (h) of §
90311	
	5 of Ord. 62662, utilities
400	business tax (5.48)
90535	Vehicle wrecker and
	towing services (Repealed
	by 109080)
90578	Abatement Revolving Fund
777	(5.76)
90653	Amends §§ 2, 7, 13, 18, 19,
	22 and 23 of Ord. 87983,
	Harbor Code (16.04, 16.20,
	16.28, 16.44)
90685	Amends § 4 of Ord. 72495,
70003	admission tax (5.40)
90695	
90093	Underground utility district
00700	(21.68)
90708	Home fallout shelters
0070	(Repealed by 111200)
90722	Amends §§ 3.16, 3.26, 6.3,
	12.2,13.1, 13.2, 22.3, 22.4
	and 24.2 of Ord. 86300,
	zoning (Repealed by
	117570)
90727	Reenacts and amends §§
	152 and 153 of Ord. 48022,
	hawkers (Repealed by
	109271)
90730	Amends § 14(f) of Ord.
	78444, retirement system
	(4.36)
90788	Amends § 1 of Ord. 86799,
70700	vacation leave (4.32)
00780	Retirement sick leave
90789	
00000	(4.24)
90880	Amends § 58-A of Ord.
	48022, cabarets (Repealed
	by 109653)
90881	Disability compensation
	(4.44)
90897	Amends § 183 of Ord.
	48022, pawnbrokers
	(Repealed by 109818)
90952	Ambulance licensing
	(Repealed by 110151)
İ	
91048	Amends & 4 of Ord 90370
91048	Amends § 4 of Ord. 90379,
91048	Amends § 4 of Ord. 903/9, garbage collection rates and charges (21.40)

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	igus and a	te file
	91058	Amends § 3 of Ord. 4953,
		annual franchise reports
	10 - m hi	(21.64)
	91059	Amends § 148-A of Ord.
		48022, public garage or
		parking lot (6.48)
	91072	Registration of hotel guests
		(6.98)
	91138	Amends §§ 14.2, 15.2,
	age crea	17.2, 20.2, 21.3, 22.4, 23.1,
У		23.2, 23.3, 25.1 and 27.1 of
	10	Ord. 86300, zoning
	91141	(Repealed by 117570) Amends § 7 of Ord. 90535,
,	91141	vehicle wrecker and towing
	1 1 - 06	services (Repealed by
	and to co	109080)
ુ	91144	Amends § 1 of Ord. 90881,
		disability compensation
	2112	(4.40)
	91201	Amends § 5.3 of Ord.
		86300, zoning (Repealed
		by 117570)
	91202	Amends § 23.3 of Ord.
		86300, zoning (Repealed
	2122	by 117570)
	91208	Adds § 4.1 to and amends
		§§ 3, 4, 5, 9 and 11 of Ord.
		84390, sewerage rates and charges (5.76, 21.28)
	91281	Amends § 207 of Ord.
	71201	48022, side sewer
		contractors (Repealed by
		111441)
	91345	Amends § 15.3 of Ord.
		86300, zoning (Repealed
		by 117570)
	91436	Amends §§ 2, 3 and 4 of
		Ord. 82583, sewer
	91478	connection fee (21.24)
	714/0	Fallout shelter licenses (10.04)
	91513	Amends § 23 of Ord.
	, 1010	87983, Harbor Code
		(16.28)
	91667	Transfer to Sewer Fund
		(5.76)
	91675	Treasurer's Office
		Operating Fund (Repealed
	01.600	by 110826)
	91689	Amends § 23 of Ord.
		48022, exhibitors or trade
	91700	shows (6.20) Amends §§ 3.22, 6.4, 7.4,
	71/00	Amends §§ 3.22, 6.4, 7.4, 8.4, 9.4, 10.4, 11.4, 12.4,
		13.4, 14.2, 14.3, 15.3, 16.3,
		17.3, 18.3, 19.3, 21.3 23.4
		and 26.2; repeals § 20.3 of
		Ord. 86300, zoning
		(Repealed by 117570)
,	461	C' COTTO

eference only.

The state of th

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	cipal of		
	- Aa	te file	
	91749	Amends §§ 6, 9, 14, 18 and	
		19 of Ord. 90047, street	
	1 Car hi	use regulations (15.02,	
	91775	15.04, 15.10, 15.12) Adds §§ 14 and 15 to and	
	71773	amends §§ 1-7 and 9 of	
		Ord. 72495, admission tax	
	01010	(5.40)	
	91819	Transfers debris collection barge operation to	
	ces crea	Engineering Department	
		(Repealed by 118409)	
	91885	Seattle Center Advisory	
	Comple	Commission (Repealed by 108936)	
	91943	Amends § 196-A of Ord.	
	and to co	48022, peddlers and	
•		solicitors (Repealed by	
	91953	109271)	
	91933	Urban Renewal Operating Fund (5.76)	
	91981	Prohibits rental of Seattle	
		Center buildings to	
		subversive organizations;	
		repeals Ords. 68445 and 81315 (Repealed by	
		109118)	
	91987	Amends §§ 3 and 6 of Ord.	
		90145, tobacco vending machines (Repealed by	
		115531)	
	92030	Amends § 23.3 of Ord.	
		86300, zoning (Repealed	
	92059	by 117570) Amends §§ 4.1, 6.3, 12.2,	
	72037	13.1, 13.2, 23.2 and 24.4 of	
		Ord. 86300, zoning	
	02005	(Repealed by 117570)	
	92085	Adds § 2-A to Ord. 61098, meat shop licenses; adds §	
		16-A to Ord. 80028,	
		preparation and sale of	
		poultry, rabbit and fish (Repealed by 110052)	
	92113	Adds §§ 4.2 and 4.3 to and	
		amends §§ 4, 4.1 and 5 of	
		Ord. 84390, sewerage rates	
	92193	and charges (21.28) Amends § 5 of Ord. 78444,	
	72173	retirement system (4.36)	
	92204	Amends § 23 of Ord.	
		48022, exhibitors or trade	
	92247	shows (6.20) Seattle Center gas line	
	/44T1	permits (17.28)	
	92252	Amends § 34 of Ord.	
		90047, street use	
	92358	regulations (15.40) Utility Clearing Fund	
		(5.76)	

reference only.

a mending
graphics,
y ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

		te tille
92383	1117430	Amends § 2 of Ord. 66591,
(300)		Special Police (Repealed
	- 101	by 109759)
92405	OF III	Amends 34 of Ord. 90047,
W P		street use regulations
		(15.40)
92478		Building Department
		Operating Fund (5.76)
92479		Seattle Center Fund (5.76)
92492		Amends §§ 12.1, 18.2,
GEO		18.3, 19.2 and 23.3 of Ord.
		86300, zoning (Repealed
	400 MIC	by 117570)
92507		Amends § 267 of Ord.
		48022, going out of
	40 C	business sales (Repealed
		by 111441)
92657		Amends § 23 of Ord.
021		48022, exhibitors or trade
92729	<u> </u>	shows (6.20)
92129		Seattle-King County Youth Commission (3.66)
92781		Amends § 331.3 of Ord.
92/01		48022, bail bondsmen
		(Repealed by 111441)
92792		Trespass on Seattle Center
72172		property (17.12)
92869		Amends § 1 of Ord. 91885,
)200)		Seattle Center Advisory
		Commission (Repealed by
		108936)
92886		Amends §§ 13.4, 15.2, 15.3
		and 16.3 of Ord. 86300,
		zoning (Repealed by
		117570)
92887		Harbor and waterfront
		uses; Lake Union
		Construction Limit Line
		(Repealed by 117570)
92909		Amends § 4.3 of Ord.
		84390, sewerage rates and
02025		charges (21.28)
92935		Amends § 1 of Ord. 71917, Municipal Light and Power
		Bond Reserve Fund (5.76)
92946		Amends § 1 of Ord. 79957,
72740		City office hours (3.102)
92986		Amends § 338 of Ord.
		48022, vehicle wreckers
72700		, ,
72780		(Repealed by 109080)
92987		(Repealed by 109080) Food-service
		Food-service
		Food-service establishments; repeals
		Food-service establishments; repeals Ord. 74360 and § 2 of Ord. 87022 (Repealed by 109910)
		Food-service establishments; repeals Ord. 74360 and § 2 of Ord. 87022 (Repealed by
92987		Food-service establishments; repeals Ord. 74360 and § 2 of Ord. 87022 (Repealed by 109910) Amends §§ 27.1 and 28.1 of Ord. 86300, zoning
92987		Food-service establishments; repeals Ord. 74360 and § 2 of Ord. 87022 (Repealed by 109910) Amends §§ 27.1 and 28.1 of Ord. 86300, zoning (Repealed by 117570)
92987		Food-service establishments; repeals Ord. 74360 and § 2 of Ord. 87022 (Repealed by 109910) Amends §§ 27.1 and 28.1 of Ord. 86300, zoning (Repealed by 117570) Amends § 90 of Ord.
92987		Food-service establishments; repeals Ord. 74360 and § 2 of Ord. 87022 (Repealed by 109910) Amends §§ 27.1 and 28.1 of Ord. 86300, zoning (Repealed by 117570)

eference only.

amending
amending
graphics,
accuracy of

For current Single (0.14)

the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

93051	Amends § 13 of Ord.
93051	
	48022, general License
93065	Code provisions (6.02)
93003	Amends § 1 of Ord. 90789,
02066	retirement sick leave (4.24)
93066	Adds § 10-A to and
	amends § 1 of Ord. 88522,
02145	sick leave (4.24)
93145	Fire protection service
	outside City limits (3.16)
93213	Amends § 6 of Ord. 83834,
10	health care program
	(Repealed by 114893)
93257	Amends § 3 of Ord. 88522,
02247	sick leave (4.24)
93267	Optional health coverage
	by Group Health
	Cooperative (Repealed by
02201	114893)
93291	Survey and Planning
	Funds, Wash. R-5 and
	R-13; abolishes Urban
	Renewal-Federally
0000	Assisted Fund (5.76)
93350	Amends § 4.1 of Ord.
	86300, zoning (Repealed
00010	by 117570)
93360	Amends §§ 3 and 5 of Ord.
	72630, business and
02251	occupation tax (5.44)
93364	Amends § 1 of Ord. 86799,
00010	vacation leave (4.32)
93369	Amends § 90-1 of Ord.
	48022, detectives and
02207	detective agencies (6.14)
93397	Amends § 23 of Ord.
	48022, exhibitors or trade
93424	shows (6.20)
93424	Amends § 14.4 of Ord.
	86300, zoning (Repealed
02572	by 117570)
93562	Poverty programs (3.22)
93603	Official bonds; repeals
	Ords. 78635 and 80664
02514	(3.90)
93614	Increase balance of
02617	Contingent Fund D (5.76)
93617	Amends §§ 6.2, 7.2 and
	10.1 of Ord. 86300, zoning
02/21	(Repealed by 117570)
93631	Amends § 1 of Ord. 90789,
02771	retirement sick leave (4.24)
93771	Prohibits sale, gift or
	coloring of baby rabbits or
	fowl (Repealed by 109348)
93789	Cabulance licenses
	(Repealed by 109348)
93836	Amends § 2-B of Ord.
	82583, sewer connection
	fees (21.24)

eference only. d amending graphics, ccuracy of

For current SM (21.24) The second of the sec the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

a anda	te tile
94021	Amends § 8 of Ord. 93789,
	cabulance licenses
- 0 - m hi	(Repealed by 109348)
94022	Amends §§ 4 and 5 of Ord.
	90379, garbage collection
	rates and charges (21.40)
94034	Amends § 18.4 of Ord.
	86300, zoning (Repealed
	by 117570)
94035	Adds § 24.7 to Ord. 86300,
GES CI CO	zoning (Repealed by
	117570)
94036	Amends §§ 2.3, 2.4, 3.02,
	3.04, 3.17, 3.20, 3.23, 3.27,
	4.1, 16.1, 16.2, 16.3, 16.4,
1 1 - 0	16.5, 16.6, 16.7, 17.1, 17.2,
and to co	17.3, 17.4, 17.5, 17.6, 17.7,
	18.2, 23.1, and 24.6; adds
	§§ 17A.1, 17A.2, 17A.3,
0210	17A.4, 17A.5, 17A.6 and
1116.	
	17A.7; repeals § 27.4 of
	Ord. 86300, zoning
0.41.01	(Repealed by 117570)
94101	Amends § 2 of Ord. 66591,
	Special Police (Repealed
	by 109759)
94116	Amends § 5 of Ord. 62662,
	utilities business tax (5.48)
94174	Adds § 5 to Ord. 87178,
	privately owned fire alarm
	systems (10.08)
94181	Franceska Ballinger
	Memorial subaccount
	under the Civic Arts
	Account of the General
	Donations and Gift Trust
	Fund (5.76)
94188	Adds § 262-1 to and
	amends §§ 11, 11-A, 11-B,
	91-1, 260, 261 and 262 of
	Ord. 48022, License Code
	(6.02, 6.14, 6.38)
94190	Investment of City funds
	(Repealed by 110749)
94209	Amends § 20 of Ord.
	72630, business and
	occupation tax (5.44)
94213	Amends §§ 2-A, 2-B and
	2-C of Ord. 82583, sewer
	connection fees (21.24)
94283	Westlake Square Fountain
	subaccount under Civic
	Arts Account of General
	Donations of Gift Trust
	Fund (5.76)
0/1280	Signatures on initiative and
94289	
04202	referendum petitions (2.16)
94302	Amends § 3; repeals §§ 5
	and 6 of Ord. 72834,
	contracts for sale of foreclosed property (20.80)

eference only. amending graphics, ecuracy of

the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
94330	Amends §§ 11, general
94330	
	License Code provisions,
1 Co# M1	and §§ 57, 58 and
d for in	58-A,cabarets, of Ord.
	48022, License Code
	(6.02)
94348	Amends § 63 of Ord.
	48022, cabarets (Repealed
	by 109653)
94354	Amends §§ 2(m) and 14(f)
CG2 C2	of Ord. 78444, retirement
	system (4.36)
94366	Amends § 3 of Ord. 72495,
COURT	admission tax (5.40)
94383	Amends §§ 19.2 and 20.2
1 1-0	of Ord. 86300, zoning
and to C	(Repealed by 117570)
94436	Amends § 41 of Ord.
77730	90047, excavations (15.44)
04446	
94446	Seattle Center Department
24455	(17.04)
94465	Meat distribution and sale;
	repeals Ord. 61098 (5.76)
94505	Amends §§ 330 and 330.1
	of Ord. 48022, panorams
	and peepshows (6.42)
94524	Frozen dairy food products
	(10.18)
94555	Amends § 9 of Ord. 72630,
	business and occupation
	tax (5.44)
94567	Amends § 11 of Ord.
	94465, distribution and sale
	of meat (Repealed by
	110052)
94568	Amends § 153 of Ord.
7.500	48022, hawkers (Repealed
	by 109271)
94586	Adds § 2-A to and amends
74300	§§ 1, 2, 3 and 4 of Ord.
	90138, signs adjacent to
	freeways (Repealed by
0.45.97	117570)
94587	Amends §§ 2, 3, 19, 27, 28
	and 31 of Ord. 87983,
	Harbor Code (16.04, 16.12,
	16.20, 16.40, 16.44)
94595	Repeals §§ 130-137, 137-A
	and 138-145; adds §§
	130-141, 141.5, 142, 142.5,
	143, 144 and 145 to Ord.
	48022, steam engineers and
	boiler firemen (Repealed
	by 111301)
94622	Amends §§ 4 and 5 of Ord.
	90952, ambulance
	licensing (Repealed by
	110151)
L	· /

reference only.

an ending

an aphics,
curacy of

Seattle Municipal Code
June 2009 co 94625 update
Amend
176.1 o

Text provide

See ordinant sections for and tables a this source

Cibas	to file
	ALE BASS
94625	Amends §§ 171, 172 and
	176.1 of Ord. 48022, massage parlors and
1 SOT 11	bathhouses (6.36)
94649	Amends § 90-3 of Ord.
74047	48022, detectives and
	detective agencies (6.14)
94711	Amends §§ 4, 5 and 9 of
	Ord. 90379, garbage
LOC CTE	collection rates and charges
CG2 CT	(21.40)
94730	Traffic Violations Bureau
94767	(11.90)  Refunds from Seattle
94707	Center Fund (5.76)
94880	Amends § 2-B of Ord.
and to	82583, sewer connection
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	fees (21.24)
94893	Amends § 1 of Ord. 94730,
file.	Traffic Violations Bureau (11.90)
94970	Amends §§ 3.23, 24.2 and
94970	28.1 of Ord. 86300, zoning
	(Repealed by 117570)
95008	Amends §§ 104, 108, 110
	and 111 of Ord. 48022,
	for-hire drivers (Repealed
	by 109648)
95144	Amends §§ 240-244 of
	Ord. 48022, vehicle rental
	agency (Repealed by 110372)
95183	Amends §§ 14(f) and
	16(a); adds § 16(e) to Ord.
	78444, retirement system
	(4.36)
95244	Reimbursement for certain
	travel expenses of
	Department of Lighting
	employees (Repealed by 119578)
95246	Amends §§ 3 and 4 of Ord.
332.0	66591, Special Police
	(Repealed by 109759)
95256	Police holidays (4.20)
95317	Amends § 242 of Ord.
	48022, vehicle rental
	agency (Repealed by
27212	110372)
95318	Amends § 11 of Ord. 48022, general License
	Code provisions (6.02)
95389	Amends § 3 of Ord. 86799,
73307	vacation leave (4.32)
95390	Amends § 1 of Ord. 81698,
	service award program
	(4.88)
95463	Amends § 1 of Ord. 66705,
	Standard Time (1.12)

d amending graphics, accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	a - anda	te tile
	95466	Group term life insurance
		program (Repealed by
ļ	Sor Hi	114893)
	95526	Renumbers and amends §
1		7-A to be 7-A-1; adds §§
		7-A-2 and 7-A-3, water
ļ		supply system (21.04)
	95659	Amends § 58 of Ord.
		48022, cabarets (Repealed
ļ	oc crea	by 109653)
	95661	Amends § 7 and repeals § 5
		of Ord. 16374, Police
l	<u> </u>	Department (3.28)
١	95698	Amends § 12.5 of Ord.
		86300, zoning (Repealed
L	140 66	by 117570)
i	95751	Payment for use of private
1		automobile on City
l		business (4.68)
ĺ	95776	Driveway regulations
l	2320.	(15.06)
ĺ	95820	Amends § 142 and repeals
		§ 142.5 of Ord. 48022,
		steam engineers and boiler
		firemen (Repealed by
l		111301)
	95823	Amends §§ 10 and 11 of
		Ord. 90047, street use
l		regulations (15.04, 15.06)
	95906	Amends §§ 1, 3, 4, 6, 8, 10,
		11, 25, 26, 28 and 32;
		repeals § 22 of Ord. 94465,
		distribution and sale of
ļ		meat (Repealed by 110052)
	95946	Amends § 1 of Ord. 91885,
		Seattle Center Advisory
		Commission (Repealed by
ļ		108936)
l	96003	Collection and disposal of
l		garbage and refuse; repeals
ļ	0.4020	Ord. 91356 (21.36)
	96020	Street vacation procedures
ļ	0.4021	(Repealed by 109740)
	96031	Amends §§ 3.08, 3.09,
	70031	
	70031	3.17, 10.2, 15.3 and 24.4 of
	70051	Ord. 86300, zoning
		Ord. 86300, zoning (Repealed by 117570)
	96110	Ord. 86300, zoning (Repealed by 117570) Amends § 1 of Ord. 94465,
		Ord. 86300, zoning (Repealed by 117570) Amends § 1 of Ord. 94465, distribution and sale of
	96110	Ord. 86300, zoning (Repealed by 117570) Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052)
		Ord. 86300, zoning (Repealed by 117570) Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052) Amends §§ 1, 3, 4, 5, 6, 7,
-	96110	Ord. 86300, zoning (Repealed by 117570) Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052) Amends §§ 1, 3, 4, 5, 6, 7, 9, 10, 16, 18, 18-1, 18-2
	96110	Ord. 86300, zoning (Repealed by 117570)  Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  Amends §§ 1, 3, 4, 5, 6, 7, 9, 10, 16, 18, 18-1, 18-2 and 20 of Ord. 80028,
	96110	Ord. 86300, zoning (Repealed by 117570)  Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  Amends §§ 1, 3, 4, 5, 6, 7, 9, 10, 16, 18, 18-1, 18-2 and 20 of Ord. 80028, preparation and sale of
	96110	Ord. 86300, zoning (Repealed by 117570)  Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  Amends §§ 1, 3, 4, 5, 6, 7, 9, 10, 16, 18, 18-1, 18-2 and 20 of Ord. 80028, preparation and sale of poultry, rabbit, and fish
	96110 96125	Ord. 86300, zoning (Repealed by 117570)  Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  Amends §§ 1, 3, 4, 5, 6, 7, 9, 10, 16, 18, 18-1, 18-2 and 20 of Ord. 80028, preparation and sale of poultry, rabbit, and fish (Repealed by 110052)
	96110	Ord. 86300, zoning (Repealed by 117570)  Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  Amends §§ 1, 3, 4, 5, 6, 7, 9, 10, 16, 18, 18-1, 18-2 and 20 of Ord. 80028, preparation and sale of poultry, rabbit, and fish (Repealed by 110052)  Amends § 2 of Ord. 90695,
	96110 96125	Ord. 86300, zoning (Repealed by 117570)  Amends § 1 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  Amends §§ 1, 3, 4, 5, 6, 7, 9, 10, 16, 18, 18-1, 18-2 and 20 of Ord. 80028, preparation and sale of poultry, rabbit, and fish (Repealed by 110052)

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

96153	cipar		
of Ord. 86300, zoning (Repealed by 117570)  Amends § 61 of Ord.48022, cabarets (Repealed by 109653)  Amends § 2.2, 3.03, 3.08, 3.21, 3.22, 3.27, 4.1 and 13.5; adds §§ 13A.1-13A.6 and 13B.1-13B.6 to Ord. 86300, zoning (Repealed by 117570)  96234  Amends § 2 of Ord. 82620, Contingent Fund B (5.76)  Amends § 1 of Ord. 3622, cancellation of certain warrants (5.16)  96252  Amends §§ 3.13, 3.17, 3.20, 22.4, 24.6, 27.1 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96278  Amends §§ 3.05 and 23.3 of Ord. 86300, zoning (Repealed by 117570)  96293  Adds § 7.1 to and amends § 14(f) of Ord. 78444, retirement system (4.36)  96298  Contracts for rental of office space in Blue Spruce Building (Repealed by 112790)  96301  Adds § 4.1 to and amends § 4 of Ord. 90379, garbage collection rates and charges (21.40)  96348  Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds § 83-975 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends § 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds § 14A.1-14A.7 to and amends § 3.3, 3.7, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	a mada	te tille	
Repealed by 117570    Amends § 61 of Ord.48022, cabarets (Repealed by 109653)   George   Amends § 2.2, 3.03, 3.08, 3.21, 3.22, 3.27, 4.1 and 13.5; adds § 13A.1-13A.6 and 13B.1-13B.6 to Ord. 86300, zoning (Repealed by 117570)   George   Amends § 2 of Ord. 82620, Contingent Fund B (5.76)     George   Amends § 2 of Ord. 86220, Contingent Fund B (5.76)     George   Amends § 3.13, 3.17, 3.20, 22.4, 24.6, 27.1 and 28.1 of Ord. 86300, zoning (Repealed by 117570)     George   Amends § 3.05 and 23.3 of Ord. 86300, zoning (Repealed by 117570)     George   Amends § 3.05 and 23.3 of Ord. 86300, zoning (Repealed by 117570)     George   Adds § 7.1 to and amends § 14(f) of Ord. 78444, retirement system (4.36)     George   Contracts for rental of office space in Blue Spruce Building and 225 Mercer Building (Repealed by 112790)     George   Ge	96153	Amends §§ 16.6 and 17.6	
Amends \( \) 61 of Ord.48022, cabarets (Repealed by 109653)			
Ord.48022, cabarets (Repealed by 109653)	- C h1		
Repealed by 109653    Amends §§ 2.2, 3.03, 3.08, 3.21, 3.22, 3.27, 4.1 and 13.5; adds §§ 13A.1-13A.6 and 13B.1-13B.6 to Ord. 86300, zoning (Repealed by 117570)   96234	96197		
Amends §§ 2.2, 3.03, 3.08, 3.21, 3.22, 3.27, 4.1 and 13.5; adds §§ 13A.1-13A.6 and 13B.1-13B.6 to Ord. 86300, zoning (Repealed by 117570)			
3.21, 3.22, 3.27, 4.1 and 13.5; adds § \$13A.1-13A.6 and 13B.1-3B.6 to Ord. 86300, zoning (Repealed by 117570)  96234			
13.5; adds §§ 13A.1-13A.6 and 13B.1-13B.6 to Ord. 86300, zoning (Repealed by 117570)	96202		
and 13B.1-13B.6 to Ord. 86300, zoning (Repealed by 117570)			
86300, zoning (Repealed by 117570)	- 2000		
by 117570    96234	age crea		
Amends § 2 of Ord. 82620,			
Contingent Fund B (5.76)	06224		
Amends § 1 of Ord. 3622, cancellation of certain warrants (5.16)	90234		
Cancellation of certain warrants (5.16)	06220		
warrants (5.16)	90239		
Amends §§ 3.13, 3.17, 3.20, 22.4, 24.6, 27.1 and 28.1 of Ord. 86300, zoning (Repealed by 117570)	ad to co		
3.20, 22.4, 24.6, 27.1 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96278	06252		
28.1 of Ord. 86300, zoning (Repealed by 117570)  96278	90232		
(Repealed by 117570)  96278  Amends §§ 3.05 and 23.3 of Ord. 86300, zoning (Repealed by 117570)  96293  Adds § 7.1 to and amends § 14(f) of Ord. 78444, retirement system (4.36)  96298  Contracts for rental of office space in Blue Spruce Building and 225 Mercer Building (Repealed by 112790)  96301  Adds § 4.1 to and amends § 4 of Ord. 90379, garbage collection rates and charges (21.40)  96348  Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	C:10		
Amends §§ 3.05 and 23.3 of Ord. 86300, zoning (Repealed by 117570)	1116.	_	
of Ord. 86300, zoning (Repealed by 117570)  96293 Adds § 7.1 to and amends § 14(f) of Ord. 78444, retirement system (4.36)  96298 Contracts for rental of office space in Blue Spruce Building and 225 Mercer Building (Repealed by 112790)  96301 Adds § 4.1 to and amends § 4 of Ord. 90379, garbage collection rates and charges (21.40)  96348 Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352 Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360 Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366 Rapid Transit Study Fund (5.76)  96372 Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395 Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400 Amends § 23 of Ord. 48022, exhibitors or trade	06278		
Repealed by 117570	90278		
Adds § 7.1 to and amends § 14(f) of Ord. 78444, retirement system (4.36)			
\$ 14(f) of Ord. 78444, retirement system (4.36)  96298	96293		
retirement system (4.36)  96298  Contracts for rental of office space in Blue Spruce Building and 225 Mercer Building (Repealed by 112790)  96301  Adds § 4.1 to and amends § 4 of Ord. 90379, garbage collection rates and charges (21.40)  96348  Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	70273		
Contracts for rental of office space in Blue Spruce Building and 225 Mercer Building (Repealed by 112790)     96301			
office space in Blue Spruce Building and 225 Mercer Building (Repealed by 112790)  96301  Adds § 4.1 to and amends § 4 of Ord. 90379, garbage collection rates and charges (21.40)  96348  Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	96298	• • • • • • • • • • • • • • • • • • • •	
Building and 225 Mercer Building (Repealed by 112790)  96301  Adds § 4.1 to and amends § 4 of Ord. 90379, garbage collection rates and charges (21.40)  96348  Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	70276		
Building (Repealed by 112790)			
112790    96301			
Adds § 4.1 to and amends § 4 of Ord. 90379, garbage collection rates and charges (21.40)   96348			
\$ 4 of Ord. 90379, garbage collection rates and charges (21.40)  96348  Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends \$ 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds \$\$ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends \$\$ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds \$\$ 14A.1-14A.7 to and amends \$\$ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends \$ 23 of Ord. 48022, exhibitors or trade	96301	,	
Collection rates and charges (21.40)   96348		=	
(21.40)  96348  Noguchi Sculpture subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade			
subaccount under Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade			
Arts Account of General Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	96348	Noguchi Sculpture	
Donations and Gift Trust Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade		subaccount under Civic	
Fund (5.76)  96352  Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade		Arts Account of General	
96352 Amends § 2 of Ord. 92729, Seattle-King County Youth Commission (3.66)  96360 Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366 Rapid Transit Study Fund (5.76)  96372 Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395 Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400 Amends § 23 of Ord. 48022, exhibitors or trade			
Seattle-King County Youth Commission (3.66)     96360			
Commission (3.66)     96360   Adds §§ 369-375 to Ord.     48022, mobile caterers     (Repealed by 109271)     96366   Rapid Transit Study Fund     (5.76)     96372   Amends §§ 3.17 and 28.1     of Ord. 86300, zoning     (Repealed by 117570)     96395   Adds §§ 14A.1-14A.7 to     and amends §§ 2.3, 3.27,     4.1, 14.1, 14.2, 15.2, 24.2     and 24.6 of Ord. 86300,     zoning (Repealed by 117570)     96400   Amends § 23 of Ord.     48022, exhibitors or trade	96352		
96360  Adds §§ 369-375 to Ord. 48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade			
48022, mobile caterers (Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade			
(Repealed by 109271)  96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	96360		
96366  Rapid Transit Study Fund (5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade			
(5.76)  96372  Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade			
96372 Amends §§ 3.17 and 28.1 of Ord. 86300, zoning (Repealed by 117570)  96395 Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400 Amends § 23 of Ord. 48022, exhibitors or trade	96366	1	
of Ord. 86300, zoning (Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	0.6050	` '	
(Repealed by 117570)  96395  Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	90372	0.0	
96395 Adds §§ 14A.1-14A.7 to and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400 Amends § 23 of Ord. 48022, exhibitors or trade			
and amends §§ 2.3, 3.27, 4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	06205		
4.1, 14.1, 14.2, 15.2, 24.2 and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade	90393		
and 24.6 of Ord. 86300, zoning (Repealed by 117570)  96400  Amends § 23 of Ord. 48022, exhibitors or trade			
zoning (Repealed by 117570)  96400 Amends § 23 of Ord. 48022, exhibitors or trade			
117570)   96400   Amends § 23 of Ord.   48022, exhibitors or trade			
96400 Amends § 23 of Ord. 48022, exhibitors or trade			
48022, exhibitors or trade	06400		
	90400		
ah owa (6 20)			
shows (6.20)	Last SM	SHOWS (0.20)	

eference only.

al amending
graphics,
ccuracy of

For current Sind, the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

96453 Parks and Recreation Department (3.26) 96464 Adds § 173.1 to and amends §§ 171, 172, 173 and 176 of Ord. 48022, massage parlors and bathhouses (6.36) 96529 Transfer to Light Fund; repeals Ord. 90558 (5.76) 96539 Amends §§ 3.05, 10.1, 11.1, 12.1 and 22.2 of Ord. 86300, zoning (Repealed by 117570) 96598 Amends § 30 of Ord. 90047, street use regulations (15.32) 96666 Amends §§ 1 and 3 of Ord. 91458 and §§ 1 and 3 of Ord. 92366, Seattle Center rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12) 96668 Amends §§ 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570) 1nspection of bridges, trestles, viaducts, and tunnels near streets (15.76) 96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052) 96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052) 96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76) 96797 South Seattle Redevelopment Project underground utility district (21.68) 96797 South Seattle Redevelopment Project underground utility district (21.68) 96839 Authorizes expenditures from Model City Fund (5.76) 96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76) 96867 Amends § 1 of Ord. 87983, Harbor Code (16.20) 96897 Seattle Design Commission (3.58) 96931 Fluoridation of water supply (10.22)	a anda	te file
Adds \$ 173.1 to and amends \$ 171, 172, 173 and 176 of Ord. 48022, massage parlors and bathhouses (6.36)	96453	Parks and Recreation
amends § § 171, 172, 173 and 176 of Ord. 48022, massage parlors and bathhouses (6.36)  96529  Transfer to Light Fund; repeals Ord. 90558 (5.76)  Amends § 3.05, 10.1, 11.1, 12.1 and 22.2 of Ord. 86300, zoning (Repealed by 117570)  96598  Amends § 3 0 of Ord. 90047, street use regulations (15.32)  96666  Amends § 1 and 3 of Ord. 92366, Seattle Center rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12)  96668  Amends § 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  96715  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  P6867  Amends § 1 of Ord. 87983, Harbor Code (16.20)  Seattle Design Commission (3.58)  96931  Fluoridation of water		Department (3.26)
and 176 of Ord. 48022, massage parlors and bathhouses (6.36)  96529  Transfer to Light Fund; repeals Ord. 90558 (5.76)  96539  Amends §\$ 3.05, 10.1, 11.1, 12.1 and 22.2 of Ord. 86300, zoning (Repealed by 117570)  96598  Amends § 30 of Ord. 90047, street use regulations (15.32)  96666  Amends §\$ 1 and 3 of Ord. 91458 and §\$ 1 and 3 of Ord. 91458 and §\$ 1 and 3 of Ord. 92366, Seattle Center Director (17.12)  96668  Amends §\$ 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  96715  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  Seattle Design Commission (3.58)  96931  Fluoridation of water	96464	Adds § 173.1 to and
and 176 of Ord. 48022, massage parlors and bathhouses (6.36)  96529  Transfer to Light Fund; repeals Ord. 90558 (5.76)  96539  Amends §\$ 3.05, 10.1, 11.1, 12.1 and 22.2 of Ord. 86300, zoning (Repealed by 117570)  96598  Amends § 30 of Ord. 90047, street use regulations (15.32)  96666  Amends §\$ 1 and 3 of Ord. 91458 and §\$ 1 and 3 of Ord. 91458 and §\$ 1 and 3 of Ord. 92366, Seattle Center Director (17.12)  96668  Amends §\$ 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  96715  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  Seattle Design Commission (3.58)  96931  Fluoridation of water	A FOR III	
massage parlors and bathhouses (6.36)	U July	
bathhouses (6.36)		l ·
Post   Post		© 1
repeals Ord. 90558 (5.76)	96529	
Amends § § 3.05, 10.1, 11.1, 12.1 and 22.2 of Ord. 86300, zoning (Repealed by 117570)	30323	
11.1, 12.1 and 22.2 of Ord. 86300, zoning (Repealed by 117570)  96598  Amends § 30 of Ord. 90047, street use regulations (15.32)  Amends § 1 and 3 of Ord. 91458 and § 1 and 3 of Ord. 91458 and § 1 and 3 of Ord. 92366, Seattle Center rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12)  96668  Amends § 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  Seattle Design Commission (3.58)  96931  Fluoridation of water	06530	
86300, zoning (Repealed by 117570)	10338 CI CO	
by 117570		
Amends § 30 of Ord. 90047, street use regulations (15.32)	10	
96666 Amends §§ 1 and 3 of Ord. 91458 and §§ 1 and 3 of Ord. 91458 and §§ 1 and 3 of Ord. 92366, Seattle Center rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12)  96668 Amends §§ 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  96715 Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	0.500	
regulations (15.32)	96398	
Amends §§ 1 and 3 of Ord. 91458 and §§ 1 and 3 of Ord. 92366, Seattle Center rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12)		
91458 and §§ 1 and 3 of Ord. 92366, Seattle Center rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12)  96668  Amends §§ 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  96715  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water	140 0	
Ord. 92366, Seattle Center rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12)  96668 Amends § 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  96715 Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96899 Authorizes expenditures from Model City Fund (5.76)  96867 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	96666	
rental agreements; amends § 2 of Ord. 92792, Seattle Center Director (17.12)  96668 Amends § 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  96715 Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
\$ 2 of Ord. 92792, Seattle Center Director (17.12)  96668  Amends § 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  Fluoridation of water		,
Center Director (17.12)   96668	<b>5116.</b>	
Amends §§ 6.3, 8.3, 8.6, 13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)	222	
13A.5, 13B.5 and 26.2 of Ord. 86300, zoning (Repealed by 117570)		Center Director (17.12)
Ord. 86300, zoning (Repealed by 117570)  96715  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water	96668	Amends §§ 6.3, 8.3, 8.6,
(Repealed by 117570)  96715  Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787  Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water		13A.5, 13B.5 and 26.2 of
Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		Ord. 86300, zoning
Inspection of bridges, trestles, viaducts, and tunnels near streets (15.76)  96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		(Repealed by 117570)
trestles, viaducts, and tunnels near streets (15.76)  96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	96715	
tunnels near streets (15.76)  96787 Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
Amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
80028, preparation and sale of poultry, rabbit and fish (Repealed by 110052)  96788 Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	96787	` /
of poultry, rabbit and fish (Repealed by 110052)  96788  Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790  Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water	30707	-
(Repealed by 110052)           96788         Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)           96790         Amends § 1 of Ord. 45599, Contingent Fund A (5.76)           96796         First Hill underground utility district (21.68)           96797         South Seattle Redevelopment Project underground utility district (21.68)           96839         Authorizes expenditures from Model City Fund (5.76)           96845         Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)           96867         Amends § 1 of Ord. 88522, sick leave (4.24)           96875         Amends § 7 of Ord. 87983, Harbor Code (16.20)           96897         Seattle Design Commission (3.58)           96931         Fluoridation of water		
Amends § 3 of Ord. 94465, distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
distribution and sale of meat (Repealed by 110052)  96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	06788	
meat (Repealed by 110052)   96790	30788	
96790 Amends § 1 of Ord. 45599, Contingent Fund A (5.76)  96796 First Hill underground utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
Contingent Fund A (5.76)  96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water	06700	
96796  First Hill underground utility district (21.68)  96797  South Seattle Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water	96790	
utility district (21.68)  96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	0.550.6	
96797 South Seattle Redevelopment Project underground utility district (21.68)  96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	96/96	U U
Redevelopment Project underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water	0.4707	
underground utility district (21.68)  96839  Authorizes expenditures from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water	96797	
(21.68)           96839         Authorizes expenditures from Model City Fund (5.76)           96845         Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)           96867         Amends § 1 of Ord. 88522, sick leave (4.24)           96875         Amends § 7 of Ord. 87983, Harbor Code (16.20)           96897         Seattle Design Commission (3.58)           96931         Fluoridation of water		
96839 Authorizes expenditures from Model City Fund (5.76)  96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
from Model City Fund (5.76)  96845  Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water		
(5.76)         96845       Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)         96867       Amends § 1 of Ord. 88522, sick leave (4.24)         96875       Amends § 7 of Ord. 87983, Harbor Code (16.20)         96897       Seattle Design Commission (3.58)         96931       Fluoridation of water	96839	
96845 Authorized investments in Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
Civic Arts Account of General Donations and Gift Trust Fund (5.76)  96867  Amends § 1 of Ord. 88522, sick leave (4.24)  96875  Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897  Seattle Design Commission (3.58)  96931  Fluoridation of water		(5.76)
General Donations and Gift Trust Fund (5.76)	96845	Authorized investments in
General Donations and Gift Trust Fund (5.76)		Civic Arts Account of
96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		
96867 Amends § 1 of Ord. 88522, sick leave (4.24)  96875 Amends § 7 of Ord. 87983, Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water		Trust Fund (5.76)
sick leave (4.24)   96875	96867	
96875 Amends § 7 of Ord. 87983, Harbor Code (16.20) 96897 Seattle Design Commission (3.58) 96931 Fluoridation of water		
Harbor Code (16.20)  96897 Seattle Design Commission (3.58)  96931 Fluoridation of water	96875	
96897 Seattle Design Commission (3.58) 96931 Fluoridation of water	70073	
(3.58) 96931 Fluoridation of water	06907	
96931 Fluoridation of water	96897	_
	0.4024	` '
supply (10.22)	96931	
		supply (10.22)

reference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	ipai -	
		te file
	96932	Classifies roads and streets
	23.5	into arterial or access
	- 101	streets (Repealed by
	a sor m	117569)
•	96956	Reimbursement to
	, 6, 2, 6	employees of cost of
		obtaining a special driver's
		license (4.68)
	96977	Amends §§ 1, 4 and 6 of
	of orea	Ord. 96897, Seattle Design
	cez cron	Commission (3.58)
	97016	Side sewer regulations;
	- ammle	repeals Ords. 82321 and
	COHIL	83316 (21.16)
	97025	Adds §§ 3-A and 3-B to
	140 00	and amends §§ 1, 2, 2-A, 3,
	ma to	4 and 7 of Ord. 86300,
•		zoning (Repealed by
		117570)
	97053	Arterial Development Fund
-	05050	(5.76)
	97059	Park Acquisition and
	07007	Development Fund (5.76)
	97085	Amends §§ 13B.4, 13B.5
		and 23.4 of Ord. 86300,
		zoning (Repealed by 117570)
	97110	Amends § 1 of Ord. 86799,
	9/110	vacation leave (4.32)
	97185	Compensation for
	77103	employees temporarily
		assigned to work at
		locations requiring
		overnight absence; repeals
		Ord. 96256 (4.20)
	97192	Amends § 7 of Ord. 93789,
		cabulance licenses
		(Repealed by 97192)
	97241	Amends § 142 of Ord.
		48022, steam engineers and
		boiler firemen (Repealed
	07297	by 111301)
	97286	Amends §§ 23, 65, 183,
		300-C, 327, 328-A and
		330.2 of Ord. 48022, License Code (6.20, 6.42)
	97288	Amends § 5 of Ord. 62662,
	77200	utilities business tax (5.48)
	97303	Amends §§ 14(f) and 14(h)
	77303	of Ord. 78444, retirement
		system (4.36)
	97330	Compensation and working
		conditions for City officers
		and employees (4.20)
	97331	Amends §§ 1 and 7 of Ord.
		95035, vehicle wrecker and
		towing service (Repealed
		by 109080)
	97354	Amends § 1 of Ord. 86799,
		vacation leave (4.32)

reference only.

s amending graphics, ccuracy of

June 2009 co 97358

Text provide

See ordinant sections for and tables a this source

	Sibar	0210
	a mada	te file
	97358	Amends § 1 of Ord. 1466,
		payment by grantee of
	1 Cost hi	franchise to defray
•	d for any	expenses (21.64)
	97366	Committee of 33
		subaccount and Doris
		Chase Sculpture
		subaccount under Civic
		Arts Account of General
	nos crea	Donations and Gift Trust
Ц	07204	Fund (5.76)
	97384	Amends §§ 26.1 and 26.2
	COMBIE	of Ord. 86300, zoning
	97394	(Repealed by 117570)
	97394	Amends §§ 5.2, 17A.6 and
	ad to co	22.4 of Ord. 86300, zoning
4	07410	(Repealed by 117570)
	97410	Amends § 46 of Ord.
	0210	97330, compensation of
	<b>4116.</b>	offices and employees
	97456	(4.20)
	97456	Amends § 4 of Ord. 90138,
		signs adjacent to freeways
	07.472	(Repealed by 117570)
	97473	Department of Community
		Development; repeals
		Ords. 90067 and 95670
	07400	(Repealed by 116007)
	97499	Adds §§ 13.0, 13A.0 and
		13B.0; amends §§ 26.3, 27.1 and 28.1 of Ord.
		86300, zoning (Repealed by 117570)
	97552	Amends § 51 of Ord.
	71332	97330, compensation of
		officers and employees
		(4.20)
	97646	Amends § 14 of Ord.
	77010	48022, general License
		Code provisions (6.02)
	97652	Amends §§ 6.2 and 14A.3
	, , <del>, , , , , , , , , , , , , , , , , </del>	of Ord. 86300, zoning
		(Repealed by 117570)
	97794	Amends § 28 of Ord.
		94465, distribution and sale
		of meat (Repealed by
		110052)
	97841	Amends § 104 of Ord.
		48022, for-hire drivers
		(Repealed by 109648)
	97850	Amends § 4 of Ord. 81599,
		impoundment (Repealed by
		110230)
	97902	Summer Youth Activities
		account of General
		Donations and Gift Trust
		Fund (5.76)
		,

d amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

a anda	te file
97956	Amends §§ 5-A, 11, 11-A
	and 19, and repeals §§ 64,
1 Cost h1	65, 65-A, 66, 67 and 68 of
dian	Ord. 48022, License Code;
	amends § 3 of Ord. 75088,
	expiration of licenses
07071	(6.02)
97971	Department of Human
-000	Rights; repeals Ord. 92191
97998	(3.14) Amends §§ 3.13 and 22.1
9/998	of Ord. 86300, zoning
	(Repealed by 117570)
98029	Amends § 5 of Ord. 91778,
	privately owned fire alarm
	systems (10.08)
98066	Amends § 6.2 of Ord.
1100	86300, zoning (Repealed
	by 110381)
98067	Amends § 15.2 of Ord.
Time	86300, zoning (Repealed
	by 117570)
98086	Arrest of probationers
	(12A.56)
98122	Amends §§ 12(a), 13 and
	14(f) of Ord. 78444,
98123	retirement system (4.36)
98123	Amends § 1 of Ord. 96839,
98130	Model City Fund (5.76)
98130	Amends §§ 4, 4.1 and 5 of Ord. 90379, garbage
	collection rates and charges
	(21.40)
98132	Amends §§ 1, 2, 3 and 4, of
	Ord. 92729, Seattle-King
	County Youth Commission
	(3.66)
98149	Amends §§ 1 and 5 of Ord.
	78076, weeds and
	vegetation nuisances
	(Repealed by 114355)
98163	Amends § 9(c) of Ord.
	78444, retirement system
00171	(4.36) Amends §§ 12.5, 13.5,
98171	13A.5 and 13B.5 of Ord.
	86300, zoning (Repealed
	by 117570)
98196	Amends §§ 903 and 262 of
	Ord. 48022, detective
	agencies and merchant
	patrolmen (6.14, 6.38)
	Amends §§ 6, 25(h) and 41
98197	
98197	of Ord. 90047, street use
98197	of Ord. 90047, street use regulations (15.02, 15.44)
98197 98216	of Ord. 90047, street use regulations (15.02, 15.44) Amends §§ 15.2, 15.3, 15.6
	of Ord. 90047, street use regulations (15.02, 15.44) Amends §§ 15.2, 15.3, 15.6 and 22.2 of Ord. 86300,
	of Ord. 90047, street use regulations (15.02, 15.44) Amends §§ 15.2, 15.3, 15.6 and 22.2 of Ord. 86300, zoning (Repealed by
98216	of Ord. 90047, street use regulations (15.02, 15.44) Amends §§ 15.2, 15.3, 15.6 and 22.2 of Ord. 86300, zoning (Repealed by 117570)
	of Ord. 90047, street use regulations (15.02, 15.44) Amends §§ 15.2, 15.3, 15.6 and 22.2 of Ord. 86300, zoning (Repealed by

eference only.

l amending

graphics,

ccuracy of

For current Size, the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	ILE Par
98247	Seattle Parking
	Commission (Repealed by
40-4 h1	110171, 110226)
98292	Amends § 46 of Ord.
	97330, compensation of
	officers and employees
	(4.20)
98295	Group accidental death and
	dismemberment insurance
oc crea	plan (Repealed by 114893)
98304	Seattle-King County
	Commission on
	Alcoholism (3.52)
98316	Adds §§ 51A and 52A to
	and amends § 49 of Ord.
140 00	97330, compensation of
ma to	officers and employees
2200	(4.20)
98327	Amends §§ 1 and 3 of Ord.
	79432, Public Health
Tilo	Department (Repealed by
	109522)
98365	Amends § 1 of Ord. 78076
	weeds and vegetation
	nuisances (Repealed by
	114355)
98403	Amends § 2 of Ord. 72495
	admission tax (5.40)
98417	Project Temporary Loan
	Prepayment Fund, Project
	No. Wash. R-13 (5.76)
98418	Duplicate Medicare
	payments (3.30)
98423	Amends §§ 4, 5, 10, 11, 12
	and 18; repeals § 13 of
	Ord. 62662, utilities
	business tax (5.48)
98426	Amends §§ 3.05, 3.21, 6.2,
	13A.5, 13B.5, 15.2, 15.3,
	17.3, 19.2, 20.2 and 24.4 or
	Ord. 86300, zoning
	(Repealed by 117570)
98511	Data processing services
	(Repealed by 109576)
98543	Amends § 14(f) of Ord.
, , , , ,	78444, retirement system
	(4.36)
98577	Amends § 101 of Ord.
70311	48022, dog kennels
	(Repealed by 111649)
98597	Amends § 4 of Ord. 78444
70371	retirement system (4.36)
0.00	
98608	Adds §§ 11A.1-11A.6 to
	and amends §§ 2.2, 3.05,
	3.13, 3.16, 3.20, 3.27, 4.1,
	10.1, 11.1, 12.1, 12.3, 22.3
	and 23.2 of Ord. 86300,
	· /D 1 11
	zoning (Repealed by 117570)

eference only.

| amending | graphics, | accuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

	late file
98639	Amends § 4 of Ord. 90952,
	ambulance licensing
1	(Repealed by 110151)
98698	Amends §§ 18.6 and 24.4
	of Ord. 86300, zoning
	(Repealed by 117570)
98708	Sales and use tax
	(Repealed by 112277)
98755	Swimming and wading
	pools; repeals Ord. 27183
ces cr	(Repealed by 110180)
98770	Amends § 1 of Ord. 96839,
70110	Model City Fund (5.76)
98817	Adds § 2.2 to and amends
70017	§§ 3, 11, 13, 17 and 18 of
	Ord. 72630, business and
and to	occupation tax; repeals
	Ord. 98533 (5.44)
98820	Weights and Measures
70020	C
1116.	Code; repeals Ords. 41014
98838	and 71087 (7.04) Amends §§ 4.1 and 22.2 of
90030	
	Ord. 86300, zoning
98839	(Repealed by 117570) Amends § 18.3 of Ord.
98839	
	86300, zoning (Repealed
00040	by 117570)
98848	Amends § 1.1 of Ord.
	90789, retirement sick
00040	leave (4.24)
98849	Amends § 49 of Ord.
	97330, compensation of
	officers and employees
98852	(4.20)
98832	Pioneer Square Historic
	District (Repealed by 110058)
98861	Amends § 9 of Ord. 78444,
90001	
00076	retirement system (4.36)
98876	Northwest Leschi
	Rehabilitation Escrow
	Fund; Program No. Wash.
00000	A-2 (5.76)
98908	Reimbursement for
	medical expenses to
	citizens aiding police
98926	officers (12A.62)
98920	Amends §§ 1 and 3 of Ord.
98920	86799, vacation leave
	86799, vacation leave (4.32)
98929	86799, vacation leave (4.32) Amends § 2 of Ord. 96020,
	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures
98929	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures (Repealed by 109740)
	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures (Repealed by 109740)  Certain payments from
98929	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures (Repealed by 109740)  Certain payments from Firemen's Pension Fund
98929 98956	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures (Repealed by 109740)  Certain payments from Firemen's Pension Fund (5.76)
98929	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures (Repealed by 109740)  Certain payments from Firemen's Pension Fund (5.76)  Biennial review of boards
98929 98956 99003	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures (Repealed by 109740)  Certain payments from Firemen's Pension Fund (5.76)  Biennial review of boards and commissions (3.98)
98929 98956	86799, vacation leave (4.32)  Amends § 2 of Ord. 96020, street vacation procedures (Repealed by 109740)  Certain payments from Firemen's Pension Fund (5.76)  Biennial review of boards

eference only. amending graphics, accuracy of

For current SMO. the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

	a conda	te file
	99040	Amends § 1 of Ord. 92590,
		segregation of
	- 0 - m hi	condemnation assessments
	A FOR III	(20.76)
	99051	Amends §§ 5.3 and 26.2 of
		Ord. 86300, zoning
		(Repealed by 117570)
	99092	Seattle/King County Office
	,, o, =	of Citizen's Complaints
	- area	(Repealed by 115345)
1	99094	Amends § 14 of Ord.
- 1	3,03.	48022, general License
	10	Code provisions (6.02)
	99108	Adds § 19-A to Ord.
,	77100	87983, Harbor Code
	4 4 - 06	(16.20)
	99109	Amends §§ 33 and 34 of
0	1313	Ord. 65877, water supply
		system (21.04)
	99122	Amends § 5 of Ord. 90379,
		garbage collection rates
		and charges (21.40)
	99127	Surcharges for water
	99127	customers in Richmond
	99155	Beach area (21.12)
	99155	Cherry Hill Fund, Program
	99195	No. Wash. N-S (5.76)
	99195	Adds § 12-A to Ord.
		90952, ambulance
		licensing (Repealed by
	99322	110151) Amends § 11 of Ord.
	99322	-
		90379, garbage collection
		rates and charges (5.76,
	99363	21.40)
	99303	Prohibits climbing on
		Wedgewood Rock (12A.54)
	00269	
	99368	Amends §§ 6.2, 10.2, 12.1,
		14A.1, 15.2, 16.4, 16.5,
		17.4, 17.5, 23.2, 23.3 and 23.4 of Ord. 86300, zoning
	00422	(Repealed by 117570) Amends § 1 of Ord. 88522,
	99422	sick leave (4.24)
	00442	
	99443	Joint City-County subcommittee on
		Shoreland and Tideland
		Planning (Repealed by 109155)
	99444	,
	77444	Amends §§ 2-A and 2-B of
		Ord. 82583, sewer
	00454	connection fees (21.24)
	99454	Amends §§ 3, 4, 4.1, 4.2
		and 4.3 of Ord. 84390,
		sewerage rates and charges
		(21.28)
	99481	Fees for certain reports and
		tests by Fire Department
		(3.16)

eference only.

cel amending
graphics,
accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

Cibos	adate file
99492	Administrative charges for
	public works contract
	(Repealed by 117390)
99503	Amends §§ 6.1, 12.1, 17.2
A COLUMNIA	and 17.3 of Ord. 86300,
	zoning (Repealed by
	117570)
99524	Amends § 6 of Ord. 63662,
77324	utilities business tax (5.48)
99554	Department of General
99334	Services (Repealed by
	109129)
00566	
99566	Amends §§ 2(m), 7, 13(c),
COSS	14(c), 14(f), 14(h), 16(a)
	and 16(b); adds § 19(c);
- 4	repeals § 14-i of Ord.
	78444, retirement system;
00505	repeals Ord. 91594 (4.36)
99585	Amends § 1 of Ord. 45599,
	Contingent Fund A (5.76)
99641	Northwest Leschi Project
	underground utility district
	(21.68)
99644	Community Development
	Operating Fund (5.76)
99650	Amends § 15 of Ord.
	90047, street use regulation
	(15.02)
99674	Adds § 49(a) to and
	amends § 6 of Ord. 90047,
	street use regulations
	(15.02, 15.16)
99685	Membership in
	Washington Public Power
	Supply System (21.56)
99704	Amends § 13.5 of Ord.
	86300, zoning (Repealed
	by 117570)
99705	Amends § 7-A-1 of Ord.
	65877, water supply
	system (21.04)
99745	Office of Economic
	Development (Repealed by
	116007)
99749	Amends § 2.010 of Ord.
	89715, mobile homes and
	mobile home parks
	(22.904)
99751	Amends § 3 of Ord. 86799,
	vacation leave (4.32)
99752	Amends § 49 of Ord.
	97330, compensation of
	officers and employees
	(4.20)
99753	Funeral leave; repeals §
	10-A of Ord. 88522 (4.28)
99754	Amends § 1.1 of Ord.
1	90789, retirement sick
	leave (4.24)
	1200.0 (1121)

eference only.

I amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te file
	99767	Community Service
		Officer program in Police
	101	Department (3.28)
	99788	Amends § 4 of Ord. 84390,
2	0.70	sewerage rates and charges
		(21.28)
	99829	Amends § 50 of Ord.
	77027	97330, compensation of
		officers and employees
	- 0403	(4.20)
	99837	Amends § 6 of Ord. 83834,
3	99837	
	10	health care program (Repealed by 114893)
	99844	
	99844	Amends § 6 of Ord. 66591,
		Special Police (Repealed
	00046	by 109759)
4	99846	Amends §§ 2, 6 and 7 of
		Ord. 98852, Pioneer
	0210	Square Historic District
	<b>F116.</b>	(Repealed by 110058)
	99856	Adds § 5; amends §§ 1, 2
		and 4; repeals and replaces
		§ 3 of Ord. 92729,
		Seattle-King County Youth
		Commission (3.66)
	99859	Amends § 46 of Ord.
		97330, compensation of
		officers and employees
		(4.20)
	99869	Amends § 9 of Ord. 90535,
		vehicle wrecker and towing
		service (Repealed by
		109080)
	99870	Fees for certain Police
		Department reports; repeals
		Ords. 89685 and 93158
		(3.28)
	99872	Amends §§ 3.05, 14A.2,
		14R.4, 14A.7, 15.2 and
		15.3 of Ord. 86300, zoning
		(Repealed by 117570)
	99890	Amends § 34 of Ord.
		65877, water supply
		system (21.04)
	99901	Division on Aging in
	///UI	Office of Human
		Resources (3.20)
	99911	Naming of parks and
	77711	<u> </u>
	00044	recreation areas (18.08) Amends § 50 of Ord.
	99944	-
		97330, compensation of
		officers and employees
	00051	(4.20)
	99951	Water supply and sewage
		disposal (Repealed by
		107099)
		A 1 00 1 12
	99963	Amends §§ 1 and 3;
	99963	repeals § 2 of Ord. 99753,
	99963	

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

Cipal		
	a mada	te file
	99982	Seattle Arts Commission;
		repeals Ords. 86692,
	a com hi	89304, 90354 and 95649
	A TOL 185	(Repealed by 110894)
	99992	Amends § 6 of Ord. 92987,
		food-service
		establishments (Repealed
		by 109910)
	99998	Permits for use of tennis
	age crea	courts for professional
У	100014	instruction (18.20)
	100014	Amends § 104 of Ord. 48022, for-hire drivers
	comple	(Repealed by 109648)
	100020	Amends § 3 of Ord. 86799,
	100020	vacation leave (4.32)
	100100	Amends §§ 3.09, 6.2, 6.4,
		7.2, 8.2, 10.2 and 12.1 of
		Ord. 86300, zoning
	2110	(Repealed by 117570)
	100112	Family emergency leave
		for certain employees
		(4.20)
	100171	Adds §§ 65-74 to and
		amends §§ 2, 3, 5, 6, 7, 8,
		10 and 28 of Ord. 87983,
		Harbor Code (5.76, 16.04,
		16.08, 16.12, 16.20, 16.32,
	1001=1	16.36, 16.40, 16.60)
	100174	Cafeteria in Seattle
		Municipal Building
	100181	(Repealed by 116007) Amends § 1 of Ord. 99492,
	100181	administrative charges for
		public works contract
		(Repealed by 117390)
	100216	Amends § 14A.2 of Ord.
	100210	86300, zoning (Repealed
		by 117570)
	100217	Amends § 15.1 of Ord.
		86300, zoning (Repealed
		by 117570)
	100219	Payroll deductions for auto
		insurance payments (4.68)
	100232	Amends §§ 6.2, 7.2, 8.3,
		and 12.1 of Ord. 86300,
		zoning (Repealed by
	10000	117570)
	100276	Amends §§ 14A.3 and 15.2
		of Ord. 86300, zoning (Repealed by 117570)
	100327	Amends § 9 of Ord. 62662,
	100341	utilities business tax (5.48)
	100336	CPR training program
	100000	(3.16, 5.76)
	100363	Adds § 1-A to Ord. 89021,
		litter control (Repealed by
		113502)
	100364	Amends § 1 of Ord. 99870,
		fees for certain Police
	4 CM	Department reports (3.28)
	MONT DIV	7

reference only. d amending graphics, accuracy of

For current SA the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipai		
		te the
1003	373	Amends § 159 of Ord.
		48022, junk shops and junk
	cor hi	wagons (Repealed by
1003	202	109818)
1003	000	Amends §§ 6.2, 7.2, 8.2 and 12.1 of Ord. 86300,
		zoning (Repealed by
		117570)
1004	104	Amends § 1 of Ord. 86799,
	se erea	vacation leave (4.32)
1004	413	Amends § 1 of Ord. 99127,
		surcharge for water
	ammle	customers in Richmond
100	Ollisha	Beach area (21.12)
1004	135	Code of ethics (Repealed
1004	145 40 CI	by 108882, 109950)
1002	145	Amends § 16 of Ord. 94465, distribution and sale
		of meat (Repealed by
	10	110052)
1004	158	Use of City owned or
100	130	leased motor vehicles;
		repeals Ords. 67585 and
		86428 (3.18, 4.68)
1004	175	Pike Place Market
		Historical District (25.24)
1004	178	Amends §§ 27-6, 41, 51,
		104, 188, 216, 331.5 and
		363 of Ord. 48022, License
		Code (Repealed by
100	-04	109648, 109818, 111441)
1005	501	Advance payments in
		partial settlement of claims against the City;
		establishes Advance
		Payment Claims Fund
		(Repealed by 108657)
1005	542	Amends §§ 3.13, 3.17,
		3.19, 3.23, 17A.2 and
		17A.6 of Ord. 86300,
		zoning (Repealed by
1001		117570)
1005	545	Amends § 267 of Ord.
		48022, going out of
		business sales (Repealed by 111441)
1005	562	Nutcracker Northwest
100.	702	Account in General
		Donations and Gift Trust
		Fund (5.76)
1005	591	Fees for copies of City
		books and records (3.104)
1006	503	Amends § 13 of Ord.
		90047, street use
		regulations (15.04)
1006	512	Amends § 15.2 of Ord.
		86300, zoning (Repealed
		by 117570)
1006	513	Amends § 23.3 of Ord.
		86300, zoning (Repealed
	-4 SM	by 117570)

eference only.

| amending |
| aphics, |
| accuracy of

For current Sind, the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

ilbar .	0210
a and	ate file
100628	FACE Programs Escrow Fund (5.76)
100648	Amends § 7 of Ord.
	100435, Code of Ethics
CI LO	(Repealed by 108882,
	109950)
100649	Amends §§ 4 and 8 of Ord.
	100435, Code of Ethics
oc crea	(Repealed by 108882, 109950)
100664	Amends § 1 of Ord. 94289,
	signatures on petitions (2.16)
100701	Adds §§ 25-1 and 25-2 to
4400	Ord. 65877, water supply system (21.04)
100708	Disclosure of unit prices
'THOS	(7.12)
100729	Amends § 14 of Ord.
<b>\$112.</b>	48022, general License
222	Code provisions (6.02)
100749	Amends § 1 of Ord. 96298,
	rental of Blue Spruce and
	225 Mercer Buildings
100750	(Repealed by 112790)
100759	Amends § 2 of Ord. 16374, Police Department
	(Repealed by 109757)
100792	Amends §§ 14(b), 14(e),
100772	14(f) and 16 of Ord. 78444,
	retirement system (4.36)
100862	Dental care program
	(Repealed by 114893,
	115159)
100877	Amends § 9 of Ord. 78444,
	retirement system (4.36)
100878	Amends § 1 of Ord. 86799,
100070	vacation leave (4.32)
100879	Amends § 1 of Ord. 88522;
	amends §§ 1 and 1.1 of
	Ord. 90789, sick leave (4.24)
100890	Amends §§ 3.04, 3.19,
	14.2, 14A.2, 15.2 and 18.2
	of Ord. 86300, zoning
	(Repealed by 117570)
100895	Transfers between budgets
	of City departments (5.08)
100903	Amends § 10 of Ord.
	98820, Weights and
100016	Measures Code (7.04)
100916	Amends §§ 1 and 3 of Ord.
100017	99753, funeral leave (4.28)
100917	Amends § 5 of Ord. 95466,
	group term life insurance
	program (Repealed by 114893)
100918	Amends § 4 of Ord. 78444,
100710	retirement system (4.36)
	Terrement system (4.50)

reference only.

Td. amending

graphics,

curacy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	te file
100946	Amends §§ 3 and 9 of Ord.
100 0-2	72630, business and
100	occupation tax (5.44)
100957	Amends § 4 of Ord. 90952,
	ambulance licensing
	(Repealed by 110151)
100960	Amends §§ 1 and 3 of Ord.
	100708, unit prices (7.12)
100965	Animal Control
	Commission; repeals §§ 1,
ces crea	2, 3 and 14 of Ord. 81599,
	(Repealed by 110230)
100988	Amends § 25 of Ord.
	65877, water supply
Cosses	system (21.04)
100990	Amends § 5 of Ord.
100990	100862, dental care
Hiller co	
	program (Repealed by 114893, 115159)
101024	
101024	Amends §§ 5 and 7 of Ord.
	78944, retirement system (4.36)
101026	( /
101026	Amends § 49 of Ord.
	97330, compensation of
	officers and employees
101022	(4.20)
101033	Amends § 3 of Ord. 72630,
	business and occupation
	tax; repeals § 7-A of Ord.
101076	62662 (5.44)
101076	Amends § 198 of Ord.
	48022, peddlers or
	solicitors (Repealed by
1010=0	109271)
101079	Amends § 4 of Ord. 81599,
	impoundment (Repealed by
101001	110230)
101084	Amends §§ 1 and 5 of Ord.
	93603, official bonds;
	repeals Ord. 100431 (3.90)
101117	Amends § 23.3 of Ord.
	86300, zoning (Repealed
	by 117570)
101151	Amends § 14.6 of Ord.
	86300, zoning (Repealed
	by 117570)
101168	Amends § 23 of Ord.
	48022, exhibitors or trade
	shows (6.20)
101169	Amends § 6.3 of Ord.
	86300, zoning (Repealed
	by 110381)
101210	Amends § 3 of Ord. 98511,
	data processing services
	(Repealed by 109576)
101212	Animal Population Control
	Clinic account in General
	Donations and Gift Trust
	Fund (5.76)
1	

eference only.

amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	cipa. do sile		
	4 - 440012	TG 111	
	101239	Amends § 193 of Ord.	
		48022, peddlers or	
	1 Cor hi	solicitors (Repealed by	
9	4 101	109271)	
	101240	Amends § 22.4 of Ord.	
		86300, zoning (Repealed	
	101266	by 117570) Amends §§ 14A.3 and 15.2	
	101200	of Ord. 86300, zoning	
	- 0002	(Repealed by 117570)	
	101267	Amends §§ 20.2 and 20.7	
3	101207	of Ord. 86300, zoning	
		(Repealed by 117570)	
	101274	Amends § 15 of Ord.	
		78444; retirement system	
	140 00	(4.36)	
	101281	Office of Economic	
		Development	
		Entrepreneurial Assistance	
	2116	Program (3.10)	
	101285	Amends §§ 3.05, 3.07,	
		3.15, 6.2, 6.3, 10.2, 12.1	
		and 14.2 of Ord. 86300,	
		zoning (Repealed by	
	101295	117570) Travel expenses; repeals	
	101293	Ord. 96129 (4.72)	
	101296	Amends § 23.4 of Ord.	
	101290	86300, zoning (Repealed	
		by 117570)	
	101347	Seattle Arts Festival	
		Account in General	
		Donations and Gift Trust	
	101251	Fund (5.76)	
	101351	Amends §§ 13 and 41(H) of Ord. 90047, street use	
		regulations (15.04, 15.44)	
	101362	Amends § 15.2 of Ord.	
	101302	86300, zoning (Repealed	
		by 117570)	
	101371	Amends §§ 305.1-305.6,	
		burglar alarms (6.08)	
	101372	Amends § 2 of Ord.	
		100458, use of City owned	
	101200	or leased vehicles (4.68)	
	101388	Amends §§ 172 and 173 of	
		Ord. 48022, massage parlors and bathhouses	
		(6.36)	
	101402	Amends § 95-c of Ord.	
	-	48022, dog and cat licenses	
		(Repealed by 110230)	
	101413	Amends §§ 23.2 and 23.3	
		of Ord. 86300, zoning	
		(Repealed by 117570)	
	101414	Amends §§ 1, 2 and 4;	
		repeals and replaces § 3 of	
		Ord. 92729, Seattle-King	
		County Youth Commission (3.66)	
I		(3.00)	

reference only.

The contraction of the contraction

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

30	to file
101432	Prevention of
101432	discrimination in City
103	contracts; repeals Ord.
a sor ni	9838 (20.44)
101475	Office of Housing Policy
101.70	(Repealed by 116007)
101476	Privately owned burglary
	alarm systems; amends § 1
	of Ord. 87178 (10.08)
101477	Amends § 2 of Ord. 97971,
ces es	Department of Human
	Rights (3.14)
101483	Adds § 3-A to and amends
Comple	§§ 1 and 2 of Ord. 95751,
	reimbursement for use of
- 4 to C	private automobiles on
101404	City business (4.68)
101484	Amends § 4 of Ord.
C:10	100458, use of City owned or leased vehicles (4.68)
101488	Amends § 336 of Ord.
101400	48022, vehicle wreckers
	(Repealed by 109080)
101523	Adds §§ 305.1-305.6 to
101020	Ord. 48022, burglar alarms
	(6.08)
101531	Beekeeping (10.36)
101532	Amends § 2 of Ord. 82583,
	sewer connection fees
	(Repealed by 116421)
101533	Entrepreneurial Assistance
	Program Fund (5.76)
101548	Affirmative action program
10110	(Repealed by 109112)
101608	Amends §§ 13A.5, 13B.5,
	24.4 and 24.5 of Ord.
	86300, zoning (Repealed by 117570)
101614	Amends § 46 of Ord.
101014	97330, compensation of
	officers and employees;
	repeals Ords. 81508 and
	101008 (4.20)
101615	Amends § 6 of Ord. 78444,
	retirement system; repeals
	§ 2 of Ord. 69816 (4.36)
101620	Amends § 1 of Ord. 99127,
1	
	surcharge for water
	customers in Richmond
101/20	customers in Richmond Beach area (21.12)
101629	customers in Richmond Beach area (21.12) Transfer of parking meter
	customers in Richmond Beach area (21.12) Transfer of parking meter duties to police (3.28)
101629 101637	customers in Richmond Beach area (21.12) Transfer of parking meter duties to police (3.28) Comprehensive Planning
101637	customers in Richmond Beach area (21.12)  Transfer of parking meter duties to police (3.28)  Comprehensive Planning Assistance Fund (5.76)
	customers in Richmond Beach area (21.12)  Transfer of parking meter duties to police (3.28)  Comprehensive Planning Assistance Fund (5.76)  Adds § 4A to Ord. 84390,
101637	customers in Richmond Beach area (21.12)  Transfer of parking meter duties to police (3.28)  Comprehensive Planning Assistance Fund (5.76)  Adds § 4A to Ord. 84390, sewerage rates and charges
101637 101660	customers in Richmond Beach area (21.12)  Transfer of parking meter duties to police (3.28)  Comprehensive Planning Assistance Fund (5.76)  Adds § 4A to Ord. 84390, sewerage rates and charges (Repealed by 110201)
101637	customers in Richmond Beach area (21.12)  Transfer of parking meter duties to police (3.28)  Comprehensive Planning Assistance Fund (5.76)  Adds § 4A to Ord. 84390, sewerage rates and charges (Repealed by 110201)  Department of Human
101637 101660	customers in Richmond Beach area (21.12)  Transfer of parking meter duties to police (3.28)  Comprehensive Planning Assistance Fund (5.76)  Adds § 4A to Ord. 84390, sewerage rates and charges (Repealed by 110201)

eference only. amending graphics, ccuracy of

the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te file
Ī	101715	Industrial Insurance Fund;
4		qualification as self-insurer
	- 101	for workmen's
		compensation (4.44, 5.76)
ŀ	101720	Amends § 1 of Ord. 88522;
1	101720	
		amends §§ 1 and 1.1 of
		Ord. 90789, sick leave
L		(4.24)
	101734	Metro Transit Payroll Fund
L	oc crea	(5.76)
I	101738	Amends § 9 of Ord. 78444,
ľ		retirement system (4.36)
Г	101739	Amends §§ 1 and 3 of Ord.
	CULIBRIO	86799, vacation leave
		(4.32)
H	101742	Federal Shared Revenue
	101742	
ŀ	101744	Fund (Not codified)
ľ	101744	Adds § 16-A to Ord.
١	0.10	90047, street use
L	<u> 5116.                                      </u>	regulations (15.06)
	101763	Amends §§ 57 and 58 of
		Ord. 48022, cabarets
		(Repealed by 109653)
Г	101777	Amends § 330.6 of Ord.
		48022, panorams and
		peepshows (6.42)
H	101794	Amends §§ 1 and 2 of Ord.
	101774	101738, retirement system
		(4.36)
H	101010	` /
	101810	Department manuals
L		(Repealed by 118912)
	101820	Amends §§ 68(e) and 70 of
		Ord. 87983, Harbor Code
L		(16.60)
	101857	Amends §§ 233, 234 and
		238 of Ord. 48022, License
		Code; amends § 1 of Ord.
		93789, cabulance licenses;
		repeals § 21.67.160 of Ord.
ĺ		91910, traffic (Repealed by
l		109348)
H	101861	Amends § 1 of Ord. 88522,
ĺ	101001	sick leave (4.24)
ŀ	101966	Amends § 7 of Ord. 87983,
1	101866	
L	101070	Harbor Code (16.20)
1	101870	Water, sewerage and
1		electric rates for Newhalem
		community (21.12, 21.56)
		Real property transfer
	101941	
		certificates (7.16)
	101941	
_		certificates (7.16) Adds § 20.3 to and amends
		certificates (7.16) Adds § 20.3 to and amends § 6.2 of Ord. 86300, zoning
	101985	certificates (7.16) Adds § 20.3 to and amends § 6.2 of Ord. 86300, zoning (Repealed by 117570)
		certificates (7.16) Adds § 20.3 to and amends § 6.2 of Ord. 86300, zoning (Repealed by 117570) Amends § 4 of Ord. 92729,
	101985	certificates (7.16) Adds § 20.3 to and amends § 6.2 of Ord. 86300, zoning (Repealed by 117570) Amends § 4 of Ord. 92729, Seattle-King County Youth
	101985	certificates (7.16) Adds § 20.3 to and amends § 6.2 of Ord. 86300, zoning (Repealed by 117570) Amends § 4 of Ord. 92729, Seattle-King County Youth Commission (3.66)
	101985	certificates (7.16) Adds § 20.3 to and amends § 6.2 of Ord. 86300, zoning (Repealed by 117570) Amends § 4 of Ord. 92729, Seattle-King County Youth Commission (3.66) Amends § 3 of Ord. 96897,
	101985	certificates (7.16) Adds § 20.3 to and amends § 6.2 of Ord. 86300, zoning (Repealed by 117570) Amends § 4 of Ord. 92729, Seattle-King County Youth Commission (3.66)

eference only.

in eference only.

in a lamending

in eference only.

a lamending

in eference only.

a lamending

in eference only.

June 2009 co Text provide

See ordinant sections for and tables a this source

il puri	te file
102112	Emergency control of
102112	drainage problems, mud
	flows and earth slides
4 502 111	(10.06)
102150	Police officers' bill of
	rights (3.28)
102151	Division of Purchases;
	repeals Ords. 67976 and
	76384 (3.18)
102180	Arboretum Teahouse
GES CI CO	Reconstruction account in
	General Donations and Gift
100016	Trust Fund (5.76)
102182	Office of Women's Rights
	(Repealed by 109114)
102205	Adds §§ 267-1, 267-2 and
TO C	270-1 to and amends §§
The same	267, 268, 269 and 270 of
	Ord. 48022, going out of
2110	business sales (Repealed
Ille.	by 111441)
102210	Municipal Arts Fund; art in
	public works construction
	(5.76, 20.32)
102228	Administrative Code (3.02)
102252	Department of Licenses
102202	and Consumer Affairs;
	repeals Ords. 70328 and
	99581 (Repealed by
	117169)
102283	Bond requirements for
	certain demolition
	contracts (Repealed by
	118833)
102290	Amends §§ 3.05, 3.19,
	24.4, 26.1, 26.2, 26.3, 27.1,
	27.2, 27.3 and 28.1; adds
	§§ 24.41-24.48, 26.21,
	26.22, 26.23, 26.4, 27.11,
	27.12, 27.4, 27.41, 27.5,
	27.51, 27.6, 27.7, 27.8,
	27.51, 27.6, 27.7, 27.8, 27.81-27.84, 28.2, 28.3,
	27.81-27.84, 28.2, 28.3,
	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43,
	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to
102301	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning
102301	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570)
102301 102308	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983,
	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983, Harbor Code (16.20)
	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983, Harbor Code (16.20) Salary Administration
	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983, Harbor Code (16.20) Salary Administration Policy Committee (Repealed by 118928) Amends § 1 of Ord. 96020,
102308	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983, Harbor Code (16.20) Salary Administration Policy Committee (Repealed by 118928)
102308	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983, Harbor Code (16.20) Salary Administration Policy Committee (Repealed by 118928) Amends § 1 of Ord. 96020,
102308	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983, Harbor Code (16.20) Salary Administration Policy Committee (Repealed by 118928) Amends § 1 of Ord. 96020, street vacation procedures
102308	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570)  Amends § 7 of Ord. 87983, Harbor Code (16.20)  Salary Administration Policy Committee (Repealed by 118928)  Amends § 1 of Ord. 96020, street vacation procedures (Repealed by 109740)  Amends § 58 of Ord.
102308	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570) Amends § 7 of Ord. 87983, Harbor Code (16.20) Salary Administration Policy Committee (Repealed by 118928) Amends § 1 of Ord. 96020, street vacation procedures (Repealed by 109740)
102308	27.81-27.84, 28.2, 28.3, 28.4, 28.41, 28.42, 28.43, 28.5 and 28.51-28.55 to Ord. 86300, zoning (Repealed by 117570)  Amends § 7 of Ord. 87983, Harbor Code (16.20)  Salary Administration Policy Committee (Repealed by 118928)  Amends § 1 of Ord. 96020, street vacation procedures (Repealed by 109740)  Amends § 58 of Ord. 48022, cabarets (Repealed

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipas		
a anda	te me	
102366	Amends §§ 1 and 1.1 of	
	Ord. 90789, retirement sick	
102390	leave (4.24)	
102390	Amends § 10 of Ord. 48022, general License	
	Code provisions (6.02)	
102404	Amends §§ 190, 192, 193,	
	194, 197 and 197-1; adds	
	§§ 192-1 and 194-1-194-5;	
pos crea	repeals §§ 191-1 and 195	
669	of Ord. 48022, peddlers or	
10	solicitors (Repealed by 109271)	
102455	Adds §§ 24.81-24.87 and	
	24.92 to Ord. 86300,	
140 0	zoning (Repealed by	
and to co	112134)	
102458	Amends §§ 23, 25 and 26,	
0210	and repeals §§ 22 and 24 of	
111e.	Ord. 16046, cruelty to animals: amends § 1 of	
	Ord. 16208, § 1 of Ord.	
	31592, § 300-E of Ord.	
	48022, § 2 of Ord. 71881,	
	mechanical music	
	machines, and § 2 of Ord.	
	97955; repeals Ords.	
102459	100282 and 101370 (6.100) Gambling Tax (5.52)	
102483	Amends § 12 of Ord.	
102403	81599, animals on public	
	beaches (Repealed by	
	110230)	
102498	Amends §§ 5 and 6 of Ord.	
	83834, health care program	
102518	(Repealed by 114893) Amends § 1 of Ord. 86799,	
102316	vacation leave (4.32)	
102528	Amends § 1 of Ord. 96298,	
	rental of Blue Spruce and	
	225 Mercer Buildings	
	(Repealed by 112790)	
102560	Amends §§ 5, 6 and 7 of	
	Ord. 53993, local improvement procedure;	
	amends §§ 1 and 2 of Ord.	
	62364, Local Improvement	
	Guaranty Fund; deferral of	
	assessment collection	
	(5.76, 20.08, 20.12)	
102562	Prohibits discrimination in	
	employment practices; repeals Ord. 100642	
	(Repealed by 109116)	
102605	Seattle-King County Drug	
- 3- 3- 3- 3- 3- 3- 3- 3- 3- 3- 3- 3- 3-	Commission; repeals Ord.	
	99466 (3.60)	
102618	Amends § 1 of Ord. 43475,	
	false and misleading advertising (7.08)	

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipal of		
	a anda	te file
	102620	Amends §§ 2, 10, 11,
		15-21 and 24 of Ord.
	1 Cost hi	62662, utilities business tax
•	4 101	(5.48)
	102622	Amends §§ 6-10 and 15 of
		Ord. 72495, admission tax
ŀ	102623	(5.40) Amends §§ 2, 5, 6, 8,
	102023	12-15, 17, 19,-24, 29 and
	- 0403	30 of Ord. 72630, business
M	ces crea	and occupation tax (5.44)
	102625	Amends §§ 5, 6 and 8 of
		Ord. 73223, fumigators
	COMPA	(Repealed by 109932)
•	102626	Amends §§ 8, 16 and 16-A
	140 00	of Ord. 80028, preparation
Δ	ma to c	and sale of poultry, rabbit
۰	, 8 8 6 7	and fish (Repealed by
		110052)
	102627	Amends §§ 5, 6 and 8 of
		Ord. 80818, tattooing
ŀ	102628	(6.102)
	102628	Amends §§ 9, 10 and 17 of Ord. 84297, refrigeration
		systems (6.82)
ŀ	102629	Amends §§ 2.020 and
		2.030 of Ord. 89715,
		mobile homes and mobile
		home parks (22.904)
	102630	Amends §§ 2, 3, 5 and 7 of
		Ord. 90145, tobacco
		vending machines
	100/01	(Repealed by 115531)
	102631	Amends §§ 2, 3, 4, 5, 6, 8,
		9 and 11 of Ord. 90535, vehicle wrecker and towing
		services (Repealed by
		109080)
ŀ	102632	Amends §§ 3-6, 8, 10, 13
	- **=	and 14 of Ord. 90952,
		ambulance licensing
		(Repealed by 110151)
j	102634	Amends §§ 3-6, 8, 10, 11,
		12, 15 and 18-21 of Ord.
		93789, cabulance licenses
	100.00	(Repealed by 109348)
	102635	Amends §§ 8 and 35 of
		Ord. 98820, Weights and
Į		Measures Code (7.04)

eference only.

d amending
graphics,
accuracy of

June 2009 co 102636 UP
Text provide d for

See ordinances cresections for compand tables and to this source file.

102636 Amends §§ 3, 5-A, 9, 11, 12-a, 13-16, 19, 20, 21, 23-b, 23-c, 23-d, 23-f, 27-A, 27-J, 30, 31, 44, 58, 58-A, 63-B, 71, 72, 73, 78-C, 81, 90, 90-3, 91, 91-1, 93, 95, 95a, 98-101, 104-108, 110, 112, 113, 116, 144, 151, 156, 165, 166, 172, 173, 173.1, 174, 176.1, 183 1/2-238, 241, 243, 244, 262, 262-1, 263, 266, 267, 267-1, 268, 269, 270-1, 282, 300, 300-A, 300-C, 300-J, 303, 309-312, 314-a, 315-318, 320, 324, 328-A, 330, 330.1, 331.4, 331.6, 334, 335, 343, 350-353, 355, 356, 357, 359, 359.1-359.5, 370, and 372; adds §§ 4.1, 4.2, 13.1-13.4, 14.1, 19.1, 21.1 and 21.2; repeals §§ 11-A and 11-B of Ord. 48022, License Code; amends §§ 2-6, 8, 9 and 11 of Ord. 90535, vehicle wrecker and towing service; repeals Ord. 64843 (6.02, 6.14, 6.36, 6.38, 6.42)102638 Amends §§ 2, 4, 8, 9, 11 and 14 of Ord. 94465. distribution and sale of meat (Repealed by 110052) 102645 Amends § 6; adds § 22-A to Ord. 90047, street use

regulations (15.02)

Amends § 2 of Ord. 100458, use of City owned or leased motor vehicles

Amends § 2-A of Ord. 82583, sewer connection

Amends § 6.3 of Ord. 86300, zoning (Repealed

Amends §§ 12.2, 13.1 and 13B.1 of Ord. 86300, zoning (Repealed by

Amends § 18.3 of Ord. 86300, zoning (Repealed

Amends § 187 of Ord.

Amends § 1 of Ord. 99482, administrative charges for public works contracts (Repealed by 117390)

(4.68)

fees (21.24)

by 110381)

117570)

by 117570)

ference only.

amending

graphics,

ccuracy of

For cur (Repealed by 109818)

The Office of the City Clerk

The Office of the City Clerk

102666

102678

102684

102685

102686

102688

102694

June 2009 co Text provide

See ordinant sections for and tables a this source

	andate file		
	102712	Head Start Program	
	40	(Repealed by 110999)	
	102719	Amends §§ 1, 2, 8 and 10	
	A FOR III	of Ord. 72495, admission	
3	0 200	tax (5.40)	
	102735	Amends § 1 of Ord.	
		101084, official bonds	
		(3.90)	
	102743	Amends § 1 of Ord.	
	oc crea	100219, payroll deductions	
X	ces es	for auto insurance	
		premiums (4.68)	
	102752	Adds § 95-E to Ord.	
P	COMPLE	48022, dog and cat licenses	
		(Repealed by 110230)	
	102770	Medic I Program account	
	ma to	in General Donations and	
Ų.	,800	Gift Trust Fund (5.76)	
	102792	Adds § 95-F to Ord. 48022,	
	<b>4116</b> .	dog and cat licenses	
	100011	(Repealed by 110230)	
	102816	Amends §§ 3.14 and 23.3	
		of Ord. 86300, zoning	
	102817	(Repealed by 117570)	
	102817	Amends §§ 5.3, 14.2, 14.3,	
		15.2, 16.3, 17.2 and 17.3 of Ord. 86300, zoning	
		(Repealed by 117570)	
	102826	Pike Place Market	
	102820	regulations; repeals Ord.	
		85384 (17.32)	
	102835	Amends §§ 23, 25 and 26	
	102033	of Ord. 16046; § 1 of Ord.	
		31592; § 300-E of Ord.	
		48022; § 2 of Ord. 71881,	
		mechanical music	
		machines; § 2 of Ord.	
		97955; § 1 of Ord. 102458;	
		and §§ 1-5 of Ord. 102459,	
		gambling tax (5.52, 6.100)	

eference only.

l amending

graphics,

ccuracy of

Text provided for hi

See ordinan sections for comp and tables and this source file.

Criminal Code; repeals §§ 1, 1A, 3-16, 16A, 17-27, 29, 30, 31, 38-44, 48, 49, 50, 50, 50-1, 51-55 and 57-61 of Ord. 16046; §§ 1 and 2 of Ord. 16208; §§ 1, 2 and 3 of Ord. 16209; §§ 1-5 of Ord. 16373; § 3 of Ord. 24487; §§ 1 and 2 of Ord. 25874; §§ 1 and 2 of Ord. 27396; §§ 2, 3 and 4 of Ord. 29205; § 2 of Ord. 29365; §§ 2-5 of Ord. 30755; §§ 1 and 2 of Ord. 31592; § 2 of Ord. 32986; §§ 1 and 2 of Ord. 34507; §§ 1, 2, 3 and 4 of Ord. 37120; §§ 1, 2 and 3 of Ord. 37147, §§ 1-4 of Ord. 37311; § 1 of Ord. 37916, §§ 1, 2, 4 and 5 of Ord. 39993; §§ 1-7, 7A, 8, 9, 9A, 10 and 12 of Ord. 40149; §§ 1-4 of Ord. 42368; §§ 1, 2 and 3 of Ord. 44347; §§ 1-5 of Ord. 45127; §§ 1, 2 and 3 of Ord. 45860; §§ 1, 2 and 3 of Ord. 52455; §§ 1, 2, 3, 3D and 4 of Ord. 55985; §§ 1, 2 and 3 of Ord. 57328; §§ 1 and 2 of Ord. 60352; §§ 1, 2, 3, 3-1, 4 and 5 of Ord. 64599; §§ 1 and 2 of Ord. 66248; §§ 1 and 2 of Ord. 70829; §§ 1-4 of Ord. 73095; §§ 1 and 2 of Ord. 73778; §§ 1-6 of Ord. 74125; § 2 of Ord. 82371, §§ 1-6 of Ord. 84803; §§ 1, 2 and 3 of Ord. 86065 § 9 of Ord. 89021; §§ 1 and 2 of Ord. 89022; § 3 of Ord. 90007; §§ 21.18.300, 21.26.020, 21.26.040, 21.52.020, 21.52.040, 21.52.060 and 21.52.080 of

Ord. 91910; §§ 1 and 2 of Ord. 92832; § 1 of Ord. 95731; §§ 1, 2 and 3 of Ord. 95759; §§ 1-21 of Ord. 95984; § 1 of Ord. 96045; §§ 1-5 of Ord. 96503; §§ 1-4 of Ord. 96588; § 1 of Ord. 96597; §§ 1-7 of Ord. 96619; §§ 1 and 2 of Ord. 96988; §§ 1-5 of Ord. 97201; §§ 1,2 and 3 of Ord. 97955; §§ 1 and 2 of Ord. 98180 and §§ 1, 2 and 3 of Ord. 10191 (12A.02, 12A.04, 12A.06,

12A.08, 12A.10, 12A.12,

12A.14, 12A.16, 12A.18, 12A.22, 12A.24, 12A.26,

12A.28)

lerk

ference only. amending

For current

June 2009 co Text provide

See ordinant sections for and tables a this source

102844		te file
Sale of unclaimed property by Police (3.28)	102844	Amends § 1 of Ord. 16374.
by Police (3.28)	Che or a	
102847	- 0 101	
by 112555     102850   Civil defense; repeals Ord. 96588 (10.02)     102902   Adds §§ 6A, 6B, 6C and 6D; amends §§ 7, 8 and 9 of Ord. 98852, Pioneer Square Historic District (Repealed by 110058)     102919   Amends §§ 2 and 6 of Ord. 39104, maintenance of healthful temperatures; amends §§ 2.30.020 of Ord. 99112, Housing Code (Repealed by 110152)     102923   Animal Control Division (Repealed by 110230)     102926   Animal Control Division (Repealed by 110230)     102926   Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)     102930   Amends §§ 2 and 3 of Ord. 86799, vacation leave (4.32)     102931   Amends §§ 7 of Ord. 78444, retirement system (4.36)     102965   Amends §§ 1.1 of Ord. 90789, retirement sick leave (4.24)     102971   Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)     102977   Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)     103000   Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)     103014   Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)     103015   Comptroller to deduct disability insurance premiums from employee earnings (4.44)     103016   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)     103026   Amends §§ 1 of Ord. 2630, business and occupation tax (5.44)     103033   Amends § 70 of Ord.	102847	
by 112555     102850   Civil defense; repeals Ord. 96588 (10.02)     102902   Adds §§ 6A, 6B, 6C and 6D; amends §§ 7, 8 and 9 of Ord. 98852, Pioneer Square Historic District (Repealed by 110058)     102919   Amends §§ 2 and 6 of Ord. 39104, maintenance of healthful temperatures; amends §§ 2.30.020 of Ord. 99112, Housing Code (Repealed by 110152)     102923   Animal Control Division (Repealed by 110230)     102926   Animal Control Division (Repealed by 110230)     102926   Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)     102930   Amends §§ 2 and 3 of Ord. 86799, vacation leave (4.32)     102931   Amends §§ 7 of Ord. 78444, retirement system (4.36)     102965   Amends §§ 1.1 of Ord. 90789, retirement sick leave (4.24)     102971   Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)     102977   Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)     103000   Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)     103014   Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)     103015   Comptroller to deduct disability insurance premiums from employee earnings (4.44)     103016   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)     103026   Amends §§ 1 of Ord. 2630, business and occupation tax (5.44)     103033   Amends § 70 of Ord.		Operating Fund (Repealed
102902		
96588 (10.02)   Adds §§ 6A, 6B, 6C and 6D; amends §§ 7, 8 and 9 of Ord. 98852, Pioneer Square Historic District (Repealed by 110058)   Amends §§ 2 and 6 of Ord. 39104, maintenance of healthful temperatures; amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152)   Animal Control Division (Repealed by 110230)   Amends §§ 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)   Amends §§ 2 and 3 of Ord. 86799, vacation leave (4.32)   Amends §§ 7 of Ord. 78444, retirement system (4.36)   Amends §§ 1.1 of Ord. 90789, retirement sick leave (4.24)   Adds §§ 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)   102977   Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)   Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)   Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)   Comptroller to deduct disability insurance premiums from employee earnings (4.44)   103016   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)   103026   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)   103026   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)   103023   Amends § 7 of Ord.	102850	
6D; amends § \$ 7, 8 and 9 of Ord. 98852, Pioneer Square Historic District (Repealed by 110058)  102919		
6D; amends § 8 7, 8 and 9 of Ord. 98852, Pioneer Square Historic District (Repealed by 110058)  102919	102902	Adds §§ 6A, 6B, 6C and
of Ord. 98852, Pioneer Square Historic District (Repealed by 110058) Amends §§ 2 and 6 of Ord. 39104, maintenance of healthful temperatures; amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152) Animal Control Division (Repealed by 110230) Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904) Amends §§ 2 and 3 of Ord. 86799, vacation leave (4.32) Amends § 7 of Ord. 78444, retirement system (4.36) Amends § 1.1 of Ord. 90789, retirement sick leave (4.24) Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719) Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719) Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76) Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20) Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76) Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44) Amends § 70 of Ord.	- oc crea	
(Repealed by 110058)  Amends §§ 2 and 6 of Ord. 39104, maintenance of healthful temperatures; amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152)  Animal Control Division (Repealed by 110230)  Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930  Amends §§ 2 and 3 of Ord. 86799, vacation leave (4.32)  102931  Amends § 7 of Ord. 78444, retirement system (4.36)  Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000  Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends §§ 1 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	ces er er	
Amends § 2 and 6 of Ord. 39104, maintenance of healthful temperatures; amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152)  102926 Animal Control Division (Repealed by 110230)  Animal Control Division (Repealed by 110230)  Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930 Amends § 2 and 3 of Ord. 86799, vacation leave (4.32)  102931 Amends § 7 of Ord. 78444, retirement system (4.36)  102965 Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund; abolishes Manpower and Youth Operating Fund; solishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends § 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		Square Historic District
39104, maintenance of healthful temperatures; amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152)  Animal Control Division (Repealed by 110230)  102926 Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930 Amends § 2 and 3 of Ord. 86799, vacation leave (4.32)  102931 Amends § 7 of Ord. 78444, retirement system (4.36)  102965 Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends § 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		(Repealed by 110058)
39104, maintenance of healthful temperatures; amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152)  Animal Control Division (Repealed by 110230)  102926 Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930 Amends § 2 and 3 of Ord. 86799, vacation leave (4.32)  102931 Amends § 7 of Ord. 78444, retirement system (4.36)  102965 Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends § 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.	102919	
amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152)  Animal Control Division (Repealed by 110230)  Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930 Amends § 2 and 3 of Ord. 86799, vacation leave (4.32)  102931 Amends § 7 of Ord. 78444, retirement system (4.36)  102965 Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends § 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		
amends § 27.30.020 of Ord. 99112, Housing Code (Repealed by 110152)  Animal Control Division (Repealed by 110230)  Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930 Amends § 2 and 3 of Ord. 86799, vacation leave (4.32)  Amends § 7 of Ord. 78444, retirement system (4.36)  102965 Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends § 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.	140 00	healthful temperatures;
(Repealed by 110152)  Animal Control Division (Repealed by 110230)  Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930  Amends § 2 and 3 of Ord. 86799, vacation leave (4.32)  102931  Amends § 7 of Ord. 78444, retirement system (4.36)  102965  Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971  Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000  Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends § 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	and to co	
(Repealed by 110152)  Animal Control Division (Repealed by 110230)  Amends § 4.020 of Ord. 89715, mobile homes and mobile home parks (22.904)  102930  Amends § 2 and 3 of Ord. 86799, vacation leave (4.32)  102931  Amends § 7 of Ord. 78444, retirement system (4.36)  102965  Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971  Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000  Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends § 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	1880	Ord. 99112, Housing Code
Animal Control Division (Repealed by 110230)		
102926	102923	
102926	777	(Repealed by 110230)
89715, mobile homes and mobile home parks (22.904)	102926	
(22.904)  Amends §§ 2 and 3 of Ord. 86799, vacation leave (4.32)  Amends § 7 of Ord. 78444, retirement system (4.36)  Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		_
(22.904)  Amends §§ 2 and 3 of Ord. 86799, vacation leave (4.32)  Amends § 7 of Ord. 78444, retirement system (4.36)  Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  Amends § 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		mobile home parks
86799, vacation leave (4.32)   102931   Amends § 7 of Ord. 78444, retirement system (4.36)   102965   Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)   102971   Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)   102977   Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)   103000   Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)   103014   Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)   103015   Comptroller to deduct disability insurance premiums from employee earnings (4.44)   103016   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)   103026   Amends § 19 of Ord. 72630, business and occupation tax (5.44)   103033   Amends § 70 of Ord.		_
86799, vacation leave (4.32)   102931   Amends § 7 of Ord. 78444, retirement system (4.36)   102965   Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)   102971   Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)   102977   Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)   103000   Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)   103014   Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)   103015   Comptroller to deduct disability insurance premiums from employee earnings (4.44)   103016   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)   103026   Amends § 19 of Ord. 72630, business and occupation tax (5.44)   103033   Amends § 70 of Ord.	102930	
(4.32)   102931   Amends § 7 of Ord. 78444, retirement system (4.36)   102965   Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)   102971   Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)   102977   Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)   103000   Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)   103014   Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)   103015   Comptroller to deduct disability insurance premiums from employee earnings (4.44)   103016   Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)   103026   Amends § 19 of Ord. 72630, business and occupation tax (5.44)   103033   Amends § 70 of Ord.		
Amends § 7 of Ord. 78444, retirement system (4.36)  102965 Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		
retirement system (4.36)  102965  Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971  Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000  Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	102931	· /
Amends § 1.1 of Ord. 90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		
90789, retirement sick leave (4.24)  102971 Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.	102965	
leave (4.24)  102971  Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000  Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
Adds § 14-A to Ord. 48022, general License Code provisions (Repealed by 112719)  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
48022, general License Code provisions (Repealed by 112719)  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.	102971	
Code provisions (Repealed by 112719)  102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		~
by 112719)  102977  Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000  Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
102977 Police Officer of the Year Award account in General Donations and Gift Trust Fund (5.76)  103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		
Donations and Gift Trust Fund (5.76)  103000  Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	102977	
Fund (5.76)  103000  Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		Award account in General
103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		Donations and Gift Trust
103000 Amends §§ 1-4 of Ord. 57142, Seattle Center Veterans' Annex (17.20)  103014 Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015 Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		Fund (5.76)
57142, Seattle Center Veterans' Annex (17.20)  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	103000	
Veterans' Annex (17.20)  103014  Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
Human Resources Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
Operating Fund; abolishes Manpower and Youth Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	103014	` ′
Manpower and Youth Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
Operating Fund (5.76)  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		
103015  Comptroller to deduct disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
disability insurance premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.	103015	
premiums from employee earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
earnings (4.44)  103016  Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026  Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033  Amends § 70 of Ord.		
103016 Amends §§ 1 and 4 of Ord. 103016, gambling tax (5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		
103016, gambling tax (5.52) 103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44) 103033 Amends § 70 of Ord.	103016	
(5.52)  103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.		~ ~
103026 Amends § 19 of Ord. 72630, business and occupation tax (5.44) 103033 Amends § 70 of Ord.		
72630, business and occupation tax (5.44)  103033 Amends § 70 of Ord.	103026	
occupation tax (5.44) 103033 Amends § 70 of Ord.		
103033 Amends § 70 of Ord.		
	103033	
	100000	87983, Harbor Code
(16.60)		
mont Siving	mont SW	1,-5.50

eference only.

a lamending graphics,
accuracy of

For current Simo, the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te me
	103037	Amends § 2; repeals §§ 3
1		and 4 of Ord. 85317, sewer
l	101	connections outside City
l		limits (21.20)
ŀ	103060	Amends § 41 of Ord.
	103000	90047, excavations and
		fills (15.44)
ŀ	102072	
	103063	Amends § 13.5 of Ord.
		86300, zoning (Repealed
L	ac crea	by 117570)
	103064	Amends § 14.5 of Ord.
		86300, zoning (Repealed
		by 117570)
Ī	103072	Amends § 4 of Ord. 66591,
		Special Police (Repealed
		by 109759)
ŀ	103077	
١	103077	PONCHO Display Case
ľ		account in General
		Donations and Gift Trust
		Fund (5.76)
	103087	Amends § 24.41 of Ord.
		86300, zoning (Repealed
		by 117570)
r	103095	Corrosion-prevention
	103073	chemicals in potable hot
		water supply systems and
L	100100	feeding equipment (21.08)
	103103	Amends §§ 7-A-1 and 25-2
		of Ord. 65877, water
		supply system (21.04)
	103105	Amends §§ 17.2, 17.3,
		17.6, 17A.2 and 17A.6 of
		Ord. 86300, zoning
		(Repealed by 117570)
r	103106	Amends §§ 18.2, 18.3 and
	103100	19.2 of Ord. 86300, zoning
		(Repealed by 117570)
L	102120	
	103120	Amends § 22.4 of Ord.
		86300, zoning (Repealed
L		by 117570)
	103140	Korean and Vietnam
١		Casualties Memorial
١		account in General
١		Donations and Gift Trust
١		Fund (5.76)
r	103167	Claims Payment Fund;
	103107	payment of claims against
		the City (Repealed by
L	102177	108657)
١	103175	Amends § 2 of Ord. 78444,
L		retirement system (4.36)
١	103276	Amends § 95-E of Ord.
١		48022, dog and cat licenses
l		(Repealed by 110230)
r	103294	Amends § 28.4 of Ord.
l	103277	86300, zoning (Repealed
l		
ı	102222	by 110381)
H		Amends § 26.1 of Ord.
	103333	
	103333	86300, zoning (Repealed by 117570)

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipa and a	te file
103334	Tony Smith Sculpture
	account in General
a com hi	Donations and Gift Trust
A TOL 132	Fund (5.76)
103364	Amends § 23.3 of Ord.
102201	86300, zoning (24.64)
103381	Amends §§ 28.5 and 28.54
	of Ord. 86300, zoning (Repealed by 109737)
103382	Adds § 4-A to and amends
103382	§§ 3-B and 4 of Ord.
	90138, signs adjacent to
	freeways (Repealed by
COMPA	110058)
103387	Public corporations;
140 0	repeals Ord. 100495
ma to co	(3.110)
103392	Amends § 1 of Ord. 98852,
0210	Pioneer Square Historic
<b>F110</b> .	District (Repealed by
102202	110058)
103393	Adds § 6A; amends §§ 2, 6 and 9 of Ord. 98852,
	Pioneer Square Historic
	District (Repealed by
	110058)
103415	Treasurer's Clearing Fund
	(5.76)
103416	Elderly Activities account
	in General Donations and
	Gift Trust Fund (5.76)
103422	Amends §§ 2, 3, 4 and 5 of
	Ord. 102562, prohibiting
	discriminatory employment
	practices (Repealed by 109116)
103429	Arts Newsletter subaccount
10372)	in Civic Arts account of
	General Donations and Gift
	Trust Fund (5.76)
103434	Burke-Gilman Trail Hiking
	and Biking Fund account in
	General Donations and Gift
	Trust Fund (5.76)
103437	Amends § 98 of Ord.
	48022, dog kennels
103444	(Repealed by 111649) Amends §§ 2 and 3 of Ord.
103444	96020, street vacation
	procedures (Repealed by
	109740)
103449	Amends § 9 of Ord. 72630,
	business and occupation
	tax (5.44)
103454	Amends § 3 of Ord.
	100965; Animal Control
	Commission (Repealed by
	110230)

eference only.

I amending graphics, ccuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	Shar	0210
	a anda	te file
	103458	Amends §§ 4, 4.1 and 5 of Ord. 90379, garbage
	a com hi	collection rates and charges
	d for m	(21.40)
9	103459	Designates old Broadway
		High School as a landmark (25.32)
	103472	Amends §§ 12A.17.010,
		12A.17.080, 12A.17.100
	age crea	and 12A.17.120 of Ord.
3	669 00	102843, Criminal Code (12A.14)
	103473	Amends § 1 of Ord. 99870,
	COMPA	fees for certain Police
•		Department reports (3.28)
	103513	Amends § 4 of Ord. 90952,
4	iller co	ambulance licensing (Repealed by 110151)
	103514	Amends § 198 of Ord.
	sile.	48022, peddlers or
5	7770	solicitors (Repealed by
	103517	109271) Amends §§ 13B.1 and 18.4
	103317	of Ord. 86300, zoning
		(Repealed by 117570)
	103539	Adds § 12A.40.030 to Ord.
		102843, Criminal Code;
		adds § 2.1 to Ord. 64599 (12A.24)
	103566	Amends § 19 of Ord.
		72630, business and
	102500	occupation tax (5.44)
	103588	Debt Management Policy Advisory Committee (3.76)
	103597	Abolishes Prisoner Benefit
		Account and transfers
		balance to Korean and
		Vietnam Casualties Memorial Account (5.76)
	103598	Amends §§ 1, 2 and 4 of
		Ord. 102459, gambling tax
	100177	(5.52)
	103655	Amends § 3 of Ord. 98852, Pioneer Square Historic
		District (Repealed by
		110058)
	103661	Temporary employment
		service in Executive Department, Personnel
		Office (Repealed by
		114811)
	103667	Washington Park (18.16)
	103672	Amends § 5 of Ord. 83834,
		health care program; amends § 5 of Ord.
		100862, dental care
		program (Repealed by
	102702	115159)
	103703	Housing Rehabilitation Holding Fund (5.76)
ı		

eference only.

The serence only

June 2009 co Text provide

See ordinant sections for and tables a this source

	Cipar		
	a anda	te tille	
	103704	Amends §§ 1 and 3 of Ord. 38390, appointment of	
	d for hi	deputies by City Comptroller (Repealed by	
	103726	110938)  Deposit to Comprehensive Planning Assistance Fund	
	103767	(5.76) Annual award program for	
(	ces crea	City employees (Repealed by 118966)	
	comple	Amends § 4 of Ord. 103167, Comprehensive Planning Assistance Fund (5.76)	
<b>A</b>	103825 to CO	Social card games (12A.50)	
•	103839	Amends § 22.4 of Ord. 86300, zoning (Repealed	
	103840	by 117570)  Amends § 23.6 of Ord.  86300, zoning (Repealed by 117570)	
	103843	Amends § 24.3 of Ord. 86300, zoning (Repealed by 117570)	
	103864	Adds §§ 23.7 and 23.8 to and amends § 23.2 of Ord. 86300, zoning (Repealed by 117570)	
	103866	Amends § 3 of Ord. 100435, Code of Ethics (Repealed by 108882)	
	103880	Amends § 7 of Ord. 93789, cabulance licenses (Repealed by 109348)	
	103881	Amends § 4 of Ord. 90952, ambulance licensing	
	103891	(Repealed by 110151)  Amends § 34 of Ord.  90047, street use	
	103892	regulations (15.40) Initiative petitions (2.08)	
	103993	Adds §§ 12A.24.020-12A.24.070, 12A.40.140, 12A.40.160, 12A.40.180, 12A.40.200, 12A.40.220, 12A.40.240	
		and 12A.40.260; amends § 12A.24.010; repeals §§ 12A.24.080, 12A.24.100, 12A.24.120, 12A.24.140,	
		12A.24.120, 12A.24.140, 12A.24.160, 12A.24.180, 12A.24.200, 12A.24.220, 12A.24.240 and 12A.24.260 of Ord.	
		102842, Criminal Code (12A.18, 12A.24)	
	104012	Amends §§ 19.2, 19.3 and 19.6 of Ord. 86300, zoning (Repealed by 117570)	
_			

rd. eference only.

y

de amending
de graphics,
de accuracy of

For current Shie, and Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
104033	Amends § 5 of Ord. 62662,
104049	utilities business tax (5.48)
104048	Amends §§ 95 and 95(a) of
ad in	Ord. 48022, dog and cat
	licenses (Repealed by
104040	110230)
104049	Amends §§ 12A.36.040, 12A.36.080 and
	12A.36.100; repeals §§ 12A.36.020 and
DOS CITES	12A.36.060 of Ord.
	102843, Criminal Code (12A,22)
104059	Amends § 3 of Ord. 72630,
140.00	business and occupation tax (5.44)
104060	Amends §§ 4 and 4.2 of
1220	Ord. 84390, sewerage rates
	and charges (21.28)
104062	Amends §§ 7, 13, 18, 33,
777	34, 36 and 38 of Ord.
	65877, water supply
	system (21.04)
104063	Adds § 3.1 to Ord. 72495,
	bowling; and amends § 23
	of Ord. 48022, exhibitors
	or trade shows (6.20)
104067	Office of Policy Planning;
	amends §§ 1 and 2 and
	repeals § 3 of Ord. 97473,
	Department of Community
	Development; amends § 1
	of Ord. 101667,
	Department of Human
	Resources; amends § 3 of
	Ord. 103726, transfer of
	function to Office of Policy
	Planning; deposit to
	comprehensive Planning
	Assistance Fund (Repealed
104069	by 108735)
104068	Adds §§ 24.921, 24.922,
	24.923, 24.924 and 24.925
	to and amends § 24.92 of
	Ord. 86300, zoning
104092	(Repealed by 112134)
104083	Unemployment
	compensation program
104097	(4.40, 5.76)
104087	Amends § 1 of Ord.
	102459, gambling tax;
	amends § 12A.36.200 of Ord. 102843, Criminal
104005	Code (5.52)
104095	Amends § 5 of Ord.
	102562, prohibiting
	discriminatory employment
	practices (Repealed by 109116)
	107110)

eference only.

| Construction of the control of th

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipus andate	file
104100 Amen	ds §§ 1 and 1.1 of
	00789, retirement sick
leave	(4.24)
104156 Amen	ds §§ 2, 6, 12, 13,
14(c)	and 14(f) of Ord.
78444	, retirement system
(4.36)	-
104184 Repea	als Ord. 104035 and
reinsta	ates text of §§ 4 and
4.2 of	Ord. 84390 to be as
	nended by Ord.
	60, sewerage rates and
	es (21.28)
	ds § 23.4 of Ord.
	, zoning (Repealed
by 117	
	ds § 3 of Ord. 86799,
	on leave (4.32)
	ng and Community
	opment Revenue
	ng Fund (5.76)
	ds § 2 of Ord. 25757,
	ng in polling places
(2.20)	
	§ 8A to Ord. 89021,
	control (Repealed by
104201 Amen	ds § 10 of Ord.
l l	, ambulance ing; amends § 19 of
	93789, cabulance
	e (Repealed by
11015	
	ds § 13.3, 13.4, 14.1,
	and 21.2 of Ord.
	, general License
	provisions (6.02)
	d of Public Health
	tment permit fees
(3.30)	
104213 Amen	ds §§ 17.2, 17.6 and
	of Ord. 86300,
zoning	g (Repealed by
11757	(0)
	§§ 376-382 to Ord.
	, housing rental
	ies (Repealed by
10967	
l l	ds § 24.92 of Ord.
	), zoning (Repealed
by 112	
104235 Amen	ds § 3 of Ord. 72630,
	ess and occupation
busine	
busine tax (5.	.44)
busine tax (5. 104244 Amen	.44) ds §§ 6 and 14(c) of
busine tax (5) 104244 Amen Ord. 7	.44)

eference only.

a deciral amending as a graphics, a ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipai		
A	te file	
104271	Adds §§ 13C.1-13C.8 to	
104271	and amends §§ 2.2, 3.04,	
- 0 - m hi	4.1, 23.1 and 24.41 of Ord.	
d for III	86300, zoning (Repealed	
	by 117570)	
104281	Amends §§ 1, 2 and 4 of	
	Ord. 57142, Seattle Center	
104207	Veterans' Hall (17.20)	
104297	Adds §§ 148-B-148-G; amends §§ 146, 147, 148	
res crea	and 148-A of Ord. 48022.	
	public parking (6.48)	
104301	Plumbing Code (Repealed	
COLLIA	by 106721)	
104311	Amends § 5 of Ord.	
140 0	103387, public	
ma to co	corporations (3.110)	
104312	Amends § 12A.12.160 of	
009	Ord. 102843, Criminal	
Sile.	Code (12A.10)	
104313	Amends § 9 of Ord.	
104317	100708, unit prices (7.12)	
104317	Amends § 12 of Ord. 66591, Special Police	
	(Repealed by 109759)	
104344	Seattle School Traffic	
	Safety Committee (3.80)	
104346	Mutual firefighting	
	assistance agreement with	
	port cities (3.16)	
104348	Amends § 4.3 of Ord.	
	84390, sewerage rates and	
101055	charges (21.28)	
104357	Amends § 5 of Ord. 62662,	
104382	utilities business tax (5.48) Amends §§ 4 and 13 of	
104362	Ord. 78444, retirement	
	system (4.36)	
104383	Amends § 10 of Ord.	
	102228, Administrative	
	Code (3.02)	
104388	Adds § 95(g) to Ord.	
	48022, dog and cat licenses	
104400	(Repealed by 110230)	
104408	Amends § 1 of Ord. 86799,	
	vacation leave (Repealed	
104423	by 110195) Amends §§ 15.2 and 17.21	
104423	of Ord. 86300, zoning	
	(Repealed by 117570)	
104424	Amends § 2 of Ord. 86927,	
	suggestion awards program	
	for City officers and	
	employees (Repealed by	
	118966)	
104430	Amends §§ 30 and 32 of	
	Ord. 48022, auctions and	
	auctioneers (Repealed by	
104434	109818) Amends § 5 of Ord. 62662,	
107737	utilities business tax (5.48)	
MONT DI	danies daniess (3.40)	

eference only.

a amending
graphics,
ccuracy of

For current Single, June 13,48)
the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

	te file
104445	Adds § 1A to Ord. 32986,
101413	accident reports by
109	physicians (10.28)
104446	Amends § 16 of Ord.
	81599, impoundment
	(Repealed by 110230)
104447	Amends §§ 1 and 2 of Ord.
101117	95751, reimbursement for
	use of private automobile
orea	or City business; repeals
ces crea	Ord. 101482 (Repealed by
	110451)
104455	Amends § 11 of Ord.
COMPA	90379, garbage collection
	rates and charges (5.76,
	21.40)
104472	Utility credits for the
They	elderly (21.76)
104485	Amends § 12A.12.160 of
Calle	Ord. 102843, Criminal
1116.	Code (12A.10)
104486	Adds § 37A to Ord. 16046,
104400	cruelty to animals
	(Repealed by 110230)
104523	Amends § 3 of Ord.
104323	103726, increased
	appropriation to
	Comprehensive Planning
	Assistance Fund (5.76)
104526	Defense of City officers
	and employees; repeals
	Ord. 99509 (4.64, 5.76)
104531	Amends § 2 of Ord.
104331	102666, use of City owned
	or leased vehicles (4.68)
104539	Adds § 1A to Ord. 29025,
104339	leaving drug or chemical
	samples on private
	property (10.32)
104540	Amends § 2 of Ord. 93771,
107340	coloring baby rabbits or
	fowl (9.20)
104542	Relocation assistance for
107374	disabled persons (20.84)
104561	Amends § 51A and adds §
107301	51B to Ord. 97330,
	compensation of officers
	and employees (4.20)
104563	Amends § 2 of Ord. 71614,
107303	registration of persons
	authorized to enter
	business establishments
	(6.96)
	Amends §§ 2, 5, 7, 10, 11,
104572	
104572	12 14(h) and 17 of Ord
104572	12, 14(h) and 17 of Ord.
104572	78444, retirement system
	78444, retirement system (4.36)
104572	78444, retirement system

eference only. amending graphics, ccuracy of

For current SM the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

Cihar	4 - 6110
a - mada	te me
104601	Amends §§ 1, 2, 3, 5, 6 and
	7 of Ord. 101295, travel
- C - 40 10 1	expenses (4.72)
104615	Office of Urban
	Conservation subaccount in
	Civic Arts account of
	General Donations and Gift
10.1.100	Trust Fund (5.76)
104620	Amends § 190 and repeals
DOS CIES	§ 191 of Ord. 48022,
CC3 C	peddlers and solicitors
104623	(Repealed by 109271) Amends § 1 of Ord.
104023	•
Compage	104346, firefighting assistance agreement (3.16)
104649	Amends §§ 104, 105 and
104049	111; repeals § 106 of Ord.
Hiller co	48022. for-hire drivers
	(Repealed by 109648)
104652	Amends § 6 of Ord. 72495,
	admission tax (5.40)
104657	Adds §§ 3A and 3B;
10.007	amends §§ 1, 2, 5, 7, 12,
	13, 15 and 16 of Ord.
	102826, Pike Place Market
	regulations (17.32)
104658	Amends § 6 of Ord.
	100475, Pike Place Market
	Historical District (25.24)
104659	Adds § 2B to Ord. 90138,
	signs adjacent to freeways
	(Repealed by 117570)
104660	Amends §§ 5.3, 15.1, 16.1,
	17.1, 17A.1, 18.1, 24.91
	and 24.921 of Ord. 86300,
	zoning (Repealed by
104667	117570) Amends § 2 of Ord.
104007	100458, use of City owned
	or leased vehicles (4.68)
104670	Practical paramedic
104070	training fee (3.16)
104675	Amends §§ 14.2 and 14.3
	of Ord. 86300, zoning
	(Repealed by 117570)
104685	Amends § 4.3 of Ord.
	84390, sewerage rates and
	charges (21.28)
104710	Adds § 5A to and amends §
	5 of Ord. 102151, Division
	of Purchases (3.18)
104725	Adds § 51C to Ord. 97330,
	compensation of officers
	and employees (Repealed
	by 111768)
104731	Fire extinguisher use
10.1707	training program (3.16)
104735	Amends §§ 15.21 and
	15.24 of Ord. 86300,
	zoning (Repealed by
461	117570)

reference only.

in lift amending graphics, ccuracy of

For current SN 117570) Clerk
the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibor	to file
111100	
104749	Amends § 1 of Ord.
	104083, unemployment
Land Mil	insurance program (4.40)
104795	Amends § 25.2 and adds §§
	25.40-25.46 to Ord. 86300,
	zoning (Repealed by
	117570)
104802	Adds § 16.A; amends § 6
	of Ord. 62662, utilities
-ac offed	business tax (5.48)
104809	Amends §§ 24.45, 24.46
	and 27.41 of Ord. 86300,
	zoning (Repealed by
COMP	117570)
104811	Pike Place Project
1 40 00	Rehabilitation Escrow
and to C	Fund, Project WASH R-17
O'BROS	(5.76)
104839	Prohibits unfair housing
210	practices; repeals §§
III.	12A.44.010-12A.44.060 of
	Ord. 102843 (14.08)
104843	Amends § 1 of Ord.
104043	100458, use of City owned
	or leased vehicles (3.18)
104849	Amends § 7 of Ord. 90535,
10404)	vehicle wrecker and towing
	service (Repealed by
	109080)
104870	Amends § 17.1 of Ord.
104870	
	86300, zoning (Repealed
104878	by 117570) Amends §§ 1, 2, 3 and 4 of
104878	' '
	Ord. 90881; amends § 3 of
	Ord. 101715, disability
104001	compensation (4.44)
104901	Amends § 19.7 of Ord.
	86300, zoning (Repealed
104007	by 117570)
104907	Amends § 3 of Ord.
	102182, Seattle Women's
	Commission (Repealed by
	109114)
104908	Salary adjustment date for
	officers and employees not
	covered by collective
	bargaining agreements
	(Repealed by 117479)
104912	Amends § 6 of Ord. 78444,
	retirement system (4.36)
104922	Amends § 34 of Ord.
	65877, water supply
	system (21.04)
104937	Amends § 1 of Ord. 99911,
	naming of parks and
	recreation areas (18.08)
104953	Amends § 2 of Ord.
	102560, deferral of
	assessment collection
	(20.12)

reference only.

ss
o.
d amending
graphics,
accuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

(	ilhar .	0210
		te tille
	104971	Amends §§ 5.3, 15.1, 16.1,
		17.1, 17A.1 and 18.1 of
	1 Cor hi	Ord. 86300, zoning
9	104972	(Repealed by 117570)
	104972	Amends § 2B of Ord. 90138, signs adjacent to
		freeways (Repealed by
		117570)
	105005	Adds § 6.1 to Ord. 88522,
	as crea	sick leave (4.24)
7	105007	Amends §§ 5, 6, 8 and 9;
		repeals § 4 of Ord. 80818, tattooing (6.102)
	105013	Christopher Columbus
		Commemorative Sculpture
	4 40 CC	account in General
4	ma to	Donations and Gift Trust
	105124	Fund (5.76)
	105124	Amends § 1 of Ord. 101295, travel expenses
	1110.	(4.72)
	105139	Amends § 3 of Ord. 72630,
		business and occupation
	107110	tax (5.44)
	105140	Adds § 3.2 to Ord. 72630, business and occupation
		tax (5.44)
	105143	Transfer to City-County
		Health Fund; abolishes
		Public Health Revolving
		Fund; repeals Ord. 99690
	105144	(Repealed by 110895) Transfer to City-County
	103144	Health Fund; abolishes
		Special Nursing Fund
		(Repealed by 110895)
	105145	Cost of immunizations
		provided by Public Health
		Department (Repealed by 110903)
	105146	Fees for health certificate
		for travelers (Repealed by
		110903)
	105147	Fees for educational
		classes in prenatal and infant care, child rearing
		and guidance (Repealed by
		108033)
	105148	Family planning services
		of Public Health
		Department (Repealed by 112567)
	105149	Fees for medical laboratory
		services (Repealed by
		110271)
	105150	Amends § 5 of Ord.
		102151, Division of
	105156	Purchases (3.18) Adds § 13C.9 to Ord.
	103130	86300, zoning (Repealed

d amending graphics, ccuracy of

For current by 117570)

For Current State City Clerk
the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipus da	te file
105176	Amends § 27.3 of Ord.
	86300, zoning (Repealed
- 0 - m hi	by 110381)
105177	Adds §§ 148-H and 148-I;
U Z	amends §§ 146 and 148-A
	of Ord. 48022, public
	parking (6.48)
105187	Parks and Recreation
	Department operation of
- oc crea	fee parking lots (18.24)
105189	Adds § 198-1 to Ord.
	48022, peddlers or
	solicitors (Repealed by
COMPA	109271)
105239	Amends § 70 of Ord.
140 0	87983, Harbor Code
ma to co	(16.60)
105260	Amends § 14.3 of Ord.
	86300, zoning (Repealed
5116.	by 117570)
105265	Amends § 2 of Ord. 90695,
	underground utility district
	(21.68)
105289	Amends § 22.3 of Ord.
	86300, zoning (Repealed
	by 117570)
105294	Amends § 14A.3 of Ord.
	86300, zoning (Repealed
	by 117570)
105296	Transfer of Data
	Processing Center to
	Executive Department,
	Office of Management and
	Budget (Repealed by
105308	109576) Amends § 9 of Ord. 72630,
103308	business and occupation
	tax (5.44)
105333	Amends § 46 of Ord.
103333	97330, compensation of
	officers and employees
	(4.20)
105338	Adds § 24.88 to and
	amends §§ 24.83-24.87 of
	Ord. 86300, zoning
	(Repealed by 112134)
105359	Amends § 24.2 of Ord.
	86300, zoning (Repealed
	by 117570)
105361	Amends §§ 95 and 95-c of
	Ord. 48022, dog and cat
	licenses (Repealed by
	110230)
105389	Amends §§ 2, 3, 4 and 5;
	repeals § 6 of Ord. 102210,
	art in public works
	construction (5.76, 20.32)
105408	Amends §§ 6.2, 12.1 and
	14.2 of Ord. 86300, zoning
	(Repealed by 117570)
	A CONTRACTOR

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

<b>90</b> ,	da	te file
105	409	Amends § 1 of Ord.
	0 0-3	105145, immunization fees
		(Repealed by 110903)
105	411	Amends § 1 of Ord. 91885,
103.		-
		Seattle Center Advisory
		Commission (Repealed by
10.5		108936)
105	423	Adds §§ 5A and 5B to and
		amends §§ 3-6 of Ord.
	of CITE	102562, prohibiting
		discriminatory employment
		practices (Repealed by
	-	109116)
105	427	CATV regulations (21.60)
105	430	Adds §§ 347.1, 347.2,
		348.1, 348.2, 348.3, 349,
44	A TO C	350, 350.2, 351-358 and
		358.1. charitable
ľ		solicitations; amends § 3,
	10	,
	115.	general provisions; repeals §§ 347-359.7 of Ord.
-		0.0
		48022, License Code
		(6.02)
105	433	Amends §§ 27.1, 27.8 and
		28.4 of Ord. 86300, zoning
		(Repealed by 110381)
105	445	Amends § 6 of Ord. 72495,
		admission tax (5.40)
105	450	Leasehold excise tax (5.56)
105	462	Ballard Avenue Landmark
		District (25.16)
105	513	Amends § 28.2 of Ord.
100.		86300, zoning (Repealed
		by 117570)
105	537	Amends §§ 3, 4, 5 and 8 of
105.	337	Ord. 104472, utility credits
		for the elderly (21.76)
105	<b>5</b> 28	• •
105	336	Amends §§ 1 and 8 of Ord.
10.5		100708, unit prices (7.12)
105	544	Amends §§ 4, 4.1 and 5 of
		Ord. 90379, garbage rates
		and charges (21.40)
105	565	Amends §§ 3.21, 5.3, 16.2
		and 17.21 of Ord. 86300,
		zoning (Repealed by
		117570)
105	566	Division on Aging Senior
		Opportunities Gifts and
		Donations account in
		General Donations and Gift
		Trust Fund (5.76)
105	584	Amends § 18.7 of Ord.
	JUT	86300, zoning (Repealed
105.		
103.		by 117570)
	(21	DC 1C ' '
105	631	Refund of certain parks and
105		recreation fees (3.106)
		recreation fees (3.106) Subdivisions; repeals Ord.
105		recreation fees (3.106)

eference only.

l amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

_	a senda	te Ille
	105637	Amends §§ 1 and 4 of Ord.
		104526, defense of City
	- 0 - 0 101	officers and employees
		(4.64, 5.76)
)	105638	Greening Sculpture:
	100 000	Gasworks Park subaccount
		in Civic Arts account in
		General Donations and Gift
		Trust Fund (5.76)
_	105654	Amends § 22.3 of Ord.
	105054	86300, zoning (Repealed
		by 117570)
_	105655	Community advisory
	103033	councils' assistance to
	Compa	Parks and Recreation
	105551	system (18.04)
	105674	Amends § 3 of Ord.
		102712, Head Start
		Program (Repealed by
	5416.	110999)
	105679	Leasehold excise tax (5.56)
	105725	Marine firefighting
		assistance agreement (3.16)
	105728	Amends § 26.3 of Ord.
		86300, zoning (Repealed
		by 117570)
	105734	Amends § 173; repeals §
		174 of Ord. 48022,
		massage parlors and
		bathhouses (6.36)
	105735	State Environmental Policy
		Act guidelines (Repealed
		by 111866)
	105781	Amends §§ 1, 2 and 3 of
		Ord. 95751, reimbursement
		for use of private auto on
		City business (Repealed by
		110451)
	105805	Amends § 4.2 of Ord.
		84390, sewerage rates and
		charges (21.28)
	105836	Amends § 4 of Ord. 72495,
		admission tax (5.40)
	105840	Michael Heizer Sculpture
	100010	subaccount in Civic Arts
		account in General
		Donations and Gift Trust
		Fund (5.76)
	105843	Amends § 46 of Ord.
	105045	97330, compensation
	105947	(4.20)
	105847	Citizen handbooks
	105054	(Repealed by 110586)
	105854	Amends § 24.43 of Ord.
		86300, zoning (Repealed
		by 117570)

eference only.

that amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

ilha.	. 6110
a anda	te ille
105876	Amends §§ 15.1, 15.24,
	16.1, 16.3, 17.1, 17.21,
1 C - 10 10 1	17.3, 17A.1, 18.1, 19-1 and
4 FOL 112	23.1; adds §§ 13B.0,
	13C.2, 13C.3, 23.31 and
	23.41 to Ord. 86300,
	zoning (Repealed by
	117570)
105887	Amends § 7 of Ord. 78444,
- or ored	retirement system (4.36)
105912	Amends § 1 of Ord. 79957,
	City office hours (3.102)
105931	Amends § 17.21 of Ord.
COMPA	86300, zoning (Repealed
	by 117570)
105936	Amends § 24.92; repeals
and to co	§§ 24.921-24.925 and adds
	new §§ 24.921-24.929 to
	Ord. 86300, zoning
C110.	(Repealed by 112134)
105939	Amends § 46 of Ord.
	97330, compensation of
	officers and employees
	(4.20)
105961	Amends § 49 of Ord.
100,01	97330, compensation of
	officers and employees
	(4.20)
105962	Antirecessionary Grant
	Fund (5.76)
105987	Public Reinvestment
	Review Board (Repealed
	by 110226)
105990	Amends § 3 of Ord.
100,50	104083, unemployment
	insurance program (4.40)
105991	Amends § 6.1 of Ord.
103771	88522, sick leave (4.24)
105998	Amends § 2 of Ord. 82583,
103770	sewer connection fees
	(Repealed by 116421)
105999	Amends § 7.08.060 of Ord.
103999	90181, sewage disposal
	systems (21.32)
106000	Amends § 1 of Ord. 99951,
100000	fee for reports on water
	supply and sewage systems
	(Repealed by 107099)
106001	
106001	Amends §§ 3-B and 4 of
	Ord. 90138, signs adjacent
	to freeways (Repealed by
10,000	117570)
106002	Amends § 1 of Ord. 99492,
	administrative charges for
	public works contracts
	(Repealed by 117390)

eference only.

amending

reference only.

amending

reference only.

amending

reference only.

June 2009 co

Text provide

See ordinant sections for and tables a this source

ospo	te file
106007	Amends § 198, peddlers and solicitors, and repeals
ed for hi	§ 145-A of Ord. 48022, License Code; repeals § 1 of Ord. 75088 and Ord.
	65447, expiration of licenses (Repealed by 109271)
106012	Amends § 1 of Ord. 84046, Seattle Fire District No. 5
ces cro	Fund (Repealed by 112611)
106013 COMPLE	Amends §§ 3, 11, 12, 13 and 17 of Ord. 72630, business and occupation
106014	Amends §§ 5 and 16 of Ord. 105636, subdivisions (Repealed by 110570)
106015	Amends §§ 24.43, 27.1, 27.8 and 28.4 of Ord. 86300, zoning (Repealed by 117570)
106017	Amends §§ 4, 4.1 and 5 of Ord. 90379, garbage collection rates and charges
106019	(21.40) Amends § 1 of Ord. 96020, street vacation procedures (Repealed by 109740)
106024	Revenue License Code;
106025	amusement devices (5.32)  Adds § 13.5 to Ord. 48022, general License Code provisions; amends § 16 of Ord. 80028, preparation and sale of poultry, rabbit and fish; adds § 9-A to Ord. 80818, tattooing; adds § 13-A to Ord. 84297, refrigeration systems; adds § 2.015 to Ord. 89715, mobile homes and mobile home parks; amends § 28 of Ord. 90047, street use regulations; adds § 4-A to Ord. 90145, tobacco vending machines; adds § 2-A to Ord. 90535, vehicle wrecker and towing service; adds § 2.1 to Ord. 93789, cabulance licenses; adds § 3.1 to Ord. 94465, distribution and sale of meat; amends § 35 of Ord. 98829, Weights and Measures Code (6.02, 6.82, 6.102, 7.04, 22.904)

eference only.

amending

apphics,

accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	0210
a anda	te file
106037	Amends §§ 22, 23 and 24;
	repeals §§ 23-1, 23-a, 23-b,
1 0 0 101	23-c, 23-c-1, 23-d, 23-e,
4 FOR 111	23-f, 25, 26, 26-A, 27,
	27-A, 27-B, 27-C, 49 and
	51 of Ord. 48022,
	amusement (6.20)
106041	Amends § 5 of Ord. 62662,
	utilities business tax (5.48)
106045	Amends § 4 of Ord. 59866,
ces cross	for-hire vehicles; amends §
	3 of Ord. 104474, utility
	credits for the elderly
COMPA	(21.76)
106050	Designates Old Firehouse
	No. 23 as a landmark
and ED C	(25.32)
106051	Designates Old Firehouse
	No. 3 as a landmark
C310.	(25.32)
106052	Designates Old Firehouse
	No. 18 as a landmark
	(25.32)
106053	Designates Old Firehouse
	No. 33 as a landmark
	(25.32)
106054	Designates Old Firehouse
	No. 25 as a landmark
	(25.32)
106055	Enforcement of
	antidiscrimination
	provisions of City
	contracts; repeals Ord.
	102812 (Repealed by
	117407)
106057	Amends §§ 10.1, 11.1 and
	22.3 of Ord. 86300, zoning
	(Repealed by 117570)
106058	Amends § 11-1 of Ord.
	72495, admission tax;
	amends § 20 of Ord.
	72630, business and
	occupation tax; amends §
	29 of Ord. 94465,
	distribution and sale of
	meat; amends §§ 1 and 3 of
	Ord. 103415, Treasurer's
	Clearing Fund; Refund
	account in General Fund;
	repeals Ords. 70792,
	81163, 90268, 92146 and
	99227 (5.40, 5.44, 5.76,
	10.12)
106060	Amends § 14 of Ord.
	80028, preparation and sale
	of poultry, rabbit, and fish
	(Repealed by 110052)

d

52, s

66, graphics, s

67, curacy of

June 2009 co 10003 Up Text provide d for

See ordinances comsections for comand tables and tables and this source file.

106063 Amends §§ 90-2 and 91-1, detectives and detective agencies; § 97, dog kennels; § 111, for-hire drivers; §§ 172, 173, 173.1 and 176.1, massage parlors and bathhouses; § 192-1, peddlers or solicitors; § 261, merchant patrolmen; §§ 301 and 302, locksmiths; and § 331.1, bail bondsmen of Ord. 48022, License Code; amends §§ 6, 6A and 7 of Ord. 73223, fumigators; amends § 13 of Ord. 84297, refrigeration system; amends § 2.010 of Ord. 89715, mobile homes and mobile home parks; amends § 5.01.180 of Ord. 104301, Plumbing Code (6.14, 6.34, 6.36, 6.38)106067 Designates Ward House as a landmark (25.32) 106068 Designates Stimson Green House as a landmark (25.32)Designates West Queen 106069 Anne Walls as a landmark (25.32)106070 Designates Arboretum Aqueduct as a landmark (25.32)106071 Designates Ellsworth Storey Houses as a landmark (25.32) 106072 Designates Chinese Community Bulletin Board as a landmark (25.32) 106073 Amends § 18 of Ord. 105735, SEPA guidelines (Repealed by 111866) 106078 Amends § 41 of Ord. 90047, street use regulations (15.44) 106087 Designates Parish Episcopal Church as a landmark (25.32)

ference only.

amending

graphics,

ccuracy of

For current SMC, contact the Office of the City Clerk

Amends § 5 of Ord. 62662, utilities business tax (5.48)

Amends §§ 3.09, 6.22, 7.2, 8.2, 10.2, 12.11, 12.12, 12.13, 12.2 and 23.3 of Ord. 86300, zoning (Repealed by 117570)

Designates St. Nicholas Cathedral as a landmark

106088

106090

106098

June 2009 co

Text provide

See ordinant sections for and tables a this source

	anda	te file
	106099	Designates St. Spiridon Cathedral as a landmark
	som hi	(25.32)
2)(	106100	Designates Seattle Buddhist Church as a
		landmark (25.32)
H	106102	Seattle Center Gifts
		account in General
		Donations and Gift Trust
L	oc crea	Fund (5.76)
4	106107	Fire Code; repeals Ords.
	10	87870, 88210, 88339, 89864, 90005, 90405,
	comple	91984, 92898, 93220,
		93733, 93826, 94931,
	140 00	96018, 96169, 96677,
N.	na to o	98466, 98572, 98868,
•		101645, 103133, 103457,
	0210	105617 (Repealed by 108678)
1	106119	Amends § 23.2 of Ord.
	10011)	86300, zoning (Repealed
		by 117570)
	106140	Amends § 24.43 of Ord.
		86300, zoning (Repealed
L.	102141	by 117570)
	106141	Designates Flatiron Building (Triangle Hotel)
		as a landmark (25.32)
F	106142	Designates Immaculate
		Conception Church as a
L		landmark (25.32)
	106143	Designates 20th Avenue
		Northeast Bridge as a landmark (25.32)
H	106144	Designates Capitol Hill
	100111	United Methodist Church
		as a landmark (25.32)
	106145	Designates First Church of
		Christ, Scientist as a
H	106146	landmark (25.32) Designates West Queen
	100140	Anne Elementary School
		as a landmark (25.32)
F	106147	Designates Old Main Street
		School as a landmark
		(25.32)
	106157	Amends § 2 of Ord. 90695,
		underground utility district (21.68)
-	106158	Amends §§ 4 and 4.2 of
		Ord. 84390, sewerage rates
		and charges; amends § 4 of
		Ord. 97016, side sewers
		(21.16, 21.28)
L		Adds §§ 4.4-4.14 and
-	106162	
	106162	5.1-5.4, sewerage rates and charges (Repealed by

d amending graphics, accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

Ciber -	to file
1011111111	
106166	Mechanical Code; repeals Ord. 52113 (Repealed by
a combi	111201)
106167	Refund of special
u so-	connection charges (21.24)
106168	Amends § 15 of Ord.
	62662, utilities business
	tax; amends § 15 of Ord.
	72495, admission tax;
oc crea	amends § 19 of Ord.
CE2 C2	72630, business and
	occupation tax (5.44, 5.48)
106181	Amends § 58 of Ord.
COMPA	48022, cabarets (Repealed
	by 109653)
106189	Amends § 103, for-hire
ma to o	drivers, and § 238 for-hire
18800	vehicles of Ord. 48022,
	License Code (Repealed by
2316.	109648)
106200	Adds Art. 21A and § 30.1
	to Ord. 86300, zoning
	(Repealed by 113466)
106213	Amends § 28.53 of Ord.
	86300, zoning (Repealed
	by 109737)
106214	Energy Office (Repealed
	by 110254)
106216	Consultants in Engineering
	Department (Repealed by
	110986)
106220	Amends § 4 of Ord.
	104472, utility credits for
	the elderly (21.76)
106230	Amends §§ 3.02 and 5.1 of
	Ord. 86300, zoning
	(Repealed by 117570)
106240	Amends § 2.010 of Ord.
	160024, Revenue License
	Code (5.32)
106272	Amends §§ 12, 13 and
	14(h) of Ord. 78444,
	retirement system (4.36)
106273	Designates San Mateo
	Steam Ferry as a landmark
	(25.32)
106274	Designates Wawona
	Schooner as a landmark
	(25.32)
106275	Designates Relief
	Lightship as a landmark
	(25.32)
106276	Designates Authur Foss
	Tug as a landmark (25.32)
106277	Designates W. T. Preston
	Snagboat as a landmark
	(25.32)
106278	Designates Virginia V
	Excursion Boat as a
	landmark (25.32)

d amending graphics, accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	te file
106282	Amends § 12.13 of Ord.
10	86300, zoning (Repealed by 117570)
106289	Amends § 13.5 of Ord.
	48022, General License
10.000	Code provisions (6.02)
106295	Amends § 1 of Ord. 106058, Refund account in
	General Fund (5.76)
106302	City Planning
ces cross	Commission; repeals Ord.
	104177 (Repealed by 109155)
106309	Amends § 6 of Ord.
	100475, Pike Place Market
1 40 C	Historical District (25.24)
106313	Amends § 2 of Ord. 85317,
	sewer connections outside City limits (21.20)
106319	Housing Code; repeals
2116.	Ords. 99112, 100641,
	101978, 103403 and
	105839 (Repealed by
	113545)
106339	Amends § 20 of Ord.
	105735, SEPA guidelines (Repealed by 111866)
106340	Amends § 25.40 of Ord.
1000.10	86300, zoning (Repealed
	by 110381)
106346	Amends § 3 of Ord. 4953,
	annual franchise reports
106348	(21.64) Landmarks preservation;
100346	repeals Ord. 102229
	(25.12)
106348-A	Amends § 3 of Ord.
	102228, Administrative
	Code (3.02)
106352	Prohibits use of City
	Treasurer's name on utility charge statements (21.72)
106356	Amends § 1 of Ord. 94446,
	Seattle Center Department
	(17.04)
106360	Noise control; repeals
	Ords. 70150, 90007,
	90726, 103609 and § 1 1/2 of Ord. 15956 (25.08)
106416	Municipal Lighting and
100110	Power System; repeals
	Ord. 104030 (Repealed by
	109218)
106418	Amends § 39 of Ord.
	65877, water supply
106436	system (21.04) Amends § 23.2 of Ord.
100430	Amends § 23.2 of Ord. 86300, zoning (Repealed
	by 117570)

eference only.

l amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
106437	Amends § 1 of Ord.
	101295, travel expenses
- 0 - m hi	(4.72)
106438	Amends § 25.40 of Ord.
	86300, zoning (Repealed
	by 110381)
106454	Amends § 1 of Ord.
	100591, fees for copies of
	City books and records
oc crea	(3.104)
106463	Amends § 21A.40 of Ord.
	86300, zoning (Repealed
91000	by 113466)
106467	Amends § 26 of Ord.
	106107, Fire Code
140 0	(Repealed by 108678)
106470	Amends § 20 of Ord.
	105735, SEPA guidelines
	(Repealed by 111866)
106471	Amends § 7 of Ord.
3330	100435, Code of Ethics
	(Repealed by 108882,
	109950)
106474	Amends §§ 1, 2, 3 and 6 of
	Ord. 99554, Department of
	General Services (Repealed
	by 109129)
106475	Amends § 5 of Ord.
	102562, prohibiting
	discriminatory employment
	practices (Repealed by
106476	109116)
106476	Amends § 4 of Ord.
	104839, prohibiting unfair
106477	housing practices (14.08)
106477	Amends § 10 of Ord.
	102228, Administrative
106470	Code (3.02)
106479	Amends § 15.1 of Ord. 86300, zoning (Repealed
	by 117570)
106480	Seattle-King County
100400	Department of Public
	Health Citizens Advisory
	_
106503	Board (3.30) Amends §§ 24.924 and
100000	24.925 of Ord. 86300,
	zoning (Repealed by
	112134)
106512	Amends § 28 of Ord.
	48022, auctions and
	auctioneers (Repealed by
	109818)
106516	Amends § 31 of Ord.
	98820, Weights and
	Measures Code (7.04)
106522	Amends § 2 of Ord.
	10584/. citizen handbooks
	105847, citizen handbooks (Repealed by 110586)

eference only.

l amending

graphics,

ccuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

44471114	
106525	Amends §§ 1-6 of Ord.
	101295, travel expenses
106526	(4.72) Amends § 5 of Ord. 62662
100320	utilities business tax (5.48)
106527	Amends § 3.2 of Ord.
100327	72630, business and
	occupation tax (5.44)
106546	Amends § 2-B of Ord.
oc crea	82583, sewer connection
CE2 es	fees (21.24)
106561	Amends § 12 of Ord.
	72630, business and
COMPLE	occupation tax (5.44)
106565	Consultant in bridge design
ad to C	for Engineering
	Department (Repealed by 110986)
106583	Adds §§ 19-A-19-I;
100000	amends § 6; repeals § 22-A
1110.	of Ord. 90047, street use
	regulations (15.02, 15.14)
106587	Amends § 19(c) of Ord.
	78444, retirement system
	(4.36)
106588	Amends § 4 of Ord. 78444
	retirement system (4.36)
106589	Amends § 21A.40 of Ord.
	86300, zoning (Repealed
	by 113466)
106594	Amends § 1 of Ord.
	105149, fees for medical
	laboratory services
106595	(Repealed by 110271) Amends § 28.2 of Ord.
100373	86300 zoning (Repealed by
	117570)
106598	Amends § 56 of Ord.
	103387, public
	corporations (3.110)
106615	Park Code; repeals Ords.
	76027, 78096, 85252,
	94550, 96454 and 98778
104400	(18.12)
106623	Amends § 4 of Ord. 96897
	Seattle Design Commission
106626	(3.58) Amends §§ 23.22, 23.28
100020	and 23.29 of Ord. 86300.
	zoning (Repealed by
	117570)
106651	Conservation Investment
	Assistance Program
	(21.52)
106653	Election campaign
	contributions; repeals Ord.
	100241 (2.04, 5.76)
106656	K-9 Unit Gifts account in
	General Donations and Gif
	Trust Fund; repeals Ord.
	105193 (5.76)

reference only.

G2,
H8)

amending
graphics,
curacy of

For current Sharp, Sharp Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

a anda	te file
106662	Amends §§ 19.4, 20.4 and
	21.4 of Ord. 86300, zoning
106688	(Repealed by 117570) Amends § 1 of Ord.
100088	106594, fees for medical
	laboratory tests (Repealed
	by 110271)
106698	Amends § 2 of Ord.
- 1000	106522, citizen handbooks
106721	(Repealed by 110586)
100/21	Plumbing Code; repeals Ord. 104301 (Repealed by
100016	109033)
106724	Amends § 10 of Ord.
	102228, Administrative
1400	Code (3.02)
106734	Amends § 24.92 of Ord.
	86300, zoning (Repealed by 112134)
106750	Amends §§ 1, 3 and 4 of
	Ord. 96020, street vacation
	procedures (Repealed by
	109740)
106751	Amends § 6 of Ord. 72495,
106764	admission tax (5.40) Amends § 1 of Ord.
100/04	105296, Data Processing
	Center (Repealed by
	109576)
106775	Amends §§ 3.05, 2.07 and
	23.3 of Ord. 86300, zoning
106780	(Repealed by 117570) Amends §§ 17 and 25 of
100780	Ord. 90047, street use
	regulations (15.08, 15.22)
106797	Urban Renewal Close Out
	subaccount in Housing and
	Community Development
	Revenue Sharing Fund (5.76)
106798	Amends §§ 19 and 24 of
	Ord. 78444, retirement
10 100 1	system (4.36)
106804	Amends § 2 of Ord. 92887,
	Lake Union construction limit line (Repealed by
	117570)
106828	Amends § 2b of Ord.
	106416, Municipal
	Lighting and Power
	System (Repealed by
106832	109218) Amends §§ 3.02, 18.3 and
100032	20.2 of Ord. 86300, zoning
	(Repealed by 117570)
106844	Traffic consultants for
	Engineering Department
	(Repealed by 110986)

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Clpar		
	a mada	te file	
	106848	Amends §§ 3.19, 14A.7,	
		15.3, 16.3, 17.3 and 19.21	
	a sor hi	of Ord. 86300, zoning	
	106862	(Repealed by 117570) Amends §§ 3.02, 3.05,	
	100802	3.17, 3.22, 16.6 and 17.61	
		of Ord. 86300, zoning	
		(Repealed by 117570)	
	106869	Economic Development	
	oc crea	Grant Fund (5.76)	
)	106874	Amends § 4 of Ord. 81599,	
	16	impoundment (Repealed by 110230)	
	106875	Amends § 2 of Ord.	
		106055, enforcement of	
	1 40 CC	antidiscrimination	
A	ma to	provisions of City contracts	
٠		(Repealed by 117407)	
	106884	Amends § 57 of Ord.	
	<b>1116</b> .	48022, cabarets (Repealed	
	106896	by 109653) Amends § 4 of Ord. 84390,	
	100890	sewerage rates and charges	
		(21.28)	
	106931	Amends § 17 of Ord.	
		72630, business and	
		occupation tax (5.44)	
	106946	Amends § 4 of Ord.	
		106651, Conservation	
		Investment Assistance	
	100000	Program (21.52)	
	106950	Amends § 14.2 of Ord.	
		86300, zoning (Repealed by 117570)	
	106952	Amends § 28.2 of Ord.	
	100/32	86300, zoning (Repealed	
		by 117570)	
	106958	P-patch program garden	
		plot fees (3.20)	
	106959	Water Demand Charge	
		Fund (Repealed by	
	10000	109398)	
	106960	Transfer to General Fund; repeals Ord. 101289,	
		Fireman's Disability and	
		Leave Allowance Fund	
		(5.76)	
	106962	Amends § 1 of Ord. 99870,	
		fees for certain Police	
		Department reports (3.28)	
	106963	Conservatory Donations	
		account in Parks and	
		Recreation Fund;	
		Conservatory donations box in Volunteer Park	
		(5.76, 18.16)	
	106965	Paper recycling program	
	100/03	(3.18)	
	106966	Amends § 1 of Ord.	
		106058, Refund account of	
	4 CM	General Fund (5.76)	
		7	

eference only.

amending

applics,
by graphics,
accuracy of

For current State City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipar		
	a mada	te ille
	106967	Amends §§ 148-A and
		148-H of Ord. 48022,
	106970	parking (6.48) Solid waste disposal
	100970	(21.44)
	106985	Amends § 24.85 of Ord.
		86300, zoning; amends § 6
		of Ord. 98852, Pioneer Square Historic District;
	- and	amends § 6 of Ord.
	ces crea	100475, Pike Place Market
		Historical District; amends
	comple	§ 12.02 of Ord. 106348,
	Compa	landmarks preservation; amends § 301 and adds §§
	440.00	303A, 303B and 303C to
	ad to co	Ord. 106106, permit fees
•	100000	(25.12, 25.24)
	106992	Amends §§ 1 and 7 of Ord. 106651, Conservation
	1112.	Investment Assistance
		Program (21.52)
	106995	Reimbursement of travel
		expenses for certain applicants for City
		employment (Repealed by
		117242)
	107012	Floating home moorage
		tenants rights (Repealed by 109280)
	107017	Amends § 10 of Ord.
		106107, Fire Code
	107001	(Repealed by 108678)
	107024	Amends § 12.2 of Ord. 86300, zoning (Repealed
		by 117570)
	107025	Amends §§ 7, 11 and 15 of
		Ord. 105427, CATV
	107055	regulations (21.60) Rental of Seattle Center
	107033	buildings; repeals Ords.
		99203, 101257, 104037
	105070	and 104936 (17.16)
	107063	Amends §§ 1, 2, 3, 4 and 5 of Ord. 104083,
		unemployment insurance
		program (4.40, 5.76)
	107069	Amends § 17.3 of Ord.
		86300, zoning (Repealed by 117570)
	107071	LID subaccount in Housing
		and Community
		Development Revenue
	107074	Sharing Fund (5.76) Amends §§ 5.3 and 23.1 of
	10/0/4	Ord. 86300, zoning
		(Repealed by 117570)
	107075	Amends §§ 3.02, 3.05,
		3.17, 3.22, 16.6 and 17.61 of Ord. 86300, zoning
		(Repealed by 117570)
	- 40 T T 171	

d amending graphics, accuracy of

For current Six (Repealed by 117570) to the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipal Cile		
	a mada	te file
	107080	Adds § 4.2 to Ord. 90379,
	4-9	garbage collection rates
	a for m	and charges (Repealed by 116419)
	107081	Amends § 34 of Ord.
		65877, water supply
	107000	system (21.04)
	107088	Amends § 24.92 of Ord. 86300, zoning (Repealed
	- 0402	by 112134)
	107092	Amends § 8 of Ord.
		100708, unit prices (7.12)
	107095	Adds §§ 400-413, for-hire
	COIIIB	vehicles, and repeals §§
		232-239 of Ord. 48022,
	ad to CC	License Code; amends §§
4	III co	1, 2, 2-i, 3, 4, 5, 13, 14, 14-6, 15, 16, 17, 20-23, 25
		and 28; adds §§ 3-a, 3-b
	C110.	and 27-a, and repeals §§ 12
	Ille.	and 26 of Ord. 59866,
		for-hire vehicles (Repealed
		by 109348)
	107096	Amends § 1 of Ord.
		105148, family planning
		services of Public Health Department (Repealed by
		112567)
	107098	Agreements for health care
		with King County
		municipalities (3.30)
	107099	Reports on condition of
		water supply and sewage
		disposal systems; repeals Ords, 99951 and 106000
		(10.25)
	107109	Amends §§ 3.02, 14.2,
		14A.2, 15.21, 16.23, 17.21,
		18.2, 19.7, 20.7, 21.6 and
		28.2, of Ord. 86300, zoning
	107128	(Repealed by 117570)
	10/120	Amends § 2 of Ord. 102182, Office of
		Women's Rights (Repealed
		by 109114)
	107133	Handicapped person and
		senior citizen identification
		program (Repealed by
	107137	109560) Amends § 3.2 of Ord.
	10/15/	72630, business and
		occupation tax (5.44)
	107153	Amends § 1 of Ord. 99870,
		fees for certain Police
		Department reports (3.28)

reference only.

d amending
praphics,
accuracy of

Seattle Municipal Code

June 2009 co 107157 up date

Amend
13.1-13
58, 9098, 107
134, 13
1773, 17
201, 20
303, 30
303, 30
Ord. 48
(6.02, 6)

See ordinan ces com sections for com and tables and tables and this source file.

Amends §§ 9, 11, 13, 13.1-13.4, 14A, 15, 16, 23, 58, 90-3, 95, 95-A, 95-F, 98, 107, 108, 130, 131, 134, 135, 137, 165, 172, 173, 173-1, 176.1, 196-A, 201, 207, 240, 261, 262, 303, 309, 312 and 372 of Ord. 48022, License Code (6.02, 6.14, 6.20, 6.36, 6.38)

Amends § 10 of Ord.

Amends § 10 of Ord.
62662, utilities business
tax; amends § 12 of Ord.
72630, business and
occupation tax; amends §§
8 and 16 of Ord. 80028,
preparation and sale of
poultry, rabbit and fish;
amends § 8 of Ord. 80818,
tattooing; amends §§ 3 and
9 of Ord. 84297,

refrigeration systems;

amends § 6 of Ord. 85744, fumigators; amends § 2.020 of Ord. 89715, mobile homes and mobile home parks; amends § 28 of Ord. 90047, street use regulation; amends §§ 2 and 3 of Ord. 90145, tobacco vending machines; amends §§ 2, 5, 7, and 9 of Ord. 90535, vehicle wrecker and towing service; amends § 3 of Ord. 90952, ambulance licensing; amends § 3 of Ord. 93789, cabulance licenses; amends §§ 4 and 6 of Ord. 94465,

distribution and sale of meat; amends § 35 of Ord. 98820, Weights and Measures Code; amends § 1.100 of Ord. 106024, Revenue License Code

(5.32, 5.44, 5.48, 6.82, 6.102, 6.104, 7.04, 22.904)

107164 Amends §§ 4, 13, and 23 of Ord. 78444, retirement system (4.36)

107172 Amends § 46 of Ord. 97330, compensation and working conditions for City officers and employees (4.20)

107186 Amends §§ 12.12, 12.13, 12.2 and 13.1 of Ord. 86300, zoning (Repealed by 117570)

ference only.

amending

raphics,

For current Smc, Courterly Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

a mda	te file
107194	Amends § 17.61 of Ord.
	86300, zoning (Repealed
10 - m h1	by 117570)
107208	Adds § 3A to and amends
	§§ 2, 3 and 10 of Ord.
	96003, collection and
	disposal of garbage and
	refuse (21.36)
107228	Amends §§ 2, 16, 18 and
- a orea	19 of Ord. 78444,
ces cross	retirement system (4.36)
107232	Amends § 12A.01.140 of
	Ord. 102843, Criminal
CULIBRIO	Code (12A.02)
107245	Amends § 12A.08.230 of
1 4 0 00	Ord. 102843, Criminal
TO C	Code (12A.08)
107246	Adds §§ 12A.36.050,
10,210	12A.36.053, 12A.36.055,
Cilo	12A.36.058, 12A.36.060,
TILE	12A.36.063, 12A.36.065
	and 12A.36.068; repeals §§
	12A.36.100, 12A.36.120,
	12A.36.140 and
	12A.36.160 of Ord.
	102843, Criminal Code
107250	(12A.22)
107230	Surcharge to water
	customers Water District
	No. 61, King County
107251	(21.12)
107251	Amends § 4.2 of Ord.
	90379, garbage collection
	rates and charges
107252	(Repealed by 116419)
107252	Real property leases for use
	by City departments and
105050	agencies (3.18)
107270	Amends §§ 410 and 411 of
	Ord. 48022, for-hire
	vehicles (Repealed by
107070	109348)
107278	Amends §§ 1, 2 and 4;
	renumbers § 6 to be § 7,
	and adds new § 6 to Ord.
	102459, gambling tax
10=01	(5.52)
107296	Amends § 1 of Ord. 90497,
	arterial City street fund
10=000	(5.76)
107309	Amends § 12A.01.140
	"Arrest-Citations" and
	renumbers § 12A.01.140
İ	"Definitions" to be §
	12A.01.150 of Ord.
	102843, Criminal Code
	(12A.02)
107312	· · · · · · · · · · · · · · · · · · ·

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

107323	Minimum maintenance
	standards for historic
a combi	buildings in Pioneer Square
4 FOL III	Historic District; repeals §§
CI ZO	6A, 6B and 6C of Ord.
	98852 (5.76, 25.28)
107350	Amends § 19.1 of Ord.
	48022, general License
	Code provisions (6.02)
107377	Amends § 701 of Ord.
ces er en	106360, noise control
	(Repealed by 112567)
107412	Amends § 1 of Ord.
COMPA	105148, family planning
	services of Public Health
	Department (Repealed by
and to co	112567)
107429	Amends § 34 of Ord.
107129	65877, water supply
C:10	system (21.04)
107442	Amends § 1 of Ord. 99685,
10/442	membership in Washington
	Public Power Supply
	System (21.56)
107447	Amends §§ 1, 3 and 4 of
10/44/	Ord. 101295, travel
	expenses (4.72)
107448	Amends §§ 27.1, 27.8 and
107446	28.4 of Ord. 86300, zoning
	(Repealed by 110381)
107453	Adds §§ 24.911-24.919 to
107433	and amends § 24.91 of
	Ord. 86300, zoning
	(Repealed by 112134)
107483	Bond for collector of City
107483	taxes (King County
	Comptroller) (Repealed by
	112072)
107501	Adds §§ 10A and 20B to
107301	
	and amends §§ 3, 4, 7, 8, 12, 13 and 15-20 of Ord.
	105735, SEPA guidelines
	(Repealed by 111866)
107513	Adds § 12A.46.130 to Ord.
10/313	102843, Criminal Code
	(Repealed by 109560)
107521	
10/321	Designates Holyoke Building as a landmark
	(25.32)
107522	Designates Myron Ogden
107322	residence as a landmark
107526	(25.32)
107526	Designates Coliseum
	Theater as a landmark
107527	(25.32)
107537	Amends §§ 3.11, 3.24,
	6.21, 14.2, 17.21, 17.3,
	17A.7, 18.3, 19.31, 20.3,
	21.6 and 23.3 of Ord.
	86300, zoning (Repealed by 117570)

reference only.

dl amending
graphics,
accuracy of

For current Shie, the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

		te file
	107539	Amends § 1 of Ord.
	000	104731, training in use of
	101	fire extinguishers (3.16)
	107560	Amends § 1 of Ord. 94446,
-		Seattle Center Department
		(17.04)
	107564	Purchase and resale of
	107304	conservation materials by
		Lighting Department
		(21.52)
	107578	
7	10/5/8	Authorizes Mayor to
	10	accept gifts of works of art
	107570	for display (20.36)
	107579	Amends § 12.2 of Ord.
		86300, zoning (Repealed
	140 66	by 117570)
	107585	Ballard Avenue Project,
Y		Miro FitzGerald mosaic
		subaccount in Civic Arts
	<b>5112.</b>	account of General
	727	Donations and Gifts Trust
		Fund (5.76)
	107599	Aquarium Donations
		account in General Gifts
		and Donations Trust Fund
		(5.76)
	107600	Aquarium Donations
		account in Parks and
		Recreation Fund (5.76)
	107613	Designates
		Thompson/LaTurner
		House as a landmark
		(25.32)
	107614	Adds § 13A to and amends
		§§ 4, 13, 14 and 15 of Ord.
		205427, Office of Cable
		Communications; repeals
		Ord. 102998 (21.60)
	107619	Seattle Center storage
		space rental (17.16)
	107621	Designates Brill Trolley as
		a landmark (25.32)
	107634	Lease agreement authority
		of Superintendent of
		Buildings; repeals Ord.
		106729 (3.18)
	107660	Amends §§ 2a, 2b, 4a and
	107000	4b of Ord. 106416,
		Municipal Lighting and
		Power System (Repealed
		by 109218)
	107661	Adds § 11(a) to Ord.
	10/001	0 ( )
		106416, Municipal
		Lighting and Power
		System (Repealed by
	107670	109218)
	107670	Adds §§ 12A.04.180 and
		12A.04.185 to Ord.
		102843, Criminal Code
		(12A.06)

eference only.

l amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
107678	Policies for implementation
63.00	of SEPA guidelines
- 101	(Repealed by 111866)
107679	Columbia City Landmark
10/0/3	District (25.20)
107707	Condominium conversion
107707	
10777	regulations (22.902)
107725	Amends § 2 of Ord. 96020,
	street vacation procedures
oc crea	(Repealed by 109740)
107750	Designates Easter Hotel as
	a landmark (25.32)
107751	Designates Cotterill House
COMPA	as a landmark (25.32)
107752	Designates Del a Mar
	Apartments as a landmark
and to co	(25.32)
107753	Designates Austin A. Bell
10,755	Building as a landmark
Calle.	(25.32)
107754	
10/734	Designates Barnes
	Building as a landmark
	(25.32)
107755	Designates Chelsea
	Apartments as a landmark
	(25.32)
107756	Designates Epiphany
	Chapel as a landmark
	(25.32)
107759	Amends § 4 of Ord.
	102562, prohibiting
	discriminatory employment
	practices (Repealed by
	109116)
107766	Amends §§ 1 and 2 of Ord.
107700	107564, contracts for
	installation of insulation
107772	materials in homes (21.52)
107772	Adds §§ 13-A-13-M to and
	amends §§ 2, 15 and 21 of
	Ord. 106653, election
	campaign contributions
	(2.04, 5.76)
107773	Adds § 17.1 to Ord. 48022,
	general License Code
	provisions (6.02)
107785	Board of Public Works
	(Repealed by 116007)
107786	Building Department and
	Superintendent (Repealed
107787	by 109124)
107787	by 109124) City Light Department and
	by 109124) City Light Department and Superintendent (3.08)
107787 107788	by 109124) City Light Department and Superintendent (3.08) Water Department and
107788	by 109124) City Light Department and Superintendent (3.08) Water Department and Superintendent (3.32)
	by 109124) City Light Department and Superintendent (3.08) Water Department and Superintendent (3.32) Department of Engineering
107788	by 109124) City Light Department and Superintendent (3.08) Water Department and Superintendent (3.32) Department of Engineering and Director (3.12)
107788	by 109124) City Light Department and Superintendent (3.08) Water Department and Superintendent (3.32) Department of Engineering and Director (3.12) Personnel system; repeals
107788	by 109124) City Light Department and Superintendent (3.08) Water Department and Superintendent (3.32) Department of Engineering and Director (3.12) Personnel system; repeals Ords. 69816, 73052,
107788	by 109124) City Light Department and Superintendent (3.08) Water Department and Superintendent (3.32) Department of Engineering and Director (3.12) Personnel system; repeals

eference only.

I amending
graphics,
accuracy of

For current Sites, and Clerk the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

4	te file
107791	Public safety civil service
	regulations (4.08)
107797	Amends § 1.02 of Ord.
a'sor m	106319, Housing Code
CI TO	(Repealed by 113545)
107828	Exemptions from civil
10/828	
	service (Repealed by
105000	109004)
107829	Amends § 3 of Ord.
age Crea	106107, Fire Code
	(Repealed by 108678)
107830	Amends § 4 of Ord. 81599,
- ammie	animal impoundment
COMPA	(Repealed by 110230)
107831	Streetcar Fund (Repealed
140 00	by 112915)
107832	Amends § 1 of Ord. 99492,
18800	administrative charges for
	public works contracts
C110.	(Repealed by 117390)
107833	Amends § 1 of Ord.
107033	106958, P-patch garden
	plot fees (3.20)
107927	
107836	Amends § 2 of Ord. 83834,
	§ 2 of Ord. 98295, and § 2
	of Ord. 100862, transfer of
	administration of health
	care program, group
	accidental death and
	dismemberment insurance
	plan, and dental care
	program to Personnel
	Department (Repealed by
	115159)
107837	Adds §§ 8-A and 8-B to
	Ord. 92987, food service
	establishment fees
	(Repealed by 109910)
107840	Rates for data processing
107070	services (Repealed by
107941	108656)
107841	Free parts for electric
	appliances for low-income
	senior citizens (Repealed
	by 114835)
107845	Adopts maps identifying
107845	environmentally sensitive
107845	
107845	environmentally sensitive
107845	environmentally sensitive areas (Repealed by 111866)
	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord.
	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code
107867	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code (Repealed by 109033)
	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code (Repealed by 109033) Office of Policy Planning,
107867	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code (Repealed by 109033) Office of Policy Planning, Program 6000 (Repealed
107867	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code (Repealed by 109033) Office of Policy Planning, Program 6000 (Repealed by 108735)
107867	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code (Repealed by 109033) Office of Policy Planning, Program 6000 (Repealed by 108735) Amends § 1 of Ord.
107867	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code (Repealed by 109033) Office of Policy Planning, Program 6000 (Repealed by 108735) Amends § 1 of Ord. 105149, fees for medical
107867	environmentally sensitive areas (Repealed by 111866) Adds § 5.02.095 to Ord. 106721, Plumbing Code (Repealed by 109033) Office of Policy Planning, Program 6000 (Repealed by 108735) Amends § 1 of Ord.

eference only.

d amending
praphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	ilber 40	to file
	4 44400	
	107890	Repeals and replaces § 2 of
		Ord. 107055, rental of
	4 Cost 111	Seattle Center buildings
	A TOL 332	(Repealed by 110896)
	107903	Adds § 4A to and amends
		§§ 2, 3, 5 and 6 of Ord.
		102228, Administrative
		Code (3.02)
	107904	Adds § 1A to and amends §
		1 of Ord. 83897, vacancies
	DES CITCO	in Charter offices (3.92)
77	107905	
	10/905	Amends § 5 of Ord. 83834,
	- mille	health care; amends §§ 1,
	COMP	4, 5, 6, 6.1, 7 and 8 and
		repeals § 9 of Ord. 88522,
	140 00	sick leave; amends § 2 of
		Ord. 90789, retirement sick
•		leave; amends §§ 1 and 3
		of Ord. 99753, funeral
	C:10.	leave (4.24, 4.28)
	107906	Amends § 1 of Ord. 99870,
	107,500	fees for certain Police
		Department reports (3.28)
	107945	Amends § 19 of Ord.
	107943	82630, business and
	107046	occupation tax (5.44)
	107946	Amends § 94 of Ord.
		48022, dog and cat licenses
		(Repealed by 110230)
	107948	Amends § 2 of Ord. 86363,
		Voucher Fund (5.76)
	107949	Amends § 2 of Ord. 77411,
		Salary Fund (5.76)
	107951	Fees for X-rays for
		asbestos-related lung
		disease (10.03)
	107977	Amends § 16 of Ord.
		107790, personnel system
		(4.04)
	107978	Amends §§ 2 and 13-C of
		Ord. 106653, election
		campaign contributions
		(2.04)
	107990	Amends § 19.22 of Ord.
	107770	86300, zoning (Repealed
		by 117570)
	107002	Designates Fremont Hotel
	107993	
	105005	as a landmark (25.32)
	107995	Designates Montlake
		Bridge as a landmark
		(25.32)
	108009	Amends §§ 5 and 14(f) of
		Ord. 78444, retirement
	<u></u>	system (4.36)
	108020	Amends §§ 6, 10, 17, 19
		and 33 of Ord. 90047,
		street use regulations
		(15.02, 15.04, 15.08, 15.12,
		15.38)
ı		- 40

eserence only.

amending

applics,

curacy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	4- 5112
	ILG PILL
108033	Repeals Ord. 105147, fees
	for health education classes
0 - 101	(Repealer)
108053	Amends § 5 of Ord.
	105636, subdivisions
	(Repealed by 110570)
108066	Stevens Neighborhood
	Strategy Area Revolving
	Development Account in
oos crea	Housing and Community
CE2 OF	Development Revenue
	Sharing Fund (5.76)
108074	Amends § 1 of Ord.
Comple	107828, exemptions from
	civil service (Repealed by
140 C	109004)
108076	Amends §§ 133-137, 140,
	141, 143 and 144 of Ord.
0210	48022, steam engineers and
1411e.	boiler firemen; amends § 6
8800	of Ord. 73223, fumigators;
	amends § 5 of Ord. 84297,
	refrigeration systems;
	amends §§ 4 and 11 of Ord. 94465, distribution
	· ·
	and sale of meat; repeals Ord. 107776 (6.54, 6.82)
108077	Adds § 23 to Ord. 107791,
108077	public safety civil service
	(4.08)
108080	Grading and drainage
100000	control; amends § 7000 and
	repeals §§ 7001-7009 of
	Ord. 106350, Building
	Code; repeals Ord. 107378
	(Repealed by 117697)
108090	Seattle Center office space
	rental agreements
	(Repealed by 112790)
108097	Amends § 1 of Ord. 98511,
	data processing services
	(Repealed by 109576)
108098	Office of
	Intergovernmental
	Relations; amends §§ 1, 2,
	and 6 of Ord. 140067,
	Office of Policy Planning
	(3.14)
108106	Amends § 2 of Ord.
	100708, unit prices (7.12)
108113	Lloyd Hamrol Sculpture
	subaccount in Municipal
	Art Fund (5.76)
108118	Amends § 1 of Ord. 16374,
	sale of unclaimed property
	by Police Department
	(3.28)
108124	Amends § 2 of Ord. 16374,
	Police Department
	(Repealed by 109757)
	The Contract

d amending graphics, accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

Cipai					
	andate file				
	108125	Amends §§ 2 and 3 of Ord.			
		107252, increase in rental			
	- 0 - m hi	payments for lease of			
	A for III	non-City-owned properties			
Ċ		(3.18)			
	108128	Amends § 1 of Ord. 67337,			
		solicitation of funds in City			
		department or division			
		(3.100)			
	108130	Amends § 13-J of Ord.			
	ces es	107772, fair campaign			
		practices (5.76)			
	108137	Amends §§ 3.21, 12.11,			
	COINE	14.3, 15.22, 15.3 and 18.23			
		of Ord. 86300, zoning			
	140 00	(Repealed by 117570)			
	108144	Amends § 5 of Ord. 90145,			
C.		tobacco vending machines			
		(Repealed by 115531)			
	108147	Amends § 4-A of Ord.			
	777	80028, preparation and sale			
		of poultry, rabbit, and fish			
		(Repealed by 110052)			
	108149	Amends § 1 of Ord. 99982,			
		Seattle Arts Commission			
	100150	(Repealed by 121006)			
	108150	Amends § 1 of Ord.			
		101531, beekeeping			
	100150	(10.36)			
	108152	Amends § 2 of Ord. 32444,			
		physical examination of			
	100157	arrested persons (10.30)			
	108157	Permit fee ordinance;			
		repeals Ord. 107379			
		(Repealed by 108729,			
	100101	109506)			
	108191	Amends § 12A.17.160 of			
		Ord. 102843, Criminal			
		Code (12A.14)			

eference only.

amending

graphics,

ccuracy of

Seattle Municipal Code ference only. date June 2009 co 108196 Adds §§ 21.3 and 500-529, and amends §§ 14, 21 and Text provided for hi 401 of Ord. 48022, License Code; amends § 36 of Ord. 98820, weights and measures; repeals Ords. amending 59866, 62610, 69144, 69377, 69585, 69680, 70429, 71905, 73762, See ordinances crea 74086, 78208, 79136, 80008, 80648, 86242, 87597, 87630, 89192, sections for comp 91012, 91670, 92658, 93542, 93887, 94021, 95032, 95501, 95715, and tables and to 96433, 97192, 97839, 99225, 100479, 102637, 103347, 103690, 103880, 106133, 106178, 106189, this source file. 106609, 107133, 107144, 107351, 107574, 107742, 107918 and 108087; repeals §§ 1-22 of Ord. 101857, § 3 of Ord. 104201, § 24 of Ord. 106025, §§ 1-6, 8, 9 and 10 of Ord. 106189, and §§ 3-10 and 12-25 of Ord. 107095, taxicabs (6.02, 7.04)108199 Amends § 3 of Ord. 106107. Fire Code

(Repealed by 108678)

June 2009 co 108200 upus Text provide d for hi

See ordinances crossections for compand tables and to this source file.

108200 Traffic Code; repeals Ords. 91910, 92153, 93078, 95875, 95969, 96032, 97339, 99722, 100929, 101361, 101605, 101762, 101840, 102053, 102361, 103203, 103366, 103770, 103927, 104047, 104126, 104532, 104604, 104682, 104905, 104952, 105385, 105673, 105853, 105895, 106782, 106882, 106926, 107156, 107601, 107736, and §§ 12A.28.020, 12A.28.040, 12A.28.060, 12A.28.100, 12A.28.120, 12A.28.140, 12A.28.160, 12A.28.165 and 12A.28.180 of Ord. 102843 (11.10, 11.12, 11.14, 11.16, 11.18, 11.20, 11.22, 11.23, 11.24, 11.25, 11.26, 11.27, 11.28, 11.30, 11.32, 11.34, 11.40, 11.42, 11.44, 11.50, 11.52, 11.53, 11.54, 11.55, 11.56, 11.57, 11.58, 11.59, 11.60, 11.62, 11.64, 11.66, 11.68, 11.70, 11.72, 11.74, 11.76, 11.80, 11.82, 11.84, 11.86) 108205 Amends §§ 2, 3, 4 and 7 of Ord. 104839, prohibiting unfair housing practices (14.08)108210 Designates Nowell House as a landmark (25.32) 108211 Designates the Victorian Group as a landmark

(25.32)

(25.32)

(25.32)

(2.08)

(25.32)

Designates Charles Bussell House as a landmark

Designates Bystrom House as a landmark (25.32)

Amends §§ 1 and 2 of Ord. 103892, initiative measures

Designates The Victorian Row Apartments as a landmark (25.32)

Designates Ballard Howe House as a landmark

Designates Italianate Victorian Pair as a landmark (25.32)

Designates P.P. Ferry Mansion (St. Mark's Deanery) as a landmark

108212

108213

108214

108216

108224

108225

108226

ference only.

amending

raphics,

curacy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

a anda	te file
108227	Designates Anhalt
	Apartments as a landmark
a can hi	(25.32)
108236	Amends § 4 of Ord.
	104472, utility credits for
	the elderly (21.76)
108270	Designates Lacey V.
	Murrow Bridge, West
	Plaza, Mt. Baker Tunnel
oc crea	and East Tunnel Portals
CE2 CT	(Mercer Island Floating
	Bridge) as a landmark
	(25.32)
108289	Amends § 330.6 of Ord.
	48022, panorams and
140 00	peepshows (6.42)
108307	Amends § 5 of Ord. 48022,
	general License Code
	provisions (6.02)
108309	Amends § 12A.17.160,
777	Criminal Code (Repealed
	by 117157)
108312	Form for referendum
	petitions (2.12)
108314	Amends § 3 of Ord.
	100862, dental care
	program (Repealed by
	115159)
108329	Control of data processing
	related contracts (Repealed
	by 118397)
108333	Collection of information
	for law enforcement
	purposes (14.12)
108352	Amends § 30 of Ord.
	86300, zoning (Repealed
	by 110381)
108354	Adds §§ 11.14.782,
	11.16.315, 11.16.317,
	11.28.010 and 11.72.351 to
	and amends §§ 11.16.120
	and 11.16.300 of Ord.
	108200, Traffic Code
	(11.14, 11.16, 11.28,
100255	11.72)
108355	Amends § 22 of Ord.
	72630, appeals from tax
	deficiency determinations
100257	(5.44)
108357	Help the Animals Account
	in General Donations and
100250	Gift Trust Fund (5.76)
108358	Amends §§ 16.21, 17.21
	and 23.3 of Ord. 86300,
	zoning (Repealed by
	117570)
108379	117570) Amends § 12-c of Ord.
108379	117570)

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipar of cite			
a and	ate me		
108382	Adds § 28-A; repeals § 28 of Ord. 90047, licensing of		
10	house movers (15.28)		
108395	Mrs. Betty Wile Levy		
Charles	Medical Aid Unit		
	Memorial account in		
	General Donations and Gift		
	Trust Fund (5.76)		
108400	Political signs; amends §		
CES CITE	4907 of Ord. 106350, Building Code; adds §		
	14-A and amends § 21 of		
	Ord. 106653, fair campaign		
COMPA	practices (2.04)		
108419	Consolidated Local		
1400	Improvement District No. 1		
ing to	Bond Redemption Fund		
100.101	(5.76)		
108421	Repeals §§ 1, 8, 8-A and		
111e.	8-B; adds §§ 1-A-1-D and 8-A-8-J to Ord. 92987,		
	food-service		
	establishments (Repealed		
	by 109910)		
108434	Amends § 1 of Ord.		
	106958, Refund account;		
	repeals Ord. 106658 (5.76)		
108443	Reimbursement of moving		
	expenses for certain employees (Repealed by		
	117242)		
108444	Sale of Metro bus passes		
	by City Treasurer		
	(Repealed by 116682)		
108453	Amends §§ 27.3 and 28.42		
	of Ord. 86300, zoning		
	(Repealed by 109737, 110381)		
108472	Amends § 12A.02.130 of		
1007/2	Ord. 102843, Criminal		
	Code (12A.04)		
108481	Amends §§ 49.3 and 49.5		
	of Ord. 97330,		
	compensation of officers		
100100	and employees (4.20)		
108482	Electrical Code; repeals		
	Ords. 105886, 106537 and 106809 (Repealed by		
	114181)		
108498	Amends § 602 of and adds		
	§ 604 to Ord. 106360,		
	noise control (25.08)		
108500	Energy Code (Repealed by		
100500	112500)		
108508	Building Code; repeals		
	Ords. 106350, 106779, 106808, 106914, 107015,		
	107380, 107866 and		
	108011 (Repealed by		
	111200)		
	A CARILLE		

eference only.

I amending graphics, ccuracy of

For current Smc, combattee City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te file
I	108517	Designates New Pacific
		Apartments as a landmark
Ļ		(25.32)
	108518	Designates Hull Building
1	100510	as a landmark (25.32)
	108519	Designates Seattle Hebrew
		Academy as a landmark
ŀ	100540	(25.32)
	108549	Amends § 1 of Ord. 81301, Cumulative Reserve Fund
1	es crea	(Repealed by 116497)
1	108552	Adds §§ 211.5 and 504 to
	108332	Ord. 106360, noise control
ı	comple	(25.08)
ŀ	108563	Sister cities designated
	100303	(1.24)
ď	108567	Adds §§ 12A.04.030 and
1	11100	12A.08.310; amends §§
		12A.04.050 and
	<b>S</b> 112.	12A.04.140 of Ord.
	11100	102843, Criminal Code
		(12A.06, 12A.08)
ľ	108608	Amends § 1 of Ord. 72495,
		admissions tax (5.40)
Ī	108612	Amends § 5 of Ord. 78444,
		employee's retirement
		system (4.36)
	108614	Establishes Vanpool
		Operating Fund in
		Engineering Department
L		Operating Fund (5.76)
	108615	Restricted parking zone in
		Montlake Neighborhood
ŀ	108632	(11.73) Adds § 19.112 to Ord.
	108032	106107, Fire Code
		(Repealed by 108678)
F	108635	Amends § 11.56.020 of
	100033	Ord. 108200, Traffic Code
		(11.56)
t	108639	Amends § 4 of Ord. 84390,
		sewerage rates and charges
		(21.28)
ľ	108647	Amends § 4.23 of Ord.
		106319 and § 904 of Ord.
		106360, housing code and
		noise (Repealed by
L		113545)
	108648	Amends § 14.1 of Ord.
		48022, License Code
L		(6.02)
	108649	Hearing Examiner fees
ŀ	100650	(3.02)
	108650	Amends § 8 of Ord.
		102228, Administrative
ļ	100651	Code (3.02)
	108651	Amends § 2 of Ord. 82583,
		sewer permit fees (Repealed by 116421)
L		(Repealed by 110421)

eference only.

A amending
graphics,
accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

Sihar .	. C110
a - made	ate tille
108652	Amends § 5 of Ord.
	105636, subdivisions
a come la	(Repealed by 110570)
108653	Amends §§ 27.1 and 28.4
	of Ord. 86300, zoning
	(Repealed by 110381)
108654	Amends § 1 of Ord. 99492,
	Board of Public Works
	(20.48)
108656	Data processing operating
CES OF	fund; repeals Ord. 107840
	(5.76)
108657	Judgment/Claims Fund;
COMPA	repeals Ords. 16117 and
	100501 (5.24, 5.76)
108658	Repeals and replaces § 2 of
male	Ord. 107055, Seattle
1880	Center rental agreements
	(Repealed by 110896)
108659	Amends § 5 of Ord. 62662,
222-	utilities business tax (5.48)
108675	Amends §§ 9, 16 and 20 of
	Ord. 80028, § 3 of Ord.
	85679 and §§ 3, 4, 7 and
	30 of Ord. 94465; repeals §
	29 of Ord. 94465; meat,
	rabbit, poultry and fish sale
	and distribution (Repealed
	by 110052, 110895)
108678	Fire Code; repeals Ord.
	106107, 106467, 107017
	and 107829 (Repealed by
	111001)
108707	Amends § 10 of Ord.
	72630, business and
	occupation tax (5.44)
108709	Amends §§ 27.3 and 28.42
	of Ord. 86300, zoning
	(Repealed by 109737,
	110381)
108729	Permit fees; repeals Ord.
	108157 (Repealed by
	109506)
108730	Designates Jolly Roger
	Roadhouse as a landmark
	(25.32)
108731	Designates Anhalt
	Apartments as a landmark
	(25.32)
108732	Designates 23rd Avenue
	Rowhouse Group (812-828
	23rd Avenue) as a
	landmark (Repealed by
	117570)
108733	Designates Ellsworth
100/33	Storey Cottages Group as a
	landmark (25.32)
108734	
100/34	Designates Brehm Brothers Houses as a landmark
	(25.32)

eference only.

and amending graphics, accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	te ille
108735	Office of Policy
0.0	Evaluation; repeals Ords.
- 103	104067 and 107879
	(Repealed by 117408)
108736	(Repealed by 117408)
108/36	Amends § 4.21 of Ord.
	106319, Housing Code
	(Repealed by 113545)
108762	Consulting services (3.114)
108807	Amends §§ 1, 2 and 3 of
	Ord. 108614, Vanpool
ces er en	Fund (5.76)
108809	Amends § 11.40.140 of
10000)	Ord. 108200, pedestrians
COMPLE	
100014	(11.40)
108814	Adds §§ 12A.20.010 and
- 4 40 C	12A.12.160 and amends §§
	12A.12.150, 12A.16.020,
	12A.17.140, 12A.17.160,
	12A.20.020, 12A.20.110,
	12A.46.100 and
22200	12A.46.120 of Ord.
	102843, Criminal Code
	(12A.10, 12A.12, 12A.14,
	12A.16)
108836	Amends § 20B of Ord.
108830	ŭ .
	107501, State
	Environmental Policy Act
	guidelines (Repealed by
	111866)
108837	Amends § 7 of Ord. 65877,
	water supply system;
	repeals §§ 7-B and 14 of
	Ord. 65877 (21.04)
108841	Amends § 1 of Ord.
100011	107828, exemptions from
	civil service (Repealed by
100046	109004)
108846	Amends § 18 of Ord.
	90047, street use
	regulations (15.10)
108850	Amends § 7-A-2 of Ord.
	65877, water supply
	system (21.04)
108860	Repeals § 21 of Ord.
	105735 (Repealed by
	111886)
108861	Amends § 1 of Ord.
100001	108649, Hearing Examiner
	, ,
100065	fees (3.02)
108865	Repeals Ord. 72585
	(Repealer)
108867	Amends § 9 of Ord.
	108814, Criminal Code
	(12A.12)
108882	Code of Ethics, repeals
108882	Code of Ethics; repeals
108882	Ords. 100435, 100648,
	Ords. 100435, 100648, 100649 and 106471 (4.16)
108882	Ords. 100435, 100648,

reference only.

All amending
graphics,
ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
108887	Amends § 11.55.100 of
	Ord. 108200, Traffic Code
100 C OF 101	(11.55)
108908	Adds § 12A.04.090 to Ord.
	102843, Criminal Code
100000	(12A.06)
108909	Amends § 95 of Ord.
	48022, dog and cat licenses (Repealed by 110230)
108934	New license code (6.202,
108334	6.214, 6.260)
108935	Amends § 12 of Ord.
- 400 40   6	106615, Park Code (18.12)
108936	Seattle Center Advisory
	Commission; repeals Ords.
140 (	91885 and 105411 (17.08)
108967	Amends § 2 of Ord.
	106166, § 2 of Ord.
0210	108482 and § 1 of Ord.
<b>4116</b> .	108508, building codes
880	(Repealed by 111200,
100004	114181)
108984	Amends § 14.2 of Ord.
	86300, zoning (Repealed
108992	by 117570) Amends §§ 1, 2 and 3 of
108992	Ord. 65482, sidewalk
	construction (15.70)
108995	Adds §§ 12A.04.195,
108993	12A.04.196, 12A.04.197
	and 12A.04.198 to Ord.
	102843, Criminal Code
	(12A.06)
108996	Amends § 4 of Ord.
	100708, unit pricing (7.12)
108998	Collection of property
	taxes and abatement of
	liens (5.04)
109004	Exemptions from civil
	service; repeals Ords.
	107828 and 108074
	(Repealed by 110229)
109019	Police firearm discharge
100000	(3.28)
109020	Spay and neuter clinic
100021	(Repealed by 117218)
109021	Amends § 4 of Ord.
	104472, utility credits for the elderly (21.76)
109022	Garbage collection rates
107022	and charges (21.40)
109031	Amends § 11.30.180 of
107031	Ord. 108200, vehicle
	impoundment fees (11.30)
109033	Plumbing Code; repeals
10/033	Ord. 106721 (22.500,
	22.504, 22.506)
109037	Adds § 12A.08.290 to Ord.
10/03/	102843, Criminal Code
	(12A.08)
	COLLEGE

est eference only.

est amending
graphics,
curacy of

June 2009 co Text provide

See ordinant sections for and tables a this source

Cipal				
	a mada	te file		
	109039	Amends § 23.3 of Ord.		
		86300, zoning (Repealed		
	- C 101	by 117570)		
•	109050	Amends § 3 of Ord.		
		108205, open housing		
		(14.08)		
	109078	Amends § 12 of Ord.		
	100000	106615, Park Code (18.12)		
	109080	Adds Ch. 7 to Ord.		
	CE CITE	108934, tow trucks; repeals		
7		§§ 8 of Ord. 106025 and 17, 18, 19 and 20 of Ord.		
	10	107158 and Ords. 90316,		
	comple	90535, 91141, 92986,		
,	Cosses	97331, 99869, 101488,		
	4 4 5 00	102631 and 104849 (6.214)		
	109081	License fees (6.204)		
•	109084	Adds § 28.21 and amends		
		§§ 6.1, 6.3, 23.1 and 28.4		
	2112	of Ord. 86300, zoning		
	7110.	(Repealed by 117570)		
	109091	Amends §§ 4.1 and 4.3 of		
		Ord. 84390, sewer utility		
		(21.28)		
	109099	Adds §§ 226.5 and 407 and		
		amends §§ 215 and 232		
		and Ch. 4 of Ord. 106360,		
	109112	noise control (25.08)		
	109112	Affirmative action plan; amends §§ 8 and 9 of Ord.		
		107790 and § 8 of Ord.		
		107791; repeals Ord.		
		101548 (4.04, 4.08, 4.80)		
	109113	Women and minority		
		business utilization (20.46)		
	109114	Office of Women's Rights		
		and Seattle Women's		
		Commission; repeals Ords.		
		102182, 104907 and 107128 (3.14)		
	109115	Amends §§ 2 and 3 of Ord.		
	107113	97971, Office of Human		
		Rights (3.14)		
	109116	Seattle Fair Employment		
		Practices Ordinance;		
		repeals Ords. 102562,		
		103422, 104095, 105423,		
		106475 and 107759 (14.04)		
	109118	Amends §§ 1 and 3 of Ord.		
		107634, lease authority of Director of Administrative		
		Services; repeals Ords.		
		105093 and 91981 (3.18)		
	109120	Amends § 1 of Ord.		
		102735, bonds of City		
		officers (3.90)		
	109124	Department of		
		Construction and Land		
		Use; repeals Ord. 107786		
		(3.06, 5.76)		

eference only.

amending

graphics,
accuracy of

For current SMC, Clerk the Office of the City Clerk

Seattle Municipal Code date ference only. Amends § 1-A of Ord. 109125 89021, §§ 2.020, 2.030, Text provided for hi 3.010 and 3.030 of Ord. 89715, §§ 4 and 5 of Ord. 90138, § 8 of Ord. 98852, §§ 6, 7 and 8 of Ord. amending 100475, §§ 1 and 2 of Ord. 101941, §§ 5 and 8 of Ord. 105462, §§ 4 and 18 of Ord. 105636, §§ 17 and 18 See ordinan of Ord. 105735, § 14 of Ord. 106166, § 3.43 of sections for comp Ord. 106319, § 1.03 of Ord. 106348, § 3.08 of Ord 107323 § 1 of Ord. 107679, §§ 3.2, 4.1, 4.2, and tables and 5.1 and 5.2 of Ord. 107707, §§ 3, 25 and 30 of Ord. 108080, § 201 of Ord. this source file. 108482, § 202 of Ord. 108500, §§ 201, 403 and 405 of Ord. 108508 and § 1 of Ord. 108729, reorganization project (7.16, 22.902, 22.904,25.12, 25.16, 25.20, 25.24) 109126 Amends §§ 3.04, 3.05, 3.20, 11.1, 13C.51, 13C.52, 13C.54, 13C.6, 13C.7, 13C.71, 13C.72, 13C.8, 17.22, 20.2, 21A.13, 21A.14, 21A.17, 21A.50, 21A.51, 21A.52, 21A.54, 21A.55, 21A.56, 21A.57, 21A.58, 21A.62, 21A.69, 21A.104, 21A.151, 21A.154, 23.21, 23.31, 23.41, 23.43, 23.44, 23.5, 23.63, 23.7, 24.41, 24.47, 24.83, 24.84, 24.85, 24.86, 24.912, 24.916, 24.922, 26.2, 26.3, 27.11, 27.12, 27.2, 27.41, 27.84, 28.2, 28.43 and 28.54 and repeals § 27.4 of Ord. 86300, zoning (Repealed by 117570) 109128 Amends § 2 of Ord. 105934, § 2 of Ord. 107007, §§ 1 and 2 of Ord. 107252, § 2 of Ord. 107298, § 2 of Ord. 107964, § 2 of Ord. 108167 and § 2 of Ord.

> For current SMC, contact the Office of the City Clerk

108553, Director of Administrative Services

(3.18)

June 2009 co

Text provide

See ordinant sections for and tables a this source

	C:10
4 - mda	te tille
109129	Department of
	Administrative Services;
a com hi	amends §§ 1, 2 and 3 of
d tor	Ord. 100174 and § 1 of
	Ord. 106965; repeals Ord.
	99554 (3.18, 5.76)
109130	Garbage and rubbish
	collection charges; repeals
	Ord. 90379 (21.40)
109131	Amends § 4 of Ord. 96003,
	garbage collection (21.36)
109132	Amends § 4 of Ord.
ammle	104472, utility credits for
COMPLE	the elderly (21.76)
109133	Amends § 1 of Ord.
1 40 CC	106058, Refund Account
ma to	(5.76)
109155	Planning Commission;
	amends §§ 11.1, 16.3, 16.5,
sile.	17.22, 24.4, 24.46, 24.7,
8880	24.83, 27.2, 27.41, 27.6,
	27.7 and 28.2 of Ord.
	86300, §§ 2 and 7 of Ord.
	98852, §§ 13.08, 18, 19
	and 22 of Ord. 105636;
	repeals §§ 24.45 and 27.4
	of Ord. 86300, §§ 3.04, 20
	and 21 of Ord. 105636 and
	Ords. 99443 and 106302,
	zoning (3.64)
109157	Amends § 3 of Ord.
	104542, relocation
	assistance for disabled
	persons (20.84)
109189	Adds Ch. 40 to Ord.
	108934, adult
	entertainment studios
	(Repealed by 114225)
109190	Adds § 12A.20.070 to Ord.
	102843, Criminal Code
	(12A.16)
109191	Amends §§ 90-3 and 262
	of Ord. 48022; repeals §
	105 of Ord. 48022; License
	Code (6.14, 6.38)
109208	Renumbers §§ 5 to 15, 6 to
	16 and 7 to 11, adds new
	§§ 5, 6, 7, 8, 9, 10 and 12,
	and amends § 4 of Ord.
	and amends § 4 of Ord. 104839, open housing
	and amends § 4 of Ord. 104839, open housing (14.08)
109218	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and
109218	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and conditions of use; repeals
109218	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and conditions of use; repeals Ords. 106416, 106828,
109218	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and conditions of use; repeals Ords. 106416, 106828, 107660 and 107661
	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and conditions of use; repeals Ords. 106416, 106828, 107660 and 107661 (Repealed by 110733)
109218	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and conditions of use; repeals Ords. 106416, 106828, 107660 and 107661 (Repealed by 110733) Amends §§ 4.15 and 4.31
	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and conditions of use; repeals Ords. 106416, 106828, 107660 and 107661 (Repealed by 110733) Amends §§ 4.15 and 4.31 of Ord. 106319, Housing
	and amends § 4 of Ord. 104839, open housing (14.08) Electricity rates, terms and conditions of use; repeals Ords. 106416, 106828, 107660 and 107661 (Repealed by 110733) Amends §§ 4.15 and 4.31

reference only.

a lamending

graphics,
curacy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

	ado	te file
	109220	Housing preservation
	10,220	(Repealed by 112342)
	109227	Adds § 75 and amends §§
	a sor m	7, 8, 10 and 72 of Ord.
÷	C LO	87983, Harbor Code
		(16.12, 16.20, 16.32,
		16.36)
	109228	Amends § 20 of Ord.
		87983, Harbor Code
	oc crea	(16.44)
V	109229	Liability for experts used
		by Fire Department (3.16)
	109237	Amends § 29 of Ord.
9	Comple	108333, police
	100250	investigations (14.12)
	109250	Amends § 68(e) of Ord.
4	iller co	87983, Harbor Code (16.60)
	109261	` '
	109201	Restricted parking zone in vicinity of Providence
	1116.	Medical Center (11.73)
	109266	Amends §§ 11.11, 24.46
	10,200	and 27.41 of Ord. 86300, §
		2 of Ord. 98852 and §§
		13.08 and 22 of Ord.
		105636, zoning and
		subdivisions (Repealed by
		117570)
	109267	Establishes repayment fund
		subaccount within Housing
		and Community
		Development Revenue
	109271	Sharing Fund (5.76) Adds § 50 to Ord. 90047
	109271	and Ch. 30 to Ord. 108934;
		amends § 6 of Ord. 90047;
		repeals §§ 152, 153, 190,
		191, 192-1, 193, 194,
		194-1, 194-3, 194-4, 194-5,
		195, 195-A, 196, 196-A,
		369, 370, 371, 372, 373,
		374 and 375 of Ord. 48022,
		§ 3 of Ord. 75087, Ords.
		75207, 80950, 90074,
		90480, 90729, 91943,
		94568, 96360, 101076 and
		102404, §§ 114, 115, 116, 117, 118, 119, 120, 121,
		122, 123, 124, 125 and 126
		of Ord. 102636, Ords.
		103514, 104620 and
		105189, § 9 of Ord.
		106063, Ord. 106077 and
		§§ 11 and 30 of Ord.
		107157, peddlers (6.260,
		15.02, 15.17)
	109272	Adds § 6 to Ord. 109081,
		adult entertainment license
		fees (Repealed by 114225)

eference only. amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

a anda	te file
109274	Amends § 6.4 of Ord.
	86300, zoning (Repealed
- 103	by 110381)
109279	Adds §§ 40.025 and 40.045
109279	
	to Ord. 108934; repeals §
	40.090 of Ord. 108934,
	business licenses (Repealed
	by 114225)
109280	Floating home moorages;
- of orea	repeals Ord. 107012
ces er er	(Repealed by 111526)
109299	Amends § 2 of Ord.
10,2,,	107785, Board of Public
COMPLE	Works (Repealed by
Comme	116007)
100212	
109313	Repeals and replaces § 1
Ma co	and amends § 2 of Ord.
	106214, Energy Office
	(Repealed by 110254)
109317	Designates
7770.	Parsons/Gerrard residence
	as historic landmark
	(25.32)
109318	Designates
107510	McFee/Klockzien
	residence as historic
100210	landmark (25.32)
109319	Designates Parsons
	Memorial Park as historic
	landmark (25.32)
109320	Designates East
	Republican Street stairway
	as historic landmark
	(25.32)
109339	Amends § 12 of Ord.
	109218, electricity service
	connections (Repealed by
	111615)
109348	Adds Chs. 5 and 6 to Ord.
109346	
	108934; repeals §§
	400-413, 500-508, 510-514
	and 520-526 of Ord. 48022
	and Ords. 93789, 107095,
	107270 and 108196;
	107270 and 108196;
	107270 and 108196; vehicles for hire (Repealed
109358	107270 and 108196; vehicles for hire (Repealed by 118341)
109358	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord.
109358	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil
	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08)
109358 109376	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08) Amends §§ 2 and 5 of Ord.
	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08) Amends §§ 2 and 5 of Ord. 95466, group term life
	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08) Amends §§ 2 and 5 of Ord. 95466, group term life insurance program (4.48)
	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08) Amends §§ 2 and 5 of Ord. 95466, group term life
109376	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08) Amends §§ 2 and 5 of Ord. 95466, group term life insurance program (4.48) Amends § 1 of Ord.
109376	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08) Amends §§ 2 and 5 of Ord. 95466, group term life insurance program (4.48) Amends § 1 of Ord. 107055, rental of Seattle
109376	107270 and 108196; vehicles for hire (Repealed by 118341)  Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08)  Amends §§ 2 and 5 of Ord. 95466, group term life insurance program (4.48)  Amends § 1 of Ord. 107055, rental of Seattle Center buildings (17.16)
109376	107270 and 108196; vehicles for hire (Repealed by 118341) Amends §§ 4 and 6 of Ord. 107791, public safety civil service regulations (4.08) Amends §§ 2 and 5 of Ord. 95466, group term life insurance program (4.48) Amends § 1 of Ord. 107055, rental of Seattle

eference only.

detail amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipal col					
	· · · · · · · · · · · · · · · · · · ·	te file			
	109398	Amends §§ 33 and 34 of	Sol		
7		Ord. 65877 and § 4 of Ord.	316		
	a sor hi	104472; repeals Ord. 106959; water rates (21.04,			
•	0 101	21.76)			
	109416	Amends §§ 102, 103, 202,			
		203, 207, 218, 219, 304,			
		402, 403, 404, 411, 412, 414, 415, 418, 420, 422,	lai		
	orea	425, Tables 4-6, 4-9 and			
1	ces erea	4-10, §§ 601, 602 and 603	7502		
		of Ord. 108500, Energy	270		
	comple	Code (Repealed by 112500)			
•	109432	Amends § 3 of Ord.	CCI		
	140.00	109220, housing			
Δ	nd to co	preservation (Repealed by			
٩	100422	112342)			
	109433	Amends §§ 12A.02.030 and 12A.04.140 of Ord.			
	1116.	102843, Criminal Code			
		(12A.04)			
	109438	Master use permits			
	109440	(Repealed by 110381) Amends § 23.28 of Ord.			
	10,7110	86300, zoning (Repealed			
		by 117570)			
	109461	Adopts final order relating			
		to PURPA Section 113, public utility regulatory			
		policies (Not codified)			
	109475	Amends §§ 11.16.420,			
		11.56.020, 11.56.130,			
		11.56.320, 11.58.005, 11.80.060, 11.84.260 and			
		11.84.300 of Ord. 108200,			
		traffic (11.16, 11.56, 11.58,			
	109476	11.80, 11.84) Adds Ch. 11.31 and §§			
	109470	11.20.020, 11.55.340 and			
		11.59.090 to Ord. 108200;			
		amends and renumbers §§			
		11.32.060 to 11.31.030, 11.32.120 to 11.31.100,			
		11.32.120 to 11.31.100, 11.32.180 to 11.31.110 and			
		11.32.200 to 11.31.080 of			
		Ord. 108200; amends §§			
		11.20.010, 11.23.380, 11.23.400, 11.30.360,			
		11.32.020, 11.32.140,			
		11.34.020, 11.54.150,			
		11.55.320, 11.58.005,			
		11.58.008, 11.60.583, 11.60.586, 11.60.589,			
		11.60.592, 11.60.665,			
		11.60.670, 11.60.675,			
		11.60.680, 11.60.685 and 11.66.200 of Ord. 108200,			
		traffic (11.20, 11.23, 11.30,			
		11.31, 11.32, 11.34, 11.54,	ct		
	ACM	11.55, 11.58, 11.59, 11.60, 11.66)			
		11.00)			

eference only.

od,

od,

amending

graphics,

accuracy of

For current Simple City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

Clhar	C:10
a - mada	te tille
109480	Amends § 3 of Ord.
103	107788 and § 3 of Ord. 107789, engineering and
a sor m	water departments (3.12,
CI TO	3.32)
109489	Amends § 1 of Ord.
	106958, P-patch program
109490	garden plot fees (3.20) Adds Exhibit B-1981 to
109490	Ord. 107055, rental of
ces crea	Seattle Center buildings
	(Repealed by 110896,
100/01	112573)
109491	Amends § 1 of Ord. 99492, administrative charges for
4 4 0 00	public works contracts
and to co	(20.48)
109492	Amends § 1 of Ord.
0210	105149, fees for medical
111E.	laboratory tests (Repealed by 110271)
109493	Amends §§ 4 and 5 of Ord.
	103095, corrosion
	prevention (21.08)
109494	Amends § 16 of Ord.
	80028, milk and milk products and § 3 of Ord.
	94465, meat distribution
	(Repealed by 110052)
109495	Amends § 6 of Ord.
	109033, Plumbing Code (22.504)
109496	Amends § 1 of Ord. 82430,
107.70	fees for health certificates
	for travelers (Repealed by
109497	110903)
109497	Amends § 3 of Ord. 107099, reports on
	condition of water and
	sewer systems (10.25)
109498	Amends § 1 of Ord. 99870,
	police department records fees (3.28)
109499	Amends § 2.100 of Ord.
107.77	106024, Revenue Code
	(5.32)
109500	Amends § 3 of Ord. 90145,
	tobacco vending machine license (Repealed by
	115531)
109501	Amends § 13 of Ord.
	84297, refrigeration license
100502	fees (Repealed by 118398)
109502	Amends §§ 23, 95, 97, 131, 136, 147, 172, 173, 206,
	263, 330.2, 331 and 331.1
	of Ord. 48022, business
	license fees (6.20, 6.36,
	6.42, 6.48)

reference only.

d amending
praphics,
curacy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te file
	109503	Amends § 3 of Ord.
		103667, Washington Park
	a combi	(18.16)
	109504	Amends § 4 of Ord. 84390,
0		Southwest Suburban Sewer
		District rates (21.28)
	109505	Amends § 21A.59 of Ord.
		86300, zoning fees
		(Repealed by 113466)
	109506	Permit fees; repeals Ord.
	ces es	108729 (Repealed by
		109720)
	109507	Amends § 41 of Ord.
	COMPA	90047, excavations (15.44)
	109508	Fees for North District
	140 00	Comprehensive Care Clinic
	ma to co	(Repealed by 110272)
•	109511	Amends §§ 24.43, 27.1,
		27.8 and 28.4 of Ord.
		86300 and §§ 5 and 16 of
	Fire	Ord. 106014, zoning and
		subdivisions (Repealed by
		117570)
	109514	Amends § 5 of Ord. 78444,
		employees' retirement
		system (4.36)
	109515	Amends § 18 of Ord.
		105735, environmental
		review fees (Repealed by
		111866)
	109517	Amends § 4.3 of Ord.
		84390, sewer rates (21.28)
	109522	Establishes Seattle-King
		County Department of
		Public Health; repeals
		Ords. 79432 and 98327
		(5.76)
	109523	Amends §§ 2, 3, 12, 13 and
		20 of Ord. 102623,
		business and occupation
		tax (5.44)
	109524	Amends § 1 of Ord.
		108649, Hearing Examiner
		fees (3.02)
	109545	Adds § 51 to Ord. 90047;
		amends § 11.72.320 of
		Ord. 108200, temporary
		parking on planting strips
		(11.72, 15.07)
	109555	Adds § 76 to and amends
		§§ 3 and 22 of Ord. 87983,
		Harbor Code (16.12, 16.28)
,		

eference only.

l amending

graphics,

ccuracy of

Seattle Municipal Code

June 2009 co 109560 up date

Code at 1.04.01
25.04.5 amends renumb
3.22.14
3.20.10
3.14.52

See ordinant sections for and tables a this source

a anda	te file
109560	Code adoption; adds §
	1.04.010; adds to §§
101	25.04.500 and 25.04.550;
A FOT III	amends § 12A.06.030;
CI TO	renumbers §§ 3.22.130 and
	3.22.140 to 3.20.090 and
	3.20.100; repeals §§
	3.14.520, 3.14.550,
	3.20.150, 3.20.160,
- 0002	3.20.170, 3.20.180,
CES CITES	6.22.010, 6.78.010,
	6.78.070, 6.78.090 and
-10	12A.06.030 (1.01, 1.04,
COMBIE	3.20, 12A.06, 25.04)
109564	
109504	Amends § 46 of Ord.
- 4 to C	97330, compensation and
ma to	working conditions for
	City officers and
	employees (4.20)
109576	Amends § 5 of Ord.
Tare	105296; repeals § 1 of Ord.
	105296 and Ord. 98511,
	data processing (Repealed
	by 117408)
109577	Amends §§ 2, 3 and 4 of
	Ord. 108329, data
	processing (Repealed by
	118397)
109586	Designates Brace/Moriarty
	residence as historic
	landmark (25.32)
109621	Amends §§ 5, 7, 10, 11, 12
	and 14 of Ord. 109218,
	electricity rates (Repealed
	by 110723, 111615)
109622	Amends § 7.140 of Ord.
103022	108934, towing operators
	(6.214)
109630	Amends § 3 of Ord.
107030	109280, floating homes
	(7.20)
109631	Amends §§ 2, 3, 4 and 5
10,0031	and repeals § 1 of Ord.
	108656, data processing
100/27	funds (5.76)
109635	Adds § 6.245 and amends
	§§ 6.020, 6.150 and 6.300
	of Ord. 108934, taxi
1	licensing (Repealed by
	118341)

reference only.

Id amending

graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

CI has	
a mada	te file
109648	Adds Ch. 4 (§§ 4.020,
	4.024, 4.030, 4.140, 4.200,
- 0 hi	4.210, 4.220, 4.240 and
d for III	4.300) to license code,
	for-hire driver licenses;
	repeals §§ 103, 104, 105
	and 107-114 of license
	code and Ords. 53281,
	60694, 66558, 72318,
- oc crea	73763, 74223, 75087,
ces er	78349, 79402, 79606,
	80689, 80738, 83281,
	88789, 89770, 95008,
COMPA	97841, 100014, 100478,
	102636, 104649, 106063,
140 0	106189, 107157 (Repealed
and to co	by 118341)
109651	Amends § 1.090 of Ord.
	108934, New License Code
2110	(6.202)
109657	Amends § 4904 of building
	code supplement, sign
	permits (Repealed by
100.500	109861)
109673	Amends §§ 6.170 and
	6.220 of Ord. 108934,
	taxicabs (Repealed by
100674	118341)
109674	Adds §§ 12A.06.025,
	12A.06.120, 12A.06.130,
	12A.06.140, 12A.06.150,
	12A.08.105, 12A.10.100,
	12A.16.050 and
	12A.16.060 and amends §§ 12A.04.020-12A.04.060,
	12A.04.020-12A.04.000, 12A.06.080, 12A.10.070,
	12A.12.010, 12A.14.080,
	12A.12.010, 12A.14.080, 12A.14.090-12A.14.120,
	12A.14.090-12A.14.120, 12A.16.010, 12A.16.040,
	12A.16.010, 12A.16.040, 12A.28.050, 12A.28.070
	and 12A.52.020, criminal
	code (12A.04, 12A.06,
	code (12A.04, 12A.00,

12A.08, 12A.10, 12A.12, 12A.14, 12A.16, 12A.28)

Comprehensive residential weatherization program

Amends § 24.38.070, subsection A of § 24.46.110, subsections A and C of § 24.48.100, and § 24.50.100, zoning (Repealed by 117570)

Amends §§ 24.62.080, 24.64.030 and 24.66.150, zoning (Repealed by

Adds §§ 11.20.040, 11.22.200, 11.22.220 and

(21.52)

117570)

eference only.

I amending

graphics,

For cur license plates (11.20, 11.22)

For cur license plates (11.20, 11.22)

The Office of the City Clerk

109675

109694

109696

109698

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
109702	Adds §§ 4.72.080 and
Che o-5	4.72.090 and amends §§
- 101	4.72.010, 4.72.020 and
	4.72.070, employee travel
C TOT	expenses (4.72)
109704	Rezone, Map 37W, to
109/04	
100700	correct error (Special)
109709	Amends §§ 22.202.040,
- 2000	22.202.060, subsection E
age cited	of § 22.202.070, §§
	22.208.050 and
	22.208.060, housing code
	(Repealed by 113545)
109715	Amends § 3312 of the
	building code supplement,
	exit signs (22.108)
109716	Amends §§ 22.300.020,
Hara o	electrical code (Repealed
	by 114181)
109717	Adds to § 24.16.050,
103/1/	
	zoning (Repealed by
100=0	110381)
109720	Permit fees; repeals Ord.
	109506 (Repealed by
	110264)
109721	Rezone, Map 37W
	(Special)
109729	Local improvement
	districts; repeals Ord.
	53493 (20.04)
109730	Designates Immanuel
107/30	Lutheran Church as a
	landmark (25.32)
109731	
109/31	Designates Temple de Hirsch Sinai as a landmark
	(25.32)
109732	Designates the Church of
	the Blessed Sacrament
	rectory and grounds as a
	landmark (25.32)
109733	Designates the Hillcrest
	Apartment Building as a
	landmark (25.32)
109734	Designates the
	Hainsworth/Gordon House
	and grounds as a landmark
	(25.32)
100727	,
109737	Amends Ord. 86300,
	zoning (Repealed by
	117570)
	Designates Salmon Bay
109738	
109738	Burlington Northern
109738	
109738	Burlington Northern Bridge No. 4 as a landmark (25.32)
109738	Bridge No. 4 as a landmark

reference only.

dlamending
graphics,
curacy of

June 2009 co

Text provided for

See ordinan 109754 Com sections for com and tables and t this source file.

Repeals and replaces Ch.
15.62, street vacation
procedures; repeals Ords.
96020, 98929, 102342,
103444, 106019, 106750
and 107725 and Res.
21099 (15.62)

Amends § 2-A of Ord.
82583, sewer connection
fees (21.24)

Amends § 15.04.010,

Amends §§ 15.04.010, 15.04.020, 15.04.030, 15.04.070, 15.04.080, 15.06.050, 15.16.010-15.16.080, 15.74.010-15.74.030, 22.100.010, 22.106.020, 22.804.150, 22.804.230, 24.04.130, 24.04.160, 24.04.190, 24.04.200, 24.80.070, 24.80.070, 24.80.080-24.80.100, 24.80.110 and

> 24.98.070-24.98.090; repeals § 15.74.040; Department of

Construction and Land Use

79.201, 80.102, 80.113 and 82.102, fire code (Repealed

by 111001)

(15.04, 15.06, 15.16, 22.106) 109757 Retired police officer commission; repeals § 2 of Ord. 16374 and Ords. 100759 and 108124 (3.28) 109758 Reserve police officers (3.28)109759 Special police officers; repeals Ords. 66591, 66888, 67137, 71537, 82230, 92383, 94101, 95246, 99844, 103072 and 104317 (3.28) 109760 Amends § 27.1 of Ord. 86300, zoning (Repealed by 110381) 109763 Adds Ch. 11 to Ord. 108934 and § 7 to Ord. 109081; repeals §§ 376, 377, 378, 379, 380, 381 and 382 of Ord. 48022 and § 1 of Ord. 104214, rental housing agencies (6.204, 6.222) 109765 Adds §§ 80.114 and 80.115 to Ord. 108678; amends §§ 3.101, 4, 4.101, 10.307, 11.203, 25.101, 25.104,

ference only.

amending

graphics,

ccuracy of

For current SMC, Constitution of the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

a and	ite file
109766	Adds § 80.116 to Ord.
	108678, fire code
- 0 hi	(Repealed by 111001)
109770	Amends §§
C Z	24.14.030-24.14.070 and
	24.44.030, zoning
	(Repealed by 117570)
109771	Rezone, Map 80E (Special)
109781	Rezone, Map 32E (Special)
109793	Rezone, Map 16E (Special)
109797	Rezone, Map 66E (Special)
109801	Adds subsection K to §
COMPLE	24.40.020, subsection C to
COMPA	§ 24.40.050 and subsection
	E to § 24.42.020; amends
- 4 40 C	subsection A of §
ma to	24.16.050, subsection A of
	§ 24.40.010, subsection A
	of § 24.42.010 and
\$116	subsection D of §
7770	24.42.050; repeals
	subsection C of §
	24.40.070; zoning
	(Repealed by 117570)
109808	Amends §§ 20.44.010,
	20.44.020, 20.44.030,
	20.44.040, 20.44.050 and
	20.44.100, fair
	employment practices
	(20.44)
109810	Adds language to §
109810	24.08.050, and adds
	subsection J to §
	Ü
	24.44.080, subsection F to
	§ 24.46.070, subsection 1
	to § 24.48.060 and
	subsection F to §
	24.54.030; amends
	subsection C of §
	24.42.050 and §§
	24.44.090, 24.46.100,
	24.48.070 and 24.50.080,
	zoning (Repealed by
	117570)
109814	Authorizes Engineering
	Department to retain
	certain consultants
	(Repealed by 110986)
109818	Used goods dealer licenses;
	adds § 7 to Ord. 109081;
	repeals Chs. 6.04, 6.32,
	6.44, 6.50 and 6.58 (6.204,
	6.288)
109823	Amends §§ 20.88.020,
	20.88.040 and 20.88.050
	and § 2 of Ord. 105934 and
	§ 2 of Ord. 107007, real
1	property leases by City
	(3.18)

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	:1puz	to file
- [	109833	Adds & 6A and amonds & 8
	109833	Adds § 6A and amends §§
		3, 6, 9 and 12 of Ord.
	1 Cor MI	109438, master use permits
	U IOI	(Repealed by 110381)
	109836	Adds subsection B to §
		24.62.100; amends
		subsection 16 of §
		24.08.130 and subsection
		A of § 24.62.080; repeals
	oc crea	subsection D of §
V	CES OF	24.62.150, zoning
		(Repealed by 117570)
	109837	Adds to § 24.08.200 and
	COMPA	amends subsection B of §
		24.56.020, zoning
	14000	(Repealed by 117570)
<u></u>	109841	Amends § 1.140 of Ord.
•	,550	108934, New License Code
		(6.202)
	109844	Adds subsection G to §
	1110.	24.16.070 and subsection B
		to § 24.38.060 and adds to
		§§ 24.08.090 24.30.080,
		24.30.100, 24.32.090,
		24.34.070, 24.40.050,
		24.40.070, 24.42.050,
		24.42.070, 24.44.090,
		24.44.110, 24.46.080,
		24.46.100, 24.48.070,
		24.48.080, 24.50.080,
		24.50.090, 24.52.120,
		24.52.130, 24.54.120 and
		24.54.130, zoning
		(Repealed by 117570)
ŀ	109861	Amends § 4904 of building
	10,001	code supplement; repeals
		Ord. 109657 (Repealed by
		111200)
ŀ	109863	Adds to Ch. 24.72, zoning
	107003	amendments (Repealed by
		110381)
ŀ	100960	Amends §§ 20.46.050 and
	109869	20.46.130, women's and
		minority business
ŀ	100002	enterprise (20.46)
	109883	Amends § 2 of Ord.
		109648, for-hire drivers
	100001	(Repealer)
	109886	Amends § 2.04.500, fair
ļ		campaign practices (2.04)
	109900	Amends subsection A of §
		24.68.270, zoning
Į		(Repealed by 112134)
	109910	Adds Ch. 10.11, Food
		Code; repeals Ch. 10.10
		and Ords. 92987, 99992,
		104034, 105464, 107837
		and 108421 (Repealed by
		117001)
•		1040

eference only.

I amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipus anda	te file
109915	Repeals and replaces §
	24.64.130, zoning
- 0 101	(Repealed by 117570)
109919	Adds § 3.92.030, vacancies
	in Charter offices (3.92)
109931	Repeals §§ 347.1, 347.2,
100001	348.1, 348.2, 348.3, 349,
	350.1, 350.2, 351, 352,
	353, 354, 355, 356, 357,
orea	358, 358.1 of Ord. 48022
ces crea	and § 3 of Ord. 105430,
	business license code
	(Repealer)
109932	Adds Ch. 1054, fumigation
	and extermination; repeals
1 4 - 01	Ch. 6.80, Ords. 73223,
and to co	75771, 85744, 88838,
Tiles o	102625, and §§ 14, 15 and
	16 of Ord. 106063, § 4 of
C110	Ord. 107158, § 2 of Ord.
1116.	107776 and §§ 1 and 2 of
	Ord. 108076 (10.54)
109936	Repeals Ch. 10.48,
	regulations for removal of
	garbage and swill
	(Repealer)
109949	Adds Ch. 10.01, Health
	Department administration
	and enforcement (10.01)
109950	Renumbers and amends §§
	1-11 of Ord. 108882 to be
	code §§ 4.16.010-4.16.110
	respectively, code of
	ethics; repeals Ords.
	100435, 100648, 100649
	and 106471 (4.16)
109951	Adds Ch. 10.03, Health
	Department fees (10.03)
109954	Adds §§ 3.110.560,
	3.110.570, 3.110.580 and
	3.110.590; amends §
	3.110.020; repeals §
	3.110.250, code of ethics
	for public corporations
	(3.110)
109957	Adds § 5.40.135; repeals §
	5.40.130, disclosure
	requirements for
	admissions tax records
	(5.40)
109969	Adds §§ 15.42.015 and
	15.42.025; amends §
	15.42.010, planting trees
	and shrubs (15.42)
109973	Amends §§ 22.210.020,
	22.210.030, 22.210.050,
	22.210.060, 22.210.070,
	22.210.080, 22.210.120
	and 22.210.170, housing
	preservation (Repealed by
	112342)
MONT DIV	

eference only.

I amending graphics, ccuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

increase (Repealed by 111526)  109988  Amends § 25.04.580, City environmental policies (Repealed by 111866)  109990  Amends § 6.60.010, tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991  Charges for premarital blood test (10.03)  Amends § 1.01.010, code adoption (1.01)  Rezone, Map 73W (Special)  110009  Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110037  Rezone, Map 20W (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)	Cipar of	te file
Citizens Housing Board (Repealed by 110226)  109986 Amends § 7.20.050, floating home moorage fe increase (Repealed by 111526)  109988 Amends § 25.04.580, City environmental policies (Repealed by 111866)  109990 Amends § 6.60.010, tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991 Charges for premarital blood test (10.03)  109994 Amends § 1.01.010, code adoption (1.01)  110001 Rezone, Map 73W (Special)  110009 Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special 110037 Rezone, Map 20W (Special))  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	109974	Amends § 22.202.070,
(Repealed by 110226)  Amends § 7.20.050, floating home moorage fer increase (Repealed by 111526)  109988		
Amends § 7.20.050, floating home moorage fe increase (Repealed by 111526)  109988		
floating home moorage feincrease (Repealed by 111526)  109988 Amends § 25.04.580, City environmental policies (Repealed by 111866)  109990 Amends § 6.60.010, tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991 Charges for premarital blood test (10.03)  109994 Amends § 1.01.010, code adoption (1.01)  110001 Rezone, Map 73W (Special)  110009 Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	109986	
111526    109988		floating home moorage fee
Amends § 25.04.580, City environmental policies (Repealed by 111866)  109990 Amends § 6.60.010, tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991 Charges for premarital blood test (10.03) Amends § 1.01.010, code adoption (1.01)  110001 Rezone, Map 73W (Special)  110009 Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)		increase (Repealed by
environmental policies (Repealed by 111866)  109990  Amends § 6.60.010, tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991  Charges for premarital blood test (10.03)  Amends § 1.01.010, code adoption (1.01)  Rezone, Map 73W (Special)  110009  Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)		111526)
(Repealed by 111866)  109990  Amends § 6.60.010, tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991  Charges for premarital blood test (10.03)  Amends § 1.01.010, code adoption (1.01)  Rezone, Map 73W (Special)  110009  Amends § § 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)	109988	Amends § 25.04.580, City
Amends § 6.60.010, tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991  Charges for premarital blood test (10.03)  Amends § 1.01.010, code adoption (1.01)  Rezone, Map 73W (Special)  110009  Amends § 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)		
tobacco retailers, and § 6.84.030, tobacco vending machines (Repealed by 115531)  109991 Charges for premarital blood test (10.03)  109994 Amends § 1.01.010, code adoption (1.01)  110001 Rezone, Map 73W (Special)  110009 Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	oc cred	
6.84.030, tobacco vending machines (Repealed by 115531)  109991 Charges for premarital blood test (10.03)  Amends § 1.01.010, code adoption (1.01) Rezone, Map 73W (Special)  110009 Amends § 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	109990	
machines (Repealed by 115531)  109991 Charges for premarital blood test (10.03)  Amends § 1.01.010, code adoption (1.01)  Rezone, Map 73W (Special)  110009 Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	10	
115531   109991   Charges for premarital blood test (10.03)   109994   Amends § 1.01.010, code adoption (1.01)   110001   Rezone, Map 73W (Special)   110009   Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)   110013   Amends § 11.31.120, parking ticket late paymer penalty (11.31)   110015   Rezone, Maps 11W and 11E (Special)   110026   Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)   110028   Rezone, Map 27E (Special 110037   Rezone, Map 20W (Special)   110038   Amends subsection B of Section 21.60.230, cable communications (21.60)	Comple	
Charges for premarital blood test (10.03)  109994 Amends § 1.01.010, code adoption (1.01)  110001 Rezone, Map 73W (Special)  110009 Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	Complete	
blood test (10.03)  Amends § 1.01.010, code adoption (1.01)  110001  Rezone, Map 73W (Special)  110009  Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110037  Rezone, Map 20W (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)	100001	
Amends § 1.01.010, code adoption (1.01)  110001  Rezone, Map 73W (Special)  110009  Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)	109991	
adoption (1.01)  Rezone, Map 73W (Special)  110009  Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110037  Rezone, Map 20W (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)	100004	
Rezone, Map 73W (Special)  110009  Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110037  Rezone, Map 20W (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)	109994	
(Special)  110009  Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013  Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110037  Rezone, Map 20W (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)	110001	
Amends §§ 3.14.810 and 3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110037 Rezone, Map 20W (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	110001	
3.14.814, Department of Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110037 Rezone, Map 20W (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	110000	
Purchases regulations (3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Specia 110037 Rezone, Map 20W (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	110009	0.0
(3.18)  110013 Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)		
Amends § 11.31.120, parking ticket late paymer penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)		
parking ticket late paymer penalty (11.31)  110015 Rezone, Maps 11W and 11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	110013	\ /
penalty (11.31)  110015  Rezone, Maps 11W and 11E (Special)  110026  Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028  Rezone, Map 27E (Special)  110038  Amends subsection B of Section 21.60.230, cable communications (21.60)		
11E (Special)  110026 Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Special 110037 Rezone, Map 20W (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)		
Establishes restricted parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Specia 110037 Rezone, Map 20W (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	110015	Rezone, Maps 11W and
parking zone near Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Specia 110037 Rezone, Map 20W (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)		
Fauntleroy Ferry Termina (11.73)  110028 Rezone, Map 27E (Specia 110037 Rezone, Map 20W (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	110026	
(11.73)  110028 Rezone, Map 27E (Special)  110037 (Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)		
110028 Rezone, Map 27E (Special 110037 Rezone, Map 20W (Special) 110038 Amends subsection B of Section 21.60.230, cable communications (21.60)		
110037 Rezone, Map 20W (Special) 110038 Amends subsection B of Section 21.60.230, cable communications (21.60)	110020	` /
(Special)  110038 Amends subsection B of Section 21.60.230, cable communications (21.60)		
Amends subsection B of Section 21.60.230, cable communications (21.60)	110037	-
Section 21.60.230, cable communications (21.60)	110029	
communications (21.60)	110038	
********************************		
110047 Adds § 25.08.820; amends	110047	Adds § 25.08.820; amends
\$\ 25.08.280, 25.08.500,	11007/	
25.08.530, 25.08.560,		
25.08.660, 25.08.710,		
25.08.730, 25.08.800 and		
25.08.810; repeals §§		
25.08.440 and 25.08.720,		
noise control (25.08)		
Solicitation of donations	110049	
		for certain personal health
services (3.30)		services (3.30)

eference only.

I amending graphics, ccuracy of

Seattle Municipal Code ference only. 110052 Adds Ch. 10.13, meat, rabbit, poultry and aquatic Text provided for hi foods code; repeals Ch. 10.12, meat inspection, Ch. 10.14, poultry, rabbit and fish inspection, and repeals amending Ords. 80028, 80601, 85679, 88241, 88581, 92085, 94465, 94567, 95906, 96110, 96125, See ordinan 96787, 96788, 97794, 100445, 102626, 102638, 106060, 108147, 108675, sections for 109494, § 6 of Ord. 106025, §§ 6 and 7 of Ord. 107158, §§ 3 and 5 of Ord. and tables and 108076, § 2 of Ord. 106025 and §§ 10 and 11 of Ord. 107158 (10.13) 110058 Amends §§ 24.68.030, this source 24.68.050, 24.68.060, 24.68.080, title of Subchapter 11, §§ 24.68.100, 24.68.110, 24.68.120, 24.68.130, 24.68.140, 24.68.160, 24.68.170, 24.68.180 and 24.68.190, special review districts; repeals §§ 25.28.010-25.28.150, Pioneer Square Historic District, and repeals Ords. 98852, 99846, 102902, 103392, 103393, 103655, § 8 of Ord. 106985, § 7 of Ord. 109125, §§ 22 and 23 of Ord. 109155 and § 4 of Ord. 109266 (Repealed by 112134) 110062 Amends § 12A.08.040, criminal trespass (12A.08)

> For current SMC, contact the Office of the City Clerk

Adds § 6.288.045; amends § § 6.288.010, 6.288.030, 6.288.070, 6.288.080, 6.288.100 and § 4 of Ord. 109818, scrap metal and used goods dealer licenses

Amends §§ 11.30.120 and 11.30.160, vehicle impoundment (11.30)

Rezone, Map 59E (Special)

Rezone, Map 64E (Special)

(6.204, 6.288)

110082

110106

110108

110109

Seattle Municipal Code

June 2009 co 110136 up date

Repeals 6.62.010 automobrepeals 8 and 3141

See ordinant sections for and tables a this source

4	te the
110136	Repeals §§
	6.62.010-6.62.090, used
	automobile dealers; and
	repeals §§ 308-314, 314a
C TO TO	and 314b of Ord. 48022;
	Ord. 77718, Ord. 77869,
	Ord. 79239, §§ 95-99 of
	Ord. 120636, §§ 37 and 38
	of Ord. 107157, and § 20
as crea	of Ord. 88789 (Repealer)
110137	Amends §§ 22.602.010 and
	22.606.020, Fire Code
	(Repealed by 111001)
110151	Repeals §§
	6.76.010-6.76.160,
	ambulances; and repeals
and to co	Ord. 90952, 94622, 98639,
	99195, 100957, 103513,
	103881, § 1 of Ord.
C:10	107158 and § 2 of Ord.
1115.	104201 (Repealer)
110152	Repeals Ch. 22.910,
110132	maintenance of healthful
	temperatures (Repealer)
110171	• • • • • • • • • • • • • • • • • • • •
110171	Repeals Ord. 98247,
	Seattle Parking
	Commission, and closes
	Seattle Parking
	Commission fund
	(Repealer)
110177	Rezone, Map 37W
	(Special)
110179	Adds § 12A.14.075 and
	amends § 12A.14.080,
	unlawful use of weapons
	(12A.14)
110180	Adds § 10.03.056 and Ch.
	10.56 on Seattle Swimming
	Pool and Spa Code; repeals
	Ord. 98755, public pools
	and spa pools (10.03,
	10.56)
110181	Amends § 3 of Ord.
110101	101476, privately owned
	burglar alarm systems
110105	(Repealed by 116702)
110195	Adds Ch. 4.34, annual
	vacations; repeals Ch. 4.32,
	vacation leave (4.34)
110199	Amends § 21.76.040,
	utility credits for
	low-income elderly persons
	low-income elderry persons

reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

Reference only.

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibar	. c:10
a and	ite Ille
110201	Adds § 21.28.040; amends
	§§ 21.28.030, 21.28.060,
a com hi	21.28.070, 21.28.080,
ad tor	21.28.200, 21.28.220 and
	21.28.240; repeals §§
	21.28.050,
	21.28.090-21.28.190 and
	21.28.210, sewer use rates
110202	(21.28) Adds new section to Title
Crea	4, release time with pay for
	City employees acting as
	Civil Service Commission
COMPA	witnesses (4.20)
110208	Consolidates 1981 water
140 0	construction funds created
and to c	by Ords. 109669 and
	109708 to be Seattle
	Municipal Water System
1511e.	1981 Construction Fund
110212	(5.76)
110212	Amends § 21.24.040, sewer connection charges
	(21.24)
110216	Amends § 2 of Ord. 85317,
110210	sewer connections outside
	City limits (21.20)
110219	Amends § 4.36.190(B) and
	4.36.200(C), City
	employees' retirement
	system (4.36)
110226	Repeals Ch. 3.42, Public
	Reinvestment Review Board; Ch. 3.62, Seattle
	Parking Commission; §§
	22.202.040, 22.202.050,
	22.202.060 and
	22.202.070, Citizens
	Housing Board, and Res.
	24807, Crime Prevention
	Advisory Commission
110220	(Repealer)
110229	Establishes exemptions
	from civil service, repeals Ch. 4.12 (Repealed by
	110329)
110230	Adds Ch. 9.24, animal
	control ordinance; repeals
	Ch. 3.54, Animal Control
	Commission; Ch. 6.16, dog
	and cat licenses; Ch. 9.04,
	leash law; Ch. 9.08,
	restraint of dangerous
	animals, and Ch. 9.16,
	cruelty to animals (Repealed by 112335)
110232	Accepts amendment to
110232	property use and
	development agreement,
	Plat 39E (Special)
110235	Zoning map amendment,
most SIV	Plat 225E (Special)
LI CIII	000

eference only.

I amending

graphics,

accuracy of

For current Sivi Plat 225E (Special)
the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

	·da	te file
11023	36	Rezone, Plat 66W (Special)
11023	37	Rezone, Plat 65E (Special)
11024	12	Human Rights Department
	tor 111	payment of discrimination
300		claims (3.22)
11024	13	Amends §§ 21.76.010,
		21.76.030, 21.76.040,
		21.76.050 and 21.76.060,
		low-income utility credit
	c crea	program (21.76)
11025	50	Adds Ch. 9.26 (§§
	-10	9.26.010-9.26.060), animal
11005		fee ordinance (9.26)
11025	3 1 3 3 3 3 3	Amends §§ 4.04.130 and
		4.04.140, personnel
11025	a to co	regulations (4.04)
11025	4.5	Repeals §§ 3.14.300-3.14.320, Energy
		Office (Repealer)
11025	56	Amends § 3.18.040,
11023	150	ancillary powers of
		Director of Administrative
		Services (Repealed by
		118397)
11025	57	Contributions to
11023	, ,	Employees Retirement
		Fund from Transit Fund
		(4.36)
11025	58	Amends §§ 15.30.030 and
		15.32.020(B), construction
		permits (15.32)
11025	59	Amends § 21.24.020,
		sewer connection and
		alteration fees (Repealed
		by 116421)
11026	50	Adds § 3.104.030, fee for
		legal property description
		research (Repealed by
<u> </u>		111218)
11026	ol .	Adds § 5.44.035 and
		amends §§ 5.44.020 and
		5.44.030, tax on certain
11026	: 1	businesses (5.44)
11026	94	Permit fee ordinance,
		repeals Ord. 109720
11026	5	(Repealed by 110880) Adds § 10.03.020 and Ch.
11026	J.J	10.58, school inspections,
		reviews of plans and
		specifications, and
		investigations (10.03,
		10.58)
11026	56	Amends §§ 2 and 3 of Ord.
11020		107099 and adds §§ 1-3 of
		Ord. 107099, as amended,
		to Title 10, water supply
		and sewage disposal
		system reports and water
		samples (10.25)
		F ( )

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	, ipus	te file	
	110267	Adds § 6.82.161, amends §	٥
		6.82.250 and repeals §	
	1 Sor 111	6.82.160, refrigeration	
•	U IOI	system permit fees	
	110260	(Repealed by 113702)	
	110268	Adds §§ 10.03.110 and	
		10.11.902; amends §§	
		10.03.010 and	
		10.11.911(B); repeals §§ 10.11.912 and 10.11.913.	
	age Crea	Health Code fees and Food	
7		Code (10.03)	K
	110269	Adds § 10.03.050, Health	3
	110209	Code fees (Repealed by	
	Compage	110887)	
	110270	Adds Ch. 1062,	Þ
	1102/0	wholesaling of drugs and	
	III o	supplies (10.62)	
	110271	Adds § 10.03.040, Public	
	Callo	Health Department	
	1116.	laboratory fees; repeals	
		Ords. 105149, 106688,	
		107888 and 109492	
		(Repealed by 110882)	
	110272	Adds § 10.03.130 and Ch.	
		10.60, personal health care	
		programs; repeals Ord.	
		109508 (10.03, 10.60)	
	110273	Amends § 3.28.070, fees	
		for Police Department	
		reports and services (3.28)	
	110274	Amends §§ 5.48.020 and	
		5.48.050, business tax on	
	110275	utilities (5.48)	
	110275	Amends § 5.40.020,	
	110299	admission tax (5.40) Adds § 22.606.140,	
	110299	Uniform Fire Code Article	
		93, standards for high-rise	
		buildings (Repealed by	
		114328, 115405)	
	110300	Amends § 6.212.060,	
		taxicab inspections	
		(Repealed by 118341)	
	110301	Adds (O) to and amends	
		(D), (E), (J), (M), (O) and	
		Table 38-A of §	
		22.100.010; amends §	
		22.102.040(D), building	
	110202	code (Repealed by 111200)	
	110302	Amends §§ 4.04.030 and	
	110303	4.04.150, personnel (4.04) Amends §§ 24.46.110 and	
	110303	24.48.100, zoning	
		(Repealed by 117570)	
	110304	Amends §§ 20.88.020 and	
	***:	20.88.030, real property	
		leases (3.18)	
,			•

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Shar	0210
	1 42	te ille
	110305	Authorizes City Treasurer
	Che or a	to establish credit for
	a com hi	cashing City warrants
	A TOK ISS	(5.16)
	110318	Amends § 21.16.060,
		side-sewer contractor
		examinations and insurance
	110329	(21.16) Establishes exemptions
	110329	from civil service, repeals
	ces crea	Ord. 110229 (Repealed by
		110656)
	110343	Designates North Queen
	COMPA	Anne Drive Bridge as a
		landmark (25.32)
	110344	Designates Cowan Park
4	ma to	Bridge as a landmark
	110245	(25.32)
	110345	Designates George Washington Memorial
	THE.	(Aurora) Bridge as a
		landmark (25.32)
	110346	Designates Schmitz Park
		Bridge as a landmark
		(25.32)
	110347	Designates Fremont Bridge
		as a landmark (25.32)
	110348	Designates Fauntleroy
		Community Church and
		Y.M.C.A. as a landmark (25.32)
	110349	Designates Beacon Hill
	110547	First Baptist Church as a
		landmark (25.32)
	110350	Designates University
		Methodist Episcopal
		Church and Parsonage as a
		landmark (25.32)
	110351	Designates Seattle First
		Baptist Church as a landmark (25.32)
	110352	Designates New Age
	110332	Christian Church as a
		landmark (25.32)
	110353	Designates the Black
		Property as a landmark
		(25.32)
	110354	Designates Langston
		Hughes Cultural Arts Center as a landmark
		(25.32)
	110355	Designates Samuel Hyde
	110000	House as a landmark (Not
		codified)
	110363	Amends §§ 11.26.020,
		11.26.060, 11.26.260 and
		11.26.280, service parking
		permit fees (11.26)
	110368	Amends § 21.28.070,
		sewer service charge
	**************************************	(Repealed by 118176)
		-

eference only.

amending
graphics,
accuracy of

For current Sind, and Clerk the Office of the City Clerk

June 2009 co Text provide

See ordinan sections for and tables a

		te file
	110372	Repeals Ch. 6.64, vehicle
	Check and	rental agencies (Repealer)
ine Zoo	110373	Amends § 6.214.010 (B)
	Tarana III	and (D), towing operator
1	in re-	and tow truck licenses
out Drugge		(6.214)
EVC P	110374	Amends § 5.40.020,
	110374	
	110201	admission tax (5.40)
	110381	Adds new Title 23, land
10 010	DOS CIES	use code, and § 24.14.05,
		zoning; amends §§
See Orussian	-10	24.12.010, 24.12.020,
	anmille	24.24.010, 24.24.040(G),
103 201	Comple	24.24.050(A) and (D),
200110115		24.30.100(B) and (C),
SECTION	140 0	24.32.090(B) and (C),
4.4.4	ing to o	24.60.010, 24.60.140,
a tables	1	24.60.285, 24.60.315,
and to	- 2 2	24.60.365, 24.60.615(A),
	Sile.	24.60.765(A), 24.62.010,
- CONTROL	1110.	24.62.040, 24.68.020 and
this sum		24.68.030(A); repeals Chs.
		24.10, 24.16, 24.20, 24.22,
		24.72, 24.84 and §§
		24.06.040, 24.12.040,
		24.12.050, 24.12.060,
		24.62.030, 24.62.100,
		24.68.080, 24.72.022,
		24.74.030, 24.74.040 and
		24.74.130, zoning (23.02,
		23.04, 23.30, 23.32, 23.34,
		23.40, 23.42, 23.44, 23.45,
		23.60, 23.66, 23.78)
	110382	Amends §§ 3.17, 16.4, 17.4
	110382	and 18.7 of Ord. 86300,
		zoning (Repealed by
	110200	117570)
	110399	Amends § 21.40.080,
		disposal site and transfer
	110110	station fees (21.40)
	110413	Petty cash account
		establishment and
		regulation (5.12)
	110421	Amends § 5.32.170,
		amusement device license
		fee (5.32)
	110422	fee (5.32) Adds Ch. 15.64, skybridge
	110422	
	110422	Adds Ch. 15.64, skybridge permits (15.64)
		Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W
	110423	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special)
		Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050,
	110423	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050, number of bundles allowed
	110423	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050, number of bundles allowed per garbage collection
	110423 110443	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050, number of bundles allowed per garbage collection service level (21.36)
	110423	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050, number of bundles allowed per garbage collection service level (21.36) Rezone, Map 16W
	110423 110443 110445	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050, number of bundles allowed per garbage collection service level (21.36) Rezone, Map 16W (Special)
	110423 110443	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050, number of bundles allowed per garbage collection service level (21.36) Rezone, Map 16W (Special) Amends § 21.76.040,
	110423 110443 110445	Adds Ch. 15.64, skybridge permits (15.64) Rezone, Map 219 W (Special) Amends § 21.36.050, number of bundles allowed per garbage collection service level (21.36) Rezone, Map 16W (Special)

eference only. amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Sihar	4 - 6110
a anda	te me
110448	Amends §§ 21.40.050 and
4-9	21.40.060, garbage
110449	collection rates (21.40)
110449	Amends § 4.20.320, meal reimbursement for
	employees during overtime
	work (Repealed by
	111768)
110450	Amends § 3.16.070, use of
7 000	fire equipment outside City
ces crea	limits (3.16)
110451	Adds §§
10.51	4.70.010-4.70.040,
COMPA	reimbursement for required
	use of personal
140 00	automobiles; repeals §§
and to co	4.68.060, 4.68.070 and
12200	4.68.080 (4.70)
110453	Amends §§ 9.24.045 and
Sile.	9.24.082, Municipal Spay
777	and Neuter Clinic
	(Repealed by 112335)
110461	Police Department Horse
	Patrol account established
	in General Donations and
110462	Gift Trust Fund (5.76)
110462	Amends §§ 12A.14.010,
	12A.14.080 and
	12A.14.090, weapons control (12A.14)
110464	Amends § 24.64.080 and
110404	supersedes § 24.72.040,
	zoning (Repealed by
	110381)
110476	Adds §§ 5.44.072,
	5.44.074 and 5.44.078;
	renumbers and amends §
	5.44.070(A) to be
	5.44.076; amends §§
	5.44.070(B) and 5.44.110,
	business and occupation
110402	tax (5.44)
110492	Designates Krans/Andersson House as
	a landmark (25.32)
110503	Adds new subsection to §
110000	4.20.300, compensation
	and working conditions
	(4.20)
110504	Amends §§ 3.114.060 and
	3.114.130, consultant
	selection (3.114)
110507	Amends § 4.20.300,
	compensation and working
	conditions (4.20)
110515	Amends § 1 of Ord.
	110026, restricted parking
	near Fauntleroy Ferry
	Terminal (11.73)
	- 104

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

e Mumie	Signal	to file
	110529	Amends §§
2009 33	a sor hi	21.48.030-21.48.090, electrical rates (Repealed by 110733)
provide	110547	Amends §§ 24.46.070 and 24.46.080, zoning
	110548	(Repealed by 117570) Adds §§ 22.206.150(A)(9)
	on orea	and 22.206.160(H), habitable building
rdinan	ces cross	requirements, and adds §§ 2.201(h) and 10.307(g) to
- Sor	comple	Uniform Fire Code sections adopted in §
ous ior	1 40 66	22.600.020 (Repealed by 111001)
tables a	110566	Adds §§ 11.14.617, 11.40.370 and Ch. 11.65,
	sile.	streetcar rules, and amends § 11.14.665, vehicles and
source	110570	traffic (11.14, 11.40, 11.65) Adds Subtitle III to Title
		23, platting requirements; adds Ch. 23.45, residential
		multi-family, and adds §§ 23.16.02(B), 23.34.36,
		23.34.38, 23.34.40, 23.34.42, 23.34.44,
		23.40.10, 23.86.06(C), 23.86.12, 23.86.14,
		23.86.16, 23.86.18, 23.86.20 and 23.86.22; adopts Official Land Use
		Map, Exhibit A; amends §§ 23.30.10, 24.46.080,
		24.48.070, 24.54.10, 23.54.020, 23.54.30,
		23.76.06, 23.76.30, 23.84.04, 23.84.06,
		23.84.18, 23.84.20, 23.84.30, 23.84.32,
		23.84.36 and 23.84.38; repeals Ch. 24.98, land use
		and zoning (23.20, 23.22, 23.24, 23.28, 23.30, 23.34,
	110572	23.45, 23.46, 23.84, 23.86) Amends §§
		14.12.010-14.12.050, 14.12.060-14.12.400 and §
		42 of Ord. 108333, collection of information
		for law enforcement purposes (14.12)
	110579	Amends §§ 5.32.170 and 5.32.200, amusement
	110585	device licensing (5.32) Adds § 2 to Ord. 110305,
		credit for cashing City warrants (5.16)
	110586	Repeals § 3.14.200, sale of Citizen's Service Bureau
for cur	rent 3M	handbooks (Repealer)
ACC	ice of th	e City
the Ott		

reference only. amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
110590	Amends § 5.48.050(I), tax
110330	on solid waste collection
103	businesses (5.48)
110614	Adds Ch. 3.16 (§§
110014	3.16.200 and
	3.16.200A-3.16.230A),
	hazardous materials
	handling (3.16)
110623	Amends § 20.48.070,
oc crea	Board of Public Works
CE2 C2	(Repealed by 117390)
110640	Amends §§ 14.12.020(E),
	14.12.030(E) and (F),
COMPATO	14.12.060, 14.12.070,
	14.12.100, 14.12.130,
	14.12.200 (A)(1),
ad to co	14.12.210(B)(5),
III CO	
	14.12.230(A) and (B),
	14.12.250(A) and (B),
<b>\$116.</b>	14.12.260, 14.12.280(B),
7770	14.12.320,
	14.12.350(A)(5),
	14.12.350(B) and
	14.12.380, collection of
	information for law
	enforcement purposes
	(14.12)
110649	Adds § 10.03.140,
11004)	influenza immunization fee
	(10.03)
110656	
110656	Establishes exemptions
	from Civil Service; repeals
	Ord. 110329 (Repealed by
	111127)
110661	Amends § 22.210.070,
	housing preservation
	(Repealed by 112342)
110669	Adds § 23.40.04; adopts
110669	Adds § 23.40.04; adopts new land use maps in place
110669	new land use maps in place
110669	new land use maps in place of maps adopted by Ord.
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A),
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20, 23.42.10, 23.42.20,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20, 23.42.10, 23.42.20, 23.44.06, 23.44.08,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20, 23.42.10, 23.42.20,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20, 23.42.10, 23.42.20, 23.44.06, 23.44.08,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20, 23.42.10, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D)
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20, 23.42.10, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C),
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.26, 23.76.14(C),
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.40.20, 23.42.10, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02, 23.84.30, 23.84.36,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C),
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C), 23.90.18(B), and makes
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C), 23.90.18(B), and makes extensive, detailed changes
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C), 23.90.18(B), and makes extensive, detailed changes to all sections of Ch. 23.44,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.20(A), 23.44.24, 23.44.26, 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C), 23.90.18(B), and makes extensive, detailed changes to all sections of Ch. 23.44, land use code; and amends
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C), 23.90.18(B), and makes extensive, detailed changes to all sections of Ch. 23.44, land use code; and amends §§ 24.14.05 and 24.60.615,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C), 23.90.18(B), and makes extensive, detailed changes to all sections of Ch. 23.44, land use code; and amends §§ 24.14.05 and 24.60.615, land use and zoning (23.04,
110669	new land use maps in place of maps adopted by Ord. 110570 § 2; amends §§ 23.04.10, 23.12.30, 23.22.52(C), 23.24.040(A), 23.40.02, 23.42.20, 23.44.06, 23.44.08, 23.44.10, 23.44.14(B), (D) and (F), 23.44.16(C), 23.44.24, 23.44.26, 23.76.14(C), 23.76.36(B), 23.84.02, 23.84.30, 23.84.36, 23.86.06(B), 23.86.08, 23.86.10(B) and (C), 23.90.18(B), and makes extensive, detailed changes to all sections of Ch. 23.44, land use code; and amends §§ 24.14.05 and 24.60.615,

reference only.

d amending

graphics,
accuracy of

For current 314, 23.44, 23.86)

The Office of the City Clerk

the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

cipar of sile			
a and	ate me		
110673	Amends §§		
	12A.20.010-12A.20.040, controlled substances and		
a sor n	legend drugs (Repealed by		
an ros	117158)		
110674	Adds Ch. 5.64 (§§		
	5.64.010-5.64.090), tax on		
110675	sale of real estate (5.64) Amends §§ 10.52.030,		
1100/3	10.52.040 and 10.52.050,		
ces cree	weeds and vegetation		
	(Repealed by 114355)		
110686	Adds §§ 11.66.035,		
Comple	11.66.050, and amends §		
	11.66.030, railroad operations (11.66)		
110695	Amends §§ 21.04.110 and		
1110	21.04.350, water		
	department (21.04)		
110701	Amends § 12A.16.010,		
3330	obstructing duties of public		
110710	officers (12A.16) Amends §§ 20.04.140(D)		
110/10	and 20.04.190, public		
	improvement districts		
	(20.04)		
110713	Establishes restricted		
	parking zone in vicinity of		
	Group Health Cooperative (11.73)		
110731	Amends § 4.12.010(A)(2),		
110/31	exemptions from civil		
	service (Repealed by		
	111127)		
110733	Adds Ch. 21.49, Seattle		
	City Light Department; repeals Ch. 21.48 and		
	Ords. 109218, 109339,		
	109621 and 110529 (21.49)		
110737	Amends § 24.08.050(1)		
	and supersedes §§		
	24.72.040-24.72.070,		
	zoning (Repealed by 111127)		
110738	Repeals and replaces §		
	10.03.040(A), laboratory		
	fees; repeals portion of § 1		
	of Ord. 110271 (Repealed		
110742	by 110882)		
110/42	Amends § 10.52.020, weeds and vegetation		
	(Repealed by 114355)		
110749	Adds Ch. 5.06, City		
	Investment Committee;		
	repeals § 5.04.010 and Ord.		
110757	94190 (5.06) Amends §§ 1.01.010 and		
110757	1.01.020, code adoption		
	(1.01)		
<u> </u>	1 ()		

eference only.

amending
graphics,
accuracy of

110785

110786

June 2009 co

Text provide

110793 See ordinan sections for and tables and

Amends §§ 12A.14.010 and 12A.14.080, weapons control (12A.14)

Adds § 12A.06.015, simple assault; repeals § 12A.06.020 (Repealed by 116872)

this source file.

ference only.

amending

Adds §§ 23.45.108 and 23.45.140(C); amends §§ 23.30.10, 23.34.34, 23.44.10(D), 23.44.12(B), 23.44.20(A), 23.44.46(A), 23.44.80(G), 23.45.04(A), 23.45.05(B), 23.45.08(C) and (D), 23.45.10(B), 23.45.14(A), (C) and (D), 23.45.16(A) and (B), 23.45.18(A)-(C), 23.45.22(C) and (D), 23.45.24(B), 23.45.26, 23.45.28(A), (C) and (D), 23.45.30, 23.45.32(A)-(C), 23.45.36(C) and (D), 23.45.38(B), 23.45.40, 23.45.42(A), (C) and (D), 23.45.44, 23.45.46(A)-(C), 23.45.50(C) and (D), 23.45.52(B), 23.45.54, 23.45.56(A), (C) and (D), 23.45.58, 23.45.60(A)-(C), 23.45.66(B) and (C), 23.45.70(B), 23.45.72(B) and (D), 23.45.76, 23.45.86(D), 23.45.90, 23.45.96(C), 23.45.98(B) and (G), 23.45.102, 23.45.106(A), 23.45.118(B), 23.45.122(D) and (E), 23.45.146(B) and (D), 23.45.154, 23.45.166(A), 23.45.180, 23.45.182, 23.45.184(A) and (D), 23.45.190(A), 23.54.10, 23.54.20(C), (F) and (G), 23.54.30(C), 23.76.14(C) and (E), 23.76.24(A), 23.76.36(B), 23.78.10, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.24, 23.84.25, 23.84.32, 23.84.36, 23.84.38, 23.84.44, 23.86.10(B), 23.86.12(A), 23.86.14, 23.86.16, 23.86.18, 23.88.20(A) and 23.90.18(A), land use code (23.30, 23.44, 23.45, 23.78, 23.84, 23.86, 23.88)

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipal costs		
a mada	te tile	
110798	Amends § 1 of Ord.	
	109261, restricted parking	
1 Con hi	zone near Providence	
d tor	Medical Center (11.73)	
110816	Amends §§ 21.04.190 and	
	21.04.470; repeals §§	
	21.04.280 and 21.04.290,	
	Water Department (21.04)	
110826	Repeals Ord. 91675,	
age crea	treasurer's office operating	
	fund (Repealer)	
110829	Amends §§ 21.49.050(E)	
comple	and (F) and 21.49.070(E)	
Compre	and (F), City Light	
	Department (Repealed by	
110007	112738)	
110837	Rezone, Plat 49W, zoning	
110000	map page 126 (Special)	
110839	Rezone, Plat 67W, zoning	
110843	map page 161 (Special)	
110843	Amends § 5.48.050, utility	
110040	business taxes (5.48)	
110848	Rezone, Plat 232W, zoning	
110952	map page 26 (Special)	
110852	Amends §§ 4.04.030 and	
	4.04.220, personnel system (4.04)	
110856	Amends §§ 12A.24.050	
110830	and 12A.24.150,	
	frequenting places where	
	liquor is unlawfully kept	
	(12A.24)	
110857	Rezone, Plats 49E and	
110037	50W, zoning map pages	
	127 and 128 (Special)	
110866	Police Department Horse	
	Patrol (5.76)	
110871	Amends § 22.606.140,	
	Seattle Fire Code	
	(Repealed by 114328,	
	115405)	
110877	Sales and use tax (5.60)	
110878	Amends §§ 5.44.020,	
	5.44.030, 5.44.120,	
	5.44.130, 5.44.140 and	
	5.44.180; repeals §	
	5.44.035, business and	
	occupation tax (5.44)	
110879	Amends § 15.62.030, street	
	vacation procedures	
110000	(15.62)	
110880	Adds Ch. 22.900, 1983 permit fee ordinance;	
	repeals Ord. 110264 (Repealed by 111422)	
110881	Adds § 10.03.013, food	
110001	permit fees; amends §	
	10.13.120(C) and repeals §	
	10.13.120( <i>C</i> ) and repeats § 10.13.130, meat, rabbit,	
	poultry and aquatic foods	
	code (10.03, 10.13)	
TOTAL SIVI	7	

eference only.

all amending

graphics,
accuracy of

For current Sim out (10.03, 10.13) - Clerk the Office of the City Clerk

Seattle Municipal Code

June 2009 co 110882 up date

Adds § 10.03
110271
medica (10.03)
110883

See ordinant sections for and tables a this source

	TO THE
110000	111 8 10 02 045
110882	Adds § 10.03.045; repeals
	§ 10.03.040 and Ords.
1 Co# 111	110271 and 110738,
U IOI	medical laboratory fees
	(10.03)
110883	Amends § 10.03.110(A),
	food-service establishment
	permit fees (10.03)
110884	Amends § 21.44.040 and
oc crea	repeals § 21.44.060, solid
CE2 CZ	waste handling fees
	(Repealed by 113167)
110885	Amends § 22.504.010,
COMPA	plumbing permit fees
	(22.504)
110886	Amends § 10.03.056,
and to C	swimming and spa pool
11100	fees (10.03)
110887	Repeals § 10.03.050,
Call O -	geriatric service fees
1110.	(Repealer)
110888	Amends §§ 6.06.020,
110000	6.06.030, 6.08.010,
	6.14.030, 6.14.040,
	6.14.100, 6.18.020,
	6.20.010, 6.34.010,
	6.34.020, 6.36.020,
	6.36.030, 6.36.040,
	6.36.070, 6.38.020,
	6.48.020, 6.52.010,
	6.54.030, 6.54.080,
	6.82.140, 6.84.020 and
	6.102.030, all pertaining to
1	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
	various business license
	fees (6.08, 6.14, 6.18, 6.20,
	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84,
110000	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)
110889	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102) Amends §§ 6.204.030,
110889	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102) Amends §§ 6.204.030, 6.204.040, 6.204.050 and
110889	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license
	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102) Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204)
110889 110890	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102) Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204) Amends § 9.26.020, animal
110890	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)
	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102) Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204) Amends § 9.26.020, animal license fees (9.26) Amends § 5.32.170,
110890	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license
110890	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)
110890	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license
110890 110891	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license
110890 110891	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)
110890 110891	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and
110890 110891 110892	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)
110890 110891 110892	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code
110890 110891 110892	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code permit and inspection fees
110890 110891 110892 110893	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code permit and inspection fees (22.602)
110890 110891 110892	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.040, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 5.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code permit and inspection fees (22.602)  Closes Seattle Arts
110890 110891 110892 110893	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 9.26.020, animal license fees (9.26)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code permit and inspection fees (22.602)  Closes Seattle Arts Commission Operating
110890 110891 110892 110893	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 9.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code permit and inspection fees (22.602)  Closes Seattle Arts  Commission Operating Fund, transfers assets and
110890 110891 110892 110893	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 9.26.020, animal license fees (9.26)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code permit and inspection fees (22.602)  Closes Seattle Arts  Commission Operating Fund, transfers assets and liabilities, and repeals §
110890 110891 110892 110893	fees (6.08, 6.14, 6.18, 6.20, 6.36, 6.38, 6.48, 6.84, 6.102)  Amends §§ 6.204.030, 6.204.050 and 6.204.080, business license fees (6.204)  Amends § 9.26.020, animal license fees (9.26)  Amends § 9.32.170, amusement device license fees (5.32)  Amends § 22.904.040, mobile home park license fees (22.904)  Repeals § 22.602.010 and adds Ch. 22.602, Fire Code permit and inspection fees (22.602)  Closes Seattle Arts  Commission Operating Fund, transfers assets and

reference only.

amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

_	4 anda	te tile
	110895	Closes City-County Health
		Fund, transfers assets and
	10-m h1	liabilities, and repeals §
	4 FOL 117	3.30.030 and § 3 of Ord.
þ		109522 (Repealer)
	110896	Adds § 17.16.025 and
		repeals Ords. 108658 § 2
l		and 109490 § 1, Seattle
		Center Department
	- 0463	(Repealed by 112573)
l	110897	Amends § 3.28.070, Police
ŀ	11007	Department fees (3.28)
ŀ	110899	Amends § 20.48.070,
	11089	Board of Public Works
	COMPAGE	
ŀ	110002	(Repealed by 117390)
١	110902	Amends §§ 10.03.130 and
	ma to	10.60.010, Public Health
ľ		Department; repeals §
L		10.60.020 (10.03, 10.60)
	110903	Amends §§ 10.03.140 and
	7770	10.03.150, Public Health
		Department fees; repeals
		Ords. 98610, 105145,
		105146, 109496 (10.03)
	110906	Amends § 17.16.010,
		Seattle Center Department
		(17.16)
Ī	110908	Amends § 3.76.030, Debt
		Management Policy
		Advisory Committee (3.76)
ŀ	110909	Amends § 2.04.480;
l	110,00	readopts §§ 2.04.350,
		2.04.360 and 2.04.370;
		repeals § 2.04.500(C),
		campaign contribution
ŀ	110010	limitations (2.04)
١	110919	Amends §§ 21.49.030,
١		21.49.040 and
١		21.49.110(P), City Light
L	110020	Department (21.49)
١	110930	Amends § 1 of Ord.
١		109814, Engineering
١		Department (Repealed by
L		110936)
١	110938	Adds Ch. 3.95, City
١		Comptroller; repeals Ch.
l		3.94 (3.38, 3.95)
Γ	110940	Adds Ch. 10.14,
١		commercial fishing boat
١		retail sales (10.14)
ŀ	110951	Amends § 15.04.070, street
l	110/31	use permits (15.04)
ŀ	110056	
١	110956	Adds § 11.76.015 and
۱		amends § 11.14.425, Traffic Code (11.14, 11.76)
ı		

d amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipar	
andate till	3
110967 Adds §§ 11.14.557	and
11.56.360; amends	
11.14.420, 11.22.08	
11.26.080, 11.31.03	
11.56.020, 11.56.22	
11.56.240, 11.56.26	
11.56.280, 11.56.30 11.56.310 and 11.7	
Traffic Code (11.14	
11.26, 11.31, 11.56	
Closes Municipal	4
Sewerage 1959 Construction Fund:	<b>56</b> 9
transfers assets and	
liabilities; repeals §	
Ord. 87653 (Repea	
110970 Rezone, Plat 70E, z	
map page 168 (Spe	cial)
110978 Amends § 3.02.175	
Hearing Examiner	filing
fees (3.02)	
110979 Amends § 4.20.250	
personnel policies ( 110982 Amends § 6.52.020	
contractors' license	
(Repealed by 1114	
110986 Repeals §§ 3.12.10	
3.12.110 and 3.12.1	
Director of Engineer	ering
(Repealer)	
110999 Terminates joint	
administration of	
Seattle-King Count Start Program, abol	
City-County Head	
Policy Board and	Start
Administrative Gro	up, and
repeals Ord. 10271	2
(Repealer)	
111001 Adds Chs. 22.600 a	
22.604, Seattle Fire	Code;
repeals §§	020
22.600.010-22.600. 22.604.010,	.030,
22.606.010-22.606	130 and
repeals Ords. 1086'	
109765, 109766, 11	
110594 and §§ 1 ar	
Ord. 110548 (Repe	aled by
114328, 115405)	
111003 Amends §§ 20.04.1	
and 20.04.190, loca	
improvement proce	aures
(20.04) 111017 Amends § 15.40.01	0
111017 Amends § 15.40.01 warning lights and	υ,
barricades (15.40)	
111021 Imposes controls of	1
Boyer/Lambert Res	idence,

reference only.

amending

graphics,
accuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinantes sections for and tables at this source

Cipar	40 Sile
4. 11002	IE 111
111022	Imposes controls on
	Satterlee House, a
- Cott Mi	landmark (25.32)
111023	Imposes controls on
	Nelson/Steinbrueck House,
	a landmark (25.32)
111024	Imposes controls on
	Handschy/Kistler House, a
	landmark (25.32)
111025	Imposes controls on Drake
(CES OF	House, a landmark (25.32)
111030	Accepts §§ 5 and 8 of Ord.
	109675, Weatherization
6 CORRELA	Program (21.12)
111043	Adds §§ 22.800.090,
1400	22.804.080(E), 22.804.135
and to c	and 22.804.262; amends §§
	22.800.080, 22.802.020,
	22.802.030(E),
Sile.	22.802.040-22.802.090,
3 3 3 3 5 .	22.802.100(A)-(C),
	22.802.140, 22.804.030,
	22.804.040, 22.804.050,
	22.804.070, 22.804.090,
	22.804.100, 22.804.110(C),
	22.804.120, 232.804.130,
	22.804.140,
	22.804.160-22.804.240;
	renumbers § 22.804.250 to
	be § 22.804.261 and
	amends § 22.804.261;
	amends §§
	22.804.260-22.804.280 and
	22.806.010-22.806.030;
	repeals and reserves
	section numbers for §§
	22.802.010, 22.804.020,
	22.804.060 and
	22.804.150, all pertaining
	to grading and drainage
	control (Repealed by
	117697)
111058	Imposes controls on First
	Avenue Groups/Waterfront
	Center Project, a landmark
	(25.32)
111068	Rezone, Plat 59E, zoning
	map page 79 (Special)
111076	Amends § 15.62.090(B),
	street vacations (Repealed
	by 113022)

eference only.

I amending graphics, ccuracy of

Seattle Municipal Code ference only. 111078 Adds §§ 24.60.046, 24.60.086, 24.60.131 and Text provided for hi 24.60.266; amends §§ 24.60.090, 24.60.100, 24.60.105, 24.60.106, 24.60.110, 24.60.126, amending 24.60.127, 24.60.265, 24.60.310, 24.60.315, 24.60.350, 24.60.355, 24.60.375, 24.60.380, See ordinan 24.60.385, 24.60.390, 24.60.395, 24.60.400, sections for comp 24.60.405, 24.60.410, 24.60.420, 24.60.440, 24.60.460, 24.60.465, 24.60.475, 24.60.480, and tables and 24.60.525, 24.60.530, 24.60.535, 24.60.540, 24.60.545, 24.60.550, this source file. 24.60.555, 24.60.560, 24.60.570, 24.60.575, 24.60.580, 24.60.585, 24.60.590, 24.60.595, 24.60.600, 24.60.615, 24.60.635, 24.60.640, 24.60.650, 24.60.655, 24.60.660, 24.60.670, 24.60.675, 24.60.680, 24.60.685, 24.60.690, 24.60.705, 24.60.710, 24.60.715, 24.60.720, 24.60.725, 24.60.730, 24.60.740, 24.60.745, 24.60.750, 24.60.755, 24.60.760, 24.60.765; repeals §§ 24.60.490, 24.60.495, 24.60.500, 24.60.505, 24.60.510 and 24.60.515, zoning (Repealed by 113466) 111080 Amends § 3.18.080, Seattle Center improvements (Repealed by 116007) 111083 Amends § 12A.24.050, frequenting places where liquor is kept unlawfully (12A.24)111098 Adds Ch. 23.68 to land use code, interim Greenbelt Overlay District (Repealed by 111568)

## For current SMC, contact the Office of the City Clerk

Adds Ch. 10.64, smoking in public places (10.64)

111099

Seattle Municipal Code June 2009 co 11100 Text provide See ordinan

10,

reference only. amending graphics, ccuracy of

A SOT III	23.34.62, 23.34.64, 23.34.66, 23.34.68 and Ch.
	25.54.00, 25.54.00 and Cil.
	23.48; amends §§ 23.30.10,
	23.54.10(B), 23.54.30,
	23.84.02, 23.84.06,
	23.84.08, 23.84.30,
	23.84.36 and 23.86.12(A);
Loc crea	and adopts major
ICES CT	institution zoning maps, all
	pertaining to land use by
	major institutions (23.30,
COMPA	23.84, 23.86)
111101	Adds Chs. 23.80 and 23.82
140 0	to Land Use Code, major
and to co	institutions (23.80)
111104	Amends § 21.49.090,
000	Seattle City Light
Sile.	Department (21.49)
111119	Amends §§ 22.900.220(A)
	and (B), 22.900.230 and
	22.900.240, building,
	mechanical and electrical
	permit fees (Repealed by
	111422)
111127	Adds §§ 4.13.010 and
	4.13.020, exemptions from
	civil service systems;
	repeals Ch. 4.12 and Ords.
	110656 and 110731
111125	(Repealed by 111929)
111135	Amends § 4.36.130(A),
	City employees' retirement
111126	system (4.36)
111136	Adds § 4.36.130(J), City
	employees' retirement
111142	system (4.36) Creates subaccount for
111143	
	Bumbershoot revenues in
	Seattle Center Fund
111170	(Repealed by 112577) Rezone, Plat 64W, zoning
1111/0	
111194	map page 155 (Special) Amends § 7.20.030,
111164	
	floating homes (Repealed by 111526)
111105	Adds § 5.24.005, claims
111195	
	for damages against the
111109	City (5.24) Amends § 2.08.010,
111190	
111200	initiative petitions (2.08)
111200	Repeals and replaces
	Subtitle I of Title 22 (§
	22.100.010), Building
	Code (Repealed by
	116012)
	111101 111127 111135 111136 111143 111170 111184 111198 1111200

June 2009 co

Text provide

See ordinant sections for and tables a this source

	cipal Cour			
	a anda	te file		
	111201	Repeals and replaces		
		Subtitle IV of Title 22 (§		
	1 Sor MI	22.400.010), Mechanical		
	d in in	Code (Repealed by 116011)		
	111214	Amends § 15.14.080,		
	111214	newsstands (15.14)		
	111218	Repeals Ord. 110260, fee		
		for legal property		
	oc crea	description research		
$\mathcal{I}$	CE2 C2	(Repealer)		
	111221	Amends § 22.606.140, Fire		
	comple	Code (Repealed by		
	111222	114328, 115405)		
	111222	Rezone, Plat 221E, zoning map page 8 (Special)		
	111223	Adds §§ 2.04.215 and		
	Harry Co.	2.04.330; amends §§		
		2.04.010(7), (8) and (17),		
	Sile.	2.04.060(B), 2.04.160,		
5	7770	2.04.170(A), 2.04.180,		
		2.04.220(B)-(D), 2.04.230,		
		2.04.250-2.04.270 and		
		2.04.500(B); repeals § 2.04.190, election		
		campaign contributions		
		(2.04)		
	111227	Amends § 21.24.020,		
		sewer permit fees		
		(Repealed by 116421)		
	111235	Amends § 25.24.060, Pike		
		Place Market Historical		
	111236	District (25.24) Repeals Ch. 17.32 and		
	111230	Ords. 102826 and 104657,		
		Pike Place Market		
		Preservation and		
		Development Authority		
	111040	(Repealer)		
	111243	Amends §§ 21.49.040, 21.56.040, 21.76.010 and		
		21.76.030, Seattle City		
		Light Department (21.49,		
		21.76)		
	111244	Establishes Pioneer Square		
		Parking and Business		
		Improvement Area and		
		Business Improvement Area Fund (5.76)		
	111257	Amends §§ 24.60.545,		
		24.60.570, 24.60.575 and		
		Table 2 following §		
		24.60.420, zoning		
	111071	(Repealed by 113466)		
	111261	Amends §§ 24.24.030,		
		24.24.040, 24.24.050, 24.24.060 and 24.24.070,		
		zoning (Repealed by		
		117570)		
	111262	Adds § 21.60.700, cable		
	- CN	television (21.60)		
	rent 2111	Clot		

d amending

d amending graphics, generacy of

For current Sime, the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

	a anda	te file
	111279	Adds § 11.58.420 and
		amends § 11.56.020,
	- 0 - m hi	Traffic Code (11.56, 11.58)
	111300	Rezone, Plat 29E, zoning
5		map page 90 (Special)
ſ	111301	Adds § 6.204.090 and Ch.
		6.230 (§§
		6.230.010-6.230.190 on
		licensing of steam
L	oc crea	engineers); repeals §§
V	ces es	6.54.010-6.54.180,
		business licenses (6.230)
	111302	Imposes controls on Old
	COMPA	Georgetown City Hall, a
		landmark (25.12)
	111303	Adds § 4.36.155; amends §
	ma to o	4.36.110(A); repeals §
٩	, 8 8 0 0	4.36.150, City Employees'
Ĺ		Retirement System (4.36)
	111309	Floating home moorage
		eviction procedures
L		(Expired)
	111310	Amends § 16.40.010,
Ļ		anchorages (16.40)
	111315	Rezone, Plat 78E, zoning
Ĺ		map page 184 (Special)
	111349	Amends § 16.32.050,
		towing of unseaworthy
ļ		vessels (16.32)
	111373	Amends §§ 4.72.120 and
		4.72.130, travel expenses
Ĺ		(Repealed by 117242)

eference only.

amending
graphics,
ccuracy of

Text provided for

See ordinan sections for and tables and this source file.

Adds §§ 23.44.72; adds (D) and (F) to and renumbers (D) to be (E) of § 23.45.06; amends § 23.86.18(E) and (F) and adds (G)-(I); renumbers § 23.94.10(F) to be (F) and (G); and amends §§ 23.04.10(D), 23.24.40, 23.40.04, 23.40.20(C), 23.44.08, 23.44.10(D), 23.44.14(B) and (D), 23.44.16, 23.44.26, 23.44.42, 23.44.80, 23.44.82, 23.45.14(A) and (C), 23.45.16(A) and (B), 23.45.18(B) and (C),

23.45.26(C), 23.45.28, 23.45.30(A) and (B), 23.45.32(B) and (C), 23.45.38(A), 23.45.40(C), 23.45.42(A) and (C), 23.45.44(A) and (B), 23.45.46(B) and (C), 23.45.54, 23.45.46(A) and (C), 23.45.58(A) and (B), 23.45.60(B) and (C), 23.45.70, 23.45.76(B) and (C), 23.45.124, 23.45.168, 23.45.180, 23.45.182, 23.45.184, 23.45.190, 23.54.10(A) and (B), 23.54.20(C) and (E), 23.54.30, 23.84.04, 23.84.12, 23.84.14, 23.84.24, 23.84.26, 23.84.32, 23.84.36, 23.86.02(B), 23.86.10(A) and (B), 23.86.14(B) and (C), 23.86.16(A) and 23.94.20, Land Use Code,

24.62.140, zoning (23.04, 23.24, 23.40, 23.44, 23.45, 23.84, 23.86) 111392 Adds § 4 to Ord. 11309, floating home moorages (Expired) 111395 Adds Ch. 3.114 (§§ 3.114.010-3.114.030), financing of industrial development facilities

(3.118)

and amends §§ 24.62.120(A) and

111396 Adds Ch. 3.116 (§§ 3.116.010-3.116.090), Industrial Development Corporation (3.116) 111401 Closes Franchise Deposit Fund and transfers assets (Repealer)

For current SMC, the Office of the City Clerk

ference only.

June 2009 co

Text provide

See ordinant sections for and tables a this source

a anda	te file
111405	Amends § 15.62.040, street
	vacation procedures
- 0 - m hi	(15.62)
111407	Amends § 1 of Ord.
	111184, floating home
	moorages (Repealed by
	111526)
111412	Closes Bond and Coupon
	Transit Fund and transfers
- co orea	assets (Repealer)
111413	Amends § 7.20.030,
	floating home moorages
	(Repealed by 111526)
111422	Replaces and amends Ch.
	22.900, 1984 permit fee
1 4 - 06	ordinance; repeals Ords.
and to co	110880 and 111119
Her or	(Repealed by 112015)
111425	Adds § 21.28.370 and
	amends §§ 21.28.030,
TILE	21.28.040 and
	21.28.040 and (B),
	sewer utility rates (21.28)
111427	Adds § 5.44.075 and
11142/	
	amends § 5.44.076,
	business and occupation
111428	tax (5.44)
111428	Amends § 5.48.150, utility
111100	tax (5.48)
111429	Amends § 5.44.040,
	business and occupation
	tax (5.44)
111430	Adds § 5.44.205 and
	amends §§ 5.44.140 and
	5.44.180, business and
	occupation tax (5.44)
111431	Amends § 5.48.140,
	business tax on utilities
	(5.48)
111432	Amends § 5.48.050,
	business tax on utilities
	(5.48)
111433	Amends § 17.16.025,
	Seattle Center rental
	agreements (Repealed by
	112573)
111434	Amends § 20.48.070,
	public works contracts
	(Repealed by 117390)
111435	Closes Design Commission
	Operating Fund; transfers
	assets and liabilities;
	repeals § 9 of Ord. 96897
	(Repealer)
111436	Amends § 15.62.030, street
	vacation procedures
	(15.62)
111438	Amends § 21.76.040,
111430	utility credits for the
111430	utility credits for the elderly (21.76)

d amending graphics, accuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinances comsections for comand tables and tables and this source file.

111440 Amends §§ 21.04.430, 21.04.440; repeals § 21.12.030 and Ord. 56463 § 1, water rates (21.04) Repeals Chs. 6.06, 6.26, 6.34, 6.52 and 6.210 and all sections of these chapters: repeals §§ 206-209, 267-270.1 and 331-331.8 of Ord. 48022; §§ 5.020, 5.024, 5.030, 5.035, 5.072, 5.140, 5.170, 5.200 and 5.210-5.230 of Ord. 108934; and repeals amending Ords. 4994, 61818, 65313, 75088, 76248, 78246, 88789, 88837, 89048, 89733, 91281, 92507, 92871, 100478, 100545, 102205, 102636, 106063, 107157, 109348 and 110888 on various business licenses (Repealer) 111443 Amends § 21.24.040, sewer connection fees (21.24)111444 Adds § 10.03.160, tuberculin test fees (10.03) 111445 Amends §§ 10.03.140 and 10.03.150, immunization fees (10.03) 111446 Amends § 10.03.130, health and dental care fees (10.03)Amends § 10.03.045, 111447 medical laboratory fees (10.03)111448 Adds § 5.44.110(F), business and occupation tax (5.44) 111449 Adds § 5.40.025; amends §§ 5.40.020 and 5.40.080(A); renumbers § 5.40.080(B) to be 5.40.085 and adds subsections (A) and (B) to § 5.40.085, admissions tax (5.40) 111458 Adds §§ 25.08.425 and 25.08.535; amends §§ 25.08.080, 25.08.530 and 25.08.540, noise control (25.08)111460 Adds § 22.700.035 and amends §§ 22.702.200, 22.704.060, 22.706.010, 22.708.010, 22.708.020 and 22.708.060, major

ference only.

amending

raphics,

curacy of

For current SM (12500) Control

The Office of the City Clerk

projects energy efficiency analysis (Repealed by

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipal oct		
	late file	
111462	Amends § 10.64.060,	
	no-smoking restrictions	
-0-41	(10.64)	
111484	Amends § 23.45.182(D)	
	and (E), Land Use Code	
	(Repealed by 120293)	
111493	Adds § 4.36.135 and	
	amends §§ 4.36.130(F),	
	(G) and (H) and	
oos cre	4.36.140(A), employees'	
469 69	retirement system (4.36)	
111505	Amends § 2 of Ord.	
	111184, floating home	
Comple.	moorages (Repealed by	
	111526)	
111507	Extends moratorium set by	
ma to	Ord. 111309, floating	
	home moorages (Expired)	
111509	Amends § 23.68.03, Land	
Sile.	Use Code (23.68)	
111526	Floating home moorage	
	evictions; repeals Ords.	
	109280, 109986, 111184	
	and 111413 (7.20)	
111537	Amends § 20.46.060(A)(4)	
	and (5) and adds (A)(6)	
	and (7), women's and	
	minority businesses (20.46)	
111550	Amends § 21.76.040,	
	utility credits for the	
	elderly (21.76)	
111568	Adds Ch. 23.70 (§§	
	23.70.10-23.70.120) on	
	greenbelt overlay districts,	
	and adds § 23.76.06(H);	
	amends §§ 23.34.28(H),	
	23.76.14(E) and 23.84.14;	
	renumbers § 23.90.24(B) to	
	be (C) and adds new	
	subsection (B); repeals Ch.	
	23.68, Land Use Code	
111577	(Repealed by 113079) Amends § 20.40.140(D),	
1113//	local improvement district	
	assessments (20.40)	
111579		
1113/9	Imposes controls on St. James Cathedral, Rectory	
	and site, a landmark	
	(25.32)	
111592	(25.32) Amends § 4.04.030(29),	
111582		
111585	personnel system (4.04)	
111363	Amends § 22.900.240(F)	
	and (H), building permit	
	fees (Repealed by 112015)	

eference only.

d amending
graphics,
accuracy of

Seattle Municipal Code
June 2009 co 111589 up date
June 2009 co 111589 up date
Text provide d for hi
1205(a)
1807(l)
2104, 3

See ordinant sections for and tables a this source

	- 40	te file
L	111589	Amends Building Code
	111369	Supplement § 511(b),
	4-3	Tables 5-A and 5-E, §§
þ	U IOI	602(a), 702(a), 801, 802(c),
ľ		1001, 1201, 1202(b), 1204,
		1205(a), 1715(c), 1717(b),
		1807(1), (m) and (p), 2004,
		2104, 3305(a), 3310(b),
		Table 33-A, §§ 3401,
l	oc crea	3802(d), 4506, 4907,
	CES OF	5114(g) and 5116 as
		adopted by § 22.100.010,
	- 400 40 16	building code (Not
	COLLINA	codified)
Γ	111590	Adds §§ 23.44.80(I) and
	1 4 - 06	23.44.82(F); amends §
l	and to co	23.44.46, Land Use Code
		(23.44)
f	111591	Adds (E) to § 23.45.190
		and amends §
	7115.	23.45.146(D), Land Use
		Code (23.45)
ŀ	111615	
	111013	Amends entirety of Ch.
		21.49, Seattle City Light
		Department; repeals Ch.
		21.48 and Ords. 109218,
		109339, 109621 and
L		110529 (21.49)
	111620	Amends § 4.28.020,
L		funeral leave (4.28)
	111621	Amends § 4.34.020,
		vacations for City
L		employees (4.34)
	111623	Rezone, Plat 39E, zoning
		map page 109 (Special)
Ī	111626	Creates Residual Cash
		Investment Fund (5.76)
f	111634	Amends §§ 10.13.210(A),
	7	10.13.220(E) and
		10.13.230, meat package
		labels (10.13)
ŀ	111639	Establishes Molly Mathews
	111037	Memorial Account in
		General Donations and Gift
ŀ	111740	Trust Fund (5.76)
	111640	Amends §§ 20.04.130,
		20.04.200, 20.12.040 and
		20.12.080(A), local
١		improvement procedures
l		(20.04, 20.12)
ſ	111644	Adds § 11.16.319 and
		amends § 11.16.315,
		restricted parking zones
		(11.16)
ŀ	111649	Adds Ch. 10.64, pet
1	111047	kennels; repeals Ch. 6.18
ı		
		(10.72)

reference only.

(c), 04, 0), 04, 1d amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipai			
and!	te file		
111650	Adds § 21.16.065 and		
	amends §§ 21.16.030,		
a com bi	21.16.060, 21.16.070,		
d tor in	21.16.140, 21.16.280 and		
	21.24.020, side sewers and		
	sewer connection charges		
111656	(21.16)		
111656	Adds Art. 78A to and repeals Art. 78 of Uniform		
- 01103	Fire Code, 1982 Edition,		
ces crea	adopted in § 22.600.020,		
	fireworks (Repealed by		
	114328, 115405)		
111661	Amends §§ 21.40.050,		
	21.40.060 and		
ad to C	21.40.120(A), solid waste		
and co	collection rates and charges (21.40)		
111662	Amends §§ 21.40.010 and		
	21.40.080, solid waste		
111c.	collection rates and charges		
	(21.40)		
111674	Adds §§ 3.110.070(D) and		
	3.110.275, public		
111675	corporations (3.110)		
1110/3	Amends § 21.76.040, utility credits for the		
	elderly (21.76)		
111702	Adds §§ 23.84.30,		
	23.45.106, 24.08.170,		
	24.48.050, 24.50.130(F),		
	24.52.100(D) and		
	24.58.080(F); amends § 24.36.030, Land Use Code		
	and zoning (23.45, 23.84)		
111712	Imposes controls on United		
	States Assay		
	Office/German House, a		
	landmark (25.32)		
111714	Adds § 12A.06.160,		
	malicious harassment (12A.06)		
111715	Amends §§ 10.64.020 and		
	10.64.060, no-smoking		
	regulations (10.64)		
111724	Adds Ch. 5.10, registration		
	of bonds and other		
111746	obligations (5.10) Implements Uniform		
111740	Unclaimed Property Act		
	providing for transfer of		
	certain property to the state		
	(3.38, 3.120)		
111751	Adds § 6.96.015 and		
	repeals § 6.96.010 and § 1		
	of Ord. 71614, business owner identification (6.96)		
111756	Amends § 4.44.020,		
111/30	disability compensation		
	(4.44)		

reference only.

I amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	4- 6110
a anda	Te ille
111757	Adds §§ 4.34.005,
	4.34.055, 4.34.065 and
1 500 M	4.34.070; amends §§ 4.34.010, 4.34.020 and
ad ros	4.34.045; repeals §
	4.34.050, annual vacations
	for City employees (4.34)
111763	Amends § 1 of Ord.
	110026, restricted parking
oc crea	zone in vicinity of
ices or	Fauntleroy Ferry Terminal
111770	(11.73)
111768	Adds § 4.20.325; repeals §
COPPER	4.20.320 and Ords. 97330 § 51(C) and 110449 § 1,
	overtime-related meal
and to co	compensation (4.20)
111782	Establishes Municipal
	Archives Program (3.122)
111783	Adds § 11.16.318,
File	restricted parking zone
	near Fauntleroy Ferry
111506	Terminal (11.16)
111786	Amends § 4.72.010, travel
111787	expenses (4.72) Establishes policy of
111/8/	allowing recognition for
	volunteers donating time
	and services to the City
	(4.96)
111793	Amends § 21.24.040,
	sewer connection charges
111017	(21.24)
111817	Amends § 21.18.260,
111818	sewer charges (21.28) Amends § 21.40.120(B),
111010	solid waste collection
	charges (21.40)
111829	Amends § 3.14.806, Office
	of Policy Planning (3.18)
111834	Amends § 1 of Ord.
	111649, pet kennel licenses
111007	(10.72)
111835	Amends Ord. 108200 § 2,
	vehicle impoundment and § 11.72.065, disabled
	parking (11.30, 11.72)
111842	Adds §§ 16.20.230 and
	16.20.240, vessel and
	watercraft operation
	regulations (Repealed by
	120023)
111844	Amends subsection (c) of §
	802 of the Seattle Building
	Code Supplement (Not
111853	codified) Amends § 12A.04.100,
111033	criminal liability (12A.04)
	Cimilal natinty (12A.04)

reference only.

Is a mending graphics, so ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Shar	0010
	a anda	te file
	111857	Adds §§
		12A.06.155-12A.06.180
	a com hi	and amends § 12A.06.120,
	4 for 112	domestic violence
		(12A.06)
	111858	Adds § 12A.06.300,
		amends §§ 12A.02.070 and
		12A.06.150 and repeals §
		12A.06.070, Criminal
	- or orea	Code (12A.02, 12A.06)
	111859	Adds § 11.58.425; amends
		§§ 11.31.050, 11.34.020,
		11.56.020, 11.56.120 and
	COMPA	11.58.420; and repeals §
<b>b</b>		11.56.130, Traffic Code
	440.00	(11.31, 11.34, 11.56,
	and to co	11.58)
	111860	Adds § 12A.12.040,
		disorderly conduct on
	2110	buses; amends §
	1110.	12A.10.030, marijuana;
		and repeals §§ 12A.06.080,
		12A.10.030, 12A.20.010,
		12A.20.020 and
		12A.20.040, Criminal
		Code (12A.10, 12A.12)
	111861	Adds §§
		11.56.410-11.56.460,
		hit-and-run accidents and §
		11.58.195, child passenger
		restraint, and repeals §§
		11.52.180, 11.56.180,
		11.56.200, 11.56.220,
		11.56.240, 11.56.260,
		11.56.280 and 11.56.300,
		traffic (11.56, 11.58)
	111866	Adds Ch. 25.05,
		environmental policies and
		procedures; repeals Ch.
		25.04 (25.05)
	111874	Amends §§ 25.28.220 and
		25.28.230(B), Pioneer
		Square Historical District
		(25.28)
	111881	Designates St.
		Nicholas/Lakeside School
		as a landmark (25.32)
	111882	Designates the Good
		Shephard Center as a
		landmark (25.32)
	111883	Designates the Times
		Square Building as a
		landmark (25.32)
	111884	Designates the Georgetown
		Steam Plant as a landmark
		(25.32)
	111885	Designates the
		Fisher/Howell House as a
		landmark (25.32)

reference only.

nd amending
is graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Designates the Stuart Balcom House and Gardens as a landmark (25.32)	C	Than	. c:10
Balcom House and Gardens as a landmark (25.32)  Designates the Bowen/Huston Bungalow as a landmark (25.32)  111888  Designates the Wallingford Fire and Police Station as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  Amends §§ 3.102.010, 4.20.190 and 11.14.490, Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  Amends § 4.04.250, Civil Service Commission (4.04)  111892  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds §§ 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.20; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.040, 6.212.100, 6.212.100, 6.212.110, 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.24, 23.84.02, 23.84.14, 23.84.16, 23.84.10, 23.84.12, 23.84.10, 23.84.12, 23.84.18, 23.84.02, 23.84.28, 23.84.28, 23.84.28, 23.84.30, 23.84.38, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.28, 23.84.38, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.28, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.28, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.48, and 23.86.06, Land Use Code (23.84, 44, 23.84.48, and 24.84, 2		a anda	te Ille
Gardens as a landmark (25.32)  Designates the Bowen/Huston Bungalow as a landmark (25.32)  Designates the Wallingford Fire and Police Station as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  I11890  Amends §§ 3.102.010, 4.20.190 and 11.14.490, Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  Amends § 4.04.250, Civil Service Commission (4.04)  I11895  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  I11904  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds §§ 6.208.090, 6.208.100 and 6.208.10 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  I11906  Adds §§ 6.212.115-6.212.119 and 6.212.040, 6.212.060, 6.212.100, 6.212.080, 6.212.110, 6.212.100, 6.212.100 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  I11926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30, amends §§ 23.54.24, 23.86.24, 23.84.25, 23.84.24, 23.84.06, 23.84.00, 23.84.10, 23.84.10, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.42, 23.84.38, 23.84.30, 23.84.30, 23.84.32, 23.	1	11886	Designates the Stuart
111887   Designates the Bowen/Huston Bungalow as a landmark (25.32)			
Designates the Bowen/Huston Bungalow as a landmark (25.32)		s com hi	
Bowen/Huston Bungalow as a landmark (25.32)  Designates the Wallingford Fire and Police Station as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  Amends §§ 3.102.010, 4.20,190 and 11.14.490, Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  Ill892  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds § 6.208.090, 6.208.100 and 6.208.110 and amends § 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.20c; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  Adds Ch. 23.49 and §§ 23.54.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.24, 23.86.24, 23.84.02, 23.84.10, 23.84.10, 23.84.10, 23.84.10, 23.84.10, 23.84.10, 23.84.10, 23.84.10, 23.84.12, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.20, 23.84.32, 23.84.32, 23.84.32, 23.84.32, 23.84.32, 23.84.32, 23.84.32, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.84.40, 23.84.40, 23.84.42, 23.	04	A FOL 333	
as a landmark (25.32)  Designates the Wallingford Fire and Police Station as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  II1890  Amends §§ 3.102.010, 4.20.190 and 11.14.490, Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  II1892  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds §§ 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  II1906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.200; amends §§ 6.212.010, 6.212.000, 6.212.000, 6.212.000, 6.212.000, 6.212.000, 6.212.100, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  II1926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.24, 23.86.24, 23.86.24, 23.86.24, 23.86.24, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.06, 23.84.02, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.38, 23.84.40, 23.84.22, 23.84.40, 23.84.22, 23.84.40, 23.84.22, 23.84.40, 2	1	11887	
Designates the Wallingford Fire and Police Station as a landmark (25.32)			S
Fire and Police Station as a landmark (25.32)  Designates the Hoge Building as a landmark (25.32)  Amends §§ 3.102.010, 4.20.190 and 11.14.490, Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds S § 6.208.090, 6.208.100 and 6.208.110 and amends § 8.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  Adds § 6.212.115-6.212.119 and 6.212.190-6.212.220; amends § 6.212.010, 6.212.040, 6.212.060, 6.212.040, 6.212.060, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.10, 23.84.16, 23.84.18, 23.84.16, 23.84.18, 23.84.16, 23.84.18, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.38, 23.84.40, 23.84.42, 23.84.38, 23.84.40, 23.84.42, 23.8	<u> </u>	11000	
Indimark (25.32)   Designates the Hoge Building as a landmark (25.32)	1	11888	
Designates the Hoge Building as a landmark (25.32)			
Building as a landmark (25.32)  Amends §§ 3.102.010, 4.20.190 and 11.14.490, Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  I11892  Amends §§ 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds §§ 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  I11906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  I11926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.24, 23.86.24, 23.84.06, 23.84.10, 23.84.10, 23.84.10, 23.84.10, 23.84.14, 23.84.16, 23.84.14, 23.84.16, 23.84.14, 23.84.16, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.22, 23.84.30, 23.84.24, 23.84.24, 23.84.28, 23.84.30, 23.84.32, 23.84.32, 23.84.32, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.44, 23.84.42, 23.84.44, 23.8	1	11000	
(25.32)   Amends §§ 3.102.010,   4.20.190 and 11.14.490,   Martin Luther King, Jr.'s birthday holiday (3.102,   4.20, 11.14)     111892   Amends § 4.04.250, Civil Service Commission (4.04)     Amends subsection (D) of   \$20.04.140, local improvement procedures (20.04)     111904   Adds Ch. 6.211 and amends § 6.204.050,   for-hire vehicles (6.204)     Adds §§ 6.208.090,   6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)     111906   Adds §§ 6.212.115-6.212.119 and   6.212.190-6.212.220;   amends §§ 6.212.010,   6.212.040,   6.212.060,   6.212.070,   6.212.080,   6.212.100,   6.212.100 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)     111926   Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.18, 23.84.20, 23.84.18, 23.84.20, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.42, 23.84.30, 23.84.42, 23.84.30, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.60, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.4	1	11889	
4.20.190 and 11.14.490, Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  111892  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  111904  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds § 6.208.090, 6.208.100 and 6.208.110 and amends § 8 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds § 6.212.115-6.212.119 and 6.212.190-6.212.220; amends § 6.212.010, 6.212.040, 6.212.060, 6.212.100, 6.212.040, 6.212.100, 6.212.100, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and § § 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends § § 23.54.24, 23.86.24, 23.84.04, 23.84.10, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.38, 23.84.30, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.38, 23.84.30, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.42, 23.84.38, 23.84.30, 23.84.42, 23.84.44, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.44, 23.84.44, 23.84.44, 23.84.44, 23.84.44, 23.84.44, 23.84.44, 23.84.44, 23.84.44		mnle	(25.32)
Martin Luther King, Jr.'s birthday holiday (3.102, 4.20, 11.14)  111892  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  111904  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  111905  Adds § 6.208.090, 6.208.100 and 6.208.110 and amends § § 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds § 8  6.212.115-6.212.119 and 6.212.190, 6.212.060, 6.212.040, 6.212.060, 6.212.100, 6.212.040, 6.212.100, 6.212.100, 6.212.100 and 6.201.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and § 23.54.24, 23.86.26, 23.86.28 and 23.86.30; amends § 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends § 23.54.24, 23.86.26, 23.84.06, 23.84.06, 23.84.06, 23.84.10, 23.84.10, 23.84.10, 23.84.16, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.24, 23.84.26, 23.84.28, 23.84.30, 23.84.24, 23.84.26, 23.84.28, 23.84.30, 23.84.24, 23.84.26, 23.84.28, 23.84.30, 23.84.24, 23.84.26, 23.84.38, 23.84.40, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.48, 23.84.20, 23.84.42, 23.84.48, 23.84.20, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84, 10.11)		11890	
birthday holiday (3.102, 4.20, 11.14)  111892  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  111904  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds § 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds § 6.212.115-6.212.119 and 6.212.190-6.212.220; amends § 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.110, 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and § § 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends § § 23.54.24, 23.86.24, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.02, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.24, 23.84.28, 23.84.30, 23.84.42, 23.84.44, 23.84.06, Land Use Code (23.84, 10.10)			
4.20, 11.14)  111892  Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  111904  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  111905  Adds §§ 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.040, 6.212.060, 6.212.106 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.10, 23.84.10, 23.84.14, 23.84.16, 23.84.14, 23.84.16, 23.84.14, 23.84.26, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.28, 23.84.40, 23.84.24, 23.84.26, 23.84.36, 23.84.42, 23.84.26, 23.84.36, 23.84.42, 23.84.26, 23.84.36, 23.84.24, 23.84.26, 23.84.28, 23.84.40, 23.84.28, 23.84.40, 23.84.24, 23.84.26, 23.84.28, 23.84.40, 23.84.26, 23.84.38, 23.84.40, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.4		ad to co	
Amends § 4.04.250, Civil Service Commission (4.04)  Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  111904 Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds § 6.208.090, 6.208.100 and 6.208.110 and amends § 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906 Adds § 8  6.212.115-6.212.119 and 6.212.190-6.212.220; amends § 6.212.010, 6.212.040, 6.212.060, 6.212.040, 6.212.080, 6.212.106 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926 Adds Ch. 23.49 and § \$ 23.54.24, 23.86.26, 23.86.28 and 23.86.30; amends § \$ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.06, 23.84.06, 23.84.06, 23.84.08, 23.84.14, 23.84.16, 23.84.14, 23.84.16, 23.84.14, 23.84.25, 23.84.24, 23.84.26, 23.84.28, 23.84.20, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.28, 23.84.40, 23.84.42, 23.84.44, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.49, 23.84.40, 23.84.42, 23.84.49, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.49, 23.84.49, 23.84.40, 23.84.42, 23.84.49, 23.84.40, 23.84.42, 23.84.49, 23.84.49, 23.84.40, 23.84.42, 23.84.49, 23.84.40, 23.84.42, 23.84.49, 23.84.40,		ild co	
Service Commission (4.04)	1	11902	, ,
Amends subsection (D) of § 20.04.140, local improvement procedures (20.04)  111904 Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  111905 Adds §§ 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906 Adds §§ 6.212.119 and 6.212.119-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.100, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926 Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.10, 23.84.10, 23.84.10, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.20, 23.84.24, 23.84.28, 23.84.20, 23.84.28, 23.84.20, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.20, 23.84.24, 23.84.24, 23.84.26, 23.84.28, 23.84.20, 23.84.24, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.20, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.24, 23.84.42, 23.84.48, and 23.86.06, Land Use Code (23.84,		11074	
\$ 20.04.140, local improvement procedures (20.04)  111904 Adds Ch. 6.211 and amends \$ 6.204.050, for-hire vehicles (6.204)  111905 Adds \$\$ 6.208.090, 6.208.100 and 6.208.110 and amends \$\$ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906 Adds \$\$ 6.212.119 and 6.212.190-6.212.220; amends \$\$ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926 Adds Ch. 23.49 and \$\$ 23.54.24, 23.86.26, 23.86.28 and 23.86.30; amends \$\$ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.10, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.28, 23.84.20, 23.84.28, 23.84.20, 23.84.28, 23.84.20, 23.84.28, 23.84.20, 23.84.28, 23.84.20, 23.84.24, 23.84.28, 23.84.20, 23.84.28, 23.84.20, 23.84.24, 23.84.48, 23.84.20, 23.84.42, 23.84.48, 23.84.42, 23.84.48, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.40, 23.84.42, 23.84.48, 23.84.48, 23.84.48, 23.84.49, 23.84.48, 23.84.49, 23.84.48, 23.84.49, 23.84.48, 23.84.48, 23.84.49, 23.84.48	1	11895	
(20.04)  Adds Ch. 6.211 and amends § 6.204.050, for-hire vehicles (6.204)  Adds §§ 6.208.090, 6.208.100 and 6.208.040, for-hire vehicles (6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.10, 23.84.16, 23.84.18, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.20, 23.84.24, 23.84.26, 23.84.28, 23.84.20, 23.84.24, 23.84.26, 23.84.28, 23.84.30, 23.84.24, 23.84.26, 23.84.28, 23.84.30, 23.84.32, 23.84.38, 23.84.40, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23.84.42, 23.84.44, 23	1	110/3	§ 20.04.140, local
amends § 6.204.050, for-hire vehicles (6.204)  111905  Adds §§ 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.38, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
for-hire vehicles (6.204)  Adds §§ 6.208.090, 6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.040, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.24, 23.86.24, 23.84.04, 23.84.02, 23.84.04, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.24, 23.84.30, 23.84.24, 23.84.30, 23.84.38, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,	1	11904	
Adds § § 6.208.090, 6.208.100 and 6.208.110 and amends § § 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds § § 6.212.115-6.212.119 and 6.212.190-6.212.220; amends § § 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and § § 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends § § 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.16, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			-
6.208.100 and 6.208.110 and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
and amends §§ 6.208.030 and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,	1	11905	,
and 6.208.040, for-hire vehicles (Repealed by 118341)  111906  Adds §\$ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §\$ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §\$ 23.54.30, 23.84.02, 23.84.06, 23.84.08, 23.84.10, 23.84.14, 23.84.16, 23.84.14, 23.84.16, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
vehicles (Repealed by 118341)  Adds §\$ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.06, 23.84.08, 23.84.10, 23.84.14, 23.84.16, 23.84.14, 23.84.16, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.24, 23.84.30, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.44, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
118341)  Adds §\$ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			· · · · · · · · · · · · · · · · · · ·
Adds §§ 6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
6.212.115-6.212.119 and 6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,	1	11906	
6.212.190-6.212.220; amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,	1	11700	
amends §§ 6.212.010, 6.212.040, 6.212.060, 6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
6.212.070, 6.212.080, 6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §\$ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §\$ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
6.212.110, 6.212.120, 6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			6.212.040, 6.212.060,
6.212.160 and 6.212.170, taxicabs and taxicab drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.11, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			6.212.070, 6.212.080,
taxicabs and taxicab drivers (Repealed by 118341)  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.04, 23.84.10, 23.84.11, 23.84.16, 23.84.14, 23.84.16, 23.84.24, 23.84.25, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.36, 23.84.42, 23.84.42, 23.84.42, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
drivers (Repealed by 118341)  111926  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.11, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.28, 23.84.29, 23.84.32, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.40, 23.84.42, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
118341)  Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
Adds Ch. 23.49 and §§ 23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.54.24, 23.86.24, 23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,	1	11026	
23.86.26, 23.86.28 and 23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.32, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,		11740	
23.86.30; amends §§ 23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.54.30, 23.84.02, 23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.30, 23.84.38, 23.84.40, 23.84.42, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.04, 23.84.06, 23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			, 00
23.84.08, 23.84.10, 23.84.12, 23.84.14, 23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.16, 23.84.18, 23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.20, 23.84.24, 23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.25, 23.84.28, 23.84.30, 23.84.32, 23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.30, 23.84.32, 23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.36, 23.84.38, 23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.40, 23.84.42, 23.84.48 and 23.86.06, Land Use Code (23.84,			
23.84.48 and 23.86.06, Land Use Code (23.84,			
Land Use Code (23.84,			
			,
25.00)			
		COST SM	20.00)

reference only.

The lamending graphics, accuracy of

For current Sme, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

30	to file
111103	
111928	Designates the First
	African Methodist
a coss hi	Episcopal Church as a
WI TOL 180	landmark (25.32)
111929	Adds §§ 4.13.010 and
	4.13.020; and repeals Ch.
	4.13 and Ord. 111127,
	exemptions from civil
	service and public safety
- a orea	civil service system
ces crea	(Repealed by 112633)
111931	Adds § 3.18.140, City's
111,51	motor vehicle fleet (3.18)
111963	Amends § 11.18.010,
111903	
111072	arterial street map (11.18)
111973	Amends § 21.28.370,
ma to	sewer rates and charges
,800	(Repealed by 118176)
111974	Amends § 21.76.040,
5116	utility credits for the
222	elderly (21.76)
111987	Amends § 24.68.100(A),
	Pioneer Square Historic
	District (Repealed by
	112134)
111992	Adds §§ 4.20.600,
111992	4.20.610, 4.36.360,
	4.36.370 and 4.36.380,
	City employees' retirement
	system (4.20, 4.36)
111995	Adds § 22.502.015 and
	repeals § 22.502.010,
	Uniform Plumbing Code
	(Repealed by 113355)
112005	Adds §§
	2.04.060(B)(14)-(16),
	2.04.400-2.04.470 and
	2.04.500(C); and amends §
	2.04.060(B)(13), election
	campaign contributions
	(Repealed by 116162)
112008	Adds Subchapter VII to
112000	Ch. 2.04, election
	campaign contributions
	(2.04)
112015	Repeals and replaces Ch.
112013	22.900, 1985 permit fees
	1 22.900. 1985 permit lees
	(Repealed by 112547)
112018	(Repealed by 112547) Amends § 20.48.070,
112018	(Repealed by 112547) Amends § 20.48.070, public works contracts
	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390)
112018 112019	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390) Amends § 5.44.180,
	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390)
	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390) Amends § 5.44.180, business and occupation
112019	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390) Amends § 5.44.180, business and occupation tax (5.44)
	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390) Amends § 5.44.180, business and occupation tax (5.44) Amends § 5.48.050, utility
112019	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390) Amends § 5.44.180, business and occupation tax (5.44) Amends § 5.48.050, utility business and occupation
112019 112021	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390) Amends § 5.44.180, business and occupation tax (5.44) Amends § 5.48.050, utility business and occupation tax (5.48)
112019	(Repealed by 112547)  Amends § 20.48.070, public works contracts (Repealed by 117390)  Amends § 5.44.180, business and occupation tax (5.44)  Amends § 5.48.050, utility business and occupation tax (5.48)  Amends § 5.48.020, utility
112019	(Repealed by 112547) Amends § 20.48.070, public works contracts (Repealed by 117390) Amends § 5.44.180, business and occupation tax (5.44) Amends § 5.48.050, utility business and occupation tax (5.48)

reference only.

d amending

graphics,

accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te file
	112024	Amends § 10.11.941, Food
		Code (Repealed by
Ĺ	40-40 101	117001)
•	112025	Amends § 10.03.150,
۲		Health Code fees (10.03)
	112026	Amends § 10.03.130,
ļ	112027	Health Code fees (10.03)
	112027	Amends §§ 21.44.020,
		21.44.030 and 21.44.040,
	DES CITE	solid waste disposal facilities (Repealed by
77		114723)
ŀ	112028	Amends § 25.08.570, noise
	112028	control (25.08)
-	112029	Amends §§ 5.44.020,
	11202)	5.44.050, 5.44.076,
	and to co	5.44.110, 5.44.200 and
•	1200	5.44.210, business and
		occupation tax (5.44)
Ī	112031	Amends § 17.16.025,
	222-	Seattle Center rental
		agreements (Repealed by
Ļ		112573)
	112032	Amends § 21.24.020,
		sewer permit fees and
		connection charges
ŀ	112034	(Repealed by 116421) Creates Water Construction
	112034	Fund (5.76)
f	112035	Amends §§ 21.04.060,
	112033	21.04.100, 21.04.110 and
		21.04.120, water rates and
		regulations (21.04)
ſ	112039	Amends § 5.32.170,
Ĺ		Revenue Code (5.32)
	112043	Adds § 20.46.140,
		women's and minority
		business utilization
ŀ	112044	(Repealed by 114082)
	112044	Amends § 3.14.806, Executive Department
		(3.18)
ŀ	112052	License fees for
	· -	amusement devices (Not
		codified)
ſ	112072	Amends §§ 3.90.010,
		3.90.020, 3.90.040,
		3.90.050 and 3.90.060;
		repeals § 3.90.030, officers'
-	112076	bonds (3.90) Amends §§ 11.14.490,
	1120/0	Amends §§ 11.14.490, 11.72.260, 11.72.300,
		11.72.200, 11.72.300, 11.72.435 and 11.74.020,
		Traffic Code (11.14, 11.72,
		11.74)
ŀ	112086	Adds § 24.08.140(M)(2)
		and renumbers former
		(M)(2) to be (M)(3); adds §
		24.54.030(G), zoning
Ĺ		(Repealed by 117570)

eference only.

d amending
graphics,
ccuracy of

For current SMC, (Repealed by 117570) tot the Office of the City Clerk

June 2009 co

Text provid

See ordinan sections for and tables a this source

	ilha.	0210	
		te tille	
	112087	Amends §§ 3.28.100 and	Campaice Oll
U	<del>(1)</del>	3.28.120(B), Department	Stelemen
	1-1	of Administrative Services	
		(Repealed by 116007)	
JE	112088	Adds §§ 4.24.005 and	
		4.24.035 and amends §	10 - 4
		4.24.040, employee sick	
		leave (4.24)	amelius
	112089	Designates University	
	oc crea	Presbyterian Church "Inn"	- bios.
	CE2 C2	as a landmark (25.32)	drabilles,
	112092	Adds §§ 11.23.210,	Br orb
	oomnie	11.23.250, 11.23.290,	TO VED OF
1	Comple	11.60.050, 11.60.130,	celleacy
		11.60.370 and 11.60.587;	
	ad to co	amends §§ 11.14.185,	
4		11.23.040, 11.23.060,	
		11.23.320, 11.23.340,	
	0210	11.60.060, 11.60.160, 11.60.240, 11.60.380,	
	1116.	11.60.420, 11.60.480,	
		11.60.520, 11.60.583,	
		11.60.589 and 11.60.660;	
		repeals §§ 11.23.200,	
		11.23.240, 11.23.260,	
		11.23.280, 11.23.300,	
		11.60.040, 11.60.080,	
		11.60.100, 11.60.120,	
		11.60.260, 11.60.280,	
		11.60.300, 11.60.320,	
		11.60.360, 11.60.400,	
		11.60.440 and 11.60.586,	
		Traffic Code (11.14, 11.23,	
	112122	11.60)	
	112102	Rezone, Plat 22W (Special)	
	112103	Adds §§ 12A.14.035,	
		12A.14.055 and 12A.14.085; amends §§	
		12A.14.010, 12A.14.020	
		and 12A.14.040; repeals §§	
		12A.14.030 and	
		12A.14.050, Criminal	
		Code (12A.14)	
	112107	Adds § 10.03.170, amends	
		§ 10.64.020(B) and repeals	
		§ 9.26.030, pet kennel	
		permits (10.03, 10.64)	
	112111	Amends § 5.48.020,	
		business taxes for utilities	
	110110	(5.48)	
	112113	Amends §§ 11.72.435 and	
		11.72.440, Traffic Code	
	112114	(11.72) Amends §§ 5.44.020,	
	112114	Amends §§ 5.44.020, 5.44.200 and 5.33.210,	
		business and occupation	
		tax (5.44)	
	112120	Amends § 25.08.570, noise	
	112120	control (25.08)	
J		(_2,00)	of

eference only. amending graphics,

June 2009 co

Text provide

See ordinant sections for and tables a this source

	cipal cour		
	- mada	te file	
	112134	Adds Subchapters VII, VIII	
		and IX, on Interim Pioneer Square Mixed, Interim	
	a sor m	International District	
	CI Tos	Mixed and Interim	
		International District	
		Residential zones,	
		respectively, to Ch. 23.49;	
		amends §§ 23.30.010, 23.49.04, 23.49.05,	
	res crea	23.54.24, 23.84.02,	
_\		23.84.16, 23.84.24,	
		23.84.25, 23.84.30 and	
	COMPA	23.84.38; repeals Ch.	
		24.68, Land Use Code	
	112127	(23.30, 23.84)	
4	112137	Adds Ch. 5.78, Gift Catalogue in General	
		Donations and Gift Trust	
	C:10.	Fund (5.78)	
	112149	Rezone, Plat 43E (Special)	
	112171	Amends §§ 21.36.010(I)	
		and (N), 21.36.020 and	
		21.36.080, solid waste collection (21.36)	
	112179	Adds subsection (B)(7) to §	
	112179	24.08.030 and renumbers	
		(B)(7)-(20) to be	
		(B)(8)-(21); amends §§	
		24.44.050, 24.46.030 and	
		24.48.030, zoning and	
		subdivisions (Repealed by 117570)	
	112180	Rezone, Plat 232E	
		(Special)	
	112203	Adds §§ 12A.14.120,	
		12A.14.130 and	
		12A.14.135, firearms banned on premises selling	
		liquor for consumption	
		(Repealed by 117157)	
	112212	Repeals and replaces §	
		24.64.030 and Plates I, IIa,	
		IIb and III, zoning (Repealed by 117570)	
	112213	Adds Ch. 2.14, election	
	112213	pamphlets for primary and	
		general elections (2.14)	
	112238	Adds §§ 12A.14.140 and	
		12A.14.150, firearms	
		restrictions (Repealed by 116872)	
	112258	Amends § 4 of Ord.	
		112042, moratorium on	
		withdrawal of certain rental	
		units from rental market	
	112250	(Not codified)	
	112259	Amends § 23.54.30(E), Land Use Code (Repealed	
		by 112777)	

eference only.

Id amending

graphics,

accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipai			
a mada	te file		
112261	Amends § 11.76.015,		
112265	parking (11.76)		
112265	Approves changes to Official Land Use Map		
0 101	pages 36E and 40E		
	(Special)		
112271	Imposes controls on		
112271	McGraw Square (McGraw		
	Place), a Landmark (25.32)		
112272	Imposes controls on The		
ces er en	Eagles Temple Building, a		
	Landmark (25.32)		
112273	Imposes controls on the		
Comple	statue, "Seattle, Chief of		
	Suquamish," a Landmark		
1 40 66	(25.32)		
112274	Imposes controls on Queen		
	Anne High School, a		
112275	Landmark (25.32) Imposes controls on the		
1122/3	Decatur Building, a		
	Landmark (25.32)		
112277	Repeals Ord. 98708, sales		
1122,,	and use tax (Repealer)		
112291	Amends portions of §§		
	23.84.30 "P" and 24.08.170		
	"P," Land Use Code and		
	zoning (23.84)		
112293	Amends § 6.96.015,		
	business owner		
112200	identification (6.96)		
112300	Amends § 22.300.020, National Electrical Code		
	(Repealed by 114181)		
112303	Adds § 23.86.32; repeals		
112303	and adds Ch. 23.49,		
	downtown zoning (§§		
	23.49.02-23.49.344);		
	amends §§ 23.04.10(D),		
	23.54.30(E), 23.66.122,		
	23.66.140, 23.66.322,		
	23.66.324, 23.66.326,		
	23.80.20, 23.84.02, 23.84.04, 23.84.06,		
	23.84.12, 23.84.16,		
	23.84.25, 23.84.26,		
	23.84.28, 23.84.30,		
	23.84.32, 23.84.36,		
	23.84.38, 23.84.48,		
	23.86.06(D), 23.86.07 and		
	23.86.28; repeals §		
	23.54.24, Land Use Code;		
	and amends § 24.52.140,		
	zoning (23.04, 23.49,		
112204	23.66, 23.84, 23.86)		
112304	Amends § 24.82.010, Lake		
	Union construction limit line (Repealed by 117570)		
112305	Rezone, Plat 23E (Special)		
114303	rezone, i iai 23E (Special)		

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a and a	te file
	112317	Approves master plan for
	(30)	the University of
	103	Washington and rezones
	1 SOT 111	property (Special)
	112325	Rezone, amends Plat 226E
	112323	<i>'</i>
ŀ	112222	(Special)
	112333	Amends § 12A.06.030,
		menacing, and §
		12A.12.010, disorderly
	accepted	conduct (12A.06, 12A.12)
Ţ	112334	Adds §§ 3.114.150,
		3.114.160 and 20.46.095;
	- ammie	amends §§ 3.114.020(C),
	COMP	3.114.030-3.114.070,
		3.114.110 and 3.114.140,
	140 00	rosters for selection of
	ma to co	architectural and
•		engineering consultants
		(3.114, 20.46)
	112335	Adds Ch. 9.25 (§§
3	7770.	9.25.010-9.25.140) and
		repeals Ch. 9.24, animal
		control (9.25)
	112342	Housing Preservation
		Ordinance; repeals Ords.
		109220, 109342, 109973,
		110661 (Repealed by
		115141)
	112345	Amends Plat 305E, rezone
		(Special)
	112359	Amends § 11.62.060
		(Route 6), truck routes
		(11.62)
	112362	Creates General Trust Fund
		in City Treasury (5.76)
	112370	Renumbers § 20.46.050(J)
		to be (K) and adds new
		subsection (J); amends §§
		20.46.060(A)(4) and (C),
		women's and minority
		business enterprises
		(20.46)
	112373	Adds Ch. 6.294, dance
		halls admitting persons
		under the age of eighteen
	110050	(Repealed by 120889)
	112379	Amends § 25.08.520(E)
		and (F) and §
		25.08.540(A), noise control
	112202	(25.08)
	112383	Downtown Housing
		Maintenance Ordinance
	112412	(22.220)
	112412	Rezone, amends Plat 221E
		(Special)

eference only.

GE

ad amending

graphics,

accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Shar	C:10
	a anda	te tile
	112421	Amends §§ 11.16.240(G), 11.16.340(M), 1.23.040,
	a for hi	11.23.060, 11.23.290(D), 11.23.380, 11.23.400,
3	CI TO	11.30.180(C), 11.60.592,
		11.72.250, 11.72.320 and
		11.84.020(A), Traffic Code (11.16, 11.23., 11.30,
		11.60, 11.72, 11.84)
Ī	112436	Rezone, amends Plat 231E
M	GG2 G2 G2	(Special)
	112441	Amends §§ 21.49.030,
	comple	21.49.040, 21.49.050, 21.49.060, 21.49.070,
	Cosses	21.49.080, 21.49.100 and
	140 00	21.49.110, Seattle City
	nd to co	Light Department rates
	112464	(21.49) Amends § 22.900.170(I)
	112404	and Table B, and §
	1116.	22.900.220(D) and Table
		G, permit fees (Repealed
ļ	112465	by 112547)
	112465	Adds §12A.06.185 and amends §§ 12A.06.040,
		12A.06.120, 12A.06.130,
		12A.06.155, 12A.06.160,
		12A.06.165, 12A.06.170,
		12A.06.175 and 12A.06.180, harassment
		and domestic violence
		protection (12A.06)
Ī	112466	Adds §§ 11.56.365 and
		11.56.445; amends §§
		11.31.010, 11.34.020 and 11.56.020; repeals §
		11.56.310, Traffic Code
		(11.31, 11.34, 11.56)
	112467	Amends § 12A.10.010,
		prostitution loitering (12A.10)
ŀ	112488	Amends §§ 4 and 8(C) of
		Ord. 109675,
		comprehensive residential
		weatherization program
F	112497	(21.52) Rezone, amends Plat 70W
	112171	(Special)
Ī	112500	Repeals and adds Ch.
		22.700, Energy Code;
		repeals Ords. 108500, 109416 and 111460
		(22.700)
ľ	112504	Establishes Zoo Parent
		Account in General
		Donations and Gift Trust
ļ	112506	Fund (5.76) Accepts facade easement
		agreement for Washington
		Court Building (Special)

reference only.

de amending
E graphics,
accuracy of

For current SMC, Clerk the Office of the City Clerk

| Text | Code | Stablishes Transportation | Systems Management | (TSM) Account in General Trust Fund (5.76) | Adopts voluntary deferred compensation plan (Repealed by 119159) | 112519 | Amends §§ 23.04.10(B) and (D), 23.30.10, 23.49.12, 23.49.14(A), |

See ordinances comsections for comand tables and tables and this source file.

Establishes Transportation (TSM) Account in General Adopts voluntary deferred 23.49.16(A), 23.49.28(E) and (F), 23.49.36(F), 23.49.46(C), 23.49.50(B), 23.49.58(B), 23.49.66(C), 23.49.70(B), 23.49.78(B), 23.49.94(A), 23.49.96(D), 23.49.100(B), 23.49.102(B), 23.49.106(D), 23.49.122(C), 23.49.126(B), 23.49.134(B), 23.49.136(B), 23.49.146(A), 23.49.148(C), 23.49.152(B), 23.49.162(B) and (E), 23.49.164, 23.49.166(A), 23.49.178, 23.49.208(D), 23.49.210(C), 23.49.214 (catchline), 23.49.238(C), 23.49.246, 23.49.248(A), 23.49.324(C), 23.49.332(A) and (B), 23.54.30(B), 23.66.170(A), 23.66.304, 23.66.306, 23.66.310, 23.66.322(C), 23.66.338(F), 23.66.342(C) and (E), 23.84.18, 23.84.25, 23.84.32, 23.84.36, 23.86.04(B), 23.86.06(A), 23.86.07, 23.86.24(A) and 23.86.26, Land Use Code (23.04. 23.30, 23.49, 23.66, 23.84, 23.86)

ference only.

amending

raphics,

Seattle Municipal Code 112522 15.04.070, streets and Text provided for hi amends §§ 23.04.10, 23.22.16, 23.22.40, 23.22.48, 23.40.02, 23.40.10, 23.40.20,

See ordinances crea sections for comple and tables and to this source file.

Amends §§ 15.04.020 and sidewalks; adds Ch. 23.06, § 23.42.42 and Ch. 23.76; 23.44.18(A), Ch. 23.44 Part 2 (heading), 23.44.34, 23.45.106, 23.49.34, 23.49.36, 23.70.50, 23.70.60, 23.81.50, 23.84.06, 23.84.10, 23.84.30, 23.88.10 and 23.88.20; adds § 23.34.02 and renumbers §§ 23.34.020, 23.34.24, 23.34.28, 23.34.32, 23.34.34, 23.34.36, 23.34.38, 23.34.40, 23.34.42 and 23.34.44 to be §§ 23.34.04, 23.34.06, 23.34.08, 23.34.10, 23.34.14, 23.34.16, 23.34.18, 23.34.20 and 23.34.22 and amends §§ 23.34.04 and 23.34.06; amends § 23.82.10(C) and renumbers §§ 23.82.70 and 23.82.80 to be §§ 23.82.20 and 23.82.30 and amends §§ 23.82.20 and 23.82.30; repeals Chs. 23.76, 23.80 and 23.94 and §§ 23.22.28-23.22.36, 23.22.44, 23.28.40, 23.34.02-23.34.18 and 23.82.20-23.82.60, Land Use Code; amends § 24.66.100, zoning; amends §§ 25.05.510 and 25.05.680, environment; and defines all references in Ord. 112522 to Ch. 23.80, Decisions Requiring Council Approval to be to Ch. 23.76, Master Use Permits and Council Land Use Decisions (15.04, 23.04, 23.06, 23.22, 23.34, 23.40, 23.42, 23.44, 23.45, 23.49, 23.76, 23.84, 23.88, 25.05)

ference only.

Text provided for

See ordinan sections for this source

112539 Adds §§ 23.44.06(G), 23.44.17, 23.45.112, Ch. 23.79 (Establishment of Development Standard Departure for Public Schools), and adds language to §§ 23.84.02, 23.84.30 and 23.86.06, and adds § 23.86.06 (E); amends §§ 23.44.16(D), 23.44.22(A), (D) and (L), 23.45.90, 23.45.102, 23.45.122 and 23.54.30(E), Land Use Code; amends §§ 24.24.010 and 24.30.020(A) and (D), zoning (23.44, 23.45, 23.79, 23.84, 23.86) 112541 Adds §§ 12A.04.190, 12A.04.200 and 12.04.210; repeals and replaces §§ 12A.04.150, 12A.04.160 and 12A.04.170; repeals § 12A.02.130, Criminal Code (12A.04) 112547 Repeals and replaces Ch. 22.900, 1986 Permit Fee Ordinance (Repealed by 113162) 112549 Amends § 21.76.040(A), utility credits for sewerage charges (21.76) 112550 Adds § 11.26.290 and amends §§ 11.26.020, 11.26.060, 11.26.080, 11.26.100, 11.26.120, 11.26.160, 11.26.180, 11.26.240, 11.26.260 and 11.26.280, service parking (11.26)112551 Amends § 10.03.160, tuberculin test fees (10.03) 112552 Amends § 5.48.050, utility business and occupation

tax (5.48)

(Repealer)

(Repealer)

Fund (5.76)

fees (10.03)

Creates Special **Employment Program** 

Amends § 10.03.130, Public Health Department

Repeals Ord, 108815. personnel benefit programs

Repeals Ord. 102847, Personnel Services Operating Fund, and transfers assets and liabilities to General Fund ference only.

For current the Office of the City Clerk

112553

112555

112556

112557

June 2009 co

Text provide

See ordinant sections for and tables a this source

a anda	te file
112559	Amends § 10.18.110,
	frozen dairy food products
- 0 - m hi	(10.18)
112561	Amends §§ 10.03.140 and
CI ZO	10.03.150, Public Health
	Department fees (10.03)
112562	Amends § 10.03.120,
	Public Health Department
	fees (10.03)
112563	Amends § 20.48.070,
	Board of Public Works
	administrative charges
	(Repealed by 117390)
112564	Closes Emergency Medical
112304	Services Fund created by
	Ord. 108539 and transfers
ad to co	contents to General Fund
110565	(Repealer)
112565	Amends § 21.24.020,
<b>1116.</b>	sewer permits (Repealed
	by 116421)
112566	Amends § 10.03.020,
	fumigator and exterminator
	permit fees (10.03)
112567	Repeals § 3.30.130, family
	planning service fees
	(Repealer)
112568	Adds Ch. 18.28, park and
	recreational facility and
	service use fees (18.28)
112569	Amends §§ 10.03.010 and
	10.03.110(A), Health Code
	fees (10.03)
112570	Amends § 25.08.570, noise
	control (25.08)
112572	Seattle Center parking fees
	(Repealed by 113171)
112573	Adds § 17.16.015 and
	repeals §§ 17.16.020,
	17.16.025 and Ords.
	109490, 110896, 111433
	and 112031, Seattle Center
	Department (Repealed by
	113718)
112577	Adds Ch. 17.40,
	Bumbershoot Festival
	Commission, and creates
	Bumbershoot Festival
	Account (5.76)
112579	Amends § 6.36.020(B),
114317	massage and bathhouse
	licenses (6.36)
112500	` /
112588	Amends § 21.28.370,
	sewer charges (Repealed
	by 118176)
112589	Rezone, Plat 23E, page 168
	(Special)
112589 112590	(Special) Amends § 21.68.080,
	(Special)

eference only.

Id amending
graphics,
al ccuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipal coa			
4 - 4440	are man		
112591	Rezone, Plat 232E, page 27 (Special)		
112602	Downtown Health and		
THE PARTY IN	Human Service Account		
0 10-	established in General		
	Trust Fund (5.76)		
112606	Amends §§		
112000	0.0		
	4.04.250(D)-(F) and		
- 000	4.08.040(D)-(F), Public		
AGG CITES	Safety Civil Service		
	Commission (4.04, 4.08)		
112608	Amends § 21.52.210,		
	residential weatherization		
COMPLE	program (21.52)		
112611	Closes Seattle Fire District		
1400	No. 5 Fund created by Ord.		
ma to	84046 and transfers assets		
PRIOR C	to Seattle Fire District No.		
	5 Account in the General		
C210	Fund (5.76)		
112632	Adds § 24.74.015, zoning		
112032	(Repealed by 117570)		
112622			
112633	Adds §§ 4.12.010 and		
	4.12.020; repeals Ch. 4.13		
	and Ord. 111929,		
	exemptions from civil		
	service and public safety		
	civil service system		
	(Repealed by 113579)		
112635	Closes U.S. Savings Bond		
	Fund created by Ord.		
	72141 and amended by		
	Ord. 102271 and transfers		
	contents to Payroll		
	Withholding Fund (5.76)		
112637	Adds § 21.49.042, City		
112037	Light Department		
	low-income assistance		
112610	program (21.49)		
112649	Rezone, Plat 52W, page		
	132 (Special)		
112651	Concession agreements for		
	Seattle Center House		
	(17.16)		
112666	Adds Ch. 4.21, standby		
	duty (4.21)		
112686	Adds § 21.04.465 and		
	amends § 21.04.410, Water		
	Department charges		
	(21.04)		
112687	Amends § 21.40.060(A)		
112007	and (E), solid waste		
	S 77		
	container collection rates		
110100	(21.40)		
112699	Adds § 4.20.025, salary of		
	elected finance officers		
	(Repealed by 116368)		
112702	Amends § 6.36.020(B),		
	massage parlor and		
	bathhouse licenses (6.36)		

d amending graphics, accuracy of

Seattle Municipal Code

June 2009 co 112705 up date 1

Adds § 12A

amends § §
12A.14.055

and 12A.14

Code provise

See ordinant sections for and tables a this source

cigo.	te file
112705	Adds § 12A.14.155 and
100 0-2	amends §§ 12A.14.035,
100	12A.14.055, 12A.14.060
4 501 111	and 12A.14.085, Criminal
	Code provisions on gun
	control (Repealed by
	117157)
112719	Amends §§ 6.02.020,
	6.42.010, 6.42.020,
occeped	6.42.040 and 6.42.070(B);
ces er er	repeals §§ 6.02.280,
	6.42.060 and 6.42.080,
- 400 40 [6	business licenses and
COMPA	panoram licenses (6.02,
	6.42)
112738	Adds §§ 21.49.050,
and to co	21.49.055 and 21.49.057;
0.88	amends §§ 21.49.020,
	21.49.030, 21.49.040,
Sile.	21.49.060, 21.49.090,
Pare	21.49.100, 21.49.110,
	21.49.120 and 21.49.130;
	repeals §§ 21.49.050,
	21.49.070 and 21.49.170,
	Seattle City Light
112=12	Department (21.49)
112749	Rezone, Plat 231E, page 25
112750	(Special)
112759	Amends § 21.24.020(D),
	sewer connection fees
	(Repealed by 116421)

eference only.

l amending

graphics,

ccuracy of

Seattle Municipal Code 112777 Adds §§ 23.16.20, 23.34.070, 23.34.072, Text provided for hi 23.34.074, 23.34.076,

See ordinan sections for and tables and this source file.

(Quantity and Design Off-street Parking); 23.44.071(E), 23.44.020(D), 23.44.022(L),

ference only.

23.34.078, 23.34.080, 23.34.082, 23.34.084, 23.34.086, 23.34.088 and 23.86.034 and Chs. 23.46 (Residential-Commercial) and 23.47 (Commercial); repeals and adds Ch. 23.54 Standards for Access and amends §§ 23.30.010, 23.40.010, 23.44.016(A), 23.44.024(H), 23.44.026, 23.45.060(A), 23.45.076(A), 23.45.080, 23.45.084, 23.45.086(C), 23.45.088(B), 23.45.098(A), 23.45.110, 23.45.112(D), 23.45.124(B), 23.45.160(D), 23.49.044, 23.49.062, 23.49.092, 23.49.118, 23.49.144, 23.49.320, 23.66.122(A), (B) and (E), 23.66.322(A), 23.66.328, 23.84.002, 23.84.004, 23.84.006, 23.84.008, 23.84,010, 23.84.012, 23.84.014, 23.84.016, 23,84.018, 23.84.020, 23.84.022, 23.84.024, 23.84.025, 23.84.026, 23.84.028, 23.84.030, 23.84.032, 23.84.036, 23.84.038, 23.84.040, 23.84.042, 23.84.044 and numerous zoning maps; repeals §§ 23.44.062, 23.44.064 and 23.44.066, Land Use Code (23.30, 23.34, 23.44, 23.45, 23.46, 23.47, 23.49, 23.54, 23.66, 23.84, 23.86) Rezone, Plat 231E, page 25 (Special) Rezone, Plat 232E, page 25 (Special) Adds §§ 17.16.033 and 17.16.036; repeals and replaces § 17.16.030; repeals § 17.16.050 and Ords. 96298, 102528 and 108090, Seattle Center Department (17.16) Rezone, Plat 78E, page 184

(Special) the Office of the City

112784

112788

112790

112794

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipar			
	a mada	te file	
	112795	Amends § 24.60.675(G),	
		zoning (Repealed by	
	112796	113466) Rezone, Plat 37E, page 105	
	112/3	(Special)	
	112799	Amends §§ 23.79.004 and	
		23.79.008, Land Use Code (23.79)	
	112801	Designates Bethany	
	occepea	Presbyterian Church as a	
X	112010	Landmark (25.32)	
	112810	Amends § 10 of Ord. 111526 (7.20.110), floating	
	comple	home moorages (7.20)	
	112812	Amends §§ 24.54.150,	
	ad to CO	24.56.070 and 24.58.050,	
4	Ha co	zoning (Repealed by 117570)	
	112813	Amends §§ 5.40.025 and	
	sile.	5.40.085, admission tax	
	110001	(5.40)	
	112821	Amends § 4.08.060, Public Safety Civil Service	
		Commission (4.08)	
	112830	Makes various changes to	
		Ch. 4900 of the Seattle Amendments to the 1982	
		Uniform Building Code	
		adopted by Ord. 111200;	
		adds Ch. 23.55, signs;	
		amends §§ 23.45.110(C), 23.76.006, 23.84.002,	
		23.84.004, 23.84.006,	
		23.84.010, 23.84.012,	
		23.84.024, 23.84.025, 23.84.030, 23.84.032,	
		23.84.036, 23.84.038,	
		23.86.004 and 23.86.042;	
		repeals §§ 23.44.017(I),	
		23.44.022(K), 23.44.056, 23.45.100(D),	
		23.45.112(H), 23.45.158,	
		23.48.016 and 23.49.040,	
		Land Use Code; and amends §§ 24.44.010,	
		24.44.090, 24.52.010,	
		24.52.120, 24.54.010,	
		24.54.120, 24.56.010,	
		24.56.050, 24.58.010 and 24.58.040, zoning (23.45,	
		23.55, 23.76, 23.84, 23.86)	
	112833	Amends § 4.36.100, City	
		Employees' Retirement	
	112840	System (4.36) Adds § 23.42.040(D) and	
	11-010	amends §§ 23.76.004	
		(Exhibit A) and	
		23.76.006(B) and (C), Land Use Code; and	
		amends § 24.74.015,	
	. 014	zoning (23.42, 23.76)	

eference only.

I amending graphics, ccuracy of

For current Six Clerk the Office of the City Clerk

Seattle Municipal Code June 2009 co Text provide See ordinan

eattle Muss	a anda	te file
	112867	Amends § 12A.08.070, bad
2009 60		checks (12A.08)
	112890	Adds Part 2 (Council
-24	d for in	Conditional Uses) to Ch.
		23.44; adds language to §
ext provide		23.84.036; amends §§
035 0		23.44.18 and 23.44.24;
		renumbers former Part 2 of
		Subchapter II of Ch. 23.44
	oc crea	and § 23.44.34 to be Part 3
- dinan	Ce2 e2	and § 23.44.36, Land Use
See Orullian		Code (23.44, 23.84)
	112895	Rezone, Plat 30W, page 91
a soft	COMBR	(Special)
ections to	112900	Adds §§ 6.42.040-6.42.090
SEC 630	1 40 C	and 6.42.130; amends §§
1-106	ing to	6.02.190(A), 6.02.210 and
and tables a		6.42.010; renumbers §§
and co-	009	6.42.040 to be 6.42.100,
	<b>5110.</b>	6.42.070 to be 6.42.110
dain collect	8880	and 6.42.050 to be
this 5000		6.42.120 and amends the
		renumbered sections,
		business licenses for
		panorams and peepshows
	112002	(6.02, 6.42)
	112903	Adds §§ 14.04.030(K), (N)
		and (P), 14.04.050(E) and
		(F), 14.08.020(J),
		14.04.120(C),
		14.08.140(C) and
		14.08.190(H)-(J); amends
		§§ 14.04.020, 14.04.100, 14.04.140(A),
		14.04.140(A), 14.04.150(B),
		14.04.180(C), 14.08.010,
		14.08.040, 14.08.060(B),
		14.08.070(A) and (B),
		14.08.080(A) and (B),
		14.08.090, 14.08.120,
		14.08.160(A) and
		14.08.180(C); renumbers
		§§ 14.04.030(K)-(M) to
		(L), (M) and (O), and
		14.08.020(J)-(W) to be
		(K)-(X), discrimination and
		fair employment practices
		(14.04, 14.08)
	112908	Amends §§
		10.54.010-10.54.110,
		fumigators and
	112015	exterminators (10.54)
	112915	Closes Waterfront Streetcar
		Fund created by Ord.
		107831 and transfers
		contents to Transit Fund
	i	(Repealer)

eference only. amending graphics, ecuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

	40 Sile
4 440012	TE Pro-
112942	Amends §§ 21.36.010,
	21.36.050, 21.40.050,
1 Cor H1	21.40.060, 21.40.080 and
d in in	21.76.040, solid waste
	collection rates and low-income, elderly and
	handicapped utility credits
	(21.36, 21.40, 21.76)
112943	Adds § 5.48.220 and
112,43	amends §§ 5.48.050,
res crea	5.48.070, utility tax on
	garbage collection (5.48)
112959	Amends § 11.56.020,
COMPATO	driving under the influence
	of intoxicating liquor or
140 00	drugs (11.56)
112960	Amends § 3.76.020(A),
'Hara	Debt Management Policy
	Advisory Committee (3.76)
112962	Amends § 2.14.010,
73700	election pamphlets (2.14)
112964	Amends §§ 6.82.020,
	6.82.040(A), 6.82.060(A),
	6.82.150, 6.82.190,
	6.82.200 and 6.82.230,
	refrigeration code and
	licensing requirements; and
	amends § 201(a) of the Seattle Mechanical Code
	Supplement adopted by
	Ord. 111201 (6.82)
112969	Amends § 23.84.040, Land
112,00	Use Code; adds §§
	24.60.171 and 24.60.427;
	amends §§ 24.60.455 and
	24.60.610, zoning and
	shoreline permits (23.84)
112970	Amends §§ 23.45.038,
	23.84.012 and
	23.86.014(D), Land Use
	Code (23.84)
112971	Amends §§ 23.44.014(D),
	23.45.014(A) and (D),
	23.45.028 (A) and (D),
	23.45.042 (A) and (D),
	23.45.056(A) and (D),
	23.45.072(A) and (D),
	23.45.146(B),
	23.45.146(B), 23.47.014(C), 23.86.006,
	23.45.146(B), 23.47.014(C), 23.86.006, 23.86.012(A) and
	23.45.146(B), 23.47.014(C), 23.86.006, 23.86.012(A) and 23.86.018(D), Land Use
	23.45.146(B), 23.47.014(C), 23.86.006, 23.86.012(A) and 23.86.018(D), Land Use Code (23.44, 23.45, 23.47,
112972	23.45.146(B), 23.47.014(C), 23.86.006, 23.86.012(A) and 23.86.018(D), Land Use Code (23.44, 23.45, 23.47, 23.86)
112972	23.45.146(B), 23.47.014(C), 23.86.006, 23.86.012(A) and 23.86.018(D), Land Use Code (23.44, 23.45, 23.47, 23.86) Adds § 23.45.048, Land
	23.45.146(B), 23.47.014(C), 23.86.006, 23.86.012(A) and 23.86.018(D), Land Use Code (23.44, 23.45, 23.47, 23.86) Adds § 23.45.048, Land Use Code (23.45)
112972 112975	23.45.146(B), 23.47.014(C), 23.86.006, 23.86.012(A) and 23.86.018(D), Land Use Code (23.44, 23.45, 23.47, 23.86) Adds § 23.45.048, Land

reference only.

amending
graphics,
caccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	40 Sile
a - mada	Te min
112976	Reenacts Ord. 111458,
	adds §§ 25.08.425 and
- 0 - m hi	25.08.535, and amends §§
A FOR III	25.08.530, 25.08.540 and
	25.08.080, noise control
	(25.08)
112977	Amends §§ 11.66.035 and
,,,	11.66.050, railroad
	crossings (11.66)
112987	Amends § 4.20.225, Civil
CL CO	Service Commission
	hearings (4.20)
112988	
112988	Adds § 11.16.280(G),
COMP	Traffic Engineer's duties,
	and establishes stadium
1 40 00	event restricted parking
	zone in Montlake
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Neighborhood (11.16)
112990	Amends § 4.20.220,
	compensation for
	employees on jury duty
	(4.20)
112993	Adds § 59A to Ord.
112,70	112777, Land Use Code
	(23.46, 23.47)
112994	Amends Plat 36E, page
112994	
112007	103, rezone (Special)
112997	Adds §§ 4.64.100 and
	4.64.110, defense of claims
	against volunteer personnel
	(4.64)
113000	Adds § 12A.04.205,
	unreasonable physical
	discipline of a child; and
	amends § 12A.04.200, use
	of force by police and
	citizens (12A.04)
113011	Official Land Use Map
	Corrections, June 1986
	(Special)
113015	Establishes a Downtown
113013	Retail Core Business
	Improvement Area and
	Downtown Seattle Retail
	Core Business
	Improvement Area
	Account (5.76)
113016	Amends § 5802 of City
	amendments to the
	Uniform Building Code
	(Not codified)
113022	Amends §§ 15.62.060,
113022	Amends §§ 15.62.060, 15.62.070 and 15.62.110,
113022	15.62.070 and 15.62.110,
113022	15.62.070 and 15.62.110, and repeals and replaces §
113022	15.62.070 and 15.62.110, and repeals and replaces § 15.62.090, street vacations
	15.62.070 and 15.62.110, and repeals and replaces § 15.62.090, street vacations (15.62)
113022 113029	15.62.070 and 15.62.110, and repeals and replaces § 15.62.090, street vacations (15.62) Establishes Capitol Hill
	15.62.070 and 15.62.110, and repeals and replaces § 15.62.090, street vacations (15.62) Establishes Capitol Hill Business Improvement
	15.62.070 and 15.62.110, and repeals and replaces § 15.62.090, street vacations (15.62) Establishes Capitol Hill Business Improvement Area and Capitol Hill
	15.62.070 and 15.62.110, and repeals and replaces § 15.62.090, street vacations (15.62) Establishes Capitol Hill Business Improvement

eference only.

delamending
eraphics,
accuracy of

For current Sine, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinances comsections for comand tables and tables and this source file.

113031 Amends § 3.44.050, Executive Director of the **Board of Public Works** (Repealed by 116007) Amends § 20.48.070, public works contracts (Repealed by 117390) 113041 Adds § 23.45.190(F); adds new subsection (C) to §§ 23.45.12, 23.45.26, 23.45.40 and 23.45.54, and renumbers former subsections (C) of these sections to be (D); amends §§ 23.16.002(B), 23.40.010(D), 23.45.14(C) and (D), 23.45.16, 23.45.25(B), 23.45.28(C) and (D), 23.45.30, 23.45.38(B), 23.45.42(C) and (D), 23.45.44, 23.45.52(B), 23.45.56(C) and (D), 23.45.58, 23.70.80(C), 23.84.020, 23.84.028, 23.86.012 and 23.86.016(C), Land Use Code (23.45, 23.84, 23.86) 113042 Adds Subchapter IX and §§ 3.14.1110-3.14.1130 to Ch. 3.14, Office of International Affairs (Repealed by 115936) 113051 Amends § 23.84.036, Land Use Code (23.84) 113059 Amends § 22.700.010, Model Energy Code (22.700)113060 Rezone, Plats 22E and 29E, pages 76 and 90 (Special) 113079 Adds Ch. 23.70, Greenbelt Overlay District, and § 23.90.024(B); amends § 22.900.240(F), Exhibit 76.004 A of § 23.76.004, and §§ 23.76.006(B) and 23.76.030; and repeals Ord. 111568, building permits and Land Use Code (23.34, 23.76, 23.90) 113084 Amends § 12A.10.070, lewd conduct (Repealed by 117158) 113085 Amends § 12A.08.090, possessing stolen property (12A.08)113088 Designates the Brooklyn Building as a landmark (25.32)113089 Designates Fire Station No. 2 as a landmark (25.32)

ference only.

amending

raphics,

ccuracy of

For current Smc, Constitution of the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	40 Sile
4 - 4440012	ILE Prize
113090	Designates Hiawatha
	Playfield as a landmark
- 0 - m 101	(25.32)
113093	Rezone; adds Plats 302E
	and 303W and amends Plat
	78W (Special)
113106	Rezone, Plat 59W, page
113100	146 (Special)
113122	Rezone, Plat 47W, page
113122	122 (Special)
113136	
113130	Amends § 4.38.402,
-10	retirement age; adds Rule
aamble	12 and amends Rules 1, 3,
Compre	13 and 31 as adopted by
	Ord. 112517, voluntary
1 40 C	deferred compensation
ma to o	(Repealed by 119159)
113144	Amends §§ 14.08.010,
	14.08.020(E) and
5116	14.08.120, discrimination
777	in housing (14.08)
113147	Amends § 3.92.010 and §
	4.20.300(C), heads of
	departments and offices
	(3.92, 4.20)
113148	Adds § 4.04.225, smoking
113110	in the work environment
	for City officers and
	employees (4.04)
113160	Carries out provisions of
113100	Initiative Measure 30 on
	responsible immigration; dissolves Citizens'
	Commission on Central
	America and repeals Res.
	27402, §§ 2 and 4 of Ord.
	111256 and portions of
	Res. 27081 (4.18 Editor's
	Note)
113162	Repeals and replaces Ch.
	22.900, 1987 permit fee
	ordinance (Repealed by
	113744)
113163	Amends § 21.40.080,
	Transfer Station and
	disposal site rates (21.40)
113164	Amends § 21.28.370,
	sewer rates (Repealed by
	118176)
113165	Amends § 21.76.040(A),
	utility rates for elderly/
	handicapped/disabled
	(21.76)
113166	Adds § 21.28.300, extends
	certain responsibilities of
	the Sewerage Utility
	(Repealed by 118176)
113167	
11310/	Amends § 21.44.030 and
	repeals § 21.44.040, annual Transfer Station fees
L-SH	(Repealed by 114723)
	,

eference only.

at

at

an amending

graphics,

accuracy of

For current Sincy Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinand sections for and tables a this source

	Cipas			
	a anda	te tille		
	113168	Amends § 10.25.020, fees		
		for reports on certain water		
	1 Cost hi	supply and sewage systems		
	dioi	(10.25)		
	113169	Adds § 10.03.100, health		
		code fees (Repealed by		
	113170	113908) Amends § 10.02.130, fees		
	113170	for personal health care and		
	- 0102	dental services (10.03)		
	113171	Adds §§		
<b>B</b> '		17.19.01 <mark>0</mark> -17.19.050,		
		Seattle Center parking		
	COMPA	facilities; repeals Ch. 17.18		
		and Ord. 112572 (17.19)		
	113172	Amends § 5.48.050,		
	ma to o	business and occupation		
•		tax (5.48)		
	113176	Amends § 20.48.070,		
	<b>4116.</b>	Board of Public Works		
		contract administration charges (Repealed by		
		117390)		
	113178	Adds § 17.16.020, fees for		
	113176	Seattle Center facilities and		
		services; repeals §		
		17.16.015 and Ord. 112573		
		(Repealed by 113740)		
	113179	Amends § 18.28.010, Parks		
		and Recreation Department		
	110101	fee schedule (18.28)		
	113181	Amends §§ 5.44.130 and		
		5.44.140, business license application fee (5.44)		
	113183	Amends § 22.904.040,		
	113163	mobile home park licenses		
		(22.904)		
	113184	Amends § 7.04.570,		
		Weighmaster license fee		
		(7.04)		
	113185	Amends §§ 6.08.010,		
		6.14.030, 6.14.040,		
		6.14.100, 6.36.020,		
		6.36.030, 6.36.040,		
		6.38.020, 6.42.030, 6.48.020, 6.60.010,		
		6.82.140, 6.84.020,		
		6.102.030, 6.204.030,		
		6.204.040, 6.204.050,		
		6.204.080 and 6.204.090,		
		various business license		
		fees (6.08, 6.14, 6.36, 6.38,		
		6.48, 6.102, 6.204)		
	113186	Amends § 11.31.120,		
		penalties for traffic		
		infractions (11.31)		

ester eference only.

I amending and graphics,

I accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	. C:10
a senda	te file
113189	Adds Subchapter IX (§§
4.2	3.14.3000-3.14.3070) to Title 3, Office for
a sor mi	Long-range Planning;
CI TOT	abolishes Land Use and
	Transportation Project in
	the Executive Department
	(Repealed by 115949)
113190	Adds Subchapter VIII (§§
soc crea	3.14.2000-3.14.2060) to
CES CE	Title 3, Office for Strategic
	Human Services Planning (Repealed by 115949)
113192	Carries out provisions of
113172	Initiative Measure 30 on
4 4 5 00	responsible immigration;
and to co	dissolves Citizens'
The co	Commission on Central
	America, and repeals Res.
sile.	27402, §§ 2 and 4 of Ord.
Range	111256 and portions of
	Res. 27081, enforcement of
	federal immigration laws
113195	(4.18) Makes changes to Arterial
113193	Street Map (11.18.010
	footnote)
113199	Amends Exhibit A of Ord.
	100475 /G§25.24.020/H,
	Pike Place Market
	Historical District (25.24)
113203	Amends §§ 23.45.014(D),
	23.45.028(D),
	23.45.042(D) and 23.45.056(D), Land Use
	Code (23.45)
113216	Repeals and replaces §
	23.44.010(B), Land Use
	Code (23.44)
113220	Amends § 5.44.170,
	business and occupation
112221	tax (Repealed by 118314)
113221	Amends § 3.104.010, fees for copies of documents
	1 (3 1()4)
113222	(3.104) Amends § 3.118.010.
113222	Amends § 3.118.010,
	Amends § 3.118.010, industrial development facilities (3.118)
113222 113240	Amends § 3.118.010, industrial development facilities (3.118) Amends §§ 11.16.120 and
	Amends § 3.118.010, industrial development facilities (3.118) Amends §§ 11.16.120 and 11.16.317, restricted
113240	Amends § 3.118.010, industrial development facilities (3.118) Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16)
	Amends § 3.118.010, industrial development facilities (3.118) Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16) Low-income elderly and
113240	Amends § 3.118.010, industrial development facilities (3.118)  Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16)  Low-income elderly and handicapped housing
113240	Amends § 3.118.010, industrial development facilities (3.118) Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16) Low-income elderly and
113240 113244	Amends § 3.118.010, industrial development facilities (3.118)  Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16)  Low-income elderly and handicapped housing program (5.76)
113240	Amends § 3.118.010, industrial development facilities (3.118)  Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16)  Low-income elderly and handicapped housing program (5.76)  Rezone, Plats 58E and
113240 113244	Amends § 3.118.010, industrial development facilities (3.118)  Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16)  Low-income elderly and handicapped housing program (5.76)  Rezone, Plats 58E and 59W, pages 145 and 146
113240 113244 113245	Amends § 3.118.010, industrial development facilities (3.118)  Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16)  Low-income elderly and handicapped housing program (5.76)  Rezone, Plats 58E and 59W, pages 145 and 146 (Special)
113240 113244	Amends § 3.118.010, industrial development facilities (3.118)  Amends §§ 11.16.120 and 11.16.317, restricted parking zones (11.16)  Low-income elderly and handicapped housing program (5.76)  Rezone, Plats 58E and 59W, pages 145 and 146

reference only.

damending
ic graphics,
accuracy of

For current 534 (5.00, 3.70)

the Office of the City Clerk

Seattle Municipal Code ference only. date 113261 Designates Ramsing House as a landmark (25.12) 113262 Adds §§ 23.44.018(F), Text provide 23.45.116(F), 23.47.006(D), 23.48.005(B), 23.49.040 amending and 24.74.010(B); amends § 24.74.010 by designating provisions as subsection A, Land Use Code (23.44, See ordinan 23.45, 23.47, 23.49) Adds §§ 23.44.035, 23.45.104, 23.47.015, sections for 23.47.028(E) and 23.47.036(D); amends §§ 23.34.084, 23.40.020(A), and tables and 23.44.022(L), 23.45.104, 23.45.184(C), 23.47.004 (Chart A), 23.47.006(B) this source file. and (C), 23.47.010, 23.47.011(B) and (F), 23.47.012, 23.47.014, 23.47.016(B)-(D), 23.47.018, 23.47.022(D), 23.47.024, 23.47.030(B) and (C), 23.47.032(B), 23.47.036(A), 23.47.038(A), 23.47.042(D), 23.47.050, 23.54.015, 23.54.020(A) and (H), 23.54.030(E) and Exhibit D, 23.55.032(A), 23.84.002, 23.84.004, 23.84.014, 23.84.016, 23.84.020, 23.84.025, 23.84.026, 23.84.028, 23.84.032, 23.84.036,

23.84.044, 23.84.048 and 23.86.004(A); repeals § 23.47.044(E), Land Use Code (23.40, 23.44, 23.47, 23.54, 23.84, 23.86)

Amends Ord. 113093, rezone, plats 78W, 302E and 303W, pages 183, 205 and 206 (Special)

Amends portion of Ord. 113186, bail for overtime parking infractions (Not

codified)

## For current SMC, contact the Office of the City Clerk

113271

113274

June 2009 co 13279 UP Text provided for

See ordinances composed sections for composed and tables and to this source file.

113279 Adds §§ 23.49.046(F), 23.49.066(F), 23.49.096(G), 23.49.122(F) and 23.49.324(F); amends §§ 23.49.008(C), 23.49.016(E), 23.49.018, 23.49.020(B), 23.49.024(B) and (C), 23.49.036(C), (E) and (G), 23.49.046(C), 23.49.052(B) and (C), 23.49.066(C), 23.49.072(B), 23.49.096(D), 23.49.102(B), 23.49.122(C), 23.49.128(B), 23.49.144, 23.49.146, 23.49.148(B) and (C), 23.49.152, 23.49.154(B), 23.49.164, 23.49.166(B), 23.49.180(B), 23.49.208(B) and (C), 23.49.210(B), 23.49.212(B), 23.49.238(A), 23.49.240(B), 23.49.246, 23.49.324(C), 23.66.170(B) and 23.66.342(B); adds § 23.54.030(G) and renumbers and amends §§ 23.54.030(G) and (H) to be

> §§ 23.54.030(H) and (I), Land Use Code (23.49,

Amends § 10.13.040(B) and (H), meat, rabbit, poultry and aquatic foods

Repeals and replaces § 23.44.010(B) and confirms passage, approval and effective date of Ord. 113216, Land Use Code

Amends § 1 of Ord. 108615 and expands certain restricted parking zones in East Shelby Street

Closes Federal Shared Revenue Fund created by Ord. 101742 (Not codified)

Amends § 12A.50.010 and repeals § 12A.22.080, social card games (12A.50)

Amends § 4.36.130(A) and (D), City Employees'
Retirement System (4.36)

23.66)

code (10.13)

(23.44)

area (11.73)

113289

113297

113300

113301

113308

113309

ference only.

amending

raphics,

curacy of

For cus Land Use Code (23.44)

The Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

Cipar		
a mada	te tille	
113319	Adds § 6 to and amends § 1 of Ord. 81301,	
a sor hi	Cumulative Reserve Fund (Repealed by 116497)	
113320	Adds § 20.04.290,	
110020	cancellation of public	
	improvement district	
110006	assessments (20.04)	
113326	Establishes West Seattle Junction Parking and	
ces crea	Business Improvement	
	Area (5.76)	
113354	Amends § 23.44.036, Land	
Comb	Use Code; adds language to § 24.08.200 and amends	
and to co	§ 24.56.030, zoning (23.44)	
113355	Repeals and adds §	
cile	22.502.015, plumbing code (Repealed by 114712)	
113356	Seattle amendments to the	
	1985 Edition of the	
	Uniform Plumbing Code	
113373	(Repealed by 114713) Adds subsection (H) and	
113373	amends subsections (C),	
	(D) and (G) of § 23.49.036,	
110074	Land Use Code (23.49)	
113374	Amends §§ 21.40.050, 21.40.060, 21.40.080 and	
	21.76.040, solid waste	
	collection and transfer	
	station fees (21.40, 21.76)	
113375	Amends §§ 5.48.050(H) and 5.48.220, utility tax on	
	garbage collection (5.48)	
113387	Adds §§ 23.47.004(F) and	
	23.47.025; amends §§	
	23.44.050, 23.45.152,	
	23.47.004 (Chart A), 23.55.028(D),	
	23.55.030(D) and portion	
	of § 23.84.006, Land Use	
	Code (23.44, 23.45, 23.47,	
113400	23.55, 23.84) Repeals and adds §	
113700	23.45.92(B) and adds §	
	23.45.92 (C) and (D), Land	
112401	Use Code (23.45)	
113401	Amends §§ 23.44.012(C), 23.44.014(D), 23.44.046	
	and 23.45.146(B) and (C),	
	Land Use Code (23.44,	
	23.45)	
113422	Designates the Guiry Hotel	
113423	as a Landmark (25.32)	
113443	Designates the Parker-Fersen Building as	
	a Landmark (25.32)	

eference only.

d amending
graphics,
id ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
113424	Designates the Louisa Building as a Landmark
a o a m hi	(25.32)
113425	Designates the Olympic
	Tower/United Shopping
	Tower as a Landmark
113426	(25.32) Designates the Liggett
113420	Building as a Landmark
se crea	(25.32)
113427	Designates the L. C. Smith
10	Building (Smith Tower) as
113428	a Landmark (25.32) Designates the Fireboat
113420	Duwamish as a Landmark
140 0	(25.32)
113429	Designates the Olympic
	Warehouse and Cold
0210	Storage Building as a
113430	Landmark (25.32) Designates the Shafer
113730	Building as a Landmark
	(25.32)
113436	Amends § 12A.52.020,
	litter; adds §§ 18.12.145,
	18.12.165, 18.12.195,
	18.12.235, 18.12.255,
	18.12.265, 18.12.275 and 18.12.277, and amends §§
	18.12.030(A), 18.12.040,
	18.12.050, 18.12.070,
	18.12.080, 18.12.100,
	18.12.120, 18.12.170,
	18.12.180, 18.12.240, 18.12.260, 18.12.270 and
	18.12.280; repeals §
	18.12.220, Park Code
	(18.12)
113442	Water system improvement
	plans and bonds; creates
	1987 Seattle Municipal Water Revenue Parity
	Bond Account (1987 Bond
	Account) (5.76)
113459	Designates the Rainier
	Club as a Landmark
113460	(25.32) Designates the Schillestad
113400	Building as a Landmark
	(25.32)
113461	Amends § 1 of Ord.
	108615 [§ 11.73.040],
	restricted parking zone in
	the Montlake neighborhood (11.73)
113462	Amends § 11.73.050(B),
110704	
	Husky Stadium Event
	Husky Stadium Event restricted parking zone

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

a seeda	te file
113464	Adds § 23.44.031 on bed
	and breakfast uses, and
101	amends §§ 23.54.015,
4 501 111	23.55.022(D) and
CI TO	23.84.024, Land Use Code
	(23.54, 23.55, 23.84)
113466	Adds § 23.04.030 and Ch.
113400	23.60, Shoreline District;
	repeals Ch. 24.60, Land
ac crea	Use Code (23.60)
113472	Amends § 3.68.030,
	Commission on Children
	and Youth (3.68)
113478	Amends § 12A.08.040,
	criminal trespass (12A.08)
113479	Amends § 12A.16.060,
	interference with police
The same	dogs or police horses
	(12A.16)
112400	Amends § 12A.08.110,
113480	
	unauthorized use of a
	motor vehicle; repeals §§
	12A.06.060 and
	12A.08.080 (12A.08)
113484	Drug Enforcement
	Forfeiture Account (5.76)
113498	Amends § 5.40.025,
	admission tax exemptions
	(5.40)
113501	Adds subsection (H) to §
113301	3.14.810, purchasing
	system (3.18)
113502	
113302	Adds §§
	21.36.021-21.36.024 and
	21.36.130-21.36.170;
	amends §§ 21.36.010,
	21.36.020, 21.36.030,
	21.36.040, 21.36.110 and
	21.36.120, solid waste
	collection; repeals Ch.
	12A.52, littering (21.36)
113513	Amends § 1 of Ord.
	112169, sale of certain real
	property (Special)
113520	Amends §§ 5.60.010 and
113320	5.60.020, sales and use tax
	(5.60)
112521	
113521	Amends § 5.06.010(B),
	Investment Committee
	meetings (Repealed by
	116368)
113532	Amends § 21.40.080,
	transfer station and
	disposal site rates (21.40)
113533	Amends § 21.40.050,
	I garnage and runnish
	garbage and rubbish collection rates (21.40)

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Than	
	and a	te file
	113545	Repeals and adds Chs.
		22.200, 22.202, 22.204,
	a coa hi	22.206, 22.208 and repeals
3)(	d tor m	Ords. 106319, 107797,
		108647, 108736, 109219,
		109709, 109974, 110226 and 110548, Housing Code
		(22.200, 22.202, 22.204,
		22.206, 22.208)
	113546	Adds § 11.40.255,
	iez ez ez	skateboards and
		rollerskates on sidewalks
ŀ	1105 47 MILE	and public paths (11.40)
1	113547	Amends §§ 12A.14.010, 12A.14.075, 12A.14.080
		and 12A.14.100, concealed
	ad to cu	or unconcealed weapons
1	Her o	(12A.14)
Γ	113548	Adds Ch. 12A.30,
	sile.	inhalation, possession and
	777	sale of substances releasing
		toxic vapors or fumes, and
		amends § 12A.01.010, Criminal Code (12A.02,
		12A.30)
F	113549	Adds § 12A.14.165,
		alteration of identifying
		marks, § 12A.18.080,
		leaving child unattended,
		and § 12A.60.105,
		tampering with fire alarm apparatus, and repeals §
		12A.14.140, Criminal
		Code (12A.18, 12A.60)
	113550	Amends § 11.56.020,
		driving while intoxicated
L	110565	(11.56)
	113565	Adds § 18.12.257 and amends § 18.12.255,
		possessing liquor in public
		parks (18.12)
r	113566	Adds § 12A.24.025,
		possession or consumption
		of liquor in public places
L	112570	(12A.24)
	113579	Adds §§ 4.13.010 and 4.13.020 and repeals Ch.
		4.12, exemptions from the
		Civil Service and Public
		Safety Civil Service
L		systems (4.13)
	113597	Adds Ch. 4.10, limited
		duty assignments during
F	112601	pregnancy (4.10)
	113601	Designates the Black Manufacturing Building as
		a Landmark (25.32)
F	113602	Designates the Puget
		Sound Bank (Bank of
		California) as a Landmark
I		(25.32)

eference only.

de de de amending
graphics,
accuracy of

For current Single, Grant Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibar	
	te file
113610	Adds subsections (K)-(M)
113010	to § 14.08.190 and amends
103	§§ 14.08.010(A),
4 FOT III	14.08.020, 14.08.030,
an re-	14.08.040, 14.08.090 and
	14.08.180, discrimination
	in places of public
	accommodation (14.08)
113611	Amends § 5.44.050 and
Loc crea	subdivides said section to
(Ces es	add §§ 5.44.052, 5.44.054
	and 5.44.056, and adds §
commile	5.44.058, business and
112612	occupation tax (5.44)
113613	Adds. Ch. 3.79 [3.63], Taxicab Commission; adds
and to co	§§ 6.212.045,
	6.212.160(D) and (E),
	6.212.250, 6.212.260 and
C:10	2.212.270, and amends §
1116.	6.212.070, taxicab licenses
	and equipment (3.63)
113614	Amends § 23.44.016, Land
	Use Code (23.44)
113615	Amends § 23.84.036, Land
	Use Code (23.84)
113636	Amends §§ 21.49.055 and
	21.49.057, Seattle City
112640	Light rates (21.49)
113640	Adds § 12A.06.190,
	violation of civil
	antiharassment protection orders (12A.06)
113641	Amends § 12A.06.155,
113041	domestic violence
	prevention (12A.06)
113649	Amends § 15.62.080,
	vacation of waterfront
	streets (15.62)
113658	Adds Ch. 23.50, Industrial,
	and 23.68, Industrial
	Overlay District, and §§
	23.34.090, 23.34.092,
	23.34.094, 23.34.096, 23.54.035 and 23.55.036;
	amends §§ 23.30.010,
	23.47.004, 23.54.010,
	23.54.015, 23.54.020(F),
	23.54.030(E), 23.84.016,
	23.84.025, 23.84.036,
	23.84.038 and 23.84.040,
	Land Use Code (23.30,
	23.34, 23.47, 23.50, 23.54,
	23.55, 23.84)
113661	Corrects mapping errors
	for Plat 38W (page 106)
	and Plat 40W (page 110),
	zoning maps (Special)

d amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	ate file
113662	Amends § 23.45.110
	introductory paragraph and
a com hi	subsection (C), Land Use
A TOF BE	Code (23.45)
113690	Amends § 5.44.020 and
	subdivides the section to
	add §§ 5.44.022, 5.44.024,
	5.44.026 and 5.44.028;
	amends §§ 5.44.072,
Loc crea	5.44.110, 5.44.180,
ICE2 CT	5.48.020, 5.48.050 and
	5.48.110, business and
	occupation tax and tax on
COLLEGE	utilities (5.44, 5.48)
113693	Rezone, Plat 41W, page
1400	112 (Special)
113697	Adds § 12A.12.015 and
d'alle	amends § 12A.12.010,
	pedestrian interference and
	disorderly conduct
7770	(12A.12)
113699	Rezone, Plat 12E, page 56
	(Special)
113700	Amends § 22.100.010,
	adopts Uniform Building
	Code, 1985 Edition
	(Repealed by 116012)
113701	Adds Chs. 1, 2 and 3 to
	Uniform Building Code,
	1985 Edition (Not
	codified)
113702	Amends § 6.82.020(N),
	refrigeration systems;
	amends § 22.400.010,
	Uniform Mechanical Code
	1985 Edition adopted by
	reference; repeals §§
	6.82.040, 6.82.080,
	6.82.090, 6.82.161,
	6.82.170, 6.82.200 and
	6.82.220-6.82.310 (6.82)
113704	Rezone, Plats 12W and
	12E, pages 55 and 56
	(Special)
113706	Rezone, Plats 12W and
	12E, pages 55 and 56
	(Special)
113707	Rezone, Plat 12E, page 56
	(Special)
113709	Rezone, Plat 15W, page 61
	(Special)
113710	Amends §§ 23.54.015,
	23.54.020(B) and
	23.54.030(A), Land Use
	Code (23.54)
113711	Amends portions of certain
	contract authorized
	pursuant to Ord. 106309,
	Seattle Housing Authority
	funding (Not codified)
	AC COMILE

eference only.

d amending
graphics,
accuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

		te file
	113713	Rezone, Plat 41W, page
		112 (Special)
	113714	Amends § 5.48.050,
	A FOR III	business tax on utilities
3		(5.48)
	113715	Amends § 94 of Ord.
		113701, Building Code §
		3306 amendment (Not
		codified)
	113718	Adds § 23.45.007, Land
V	CES OF	Use Code (Repealed by
		114196)
	113727	Amends § 5.60.010, sales
	COMPLE	and use tax (5.60)
	113729	Amends § 21.76.040(A),
	- 4 to C	elderly/handicapped/disabl
4	nu to o	ed utility credits (21.76)
	113730	Amends § 21.28.370,
	0010	sewer service charges
	Sile.	(Repealed by 118176)
	113733	Amends § 10.03.130,
		Health Department fees for
		personal health services
	112724	(10.03)
	113734	Amends § 22.602.040, Fire
		Department fees for hazardous materials and
		processes (22.602)
	113735	Amends § 9.26.020, cat
	113/33	and dog license fees (9.26)
	113736	Amends § 18.28.010, Parks
	113730	and Recreation Department
		fee schedule (18.28)
	113737	Amends § 20.48.070,
	113737	Board of Public Works
		daily administrative
		charges (Repealed by
		117390)
	113738	Amends § 3.20.210,
		P-patch program garden
		plot fee schedule (3.20)
	113740	Adds § 17.16.015 and
		repeals § 17.16.020 and
		Ord. 113718, fees for
		Seattle Center facilities and
		services (17.16)
	113741	Adds § 17.19.030(E);
		amends §§
		17.19.010(B)-(D),
		17.19.020 and
		17.19.030(B), Seattle
		Center Department fees
	112742	and charges (17.19)
	113742	Amends § 21.24.020,
		sewer connection permits
	113743	(Repealed by 116421)
	113/43	Adds § 22.900.205, special
		valuation program for historic properties
		(Repealed by 117405)
		(Repealed by 117403)

d amending graphics, accuracy of

For current SM<del>C, complact</del> the Office of the City Clerk Seattle Municipal Code

June 2009 co 113744 up a Repeals
22,900, ordinan
114243
Text provide 113747 Amend

See ordinant sections for and tables at this source

cipus da	te file
113744	Repeals and replaces Ch.
	22.900, 1988 permit fee
1 - 0 - 10	ordinance (Repealed by
A SOT III	114243)
113747	Amends §§ 3.56.030(C)
	and 3.56.040, Seattle Arts
	Commission (Repealed by
	121006)
113750	Adds § 3.14.580 and adds
	Subchapter XII (§§
ces crea	3.14.4000-3.14.4050) to
	Ch. 3.14, Office of
	Neighborhoods; amends §
COMPLE	3.20.030, Department of
Cosses	Human Resources (3.20)
113757	Adds § 6.230.210 and
113/3/4	amends §§ 6.230.050,
HILL CO	
	6.230.060, 6.230.090,
0210	6.230.150 and 6.230.160,
111C.	steam engineer and boiler fireman licensing (6.230)
113759	<u> </u>
113/39	Makes changes to Plat 35E
	(page 101) and Plat 102
	(page 102) to implement
	South Lake Union area
112764	(Special)
113764	Amends § 2 of Ord.
	113466 and §§
	23.60.002(B), 23.60.018,
	23.60.020, 23.60.184,
	23.60.196(C),
	23.60.246(K),
	23.60.306(K),
	23.60.368(K),
	23.60.424(I), 23.60.430(L),
	23.60.480, 23.60.542(D),
	23.60.544(J),
	23.60.602(D),
	23.60.606(J), 23.60.642,
	23.60.668(J), 23.60.722(E),
	23.60.728(A), (B) and (K),
	23.60.730(A), 23.60.760,
	23.60.782(D), 23.60.784,
	23.60.786(F), 23.60.842
	(introductory paragraph
	and (D)) and 23.60.848(J);
	repeals §§
	23.60.426(D)(1)-(4),
	23.60.604(D)(1)-(4),
	23.60.724(D)(1)-(4) and
	23.60.844(C)(1)-(4),
	Shoreline District Land
	Use regulations (23.60)
113779	Amends § 3.44.040,
	contract duties of Board of
	Public Works (Repealed by
	116007)
113786	Adds Ch. 3.33, Seattle
	Municipal Court (3.33)
	40

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipal Cilo		
a mada	te tille	
113797	Adds § 3.114.145 and amends §§ 3.114.150 and	
ed for hi	3.114.160, certified consultants rosters for contracts (3.114)	
113799	Rezone, Plat 4W, page 40 (Special)	
113800	Adds § 23.44.051 and repeals § 23.44.031, Land	
113815 <b>CPC</b> 3	Use Code (23.44) Adds § 11.201(b) to	
male	Uniform Fire Code as adopted by Ord. 111001	
113821	(Repealed by 114328) Twelve-month moratorium	
and to co	on issuance of permits for demolition or change of use of low-income housing	
113824	(Expired) Amends § 11.56.320, driving while license is suspended or revoked	
113826	(11.56) Adds §§ 6.294.075 and 6.294.085; amends §§ 6.294.010, 6.294.040, 6.294.050, 6.294.055, 6.294.060, 6.294.070, 6.294.080, 6.294.090, 6.204.105 and 6.204.160	
	6.294.105 and 6.294.160, dancehalls admitting persons under the age of eighteen (Repealed by 120889)	
113830	Amends § 5.06.010, Investment Committee (5.06)	
113834	Establishes the low-income housing fund; amends Ch. 5.76, Table of Funds; amends § 14 of Ord. 112342 (§ 22.210.190), downtown housing preservation account; amends § 9(A) of Ord. 112383 (§ 22.220.100), downtown housing maintenance account; amends § 8, 1, 2 and 5 of Ord. 113752, § 8, 4, 6(E) and 9 of Ord. 112904, § 8, 2, 3 and 6 of Ord. 113573, § 4 of Ord. 113751; closes funds created by Ords. 112342 and 112904, and closes, transfers, establishes and reappropriates certain funds, all pertaining to low-income housing (5.76, 22.220)	

d amending graphics, ccuracy of

For current SM 22.220) College of the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
113836	Amends § 4.04.225, smoking in the work
100	environment (4.04)
113843	Amends § 12A.10.010,
	prostitution loitering
	(12A.10)
113847	Adds (D) to § 4.36.110 and
	amends § 4.36.120(A), city
	employees' retirement
oc cred	system (4.36)
113858	Adds § 23.45.006.5,
	interim zoning controls for
	certain developments
COMBRA	(Repealed by 114887)
113860	Adds §§ 18.12.042,
- 4 to C	18.12.045 and 18.12.047,
ing to	permit system for use of
	park and recreational
44004	facilities (18.12)
113861	Rezone, Plat 58E, page 145
112064	(Special)
113864	Amends §§ 22.900.170,
	Tables A and B;
	22.900.180(A) and (B); 22.900.220, Table G;
	22.900.220, Table G; 22.900.230, Table I;
	22.900.240(B), (D)-(H)
	and (J)-(O); 22.900.250
	and 22.900.260(4) and (5),
	Permit Fee Ordinance
	(Repealed by 114243)
113865	Amends § 21.52.240, home
	energy loan program
	(21.52)
113872	Adds §§ 4.20.700,
	5.20.050, 5.12.060 and
	5.24.035, all pertaining to
	warrants, payments and
	billings under the amount
	of one dollar (4.20, 5.12,
	5.20, 5.24)
113882	Amends § 21.40.080, solid
	waste collection rates and
112002	charges (21.40)
113883	Adds (C) to and changes
	title of § 23.86.008;
	amends §§ 23.44.010(B),
	23.44.012(A) and 23.44.014(D), Land Use
	Code (23.44, 23.86)
	Coue (23.44, 23.80)

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

2	a anda	te file
	113892	Adds §§ 23.47.008,
		23.47.013, 23.47.023 and
	a com hi	23.47.027; adds (F) to and
5	d for man	amends (B)-(E) of §
		23.47.014; amends §§
		23.47.004 (Chart A),
		23.47.006(B), 23.47.024
		and 23.86.007(A); deletes
		(E) from and reletters
	oc crea	remaining subsections of §
	CE2 CZ	23.47.012, Land Use Code
		(23.47, 23.86)
	113894	Harborview Medical
	COMPA	Center master plan and
		rezone (Special)
	113895	Corrections to land use
7	ma to co	maps for Plats 11W, page
•	18800	53; 21W, page 73; 21E,
		page 74; 222, page 9; 38W,
- 1		404 140

eference only.

l amending

graphics,

ccuracy of

page 106; and 63E, page 154 (Special) 113898 Adds § 11.72.145; amends § 11.22.080, vehicle license plates (11.22, 11.72) 113906 Adds § 11.14.597, State Commission on Equipment; adds §§ 11.84.090, 11.84.150 and 11.84.290 and amends § 11.57.100, all pertaining to vehicles and equipment (11.14, 11.57, 11.84) 113907 Amends § 10.64.060, no-smoking areas (10.64) 113908 Amends §§ 10.03.110, 10.11.020, 10.11.040 and 10.11.060, food service establishments; repeals Ord. 113169 (10.03) 113913 Exempts certain subsidized low-income housing projects from interim zoning controls in multi-family zones (Repealed by 114887) 113925 Amends § 21.04.570, unlawful connections to water supply system (21.04)113957 Adds Ch. 11.69 (§§ 11.69.010-11.69.040), traffic congestion areas (11.69)113977 Amends § 403 of city amendments to the Uniform Building Code; and amends §§ 23.76.026, 23.84.002, 23.84.030 and 23.84.036, Land Use Code (23.76, 23.84)

For current Sind, 5 the City Clerk the Office of the City Clerk

Seattle Municipal Code June 2009 co Text provide

eattle Munic		te file
	113978	Amends §§ 23.42.010,
	113/13	23.42.020, 23.44.040 and
Inne Zuur	4.3	23.45.140; amends Ch.
, 10	a sor m	23.90 by amending §
royt provide	CI TO	23.90.002, renumbering §§
Post Dru		23.90.008, 23.90.012 and
I CW R		23.90.014 to be §§
		23.90.004, 23.90.006 and
		23.90.008, with
	oc crea	amendments; adds new §§
- dinan	ces es	23.90.010, 23.90.012 and
See Orumina		23.90.014; repeals §§
366	ammile	23.90.016 and 23.90.018;
rections for	Comple	and renumbers and amends
sections for		§§ 23.90.020, 23.90.024, 23.90.028 and 23.90.030 to
Second	ad to co	be 23.90.016, 23.90.018,
and tables a	III o	23.90.020 and 23.90.022
and tables		respectively, Land Use
Siles on	C:10	Code (23.42, 23.44, 23.45,
939900	1116.	23.90)
this source	113979	Rezone, Plat 59W, page
		146 (Special)
	113985	Amends § 23.47.013, Land
		Use Code (Repealed by
		117430)
	113987	Designates the J. S.
		Graham Store/Doyle
		Building as a Landmark (25.32)
	113988	Designates the San Remo
	113766	Apartment Building as a
		Landmark (25.32)
	113990	Designates the 84 Union
		Building (U.S. Immigration
		Building) as a Landmark
		(25.32)
	113991	Amends § 3 of Ord.
		109267, Southeast Seattle
		Economic Development
	114007	Revolving Account (5.76)
	114007	Amends §§ 23.49.152 and 23.49.164, Land Use Code
		23.49.164, Land Use Code (Repealed by 114079)
	114011	Creates the Fort Lawton
	111011	Landmark District with
		administrative and
		enforcement procedures
		(25.32)
	114034	Adds Ch. 3.124,
		solicitations for charitable
		contributions from city
		employees (3.124)
	114041	Amends § 23.76.024, Land
		Use Code (23.76)

reference only. d amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	to file
<del>4 1111</del> 02	LE Bass
114046	Adds §§ 23.45.015,
	23.45.017, 23.45.029,
1 Cost h1	23.45.031, 23.45.043,
d ior	23.45.045, 23.45.057,
	23.45.059, 23.45.073 and
	23.45.075; amends §§
	23.45.018(D),
	23.45.032(D),
	23.45.046(D),
on orea	23.45.060(D), 23.47.016
ces er en	and 23.84.024; adds (C) to
	§ 23.47.022 and renumbers
	former (C) and (D) to be
COMPLE	(D) and (E); repeals §
Compage	23.86.022, Land Use Code
	(23.45, 23.47, 23.84)
114052	
114052	Establishes Kubota
	Gardens Account in the
	General Donations and Gift
5116-	Trust Fund (5.76)
114056	Adds Ch. 4.94, safety and
	performance awards;
	repeals Ord. 112928 (4.94)
114057	Amends entirety of Ch.
	25.05, environmental
	policies and procedures
	(25.05)
114074	Designates Martha
	Washington School as a
	Landmark (25.32)
114079	Amends §§ 23.49.152 and
1140/9	
	23.49.164; repeals Ord.
	114007, Land Use Code
11.4000	(23.49)
114082	Amends §§ 20.46.050,
	20.46.060, 20.46.070,
	20.46.080, 20.46.090,
	20.46.110 and 20.46.120;
	repeals § 20.46.140,
	utilization of women's and
	minority business
	enterprises by the City and
	City contractors (20.46)
114090	Amends §§ 25.05.680 and
	25.05.800, environmental
	policies and procedures
	(25.05)
114095	Amends §§ 4.38.401,
1140/3	4.38.402, 4.38.404 and
	4.38.504, voluntary
	deferred compensation plan
111000	(Repealed by 119159)
114098	Amends § 2.08.020,
	preparation of ballot title
	(2.08)
	A 11 C 1 1 4 TH 100
114107	Adds Subchapter III and 99
114107	Adds Subchapter III and §§ 3.12.210 and 3.12.220,
114107	3.12.210 and 3.12.220,
114107	3.12.210 and 3.12.220, agreements and payments
114107	3.12.210 and 3.12.220,

d amending graphics, ccuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

	Silber	ci10	
	a - mada	te Ille	
	114122	Amends § 17.40.020,	sorence of
		Bumbershoot Festival	STELCIO
	1 C - m 101	commission (Repealed by	
	A TOP III	118507)	
	114149	Rezone, Plat 36E, page 103	
		(Special)	42.00
	114155	Drainage utilities and	and the second
		storm drainage service	anience
		charge; amends § 5.48.050,	
	oos crea	utility businesses (5.48,	- hics
		5.76, 21.33)	drapilies,
	114156	Amends §§ 21.28.370 and	5,00
	ammle	21.76.040, sewer charges	
	COMPARA	and credits (21.76)	ocilracy
	114158	Amends §§ 21.24.030 and	CCC
	- 4 to CO	21.24.040, special sewer	
A	MU LO O	connection charges (21.24)	
	114160	Adds Ch. 4.100, dependent	
	000	care assistance program	
	<b>\$110.</b>	(4.100)	
5	114163	Approves Group Health	
		Cooperative Central	
		Campus Master Plan for	
		certain area (Special)	
	114171	Rezone, Plat 234W, page	
	114170	30 (Special)	
	114179	Adds § 17.18.010, Seattle	
		Center ice rink fees, and	
		authorizes Winterfest Ice Rink Concession	
		Agreement (Repealed by	
		115934)	
	114180	Amends §§	
	114180	17.16.100-17.16.120,	
		Seattle Center concession	
		agreements and fee	
		schedules (Repealed by	
		117776)	
	114181	Adds § 22.300.010 and	
		repeals § 22.300.020,	
		adopts National Electrical	
		Code, 1987 Edition	
		(22.300)	
	114182	Makes various changes to	
		National Electrical Code,	
		1987 Edition, including	
		addition of Chs. 1-3, §§	
		110-23, 215-10, 220-5,	
		220-23, 230-5, 230-33,	
		Tables 300(A) and (B), §§	
		300-18, 555-12 and 620-	

eference only. amending graphics,

Seattle Municipal Code Aate ference only. -102; amends Art. 100, §§ 110-13, 110-16, 210-8(a), Text provided for hi 215-2, 220-15, 220-17, 220-19, Table 220-20, §§ 230-1, 230-3, 230-28, 230-29, 230-40-230-43, amending 230-52, 230-62, 230-82, 250-1, 250-80, 250-81(a), 250-84, 300-1, 300-21, See ordinances 300-22, 310-4, 324-4, 331-4, 333-6, 334-3, 334-10, 336-3, 336-4, 336-10, 338-2, 347-2, sections for 348-1, 348-8, 370-1, 373-3. 374-1, 380-3, 380-10, 380-13, 384-14, 450-10, and tables and 450-13, 550-21, 555-4, 555-6, 620-21, 620-22, 620-37, 620-44, 620-71, this source file. 620-72, 620-81, 700-4, 700-12 and 700-16; repeals and adds §§ 90-1 and 90-2; repeals §§ 220-30, 220-31, 336-12, 336-16, Art. 342 and § 348-4; and incorporates Appendix 8, Emergency Illumination, of the Seattle Building Code (Not codified) 114183 Establishes City Light Department's Upward **Mobility Training Program** (Not codified) 114196 Adds §§ 23.34.013, 23.45.006.6 and 23.45.007.0-23.45.007.6; amends §§ 23.30.010, 23.45.002, 23.45.004, 23.45.006.5, 23.45.060(B), 23.45.076(B), 23.45.090(A), 23.45.112, 23.45.180, 23.45.182, 23.45.184, 23.45.190, 23.54.020(A), 23.84.030 and 23.86.018(H), Land Use Code and new SFA

> For current SMC, contact the Office of the City Clerk

114202

zone regulations (23.34, 23.45, 23.54, 23.84, 23.86)

Amends §§ 23.49.046(C),

23.49.148(E), 23.49.166(B) and 23.49.324(C), Land Use Code (23.49)

23.49.066(C), 23.49.096(D), 23.49.122(C), Seattle Municipal Code

June 2009 co 114205 up Amend subdivi 21,36.0

Text provide d for hi 21,36.0
adds §§ 21,36.0
amends 21,36.0
21,40.0
21,40.0
See ordinan ces crea 21,40.0
solid w

See ordinances creasections for this source this source that the second tables and to contain the source of the so

Amends § 21.36.010 by subdividing it into § § 21.36.010, 21.36.012, 21.36.014 and 21.36.016; adds § § 21.36.085, 21.36.087 and 21.36.095; amends § § 21.36.080, 21.36.070, 21.36.080, 21.40.010, 21.40.050, 21.40.060, 21.40.080 and 21.76.040, all pertaining to solid waste collection (21.36, 21.40, 21.76)

Amends § § 6.02.190A, 6.36.010, 6.36.020,

21.40.060, 21.40.080 and 21.76.040, all pertaining to solid waste collection (21.36, 21.40, 21.76)

Amends §§ 6.02.190A, 6.36.010, 6.36.020, 6.36.030, 6.36.050, 6.36.060, 6.36.070 and 6.36.080; repeals § 6.36.040, all pertaining to massage parlor licensing (6.02, 6.36)

Amends §§ 3.28.005 and 3.28.010, disposition of unclaimed property by the Police Department (3.28)

and Recreation Department facility fee schedule

(18.28)

114211 Rezone, Plat 66E, page 160 (Special) 114212 Amends § 5.48.050, business tax on utilities (5.48)114213 Rezone, Plat 16W, page 63 (Special) 114214 Rezones, Plats 16W and 16E, pages 63 and 64 (Special) 114215 Rezone, Plat 53W, page 134 (Special) 114220 Rezone, Plat 76W, page 179 (Special) 114222 Moratorium on acceptance of Master Use Permit and building permit applications for development in the Downtown Office Core 1 zone (23.76 footnote) 114225 Adds Ch. 6.270, adult entertainment; repeals § 6.204.060 and Ch. 6.280 (6.270)114227 Amends § 16.08.050, waterway operations and maintenance account (5.76, 16.08) 114228 Amends § 18.28.010, Parks ference only.

amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	0210
a and	ite ille
114229	Amends § 20.48.070,
	Board of Public Works
a combi	daily administrative
M FOL IN	charges (Repealed by
	117390)
114231	Amends §§ 10.03.140 and
	10.03.150, Department of
	Public Health
	immunization fees (10.03)
114232	Amends § 10.03.130,
CES OF	Department of Public
	Health personal health and
11111111	dental care fees (10.03)
114233	Amends § 17.16.015,
	Seattle Center facility fees
11/1007	(17.16)
114234	Establishes Data
	Processing Development
0010	Account in the Cumulative
111C.	Reserve Fund (Repealed by
114235	116497) Amends § 3.28.070, Police
114233	-
114236	Department fees (3.28)
114230	Amends § 5.44.030, business and occupation
	tax rate (5.44)
114241	Amends § 10.08.170(3),
114241	fee for certain responses to
	false alarms (Repealed by
	116702)
114243	Repeals and replaces Ch.
	22.900, 1989 permit fees
	(Repealed by 117405)
114244	Adds § 21.24.020(M),
	sewer permit fees
	(Repealed by 116421)
114245	Amends § 5.44.130(B),
	business and occupation
	tax; § 6.02.170, general
	business licenses; and §
	6.82.140(C), refrigeration
	service business licenses
111010	(5.44, 6.02)
114246	Amends § 22.602.040, fees
	for Fire Department
114240	permits (22.602)
114249	Amends § 5.48.220, allocation of revenues for
	landfill closure costs
	(Repealed by 117183)
114250	Adds § 3.58.090, fees and
117450	charges for Design
	Commission review (3.58)
	Commission Teview (3.36)

reference only.

reference only.

defined amending

graphics,

selections of

June 2009 co

Text provide

See ordinan sections for and tables a this source

Cihar	4- 5112
4 4400	Te me
114251	Adds §§ 11.14.113,
	11.14.703, 11.23.030,
a sor m	11.23.032, 11.28.020 and 11.72.075; amends §§
an res	11.14.115, 11.16.280,
	11.16.319, 11.26.260 and
	11.72.020, all pertaining to
	vehicles, traffic and
on orea	parking (11.14, 11.16, 11.23, 11.26, 11.28, 11.72)
114254	Moratorium on issuance of
	adult entertainment
comple	licenses except in certain
114256	locations (6.270 footnote)  Rezone, Plat 52W, Page
114230	132 (Special)
114260	Adds §§ 5.78.150 and
13300	5.78.160; amends §
	5.78.010, Gift Catalogue
2112.	Fund (5.78)
114264	Amends §§ 3.124.010,
	3.124.020, 3.124.030 and 3.124.040(E), City
	employee charitable
	contributions (3.124)
114267	Adds § 4.04.230(I) and
	amends § 4.04.260(B),
	personnel regulations
	(4.04)
114275	Moratorium on issuance of
	permits which would result in removal of mobile
	homes from certain sites
	(Expired)
114298	Adds §§ 21.16.390;
	amends §§
	21.16.020-21.16.060,
	21.16.065,
	21.16.070-21.16.110, 21.16.130-21.16.220,
	21.16.250-21.16.270,
	21.16.300, 21.16.310,
	21.16.330, 21.16.340,
	21.16.350, 21.16.360 and
	21.16.370; repeals and
	reserves §§ 21.16.120, 21.16.230, 21.16.240 and
	21.16.230, 21.16.240 and 21.16.320, sewers and
	drains (Expired)
114314	Adds new subsections (18)
	and (19) to § 4.04.030 and
	amends and renumbers §
	4.04.030 (18)-(31) to be
	(20)- (33); adds §§
	4.04.125 and 4.04.255, and
	amends § 4.13.020, intermittent positions and
	Civil Service exemptions
	(4.04, 4.13)

d to amending amphics, graphics, accuracy of

June 2009 co

Text provide

See ordinantes sections for and tables at this source

	a mada	te file
	114326	Amends § 17.16.015, 1989
		Seattle Center facility
	- 0 - m hi	license fee schedule (Not
	d for III	codified)
3	114328	Adds Ch. 22.600 (§§
		22.600.010-22.600.040),
		Seattle Fire Code; repeals
		former Ch. 22.600, §
		22.604.010, and Ords.
	oc crea	110871, 111001, 111221,
)	CE2 CZ	111656, 113815 (Repealed
		by 115405)
	114329	Seattle Fire Code
	Colliba	Supplement; amends
		specific sections of the
	1 40 C	Uniform Fire Code, 1985
	mato	Edition (Repealed by
٠	,88	115405)
	114330	Vacates a portion of Elliott
	5116.	Avenue (Special)
	114339	Amends §§ 20.46.020,
		20.46.030, 20.46.060(A),
		20.46.070 and 20.46.090
		(B)(1), women's and
		minority business
	111017	utilization (20.46)
	114345	Adds Ch. 3.121, charges
		for dishonored checks
	114247	(3.121)
	114347	Adds Ch. 5.14, procedures
		and documentation-City
	114349	moneys (5.14)  Moratorium on issuance of
	114349	permits for demolition or
		change of use of
		low-income housing
		downtown (Expired)
	114352	Amends § 6.212.045,
	114332	temporary maximum limit
		on taxicab licenses
		(Repealed by 118341)
	114353	Transportation Transportation
		Improvement Fund (5.76)
	114355	Repeals and replaces Ch.
		10.52, weeds and
		vegetation (10.52)
	114370	Amends §§ 21.04.430,
		21.04.440 and 21.76.040,
		water rates and utility
		credits for the elderly
		(21.04, 21.76)
	114371	Amends § 5.48.050, utility
		business tax on the sale of
		water (5.48)
	114272	A 1- 8 22 45 006 5

eference only.

l amending

graphics,

ccuracy of

For current SMC, contact the Office of the City Clerk

Amends § 23.45.006.5, Land Use Code (Repealed

Low Income Housing Fund

by 114887)

114372

114376

Seattle Municipal Code

June 2009 co 114377 up Adds st
9.25.03
Park Co
control
Text provide 114378 Amend

See ordinan sections for and tables a this source

	il bear	40 Sile
	4 - 440012	Te m
	114377	Adds subsection D to §
		9.25.030, enforcement of
	s cost hi	Park Code by animal
P	CI TOL 133	control officers (9.25)
	114378	Amends § 6.204.050,
		license fees for taxicabs
		and for-hire vehicles
	114382	(6.204) Adds § 23.47.012 (J) and
	114382	amends §§ 23.47.002 (F),
	ces crea	23.47.010 (A) and Chart B,
		and § 23.47.042 (D), Land
	100010	Use Code (23.47)
10	114392	Approves Seattle
		University Master plan, as
	140 00	amended, and
	and to co	corresponding portions of
0	'All on	Official Land Use Map
		(Special)
	114395	Adds §§ 22.804.020,
5	777	22.900.300 and 23.62.002
		(D); adds (c) to § 301 of
		Seattle Uniform Building
		Code amendments adopted
		by Ord. 113701, § 301(c)
		of Seattle National Electrical Code
		amendments adopted by
		Ords. 114181 and 114182,
		§ 301(d) of Seattle
		Uniform Mechanical Code
		amendments adopted by
		Ord. 113702, all pertaining
		to flood hazard protection,
		and adds Ch. 25.06,
		floodplain development
		(22.900, 25.06)
	114396	Adds §§ 11-15 to and
		amends §§ 2, 4 and 6 of
		Ord. 111244, Ratepayers
		Advisory Board for Pioneer Square Parking
		and Business Improvement
		Area (Not codified)
	114397	Rezone, Plat 225E, page 14
	114377	(Special)
	114400	Rezone, Plat 36E, page 103
	11.100	(Special)
	114415	Rezone, Plat 21E, page 74
		(Special)
	114416	Rezone, Plat 41W, page
		112 (Special)
	114425	Amends § 6.230.070 (F),
		steam engineer and boiler
		firemen's licenses (6.230)

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a mada	te file
	114428	Adds §§ 23.49.053,
		23.49.057, 23.49.073,
	- 0 - 10 h	23.49.077 and 23.49.136
	A FOR III	(D) and (E); amends §§
=	C ZO	23.49.008, 23.49.050,
		23.49.052 (A) and (B),
		23.49.054, 23.49.056,
		23.49.068, 23.49.070,
		23.49.072 (A) and (B),
	- 0403	23.49.074, 23.49.076,
M	res crea	23.49.096, 23.49.098 (B),
7		23.49.100, 23.49.102 (B)
	10	and (C), 23.49.106,
	COMPLE	23.49.136, 23.84.016,
	Com-2	23.84.030, 23.84.032 and
		23.84.036, and amends
	ad to Co	Plats 39E, 40W and 44W,
4	ma co	
7		Land Use Code (Not codified)
	111126	
	114436	Amends § 5.32.170, license
	8880	fees for amusement devices
		(5.32)
	114443	Amends § 3.124.040,
		annual charitable campaign
		(3.124)
	114446	Rezone, Plat 22W, page 25
		(Special)
	114450	Amends §§ 23.45.066 (B),
		23.49.050, 23.49.070,
		23.49.100 and 23.49.152,
		Land Use Code (23.45,
		23.49)
	114459	Adds § 21.49.058 and
		amends §§
		21.49.010-21.49.040,
		21.49.052, 21.49.055,
		21.49.057, 21.49.060 and
		21.49.080-21.49.140,
		Seattle City Light
		Department (21.49)
	114468	Adds (E) to § 5.44.030,
	 	renumbers former
		subsections (E) and (F) to
		be (F) and (G) and amends
		§ 5.44.130 (B), all to
		correct technical errors in
		Ord. 114236, Business and
		Occupation Tax (5.44)
	114473	Amends § 303(d) of Seattle
	1144/3	
		amendments to the 1985
		Uniform Building Code
		and Ord. 113701, and
		amends §§ 23.76.010 (F),
		23.76.032 and 23.76.060,
		Land Use Code (23.76)
	114470	D DI 4 10337 55

eference only.

amending

graphics,

ccuracy of

For cur restrictions and rates (21.04)

For cur restrictions and rates (21.04)

The Office of the City Clerk

Rezone, Plat 12W, page 55

Adds § 21.04.505 and amends § 21.04.500,

(Special)

114478

114479

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a conda	te file
	114484	Amends § 23.49.016 (B),
		Land Use Code (Not
	101	codified)
	114486	Amends §§ 23.49.050,
•		23.49.070, 23.49.100,
		23.49.152, 23.84.016 and
		23.84.024, Land Use Code
		(23.49, 23.84)
	114500	Adds Ch. 21.43, infectious
	114300	
	5/16/89	waste management (21.43) Initiative 31 Ordinance:
7	3/10/89	
	-10	Adds § 23.49.011 and
	comple	amends §§ 23.49.008(A),
	COMPLE	23.49.048(B),
		23.49.068(B), 23.49.096
	40 00	(B)(4), 23.49.098(B) and
		23.84.040, Land Use Code
•		CAP restrictions (23.49,
		23.84)
	114508	Amends § 10.52.010,
	777	vegetation nuisances
		obstructing sidewalks,
		streets and alleys (Not
		codified)
	114513	Amends § 4.13.010,
		exemptions from civil
		service (4.13)
	114515	Amends § 5.44.090 (A),
		exemption from business
		and occupation tax (5.44)
	114517	Adds § 5.40.075 and
	111317	amends § 5.40.070,
		admissions tax (5.40)
	114518	Adds § 11.74.030 and
	114310	amends §§ 11.14.113,
		11.23.030, 11.28.020,
		11.30.040, 11.72.020,
		11.72.075 and 11.74.010,
		*
		vehicles and commercial
		loading zones (11.14,
		11.23, 11.28, 11.30, 11.72,
	114502	11.74)
	114523	Amends § 3.28.070, fees
		for Police Department
	11.4505	records and services (3.28)
	114525	Amends § 23.45.006.5,
	11170	Land Use Code (23.45)
	114529	Adds §§ 4.20.082 and
		4.20.085 and amends §
		4.20.010, City employment
		and compensation (4.20)
	114531	Moratorium on issuance of
		permits for certain adult
		entertainment premises
		(6.270 footnote)
		\

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
114534	Suspends application of
114334	marital status
- 103	antidiscrimination
4 504 111	provisions in §§ 14.04.030,
C TO	14.04.040, 14.08.020,
	14.08.030 and 14.08.040 to
	3/1/90 pending further
	information from the
	Internal Revenue Service
- of orea	(footnotes in 14.04, 14.08)
114542	Continues moratorium on
	demolition or change of
	downtown low-income
COMPA	housing established by
	Ords. 113821 and 114349
4400	(Expired)
114543	Amends §§ 4.36.320 (B)
188	and 4.36.330 (B) and (D),
000 -	City Employees Retirement
5116.	System (4.36)
114547	Establishes an Abused
	Women's Shelter Fund in
	the General Donations and
11.45.40	Gift Trust Fund (5.76)
114549	Adds § 502.4.7 to and
	amends §§ 101.3.1,
	101.3.1.3, 201.1, 301.1,
	402.3, 502.2.1.2, 502.3.1.1 and renumbers former §
	502.4.7 to be 502.4.8 of the
	Seattle Energy Code
	adopted by Ords. 113058
	and 113059 (22.700.010
	footnote)
114561	Repeals and adds Ch.
	23.64, Airport Height
	Overlay District, amends §
	23.84.002, and amends
	portions of official land use
	map, Land Use Code
	(23.64, 23.84)
114571	Amends § 5.04.040, City
	participation in pooled
	investment system of King
	County (5.04)
114573	Adds § 3.33.200, retention
	of collection agency
	authorized for Municipal
	Court (3.33)
114582	Adds § 10.11.696, warning
	notices required for certain
	alcohol and cigarette sales
111500	(Repealed by 117001)
114592	Adds §§ 4.36.100 (K),
	4.36.103, 4.36.125,
	4.36.195, 4.36.197 and
	4.36.210 (I) and amends §
	4.36.170, City Employee's
114505	Retirement System (4.36)
114595	Amends § 5.48.170, utility
La CN	business tax (5.48)

eference only.

amending

graphics,

ccuracy of

For current Sing, State City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Clhar	calo
a - mada	te me
114606	Extends to 12/31/89
	moratorium on master use
1 0 - m hi	permit issuance for certain
M FOR III	changes affecting mobile
	homes (Expired)
114623	Adds §§ 23.49.046 (G),
	23.49.066 (G), 23.49.066
	(H), 23.49.122 (G),
	23.49.148 (F) and
- 0002	23.49.324 (G) and amends
CES CITE	§§ 23.44.036 (E),
	23.45.106 (C), 23.47.004
a minimit	(Chart A), 23.47.006 (C),
COMPL	23.49.144, 23.49.320,
	23.49.338 (A), 23.50.012
140 0	(Chart A), 23.54.015
ma to o	(Chart A), 23.66.122 (A),
1,8800	23.66.322 (A), 23.84.016,
	23.84.020 and 23.84.044,
C110	Land Use Code regulations
1110.	pertaining to work release
	centers and jails (23.44,
	23.45, 23.47, 23.49, 23.50,
	23.54, 23.66, 23.84)
114630	Adds Ch. 25.07, interim
111030	traffic standards for the
	Northgate area (Repealed
114625	by 116795)
114635	Amends §§ 12A.02.070,
	12A.04.120, 12A.08.020,
	12A.08.040, 12A.10.020,
	12A.10.040, 12A.10.060,
	12A.16.010 and
	12A.16.050, pertaining to
	punishment of crimes
	designated as
	misdemeanors or gross
	misdemeanors (12A.02,
	12A.04, 12A.08, 12A.10,
	12A.16)
114636	Amends § 12A.06.130 and
	repeals § 12A.06.140,
	violation of domestic
	violence no-contact orders
	(12A.06)
114645	Amends § 5 of Ord.
114043	114542, Downtown
	housing demolition
114640	moratorium (Expired)
114648	Adds Ch. 4.30,
	documentation of
	eligibility for and transfer
	of certain uses of sick leave
	and funeral leave, and adds
	§ 4.24.100; amends §§
	4.24.005, 4.24.035,
	4.24.040 and 4.28.020, sick
	leave and funeral leave
	(4.24, 4.28, 4.30)

eference only.

l amending

graphics,

ccuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	186	calo
	a - anda	te me
	114651	Amends § 2 of Ord.
		114531, moratorium on
	a com hi	issuance of permits for
7	4 FOL 132	certain adult entertainment
		premises (6.270 footnote)
	114655	Amends § 3.33.040,
		Municipal Court judicial
		departments (3.33)
Ī	114656	Adds § 25.08.515 and
	- or orea	amends §§ 25.08.500,
	SG2 CI On	25.08.660 and 25.08.800,
		noise control (25.08)
Ī	114658	Amends § 23.45.006.5,
	COMPA	Land Use Code (Repealed
		by 114887)
Ī	114680	Amends § 23.84.016, Land
	and to co	Use Code (23.84)
	114697	Amends §§ 4.38.401 and
		4.38.402, deferred
	2110	compensation plan for City
	Ille	employees (Repealed by
		119159)
Ī	114702	Amends § 5.44.030,
		business and occupation
		taxes (5.44)
	114708	Amends § 5.40.025 and
		creates §§ 5.40.026,
		5.40.027 and 5.40.028
		from subdivisions of §
		5.40.025, admission taxes
		(5.40)
	114712	Repeals and adds §
		22.502.015, adoption of
		Uniform Plumbing Code,
		1988 Edition and related
		standards as City Plumbing
		Code (Repealed by
Į		116594)
J	114713	Adopts Seattle
		Amendments to the 1988
		Uniform Plumbing Code;
J		repeals Ord. 113356
		(Repealed by 116594)

eference only.

amending

graphics,

ccuracy of

Seattle Municipal Code

June 2009 co 114723 up da te

Adds §
21.36.0
21.36.0
21.36.1
of Chs.
and am
21.36.0
21.36.0
21.36.0
21.36.0

See ordinances composections for compond tables and to this source file.

114725

Adds §§ 21.36.025, 21.36.026, 21.36.027, 21.36.028, 21.36.029, 21.36.096, 21.36.185 and 21.36.190; amends names of Chs. 21.36 and 21.44 and amends §§ 21.36.010, 21.36.012, 21.36.014, 21.36.016, 21.36.020, 21.36.021, 21.36.022, 21.36.023, 21.36.060, 21.36.085, 21.36.110, 21.36.120 and 21.36.180; renumbers § 21.44.070 to be § 21.44.090 and amends said section; repeals and replaces § 21.43.090; repeals §§ 21.44.010, 21.44.020, 21.44.030 and 21.44.050 and adds §§ 21.44.010-21.44.030, 21.44.040, 21.44.050, 21.44.060, 21.44.065, 21.44.070 and 21.44.080; and repeals § 21.36.024, solid waste management and handling (21.36, 21.43, 21.44)

Amends §§ 23.34.008 (D), 23.49.036 (E), 23.49.126

and 23.84.025, Land Use Code (23.34, 23.49, 23.84) 114736 Amends § 11.23.030, commercial loading permit requirements (11.23) 114738 Adds § 3.14.840, contracts with sheltered workshops for provision of janitorial services (3.18) 114745 Amends §§ 23.48.002, 23.82.010 and Map 44E, page 117 of zoning maps, Land Use Code (Repealed by 115002) 114767 Adds subsections  $(\overline{C})$ - $(\overline{E})$ to § 3.14.4050, Neighborhood Matching Fund (Repealed by 115345) 114776 Amends § 23.45.006.5, Land Use Code (Repealed by 114887) 114777 Rezones portions of Alki neighborhood (Special) 114779 Amends § 5.48.050, business tax on utilities (5.48)114781 Amends § 21.24.020, side sewer or drain connections (Repealed by 116421)

ference only.

amending

raphics,

curacy of

For current SMC, Constitution of the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipai co	e sile
4 - 440012	Ife Page
114782	Amends §§ 21.28.370,
	21.33.030 and 21.76.040;
101	repeals § 21.33.040,
4 504 111	sewers, drainage and
C To	wastewater charges (21.33,
	21.76)
114795	Adds subsections (D) and
114793	
	(E) to § 23.45.006.5, Land
	Use Code (Repealed by
THE CIES	114887)
114804	Extends to 3/31/90
	moratorium on master use
	permit issuance for certain
COMPA	changes affecting mobile
	homes (Expired)
114810	Amends Ch. 22.900, 1990
ad to co	permit fees (Repealed by
12202	117405)
114811	Adds § 4.04.045,
114011	temporary employment
7116.	
	service for the City; repeals
111010	§ 3.14.600 (4.04)
114812	Amends § 18.28.010, fee
	schedule for Parks and
	Recreation Department
	facilities (18.28)
114813	Amends § 9.26.020, cat
	and dog license fees (9.26)
114814	Closes the Fire Protection
	Facilities Fund established
	by Ord. 96326 and
	transfers remaining
	balances to Fire Protection
	Facilities Account (5.76)
114815	Amends § 22.202.050,
114613	Housing and Abatement
	Account in the
	Construction and Land Use
	Fund; closes the Housing
	and Abatement Revolving
	Fund created by Ord.
	106319 (5.76, 22.202)
114816	Amends § 3.33.040, Seattle
	Municipal Court (3.33)
114818	Closes the Arterial
	Development Fund
	established by Ord. 97053
	and transfers balances to
	subfund in the
	Transportation
	1
	I Improvement Fund (5 76)
114823	Improvement Fund (5.76)
114823	Amends § 5.44.130 (A),
114823	Amends § 5.44.130 (A), Business and Occupations
	Amends § 5.44.130 (A), Business and Occupations Tax exemptions (5.44)
114823 114825	Amends § 5.44.130 (A), Business and Occupations Tax exemptions (5.44) Amends § 20.48.070,
	Amends § 5.44.130 (A), Business and Occupations Tax exemptions (5.44) Amends § 20.48.070, Board of Public Works
	Amends § 5.44.130 (A), Business and Occupations Tax exemptions (5.44) Amends § 20.48.070, Board of Public Works daily administration
	Amends § 5.44.130 (A), Business and Occupations Tax exemptions (5.44) Amends § 20.48.070, Board of Public Works

ference only.
amending
graphics,
ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipar -	0.10
a mada	te file
114826	Amends §§
	3.14.2010-3.14.2060 by
a com hi	repealing § 3.14.2010,
d tor	renumbering §§
	3.14.2020-3.14.2060 to be
	3.14.2010-3.14.2050
	respectively, and amending new § 3.14.2020, Office
	for Strategic Human
- 0402	Services Planning
res crea	(Repealed by 115949)
114828	Amends § 10.25.020, fee
	for reports on water supply
COMPA	or sewage disposal systems
,	(10.25)
114829	Amends §§ 22.602.040 (A)
and to co	and (B), 22.602.050,
	22.602.070 and
	22.602.080, fees for Fire
file.	Department permits for hazardous materials and
	processes (22.602)
114831	Amends § 17.16.015, fees
114031	for use of certain Seattle
	Center facilities (17.16)
114832	Adds § 25.08.426 and
	amends § 25.08.570,
	variance and plan review
	fees for work involving
	control of noise (25.08)
114833	Amends § 10.03.130,
	Public Health Department
	fees for personal health
	care and dental services (10.03)
114834	Adds § 22.202.060 and
114634	amends §§ 22.206.160
	(C)(1) and (4), 22.206.230
	(B), 22.206.240 (A) and
	22.206.280 (A), rental
	housing registration
	program (22.206)
114835	Amends §§ 21.49.020,
	21.49.030, 21.49.040,
	21.49.052, 21.49.055,
	21.49.057, 21.49.058,
	21.49.060 and 21.49.110, and repeals § 21.56.040,
	City Light Department
	electrical rates (21.49)
114837	Amends §§ 10.03.020,
	10.03.110 and 10.03.120,
	Health Code fees (10.03)
114838	Adds §§ 10.11.943,
	10.11.944 and 10.11.945,
	and amends § 10.11.941,
	fees pertaining to
	food-service
	establishments (Repealed
	by 117001)

reference only.

ng amending graphics,
ly ccuracy of

June 2009 co Text provide

See ordinan

sections for and tables a this source

C	The	. 6310
	a - anda	te file
	114839	Amends § 11.31.120,
		monetary penalties for
	a com hi	failure to pay parking
	4 FOL 188	tickets (11.31)
	114841	Amends § 11.66.120, speed
		limits for trains (11.66)
	114847	Amends § 5.48.150, Utility
		Business Tax overpayment
		or underpayment (5.48)
	114848	Amends § 6.202.460,
X	CES CS	liability insurance for
		certain businesses (6.202)
	114850	Adds § 5.48.068 and
	COMPA	amends § 5.48.070, utility
		tax on network telephone
ļ	140 (1	service (5.48)
	114858	Plat 28W, page 87, rezone
Ů		(Special)
	114862	Moratorium on issuance of
	<b>4116</b> .	use permits for certain
	880	adult uses (6.270 footnote)
	114863	Amends Exhibit A of Ord.
		100475 and Map XIIA,
		boundaries of Pike Place
		Historical District (23.49
ŀ	114074	footnote)
	114864	Amends §§ 14.08.020,
		14.08.040-14.08.090 and 14.08.190, discrimination
		in housing and public
		accommodations (14.08)
ŀ	114865	Amends §§ 22.220.060 (A)
	114003	and (C), 22.220.070 (B)
		and (C), 22.220.070 (B) and (C), 22.220.080 (B)
		and (E), 22.220.090
		(B)-(E), 22.220.130
		(A)-(D) and (G),
		22.220.140 (B), 22.220.160
		(A), 22.220.170,
		22.220.180 and 22.220.190
		(B), Downtown Housing
		Maintenance (22.220)
	114866	Adds § 23.40.006 and
		amends § 23.84.016 (H),
		Land Use Code (23.84)
	114872	Adds §§ 3.14.960 and
		3.14.970; amends §§
		3.14.900, 3.14.910 and
		3.14.940, Seattle
		Commission for Lesbians
		and Gays (3.14)
	114874	Moratorium on decisions
		and permits involving
		low-income housing
		downtown (Expired)

reference only.

reference only.

and ity
amending
praphics,
curacy of

114887

Text provided for

See ordinan sections for and tables this source

114875 Adds §§ 23.44.050 (L) and 23.45.152 (L); amends §§ 23.44.006 (G) and (H), 23.44.022 (A) and (C)-(L), 23.44.050 (G), 23.45.090 (A)-(D), 23.45.096 (E), 23.45.098 (A), 23.45.102, 23.45.112, 23.45.122 (B), 23.45.152 (F), 23.47.004, 23.50.012, 23.54.015 (A) and Chart A, 23.84.008, 23.84.012 and 23.84.018, Land Use Code (23.44, 23.45, 23.47, 23.50, 23.54, 23.84) 114881 **Establishes Family** Violence Project Account in the General Donations

and Gifts Trust Fund (5.76) 114886 Repeals and adds § 23.34.013 and repeals §§ 23.34.014, 23.34.016, 23.34.018, 23.34.020 and

(23.34)

Adds §§ 23.45.009 and 23.45.010; amends §§ 23.30.010, 23.45.004, 23.45.006, 23.84.024, 23.84.028, 23.84.038, 23.84.048, 23.86.012 and 23.86.016; repeals and adds §§ 23.45.008, 23.45.010, 23.45.012, 23.45.014, 23.45.015, 23.45.016 and 23.45.018; changes all Title 23 references to the SFA zone to be to the LDT zone; and repeals §§ 23.45.006.5, 23.45.006.6, 23.45.007.0-23.45.007.6, 23.45.017, 23.45.022, 23.45.024, 23.45.026, 23.45.028-23.45.032, 23.45.036, 23.45.038, 23.45.040 and 23.45.042-23.45.046, Land Use Code (23.30, 23.45,

restrictions (Repealed by

23.34.013, Land Use Code

23.84, 23.86) 114888 Amends §§ 23.30.010, 23.45.008, 23.45.009, 23.45.010, 23.45.011, 23.45.012, 23.45.014, 23.45.016, 23.45.018 and 23.84.048, Land Use Code (23.30, 23.45, 23.84)114889 Amends § 21.36.022, solid waste receptacle use

ference only. amending

117441) For current the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	, ipus	te file	
	114893	Personnel Director	1
	114895		0
		employee benefits duties;	
	1 Co# 111	creates the Health Care	
•	U IOI	Fund and Group Term Life	
		Insurance Fund and repeals	
		Chs. 4.48, 4.52, 4.56 and	
		Ords. 83834, 95466, 98295	L
		and 100862 (Repealed by	
		115365)	
	114895	Amends §§ 5.32.140 and	•
V	ces cross	5.32.160, Revenue Code,	
7		and §§ 6.42.010-6.42.040	1
	10	and 6.42.060, panorams	Ŀ
	COMPLE	and peepshows (5.32, 6.42)	
	114898		۱
	114898	Amends §§ 21.28.370,	Ь
	1 40 C	21.33.030 and 21.76.040,	
	ma to o	sewers and utility credits	
•		for the elderly (21.33,	
		21.76)	
	114899	Adds § 25.07.090 and	
	Ille.	amends §§ 25.07.040 and	
		25.07.060(B), interim	
		traffic standards for the	
		Northgate area (Repealed	
		by 116795)	
	114900	,	
	114900	Creates Department of	
		Parks and Recreation Open	
		Space and Trails Bond	
		Program and Open Spaces	
		and Trails Bond Fund	
		(5.76)	
	114903	Amends § 6.212.045,	
		temporary maximum limit	
		on taxicab licenses	
		(Repealed by 118341)	
	114909	Amends § 23.45.009, Land	
		Use Code (23.45)	
	114910	Amends §§ 23.45.090 and	•
	114910	23.45.092, Land Use Code	
		(23.45)	
	11/013	,	1
	114913	Amends § 5 of Ord.	
		114874, moratorium on	
		demolition of certain	
		low-rental housing	
		(Expired)	
	114917	Rezone, Map 15W, page	
		61 (Special)	
	114918	Rezone, Map 44W, page	
		116 (Special)	
	114928	Rezones in Southwest	•
	11.520	District, Maps 48W, 48E,	
		54W, 54E, 55W, 55E, 60E,	
		61W, 61E, 62W, 62E,	
		68W, 68E, 69W, 69E,	
		70W, 75W, 76W, 76E and	
		77W (Special)	
	114947	Establishes Discovery Park	
		Memorial Account in	
		General Donations and Gift	
		Trust Fund (5.76)	b
Į.			1

eference only.

The second of the second of

June 2009 co

Text provide

See ordinantes sections for and tables at this source

- Ag	te file
114965	Adds § 4.70.025, amends §
	4.70.010 and repeals §§
103	4.70.020, 4.70.030 and
	4.70.040, use of passenger
Cl 102	automobiles for City
	business (4.70)
114966	Adds § 5.78.170, donations
114900	
	administered by the City
11.1002	Comptroller (5.78)
114993	Designates the Colman
	Building as a Landmark
111001	(25.32)
114994	Designates Summit
Collina	School/Northwest School
	as a Landmark (25.32)
114995	Designates the Maryland
ma to	Apartments as a Landmark
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	(25.32)
115001	Rezone, Plats 69W and
Sile.	69E, pages 165 and 166
71100	(Special)
115002	Adds §§ 23.04.040,
	23.54.015(K), 23.86.036
	and Ch. 23.69 (Major
	Institution Overlay
	District); amends §§
	23.30.010, 23.44.022,
	23.45.004, 23.45.090(B),
	23.45.122, 23.47.004 Chart
	A, 23.47.006(B), 23.50.012
	Chart A, 23.50.014(B),
	23.54.010(B),
	23.54.015(A), (D) and (G),
	23.56.010(A), 23.76.036,
	23.76.042(B),
	23.76.046(B),
	23.76.062(B), 23.84.018,
	23.84.025 and 23.86.012;
	and repeals §§ 23.34.058,
	23.34.060, 23.34.062,
	23.34.064, 23.34.066,
	23.34.068, 23.55.026 and
	Chs. 23.48, 23.81 and
	23.82, Land Use Code
	(23.04, 23.30, 23.44, 23.45,
	23.47, 23.50, 23.54, 23.69,
115008	23.76, 23.84, 23.86) Extends to 5/31/91
113006	moratorium on master use
	permit issuance for certain
	changes affecting mobile
115010	homes (Expired)
115010	Adds § 5.16.050, contract
	to provide for handling
	City's warrant overdrafts
	(5.16)
115014	Amends § 6.212.045,
113014	
113014	taxicab license moratorium
113014	taxicab license moratorium procedures (Repealed by 118341)

reference only.

The state of t

June 2009 co

Text provide

See ordinant sections for and tables a this source

		te file
	115016	Amends § 3.33.040, Seattle
	0,0	Municipal Court
	- 0 101	departments (3.33)
	115017	Amends § 2.04.430,
Š		candidate campaign
		expenditure limitations (2.04)
	115035	Amends § 2 of Ord.
		114777, rezones portions
	soc crea	of Alki neighborhood
		(Special)
	115036	Designates the C.H. Black
	- ammle	House and Gardens as a
	COMPLE	Landmark (25.32)
	115037	Designates the Mann
	1 40 CC	Building as a Landmark
	ma to	(25.32)
C.	115038	Designates the Exchange
	0210	Building as a Landmark
	Sile.	(25.32)
5	115039	Rezone, Plat 19E, page 70
		(Special)
	115040	Adds §§ 11.20.340 and
		11.20.350 on financial
		responsibility of vehicle
		operators, and amends §§
		11.31.010, 11.34.020,
		11.59.040 and 11.59.090,
		Traffic Code (11.20, 11.31,
	115041	11.34, 11.59)
	115041	Amends §§ 25.08.100 and
		25.08.425, noise control
	115043	(25.08)
	113043	Adds § 23.45.017, amends
		§§ 23.44.022(B)-(D), 23.45.006(A),
		23.45.009(B),
		23.45.014(C)-(E),
		23.45.016(B), 23.45.048,
		23.45.090, 23.45.146(C),
		23.47.004 Chart A,
		23.50.012 Chart A,
		23.54.015 Chart A,
		23.55.026(B) and (C),
		23.69.012(B),
		23.69.022(A) and
		23.84.018, and repeals §
		23.16.010, Land Use Code
		(23.44, 23.45, 23.47, 23.50,
		23.54, 23.55, 23.69, 23.84)
	115050	Amends § 2.08.040, form
		of petition for enactment of
		an ordinance (2.08)
	115055	Amends §§ 5.48.050(H)
		and 5.48.220, solid waste
		collection tax and
		apportionment of revenue
		for landfill closure costs
		(5.48)

d amending
raphics,
accuracy of

Seattle Municipal Code

June 2009 co 115058 update

Adds § amends 23.84.0 (23.40, 115060 Rezone

See ordinant sections for and tables a this source

	. 0210
a mada	te tille
115058	Adds § 23.40.006 and
	amends §§ 23.84.016 and
1 0 - m h1	23.84.030, Land Use Code
4 FOL 111	(23.40, 23.84)
115060	Rezone; changes zoning
	maps (Special)
115087	Moratorium on issuance of
	use permits for certain
	adult uses (6.270 footnote)
115089	Low Income Housing Fund
ces crea	(5.76)
115103	Adds Ch. 22.903,
1000016	condominium conversion
COMPA	(22.903)
115105	Amends §§
113103	22.902.010-22.902.040,
and to co	22.902.060, 22.902.080,
The same	22.902.120, 22.902.130
	and 22.902.160-22.902.180
C:10	and repeals § 22.902.050,
THE	cooperative conversion
	(22.902)
115118	Amends § 5.44.030(D),
113116	business and occupation
	-
115135	tax (5.44)
115155	Amends §§ 23.50.014(B)
	and 23.60.730(C), Land
115120	Use Code (23.50, 23.60)
115138	Rezone; amends zoning
115139	maps (Special)
115139	Amends §§ 10.03.013(A),
	10.13.100(A), 10.13.120
	(A) and (F), 10.13.150(B),
	10.13.160(G), 10.13.400
	and 10.13.410, and repeals
	§ 10.13.160(F), health code
	and meat, rabbit, poultry
	and aquatic foods (10.03,
115141	10.13)
113141	Adds §§
	22.210.010-22.210.200, tenant relocation
	***************************************
	assistance, and repeals former Ch. 22.210, housing
115142	preservation (22.210)
115142	preservation (22.210) Amends § 6.212.117,
115142	preservation (22.210) Amends § 6.212.117, taxicab charges for extra
115142	preservation (22.210) Amends § 6.212.117, taxicab charges for extra passengers and baggage
	preservation (22.210) Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)
115142 115157	preservation (22.210) Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341) Amends § 21.04.505,
	preservation (22.210) Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341) Amends § 21.04.505, charges for water use
	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations
115157	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations (21.04)
	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations (21.04)  Long-term disability
115157	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations (21.04)  Long-term disability insurance program; amends
115157	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations (21.04)  Long-term disability insurance program; amends § 3 of Ord. 114893, Group
115157	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations (21.04)  Long-term disability insurance program; amends § 3 of Ord. 114893, Group Term Life Insurance Fund,
115157	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations (21.04)  Long-term disability insurance program; amends § 3 of Ord. 114893, Group Term Life Insurance Fund, and repeals Ch. 4.60,
115157	preservation (22.210)  Amends § 6.212.117, taxicab charges for extra passengers and baggage (Repealed by 118341)  Amends § 21.04.505, charges for water use restriction violations (21.04)  Long-term disability insurance program; amends § 3 of Ord. 114893, Group Term Life Insurance Fund,

eference only.

of amending aphics, accuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

	har	. 6210
		late Ille
115	160	Adds Ch. 5.68, use tax on
		natural or manufactured
	0-41	gas (5.68)
115	164	Amends §§ 22.47.016,
		23.47.036(B) and
		23.50.016, Land Use Code
115	1.65	(23.47, 23.50)
115	165	Adds Subchapter V heading before § 23.69.022
		(Development Outside a
CE	s cre	Major Institution Overlay
		District) and adds §§
		23.64.021 and 23.69.023;
	OHR	amends § 23.54.015(A)
		and redesignates §
	140	23.54.015(K) to be new §
111		23.54.016; renumbers
		Subchapter V to be
		Subchapter VI of Chapter
E	110.	23.69; amends §§
		23.69.022(E),
		23.69.024(B), 23.69.026(B),
		23.69.026(B), 23.69.028(B) and (D),
		23.69.034(B),
		23.69.035(B) and
		23.76.036(A), Land Use
		Code (23.54, 23.69, 23.76)
115	171	Adds § 12A.20.050,
		drug-traffic loitering
115	102	(12A.20)
115	183	Adds §§
		22.904.400-22.904.470 and amends § 22.904.010,
		mobile homes and mobile
		home parks (22.904)
115	193	Amends § 17.19.010(B),
		Seattle Center parking for
		Goodwill Games (17.19)
115	197	Rezone; Plats 4W, 4E, 5W,
		5E, 6W, 6E, 7W, 10E,
		11W, 14W, 14E, 15W,
		16W, 16E, 17W, 22E, 25W, 27W, 30E, 32E, 36E,
		37W and 37E, pages 40-46,
		52, 53, 59-65, 76, 81, 85,
		92, 96 and 103-105
		(Special)
115	229	Amends § 3 of Ord.
		115055, removal of
		business tax on collection
		of recyclables (5.48)
115	231	Adds §§
		21.40.080(A)(14)-(A)(17)
		and amends §§ 21.36.012,
		21.36.016(11) and
		21.40.050(A)(5) and $(A)(9)$ solid waste utility
		(A)(9), solid waste utility
115	237	(A)(9), solid waste utility (21.36, 21.40)
1150	237	(A)(9), solid waste utility

eference only.

I amending graphics, ccuracy of

For current 31/1 Ord. 115060 (Special)
the Office of the City Clerk

June 2009 co

Text provide

See ordinand sections for and tables a this source

Adds § 20.46.140 and amends § § 20.46.020(E)-(H) and 20.46.090(B), women's and minority business enterprises (20.46)  115244	Cale		te file
amends § \$ 20.46.020(E)-(H) and 20.46.090(B), women's and minority business enterprises (20.46)  115244	115239	11 Deser	Adds § 20.46.140 and
20.46.020(E)-(H) and 20.46.090(B), women's and minority business enterprises (20.46)  115244 Amends § 23.76.012(B), Land Use Code  115245 Corrects maps of Plat 49W, page 126, attached to Ord. 113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  115266 Repealed by 118314)  115281 Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283 Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303 Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323 Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.100, 11.50.150, 11.50.1100, 11.50.150, 11.50.1100, 11.50.150, 11.50.180, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
20.46.090(B), women's and minority business enterprises (20.46)  115244 Amends § 23.76.012(B), Land Use Code  Corrects maps of Plat 49W, page 126, attached to Ord. 113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  11523 Adds § 11.14.266 and amends §§ 11.40.040, 11.40.040, 11.50.040, 11.50.040, 11.50.100, 11.50.100, 11.50.150, 11.50.140, 11.50.150, 11.50.140, 11.50.150, 11.50.180, 11.50.150, 11.50.180, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,		- 101	
minority business enterprises (20.46)  115244  Amends § 23.76.012(B), Land Use Code  Corrects maps of Plat 49W, page 126, attached to Ord. 113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.150, 11.50.100, 11.50.150, 11.50.1100, 11.50.150, 11.50.1100, 11.50.150, 11.50.110, 11.50.150, 11.50.1100, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	4		
enterprises (20.46)     115244			* * *
Amends § 23.76.012(B), Land Use Code  Corrects maps of Plat 49W, page 126, attached to Ord. 113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.100, 11.50.120, 11.50.140, 11.50.120, 11.50.140, 11.50.120, 11.50.140, 11.50.1270 and 11.50.280, Traffic Code (11.14, 11.40,			-
Land Use Code  Corrects maps of Plat 49W, page 126, attached to Ord. 113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.040, 11.40.040, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.120, 11.50.180, 11.50.120, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115244		
Corrects maps of Plat 49W, page 126, attached to Ord. 113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends § § 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends § § 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends § § 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.120, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
page 126, attached to Ord. 113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends § § 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends § § 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends § § 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.100, 11.50.120, 11.50.140, 11.50.120, 11.50.140, 11.50.120, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115245		
113658 and Plat 54E, page 137, attached to Ord. 114928, rezones (Special)  Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  115266 Repealed by 118314)  115266 Repeales § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303 Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323 Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.100, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	1102.0	- orea	
137, attached to Ord. 114928, rezones (Special) Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405) Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314) Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer) Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified) Amends § § 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14) Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323 Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.100, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	TOP.	2 CI co	
115253			
Amends Ch. 22.900, 1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.120, 11.50.140, 11.50.120, 11.50.180, 11.50.120, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,		10	
1990/1991 permit fees (Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  115266 Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  115281 Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283 Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303 Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323 Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115253		` <u> </u>
(Repealed by 117405)  Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  115266  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  115281  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	113233	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
Adds § 5.44.320, business and occupation tax refunds (Repealed by 118314)  115266 Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  115281 Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283 Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303 Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323 Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
and occupation tax refunds (Repealed by 118314)  115266  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  115281  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115250	<del>1 to ct</del>	
(Repealed by 118314)  Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	113239		-
Repeals § 4.10.050, limited duty assignments for pregnant employees (Repealer)			
duty assignments for pregnant employees (Repealer)  115281  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115000		
pregnant employees (Repealer)  115281  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115200	160	1 0
(Repealer)  115281  Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			3 0
Places proposed amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
amendment to § 24 of Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115001		
Charter Art. IV on 11/6/90 general election ballot, claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115281		
general election ballot, claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
claims on city for damages (Not codified)  115283  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
(Not codified)  Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			2
Amends §§ 2.14.050(B) and 2.14.060, election pamphlets and write-in candidates (2.14)  115303  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			•
and 2.14.060, election pamphlets and write-in candidates (2.14)  115303  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115000	1	
pamphlets and write-in candidates (2.14)  115303  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115283		
candidates (2.14)  Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			*
Amends §§ 4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			1 1
4.96.010-4.96.030, recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115202	1	` ′
recognition ceremonies (4.96)  115323  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115303	i	0.0
(4.96)  Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
Adds § 11.14.266 and amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			<u> </u>
amends §§ 11.40.040, 11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	11.7000		
11.40.060, 11.40.220, 11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,	115323	}	_
11.40.240, 11.50.040, 11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			,
11.50.060, 11.50.100, 11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
11.50.120, 11.50.140, 11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
11.50.150, 11.50.180, 11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
11.50.270 and 11.50.280, Traffic Code (11.14, 11.40,			
Traffic Code (11.14, 11.40,			
			· · · · · · · · · · · · · · · · · · ·
11.50)			` '
			11.50)

eference only.

amending

amending

graphics,

curacy of

Text provided for hi

See ordinan sections for comp and tables and this source file.

Amends § 302(c)II of the Seattle Amendments to the Uniform Building Code, 1985 Edition; adds §§ 23.22.052(F), 23.24.035, 23.40.030, 23.47.014(G), 23.49.002(D), 23.50.029, 23.50.030(H), 23.50.032(H) and Ch. 23.53 (requirements for streets, alleys and

easements); amends §§ 23.22.052(C), 23.44.014, 23.44.016, 23.45.006(E), 23.45.008(D), 23.45.014(A),

115337

23.45.076(B), 23.47.002(D), 23.47.032(D),

23.45.018(B), 23.45.056, 23.45.060(B), 23.45.072, 23.47.048(A), 23.49.022, 23.50.002(D), 23.50.016, 23.54.030, 23.76.006(B), 23.84.002, 23.84.006, 23.84.008, 23.84.016, 23.84.030, 23.84.032, 23.84.036, 23.86.007(B), 23.86.010(B) and 23.86.012; repeals §§ 23.50.040 and 23.54.010, Land Use Code (23.22, 23.24, 23.40, 23.44, 23.45, 23.47, 23.49, 23.50, 23.53, 23.54, 23.76, 23.84, 23.86) Adds §§ 3.56.060 and 3.56.070, deaccessioning and disposition of surplus artworks (Repealed by 121006) Amends §1 of Ord.

3.14.1000, 3.14.1010 and

115341 109261, restricted parking zone in the vicinity of Providence Medical Center (11.73)115345 Adds Ch. 3.35, Department of Neighborhoods; amends § 3.20.030; and repeals §§

3.14.4000-3.14.4050, Department of Human Resources (3.20, 3.35) 115352 Establishes First Hill/Capitol Hill neighborhood Restricted Parking Zone (11.73) 115365 Adds § 4.50.010; repeals § 1 of Ord. 114893 and § 2 of Ord. 115159, City officer and employee

ference only.

amending

health and welfare benefits (4.50)Clerk the Office of the City

Seattle Municipal Code
June 2009 co 115376 update
Amend
drainag

Text provide

See ordinan sections for and tables a this source

Cihar	40 5112
4 4400	IE III
115376	Amends § 21.33.030,
	drainage service charges,
1 Sor hi	and § 21.76.040, utility
GO Int	low-income rate assistance (21.33, 21.76)
115377	Amends § 21.24.020, side
	sewer or drain connection
	permits (Repealed by
	116421)
115385	Adds Ch. 25.09, interim
	regulations for critical
115386	areas (Expired) Amends § 5.48.050,
113380	business tax on utilities
Cosses	(5.48)
115388	Adds § 3.114.170; amends
and to co	§§ 3.114.020(C) and
	3.114.030, selection of
000 -	consultants to write
sile.	environmental impact
115200	statements (3.114)
115390	Adds §§ 7.20.115 and 7.20.125; amends §§
	7.20.125, amends gg 7.20.030, 7.20.040 and
	7.20.050; 7.20.040 and 7.20.150; repeals and
	replaces § 7.20.020,
	floating homes and floating
	home moorages (7.20)
115400	Adds § 3802(h)(6) to
	Seattle Building Code
	Amendments, sprinkler systems (Not codified)
115405	Repeals and replaces Ch.
113403	22.600; repeals §
	22.604.010 and Ords.
	114328 and 114329,
	Uniform Fire Code and
	Uniform Fire Code
115104	Standards (22.600)
115406	Seattle Fire Code
	Standards (Repealed by 116334)
115411	Rezone, Plat 13E, page 58
	(Special)
115414	Extends to 9/16/91
	expiration date of certain
	Alki and Southwest Seattle
	neighborhood interim zoning as set by Ords.
	114928 § 2 and 115035 § 1
	(Not codified)
115418	Amends § 6.212.045,
	moratorium on issuance of
	new taxicab licenses
	(Repealed by 118341)
115420	Amends § 4.13.010,
	procedure for exemption of
	various positions from the
	civil service system (4.13)

reference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinanger sections for and tables a this source

Clhar	0210
a mada	te tile
115422	Amends § 5.48.050(I),
	business tax on utilities
115122	(5.48)
115423	Amends § 10.03.130, charges for Public Health
	Department services
	(10.03)
115424	Amends §§ 21.28.080 and
	21.28.370, water service
DOS CITE	charges; amends name of
	Ch. 21.76 to be "Low-Income Utility"
	Credits for Qualified
COMPLE	Customers," and
	amends §§ 21.76.010 and
1 40 C	21.76.030-21.76.050
	(21.28, 21.76)
115425	Adds § 17.19.010(F) and (G), amends §§
C110.	17.19.010(C), (D) and
1116.	17.19.020(B), and amends
	name of Ch. 17.19, Seattle
	Center Parking and
117121	Monorail facilities (17.19)
115426	Adds §§ 22.900.220(E) and
	(F) and 22.900.255, and amends §§
	22.900.030-22.900.080,
	22.900.170(A),
	22.900.205, 22.900.220(E)
	and (F) and 22.900.240(A),
	(B), (E)-(G), (I), (J), (L) and (M), 1990-91 permit
	fees (Repealed by 117405)
115427	Rezone, Plat 14E, page 60
	(Special)
115429	Amends §§ 3.18.200(A)
	and 3.18.240, Department of Administrative Services
	(3.18)
115434	Amends § 5.44.030,
	business and occupation
	tax (5.44)
115435	Amends §§ 5.44.130 and
	5.44.140, business and occupation tax (5.44)
115437	Amends § 5.16.050,
113 137	extends bank agreement for
	the holding of warrant
	overdrafts (5.16)
115438	Amends overlegal
	trucking/parking fee schedule authorized by §
	11.23.120 (Not codified)
115439	Amends § 10.03.013(A),
	Health Code fees (10.03)
115440	Amends § 18.28.010, fee
	schedule for use of park
	and recreational facilities (18.28)
	(10.20)

reference only.
d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Silber	40 Sile
4 4400	ILE 11110
115441	Amends §§ 10.03.056 and
	10.03.110, Health Code
see hi	fees (10.03)
115442	Amends § 21.44.050(B),
	solid waste permit renewal
117110	fees (21.44)
115443	Amends § 22.504.010,
	plumbing permit fees (22.504)
115445	Amends § 20.48.070,
	Board of Public Works
-10	contract administration
comple	charges (Repealed by
	117390)
115446	Amends § 17.16.015, 1991
ad to co	charges for use of Seattle
115448	Center facilities (17.16)
115448	Amends §§ 3.20.010,
0210	3.20.030 and 3.20.040,
1116.	Department of Human
115450	Services (3.20) Amends §§ 21.52.230 and
113430	
	21.32.250, low-income weatherization program
	(21.52)
115457	Amends § 3.124.020(C)
113437	and (D), combined
	charities campaign
	participation conditions
	(Not codified)
115460	Adds § 4.36.390,
113 100	agreement for portability of
	retirement benefits
	(Repealed by 118498)
115464	Adds § 4.04.070(I),
	personnel regulations; adds
	§ 4.08.140(I), public safety
	civil service, and adds
	Subchapter III to Ch. 4.20
	(4.20.600[800]-4.20.650[8
	50]), reporting and
	whistleblower protection
	(4.04, 4.08, 4.20)
115475	Adds § 21.40.085,
	commercial railyard rate
	for solid waste transport
117101	and disposal (21.40)
115486	Amends § 11.72.351,
	restricted parking zones
115402	(11.72)
115492	Adds § 6.212.070(W),
	taxicabs (Repealed by
1	118341)

deference only.

deference only.

deference only.

deference only.

June 2009 co 115495 UP Text provided for

See ordinant sections for and tables a this source

115495 Adds §§ 3.02.125(C), 6.208.040(L), 6.208.130-66.208.160, 6.212.040(G), 6.212.042, 6.212.044, 6.212.055, 6.212.065, 6.212.280, 6.212.290 and 6.212.310, amends §§ 6.204.050, 6.208.060, 6.208.090, 6.208.110, 6.212.010, 6.212.060(A), 6.212.100, 6.212.120(B), 6.212.160(D), 6.212.220, 6.212.260 and 6.212.270(A) and (B), and amends § 6.212.045 by dividing it into §§ 6.212.045 and 6.212.047, all pertaining to taxicabs and related regulations (3.02, 6.204)115498 Adds Subchapter V to Ch. 3.20 (3.20.300-3.20.310), Office for Education, and establishes staffing and funding for the office (3.20)115501 Amends § 4.13.010(E), exemption of personnel from the Civil Service (4.13)115512 Rezone: Plats 21E, 22W and 29W, pages 74, 75 and 89 (Special) 115521 Establishes a Chinese Garden Memorial Account in the General Donations and Gift Trust Fund (5.76) 115529 Amends §§ 21.04.430 and 21.04.440, water rates, and amends §§ 21.76.030 and 21.76.040, low-income solid waste, water, drainage and wastewater rate assistance (21.04, 21.76) 115531 Adds Ch. 6.240 (§§ 6.240.010-6.240.130, 6.240.200-6.240.220), retail sale of tobacco products, adds § 6.204.120, tobacco product retailing, and repeals Chs. 6.60, tobacco retailers, and Ch.

ference only.

amending

raphics,

curacy of

For current (5.24)

The Office of the City Clerk

115532

6.84, tobacco vending machines (6.204, 6.240)

Amends § 4.13.010(E), exempting several positions from the Civil

Service (4.13)

Seattle Municipal Code

June 2009 co 115545 up date

Amend makes y
position
Depart
Text provide 115548 Adds §
amends

See ordinant sections for and tables a this source

	, made	te file
	115545	Amends § 4.13.010(B) and
	000	makes various changes in
	101	positions in City Light
	a for in	Department budget (4.13)
=	115548	Adds § 4.16.020(C),
		amends §§ 4.16.070(A)
		and 4.16.090 and
		renumbers § 4.16.070(B)
		and (C) to 4.16.075, Code
	- oc crea	of Ethics (4.16)
	115549	Amends §§ 5.48.050(H)
		and 5.48.220, solid waste
	eamnle	collection taxes and landfill
	COMPLE	closure costs (5.48)
	115552	Amends § 4.16.030(D),
	- 4 to CO	definition of "immediate
A	ma to	family" in Code of Ethics
U	11555	(4.16)
	115557	Amends Ord. 114630, as
	<b>1116</b> .	amended by Ord. 114899,
	880	to remain in force until
		March 31, 1992, interim
		traffic standards for the
		Northgate Area (Repealed
	115568	by 116795) Amends §§ 23.22.052,
	113306	23.24.035(D) and (E),
		23.53.005(A),
		23.53.005(A), 23.53.015(C) and (D),
		23.53.020(B)-(E),
		23.53.025(C),
		23.53.030(B) and (E)-(G)
		and 23.54.030(B), Land
		Use Code (23.22, 23.24,
		23.53, 23.54)
	115571	Rezone, Plat 20E, page 72
		(Special)
	115572	Amends § 3.124.020(C)
		and (D), city employee
		charitable contributions
		(3.124)
	115574	Approves Seattle Pacific
		University Master Plan and
		amends Official Land Use
	115502	Map (Special)
	115583	Redesignates the Downtown Housing
		Replacement Account of
		the Low Income Housing
		Fund as the Housing
		Mitigation Account of the
		Low Income Housing
		Fund, amends §§ 1 and 2
		of Ord. 115346, and
		reappropriates repayments
		of loans or grants made
		from funds received under
		Ord. 115436 (5.76)

eference only.

and amending graphics, fill accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipar	cilo
a mada	te tille
115589	Adds §§
	21.36.112-21.36.116 and
1 Co# 111	amends §§ 21.36.012,
u ioi	21.36.014 and 21.36.016,
	solid waste utility (21.36)
115590	Adds § 21.36.029(C),
	disposal of polystyrene
	packaging (21.36)
115600	Amends § 4.36.220(A),
age crea	City Employees Retirement
368	System (4.36)
115601	Adds §§ 3.95.060,
oomale	3.95.070, 20.48.080 and
Collina	20.52.060, all pertaining to
	fiscal audits (3.40, 20.48)
115606	Rezone, Plats 45W, 45E,
ma to	52W and 52E, pages 118,
,800	119, 132 and 133 (Special)
115620	Adds Ch. 10.76, Hazardous
<b>\$116.</b>	Waste Management
3330	Coordination Committee;
	amends § 21.44.060, solid
	waste
	collection/transportation
	vehicle permits (10.76,
	21.44)
115628	Rezone, Plat 234W, page
	30 (Special)
115632	Amends § 4.13.010(E),
	positions exempted from
115624	civil service (4.13)
115634	Amends §§ 11.30.120,
	11.30.160 and 11.30.180;
	repeals § 11.30.140,
	impounding of vehicles
115626	(11.30)
115636	Adds § 22.600.050,
	certificate of examination;
	amends §§ 22.602.010(B),
	22.602.020, 22.602.030,
	22.602.040 and
	22.602.080, Fire Department fees (22.600,
	22.602)
115637	Extends to 6/30/92
11303/	moratorium on issuance of
	use permits for certain
	adult entertainment uses
	(6.270 footnote)
115620	,
115638	Rezone, Plats 40E, 41E and
	41W, pages 111, 113 and
	114 (Special)

eference only.

l amending

graphics,

ccuracy of

Seattle Municipal Code

Text provide

Text p

and tables and

this source file.

ference only.

amending

raphics,

curacy of

adds §§ 1720 and 1721 to and amends §§ 605, 705, 905 and 1205 of the Seattle Building Code as adopted by Ords. 113700 and 113701; amends § 22.400.010, adds Ch. 10.1 and Tables 10.1-A-10.1-D to and amends §§ 403, 404, 405, 406, 408, 410, 415, 417, 420, 421, 422, 423, 424, 425, 426, 601, 602 and 604 of the Seattle Mechanical Code as adopted by Ords. 113702 and 113703; amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: adds §§ 108, 109, 202, Ch. 8, Ch. 9 and Tables 9-1, 9-2, 9-3a-9-3j and 9-4, Ch. 10 and Tables 10-A and 10-1-10-10, amends §§ 103, 104, 105, 106, 107, 201, 302, 303, heading of Ch. 4, §§ 401, 402, Ch. 5 heading and §§ 502, 503, 504 and 505, repeals Tables 502.1.2a, 502.1.2b, 502.1.2c, 502.2.1, 502.4.2 and Equations 1-4, and adds Tables 5-1, 5-2, 6-1, 6-2, 6-5, 6-6 and Equations 1-5, repeals §§ 503.4.2, 503.4.2a, 503.4.3, 503.4.5, 503.4.5a, 503.4.6, 503.4.6a, 503.4.7, 503.9.1 and 503.11 and adds Tables 5-6, 5-7, 5-8, 5-9, 5-11 and 5-12, repeals Tables 505.3.2.1a, 505.3.2.1b, 505.3.2.1c and 505.3.3.3 and adds Table 5-13 and Equations 7 and 8; amends Ch. 7 and repeals Figures 1-13, and repeals § 403, Ch. 6 and the Appendix to the Model Energy Code (22.700, 22.700.010 footnote) Establishes the Weatherization Account in Low-Income Housing Fund Amends §§ 12A.02.150, 12A.04.030, 12A.08.010 and 12A.08.040 and repeals § 12A.04.070. Seattle Criminal code (12A.02, 12A.04, 12A.08)

For current 510 (12A.04, 12A.04) |
the Office of the City Clerk

115647

115649

June 2009 co

Text provide

See ordinantes sections for and tables at this source

	a anda	te file
	115654	Rezone (Special)
	115657	Adds § 23.49.037, Land Use Code (23.49)
	115660	Amends § 21.04.070, water
		system cross-connection
	115663	control program (21.04) Rezone, Plat 40W, page
	113003	110 (Special)
	115664	Rezone, Plat 74W, page 75
	115671	(Special) Adds §§ 22.206.050(H),
<b>B</b> '	1130/1-	22.206.080(I), 22.206.130
	comple	(J)-(L), 22.206.215,
5	Comple	22.206.260(B), (C), (F) and (G), 22.206.270(C) and (D)
	140 00	and 22.206.280(B)-(D);
	nd to co	amends §§
C		22.200.020(D)-(F),
	C110.	22.204.040-22.204.060, 22.204.090, 22.204.100,
3	1116.	22.204.190, 22.204.230,
		22.206.020(A)-(E),
		22.206.040(A)-(F),
		22.206.050(B) and (D)-(G), 22.206.080(A),
		22.206.090(A) and (B),
		22.206.110(B)-(D),
		22.206.130(A), (B), (E)
		and (G), 22.206.140(A) and (B), 22.206.160(A) and
		(B), 22.206.200(A) and
		(F), 22.206.220(A) and
		(C), 22.206.230(B),
		22.206.250(A),
		22.206.260(A) and (D), 22.206.270(A) and (B),
		22.206.280(A) and
		22.206.290(B); repeals §
		22.206.140(C)-(I) and
		renumbers (J) to be (C); renumbers former §
		22.206.260(B) to be (E)
		and § 22.206.280(B) and
		(C) to be (E) and (F), all
		pertaining to the Housing and Building Maintenance
		Code (22.200, 22.204,
		22.206)
	115673	Amends § 4.13.010(C),
		positions exempted from civil service (4.13)
	115676	Adds §§ 17.16.101,
		17.16.103-17.16.105,
		17.16.107 and 17.16.115;
		amends §§ 17.16.110 and 17.16.120, and repeals §
		17.16.120, and repeats §
		lease and use agreements at
		Seattle Center (Repealed
		by 117776)

eference only.

I amending
graphics,
curacy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
4140	111 8 10 20 100
115678	Adds § 18.28.100,
	Municipal Golf
10-m 101	Improvement Subaccount,
A FOR III	and adds said subaccount
CI TO	to list of funds in Ch. 5.76
	(5.76, 18.28)
115000	
115686	Amends § 23.44.010(B),
	Land Use Code (23.44)
115687	Amends §§ 23.44.082(G),
0123	23.45.190(H),
ces crea	23.47.038(B),
	23.49.030(B) and
	23.50.010(B), Land Use
COMPLE	Code (Repealed by
	120293)
115693	Creates positions in
TO CO	Personnel Department and
	amends § 4.13.010(F),
	Personnel Department
2112	(4.13)
115718	Divides § 2.04.060(A) into
	(A)(1) and (A)(2) and
	amends the new
	subsection, Office of
	Election Administrator
	(2.04)
115719	Repeals § 23.54.015(J) as
	set out in Ord. 113710,
	,
11.5500	Land Use Code (23.54)
115739	Amends § 1 of Ord.
	115414, extending duration
	of interim zoning for
	portions of the Alki and
	Southwest Seattle
	neighborhoods (Not
	codified)
115749	Reclassifies and creates
	positions in the 1991
	budget of the City Light
	Department, creates new
	1 /
	titles and salaries, and
	amends § 4.13.010(B),
	Civil Service System
	exemptions (4.13)
115751	Amends §§ 23.76.010(F),
	23.76.026(A), 23.76.028
	and 23.76.032, Land Use
	· ·
	Code, and amends §§
	302(a) and 303(a) of the
	Seattle Amendments to the
	1985 Uniform Building
	_
115756	Code (23.76)
115756	Code (23.76) Amends §§ 5.48.020 and
115756	Code (23.76) Amends §§ 5.48.020 and 5.48.050(H), business tax
	Code (23.76) Amends §§ 5.48.020 and
115756 115757	Code (23.76) Amends §§ 5.48.020 and 5.48.050(H), business tax
	Code (23.76) Amends §§ 5.48.020 and 5.48.050(H), business tax on utilities (5.48) Amends §§ 11.34.020,
	Code (23.76)  Amends §§ 5.48.020 and 5.48.050(H), business tax on utilities (5.48)  Amends §§ 11.34.020, 11.56.120, 11.56.320,
	Code (23.76)  Amends §§ 5.48.020 and 5.48.050(H), business tax on utilities (5.48)  Amends §§ 11.34.020, 11.56.120, 11.56.320, 11.56.360 and 11.58.005;
	Code (23.76)  Amends §§ 5.48.020 and 5.48.050(H), business tax on utilities (5.48)  Amends §§ 11.34.020, 11.56.120, 11.56.320,

eference only.

d amending
graphics,
accuracy of

For current SMC, Code (11.34, 11.56, 11.58) ct the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

a anda	te file
115759	Adds §§ 17.40.090 and
	17.40.095; repeals Ord.
1 c - m h1	112578, Bumbershoot
ad tor in	Festival admission fees
	(Repealed by 118507)
115760	Rezone, Plat 78E, page 184
	(Special)
115765	Submits to qualified
	electors for ratification or
6 6163	rejection the amendment of
res er er	Charter Art. VII §§ 1, 2
	and 3, and repeal of Art.
	VII §§ 4, 5 and 6, and Art.
COMPA	VIII § 16, regarding
	contracting and purchasing
1 40 00	(Not codified)
115766	Submits to qualified
913162	electors for ratification or
	rejection the amendment of
C310.	Charter Art. VIII §§ 1, 2
1116.	and
	3 and Art. XIX §§ 1 and 3,
	and the repeal of Art. VIII
	§§ 4-9, regarding financial
	management (Not codified)
115771	Amends § 4.44.020,
	disability compensation
	(4.44)
	( /

eference only.

amending

graphics,

ccuracy of

115781

115796

115797

115835

115843

115851

115858

Text provided for

See ordinan sections for comp and tables and this source file.

eference only. Repeals and replaces § 22.300.010, National Electrical Code, and adds Chs. 1, 2 and 3, and §§ 110-23, 220-5, 220-23, 230-5, 230-33, 343-1, 450-20, 555-10 and 620-102, and Art. 343; amends Arts. 100 and 110-13 and §§ 110-16, 210-8, 215-2, 220-15, 220-17, 220-19, 220-20, 230-1, 230-3, 230-28, 230-29, 230-40, 230-42, 230-43, 230-52, 230-62, 230-82, 250-1, 250-80, 250-81, 250-84, 300-1, 300-21, 300-22, 310-4,

373-3, 374-1, 380-3, 380-10, 380-13, 384-14, 450-10,450-13, 555-4, 555-6, 600-5, 620-21, 620-37, 620-44, 620-71, 620-72, 620-81, 700-4, 700-12 and 700-16, and repeals §§ 90-1, 90-2, 220-30, 220-31, 333-10, 336-12, 336-16, 348-4, 450-41-450-43 and 450-45-450-48 and Art. 342 of the National Electrical Code (Repealed

by 118553)

Amends § 4.13.010, supported employment program (4.13)

Amends §§ 4.20.360 and 4.34.055, sick leave and vacation leave (4.34)

Amends § 12A.06.130, no-contact orders (12A.06)

6.288.030(B), 6.288.050-6.288.090, 6.288.100 and

(6.288)

(4.24)

Amends §§ 6.288.010(B),

6.288.180(A), used goods dealers and pawnshops

Amends § 4.24.100, sick leave for City employees

Amends §§ 17.40.030 and

17.40.060(B) and (E), **Bumbershoot Festival** Commission (Repealed by

amending 324-4, 331-4, 333-6, 334-3, 334-10, 336-3, 336-4, 336-10, 338-2, 338-3, 347-2, 348-1, 348-8, 370-1,

June 2009 co

Text provide

See ordinant sections for and tables a this source

ilhar .	4 - 6110	
a anda	te me	
115864	Amends § 11.18.010,	0
	classifies portions of South	
10-m h1	Massachusetts Street as an	
A FOR III	arterial (11.18)	
115871	Rezones certain property	
	(Special)	
115875	Amends § 23.20.004, Land	
	Use Code (23.20)	
115877	Amends §§ 22.206.160 (A)	
	and (C), 22.206.230(B) and	
ces er en	22.206.280(A), Housing	
	and Building Maintenance	4
	Code (22.206)	
115895	Amends §§ 11.20.340 and	
	11.20.350; renumbers and	
	amends §§ 11.20.340(C)	
and to C	and 11.20.340(D) to new	
	§§ 11.20.345 and	
	11.20.347, driver licenses	
C110.	(11.20)	
115906	Amends §§ 23.69.032,	
113700	23.78.002, 23.78.010,	
	23.78.012, 23.78.014 and	
	23.79.004, Land Use Code	
	(23.69, 23.78, 23.79)	
115908	Amends §§ 5.48.020(F)	
113700	and 5.48.050, business tax	
	on utilities (5.48)	
115916	Adds § 5.52.060; amends	
113910	§§ 5.52.010 and 5.52.020;	
	repeals §§ 12A.22.010 and	
	12A.22.120, gambling tax	
	(5.52)	
115917	Amends § 25.09.220 and §	
113917	5 of Ord. 115385, interim	
	regulations for critical	
	areas (Expired)	
115919	Amends § 18.28.010, fee	
113717	schedule for use of park	
	and recreational facilities	
	and services (18.28)	
115920	Amends § 10.08.170(B)(3),	
113720	fee for false alarm response	
	(Repealed by 116702)	
115921	Adds §§ 17.19.010(H) and	
113721	17.19.030(F)-(J); amends	
	§§ 17.19.010(B), (C) and	
	(F), 17.19.020 and	
	17.19.030(B), Seattle	
	Center Parking and	
	Monorail facilities (17.19)	
	mandan manifes (17.17)	
115000	1 8 20 40 050	
115922	Amends § 20.48.070,	
	contract administrative	
	charges of the Department	
	of Administrative Services	
115000	(Repealed by 117390)	
115923	Amends § 21.24.020, side	
	sewer and drainage permit	
	fees (Repealed by 116421)	
mont SM	C, CUITO	

eference only.

de lamending amphics, laccuracy of laccur

For current SMC, Correlling Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Sihar	4- 8112
a - moda	TG PIL
115925	Amends § 3.02.125,
	hearing examiner filing
101	fees (3.02)
115926	Amends § 21.40.080(A),
	transfer station and
	disposal site rates (21.40)
115927	Amends § 11.31.120,
	traffic infraction penalties
	(11.31)
115929	Amends § 3.20.210, garden
ces crea	plot fee schedule (3.20)
115930	Amends § 17.16.015,
100016	Seattle Center facilities
COMPA	fees (17.16)
115932	Adds § 5.64.100, real
113,32	estate excise tax (5.64)
115933	Adds § 22.900.254;
113733	amends §§ 22.900.234;
	22.900.020, 22.900.030,
0210	*
1116.	22.900.040, 22.900.050,
	22.900.070, 22.900.110,
	22.900.120, 22.900.140,
	22.900.150, 22.900.170,
	22.900.200, 22.900.205,
	22.900.210, 22.900.220,
	22.900.230, 22.900.240
	and 22.900.280; repeals §
	22.900.160, 1992 permit
	fees (Repealed by 117405)
115934	Amends §§ 17.16.030,
	17.16.101(B), 17.16.107
	and 17.16.115; repeals §
	17.18.010, Seattle Center
	Department (17.16)
115936	Adds § 3.14.425; repeals
	§§ 3.14.110-3.14.1130,
	international affairs (3.14)
115938	Amends §§ 9.26.050 and
	9.26.060, animal fees
	(9.26)
115940	Amends Subchapter VII of
	Ch. 3.14 (§§
	3.14.900-3.14.970), Seattle
	Women's Commission and
	Commission for Lesbians
	and Gays (3.14)
115941	Amends § 4.13.010, Civil
	Service System (4.13)
115942	Adds Ch. 15.35; amends §
	15.04.070(C), filming
	permits (15.04, 15.35)
115944	Adds § 10.03.180,
/	construction and
	environmental review fees
	environmental review fees
1150/15	(10.03)
115945	(10.03) Amends §§ 10.03.010,
115945	(10.03) Amends §§ 10.03.010, 10.03.110 and 10.11.942,
	(10.03) Amends §§ 10.03.010, 10.03.110 and 10.11.942, health code fees (10.03)
115945 115946	(10.03) Amends §§ 10.03.010, 10.03.110 and 10.11.942, health code fees (10.03) Amends § 22.504.010,
	(10.03) Amends §§ 10.03.010, 10.03.110 and 10.11.942, health code fees (10.03)

eference only.

d amending
graphics,
accuracy of

For current Shie, List Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	40	to file
	115948	A
	113948	Amends §§ 6.204.120,
		6.240.010(D), 6.240.020, 6.240.040, 6.240.050,
	a for hi	6.240.120 and 6.240.200:
2	C 101	repeals §§ 6.240.030,
		6.240.060 and 6.240.130,
		tobacco product retailing
		(6.204, 6.240)
	115949	Adds Ch. 3.37; repeals §§
		3.14.2000-3.14.2050 and
V	ces crea	3.14.3000-3.14.3070,
3		Department of Planning
	10	(Repealed by 117408)
	115951	Amends §§ 21.49.030,
		21.49.040, 21.49.052,
	4 4 - 06	21.49.055, 21.49.057 and
	and to co	21.49.058, City Light
	The co	Department (21.49)
	115953	Amends § 5.16.050,
	C110.	warrant overdrafts (5.16)
	115954	Amends §§ 5.48.050(C),
		(F), (G) and (H) and
		5.48.220, business tax on
		utilities (5.48)
	115955	Amends §§ 21.28.080,
		21.28.370, 21.76.040 and
		21.76.050, drainage and
		wastewater, solid waste
		and water rates (21.28,
		21.76)
	115956	Adds §§ 22.602.040(C)
		and 22.602.050(D) and (E);
		amends §§ 22.602.010(B),
		22.602.040(A) and (B) and
		22.602.080; redesignates §
		22.602.040(C)-(G) to
		(D)-(H), Building and
		Construction Code fees
ŀ		(22.602)
	115957	Adds §§ 5.40.053 and
		5.40.056; amends §§
		5.40.020 and 5.40.025,
		complementary and
		discount admission taxes
Į		(5.40)

eference only.

l amending

graphics,

ccuracy of

Seattle Municipal Code eference only. 115958 Amends §§ 3.06.070, 3.118.010(B), 3.20.010, Text provided for hi 3.20.030, 3.20.040, 3.20.080, 3.20.120, 5.78.060, 16.60.030(C), 20.84.030, 21.52.020, 21.52.230, 21.52.250, 21.52.260(E), 21.76.010, 21.76.030(C), 21.76.040, 21.76.050, 21.76.070, See ordinan 22.220.080(E), 22.220.090, 22.220.100(C), 22.220.130, sections for comp 25.12.320, 25.16.040, 25.16.060, 25.24.030, 25.24.050 and 25.24.080; repeals §§ 3.10.010, and tables and 3.10.020, 3.10.050, 3.10.100, 3.10.110, 3.10.120, 3.10.130 and this source file. 3.14.730, reorganizes various departments, projects and services (3.06, 3.20, 3.118, 5.76, 16.60, 20.84, 21.52, 21.76, 22.220, 25.12, 25.16, 25.24) 115960 Amends § 5.68.010, natural or manufactured gas use tax (5.68) 115962 Adds Subchapter II to Ch. 4.76 (§ 4.76.100); renames Ch. 4.76, moving and security expenses (4.76) 115975 Adds Subchapter VII to Ch. 3.28 (§§ 3.28.600-3.28.660),

internal investigations auditor (Repealed

Adds Ch. 15.52, crowd control events (15.52)

Amends §§ 21.43.010(G) and 21.43.040(K), infectious waste (21.43)

Abrogates, reclassifies and creates positions in City Light Department; amends § 4.13.010, Civil Service

(4.13)

## For current SMC, contact the Office of the City Clerk

by 120728) 115982

115983

115987

Seattle Municipal Code date ference only. June 2009 co 115994 Adds Ch. 15.90, Street Use Appeals Board; amends §§ Text provided for hi 15.04.010, 15.04.020, 15.04.030, 15.04.040, 15.04.050(B) and (C), 15.04.070, 15.04.080, amending 15.06.020, 15.06.030, 15.08.010, 15.08.080, 15.10.020, 15.10.030, See ordinances crea 15.12.010, 15.12.030, 15.12.040(F), 15.14.010(B), 15.14.040, sections for comple 15.14.050, 15.14.070, 15.14.080, 15.16.060, 15.22.020, 15.24.020, and tables and to 15.28.030, 15.28.060, 15.30.010, 15.30.020, 15.30.030, 15.30.050, 15.32.010, this source file. 15.32.020(A)-(D), (F)-(H) and (J), 15.34.010, 15.42.030, 15.42.050, 15.42.060, 15.44.130(A), 15.44.140, 15.44.150, 15.44.170(B), 15.50.010, 15.50.020, 15.50.040, 15.64.030, 15.64.040, 15.64.060, 15.64.065(A), 15.64.070, 15.64.080, 15.72.010, 15.72.020, 15.72.040, 15.74.030, 15.78.010, 16.60.010, 16.60.020, 16.60.030(A),

> (B) and (D)-(G), 16.60.040, 16.60.050(C),

22.802.020 and 22.802.100(G), street, sidewalk and waterway use (15.05, 15.06, 15.08, 15.10, 15.12, 15.14, 15.16, 15.22, 15.24, 15.28, 15.32, 15.42, 15.44, 15.50, 15.64, 15.72,

16.60)

Seattle Municipal Code ference only. Aate. 115995 Adds §§ 11.16.125, 11.24.030 and 11.73.200; Text provided for hi amends §§ 11.16.120, 11.16.220, 11.16.300(E), 11.16.315, 11.16.317, 11.16.340(M), 11.23.020, 11.23.120, 11.23.390, 11.23.420, 11.24.020, 11.54.120, 11.66.180, 11.66.200, 11.73.050(B) See ordinan and 11.74.060; repeals §§ 11.24.040, 11.24.060, 11.24.080, 11.24.120, sections for 11.73.010(B), 11.73.<mark>020(B),</mark> 11.73.030(B), and tables and 11.73.040(B), 11.73.070, 11.73.080(B) and 11.93.010-11.93.030, this source file. vehicles and traffic (11.16, 11.23, 11.24, 11.54, 11.66, 11.73, 11.74) Amends §§ 3.114.020(A) and (B), 3.114.040, 3.114.050, 3.114.060(A) and (B), 3.114.070, 3.114.080, 3.114.090, 3.114.100, 3.114.110, 3.114.120, 3.114.130, 3.114.140, 3.114.150, 3.114.160(B), (D)-(I) and (K) and 3.114.170. consultant selection (3.114) 116004 Amends §§ 3.14.810(A) and 3.14.814, purchasing agent (3.14) 116005 Adds Ch. 3.70; amends §§ 2.04.610, 2.04.010(2) and (18), 2.14.010(A), 2.24.030(A), 2.24.040, 4.13.010(C) and (G), 4.16.030, 4.20.800,

> For current SMC, contact the Office of the City Clerk

116006

4.20.810(C)(1)(a), 4.20.830(E) and 4.20.850(A); repeals §§ 2.04.050, 2.04.060(A) and

(B)(1), 2.04.080, 2.04.150(C), 4.16.040, 4.16.050 and 4.16.060, Ethics and Elections Commission (2.04, 2.14, 3.70, 4.13, 4.16, 4.20)

Amends § 3.76.010, Debt Management Policy Advisory Committee (3.76)

Text provided for

See ordinan sections for and tables and this source file.

116007 Adds Subchapter II-A to Ch. 3.18 (§§ 3.18.101, 3.18.111, 3.18.121), Contracting Appeals Board, and adds § 3.18.030(H) - (K); amends §§ 3.18.010, 3.14.804-3.14.834, 3.14.840, 20.48.020, 20.48.030, 20.48.040, 20.48.060 and 20.48.070; renumbers § 3.18.030(H) to (L) and Subchapter VI of Ch. 3.14 (§§ 3.14.804-3.14.840) to Subchapter VIII of Ch. 3.18 (§§ 3.18.800-3.18.832); repeals Ch. 3.44 and §§ 3.14.800, 3.14.802, 3.14.836, 3.14.838, 3.18.070, 3.18.080, 3.18.090, 3.18.100, 3.18.110 and 3.18.120, Department of Administrative Services (3.18, 20.48)116011 Repeals and replaces § 22.400.010 and makes extensive amendment to Uniform Mechanical Code, 1991 Edition (Repealed by 117722)116012 Repeals and replaces §§ 22,100,010 and makes extensive amendments to Uniform Building Code, 1991 Edition (Repealed by 117721) 116015 Rezone, Plats 47W, 47E, 48W, 53W and 53E, pages 122, 123, 124, 134 and 135 (Special) 116023 Amends § 9.25.100, animal control (9.25) 116037 Amends §§ 4.04.030 and 4.04.230(F), Personnel System (4.04) 116039

Adds Subchapter III to Ch.

10.28.200-10.28.230), health care records (10.28)

23.60.090(F)-(I) and 23.60.916(H), Land Use

Designates the James W. Washington, Jr., Home and Studio as a Landmark

Designates the Harvard

10.28 (§§

Amends §§

Code (23.60)

(25.32)

ference only.

Mansion as a Landmark (25.32)the Office of the Cit

116051

116052

116053

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a mada	te ille
	116054	Designates the Fremont
	130	Trolley Barn/Red Hook
		Ale Brewery as a
l		Landmark (25.32)
j	116055	Designates the Pacific
	110000	Medical Center/U.S.
		Marine Hospital as a
		Landmark (25.32)
F	116056	Designates the YMCA
	110050	Central Branch (South
ſ	DE CLE	`
ŀ		Building) as a Landmark
ŀ	11/057	(25.32)
	116057	Adds Ch. 25.10,
	Compa	radiofrequency radiation,
		and amends § 3.02.125,
	1 40 CC	Hearing Examiner filing
L	an to	fees (3.02, 25.10)
B	116061	Amends § 17.16.015,
		Seattle Center facility use
		fees, terms and conditions
L	7370	(17.16)
	116066	Adopts Industrial zoning
		maps showing the
		boundaries of the General
		Industrial 1 and 2 (IG1 and
		IG2), Industrial
		Commercial (IC) and
		Industrial Buffer (IB)
		zoning districts as
		amendments to Offi-
r		cial Land Use Map and
		corrects Plat 22W, page 75
		of the Official Land Use
		Map (Special)
H	116069	Adds §§ 4.36.270(D) and
	11000)	4.36.295, and amends §§
		4.36.230(E) and 4.36.260,
		City Employees'
		Retirement System (4.36)
H	116070	Adds § 4.36.145, City
	110070	l
		Employees' Retirement System (4.36)
H	116083	Amends Ord. 114630, as
	110083	
		amended by Ords. 114899
		and 115557, to remain in
		force until July 31, 1992,
		interim traffic standards for
		the Northgate Area
		(Repealed by 116795)
	11,000	
	116085	Closes Sewer Improvement
	116085	Fund, transfers assets and
	116085	Fund, transfers assets and liabilities to Drainage and
	116085	Fund, transfers assets and liabilities to Drainage and Wastewater Fund and
	116085	Fund, transfers assets and liabilities to Drainage and Wastewater Fund and abandons remaining
	116085	Fund, transfers assets and liabilities to Drainage and Wastewater Fund and
	116085	Fund, transfers assets and liabilities to Drainage and Wastewater Fund and abandons remaining
	116085	Fund, transfers assets and liabilities to Drainage and Wastewater Fund and abandons remaining unencumbered appropriation balances (Special)
	116085	Fund, transfers assets and liabilities to Drainage and Wastewater Fund and abandons remaining unencumbered appropriation balances

reference only.

d amending
, graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	to file
11,007	A11 1 (C) ( 8
116097	Adds subsection (C) to §
	3.02.125, Administrative
40-44 h1	Code (3.02)
116099	Amends §§ 5.44.010 and
	5.44.090, occupation taxes
	for gambling activities of
	for-profit organizations
	(5.44)
116131	Adds § 3.18.806(H);
110131	
age crea	amends §§ 3.18.806(E),
	3.18.810 and 3.18.826,
	Department of
- commile	Administrative Services
COMPA	(3.18)
116133	Authorizes revision to
1 1 - 00	SCTP-Street Classification
and to co	Maps (Special)
116124	
116134	Amends § 11.18.010,
	arterial streets (11.18)
116142	Amends § 25.05.675(G)
1110.	and (J),
	Telecommunications
	Facility Policy (25.05)
116143	Waives certain zoning
110145	
	restrictions for contract
	rezone of Pacific Medical
	Center (Special)
116145	Adds Ch. 23.67; amends §§
	23.47.004(A) and
	23.76.042(C), Southeast
	Seattle Reinvestment Area
	(23.47, 23.67, 23.76)
116146	
116146	Amends § 23.44.022 and
	Chart A of § 23.54.015,
	Land Use Code (23.44,
	23.54)
116152	Rezone, Plats 55W, 55E,
	62W, 62E, 69W, 69E,
	70W, 75E, 76W, 76E and
	77W, pages 138, 139, 151,
	152, 165, 166, 167, 178,
	179, 180 and 181 (Special)
116153	179, 180 and 181 (Special) Rezone, Plat 34E, page 99
116153	179, 180 and 181 (Special)
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special)
116153 116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows:
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote)
	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to Ch. 2.04, designation for
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to Ch. 2.04, designation for campaign matching fund
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to Ch. 2.04, designation for campaign matching fund account (Repealer)
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to Ch. 2.04, designation for campaign matching fund account (Repealer) Amends Chart A of §
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to Ch. 2.04, designation for campaign matching fund account (Repealer) Amends Chart A of § 23.54.015, Land Use Code,
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to Ch. 2.04, designation for campaign matching fund account (Repealer) Amends Chart A of §
116159	179, 180 and 181 (Special) Rezone, Plat 34E, page 99 (Special) Amends § 22.700.010, and portions of the 1989 Model Energy Code as follows: amends §§ 502-504 and Tables 5-6 through 5-9 (22.700, 22.700.010 footnote) Repeals Subchapter VII (§§ 2.04.600-2.04.630) to Ch. 2.04, designation for campaign matching fund account (Repealer) Amends Chart A of § 23.54.015, Land Use Code,

eserence only.

amending

graphics,

accuracy of

For current Sine, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibar	0210
a mada	te file
116172	Amends § 5 of Ord.
	115385, extending
s con hi	effective date of interim
d in in	regulations for critical
116185	areas (Expired)
116185	Amends §§ 21.04.430, 21.04.440 and 21.76.040,
	water supply emergency
	restrictions (21.04, 21.76)
116186	Amends §§ 5.48.050(G)
ces crea	and 5.48.220, solid waste
	collection tax and landfill
	closure costs (5.48)
116187	Amends §§ 21.36.050,
	21.40.050, 21.40.060,
ad to C	21.40.080 and
liller co	21.76.040(B), solid waste utility collection rates and
	charges (21.36, 21.40,
C:10	21.76)
116188	Amends § 21.40.085, solid
	waste utility rates (21.40)
116196	Extends to 6/30/93
	moratorium on issuance of
	use permits for certain
	adult entertainment uses
11(100	(6.270 footnote)
116199	Adds § 3.18.030(L);
	amends § 17.04.030; reletters § 3.18.030(L) to
	(M), natural gas delivery to
	Seattle Center (3.18, 17.04)
116205	Amends §§ 23.44.010 and
	23.84.036(S), Land Use
	Code (23.44, 23.84)
116220	Amends § 21.36.030,
	authorized garbage haulers
11.000	(21.36)
116223	Rezones portions of
116225	Southeast Seattle (Special) Fireworks restrictions for
110223	1992; amends §
	78A.106(b) of Seattle Fire
	Code adopted by Ord.
	115406, fireworks
	(22.600.020 footnote)
116226	Amends § 3.14.920, Seattle
	Women's Commission
11 (227	membership (3.14)
116227	Extends to 1/31/93 the
	expiration date for Northgate area traffic
	impact mitigation interim
	standards (Repealed by
	116795)
116231	Rezones portions of certain
	West Seattle
	neighborhoods (Special)
116232	Amends § 23.50.014(B),
	Land Use Code (23.50)

d amending graphics, accuracy of

For current SMC,
tand Use Code (23.50)
the Office of the City Clerk

June 2009 co Text provide

See ordinant sections for and tables a this source

C	ilber 1	to file
	111104	
	116236	Amends § 4.13.010(A)-(C)
	103	and (G), exemptions from civil service (4.13)
	116237	Repeals §§ 3.18.804 and
	110237	3.18.828, annual
		purchasing plans
		(Repealer)
	116238	Amends § 4.36.390(B)(10),
		City Employees'
	age crea	Retirement System
7	116239	(Repealed by 118498) Amends § 3.33.190, Seattle
	110239	Municipal Court, and §
	COMPLE	4.13.010(E) and (G),
<b>b</b>		exemptions from civil
	140 06	service (3.33, 4.13)
	116242	Continues validity of
Ů,		prosecutions following
	0210	expiration or repeal of Ord. 115171, controlled
	1116.	substances (12A.20.050
		footnote)
	116243	Amends § 25.05.675(P)
		and (Q), specific
		environmental policies
		(25.05)
	116247	Moratorium on granting
		certain City Light pole attachment agreements
		until 4/1/93 or date City
		adopts comprehensive pole
		attachment policy (15.32
		footnote)
	116250	Adds § 21.36.087(D) and
		amends §§ 21.36.087(B) and 21.40.080 (A) and (D),
		solid waste collection and
		collection rates (21.36,
		21.40)
	116251	Corrects clerical error in
		rezone Ord. 116231, Map
	116253	68W, page 163 (Special) Adds Ch. 25.09,
	110200	regulations for
		environmentally critical
		areas (25.09)
	116254	Adds § 25.05.747 and
		amends §§
		25.05.675(B)-(D), (J) and (S), 25.05.748, 25.05.800
		and 25.05.908,
		environmental policies and
		procedures (25.05)
	116255	Amends § 25.06.110,
		floodplain development
	116256	(25.06) Moratorium on issuance of
	110430	permits for certain
		advertising signs (Repealed
		by 116581)
		C' CATTER

eference only.

in the lamending araphics, accuracy of

See ordinantes sections for and tables at this source

Cibon	4 - 6110
a and a	TG 1110
116258	Amends § 22.602.040(A)
	and Table A and (C) and
1 Com h1	Table C, Fire Code permit
ad tor	and inspection fees
	(22.602)
116259	Adds § 22.900.290,
	surcharge on building, land
	use and related permit fees
11/2/2	(Repealed by 116949)
116262	Adds § 23.22.058; amends
	§§ 23.20.008, 23.22.058,
10	23.24.040, 23.28.030,
COMPLE	23.42.042, 23.44.008(G), 23.44.010(A) and (B),
L COMPAGE	23.44.014(D),
	23.44.024(A)-(C),
and to co	23.44.034, 23.45.014(G),
	23.45.056(D),
	23.45.072(D),
C:10	23.53.010(B),
1116.	23.53.015(A) and (D),
	23.53.020(A),
	23.53.030(A) and (G),
	23.70.010(A) and (B),
	23.76.032(B), 23.84.010
	and 23.84.036; repeals Ch.
	23.62, Land Use Code
	(23.20, 23.22, 23.24, 23.28,
	23.42, 23.44, 23.45, 23.53,
	23.84)
116265	City Light Department
	personnel changes; amends
	§ 4.13.010, exemptions
	from Civil Service Systems
11,070	(4.13)
116270	Adds Subchapter IX (§§
	3.18.900-3.18.926 on
	recycled content product
	procurement program) and amends §§ 3.18.806,
	3.18.810 and 3.18.826,
	Department of
	Administrative Services
	(3.18)
116291	Increases City Light rates
110271	in Ch. 21.49 by five
	percent for Schedule 26
	and ten percent for
	Schedules 3, 21, 31, 34, 35,
	38, 39, 42, 43, 48 and 81
	(21.49)
L	

eference only.

defined amending seraphics, accuracy of

Seattle Municipal Code eference only. 116295 Adds Ch. 23.57, communications Text provided for hi regulation; amends §§ 23.44.002, 23.44.012(C), 23.45.002, 23.45.009(C), 23.45.050(D), 23.45.066(C), 23.47.002(D), 23.47.004 (Chart A), 23.47.006(B), 23.47.012(G) and (I), See ordinan 23.47.014(F), 23.49.002(E), 23.49.008 sections for comp (intro) and (C), 23.49.046(E), 23.49.066(E), 23.49.096(F), and tables and 23.49.122(E), 23.49.148(E), 23.49.338(A), this source file. 23.50.002(D), 23.50.012 (Chart A), 23.50.020(A), 23.84.006, 23.84.040 and 23.84.048; repeals §§ 23.44.054, 23.45.104 and 23.45.156, Land Use Code (23.44, 23.45, 23.47, 23.49, 23.50, 23.57, 23.84) 116307 Drug traffic loitering; sets out provisions for § 12A.20.050 to succeed those of Ord. 115171 § 1 expiring 8/5/92 (12A.20) 116308 Amends §§ 3.18.010 and 3.18.030(L) and (M), Department of Administrative Services; amends §§ 21.60.040, 21.60.060, 21.60.230(B), 21.60.330, 21.60.350, 21.60.360 and 21.60.380, Office of Cable Communications (21.60) 116310 City Light Department personnel changes; amends

## For current SMC, contact the Office of the City Clerk

116315

§ 4.13.010(B), exemptions from Civil Service systems

Amends §§ 22.206.270(A) and (B) and 22.206.280, rental housing registration and inspection program

(4.13)

(22.206)

June 2009 co

Text provide

See ordinant sections for and tables a this source

Sibon	4- 5112
a mada	Le me
116325	Adds § 23.60.365; amends
	§§ 23.60.014(C),
a com hi	23.60.152(H) and (R),
d tor	23.60.365, 23.60.394(B),
	23.60.396, 23.60.956
	(intro) and (B) and
	23.60.958 (intro) and (A);
	amends § 23.60.360 by
	repealing (A) and
	relettering (B) and (C) to
ces cross	be (A) and (B), Land Use
	Code (23.60)
116328	Amends §
COLLIBATO	23.60.090(F)-(I), Land Use
	Code (23.60)
116334	Repeals and adds Ch.
and to C	22.600, Seattle Fire Code,
The co	and repeals Ord. 115405
	(Repealed by 118019)
116335	Seattle Fire Code
IIIC.	Supplement; adopts Seattle
	amendments to the
	Uniform Fire Code, 1991
	Edition (Repealed by
	118019)
116346	Amends § 21.40.060, solid
110310	waste container pickup
	rates for detachable
	containers (21.40)
116349	Adds Ch. 10.09, public
110319	nuisances (10.09)
116350	Amends § 22.602.040(C)
110330	and Table C, Fire Code
	permit and inspection fees
	(22.602)
116364	Amends § 22.206.270,
110304	violation of Housing and
	Building Maintenance
	Code (22.206)
	Code (22.200)

d amending graphics, ecuracy of

Seattle Municipal Code 116368 Text provided for hi

See ordinances sections for comp and tables and this source file.

Adds § 3.14.590, Ch. 3.38 (Finance Department), Ch. 3.40 (office of City Auditor), Ch. 3.42 (City Clerk) and § 20.76.098; amends §§ 2.04.060(B)(14), 2.04.160, 2.04.170, 2.04.220, 2.04.230(B), 2.04.250-2.04.270, 2.04.330, 2.04.350-2.04.370, 2.04.410, 2.04.430, 2.04.440,

2.08.010-2.08.030, 2.14.010, 2.14.080-2.14.100, 3.02.050-3.02.070,

3.02.090, 3.12.220, 3.14.590, 3.16.080, 3.18.040, 3.18.180, 3.26.040, 3.28.180,

3.30.110, 3.30.120, 3.33.060, 3.33.090, 3.33.160, 3.33.170,

3.68.070, 3.76.010, 3.76.020, 3.90.010, 3.106.010, 3.110.010,

3.110.020, 3.110.040, 3.110.090, 3.110.110,

3.110.200(H), 3.110.220(B),

3.110.390(B), 3.110.400, 3.110.410, 3.110.430,

3.110.450(A)(5),

3.110.470, 3.110.520, 3.110.550, 3.110.570(D),

3.114.100, 3.114.130,

3.116.080, 3.121.020,

3.121.040(B), 3.124.020(B),

3.124.030-3.124.050,

4.04.250(D) and (J), 4.08.040(D), 4.08.150,

4.20.110(A), 4.20.160,

4.20.325(A), 4.20.800,

4.20.810(C), 4.20.830,

4.20.850, 4.24.080,

4.24.090, 4.28.030,

4.36.110(B), 4.36.126(D), 4.36.130, 4.36.135,

4.36.140(A), 4.36.195(G),

4.38.306, 4.40.020(A),

4.40.040, 4.44.110,

4.44.060, 4.44.070,

4.68.140, 4.72.010(A) and (B), 4.72.020(A), 4.72.050,

4.72.070(A), 4.72.080,

4.72.090, 4.72.110,

4.72.120, 4.76.010,

4.76.100, 4.92.020(A),

5.04.030, 5.04.040,

5.06.010, 5.06.030, 5.06.040(C), 5.06.050,

5.08.040.

5.10.050-5.10.080,

5.12.010-5.12.040, 5.14.010-5.14.060,

5.16.010,

ference only. amending lerk

For current the Office of

Text provided for hi

See ordinan sections for and tables and this source file.

21.76.060, 22.100.010, 22.208.110(D), 22.220.100 (C), 22.300.010, 22.400.010, 22.600.020, 22.700.010, 22.802.100(D) and (E), 22.806.030, 22.906.030, 23.22.066, 23.45.094(A)(b)(9), 23.45.096(E)(1)(c), 24.64.160(A)(4), 24.66.070(intro), 24.66.100(C), 25.05.914(D) and 25.28.350(B); amends § 11

of Ord. 112904 and §§ 1 and 2 of Ord. 116208; renames Ch. 5.06 to be City Investments; renumbers and amends §§

3.95.020, 3.95.060, 3.95.070 and 3.120.010 to be §§ 3.38.070, 3.40.040, 3.40.050 and 3.38.080 respectively; and repeals §§ 3.12.110, 3.95.010, 3.95.030-3.95.050, 3.120.020, 3.120.030, 4.20.025, 4.20.370, 5.06.020, 5.12.020,

5.124.010(B), 5.44.022(6) 5.44.028(5), 20.76.100, 20.80.070, 20.80.080 and 21.72.010, all pertaining to establishment of the Finance Department and corresponding changes to

city officers' duties (2.04, 2.08, 2.14, 3.02, 3.12, 3.14, 3.16, 3.18, 3.26, 3.28, 3.30, 3.33, 3.38, 3.40, 3.42, 3.68,

3.76, 3.90, 3.106, 3.110, 3.114, 3.116, 3.120, 3.121, 3.124, 4.04, 4.08, 4.20,

4.24, 4.28, 4.36, 4.40, 4.44, 4.68, 4.72, 4.76, 5.04, 5.06, 5.08, 5.10, 5.12, 5.14, 5.16,

5.20, 5.24, 5.32, 5.40, 5.44,

5.48, 5.52, 5.56, 5.64, 5.78, 6.02, 6.14, 6.36, 6.38,

6.202, 6.222, 6.230, 6.260, 7.04, 7.20, 10.02, 10.08, 10.16, 10.18, 10.46, 10.54,

11.16, 11.23, 11.30, 11.66, 15.04, 15.16, 15.28, 16.44, 17.28, 20.04, 20.08, 20.12,

20.32, 20.36, 20.46, 20.48, 20.76, 20.80, 21.04, 21.16, 21.24, 21.28, 21.40, 21.60,

21.76, 22.208, 22.222, 22.700, 22.806, 23.22,

23.45, 25.05, 25.28)

ference only.

For current the Office of the City Clerk Seattle Municipal Code ference only. June 2009 co 116377 Amends §§ 3.70.100(B) and (E), 3.70.160(C) and Text provided for 3.70.200, Ethics and Elections Commission, and amends §§ 4.16.030(E), 4.16.070(4), 4.16.075 and amending 4.16.090(B)-(F), Code of Ethics (3.70, 4.16) 116393 Amends §§ 21.28.070, 21.28.080, 21.28.370 and See ordinan 21.33.030, sewer and wastewater charges (21.28, 21.33) sections for 116398 Adds § 23.60.633 and amends §§ 20.60.638(B) and 23.60.958(B), Land and tables a Use Code (23.60) Volunteer Park Conservatory Account in the General Donations and this source Gift Trust Fund (5.76) 116401 Amends § 4.13.010(B), positions exempt from Civil Service System (4.13)

Seattle Municipal Code ference only. 116419 Amends Ch. 21.36 by adding Subchapter I Text provided for (General Provisions) before § 21.36.010, adding § 21.36.017, amending §§ 21.36.010, 21.36.012, 21.36.014 and 21.36.016 and renumbering and amending 21.36.110 to be 21.36.018; adds Subchapter See ordinan II (Solid Waste Collection) before § 21.36.025 and § 21.36.042, and amends §§ sections for 21.36.029, 21.36.030, 21.36.040, 21.36.050 and 21.36.095; adds Subchapter and tables and III (Flow Control Special Provisions) before § 21.36.112 and amends §§ this source file. 21.36.112, 21.36.113 and 21.36.115; adds Subchapter IV (Miscellaneous Provisions) before § 21.36.180, Subchapter V (Litter Control Code) and §§ 21.36.400 and 21.36.410 of Subchapter V; renumbers and amends §§ 21.36.020, 21.36.021, 21.36.023 and 21.36.100 to be §§ 21.36.420, 21.36.430, 21.36.425 and 21.36.440 respectively; adds Subchapter VI (Penalties and Enforcement) before § 21.36.920, adds § 21.36.965 and amends and renumbers §§ 21.36.120, 21.36.130, 21.36.140, 21.36.150, 21.36.160 and 21.36.170 to be §§ 21.36.920, 21.36.930, 21.36.940, 21.36.950, 21.36.960 and 21.36.970 respectively, solid waste collection; amends § 21.40.010 and repeals §

> For current SMC, contact the Office of the City Clerk

116420

116421

21.40.070, solid waste collection rates and charges

22.208.010-22.208.040 and 22.208.080-22.208.120, buildings unfit for human habitation (22.208)

Adds § 21.24.021, side sewer permits for sewage and drainage (21.24)

(21.36, 21.40)

Amends §§

June 2009 co

Text provided for

See ordinant sections for and tables a this source

116423 Amends §§ 21.04.430 and 21.04.440, water rates and regulations, and amends §§ 21.76.040 and 21.76.050, low-income utility credits for qualified customers (21.04, 21.76)116425 Adds Chs. 22.801 (Definitions) and 22.806 (Administration and Enforcement) and repeals and replaces Chs. 22.800 (Title, Purpose and Authority), 22.802 (Stormwater, Drainage, and Erosion Control) and 22.804 (Grading), all relating to the public drainage control system (22.800, 22.801, 22.802, 22.804, 22.808) 116426 Amends § 21.40.080, solid waste transfer station rates (21.40)116429 Amends § 5.48.050, business tax on utilities (5.48)116431 Amends § 6 of Ord. 116205, extends expiration date to 6/3/93 for interim controls on development of certain submerged lots (23.44.010 footnote) 116434 Amends § 10.13.120(F), meat cutter or wrapper salesperson permits (10.13) 116435 Amends § 10.03.056, swimming and spa pool fees (10.03) 116436 Amends § 10.03.110, food-service establishment permit fees (10.03) Amends §§ 10.11.941 and 116437 10.11.944, food-service establishment plan review fees (Repealed by 117001) 116438 Amends §§ 21.44.050 and 21.44.060, solid waste handling facility fees (21.44)116439 Authorizes the Director of Engineering to collect certain street use permit fees (Not codified)

ference only.

amending

raphics,

curacy of

116442

116443

116447

116448

116449

116450

116451

116454

file.

June 2009 co 1644 up Text provided for

See ordinant sections for and tables a this source

Establishes Engineering Services Fund, Transportation Fund; reestablishes the City Street Fund, Arterial City Street Fund and the Transportation Fund as subfunds in the City Street Fund; amends §§ 1 and 2 of Ord. 67365, § 1 of Ord. 98497 and § 16.08.050, Harbor Code; abolishes the **Engineering Department** Operating Fund and repeals Ord. 89913, Table of funds (5.76, 16.08)Adds §§ 22.900.035,

22.900.245 and 22.900.265; amends §§ 22.900.010, 22.900.020, 22.900.025, 22.900.030, 22.900.040, 22.900.045, 22.900.050-22.900.150, 22.900.170, 22.900.180, 22.900.185, 22.900.190, 22.900.210, 22.900.220-22.900.260; renumbers and amends §§ 22.900.200 and 22.900.205 to be subsections (A) and (B) respectively of § 22.900.280, and repeals §§ 22.900.254 and 22.900.255, 1993 Permit Fee Ordinance (Repealed by 117405) Amends § 18.28.010, Parks and Recreation Department fee schedule (18.28) Adds § 17.19.060 and amends §§ 17.19.010(C) and (H) and 17.19.020, Seattle Center and Monorail fees and parking charges (17.19) Amends § 17.16.015, Seattle Center facility use fees (17.16) Amends § 17.40.090, **Bumbershoot Festival** admission fees (Repealed by 118507) Amends § 7.04.570, Weighmaster license fee (7.04)Adds Subtitle VIII (Street Utility) and §§ 21.100.010-21.100.060, Street Utility (21.100) Amends § 21.36.112, Solid Waste Utility (21.36)

ference only.

amending

graphics,

ccuracy of

For current Smc, the City Clerk the Office of the City Clerk

June 2009 co Text provide

See ordinand sections for and tables a this source

	Cipar		
	a mada	te tille	
	116455	Adds Ch. 21.101 (Street	
		Utility Rates) and amends	
	a com hi	§ 21.33.070, storm	
	d for 112	drainage utility, and §§	
_		21.76.010 and	
		21.76.030-21.76.060,	
		low-income utility credits	
	116456	(21.33, 21.76)	
	116456	Adds § 17.30.010, Bite of	
	CE CIE	Seattle Festival admission fee (Repealed by 116756)	
3	116457	Adds Subchapter IV and §§	
	110437	3.14.600-3.14.620, office	
	comple	of Economic Development	
	6033-8	(3.14)	
	116458	Exempts certain City	
	TO C	employees from Chs. 4.04	
	11100	and 4.08 and certain rules	
		and regulations	
	116459	Amends § 5.44.130,	
	1110.	business license fees (5.44)	
	116460	Amends §§ 5.48.050 and	
		5.48.060, business tax on	
		utilities (5.48)	
	116461	Amends § 5.44.030,	
ļ		business license fees (5.44)	
	116462	Amends § 5.48.070,	
		business tax on utilities	
ŀ	116463	(5.48)	
	110403	Amends §§ 6.204.020-6.204.050,	
		6.204.080 and 6.204.090,	
		business license fees	
		(6.204)	
Ì	116464	Amends §§ 6.08.010,	
		6.36.020(B), 6.42.030,	
		6.48.020, 6.82.140,	
		6.102.030 and 6.204.120,	
		fees for various businesses	
		(6.08, 6.36, 6.42, 6.48,	
ļ	116467	6.102, 6.204)	
	116465	Amends § 6.20.010, license	
		fee for exhibitors and trade shows (6.20)	
ŀ	116466	Amends § 6.270.060, adult	
	110700	entertainment premises	
		license fees (6.270)	
Ì	116467	Amends § 22.904.040,	
		mobile home park license	
		fees (22.904)	
	116468	Amends § 5.32.170,	
		amusement device license	
ļ	116460	fees (5.32)	
	116469	Amends § 15.62.030, fee	
		for street vacation petitions	
ŀ	116471	(15.62) Amends §§ 5.44.140 and	
	1104/1	5.44.180, business and	
		occupation tax return filing	
		dates (5.44)	
L			

eference only.

amending

amending

applics,

curacy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

cipal			
a anda	te file		
116473	Extends to 7/31/93		
	effective date of Northgate		
a com hi	Area interim Traffic		
y for my	Standards set by Ord.		
	114630 (Repealed by		
11117	116795)		
116475	Amends § 4.20.190,		
	holidays for City		
116497	employees (4.20) Adds Ch. 5.80, Cumulative		
THE CITES	Reserve Fund, and §§		
	5.80.010-5.80.030; amends		
100010	§ 5.64.100, real estate		
COMPA	excise tax; repeals Ords.		
	108549, 113195 § 2 and		
1 40 00	114234 (5.64, 5.80)		
116500	Adds §§ 11.16.340(P) and		
	§ 11.58.015, street use		
000 -	restrictions for vehicles		
<b>File.</b>	with compression brakes		
11.6501	(11.16, 11.58)		
116501	Rezone, Plat 221 W, page		
11/512	7 (Special)		
116513	Adds Subchapter VI and § 3.20.320, TDR Bank; adds		
	§ 23.49.033 and amends §§		
	23.49.035, 23.49.048,		
	23.49.050, 23.49.052,		
	23.49.068, 23.49.070,		
	23.49.072, 23.49.098,		
	23.49.100, 23.49.102,		
	23.49.124, 23.49.126,		
	23.49.128, 23.49.152,		
	23.49.154, 23.84.024 and		
	23.84.030, Land Use Code		
116538	(3.20, 23.49, 23.84)		
110338	Adds §§ 11.16.122 and 11.40.430, no-admittance		
	areas under bridges,		
	viaducts and overpasses		
	(11.16, 11.40)		
116540	Amends § 3.02.125, filing		
	fee for landmark appeals;		
	adds § 23.40.008, Land		
	Use Code; adds §		
	25.12.835, landmark		
	demolition restrictions		
11.65.41	(3.02, 23.40, 25.12)		
116541	Amends §§ 6.270.010(D),		
	6.270.030, 6.270.070, 6.270.100, 6.270.110,		
	6.270.120 and 6.270.140,		
	including the addition of §§		
	6.270.030(C), (G) and (K),		
	6.270.070(A)(8),		
	6.270.100(A)(5) and (12),		
	(B)(3) and (E),		
	6.270.110(C) and (D),		
	6.270.120(B) and		
	6.270.140(B), licensing of		
461	adult entertainment		
wont JM	businesses (6.270)		
7 0320	Cles		

d amending graphics, accuracy of

For current 31 pusinesses (6.270)

the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Amends §§ 22.900.050, 22.900.170, 22.900.180, 22.900.220-22.900.240, 22.900.245 and 22.900.265, 1993 Permit Fees (Repealed by 117405)  Amends § 5.40.025, admission tax exemptions and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  Amends §§ 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020, 23 84 0440 and 23 84 048	Cibes	40 Sile
22.900.080, 22.900.170, 22.900.180, 22.900.245 and 22.900.265, 1993 Permit Fees (Repealed by 117405)  Amends § 5.40.025, admission tax exemptions and discounts (5.40)  116581  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  Amends § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds § 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds § 401 and 616 to the UPC, and amends § 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends § 8 4.92.010-4.92.040, suggestion award program (Repealed by 118966) 116596  Amends § 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	4. 11002	Te m
22.900.180, 22.900.220-22.900.240, 22.900.265, 1993 Permit Fees (Repealed by 117405)  Amends § 5.40.025, admission tax exemptions and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  Amends § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds § 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	116567	
22.900.220-22.900.240, 22.900.245 and 22.900.265, 1993 Permit Fees (Repealed by 117405)  Amends § 5.40.025, admission tax exemptions and discounts (5.40)  116581  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  116585  Amends § § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds § 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds § 401 and 616 to the UPC, and amends § 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends § § 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends § \$ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
22.900.245 and 22.900.265, 1993 Permit Fees (Repealed by 117405)  Amends § 5.40.025, admission tax exemptions and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  Amends § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	1 Cor 111	
22.900.265, 1993 Permit Fees (Repealed by 117405)  Amends § 5.40.025, admission tax exemptions and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  116585  Amends § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	bu in	· · · · · · · · · · · · · · · · · · ·
Fees (Repealed by 117405)  Amends § 5.40.025, admission tax exemptions and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  116585  Amends § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
Amends § 5.40.025, admission tax exemptions and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  116585  Amends § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds § 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds § 401 and 616 to the UPC, and amends § 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends § 8  4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends § 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
admission tax exemptions and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  116585  Amends § 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	116577	
and discounts (5.40)  Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle  Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	1165//	
Sets moratorium on certain advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  116585  Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  116594  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle  Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
advertising signs until 8/1/93; repeals Ord. 116256 (23.84.036 footnote)  Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle  Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	116591	` ′
8/1/93; repeals Ord. 116256 (23.84.036 footnote)  Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	110381	
116256 (23.84.036 footnote)  Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	-10	
footnote)  Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	COMPLE	•
Amends §§ 1 and 2 of Ord. 81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	Company	
81698, service award program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle  Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	116505	
program (4.88)  Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	110363	
Amends § 3.02.125, hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		1
hearing examiner filing fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	116504	
fees; repeals and replaces § 22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	110394	
22.502.015; adds §§ 10.4, 10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	1116.	0
10.5, 20.1, 20.2, 20.4-20.6, 20.8, 20.10-20.12, 20.16, 20.17 to Seattle  Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
20.8, 20.10-20.12, 20.16, 20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
20.17 to Seattle Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
Amendments to the Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
Uniform Plumbing Code; adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
adds §§ 401 and 616 to the UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
UPC, and amends §§ 102, 103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
103, 105, 113, 117, 203, 308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
308, 310, 315, 318, 411, 503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
503, 608, 803, 910, 1003, 1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
1004, 1007, 1008, entirety of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
of Appx. C and § D.1(a) of Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
Appx. D, Uniform Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
Plumbing Code, and repeals Ords. 114712 and 114713 (3.02)  116595  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
114713 (3.02)  Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
116595 Amends §§ 4.92.010-4.92.040, suggestion award program (Repealed by 118966)  116596 Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		repeals Ords. 114712 and
4.92.010-4.92.040, suggestion award program (Repealed by 118966) 116596 Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		114713 (3.02)
suggestion award program (Repealed by 118966)  116596  Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,	116595	
(Repealed by 118966)  116596 Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
116596 Amends §§ 23.44.035, 23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
23.47.014(F), 23.50.012 (Chart A), 23.50.020,		
(Chart A), 23.50.020,	116596	,
23.84.040 and 23.84.048.		
		23.84.040 and 23.84.048,
Land Use Code (23.44,		-
23.47, 23.50, 23.84)		
116598 Adds §§ 3.14.915 and	116598	
3.14.918, Office for		
Women's Rights public		
information and education		
projects (3.14)		projects (3.14)

reference only.

solution amending
in graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibon	Cilo
a anda	te tille
116616	Amends §§ 23.47.006(C),
	23.49.046(F),
1 - 0 - 10	23.49.066(F),
4 501 111	23.49.096(G),
am ros	23.49.122(F),
	23.49.324(F), 23.50.014(B)
	and (D), 23.60.092(A) and
	(B), 23.60.664, 23.60.730,
	23.68.014(B) and
- 0002	24.54.060, zoning (23.47,
PES CITE	23.49, 23.50, 23.60)
116619	Adds § 21.49.024; amends
11001)	§§ 21.49.010-21.49.040,
COMPLE	21.49.042, 21.49.052,
Compage	21.49.055, 21.49.057,
	21.49.058, 21.49.060 and
and to Co	21.49.080-21.49.160,
and co	Seattle City Light
	Department (21.49)
116620	Extends to 4/12/93
110020	
3 8 8 9 9	moratorium on pole
	attachments set out in Ord.
11.601	116247 (Not codified)
116621	Amends §§ 25.08.060 and
	25.08.520, noise control
11.500	(25.08)
116633	Adds § 15.32.030 and
	amends § 15.32.020, utility
	street use permits and pole
	attachment policy (15.32)
116641	Amends § 4.13.010,
	exemptions from the Civil
	Service (4.13)
116642	Creates Emergency Fund
	with Emergency Subfund
	and Supplemental
	Appropriation Subfund
	(5.76)
116643	Adds §§ 4.20.315 and
	4.20.320 and repeals §
	4.20.310, overtime
	provisions for certain City
	employees (4.20)
116647	Amends §§ 3.68.020 and
	3.68.030, Commission on
	Children and Youth (3.68)

eference only.

l amending

graphics,

ccuracy of

Seattle Municipal Code

Seattle Municipal Code

[116655] Update Sile

Seattle Building Code

June 2009 co liess up Text provided for

See ordinances comsections for comand tables and tables and this source file.

103(c) and (d), 514, 515 Part II of Ch. 12 (§§ 1217-1228), 1723, 2905(e), 3301(e), Ch. 55 (transformer vaults, §§ 5501-5510) and Appx. Ch. 55 (utility transformer vaults §§ 5511-5521); and amends §§ 103(a), 401(a), 402-405, 407-409, 414, 415 417, 419, 420, 423, Table 5-A, Table 5-A-1. Table 5-B, Table 5-E, §§ 604(a), 702(b), 704, 801, 804, 901(a), 902(g) and (h), 904, Table 9-B, §§ 1001, 1004, 1201, 1204, 1214, 1716(c), 1720-1722, 2004, 2104, 2605(e),

Supplement; adds §§

2621(j), 2710(g), 2905(b) and (c), 3301(f), 3303(a) and (e), 3304(a), (b), (h) and (i), 3305(a), 3306(g), (i) and (k), 3307(a), (c) and (d), 3314(a), 3315(a) and (e), Table 33-A, § 3802(e), Table 38-A, §§ 3901(b), 3903(e), 4206(a) and (b), Table 42(c), §§ 5103.

ference only.

amending

graphics,

ccuracy of

5104, 5116(e) and 5130; and amends and renumbers § 2905(e) and (f) to be (f) and (g), all pertaining to the 1991 Uniform Building Code (22.100.010 footnote) 116656 Adds Table 10.1-D and amends and renumbers former Table 10.1-D to be 10.1-E; amends §§ 301(c), 415, 421, 423, Table 6-A, §§ 914(a), 10.102-10.106, Table 10.1-B, §§ 1107(j) and 1508 and amends Appendix B § 2508 of the Uniform Mechanical Code (22.400.010 (footnote)) 116657 Adds Ch. 25.02, commute trip reduction (25.02) 116660 Extends to 6/30/94 moratorium on issuance of use permits for adult cabarets (Ch. 6.270 footnote) 116666 General Trust Fund Vice Enforcement/Money Laundering Forfeiture Account (5.76) 116681 Amends § 3.18.140, motor vehicle fleet (3.18)

For current Sing, 500 the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Shaz	. C:10
	a mada	te tille
	116682	Transit and ferry passes
		[adds § 4.20.370]; repeals
ļ	1 Con h1	Ord. 108444 (4.20)
9	116694	Adds §§ 9.25.052 and
		9.26.025; amends §§
		9.25.023, 9.25.030, 9.25.050, 9.25.082,
		9.25.084, 23.44.048,
		25.45.148 and 23.47.026,
	- a orea	animals (9.25, 9.26, 23.44,
	ces er ea	23.45, 23.47)
	116695	Rezone, Plat 58E, page 145
	-	(Special)
	116701	Adds §§ 11.84.345 and
		11.31.125 and amends §
	- 4 to CO	11.31.080, vehicles and
A		traffic (11.31, 11.84)
	116702	Adds §§ 10.08.135,
	C:10	10.08.140, 10.08.145, 10.08.155, 10.08.165,
	1116.	10.08.168, 10.08.173 and
		10.08.178; amends §§
		10.08.150, 10.08.160 and
		10.08.180 and changes title
		of Ch. 10.08, Subchapter
		III; repeals § 10.08.170,
ļ		alarms (10.08)
	116708	Amends §§ 11.74.150,
		11.74.160 and 21.36.410,
		vehicles and litter (11.74, 21.36)
ŀ	116710	Extends to 1/31/94
	110/10	effective date of Northgate
		Area interim Traffic
		Standards set by Ord.
		114630 (Repealed by
		116795)
	116711	Amends City's Agency
ļ		Service Agreement with
		the Seattle-King County
		Convention and Visitors Bureau (3.14 footnote)
ŀ	116714	Suspends §§ 4.20.315 and
		4.20.320 (4.20 footnote)
ŀ	116717	Franchise extension
		agreement with Viacom
		Cable; repeals § 21.60.610
Į		(Repealer)
	116718	Franchise extension
		agreement with TCI Cable;
		repeals § 21.60.610
ļ	116722	(Repealer)
	116723	Amends § 3.20.010, department of housing and
ļ		human services (3.20)
ŀ	116725	Amends § 21.40.050, waste
		collection rates (21.40)
L		(,

eference only.

d amending
araphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	ilbar -	
	a mada	te file
	116726	Adds §§ 3.18.907 and
		21.36.088; amends §§
	10 - m h1	3.18.904, 3.18.906,
	4 FOL 122	3.18.908, 3.18.918 and
		3.18.922, recycling (3.18,
		21.36)
	116731	Adds a new subchapter to
		Ch. 18.16 and adds §§
		18.16.150 and 18.16.160,
	oc cred	Westlake Park (18.16)
	116737	Amends § 18.12.040,
		prohibitions in parks
	91000	(18.12)
	116743	Extends time limit of
		rezone in Ord. 115663
	1 40 66	(Special)
9	116744	Amends §§ 23.04.040,
		23.45.015, 23.45.057,
	0010	23.45.073, 23.45.112,
	<b>4116.</b>	23.47.006, 23.47.016,
		23.47.027, 23.49.036,
		23.49.046, 23.49.056,
		23.49.066, 23.49.076, 23.49.096, 23.49.106,
		23.49.122, 23.49.134,
		23.49.162, 23.49.180,
		23.49.212, 23.49.214,
		23.19.240, 23.49.332,
		23.50.016, 23.50.034,
		23.53.020, 23.60.196,
		23.60.704, 23.66.020,
		23.66.025, 23.66.030,
		23.66.035, 23.66.110,
		23.66.115, 23.66.122,
		23.66.124, 23.66.130,
		23.66.140, 23.66.150,
		23.66.155, 23.66.160,
		23.66.170, 23.66.180,
		23.66.190, 23.66.310,

23.66.314, 23.66.316, 23.66.318, 23.66.326, 23.66.334, 23.66.336, 23.66.338, 23.66.342, 23.68.022, 23.69.023, 23.69.034, 23.84.006, 23.84.008, 24.08.040, 24.24.010, 24.26.010, 24.66.050, 24.70.060 and 24.74.028, land use and zoning (23.04, 23.45, 23.47, 23.49, 23.50, 23.53, 23.60, 23.66, 23.69,

Amends § 4.13.010(E), City Light positions (4.13)

Adds § 23.16.050, land use (Repealed by 117929)

Amends § 21.40.050, waste collection fees (21.40)

23.84)

eference only.

l amending

graphics,

For current SMC, contact the Office of the City Clerk

116749

116752

116754

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipai o	
A	te file
116756	Amends § 2 of Ord. 116456 and repeals § 17.30.010. Bite of Seattle
d for h	agreement (Repealer)
116761	Adds Ch. 4.26 and amends
	§§ 4.30.010(A) and (B) and 4.30.020, family leave (4.26, 4.30)
116763	Acceptance of applications
ces crea	for advertising signs (Special)
116766	Rezone, Plat 303W, page 206 (Special)
116769	Amends § 17.16.015, Seattle Center license fees
116770	(17.16) Adds § 23.16.060 and
	amends § 23.16.002(B)(3),
C110.	land use and zoning (Repealed by 117929)
116780	Amends §§ 23.55.014,
	23.55.030, 23.55.034, 23.55.042, 23.84.036 and
	23.86.004, zoning (23.55,
	23.84, 23.86)
116790	Repeals and replaces § 22.300.010 and amends
	various electrical code
	sections, electrical code
116794	(Repealed by 118553) Rezones, Plats 220E, page
	6; 221W, page 7; 225E,
	page 14; 226W, page 15; 226E, page 16; 227W, page
	17; 223W, page 28; 233E,
	page 29; and Exh. 1
116795	(Special) Adds Ch. 23.71 and §
	23.45.047 and amends
	official land use map by adding Northgate Overlay
	District; amends §§
	23.34.013, 23.34.078,
	23.45.002, 23.46.002, 23.46.012, 23.47.002,
	23.47.008, 23.47.009,
	23.47.023, 23.76.006, 23.84.028, 23.84.048 and
	23.90.018 and Exh.
	23.76.004A; repeals Ch.
	25.07, land use and zoning (23.34, 23.45, 23.46, 23.47,
	23.71, 23.76, 23.84, 23.90)
116802	Adds § 4.36.270(E) and
	amends § 4.36.230(D) and (E), city employees'
	retirement system (4.36)
116810	Adds § 4.36.400,
116816	retirement system (4.36) Rezone, Plat 232W, page
110010	26 (Special)

eference only.

Is a lamending graphics, a ccuracy of

For current Sine, \_\_\_\_\_\_\_
the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

-aanda	te file
116817	Rezone, Plat 29E, page 90
	(Special)
116818	Adds §§ 14.08.045,
ad for III	14.08.055, 14.08.095,
200	14.08.135, 14.08.165,
	14.08.185 and 14.08.187
	and amends §§
	14.08.010-14.08.040,
	14.08.050, 14.08.060-
- co orea	14.08.090,
ices er ou	14.08.100-14.08.130,
	14.08.150, 14.08.160,
	14.08.170, 14.08.180 and
COMPA	14.08.190(E), open
	housing and public
1 40 00	accommodations (14.08)
116825	Suspends §§ 4.20.315 and
0,8300	4.20.320 until October 31,
	1993, employee provisions
<b>C</b> 112	(4.20)
116832	Amends §§ 6.204.050(C)
	and 6.212.115, business
	regulations (6.204)
116841	Amends § 10.76.060,
	hazardous waste
	management fees (10.76)
116843	Adds a new chapter to Title
	7, rental agreement
	regulation (7.24)
116850	Amends § 16.60.050, use
	of waterways-permits
	(16.60)
116855	Amends § 14 of Ord.
	116616, land use and
	zoning (Not codified)
116858	Repeals § 21.49.024,
	surcharge on electric rates
	(Repealer)
116859	Abrogates certain City
	Light positions and amends
	§ 4.13.010(E), City Light
115054	Department (4.13)
116864	Amends §§ 4.20.315 and
	4.20.320, employee
116067	compensation (4.20)
116867	Amends §§ 4.20.170,
	4.20.210, 4.20.250 and
	4.20.300, compensation
	and working conditions
116071	(4.20)
116871	Amends §§ 11.16.125 and
	11.40.430, closure of alleys
	(11.16, 11.40)

d amending graphics, accuracy of

Seattle Municipal Code
June 2009 co 116872 UP Adds §
12A.02
12A.06
11.20.0
11.56.0
12A.06
12A.12
12A.14

See ordinant sections for and tables a this source

	ipai co	te file
	116872	Adds §§ 11.10.130,
		12A.02.035 and
	- 0 - m hi	12A.06.035 and amends §§
	4 FOL 111	11.20.010, 11.34.020,
		11.56.020, 11.56.320,
		12A.06.010, 12A.06.030,
		12A.12.040, 12A.14.010,
		12A.14.020, 12A.14.035 and 12A.14.080 and
		repeals §§ 12A.04.020,
	res crea	12A.04.040, 12A.06.015
}		and 12A.14.150, various
		provisions (11.10, 11.20,
	COMPA	11.34, 11.56, 12A.02,
,		12A.06, 12A.12, 12A.14)
	116873	Amends § 4.13.010 and
	na to to	establishes titles, retitles,
		reclassifies and sets
	0210	compensation for certain positions, employees (4.13)
	116880	Amends § 11.56.020,
	110000	driving while intoxicated
		(11.56)
İ	116885	Adds §§ 15.48.040 and
		15.48.050, sitting or lying
		down on public sidewalks
	11.000	(15.48)
	116893	Amends § 18.12.040,
		superintendent authority (18.12)
ŀ	116896	Amends §§ 12A.10.100
	110070	and 12A.24.025, urinating
		or consuming liquor in
		public (12A.10, 12A.24)
	116987	Amends §§ 12A.12.015,
		12A.24.100 and
		12A.24.150, pedestrian interference and liquor
		consumption (12A.12,
		12A.24)
ı	116903	Amends arterial street map
		(11.18)
	116907	Amends §§ 23.47.006,
		23.49.046, 23.49.066,
		23.49.096, 23.49.122, 23.49.324, 25.50.012
		(Chart A), 23.50.012
		23.60.092, 23.60.664,
		23.60.730, 23.68.014,
		24.44.070, 24.44.080,
		24.46.060, 24.46.070,
		24.48.050, 24.48.060,
		24.52.100, 24.54.060 and
		24.54.100, zoning (23.47, 23.49, 23.50, 23.60)
L		23.77, 23.30, 23.00)

reference only.

d amending
paphics,
accuracy of

June 2009 co 1169

Text provide

See ordinant sections for and tables a this source

	Shar	0210
		te ille
	116909	Adds Ch. 23.41 and §
	0.0	23.76.011, and amends §
	- 101	3.06.030, 23.60.154,
	4 501 111	25.76.004A, 23.76.006,
-	CI TO	23.76.008, 23.76.012,
		23.76.024 and 25.05.675
		and repeals § 23.40.010,
		zoning and other
		provisions (3.06, 23.41,
	- 0003	23.60, 23.76, 25.05)
	116912	Rezone, Plat 3W, page 38
3	110)12	(Special)
ŀ	116914	Amends § 4.13.010(I),
		employees (4.13)
5	116917	Adds § 21.60.295 and
	110717	amends §§ 21.60.300 and
	and to co	21.60.320, cable
4	Her co	communications (21.60)
	116924	
	110724	Adds §§ 6.288.010(I), 6.288.065, 6.288.085,
	1116.	6.288.130 and amends §§
		6.288.030(A), 6.288.040
		and 6.288.120(C) and (D),
-	11/025	used goods dealers (6.288)
	116925	Rezone, Plat 234W, page
ŀ	11.002	30 (Special)
	116926	Amends §§ 9.26.050 and
		9.26.060, animal fees
ŀ	11.007	(9.26)
	116927	Adds § 21.36.450, fee on
ļ	11 (020	unsecured loads (21.36)
	116928	Amends §§ 21.28.370 and
		21.76.040, drainage and
ļ	11.000	wastewater utility (21.76)
	116929	Amends § 18.28.010, fee
ļ	11 (022	schedule (18.28)
	116932	Amends § 17.16.015, fees
ļ	11.000	and charges (17.16)
	116933	Amends § 4.13.010 (B),
		(G), (I), (J), (M), (N), (S),
		(U) and (BB), exemptions
		from the Civil Service
ŀ	116024	(4.13)
	116934	Amends § 10.03.110, food
ļ	116025	service fees (10.03)
	116935	Amends § 10.03.130,
ļ	116026	health care fees (10.03)
	116936	Amends § 10.03.013, food
ļ	11,007	fees (10.03)
	116937	Amends § 10.03.120,
		school plan review fees
ļ	11,000	(10.03)
	116938	Amends §§ 10.25.020 and
		10.25.030, water and
ļ	11.1000	sewage fees (10.25)
	116939	Amends § 22.504.010,
		plumbing permit fees
ļ	11.10.10	(22.504)
	116940	Adds §§ 10.03.200 and
		10.03.210, sale of drugs to
Į	-4 SM	clinics (10.03)

d amending graphics, accuracy of

For current State of drugs to clinics (10.03)

The Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Amends §§ 10.03.140 and 10.03.150, immunizations (10.03)  Amends § 10.03.056, swimming and spa pool permit fees (10.03)  Amends § 10.03.045, laboratory fees (10.03)  Amends §§ 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  Amends § 17.19.010, parking charges (17.19)  Adds §§ 22.900.200 and 22.900.300 and amends § 22.900.300 and amends § 22.900.010, 22.900.010, 22.900.025, 22.900.035, 22.900.040, 22.900.050, 22.900.060, 22.900.070, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.180, 22.900.180, 22.900.180, 22.900.185, 22.900.240, 22.900.245, 22.900.240, 22.900.245, 22.900.260, 22.900.245, 22.900.270 and 22.900.28 and repeals §§ 22.900.280 and 22.900.290, Department of	
10.03.150, immunizations (10.03)  116942  Amends § 10.03.056, swimming and spa pool permit fees (10.03)  116943  Amends § 10.03.045, laboratory fees (10.03)  116945  Amends § 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds § 22.900.200 and 22.900.010, 22.900.010, 22.900.010, 22.900.025, 22.900.035, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.080, 22.900.090, 22.900.110, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.240, 22.900.265, 22.900.270 and 22.900.280 and repeals §§ 22.900.250 and 22.900.290, Department of	_
10.03.150, immunizations (10.03)  116942  Amends § 10.03.056, swimming and spa pool permit fees (10.03)  116943  Amends § 10.03.045, laboratory fees (10.03)  116945  Amends § 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds § 22.900.200 and 22.900.010, 22.900.010, 22.900.010, 22.900.025, 22.900.035, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.080, 22.900.090, 22.900.110, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.240, 22.900.265, 22.900.270 and 22.900.280 and repeals §§ 22.900.250 and 22.900.290, Department of	
116942  Amends § 10.03.056, swimming and spa pool permit fees (10.03)  116943  Amends § 10.03.045, laboratory fees (10.03)  116945  Amends § 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds § 22.900.200 and 22.900.010, 22.900.010, 22.900.010, 22.900.025, 22.900.035, 22.900.044, 22.900.045, 22.900.070, 22.900.080, 22.900.070, 22.900.100, 22.900.110, 22.900.110, 22.900.110, 22.900.120, 22.900.150, 22.900.150, 22.900.185, 22.900.180, 22.900.240, 22.900.245, 22.900.266, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.290, Department of	- 4
swimming and spa pool permit fees (10.03)  Amends § 10.03.045, laboratory fees (10.03)  Amends §§ 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  Amends § 17.19.010, parking charges (17.19)  Adds §§ 22.900.200 and 22.900.300 and amends § 22.900.300 and amends § 22.900.010, 22.900.025, 22.900.040, 22.900.045, 22.900.050, 22.900.045, 22.900.070, 22.900.100, 22.900.110, 22.900.110, 22.900.120, 22.900.110, 22.900.180, 22.900.185, 22.900.180, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
swimming and spa pool permit fees (10.03)  116943  Amends § 10.03.045, laboratory fees (10.03)  116945  Amends § 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds § 22.900.200 and 22.900.010, 22.900.010, 22.900.010, 22.900.004, 22.900.025, 22.900.044, 22.900.045, 22.900.090, 22.900.080, 22.900.090, 22.900.100, 22.900.110, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.150, 22.900.185, 22.900.180, 22.900.240, 22.900.245, 22.900.240, 22.900.265, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
permit fees (10.03)  Amends § 10.03.045, laboratory fees (10.03)  116945  Amends § 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  Amends § 17.19.010, parking charges (17.19)  Adds § 22.900.200 and 22.900.010, 22.900.010, 22.900.010, 22.900.025, 22.900.035, 22.900.044, 22.900.045, 22.900.070, 22.900.080, 22.900.070, 22.900.100, 22.900.110, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.150, 22.900.185, 22.900.180, 22.900.240, 22.900.245, 22.900.266, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
laboratory fees (10.03)  Amends §§ 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds §§ 22.900.200 and 22.900.300 and amends § 22.900.010, 22.900.010, 22.900.025, 22.900.030, 22.900.035, 22.900.040, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.100, 22.900.110, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.20, 22.900.210, 22.900.20, 22.90	
Amends § 5.44.140 and 5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds § 22.900.200 and 22.900.010, 22.900.010, 22.900.020, 22.900.025, 22.900.035, 22.900.044, 22.900.045, 22.900.070, 22.900.080, 22.900.070, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.150, 22.900.185, 22.900.180, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.266, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	_
5.44.130(A) and repeals § 5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds §§ 22.900.200 and 22.900.300 and amends § 22.900.010, 22.900.025, 22.900.035, 22.900.044, 22.900.045, 22.900.070, 22.900.080, 22.900.070, 22.900.100, 22.900.110, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.185, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.265, 22.900.270 and 22.900.280 and repeals §§ 22.900.290, Department of	
5.44.120, business and occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds §§ 22.900.200 and 22.900.300 and amends § 22.900.010, 22.900.025, 22.900.035, 22.900.044, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.080, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.185, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.250, 22.900.265, 22.900.270 and 22.900.280 and repeals §§ 22.900.250 and 22.900.290, Department of	
occupation tax (5.44)  116946  Amends § 17.19.010, parking charges (17.19)  116949  Adds §§ 22.900.200 and 22.900.300 and amends §; 22.900.010, 22.900.025, 22.900.030, 22.900.035, 22.900.040, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.080, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.170, 22.900.180, 22.900.185, 22.900.20, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.280 and repeals §§ 22.900.250 and 22.900.290, Department of	
116946 Amends § 17.19.010, parking charges (17.19)  116949 Adds §§ 22.900.200 and 22.900.300 and amends §; 22.900.010, 22.900.025, 22.900.035, 22.900.044, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.080, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.185, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.250, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
parking charges (17.19)  Adds § 22.900.200 and  22.900.300 and amends § 22.900.010, 22.900.025, 22.900.030, 22.900.035, 22.900.040, 22.900.045, 22.900.070, 22.900.080, 22.900.070, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.170, 22.900.180, 22.900.185, 22.900.20, 22.900.230, 22.900.240, 22.900.245, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	_
Adds §§ 22.900.200 and 22.900.300 and amends § 22.900.010, 22.900.010, 22.900.025, 22.900.035, 22.900.044, 22.900.045, 22.900.070, 22.900.080, 22.900.090, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.250, 22.900.270 and 22.900.280 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.300 and amends \$\frac{1}{2}2.900.010, 22.900.010, 22.900.025, 22.900.033, 22.900.045, 22.900.050, 22.900.050, 22.900.050, 22.900.060, 22.900.070, 22.900.080, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.150, 22.900.185, 22.900.180, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.250, 22.900.265, 22.900.270 and 22.900.28 and repeals \$\frac{1}{2}2.900.290, Department of	
22.900.010, 22.900.010, 22.900.025, 22.900.030, 22.900.035, 22.900.044, 22.900.045, 22.900.050, 22.900.050, 22.900.060, 22.900.070, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.185, 22.900.180, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.20, 22.900.245, 22.900.266, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.020, 22.900.025, 22.900.030, 22.900.035, 22.900.040, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.170, 22.900.180, 22.900.185, 22.900.20, 22.900.20, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.280 and repeals §§ 22.900.250 and 22.900.290, Department of	5
22.900.030, 22.900.035, 22.900.044, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.100, 22.900.110, 22.900.120, 22.900.150, 22.900.150, 22.900.150, 22.900.150, 22.900.150, 22.900.150, 22.900.185, 22.900.190, 22.900.210, 22.900.20, 22.900.245, 22.900.266, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.290, Department of	
22.900.040, 22.900.045, 22.900.050, 22.900.060, 22.900.070, 22.900.080, 22.900.090, 22.900.100, 22.900.110, 22.900.120, 22.900.150, 22.900.150, 22.900.150, 22.900.150, 22.900.185, 22.900.190, 22.900.210, 22.900.20, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.070, 22.900.080, 22.900.090, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.150, 22.900.150, 22.900.150, 22.900.150, 22.900.185, 22.900.190, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.090, 22.900.100, 22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.170, 22.900.180, 22.900.185, 22.900.190, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.110, 22.900.120, 22.900.130, 22.900.140, 22.900.150, 22.900.170, 22.900.185, 22.900.190, 22.900.210, 22.900.220, 22.900.230, 22.900.244, 22.900.245, 22.900.266, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.130, 22.900.140, 22.900.150, 22.900.170, 22.900.180, 22.900.185, 22.900.190, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.150, 22.900.170, 22.900.180, 22.900.185, 22.900.190, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.180, 22.900.185, 22.900.190, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.190, 22.900.210, 22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.220, 22.900.230, 22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.240, 22.900.245, 22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
22.900.260, 22.900.265, 22.900.270 and 22.900.28 and repeals §§ 22.900.250 and 22.900.290, Department of	
and repeals §§ 22.900.250 and 22.900.290, Department of	
and 22.900.290, Department of	
Department of	
Construction and Land Us	
fees (Repealed by 117405 116951 Amends §§ 5.44.026,	,
5.44.030, 5.44.110 and	
5.48.070, excise taxes	
(5.44, 5.48)	
116955 Adds § 5.48.072 and	
amends §§ 5.48.020 and	
5.48.050, business	
tax-utilities (5.48)	
Adds Arctic Building to	
Ch. 25.32, table of historical landmarks	
(25.32)	
116970 Adds Dexter Horton	-
Building to Ch. 25.32,	
table of historical	
landmarks (25.32)	
116971 Adds James A. Moore	
Mansion and its site to Ch	
25.32, table of historical	
landmarks (25.32)	_
Adds Admiral Theater to	
Ch. 25.32, table of historical landmarks	
(25.32)	
mont SMUT	

eference only.

al amending
graphics,
accuracy of

For current Sme, come, the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Sihar	0210
a and	ite ille
116973	Adds Rainier Cold Storage
	and Ice/Seattle Brewing
1 Co# h1	and Malting Company
din	Building and its site to Ch.
	25.32, table of historical landmarks (25.32)
116976	Amends §§ 25.05.908,
1107/0	25.09.040, 25.09.060,
	25.09.080, 25.09.140,
on orea	25.09.160, 25.09.180,
ces er er	25.09.200, 25.09.240,
	25.09.260, 25.09.280,
	25.09.300, 25.09.320,
Comba	25.09.340 and 25.09.420,
	environmentally critical
116977	areas (25.09) Adds § 11.58.275,
1109//	right-of-way of transit
	vehicles (11.58)
116980	Amends § 4.70.025,
Tile.	personnel director (4.70)
116991	Adds § 12A.10.110,
	mandatory counseling
	(Repealed by 117074)
117000	Amends §§ 10.01.180,
	10.01.190, 10.03.010 and
	10.03.110, health fees
	(10.01, 10.03)
117001	Adds Ch. 10.10 and repeals
	Ch. 10.11 and Ords. 109910, 112024, 114582,
	114838 and 116437,
	Seattle Food Code (10.11)
117002	Adds §§ 5.44.130(D),
	6.20.030-6.20.070 and
	6.20.100, and amends §
	6.20.010, trade shows and
	street fairs (5.44, 6.20)
117017	Amends Ch. 5.76, table of
	funds (5.76)
117018	Amends § 4.36.100,
117010	retirement system (4.36)
117019	Amends §§ 4.04.030 and 4.04.150, and repeals §
	4.20.180, personnel
	regulations (4.04)
117035	Amends Ch. 5.76, table of
117033	funds (5.76)
117039	Adds § 4.20.860 and
	amends Subchapter III of
	Ch. 4.20 (§§
	4.20.800-4.20.850),
	compensation and working
	conditions generally (4.20)
117043	Adds § 11.72.051 and
	amends § 11.70.100,
	vehicles and traffic (11.70,
117056	11.72)
117056	Amends § 4.16.080,

eference only.

l amending

graphics,

ccuracy of

For current SM interest (4.16)

The Office of the City Clerk

Amends § 4.16.080, statements of financial

June 2009 co 117066 upd

Text provide

See ordinant sections for and tables a this source

Gihar	C210
a and	ate tille
117066	Adds §§
	15.48.100-15.48.130,
1 0 - m h	15.50.050 and
d for II	15.90.020(P), posting of
	handbills (15.48)
117074	Repeals and replaces §
	12A.10.110, counseling
	and costs for convicted
	persons (12A.10)
117075	Amends § 23.41.008,
ces cro	zoning (23.41)
117076	Rezone, Plat 221E, page 8
10000	(Special)
117080	Adds § 20.46.001 and Ch.
	20.46A, women's and
	minority business
TO C	enterprise utilization
Tites o	(20.46, 20.46A)
117081	Amends § 22.700.010 and
117001	certain sections of the
1116.	Washington State Energy
	Code (22.700)
117089	Amends § 17.40.090,
117089	Bumbershoot Festival
	admission fees (Repealed
	by 118507
117092	Adds Ch. 10.12, and
117092	·
	repeals Ch. 22.906, water
	recreation facility code
117094	(10.12)
117094	Amends §§ 22.210.040,
	22.210.060, 22.210.090,
	22.210.100,
	22.210.110-22.210.150 and
	22.210.180, and suspends § 22.210.110, tenant
	relocation assistance
	(22.210)
117007	,
117097	Amends Ch. 25.32, Table of Historical Landmarks
117102	(25.32)
117103	Amends § 15.48.040,
	sitting or lying down on public sidewalks (15.48)
117104	
11/104	Amends § 12A.12.015,
	pedestrian interference
117100	(12A.12)
117108	Amends § 11.16.317,
	restricted parking zones
117115	(11.16)
117115	Amends § 21.49.005, rate
	surcharge (Repealed by
	119747)
117116	Amends § 17.16.105 (L),
	rental agreements for
	Seattle Center (Repealed
	by 117776)
117117	Rezone, Plat 77E, page 182
	(Special)
-	- 104

reference only.

d amending

graphics,
accuracy of

June 2009 co 117123 1119

Text provide

See ordinant sections for and tables a this source

		te file
	117123	Amends §§
	11/123	17.19.040-17.19.060 and
	1-3	repeals §§ 17.19.020 and
	1 50° III	17.19.030, parking (17.19)
9	117132	
	117132	Amends § 4.13.010 (C),
		exemption from civil service (4.13)
	117141	Extends to 6/30/95
	11/141	moratorium on permits for
	DOS CITE	adult entertainment
7		premises (Ch. 6.270 footnote)
	117145	
	117143	Rezone, Plat 41W, page
5	117146	112 (Special)
	11/140	Amends § 3.18.832,
	ad to co	contracting with sheltered
4	117149	workshops (3.18)
	T1 / 149	Rezone, Plat 23E, page 78
	117151	(Special)
	117151	Adds § 4.38.208 and
		amends §§ 4.38.204,
		4.38.401, 4.38.402,
		4.38.406, and §§ 8-11 of
		Ord. 112517, voluntary
		deferred compensation plan
	117152	(Repealed by 119159)
	11/152	Rezone, Plat 40E, page 111 (Special)
	117154	Amends §§ 9.25.081 and
	11/134	9.25.100, offenses relating
		to cruelty (9.25)
	117155	Adds §§ 11.14.023 and
	11/133	11.14.490, and amends §
		11.56.020, traffic code
		(11.14, 11.56)
	117156	Amends §§ 11.58.005,
	117130	12A.06.110, 12A.16.020,
		12A.24.025 and
		12A.30.050. traffic code
		and criminal code (11.58,
		12A.06, 12A.16, 12A.24,
		12A.30)
	117157	Adds §§ 12A.06.195,
	11.10,	12A.14.071 and
		12A.14.081; amends §§
		12A.14.010, 12A.14.075
		and 12A.14.080; repeals §§
		21A.14.020, 12A.14.035,
		12A.14.040, 12A.14.055,
		12A.14.060, 12A.14.070,
		12A.14.085, 12A.14.090,
		12A.14.110, 12A.14.125,
		12A.14.130, 12A.14.135,
		12A.14.155, 12A.14.165,
		12A.18.040, 12A.18.050,
		12A.18.060 and
		12A.28.050, criminal code
		(12A.14, 12A.18)
,		, ,

eference only.

deference only.

and amending

graphics,

accuracy of

Seattle Municipal Code ference only. 117158 Amends §§ 12A.06.035 and 12A.16.010, and repeals §§ 12A.10.070 and 12A.20.030, criminal code (12A.06, 12A.16) Adds § 20.46A.300 and amends §§ 3.18.802, 3.18.806 (A), 3.18.810, 20.46A.020 (A), 20.46A.020 (F), See ordinan 20.46A.180 (A), and 20.46A.220, women and minority business sections for enterprise preference program (3.18, 20.46A) and tables and this source file.

June 2009 co 117169 Text provided for hi

See ordinances sections for comp and tables and this source file.

Adds § 3.38.030 (H) and 6.230.030 (15)-(17), renumbers previous § 3.38.030 (H) to be (I), renumbers and amends § 3.24.040 to be 3.02.115 and amends §§ 3.38.010, 3.63.020, 3.90.010, 4.13.010 (J), 5.32.030 (A), 5.40.028 (B), 5.40.060 (A), 5.40.070, 5.40.080,

5.40.085, 5.40.090, 5.40.100, 5.44.022 (5), 5.44.024 (1), 5.44.028 (3),

(7), 5.44.060 (B), 5.44.070, 5.44.075 (C), 5.44.076,

5.44.078, 5.44.080 (B), 5.44.130 (A), CCS, 5.44.140 (A), (C),

5.44.150, 5.44.160,

5.44.180, 5.44.200, 5.44.205, 5.44.210,

5.44.220, 5.44.230,

5.44.240, 5.44.250,

5.44.300, 5.44.310 (A), 5.48.020 (D, 5.48.080,

5.48.090 (B), 5.48.120,

5.48.130, 5.48.140 (A), 5.48.150, 5.48.160,

5.48.170, 5.48.180,

5.48.210, 5.52.020 (A),

5.52.030, 5.52.040,

5.52.060, 5.78.150, 6.02.030 (C), 6.02.040,

6.02.050, 6.02.060,

6.02.080 (A), 6.02.090,

6.02.120, 6.02.150,

6.02.170, 6.02.190 (A), (B), 6.02.200, 6.02.210,

6.02.290, 6.14.020,

6.14.090, 6.14.100,

6.36.020 (A), (C),

6.36.030, 6.36.040 (B), (E),

6.38.040, 6.42.010,

6.42.020, 6.42.050 (A), 6.48.040 (E), 6.48.050 (B),

6.48.080, 6.82.020 (E),

6.82.030 (D), 6.82.060,

6.82.100, 6.82.110, 6.82.150 (B), 6.82.180,

6.82.190, 6.102.020,

6.102.070, 6.102.090,

6.202.060 (A)(6), (A)(7), 6.202.460 (B), 6.212.042

(A), (E), 6.212.150.

6.212.160 (B), (E), 6.222.010 (B), (C),

6.230.030, 6.230.040,

6.230.070, 6.230.080,

6.230.100, 6.230.130,

6.230.170, 6.230.180,

6.270.030, 6.270.040, 6.270.050, 6.270.070,

lerk

6.294.040 (A), (B), 6.294.140 (B), (C),

7.04.130, 7.04.570, 7.08.010, 9.22.020 (B),

9.22.030, 9.25.020 (E), (F), 9.25.025 (C), 9.25.046,

ference only. amending

For current the Office of th

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipus	te file
117173	Amends §§ 23.45.005,
	23.45.006, 23.45.008,
1 Co# 111	23.45.009, 23.45.010,
ad IOI	23.45.014, 23.45.016,
	23.45.018, 23.84.006, land
	use and zoning (23.45)
117175	Amends § 22.900.150,
	housing fees (Repealed by
	117405)
117178	Adds § 1.04.020 and §
LES CICO	3.42.020 (J), renumbers
	former § 3.42.020 (J) to be
10	
COMPLE	(K), effective dates of
COLLEGE	ordinances (1.04, 3.42)
117183	Subdivides and amends §
140 00	5.48.050 to be §§ 5.48.050
and to co	and 5.48.055, amends §
0,88	5.48.060 and repeals §
	5.48.220, utility tax on
C210.	garbage transfer (5.48)
117184	Amends §§ 21.40.050,
	21.40.060, 21.40.080,
	21.40.085 and 21.76.040,
	solid waste utility (21.40,
	21.76)
117189	Amends § 11.73.010,
	restricted parking areas
	(11.73)
117202	Adds §§ 23.40.040 and
	23.44.015, amends §§
	23.44.006, 23.44.072,
	23.45.004, 23.45.080,
	23.45.088, 23.45.168,
	23.47.004, 23.49.026,
	23.50.012, 23.54.015,
	23.76.004, 23.76.006,
	23.84.002, 23.84.004,
	23.84.006, 23.84.008,
	23.84.012, 23.84.014,
	23.84.016, 23.84.026,
	23.84.032 and 23.84.036,
	and repeals §§ 23.44.020,
	23.45.084, 23.45.086,
	23.45.118 and 23.45.120,
	land use and zoning (23.40,
	23.44, 23.45, 23.47, 23.49,
	23.50, 23.54, 23.84)
117203	Adds §§ 23.44.025 and
11,203	23.90.019, and amends §§
	23.44.006, 23.45.005,
	23.47.004, 23.76.032,
	23.84.008, 23.84.028,
	23.84.032 and 23.90.006,
	land use and zoning (23.44,
	23.45, 23.47, 23.84, 23.90)
117214	Rezone, Plat 75W, page
	177 (Special)

reference only.

reference only.

and

yel amending

peraphics,

peraphics,

curacy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

Cihar	4 Sile
a anda	te me
117218	Adds §§
	9.25.035-9.25.037, amends §§ 9.25.020, 9.25.025,
1 50° M	9.25.030,
50 101	9.25.045-9.25.047,
	9.25.050, 9.25.054,
	9.25.080, 9.25.081,
	9.25.091(D), 9.25.110 and
	9.25.120, and repeals Ch.
117210	9.22, animal control (9.25)
117219	Amends § 4.13.010(X), community policing
10	director (4.13)
117230	Adds § 23.60.724(D) and
	amends § 23.60.728(B),
140 00	land use and zoning
and to co	(23.60)
117234	Amends § 20.48.070, daily
0210	administration charges (Repealed by 117390)
117235	Amends §§ 11.16.315 and
117255	11.16.317, and repeals §
	11.73.200, restricted
	parking zones (11.16)
117238	Rezone, Plat 41W, page
117000	112 (Special)
117239	Adds §§ 11.66.025 and
	11.66.230, amends § 11.66.020 and repeals §§
	11.66.030 and 11.66.050,
	railroad company (11.66)
117241	Amends §§ 4.36.040(B)
	and (E), 4.36.120A,
	4.36.125(C) and
	4.36.330(B), retirement
117242	system credit (4.36) Amends §§ 2.04.060,
11,2,2	3.18.806(G), 3.26.040(L),
	3.28.010(D), 3.28.640,
	3.70.100(H), 4.04.250(D)
	and (E), 4.08.040(E),
	5.12.050, 5.14.060(B),
	6.214.070, 10.08.080(D)(1),
	12A.08.105, 12A.62.020,
	14.12.320(D),
	14.12.330(B)(1),
	14.12.380, 15.04.060(B),
	15.62.110, 18.08.020,
	20.76.060, 21.16.330,
	21.28.020, 21.72.010, 22.208.110(C), 22.806.030,
	23.76.058(C)(1), duties of
	city officials (2.04, 3.18,
	3.26, 3.28, 3.70, 4.04, 4.08.
	5.12, 5.14, 6.214, 10.08,
	12A.08, 12A.62, 14.12,
	15.04, 15.62, 18.08, 20.76, 21.16, 21.28, 21.72,
	22.208, 23.76)
	C COMITO

eference only.

Identify amending

Solution of a ceuracy of

Seattle Municipal Code

Seattle Municipal Code

Rezone,

June 2009 co li<sup>17245</sup> <del>upus</del> Text provided for hi

See ordinances comsections for comand tables and tables and this source file.

ces createmples and to consider

117251

117256

117257

117258

17259

Rezone, Plats 219W, page 3; 221E, page 8; 226E, page 16; 231E, page 25; 232E, page 27; 233E, page 29; 3W, page 38; 5W, page 42; 5E, page 43; 6W, page 44; 10E, page 52; 11W, page 53; 12W, page 55; 12E, page 56; 14W, page 59; 14E, page 60; 15W, page 61; 15E, page 62; 16W, page 63; 16E, page 64; 20E, page 72; 21E, page 74; 22W, page 75; 23W, page 77; 27E, page 86; 28E, page 88; 29W, page 89; 29E, page 90; 30W, page 91; 32E, page 96; 33, page 97; 34W, page 98; 34E, page 99; 35W, page 100; 35E, page 101; 36W, page 102; 36E, page 103; 37W, page 104; 38E, page 107; 39W, page 108; 40W, page 110; 40E, page 111; 41E, page 113; 42W, page 114; 44E, page 117; 45W, page 118; 46W, page 120; 47W, page 122; 47E, page 123; 48W, page 124; 48E, page 125; 49W, page 126; 51E, page 131; 52W,

page 132; 55E, page 139; 56W, page 140; 57E, page 143; 59W, page 146; 60E, page 148-a; 61E, page 150; 63W, page 153; 63E, page 154; 64E, page 156; 66W, page 159; 68W, page 163; 69W, page 165; 73E, page 174; 76W, page 179; 78W, page 183; 78E, page 184; 80E, page 188; 81W, page 188-a; 81E, page 188-b

(Special)

(25.32)

Adds Lake Unit Steam Plant and Hydro House and its site to Ch. 25.32, historical landmarks

Amends Ch. 5.76 and § 5.80.020, cumulative reserve fund (5.76, 5.80)

Amends § 4.20.030, salary upon appointment (4.20)

Adds Ch. 4.14 and repeals §§ 4.72.100-4.72.130 and 4.76.010-4.76.030, executive recruiting (4.14)

Amends §§ 4.20.315 and

eference only.

amending

graphics,

ccuracy of

For current SM 4.20.320, executive and merit leave (4.20)

The Office of the City Clerk

Seattle Municipal Code

Seattle Municipal Code

Amends §§ 10.76.020,
10.76.060 and 21.44.060,
hazardous waste
management (10.76, 21.44)

Adds §§ 23.44.008(H) and
23.50.012(C) amends §°

See ordinances crises sections for compand tables and to this source file.

3.20.320, 23.22. 23.24.040, 23.42 23.42.040, 23.44 23.44.010, 23.44 23.44.014, 23.44 23.44.022, 23.44 23.44.050, 23.45 23.45.060, 23.45 23.45.152, 23.45 23.45.184, 23.45 23.47.006, 23.47 23.47.014, Exh. ference only.

amending

raphics,

management (10.76, 21.44) Adds §§ 23.44.008(H) and 23.50.012(C), amends §§ 3.20.320, 23.22.024, 23.24.040, 23.42.020, 23.42.040, 23.44.006, 23.44.010, 23.44.012, 23.44.014, 23.44.016, 23.44.022, 23.44.040, 23.44.042, 23.44.044, 23.44.050, 23.45.004, 23.45.014, 23.45.054, 23.45.060, 23.45.076, 23.45.152, 23.45.182, 23.45.184, 23.45.190, 23.47.006, 23.47.011, 23.47.016D, § 23.47.036, Chart E in § 23.74.044, §§ 23.49.033, 23.49.035, 23.49.052, Table for Section 23.49.070, §§ 23.49.072, 23.49.074, 23.49.076, Public Benefit Feature Bonus Table for Section 23.49.126, §§ 23.49.128, 23.49.134, 23.49.152, 23.49.160, 23.49.162, 23.49.166, 23.49.248, Maps 1B, IIID, VIA and VID in Ch. 23.49, §§ 23.53.025, 23.53.030, 23.54.020, 23.54.030, Exh. 23.76.004A, §§ 23.76.006, 23.76.010, 23.76.022, 23.76.034, 23.78.014, 23.79.012, 23.84.002, 23.84.006, 23.84.008, 23.84.014, 23.84.024, 23.84.028, 23.84.030, 23.84.032, 23.84.036, 23.84.046, 23.86.002, Exh. 23.86.008D, §§ 23.86.010, 23.90.006 and 23.90.008, and repeals §§ 23.47.023(E), land use and zoning (23.20, 23.22, 23.24, 23.42, 23.44, 23.45, 23.47, 23.49, 23.50, 23.53, 23.54, 23.76, 23.78, 23.79, 23.84, 23.86, 23.90) Amends §§ 21.28.370 and 21.76.040(A), drainage and wastewater utility (21.76) Adds § 17.16.040 and amends § 18.28.010, vending (17.16, 18.28) Amends Ch. 5.76, table of funds (5.76)

For cur<del>rent Strands (3.70)</del>
The Office of the City Clerk

117265

117270

117273

June 2009 co

Text provide

See ordinantes sections for and tables at this source

	;1pu-	to file
	117200	Amends § 23.84.025, land
	117280	
	117287	use and zoning (23.84)
	11/28/2019	Rezone, Plat 7W, page 46 (Special)
	117290	Suspends § 22.210.110 to
	117290	October 2, 1995 and
		continues amendment of §
		22.210.130 from and after
		October 4, 1994 (22.210)
	117296	Amends § 21.04.440, water
		rates (21.04)
<b>B</b> '	117301	Adds §§ 4.04.145,
	117501	4.20.055 and 4.34.100, and
10	COMPA	repeals §§ 4.04.125 and
		4.34.070, temporary
	1 40 06	employees (4.04, 4.20,
	and to co	4.34)
	117302	Adds § 4.36.190
		[4.36.192], retirement
	<b>C</b> 110.	system (4.36)
9	117306	Adds §§ 11.30.030 and
		11.30.290; amends §§
		11.30.010, 11.30.040,
		11.30.060(B) and (C),
		11.30.080, 11.30.100(A)
		and (C), 11.30.120,
		11.30.180, 11.30.200,
		11.30.240 and
		11.30.360(B); repeals
		11.30.340, vehicle
		impounding (11.30)
	117308	Adds §§ 2.04.010(28),
		2.04.075, 2.04.250(H) and
		2.04.375; Adds § 2.04.480
		to comprise a new
		Subchapter VI and
		renumbers former
		subchapter VI to be
		Subchapter VII; amends §§
		2.04.010(7), (13) and (14),
		2.04.070, 2.04.090,
		2.04.170, 2.04.220,
		2.04.250(B) and (D),
		2.04.300, 2.04.330, 2.04.370, 2.04.500,
		2.04.510 and 2.04.520;
		repeals § 2.04.100, city elections (2.04)
	117319	Rezone, Plat 16W, page 63
	11/317	(Special)
	117321	Amends Ch. 4.21, standby
	11/321	duty (4.21)
	117327	Amends § 4.04.220, layoff
	11/34/	(4.04)
	117331	Amends § 5.44.090,
	11/331	exemptions (5.44)
	117342	Development rights fund
	11,572	(5.76)
		(0.70)

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
117350	Adds Ch. 23.72 and
	amends §§ 23.47.002(C),
19	23.47.008(A),
A FOR III	23.47.012(A) and
	23.47.013; rezone, Plat
	40E, page 111, 12th
	Avenue Overlay District
	(23.47)
117360	Amends § 4.04.220(C),
- co orea	layoff (4.04)
117371	Amends § 6.288.065,
	pawnbrokers (6.288)
117380	Amends § 5.44.110(B),
COMPA	license fees (5.44)
117383	Adds §§ 23.22.060;
1 40 C	23.,44.015 [23.44.013];
	23.45.007; 23.46.024;
	23.47.033; 23.49.021;
	23.50.050; and Ch. 23.52,
	transportation concurrency
8820	project review system;
	amends §§ 23.40.020 and
	23.69.020(B), land use and
	zoning (23.22, 23.40,
	23.44, 23.45, 23.46, 23.47,
117207	23.49, 23.50, 23.52, 23.69)
117386	Amends §§ 21.04.430,
	21.04.440 and 21.76.030,
	water rates and charges
117388	(21.04, 21.76) Amends § 18.28.010, fee
11/300	=
117390	schedule (18.28) Repeals § 20.48.070
117390	(Repealer)
117391	Amends §§ 21.28.370,
117371	21.33.030(D), (F) and
	21.76.040(A), drainage and
	wastewater utility (21.33,
	21.76)
117392	Amends § 22.602.050,
	inspections (22.602)
117393	Revises overlegal
	trucking/parking permit fee
	schedule (Not codified)
117395	Amends § 4.13.010,
	exemptions from civil
	service (4.13)
117397	Amends § 3.20.010, garden
	plot fee schedule (3.20)
117398	Amends Ch. 5.32, table of
	historical landmarks (5.32)
	A J. S 17 16 015
117399	Amends § 17.16.015,
117399	Seattle Center Department
117399	
117399	Seattle Center Department
	Seattle Center Department (17.16) Adds §§ 5.48.020(P) and
	Seattle Center Department (17.16) Adds §§ 5.48.020(P) and
	Seattle Center Department (17.16) Adds §§ 5.48.020(P) and 5.48.250-5.48.270, taxation
	Seattle Center Department (17.16) Adds §§ 5.48.020(P) and 5.48.250-5.48.270, taxation of cellular telephone
117401	Seattle Center Department (17.16) Adds §§ 5.48.020(P) and 5.48.250-5.48.270, taxation of cellular telephone industry (5.48)

reference only.

d amending
graphics,
accuracy of

For current Sincy the Office of the City Clerk Seattle Municipal Code ference only. 117403 Amends §§ 20.32.020 and 20.32.050, Seattle Arts Commission (20.32) Text provide 117405 Adds new Subtitle IX to Title 22, consisting of new Chapters amending 22.901A-22.901T. renumbers former Subtitle IX to be Subtitle X, and repeals Ch. 22.900, fees for See ordinan Department of Construction and Land Use (Repealed by 119255) sections for 117407 Adds § 3.18.070, amends §§ 3.18.010, 3.18.030, 3.18.111(E), 20.44.010(F), and tables and 20.44.030, 20.44.040(B), 20.44.060, 20.44.090, 20.44.100(C) and (F), this source file. 20.46.050(D), 20.46.060, 20.46.095(A) and (B), 20.46.100, 20.46A.050(J), 20.46A.060, 20.46A.070(D) and (E), 20.46A.300(B) and 21.60.660(B) and (C), and repeals Ch. 20.44, Subchapter II, §§ 20.44.130-20.44.160, women's and minority business utilization (3.18, 20.44, 20.46, 20.46A, 21.60) 117408 Adds §§ 3.14.500-3.14.540 and Subchapter VI, §§ 3.14.800-3.14.830; amends title of Subchapter III of Ch. 3.14; amends §§ 3.40.050, 3.90.050, 3.96.030-3.96.050, 3.116.030, 4.94.030-4.94.070, 5.44.200, 5.48.120(B) and 10.02.080; repeals §§ 3.14.515, 3.14.560, 3.14.570, 3.14.700, 3.14.710, 3.14.720, 3.14.740. 3.37.010-3.37.050 and the introductory material, "The City Budget," in Title 5, executive branch of city

> For current SMC, contact the Office of the City Clerk

117410

government (3.14, 3.40, 3.90, 3.116, 4.94, 5.44,

Amends § 23.84.040, land use and zoning (23.84)

5.48, 10.02)

Seattle Municipal Code ference only. June 2009 co 117411 Amends §§ 23.47.004, 23.47.010 and 23.47.011; rezone, Plats 35W, page for Text provide 100, and 48E, page 125; land use and zoning (23.47)amending 117418 Adds Ch. 4.77, drug-free workplace and drug and alcohol testing (4.77) 117424 Adds § 4.20.065, See ordinan administrative reassignment (4.20) 117425 Amends §§ 6.82.020(E), sections for 6.82.030(D), 6.82.060, 6.82.100, 6.82.110 and 6.82.150, Department of and tables Construction and Land Use (6.82)this source file.

Seattle Municipal Code 117430 Adds §§ 23.24.045, 23.34.007, 23.34.009, Text provided for hi 23.34.012, 23.34.075, 23.34.077, 23.34.079, 23.34.093, 23.34.100, 23.34.102, 23.34.104, 23.34.106, 23.34.108, 23.34.110, 23.34.112, 23.34.114, 23.34.116, See ordinances 23.34.118, 23.34.120, 23.34.122, 23.34.124, Ch. 23.43, §§ 23.44.036(F), sections for comp 23.45.106(G), 23.49.009, 23.49.042(C), 23.49.060(C), and tables and 23.49.116(C), 23.49.142(C), 23.49.318(C), this source file. 23.49.336(C), 23.50.027, 23.66.120(C), 23.76.019, 23.76.049 and Ch. 23.80; amends §§ 23.24.040(A), 23.30.010, 23.34.008, 23.34.010, 23.34.070, 23.34.072, 23.34.074, 23.34.076, 23.34.078, 23.34.080, 23.34.082, 23.34.090, 23.34.092, 23.34.094(A)-(C), 23.41.004, 23.45.006(I), 23.45.010(A), 23.45.014(A), 23.47.002, 23.47.004, 23.47.004(E),

eference only. amending

For current

23.34.084, 23.47.013 and lerk

23.47.006(B), 23.47.008, 23.47.009, 23.47.010, 23.47.012(A), 23.47.014(A) and (B), 23.47.016(E), 23.47.023,

23.47.028(A), 23.49.050, 23.49.070, 23.49.126, 23.49.152, 23.49.300, 23.49.330, 23.50.008, 23.50.010(A), 23.50.012, 23.50.028, 23.84.010, 23.84.030, 23.84.036, 23.84.038, 23.84.040; renumbers and amends § 23.34.012 to be 23.34.011 and 23.68.006 to be 23.34.098; renumbers and amends § 23.34.013 to be §§ 23.34.013-23.34.028; renumbers § 23.76.010(F) to be (G), adds a new (F) and amends (E); runumbers § 23.76.040(E) to be (F), adds a new (E) and amends (D); repeals §§ 23.34.006,

Ch. 23.72, land use and

23.34, 23.41, 23.43, 23.44, 23.45, 23.47, 23.49, 23.50, 23.66, 23.68, 23.76, 23.80,

zoning (23.24, 23.30,

23.84)

23.47.024(A),

Seattle Municipal Code 117432 Amends §§ 21.16.030, 21.16.040, 21.16.070, Text provided for hi 21.16.260, 21.16.350, 21.24.021(D) and (G), 22.800.080(A), 22.801.050, 22.801.170, 22.802.012(A), (D) and (F), 22.802.013(A) and

sections for and tables and to this source file.

See ordinances crea

21.16.150(A), 21.16.200, (B), 22.802.015, 22.802.020, 22.802.060, 22.802.070(A), 22.802.090(C) and (E), 22.804.030(A), 22.808.020(B) and (C), 22.808.030, 22.808.040, 22.808.050, 22.808.070, 22.808.080(E) and (F), 22.808.090(B), 22.808.110, 22.808.120, 22.808.130, 23.22.016(A), 23.47.006(B), 23.47.028(C), 23.47.044(C), 23.49.018, 23.53.015(D), 23.53.020, 23.53.030(F) and (G), 23.54.030(D) and (F), 23.54.035(C), 23.71.016(B) and 23.71.018(D) and (G), and repeals §21.24.060, development permits (21.16, 22.800, 22.801, 22.802, 22.804, 22.808, 23.22, 23.47, 23.49, 23.53, 23.54, 23.71)

eference only. amending

Seattle Municipal Code ference only. 117434 Rezone, Plats 219W, page 3; 219E, page 4; 221E, Text provided for hi page 8; 225W, page 13; 225E, page 14; 226W, page 15; 226E, page 16; 227E, page 18; 230E, page 23, amending 231W, page 24; 231E, page 25; 232W, page 26; 232E, page 27; 233W, page 28; 233E, page 29; 2E, page See ordinan 37; 3W, page 38; 3E, page 39; 4W, page 40; 5W, page 42; 5E, page 43; 11W, sections for page 53; 11E, page 54; 12W, page 55; 13E, page 58; 14W, page 59; 14E, and tables and page 60; 15W, page 61; 15E, page 62; 21E, page 74; 22W, page 75; 22E, this source file. page 76; 23W, page 77; 23E, page 78; 24W, page 79; 24E, page 80; 28E, page 88; 29E, page 90; 30W, page 91; 34W, page 98; 34E, page 99; 35W, page 100; 35E, page 101; 36W, page 102; 36E, page 103; 37W, page 104; 37E, page 105, 40W, page 110; 40E, page 111; 41W, page 112; 41E, page 113; 44W, page 116; 44E, page 117; 45W, page 118; 45E, page 119; 48W, page 124; 48E, page 125; 51E, page 131; 52W, page 132; 52E, page 133; 54W, page 136; 54E, page 137; 55W, page 138; 58E, page 145; 59W, page 146; 61W, page 149; 61E, page 150; 65E, page 152; 66W, page 159; 66E, page 160; 68W, page 163; 68E, page 164; 73W, page 173;

> For current SMC, contact the Office of the City Clerk

117437

117438

73E page 174; 76W, page 179; 76E, page 180; 77W, page 181; 78W, page 183; 78E, page 184; 80E, page 188; 81W, page 188-a; 306W, page 191 (Special)

Rezone, Plats 234W, page 30; 29E, page 87; 29W, page 89; 28W, page 90

Amends §§ 5.44.130(A) and 5.44.140(C), business and occupation tax (5.44)

(Special)

June 2009 co

Text provide

See ordinan sections for and tables a this source

apar co	0210
a seeda	te file
117441	Adds §§ 21.36.922,
	21.36.924, 21.36.975,
s cost hi	21.36.980 and 21.36.044,
u ioi	amends §§ 21.36.018,
	21.36.042, 21.36.050,
	21.36.420, 21.36.425,
	21.36.430, 21.36.440,
	21.36.920 and 21.36.970,
	and repeals §§ 21.36.022, 21.36.930-21.36.960, solid
nes crea	waste utility and litter
	control (21.36)
117479	Repeals § 4.20.500
11/4/9	(Repealer)
117483	Amends § 4.04.230,
117463	discipline and discharge
ad to co	(4.04)
117484	Amends § 4.04.030,
11/404	personnel (4.04)
117490	Adds § 21.49.065, amends
117430	§§ 21.49.010, 21.49.020,
	21.49.030, 21.49.040,
	21.49.042, 21.49.052,
	21.49.055, 21.49.057,
	21.49.058, 21.49.060,
	21.49.080-21.49.160, and
	repeals § 21.49.024,
	electricity (21.49)
117492	Adds §§ 22.202.060(G)
11, 1,52	and 22.901P.030(B),
	renumbers previous
	22.901P.030(B) to be (C),
	and amends §§ 22.900.080,
	22.900.150(C),
	22.901A.050 and
	22.901B.100, rental
	housing registration
	program (Repealed by
	119255)
117499	Amends § 11.18.010,
	arterial street (11.18
	footnote)
117507	Amends Ch. 25.32, table of
	historical landmarks
	(25.32)
117512	Adds § 6.208.115, for-hire
	drivers (Repealed by
	118341)
117514	Adds Ch. 23.73 and
	amends § 23.47.004, land
	use and zoning; rezone,
	Plat 40W, p. 110; Plat 40E,
	p. 111 (23.47, 23.73)
11555	1 00 00 77 010

ference only.

amending

raphics,

curacy of

For current SMC, contact the Office of the City Clerk

117555

Amends §§ 23.55.012,

23.60.180(B), 23.66.160(D) and 23.66.338(F), and repeals Ch. 2.24, land use and zoning (23.55, 23.60,

23.66)

Seattle Municipal Code date eference only. June 2009 co 117569 Adds §§ 15.02.015, 15.02.027, 15.02.035, Text provided for hi 15.02.060, 15.02.070, 15.02.080, 15.02.085, 15.02.090, 15.02.100, 15.02.110, 15.04.015, amending 15.04.017, 15.04.037, 15.04.100, 15.04.110, 15.08.005, 15.17.005, See ordinances crea 15.17.100, 15.17.150, 15.17.200, 15.22.085, 15.38.040-15.38.070, 15.44.003, 15.44.006, two sections for subchapter headings for Ch. 15.44, §§ 15.46.040, and tables and to 15.46.050, 15.50.023, 15.50.025, 15.50.028, 15.50.060-15.50.090, 15.60.015, 15.64.010(C), this source file. 15.64.065(C), 15.64.100, 15.76.050 and 15.76.060; amends §§ 15.02.050, 15.04.010, 15.04.050(A), 15.04.060, 15.04.090, 15.06.010, 15.06.020, 15.06.040, 15.06.050, 15.08.010, 15.08.050-15.08.070, 15.08.110, 15.10.010, 15.10.020, 15.10.040, 15.10.150, 15.12.010,

15.12.020,

June 2009 code <del>upus</del> Text provide<mark>d for hi</mark>

See ordinances crea sections for comple and tables and to comple this source file.

15.12.030, 15.12.040(B), 15.14.010(B), 15.14.020, 15.14.030-15.14.050, 15.14.070-15.14.090, 15.16.020-15.16.040, 15.16.070, 15.18.010, 15.20.020, 15.20.010, 15.20.020, 15.20.040, 15.20.050, 15.20.070, 15.20.080, 15.20.010

15.20.080, 15.22.010, 15.22.030-15.22.080, 15.22.100, 15.22.110, 15.24.010, 15.24.020, 15.28.010-15.28.070,

15.28.010-15.28.070, 15.32.010, 15.35.010, 15.36.010, 15.40.010, 15.40.030-15.40.050,

15.44.010-15.44.140, 15.44.160, 15.44.170, 15.46.010-15.46.030, 15.48.010, 15.48.020,

15.50.010, 15.46.020, 15.50.010, 15.50.020, 15.52.040, 15.62.090(A), 15.64.040(E), 15.64.080, 15.76.010, 15.76.020 and

15.76.010, 15.76.020 and 15.90.020; subdivides and amends §§ 15.02.040 (to be 15.02.040, 15.02.042, 15.02.044, 15.02.046 and

15.02.044, 15.02.046 and 15.02.048), 15.04.020 (to be 15.04.020 and 15.04.025), 15.04.030 (to

15.04.025), 15.04.030 (to be 15.04.030 and 15.04.035), 15.04.040 (to be 15.04.040, 15.04.042,

5.04.044 and 15.04.045), 15.04.070 (to be 15.04.074), 15.04.072 and 15.04.074),

15.17.010 (to be 15.17.010, 15.17.020 and 15.17.050), 15.22.020 (to be 15.22.020,

15.22.022, 15.22.024 and 15.22.026) 15.22.120 (to be 15.22.120, 15.22.122

and 15.22.125), 15.26.010 (to be 15.26.010 and 15.26.020), 15.32.020 (to

be 15.32.020-15.32.040 and 15.32.060-15.32.150), 15.38.010 (to be 15.38.010,

15.38.020 and 15.38.030) and 15.50.040 (to be

15.50.040 and 15.50.045); renumbers and amends §§ 15.04.050(B) and (C) to be 15.04.042(C) and (D),

15.20.060 to be 15.24.030, 15.30.040 to be 15.04.012,

15.30.040 to be 15.04.012, 15.30.050 to be 15.32.050, 15.48.030 to be

12A.14.083, 15.74.020 to be 15.06.060 and

15.74.030 to be 15.06.070; renumbers and amends § 15.32.030 to be 15.32.300 and adds a new §

and adds a new § 15.32.030; renames Chs. 15.10, 15.17, 15.18, 15.22, 15.28, 15.32, 15.48 and ference only.

l amending graphics, ccuracy of

act lerk

For current SM the Office of the

June 2009 co 117570 Text provided for hi

See ordinances sections for comp and tables and this source file.

Amends § 22.206.160(C), intro. to land use code, § 23.02.020, Ch. 23.04, §§ 23.24.040(A), 23.30.020, 23.32.006, 23.40.020, 23.42.020, 23.42.042, 23.45.182(G), 23.49.036(E), 23.49.037(B), 23.50.027(A) and (B), 23.53.030(G), 23.56.010, 23.76.010(E)-(G), 23.76.016(B), 23.76.028(C), 23.76.036, 23.76.040, 23.76.068, 23.84.014, 23.84.036 and 23.90.002; repeals §§ 23.04.030, 23.34.006 and Title 24, land use, zoning and housing (22.206, 23.02, 23.04, 23.24, 23.30, 23.32, 23.40, 23.42, 23.49, 23.50, 23.53, 23.76, 23.84, 23.90) 117571 Amends §§ 23.60.014, 23.60.156(C), 23.60.368(A) and 23.60.633, land use and zoning (23.60) 117573 Amends Seattle Building Code (22.100.010 footnote) 117577 Rezone, Plat 36W, page 102 (Special) 117578 Rezone, Plat 52W, page 132 (Special) 117579 Rezone, Plat 232W, page 26 (Special) 117580 Rezone, Plat 232W, page 26 (Special) 117583 Amends §§ 3.114.150 and 3.114.160, consultant selection (3.114) 117585 Amends § 6.42.080, panoram licenses (6.42) 117586 Amends §§ 6.202.280 and 6.202.310, and reenacts and ratifies § 6.202.230, adult entertainment licenses (6.202) 117598 Adds §§ 23.47.007 and 23.50.015, and amends Exh. 23.76.004A and §§ 23.76.006, 23.76.026(B),

ference only. amending

For current SMC, contact the Office of the City Clerk

117613

23.76.032 and 23.84.025, land use and zoning (23.47, 23.50, 23.76, 23.84)

Amends § 4.13.010(A) and (I), exemptions from

classified service (4.13)

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	
a anda	te me
117615	Amends §§
	14.04.140-14.04.160,
a com hi	14.04.180, 14.08.160 and
d tor III	14.08.180, fair
	employment practices and
	open housing and public
	accommodations (14.04,
	14.08)
117620	Adds § 3.35.040(D) and
- co orea	renumbers former (D) to be
ces cr	(E), and amends §
	3.35.030(C), director
100016	assistance service (3.35)
117625	Extends to 6/30/96
	moratorium on permits for
4 4 - 01	adult entertainment
and to ce	premises (Ch. 6.270
	footnote)
117627	Amends § 15.62.110, street
Callo	vacation revenues (15.62)
117629	Amends §§ 2.14.010,
11/029	2.14.020, 2.14.030(B),
	2.14.020, 2.14.030(B), 2.14.040 and
	2.14.040 and 2.14.060-2.14.100, election
117641	pamphlet (2.14)
117641	Amends § 5.06.030, funds
117642	within city treasury (5.06)
117642	Amends § 11.56.020,
115545	traffic code (11.56)
117645	Adds § 18.12.245 and
	amends §§ 18.12.040,
	18.12.235 and 18.12.277,
	operating hours for parks
	(18.12)
117655	Adds §§
	4.64.015-4.64.017, defense
	and indemnity of city
	officers and engineers
	(4.64)
117663	Amends § 18.28.100,
	municipal golf facilities
	subaccount (18.28)
117673	Amends §§ 12A.06.120,
	12A.06.130, 12A.06.155,
	12A.06.160, 12A.06.165,
	12A.06.170, 12A.06.175,
	12A.06.180 and
	12A.06.185, criminal code
	(12A.06)
117691	Amends § 17.16.015,
	Seattle Center Department
	(17.16)
117693	Amends Seattle Building
	Code (Repealed by
	117721)
	1 ( )

reference only.

de lamending lamending graphics, ord ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

a anda	te file
117697	Adds § 22.808.025;
	amends §§ 22.801.170,
	22.802.015(C),
a for m	22.804.030(A) and (C), and
U IOI	22.808.150; repeals Ch.
	-
	22.806, stormwater,
	grading and drainage
	control (22.801, 22.802,
	22.804, 22.808)
117698	Amends § 22.700.010,
LES CI CO	Washington State Energy
	code (22.700)
117702	
117703	Amends § 21.60.060 and
COMPLE	repeals § 21.60.070,
	telecommunications
	(21.60)
117713	Amends Ch. 5.76, table of
	funds (Repealed by
	118385)
117721	Repeals and replaces §
222-	22.100.010 and amends
	Uniform Building Code
	(Repealed by 119079)
117722	Repeals and replaces §
11//22	22.400.010 and amends
	Uniform Mechanical Code
	(22.400)
117723	Adds a new Subtitle IVA
	to Title 22, A new Ch.
	22.450 and a new §
	22.450.010, Seattle Boiler
	and Pressure Vessel Code
	(22.450)
117728	Amends § 4.13.010(P),
	security measures (4.13)
117730	Adds § 6.288.010(J) and
	amends §§ 6.288.040 and
	6.288.065, pawnbroker
115521	endorsements (6.288)
117731	Amends § 21.04.465,
	Seattle Water Department
	(21.04)
117734	Amends §§ 11.14.023 and
	11.56.020, traffic code
	(11.14, 11.56)
117740	
117748	Amends § 4.13.010(H), (J),
	(L) and (Z), exemptions
	from classified service
	(4.13)
117750	Amends § 4.04.200(A),
	training (4.04)
117770	Amends § 21.36.025, solid
117760	
117760	waste utility (21.36)
117760 117763	waste utility (21.36) Suspends obligation to pay
	Suspends obligation to pay
	Suspends obligation to pay rental housing registration
	Suspends obligation to pay rental housing registration fees set forth in §
	Suspends obligation to pay rental housing registration

reference only.

d amending
graphics,
accuracy of

June 2009 co

Text provided for

See ordinan 1177895 Com sections for com and tables and tables and this source file.

117776 Repeals §§ 17.16.101, 17.16.103, 17.16.104, 17.16.105, 17.16.107, 17.16.110, 17.16.115, 17.16.120 and 17.16.130, Seattle Center Department (Repealer) 117785 Amends § 16.20.130, speed regulations for watercraft and vessels (16.20) 117789 Amends §§ 22.208.060, 22.808.040(B), 23.60.070, 23.60.072, 23.60.076(E), 23.60.924, 23.60.936, 23.60.944, 23.76.012(B) and (D), 23.76.022(C), 23.76.024(K) and (L), 23.76.056(C) and (D), 23.90.019(G), 25.05.680(D) and (F), 25.05.908(A), 25.09.020(B) and 25.28.310(B), land use and environmental protection (22.208, 22.808, 23.60, 23.76, 25.05, 25.09, 25.28) 117800 Amends § 5.44.140, business and occupation taxes (5.44) 117801 Amends §§ 5.44.026(1), (4)(h) and (9), 5.44.028, 5.44.030, 5.44.076, 5.44.090(C), 5.44.110(B), 5.44.160(A), 5.44.180, 5.44.220, and 5.44.230, and amends and subdivides § 5.44.210(A) and (C), business and occupation taxes (5.44) 117806 Rezone, Plat 29E, page 90 (Special) Amends §§ 21.40.050(A), 117813 21.40.080(A) and 21.76.030(A), solid waste utility (21.40, 21.76) 117844 Amends subsection (D). (I), (M) and (AA) of § 4.13.010, employee positions (4.13) 117852 Amends §§ 22.804.160(A) and 22.804.180(C), and renumbers § 22.808.020 to be 22.808.025, Stormwater, Grading and Drainage Control Code (22.804, 22.808) 117859 Amends § 17.16.015, Seattle Center Department (17.16)

eference only.

I amending graphics, ccuracy of

Seattle Municipal Code

June 2009 co 117861 up date

Amend 22.204, 22.208, 22.

See ordinant sections for and tables a this source

osp.	te file
117861	Amends §§ 22.204.080, 22.204.170, 22.206.200,
103	22.208.010, 22.208.020,
a for m	22.208.030, 22.208.040,
30 102	22.208.050, 22.208.070,
	22.208.080, 22.208.090,
	22.208.100, 22.208.110,
	22.208.150 and
	22.208.160, Housing and
01023	Building Management
ces ci ca	Code (22.204, 22.206,
	22.208)
117863	Amends 1994 Seattle
COMPA	Mechanical Code
	(22.400.010 footnote)
117864	Amends §§
and to C	6.230.020-6.230.100
18 and 18	[amendment to § 6.230.030
	not codified] and
<b>C</b> 112	6.230.120-6.230.160,
Pile	regulation and licensing of
	steam engineers and boiler
	firemen (6.230)
117865	Amends Seattle Building
	Code (Repealed by
	119079)
117869	Adds § 23.54.020(I),
	amends Chart A for §
	23.54.015, and amends §
	23.84.018, zoning (23.54,
	23.84)
117887	Amends § 18.28.010, Parks
	and Recreation Department
117004	(18.28.010)
117904	Street utility residential and
	business charges (Repealed
117007	by 120794)
117907	Amends § 4.13.010,
	exemptions from civil
	service (4.13)

eference only.

I amending graphics, ccuracy of

Seattle Municipal Code

Seattle Municipal Code

Adds § 2

June 2009 co 117908 UP
Text provided for

See ordinances cr sections for comp and tables and to this source file.

117913

117914

117917

117919

Adds § 22.901L.050 and new Table 22; amends §§ 22.901A.010, 22.901A.050. 22.901B.010, 22.901B.020, 22.901B.040, 22.901B.050 and Table 1, 22.901B.100(B), (C) and Table 2, 22.901C.010 and Tables 3 and 4, Chapter 22.901D and Table 5, §§ 22.901E.010 and Table 6, 22.901H.010 and Table 15, 22.901J.050(A) and (C), 22.901M.010(A) (and renumbers Table 22 to be 23), 22.901P.010 (and renumbers Table 23 to be 24), 22.901P.030(A) (and renumbers Table 24 to be 25), 22.901Q.010 (and renumbers Table 25 to be 26), 22.901R.010 and 22.901S.010(A) (and renumbers Table 26 to be 27); repeals Table 20A of § 22.901K.010 (and

renumbers and amends
Table 20B to be 20), Table
21A of § 22.901K.020 (and
renumbers and amends
Table 21B to be 21) and
22.901P.030(C), fees
(Repealed by 119255)

Amends § 17.19.010, Seattle Center Department

Adds § 5.44.024(2) (and renumbers (2) and (3) to be

5.44.110(K), and amends §§ 5.44.024, 5.44.090(L) and 5.44.110(J), business and occupation taxes (5.44)

Adds §§ 7.04.085 and 7.04.645, and amends §§ 7.04.080 and 7.04.650, weights and measures

Amends § 23.47.010(A),

land use and zoning; rezone, pages 3, 4, 8, 13, 14, 18, 23-27, 37-40, 42, 43, 53-55, 58-60, 74-78, 88, 90, 91, 99-102, 104, 105, 112, 113, 117-119, 124, 125, 131-133, 136-138, 145, 146, 149, 150, 158-160, 163, 164, 173, 174, 179-181, 183,

(3) and (4)) and

(17.19)

(7.04)

eference only.

l amending

graphics,

ccuracy of

For current SN (23.47)

For Current SN (23.47)

The Office of the City Clerk

June 2009 co Text provide

117923 Amends § 11.18.010, Exh. 11.18.010 A and § 11.62.060, Traffic Code (11.18, 11.62)Adds §§ 23.12.025 and

23.12.050-23.12.150, amends §§ 23.34.008(J),

23.34.124(D), 23.69.002, 23.76.050(A), 23.84.024, 25.05.675(G) and (J), and

repeals §§

See ordinances 23.12.010-23.12.040, 23.16.002, 23.16.006, 23.16.020, 23.16.050,

sections for comp 23.16.060 and Resos. 25968, 26684, 27320,

and tables and 27353, 27768, 28894, 26579, 26763, 27261, 27320, 27353, 27708, this source file.

27801, 27886, 28071, 28110, 28111, 28167, 27156, 27769, 27341,

27186, 27705, 25713, 27618, 28682, 28846,

28484, 28401, 28500, 28081 and 28191, land use

policies (23.12, 23.16, 23.34, 23.69, 23.76, 23.84, 25.05)

117942 Amends §§ 22.204.200 and 22.206.160(C), just cause eviction (22.204, 22.206)

117943 Amends § 23.41.012, land use and zoning (23.41) 117945 Amends §§ 25.09.020(A),

25.09.180(D) and 25.09.300(D), environmentally critical

areas (25.09) 117954 Amends §§ 23.49.033(E), 23.49.048(A) and (B), 23.49.052, 23.49.068(A) and (B), 23.49.072,

23.49.098(A), 23.49.102(B), (C) and (D), 23.49.128 and 23.84.025,

land use and zoning (23.49, 23.84)

117958 Amends § 4.24.035, paid sick leave (4.24) 117959 Adds § 4.04.235 and

amends § 4.04.230, employee discipline (4.04) 117961 Amends § 4.20.300, payment for performance

of out-of-class duties (4.20) 117977 Amends §§ 3.35.050, 4.40.020, 4.44.060,

5.24.010-5.24.030, Ch. 5.80.020, funds (3.35 4.40, 4.44, 5.24 5.76 5.20

4.44, 5.24, 5.76, 5.80) the Office of the Cit

ference only. amending

## Seattle Municipal Code ference only. June 2009 co 117978 Adds §§ 4.36.193 and 4.36.385, and amends § 4.36.040(C), employee retirement system (4.36) 118003 Adds § 11.16.170 and repeals § 11.16.160, **Engineering Department** (11.16)118006 Amends § 21.60.060, telecommunications and See ordinan technology (21.60) sections for complete and tables and to confirm accuracy this source file.

June 2009 co 118012 Text provided for hi

See ordinances sections for comple and tables and to this source file.

Adds §§ 23.66.018, 23.66.115(C) and (D), 23.76.004(G), 23.76.005, 23.76.015, 23.76.023, 25.12.375, 25.12.535, 25.12.845, 25.16.115,

25.20.115, 25.22.135, 25.24.015 and 25.24.085; amends Table 6 of § 22.901E.010, §§ 23.22.016, 23.22.020, 23.22.052,

23.22.054, 23.22.064, 23.22.070, 23.22.072, 23.22.074, 23.22.078, 23.24.010(A), 23.41.004, 23.41.006, 23.41.008(B)

and (C), 23.41.010, 23.41.012(B), 23.49.037(B) and (C), 23.66.030, 23.66.132,

23.66.316, 23.76.002, 23.76.004(A), (B) and (F), 23.76.006(B)-(D),

23.76.008(A), 23.76.010(A), (C)-(G), 23.76.012(A)-(D),

23.76.014, 23.76.016, 23.76.019, 23.76.020(A)

and (C), 23.76.022(A) and (C), 23.76.024, 23.76.026,

23.76.028, 23.76.032(A), 23.76.036(A), 23.76.038,

23.76.040(B)-(F), 23.76.046(A), 23.76.049,

23.76.050(A) and (C),

23.76.052, 23.76.054(B), (D) and (E), 23.76.056(A),

(C) and (D), 23.76.058(B) and (D), 23.76.060(C)-(E),

23.76.066, 23.84.002,

23.84.008, 23.84.016, 23.84.025, 23.84.030,

23.88.020, 25.05.310(C),

25.05.665(E),

25.05.680(A)-(C),

25.05.714(A), 25.12.160, 25.12.180, 25.12.210,

25.12.280, 25.12.300,

25.12.310, 25.12.320,

25.12.340, 25.12.360,

25.12.370, 25.12.380, 25.12.390, 25.12.400,

25.12.410, 25.12.420,

25.12.440, 25.12.450, 25.12.490, 25.12.500,

25.12.520, 25.12.530,

25.12.540, 25.12.560, 25.12.600, 25.12.610,

25.12.620, 25.12.630,

25.12.640, 25.12.650, 25.12.660, 25.12.670,

25.12.680, 25.12.690,

25.12.700, 25.12.720,

25.12.730, 25.12.740, 25.12.760, 25.12.770,

lerk

25.12.835, 25.12.840, 25.12.860, 25.12.900,

25.12.910, 25.16.080, 25.16.090, 25.16.100, ference only. amending

For current the Office of th

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
118013	Amends § 4.72.020, meals
	at City events (4.72)
118014	Amends § 4.13.010,
d for III	exemptions from classified
	service (4.13)
118015	Rezone, Plat 221E, page 8
	(Special)
118019	Repeals and replaces Ch.
	22.600, Seattle Fire Code
acc crea	(Repealed by 119124)
118022	Amends §§ 5.44.110(F),
10	5.44.140(C), 5.44.180 and
comple	5.44.200, business and
110045	occupation taxes (5.44)
118045	Amends Ch. 25.32, table of
ad to co	historical landmarks
118046	(Special)
118040	Amends Ch. 25.32, table of historical landmarks
C:10	
118047	(Special) Amends Ch. 25.32, table of
11004/	historical landmarks
	(Special)
118048	Amends Ch. 25.32, table of
110040	historical landmarks
	(Special)
118049	Adds § 6.230.030, amends
110019	§§ 6.230.060(A),
	6.230.070, 6.230.110 and
	6.230.160(C), and repeals §
	6.230.030, steam engineers
	and boiler firemen (6.230)
118050	Rezone, Plat 35E, page 101
	(Special)
118066	Amends § 3.114.160(B),
	consultant roster program
110002	(Repealed by 119651)
118082	Adds Ch. 10.07, graffiti
118087	nuisance code (10.07) Amends § 20.48.030,
110067	
	public works contracting procedures (20.48)
118088	Adds § 20.46A.130(F),
110000	public works contracting
	procedures (20.46A)
118093	Amends § 3.28.620(A),
110070	Police Department
	(Repealed by 120728)
118096	Amends §§ 9.25.050(E)
	and 9.26.020, pets (9.25,
	9.26)
118098	Adds § 18.12.085, pets
	(18.12)
118099	Amends §§ 18.12.030(A)
	and 18.12.080, Parks
	Department (18.12)
118101	Amends Seattle
	Mechanical Code
	(22.400.010 footnote)
	o conti

ed
8
1 amending
graphics,
of ccuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

Cihar	cilo
a - mada	te tille
118102	Extends to 6/30/97
	moratorium on permits for
1 Sor MI	adult entertainment
ad in	premises (Ch. 6.270 footnote)
118105	Amends §§ 11.20.010,
110103	11.23.400, 11.34.020,
	11.56.020 and 11.58.005,
	traffic code (11.20, 11.23,
ass cred	11.34, 11.56, 11.58)
118106	Adds § 12A.10.070 and amends § 12A.08.020,
10	criminal code (12A.08,
comple	12A.10)
118107	Adds § 12A.06.187 and
140 0	amends §§ 12A.06.120,
and to co	12A.06.160, 12A.06.165,
	12A.06.170 and
0210	12A.06.195, domestic
118114	violence (12A.06) Adds § 16.40.040, harbor
110114	code (Repealed by 118570)
118115	Amends § 18.12.235, Parks
	Department (18.12)
118120	Amends § 20.46A.090(B),
	women and minority
	business enterprises
118121	(20.46A) Amends § 4.04.220,
118121	personnel system (4.04)
118138	Adds §§ 5.06.030(F) and
	20.04.145, local
	improvement districts
	(5.06, 20.04)
118141	Adds § 5.40.086 and
	amends § 5.40.085(A) and (B), admission tax 5.40)
118145	Exemptions to Ch. 3.114
	an §§ 5.44.040 and
	5.44.130, neighborhood
	planning office (Ch. 3.14,
110176	Subch. VI footnote)
118176	Adds §§ 21.28.005, 21.28.040(B), 21.28.090,
	21.28.100 and 21.28.400;
	recaptions Ch. 21.28 and
	amends §§
	21.28.010-21.28.040,
	21.28.080, 21.28.200,
	21.28.220, 21.28.250-21.28.290 and
	21.28.250-21.28.290 and 21.28.350; repeals
	severability note to Ch.
	21.28 and §§ 21.28.060,
	21.28.070, 21.28.230,
	21.28.240, 21.28.300,
	21.28.360 and 21.28.370,
	wastewater rates and charges (21.28)
	Charges (21.28)

d amending graphics, accuracy of

Text provided for

See ordinan sections for and tables and this source file.

118181 Adds §§ 23.66.318(C) and (D), 25.16.080(C), a new 25.20.080(B) and a new 25.22.100(B); amends §§ 23.66.030(C), 23.66.030(E), 23.76.012(D), 23.76.024(C) and (J), 23.76.032(A), 23.76.056, 23.88.020(A), 25.05.680(A), 25.12.120, 25.12.210, 25.12.430, 25.12.680(E), 25.12.690, 25.16.080(A) and (B), 25.16.100(A) and (B), 25.20.080(A)-(C), 25.22.100(A)-(D) and the Seattle Building Code; renumbers §§ 25.20.080(B) and (C) to be (C) and (D), and 25.22.100(B)-(D) to be (C)-(E), various provisions (23.66, 23.76, 23.88, 25.05, 25.12, 25.16, 25.20, 25.22) 118182 Adds § 17.04.060, drug-free zone (17.04) 118201 Amends § 18.28.010, Parks and Recreation Department (18.28)118208 Adds § 3.20.230, leases of property for certain open space uses; repeals § 1 of Ord. 105688 (3.20) Amends Ch. 25.32, table of 118235 historical landmarks (25.32)118236 Amends Ch. 25.32, table of historical landmarks (25.32)118237 Amends Ch. 25.32, table of historical landmarks (25.32)118238 Adopts master filming permit fees; amends §§ 11.23.120, 15.04.074, 15.35.010, 21.04.530, 22.602.040(C) and 22.901Q.010; repeals film fees schedule adopted by Ord. 116176, film permits (11.23, 15.04, 15.35, 21.04, 22.602) 118240 Rezone, Plat 220E, page 6 (Special) 118265 Amends § 4.34.055 and repeals § 4.20.360, vacation balances (4.20, 4.34)118266 Amends § 4.20.060, credit

ference only.

the Office of the City Clerk

for unpaid absences (4.20)

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	C:10
a - mada	te tile
118271	Amends Seattle Building
	Code (Repealed by
110000	119079)
118278	Sets carpool parking fees
	for certain lots; adds §
	11.16.310(C); amends §§ 11.14.085 and
	11.14.003 and 11.16.310(A), carpools
	(11.14, 11.16)
118279	Amends § 21.49.058, high
ces croa	demand general service
	(21.49)
118294	Amends §§ 25.05.675(M)
COMPA	and 25.05.800(A),
	environmental policies and
1400	exemptions (25.05)
118302	Adds §§ 23.34.126,
	23.34.128 and Ch. 23.48;
0210	amends §§ 23.30.010,
111C.	23.32.016, 23.41.004(A),
	23.41.012(B), 23.47.036(B) and (D),
	23.47.038(E),
	23.53.010(B),
	23.53.015(A),
	23.53.030(B)-(D),
	23.54.015(I),
	23.54.020(D)-(F) and (H),
	23.54.030(F), 23.55.030
	and 23.84.009 [23.84.012],
	Seattle Cascade mixed
	zone (23.30, 23.32, 23.34, 23.41, 23.47, 23.48, 23.53,
	23.54, 23.55, 23.84)
118307	Authorizes sale of TDRs
11000,	and renames Housing
	Mitigation Account to be
	TDR/Mitigation Subfund;
	appropriation for
	development and
	preservation of low-income
110212	housing (5.76)
118313	Adds §§ 5.52.010,
	5.52.020, 5.52.050 and 5.52.090-5.52.150; amends
	§§ 5.52.010-5.52.060;
	renumbers former §§
	5.52.010 to be 5.52.030,
	5.52.020 to be 5.52.040,
	5.52.030 to be 5.52.060,
	5.52.040 to be 5.52.070,
	5.52.050 to be 5.52.080,
	and 5.52.060 to be
	5.52.160 (5.52)

d amending graphics, ecuracy of

Seattle Municipal Code ference only. date 118314 Adds Subchapters I, II and III to Ch. 5.44, and §§ Text provided for hi 5.44.050, 5.44.070, 5.44.310, 5.44.320, 5.44.340, 5.44.350 and 5.44.480; amends §§ amending 5.44.010-5.44.030, 5.44.040, 5.44.050-5.44.060, 5.44.068-5.44.110, See ordinances 5.44.130-5.44.160, 5.44.180, 5.44.190-5.44.210, sections for 5.44.230-5.44.310; renumbers former §§ and tables and to 5.44.030 to be 5.44.400, 5.44.040 to be 5.44.030, 5.44.050 to be 5.44.410, 5.44.052 to be 5.44.412, this source file. 5.44.054 to be 5.44.414, 5.44.056 to be 5.44.416, 5.44.058 to be 5.44.418, 5.44.060 to be 5.44.440, 5.44.068 to be 5.44.444, 5.44.070 to be 5.44.420, 5.44.072 to be 5.44.422, 5.44.074 to be 5.44.424, 5.44.075 to be 5.44.426. 5.44.076 to be 5.44.428. 5.44.078 to be 5.44.430, 5.44.080 to be 5.44.442, 5.44.090 to be 5.44.450, 5.44.100 to be 5.44.460, 5.44.110 to be 5.44.470, 5.44.130 to be 5.44.300, 5.44.140 to be 5.44.040, 5.44.150 to be 5.44.060, 5.44.160 to be 5.44.080, 5.44.180 to be 5.44.090, 5.44.190 to be 5.44.110. 5.44.205 to be 5.44.210,

> 5.44.210 to be 5.44.100, 5.44.230 to be 5.44.120, 5.44.240 to be 5.44.130, 5.44.250 to be 5.44.140, 5.44.260 to be 5.44.150, 5.44.270 to be 5.44.160, 5.44.280 to be 5.44.170, 5.44.290 to be 5.44.180, 5.44.300 to be 5.44.330, and 5.44.310 to be 5.44.170, 5.44.220 and 5.44.320, business licenses

and business and occupation taxes (5.44)

June 2009 co Text provide

See ordinant sections for and tables a this source

	te file
118315	Adds §§ 5.48.095,
(1) C. C. C. C. C. C. C. C. C. C. C. C. C.	5.48.135 and 5.48.220;
103	amends §§
a sor m	5.48.010-5.48.030,
in roz	5.48.050-5.48.090,
	5.48.100-5.48.120,
	5.48.140-5.48.170,
	1
	5.48.200 and 5.48.210;
	repeals §§ 5.48.130,
pos crea	5.48.190 and 5.48.250,
CES OF	occupational utility taxes
	(5.48)
118321	Amends Ch. 4.100,
COMPA	dependent care assistance
	program (4.100)
118324	Amends § 4.14.150(A),
MA TO CO	moving expenses (4.14)
118327	Amends §§ 1, 3, 6 and 8 of
	Ord. 115911, §§ 7 and 8 of
C310.	Ord. 117151, and §
THE.	4.38.208, transfers within
	an investment selection in
	the City's deferred
	compensation plan
	(Repealed by 119159)
118337	Amends §§ 4.04.250(F)
116557	and 4.08.040(F), dates of
	* * * *
	runoff elections (4.04,
110220	4.08)
118338	Adds §§ 3.18.840 and
	20.46A.060(D); amends §§
	3.18.030, 3.18.800,
	3.18.806(A), 3.18.810 and
	3.18.812; repeals §§
	3.18.101, 3.18.111,
	3.18.121 and
	20.46A.130(E), contracting
	(3.18, 20.46A)
118339	Adds Ch. 18.30,
	mechanisms to abate
	unauthorized uses of park
	property (18.30)
118341	Adds Ch. 6.310, taxicabs
	and for-hire vehicles;
	repeals Chs. 6.208, 6.211
	and 6.212 (6.310)
118342	Amends § 11.60.587,
110342	
	penalties for violations of
	maximum gross vehicle
	weight (11.60)

reference only.

d amending
graphics,
accuracy of

June 2009 co 118362 up Text provided for

See ordinances comsections for comand tables and tables and this source file.

118362 Amends §§ 23.04.040(A)-(D), 23.12.120, 23.41.004(A), 23.41.012(A), 23.45.004(A), 23.47.004, 23.54.016(C), 23.54.020(A), 23.69.008, 23.69.012(B), Ch. 23.69 Subchapter V heading, 23.69.021(D), 23.69.022, 23.69.023, 23.69.026, 23.69.028, 23.69.030, 23.69.032, 23.69.034(H) and (I), 23.69.035, 23.69.036(D) and Plat 40W, p. 110 of zoning maps; repeals §§ 23.04.040(E)-(G), 23.69.033(C) and 23.69.036(B) and (C), land use and zoning (23.04, 23.12, 23.41, 23.45, 23.47, 23.54, 23.69) 118369 Adds § 15.04.112; amends §§ 11.16.120, 11.16.125, 11.16.317, 15.04.037, 15.04.045, 15.14.050, 15.14.080, 15.22.026, 15.28.060, 15.44.130, 15.50.023, 15.50.025, Ch. 15.64, 15.64.030, 15.64.060, 15.64.065, 15.64.070 and 16.60.030: repeals Ch. 15.90, Seattle **Engineering Department** (11.16, 15.04, 15.14. 15.22, 15.28, 15.44, 15.50, 15.64, 16.60) 118375 Amends § 18.28.010; and repeals § 18.28.020(C), Parks and Recreation Department fee schedule (18.28)118378 Amends general terms and conditions and facility license fee schedules for nonprofit performing arts events at the Seattle Center, and § 17.16.015,

eference only.

l amending

graphics,

ccuracy of

For current Six Clerk the Office of the City Clerk

118380

118385

118390

use fees, terms and conditions (17.16)

(5.76)

Amends §§ 21.28.040(B) and 21.33.030(D) and (F), drainage and wastewater rates (21.28, 21.33)

Amends Ch. 5.76, table of funds; repeals Ord. 117713

Amends § 17.19.010, Seattle Center parking Seattle Municipal Code ference only. 118391 Amends §§ 22.602.010 and 22.602.040, Fire Code (22.602)Text provide 118392 Abolishes Department of Human Rights and Office for Women's Rights, transfers powers, duties and resources to Office for Civil Rights and transfers certain responsibilities in See ordinan the areas of domestic violence and sexual assault sections for comp to Department of Housing and Human Services; amends §§ 3.14.900, 3.14.910, 3.14.915(A) and and tables and (B), 3.14.918, 3.14.920, 3.14.930, 3.14.940, 3.14.970, 3.22.040, this source file. 3.22.050, 3.22.080, 3.22.250, 3.110.260, 4.13.010(M), 4.20.810(C), 4.20.850(A), 6.212.042(B), 14.04.020(A), 14.04.030(D), (E) and (L), 14.04.050(C), 14.04.060, 14.04.080(A), 14.04.090, 14.04.100, 14.04.110(D), 14.04.120(A), 14.04.130, 14.08.010, 14.08.020(D)-(F), (Q) and (EE), 14.08.110, 14.08.120(B). 14.08.130(D), 14.08.140(A), 14.08.150, 20.46.070 and 20.46A.060(A); renumbers former §§ 3.22.080 to be 3.14.931, 3.14.970 to be 3.14.932, 3.22.040 to be 3.14.911, 3.22.050 to be 3.14.912 and 3.22.250 to be 3.14.913; repeals §§ 3.14.950, 3.14.960, 3.22.010, 3.22.020,

For current SMC, contact
the Office of the City Clerk

118393

118394

3.22.030, 3.22.060, 3.22.070, 3.114.170(F), 20.46.060(D) and 20.46A.070, Office for Civil Rights (3.14, 3.22, 3.110, 4.13, 4.20, 6.212, 14.04, 14.08, 20.46,

Amends §§ 21.04.430 and 21.04.440, water rates

Amends § 4.13.010, exemptions from civil

20.46A)

(21.04)

Seattle Municipal Code eference only. 118395 Amends §§ 5.44.300, 6.08.010, 6.20.050, d for hi 6.36.020, 6.36.030, Text provide 6.36.040, 6.42.030, 6.48.020, 6.102.030, 6.204.030, 6.204.040, amending 6.204.070, 6.204.080, 6.270.060, 6.294.105, 7.04.570, 9.26.025, 9.26.040, 9.26.050, See ordinan 9.26.060 and 22.904.040, business licenses and regulatory and personal sections for licenses (5.44, 6.08, 6.20, 6.36, 6.42, 6.48, 6.102, 6.204, 6.270, 7.04, 9.26, and tables a 22.904) this source file.

Seattle Municipal Code ference only. 118396 Abolishes Water Department and transfers Text provided for hi powers, duties and resources to Seattle Public Utilities, transfers certain functions from Engineering amending Department and City Light Department, and continues Water Fund, Drainage and Wastewater Fund, Solid See ordinan Waste Fund and Engineering Services Fund; amends Ch. 3.32 and §§ sections for 3.121.010, 5.78.080, Ch. 10.06, Ch. 10.46. 10.76.040, 10.76.060, and tables and 15.46.040, 20.16.070, 21.04.010, 21.04.020, 21.04.050, 21.04.060, this source file. 21.04.070, 21.04.080, 21.04.090, 21.04.100, 21.04.110, 21.04.120, 21.04.160, 21.04.170, 21.04.180, 21.04.190, 21.04.200, 21.04.210, 21.04.230, 21.04.240, 21.04.260, 21.04.270, 21.04.330, 21.04.340, 21.04.350, 21.04.360, 21.04.370, 21.04.380, 21.04.390, 21.04.400, 21.04.410, 21.04.420, 21.04.450, 21.04.465, 21.04.470, 21.04.480,

21.04.490, 21.04.500, 21.04.500, 21.04.505, 21.04.510, 21.04.530, 21.04.540, 21.04.570, 21.04.590, 21.04.600, 21.08.010, 21.08.030, 21.08.040, 21.08.070, 21.08.120, 21.08.130, 21.08.170, 21.08.190, 21.08.210, 21.08.220, 21.08.230, 21.08.240, 21.08.250,

Seattle Municipal Code eference only. date 21.08.270, 21.08.280, 21.08.290, 21.08.320, Text provided for hi 21.08.330, 21.08.340, 21.08.350, 21.08.360, 21.08.370, 21.12.010, 21.12.020, 21.12.050, amending 21.12.060, 21.16.030, 21.16.040, 21.16.050, 21.16.060, 21.16.065, See ordinances crea 21.16.070, 21.16.080, 21.16.090, 21.16.100, 21.16.110, 21.16.140, sections for comple 21.16.150, 21.16.160, 21.16.170, 21.16.180, 21.16.210, 21.16.220, and tables and to 21.16.250, 21.16.260, 21.16.270, 21.16.280, 21.16.300, 21.16.310, 21.16.330, 21.16.340, this source file. 21.16.350, 21.16.360, 21.16.370, 21.20.010, 21.20.020, 21.24.010, 21.24.021, 21.24.030, 21.24.040, 21.24.070, 21.24.090, 21.24.100, 21.28.005, 21.28.010, 21.28.020, 21.28.040, 21.28.080, 21.28.090, 21.28.100, 21.28.200, 21.28.220, 21.28.260, 21.28.280, 21.28.350, 21.33.010, 21.33.080, 21.36.012, 21.36.014, 21.36.016, 21.36.018, 21.36.026, 21.36.028, 21.36.029, 21.36.030, 21.36.040, 21.36.060, 21.36.080, 21.36.087, 21.36.112, 21.36.113, 21.36.114, 21.36.420, 21.36.425, 21.36.430, 21.36.922, 21.36.975, 21.36.980, 21.40.030, 21.40.050, 21.40.060, 21.40.085, 21.40.090, 21.40.120, 21.40.130, 21.76.030, 21.76.040, 21.76.050, 22.204.020, 22.204.080, 22.206.200, 22.208.120, 22.800.080, 22.801.020, 22.801.050, 22.801.200, 22.802.012, 22.802.013, 22.802.015, 22.802.040, 22.802.060, 22.802.070, 22.802.090, 22.808.020, 22.808.030, 22.808.040, 22.808.050, 22.808.070, 22.808.080, 22.808.110, 22.808.150, 22.900.030, 25.06.050, 25.06.110 and 25.09.120, Seattle Public Utilities For current (3.32, 3.121, 5.78, 10.06,10.46, 10.76, 15.46, 20.16, the Office of th 21.04, 21.08, 21.12, 21.16,

21.20, 21.24, 21.28, 21.33, 21.36, 21.40, 21.76, 22.204, 22.206, 22.208, 22.800, 22.801, 22.802,

Seattle Municipal Code June 2009 co 118397 Text provided for hi

See ordinances crea sections for comple and tables and to this source file.

Abolishes Department of Administrative Services and Personnel Department and transfers powers, duties and resources to **Executive Services** Department, designates Finance Division of **Executive Services** Department as successor to Finance Department, and confirms incumbents as Finance Director and Personnel Director: adds Ch. 3.24 and § 21.60.030(O); renames Ch. 3.18 and amends §§ 3.18.060, 3.18.070, 3.18.140, 3.18.160, 3.18.180, 3.18.260, 3.18.280, 3.18.800, 3.18.802, 3.18.806, 3.18.808, 3.18.810, 3.18.812, 3.18.814, 3.18.816, 3.18.818, 3.18.820, 3.18.822, 3.18.824, 3.18.826, 3.18.830, 3.18.832, 3.18.840, 3.18.900, 3.18.902(1), (7), (12) and (19), 3.18.904(A) and (D)-(G), 3.18.906, 3.18.908, 3.18.910, 3.18.912, 3.18.914, 3.18.918, 3.18.920, 3.18.924, 3.35.040, renames Ch. 3.38,

3.38.010, 3.38.020, 3.76.010, 3.114.020, 3.114.040, 3.114.050, 3.114.060, 3.114.090, 3.114.100, 3.114.110, 3.114.120, 3.114.140, 3.114.150, 3.114.160, 4.04.030(9), 4.04.040(A), 4.04.050, 4.04.200, 4.04.225(B)(1), 4.04.250(L), 4.04.255, 4.10.040, 4.20.140, 4.44.070, 4.68.020, 4.68.030, 4.68.100, 4.68.110, 4.68.120, 4.68.130, 4.84.040, 4.84.050, 4.88.010, 4.88.020, 4.92.020(C), 4.100.020, 5.14.060.

ference only. amending

Seattle Municipal Code date 5.32.030(A)(3), 5.44.022, 5.48.020(A)(6) and (7), Text provided for hi 5.52.010(C) and (D), 5.78.130, 5.78.150, 6.02.030, 6.202.060, 6.222.010, 6.270.030, 6.294.040, 6.310.110, 7.04.130, 7.04.645, 7.04.650(A), 9.25.020(F)

See ordinances crea sections for comple and tables and to this source file.

and (G), 10.02.090, 10.08.165(A), 11.30.280, 11.30.290, 15.32.300(B)(3), 15.38.030(A), 16.60.030(C), 20.44.010(F), 20.44.030, 20.44.040, 20.44.060, 20.44.090, 20.44.100, 20.46.050(D), 20.46.060(D), 20.46.095, 20.46.100, 20.46A.300, 20.48.020, 20.48.030, 20.48.040, 20.48.060, 20.84.030, 21.52.100, 21.52.230(C), 21.52.250, 21.60.040, 21.60.230, 21.60.330, 21.60.360, 21.60.380, 21.60.660(B) and (C) and 22.900.245(A); renumbers former § 3.18.070 to be 3.24.070; repeals §§ 3.18.010, 3.18.020, 3.18.030, 3.18.040, 3.18.050, 3.38.030, 20.52.010, 20.52.020, 20.52.030, 20.52.040, 20.52.050 and 20.52.060, **Executive Services** Department (3.18, 3.35, 3.38, 3.76, 3.114, 4.04, 4.10, 4.20, 4.44, 4.68, 4.88, 4.100, 5.14, 5.32, 5.44, 5.48, 5.52, 5.78, 6.02, 6.202, 6.222, 6.270, 6.310, 7.04, 9.25, 10.02, 10.08, 11.30, 15.32, 15.38, 16.60,

> 20.44, 20.46, 20.46A, 20.48, 20.84, 21.52, 21.60)

ference only. amending

Seattle Municipal Code ference only. Aate June 2009 co 118398 Adds § 22.901T.020; amends §§ 6.82.130, Text provided for hi 6.230.070, 22.901A.030, 22.901B.010. 22.901B.020(A), 22.901B.030, 22.901B.040, amending 22.901B.050(A), 22.901B.080(B), 22.901B.100, 22.901C.010, 22.901D.010, Ch. 22.901E, See ordinances 22.901F.010, 22.901G.020(A), sections for comp 22.901H.010 (Tables 14 and 15), 22.901J.010, 22.901J.020, 22.901J.060, Ch. 22.901N, and tables and 22.901N.010, 22.901P.030 and 22.901S.010(A); repeals §§ 6.82.140, this source file. 6.204.090, 6.230.090 and 22.901N.010(E), fees and related provisions for City permits, inspections, examinations and licenses (6.82, 6.230)118401 Amends § 11.18.010(A), Traffic Code (11.18) 118404 Amends § 4.04.045, temporary employment service (4.04) 118406 Amends § 3.14.520, Sand Point Naval Station (Repealed by 118912) 118408 Adds § 23.34.007(F);

amends §§ 23.34.007(D), 23.34.008(F) and (G), 23.60.002(B), 23.60.004, 23.60.012, 23.60.060(B), 23.60.74, 23.60.220, 23.60.724(C), 23.60.936 and 25.05.675(G) and (J); repeals § 23.12.130, Seattle shoreline master program (23.12, 23.34, 23.60,

25.05)

Seattle Municipal Code ference only. June 2009 co 118409 Abolishes Engineering Department and transfers Text provided for hi powers, duties and resources to Seattle Transportation Department; amends Ch. amending 3.12 and §§ 3.80.010, 5.78.040, 7.16.010, 11.16.020, 11.16.120, See ordinances crea 11.16.122, 11.16.125, 11.16.170, 11.16.180, 11.16.200, 11.16.220, sections for comp 11.16.240, 11.16.280, 11.16.300, 11.16.310, 11.16.315, 11.16.317, and tables and to 11.16.319, 11.16.320, 11.16.340, 11.16.360, 11.16.380, 11.16.400, 11.16.420, 11.66.020, this source file. 11.66.180, 11.66.200, 15.02.042, 15.02.044, 15.02.085, 15.04.010, 15.04.015, 15.04.020, 15.04.072, 15.04.074, 15.06.010, 15.06.050, 15.06.060, 15.06.070, [15.06.080], 15.10.020, 15.10.030, 15.12.010, 15.12.040, 15.14.050,

15.14.070,

Seattle Municipal Code date file eference only. June 2009 co 15.14.080, 15.16.040, 15.16.060, 15.16.070, Text provided for hi 15.16.080, 15.17.150, 15.17.200, 15.18.010, 15.18.020, 15.20.010, 15.22.020, 15.22.022, amending 15.22.026, 15.22.030, 15.22.080, 15.22.122, 15.24.020, 15.24.030, See ordinances crea 15.26.010, 15.28.010, 15.28.020, 15.28.030, 15.28.050, 15.28.060, sections for comple 15.28.070, 15.28.080, 15.32.010, 15.32.030, 15.32.300, 15.35.010, and tables and to 15.36.010, 15.38.010, 15.38.040, 15.38.050, 15.40.030, 15.42.030, 15.42.050, 15.42.060, this source file. 15.44.003, 15.44.090, 15.44.130, 15.44.140, 15.44.150, 15.46.010, 15.46.020, 15.48.120, 15.50.010, 15.50.023, 15.50.025, 15.50.028, 15.50.060, 15.50.080, 15.52.020, 15.52.040, 15.60.015, 15.62.030, 15.62.100, 15.62.110, 15.62.120, 15.64.040, 15.64.050, 15.64.060, 15.64.070, 15.64.080, 15.70.030, 15.72.010, 15.72.020, 15.72.040, 15.76.010, 15.76.020, 15.76.050, 15.76.060, 15.90.010, 15.90.020, 15.90.030, 16.08.050, 16.32.050, 16.36.040, 16.60.010, 16.60.020, 16.60.030, 16.60.040, 16.60.050, 20.04.030, 20.04.060, 20.04.070, 20.04.080, 20.04.090, 20.04.280, 20.12.080, 20.76.280, 20.84.030, 21.100.010, 21.100.020, 21.100.030,

Seattle Municipal Code ference only. date June 2009 co 21.100.050, 21.100.060, 21.101.020, 21.101.060, Text provided for hi 21.101.070, 21.101.090, 21.101.100, 22.806.020, 22.900.030, 22.900.070, 22.900.245, 22.900.265, amending 23.12.060, 23.22.024, 23.22.028, 23.22.064, 23.22.066, 23.22.068, See ordinances crea 23.22.070, 23.22.072, 23.22.074, 23.22.094, 23.22.096, 23.22.100, sections for comple 23.22.102, 23.24.040, 23.44.016, 23.45.015, 23.45.057, 23.45.073, and tables and to 23.45.112, 23.47.016, 23.49.018, 23.49.020, 23.49.022, 23.49.056, 23.49.076, 23.49.106, this source file. 23.49.134, 23.49.162, 23.49.332, 23.50.016, 23.50.034, 23.50.046, 23.53.010, 23.53.015, 23.53.020, 23.53.030, 23.54.015, 23.54.016, 23.54.030, 23.54.035, 23.55.012, 23.60.196, 23.60.902, 23.66.030, 23.66.170, 23.68.022, 23.69.032, 23.71.008, 23.71.010, 23.71.012, 23.71.018, 23.76.024, 23.84.036, 23.90.004, 25.02.030, 25.05.675 and 25.08.520; repeals § 10.04.010, Seattle Transportation Department (3.12, 3.80, 5.78, 11.16,11.66, 15.02, 15.04, 15.06, 15.10, 15.12, 15.14, 15.16, 15.17, 15.18, 15.20, 15.22, 15.24, 15.26, 15.28, 15.32, 15.35, 15.36, 15.38, 15.40, 15.42, 15.44, 15.46, 15.48, 15.50, 15.52, 15.60, 15.62, 15.64, 15.70, 15.72, 15.76, 15.90, 16.08, 16.32, 16.36, 16.60, 20.04, 20.12, 20.76, 20.84, 21.100, 22.806, 23.12, 23.22, 23.24, 23.44, 23.45, 23.47, 23.49, 23.50, 23.53, 23.54, 23.55, 23.60, 23.66, 23.69, 23.71, 23.76, 23.84, 23.90, 25.02, 25.05, 25.08) 118411 Amends §§ 5.32.020, 5.32.030, 5.32.050, 5.32.080, 5.32.110, 5.32.130, 5.32.160, 5.32.170 and 5.32.200, Revenue License Code

For current on the City Clerk the Office of the City Clerk

Seattle Municipal Code eference only. Adds Map B to § 118414 23.54.015, Ch. 23.59, Text provided for hi Exhibit 2 to § 25.05.675; amends §§ 23.12.060, 23.24.040, 23.24.045, 23.34.094(B), 23.40.006(A), 23.44.010(D), 23.44.012(B), 23.44.014(D), See ordinan 23.44.016(A) and (C), 23.44.017(B) and (D), sections for comp 23.44.022(K), 23.44.032, 23.44.051, 23.45.002(C), 23.45.006(A) and (F), 23.45.010(A) and (C), and tables and 23.45.014(G), 23.45.016, 23.45.018(B), 23.45.056(D), this source file. 23.45.058(B), 23.45.060(B), 23.45.072(D), 23.45.076(B), 23.45.092, 23.45.112(A), 23.46.002(E), 23.47.008, 23.47.009, 23.47.014(E), 23.47.016(A), 23.47.023(A), 23.47.024, 23.47.040(C), 23.53.015(D), 23.53.025(F), 23.53.030(E) and (G). 23.54.015 (Chart A). 23.54.030(F), 23.55.034(C) and (D), 23.57.001, 23.66.025(A), 23.67.020, 23.68.002, 23.69.004, 23.71.004, 23.71.008(D), 23.71.038, 23.73.004, 23.73.008(C), 23.84.004, 23.84.014, 23.84.030, 23.86.010(B), 23.86.014(C),

> 23.86.016(A), 23.86.018(H), 23.90.006(B),

23.44.044(E),

23.90, 25.05)

23.90.020(B); repeals §§

23.45.048(B), Ch. 23.56, § 23.66.122(C)-(F), Ch. 23.70, Map B and Exhibit 2 of Ord. 116168, various zoning and environmental protection provisions (23.12, 23.24, 23.34, 23.40, 23.44, 23.45, 23.46, 23.47, 23.53, 23.54, 23.55, 23.57, 23.59, 23.66, 23.67, 23.69, 23.71, 23.73, 23.84, 23.86,

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipas			
a - mada	te me		
118415	Amends §§ 23.60.198(B),		
	23.60.544 and		
1 50° M	23.60.666(B), zoning (23.60)		
118416	Amends § 4.72.020(A),		
110410	invoices and receipts for		
	food expenses incurred in		
	travel outside the City		
	(4.72)		
118433	Amends § 21.76.040(A),		
Ces es	water, drainage and		
10	wastewater credits for		
comple	qualified low-income		
118436	customers (21.76) Amends § 22.600.020 and		
118430	Ord. 118018, Uniform Fire		
and to co	Code (Repealed by		
Willer on	119124)		
118441	Amends §§ 22.206.160(C),		
S112.	22.206.230, 22.206.240		
2770.	and 22.206.280; repeals §		
	22.202.060, Housing and		
	Building Maintenance		
110442	Code (22.206)		
118443	Adds Ch. 6.315,		
	horse-drawn carriages, horses and drivers (6.315)		
118466	Adds §§ 23.76.026(D) and		
110.00	(E) and 25.09.010,		
	moratorium on		
	development in		
	landslide-prone areas		
110.110	(23.76, 25.09)		
118469	Amends § 4.20.030, salary		
	upon initial appointment		
118470	(4.20) Amends § 4.20.300,		
110770	payment for performance		
	of out-of-class duties (4.20)		
118472	Amends §§ 22.901E.010,		
	22.901P.040,		
	23.45.005(D), 23.47.004		
	(Chart A), 23.76.032(A),		
	23.84.008(D),		
	23.84.028(O), 23.90.006(B) and		
	23.90.000(B) and 23.90.019; amends and		
	renumbers § 23.44.025 to		
	be 23.44.041, land use and		
	zoning (23.44, 23.45,		
	23.47, 23.84, 23.90)		
118474	Amends § 18.28.010, fee		
	schedule for park and		
	recreational facilities and		
	services (18.28)		

reference only.

amending

amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te ille
118475	Amends §§ 21.49.010,
	21.49.020, 21.49.030,
101	21.49.040, 21.49.042,
4 501 111	21.49.052, 21.49.055,
C TO	21.49.057, 21.49.058,
	21.49.060, 21.49.065,
	21.49.080, 21.49.090,
	21.49.100, 21.49.110,
	21.49.120, 21.49.130,
oos crea	21.49.140, 21.49.150 and
CES OF	21.49.160, Seattle City
	Light Department (21.49)
118497	Adds § 4.36.410; amends
COMPA	§§ 4.36.030 and 4.36.260,
	City Employees'
1 4 0 00	Retirement System (4.36)
118498	Amends § 4.36.400 and
The co	repeals § 4.36.390, City
	Employees' Retirement
0210	System (4.36)
118499	Reclassifies and retitles
118499	
	certain positions,
	establishes new titles and
	salaries, corrects certain
	salaries, restores certain
	positions and amends §
	4.13.010, Civil Service
	Systems (4.13)
118503	Establishes Transportation
	Bond Fund and amends
	Ch. 5.76, Table of Funds
	(5.76)
118507	Adds § 17.16.200; repeals
110307	Ch. 17.40 and Ords.
	112577 and 115759.
	Bumbershoot Festival
110520	(17.16)
118529	Amends §§ 5.44.100 and
	5.44.120, business and
	occupation taxes (5.44)
118530	Adds § 3.18.150, review of
	Contracting Services
	Division recommendations
	for contract awards by the
	Hearing Examiner (3.18)
118533	Amends § 3.14.540, work
	program to be adopted by
	City Council (Repealed by
110520	120773)
118539	Adds §§ 25.09.345,
	25.09.350, 25.09.352,
	25.09.355 and 25.09.356;
	amends § 23.76.026,
	regulations for
	environmentally critical
	areas (23.76, 25.09)
118540	T Amends o ZI 49 una amer
118540	Amends § 21.49.065, duct, vault and pole rental rates
118540	vault and pole rental rates (21.49)

d amending graphics, accuracy of

Seattle Municipal Code 118546 Establishes P-Patch Gardenship Account, Text provided for accepts certain conservation easements and transfers jurisdiction of certain real property; adds § 5.78.180; amends § 3.35.060; amends and renumbers § 3.20.230 to be 3.35.080; and repeals § See ordinan 3.20.220, Department of Neighborhoods (3.35, 5.76, 5.78) sections for 118553 Adopts and amends National Electrical Code, 1996 Edition, and amends and tables a Seattle Building Code, 1994 Edition; amends § 22.300.010, Seattle Electrical Code (22.300, this source 22.300.015 footnote) Adopts new titles and salaries, reclassifies and retitles certain positions, corrects certain titles and amends § 4.13.010, Civil Service Systems (4.13) 118566 Amends § 23.22.106, land use and zoning (23.22) 118569 Adds §§ 2.04.010(29) and (30), 2.04.155, 2.04.156, 2.04.265, 2.04.275, 2.04.340 and 2.04.500(D); amends §§ 2.04.010(7), (8), (13) and (21), 2.04.075(B), 2.04.160(A), 2.04.170(B)-(D), 2.04.180, 2.04.215, 2.04.230(D), 2.04.250(A)-(D), 2.04.260(A), 2.04.270(A), 2.04.330(A)-(C), 2.04.370(B) and (D), 2.04.375(B) and

ference only.

For current SMC, contact the Office of the City Clerk

118570

118572

2.04.500(A); and repeals § 2.04.310, Elections Code

Amends § 16.40.030 and repeals § 16.40.040, Harbor Code (16.40)

Implements and regulates pilot compensation program in Executive Department, Office of Management and Planning, and retitles certain positions; amends §§ 4.13.010 and 4.20.040, personnel (4.13, 4.20)

(2.04)

June 2009 co

Text provide

See ordinan sections for and tables a this source

	te file
118574	Adopts and amends
	Uniform Fire Code, 1994
- 101	Edition, and amends §
a sor m	22.600.020, Seattle Fire
G TO	Code (Repealed by
	119124)
118580	Amends § 3.33.020, Seattle
110300	Municipal Court (3.33)
118582	Extends to 6/30/98
110302	moratorium on permits for
CES CITES	adult entertainment
	premises (Ch. 6.270
10	footnote)
118605	
118003	Adds § 21.04.280 and
	amends §§ 21.04.060(B)
ad to Co	and 21.16.270(B), water
Ind to	and sewer infrastructure
110.605	extensions (21.04, 21.16)
118607	Adds §§ 18.12.025,
<b>\$110.</b>	18.12.278 and 18.12.279;
8880	amends § 18.12.030; and
	repeals § 18.12.277, Parks
	Code (18.12)
118617	Creates Emergency
	Management Sub-Fund in
	the General Trust Fund
	(5.76)
118624	Adds Ch. 23.72 and
	amends §§ 23.32.016,
	Chart A for 23.54.015,
	23.84.018 and 23.84.025,
	Sand Point Overlay District
	(23.32, 23.54, 23.72,
	23.84)
118636	Establishes new fund
	structure for Executive
	Services Department and
	retitles Administrative
	Services Fund to be
	Executive Services Fund
	(5.76)
118638	Amends § 18.12.080(B)
	and Exhibit 18.12.080(E)
	(18.12)
118659	Adds § 6.230.045; amends
	§§ 6.230.030, 6.230.040,
	6.230.060, 6.230.070,
	6.230.140 and 6.230.160,
	Seattle Steam Engineers
	and Boiler Firemen Code
	(6.230)
	(0.230)

reference only.

tle amending graphics, accuracy of

Seattle Municipal Code June 2009 co Text provide See ordinan sections for

esttle Munic	ilhar .	4- 6110
eattle	a seeda	Te III
	118663	Adds §§ 23.60.248,
2009 60		23.60.308, 23.60.370,
111116 2007	- 0 - m hi	23.60.436, 23.60.490,
مهره	4 FOR III	23.60.550, 23.60.612,
ext provide		23.60.734, 23.60.795 and
ext provide		23.60.854; amends §§
6350 8		23.60.090(I),
		23.60.246(H),
		23.60.306(H), 23.60.365,
	- oc crea	23.60.368(H),
Jim 211	ces er	23.60.430(I),
cae oralliass		23.60.544(G),
366 02 0	- 400 10 16	23.60.606(G), 23.60.664,
202	COMPA	23.60.666, 23.60.728(H)
estions for		and 23.60.848(G); repeals
Secritoria	140 00	§ 23.60.366, shoreline
	and to co	district (23.60)
and tables a	118665	Amends § 5.52.030(A) and
and tobe		(B), gambling taxes (5.52)
	118672	Adds § 23.76.058(D);
this source	777	amends §§ 23.41.008(D),
this sum		Ch. 23.44, Part 3 title,
		23.44.036, 23.44.041(D),
		23.45.106, 23.47.004(E),
		23.47.006(C),
		23.49.008(C),
		23.49.042(C),
		23.49.046(D) and (E),
		23.49.060(C),
		23.49.066(D) and (E),
		23.49.090(C),
		23.49.096(B), (E) and (F), 23.49.116(C),
		23.49.122(D) and (E),
		23.49.142(C),
		23.49.148(D) and (E),
		23.49.318(C),
		23.49.324(D) and (E),
		23.49.336(A),
		23.50.012(C) and (D),
		Exhibit 23.76.004(A),
		23.76.011, 23.76.012(B)
		and (C), 23.76.036(B),
		23.76.052(C),
		23.76.062(B), 23.76.064,
		23.78.006, 23.79.006,
		23.84.024, 23.84.025,
		23.84.036 and 25.09.080,
		land use and environmental
		protection (23.41, 23.44,
		23.45, 23.47, 23.49, 23.50,
		23.76, 23.78, 23.79, 23.84,
		25.09)
	118678	Amends §§ 5.10.060 and
		5.10.080, fiscal agents
		(5.10)
	118685	Amends § 4.34.055, use
		and scheduling of
		vacations (4.34)

reference only. d amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipai			
A	te file		
118686	Establishes new titles and		
	salaries, reclassifies and		
1 0 - m hi	retitles certain positions,		
ad for in	changes certain salaries,		
500	corrects certain position		
	numbers and ordinance		
	titles; amends § 4.13.010,		
110.500	exemptions (4.13)		
118693	Establishes certain		
DOS CITE	positions and salaries and		
	provides for salary step		
-10	placement calculations and return to previous		
comple	positions; amends §		
COMPAGE	4.13.010, exemptions		
	(4.13)		
118696	Amends § 21.49.058, high		
HIRON CO	demand general service		
	(21.49)		
118698	Amends § 5.44.450(A),		
1116.	business and occupation		
	tax exemption on gambling		
	receipts (5.44)		
118703	Adds § 4.44.080; amends §		
	4.44.020(B), sick leave;		
	repeals § 4.24.020 (4.44)		
118709	Amends § 4.08.070(C),		
	Public Safety Civil Service		
110712	Commission (4.08)		
118713	Amends §§ 6.42.110 and		
	6.42.130, regulations for panoram facilities (6.42)		
118716	Amends Ch. 25.32, Table		
110/10	of Historical Landmarks		
	(25.32)		
118717	Amends § 4.20.065,		
	allowance for initial		
	administrative		
	reassignment (4.20)		
118720	Amends §§ 23.44.058,		
	23.84.006, 23.84.014 and		
	23.84.025, land use and		
110724	zoning (23.44, 23.84)		
118724	Amends §§ 9.25.084(A)		
	and 18.12.080, animal control (9.25, 18.12)		
118727	Amends § 23.40.020,		
110/2/	review of variances from		
	Land Use Code		
	requirements (23.40)		
118735	Adds § 4.16.030(H);		
	amends §§ 4.16.030(C) and		
	(E), 4.16.090 and 4.16.100,		
	Ethics and Elections		
	Commission (4.16)		
118742	Amends § 3.18.140, city		
	motor vehicle fleet (3.18)		
118750	Amends § 5.44.300(A) and		
	(E), business licenses		
	(5.44)		

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	0.10
a anda	te file
118751	Adds §§ 15.26.030 and
	15.32.160; amends §
1 C - 10 101	15.32050, utility cuts and
A FOL 112	other openings in city
	streets, alleys and other
	public places (15.26,
	15.32)
118762	Amends § 17.16.015,
	Seattle Center schedule of
oc crea	license fees and terms and
ces er	conditions for 1997 and
	1998 (17.16)
118766	Establishes new titles and
COMPA	salaries, reclassifies and
	retitles certain positions,
140 00	changes certain salaries,
and to co	corrects certain title and
	amends § 4.13.010,
	exemptions (4.13)
118769	Suspends temporarily
111c.	operation of § 9.26.020(D),
	amends § 9.25.100(D),
	animal licensing (9.25)
118774	Amends § 22.602.040, fees
	(22.602)
118775	Adds § 4.20.420, strategic
	advisor compensation
	program (Repealed by
	119705)
118776	Amends §§ 4.20.030,
	4.20.040(A), 4.20.080 and
	4.20.320(A)(1),
	compensation and working
	conditions (4.20)
118778	Amends §§ 4.14.110 and
	4.14.140, executive
	recruiting (4.14)
118779	Amends § 4.13.010(A),
	exemptions from the civil
	service and public safety
	civil service systems (4.13)
118782	Adds § 4.20.380,
	accountability pay for
	executives program (4.20)
118783	Adds § 4.20.400, manager
	compensation program
	Compensation program
118786	(Repealed by 119705) Amends § 4.13.010,

reference only.

a amending graphics, securacy of

Seattle Municipal Code eference only. Aate June 2009 co 118793 Adds 23.60.065; amends §§ 23.60.002(B), Text provided for hi 23.60.012, 23.60.016, 23.60.020, 23.60.030, 23.60.034, 23.60.036, 23.60.060(B), amending 23.60.062(B), 23.60.066(B) and (E), 23.60.068(D), 23.60.070, See ordinances crea 23.60.072(A), 23.60.074, 23.60.076(A) and (C)-(E), 23.60.122(D), 23.60.156(A), sections for 23.60.196(A), 23.60.210, 23.60.244, 23.60.304, and tables and to 23.60.365, 23.60.366(B), 23.60.426, 23.60.428(A), 23.60.486, 23.60.540, 23.60.574(B), 23.60.604, this source file. 23.60.608(A), 23.60.664, 23.60.724, 23.60.728(B), 23.60.730(C) and (D), 23.60.784, 23.60.844, 23.60.846, 23.60.902, 23.60.912, 23.60.932, 23.60.934, 23.60.936, 23.60.940, 23.60.944, 23.60.950 and 23.60.952. land use and shorelines

(23.60)

Seattle Municipal Code ference only. date 118794 Adds §§ 23.44.018(F) and 23.45.006(J) and (K); Text provided for hi amends §§ 23.22.028, 23.22.056, 23.22.058, 23.22.100(A), 23.22.106, 23.24.045, 23.24.060, 23.34.016(B), 23.34.018(B), 23.34.020(B), 23.34.022(B), See ordinan 23.34.024(B), 23.34.028(B), 23.40.002, 23.42.010, 23.44.014(D), sections for 23.44.016(D), 23.44.022(D), 23.44.041(A), and tables and 23.45.005(A), 23.45.006(I), 23.45.010(C), 23.45.014(D), (F) and (G), this source file. 23.45.016(B), 23.45.058(B), 23.45.060(B), 23.45.076(B), 23.45.112, 23.45.154, 23.47.004 (Chart A), 23.47.006(B), 23.47.010 (Chart B), 23.47.014(E), 23.47.024(B), 23.50.012 (Chart A), 23.54.020(A), 23.69.030(E), 23.69.032(E), 23.72.004, 23.72.008(A) and (B), 23.72.010(C), 23.76.010(D), 23.76.012(D), 23.76.020(C), 23.76.022(B), 23.76.032(A), 23.84.004, 23.84.008, 23.84.024, 23.84.032, 25.05.680(B), 25.05.908(A) and (C) and 25.09.020(B); repeals and replaces Exhibit 23.50.014(A) and Exhibit 1 of § 25.05.675; repeals §§ 23.45.168 and 25.05.680(C) and removes Greenbelt (GB) overlay designation from official land use map, land use and environmental protection (23.22, 23.24, 23.34, 23.40, 23.42, 23.44, 23.45, 23.47, 23.50, 23.54, 23.69, 23.72, 23.76, 23.84, 25.05) 118803 Amends §§ 21.33.030(D) and (F) and 21.76.040(A), drainage rates (21.33, 21.76)

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a anda	te file
	118804	Amends §§ 3.14.520 and 17.04.030, management of
	a sor m	public works projects (17.04)
٥	118810	Amends § 22.901E.010,
		land use and zoning review
		fees (Repealed by 119255)
	118812	Amends § 21.08.110,
		corrosion prevention
	oc crea	equipment annual
$\chi$	CE2 CI	operating permit fees
		(21.08)
	118813	Amends § 22.504.010,
9	COMPANY	permit fees (22.504)
	118815	Amends § 18.28.010, fee
	- 4 to CO	schedule for the use of park
4	ma to	properties and facilities
	110016	(18.28)
	118816	Amends Table 16 of Ch. 22.901J, permit fees for
	1116.	mechanical equipment
		(Repealed by 119255)
	118822	Adds § 5.44.470(I),
	110022	deductions allowed in
		computing tax or fees
		(5.44)
	118826	Amends § 4.24.200, use of
		sick leave accounts after
		retirement (4.24)
	118829	Amends §§ 4.84.020,
		4.92.010 and 4.94.040,
		awards programs (4.94)
	118830	Amends § 4.04.090,
		progressive discipline
	110021	(4.04)
	118831	Amends § 6.288.065,
		pawnbrokers endorsements (6.288)
	118833	Adds Ch. 20.38 [20.40];
		amends §§ 3.18.812,
		3.24.010, 3.24.040 and
		20.48.010; repeals §
		20.48.020, public works
		projects and purchasing
		(3.18, 20.40, 20.48)
	118834	Adds Ch. 20.38,
		apprenticeship program;
		extends position authority
		for human services analyst
		position and authorizes
		participation in
		construction advisory
		committee (20.38)

eference only.

we saw so a mending amending graphics, accuracy of ark

Seattle Municipal Code ference only. 118839 Adds § 22.210.130(F); amends §§ 22.210.060, Text provided for 22.210.090(A), 22.210.100(C) and (D), 22.210.120, 22.210.140(A) and 22.210.150(A); repeals §§ 22.210.190 and 22.210.200, tenant relocation assistance (22.210)See ordinan 118841 Closes and eliminates neighborhood improvement fund, Seattle sections for Center improvement fund and museum development fund: transfers assets. and tables a liabilities and residual fund balance of museum development fund to general bond interest and this source redemption fund (Repealer) Amends §§ 4.20.040(A) and 4.20.080, new compensation programs for certain exempt positions in the legislative department (4.20)118877 Amends §§ 11.59.090 and 12A.02.140, warrants checks (11.59, 12A.02) 118888 Amends § 23.55.040, special exception criteria for signs; deletes heading of § 23.55.038 (23.55)

> Amends Ch. 10.06, Emergency Control of Drainage Problems, Earth Movement, Mud Flows, Windstorm Damage and other Hazards (10.06)

Amends §§ 3.20.040, 3.20.080, 3.20.100 and 3.20.120, Department of Housing and Human Services (3.20)

118896

118906

Seattle Municipal Code

Seattle Municipal Code

Abolishes Office of Management and Planning and designates Executive Services Director as Budget Director; creates Strategic Planning Office within Executive Department and trace of

See ordinances cr sections for comp and tables and to this source file. Services Director as Budget Director; creates Strategic Planning Office within Executive Department and transfers Office for Education from Department of Housing and Human Services to Strategic Planning Office; provides transitional provisions for appropriations and positions; continues Oversight Committee; abolishes Risk Management Advisory

Oversignt Committee; abolishes Risk Management Advisory Committee; redesignates Subchapter III of Chapter 3.14 as Strategic Planning Office and adds §§ 3.14.501, 3.14.521, 3.24.030(D); amends § 3.14.510, 3.14.530, 3.14.540, 3.14.800, 3.20.300, 3.20.310, 3.24.010, 3.24.030(A),

3.24.040(D) and (L), 3.40.050, 3.76.010, 3.90.050, 3.116.030, 4.13.010, 4.20.050, 4.94.030, 4.94.040, 4.94.050, 4.94.060, 4.94.070, 5.44.200, 5.48.120, 10.02.080, 22.901R.010, 23.44.041(F),

23.69.032(K); renumbers existing §§ 3.14.580 to be 3.24.401, 3.14.590 to be 3.24.402, 3.24.030(D) to be (E), 3.24.040(L)-(S) to be (M)-(T), various provisions relating to organization of

executive branch of City government; repeals §§ 3.14.500, 3.14.520 and Ch. 3.96 (3.14, 3.20, 3.40, 3.76, 3.90, 3.116, 4.13, 4.20, 4.94, 5.44, 5.48, 10.02, 23.44, 23.69)

Amends § 5.24.005(A), claims for damages (5.24)

Repeals Ch. 3.78, salary administration policy committee (Repealer)

Amends § 3.104.010, fees for copies (3.104)

ference only.

amending

raphics,

For current SMC, contact the Office of the City Clerk

118923

118928

118933

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipus and a	te file
118934	Amends §§ 2.08.040,
110,54	2.12.020(A) and
- 101	2.16.010(A), City
14 FOT III	referendum, initiative or
	Charter amendment
	petitions (2.08, 2.12, 2.16)
118955	Amends § 21.68.080, First
	Hill underground utility
	district (21.68)
118956	Adds §§ 4.38.420 and
ces cross	4.38.425; amends §§
	4.38.010,
- 400 40 [6	4.38.204-4.38.208,
COMPA	4.38.301-4.38.303,
	4.38.401-4.38.404,
140 00	4.38.406-4.38.408,
and to co	4.38.504, 4.38.1005,
, Hade	4.38.1007, 4.38.1010 and
	4.38.1011; repeals §
\$110	4.38.404, deferred
7770	compensation plan
	(Repealed by 119159)
118957	Amends §§ 3.40.010 and
	3.92.010, filling appointive
	City offices (3.40, 3.92)
118958	Adds Ch. 4.33, sabbatical
	leave program (4.33)
118966	Adds Ch. 4.86, Seattle
	Works! excellence awards
	program; repeals Chs. 4.84
	and 4.92 (4.86)
118970	Reclassifies and retitles
	certain positions,
	establishes new title and
	salary rate, corrects certain
	titles, approves certain positions for inclusion in
	accountability pay for
	executives, allocates
	certain classified service
	positions to manager
	classification system and
	amends § 4.13.010, exempt
	positions (4.13)
118980	Adds § 23.41.016; amends
	§§ 23.41.004, 23.41.006,
	Exhibit 23.41.006A,
	23.41.008, 23.41.014,
	23.76.011, 23.76.012 and
	23.76.026, land use and
	zoning (23.41, 23.76)
118981	Adds Policy 7 to §
	23.12.120; amends Policy
	6 of § 23.12.120 and §§
	23.69.006 and
	23.69.032(I), land use and
	zoning (23.12, 23.69)
118983	Amends Ch. 25.32, table of
	historical landmarks
	(25.32)
4 GM	C' COlles

reference only.

did amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

cipal cost			
4 - 4440	ate Ille		
118984	Amends §§ 23.44.006,		
	23.45.004 and 23.47.004, cemeteries (23.44, 23.45,		
a sor n	23.47)		
118992	Amends § 11.56.020(N),		
	persons under the influence		
	of intoxicating liquor or		
119000	drugs (11.56) Amends § 4.04.030,		
117000	sabbatical leave (4.04)		
119010	Adds § 12A.06.310;		
	amends §§ 12A.06.030,		
aomin Di	12A.06.035, 12A.06.040,		
Comple	12A.06.090, 12A.06.115, 12A.08.050, 12A.14.071,		
440	12A.14.075, 12A.14.083		
and to	and 12A.24.150; repeals §		
9,88	12A.02.120, criminal code		
0210	(12A.06, 12A.08, 12A.14,		
119011	12A.24) Adds §§ 11.14.532,		
117011	11.22.025, 11.22.070,		
	11.22.090, 11.31.115,		
	11.52.110 and 11.56.130;		
	amends §§ 11.14.670, 11.20.347, 11.23.400,		
	11.27.020, 11.34.020,		
	11.52.100, 11.53.440,		
	11.55.060, 11.55.100,		
	11.55.120, 11.58.195 and		
	11.72.065, traffic code (11.14, 11.20, 11.22, 11.23,		
	11.27, 11.31, 11.34, 11.52,		
	11.53, 11.55, 11.56, 11.58,		
	11.72)		
119016	Creates certain positions		
	relating to City Light and amends § 4.13.010,		
	exemptions from Civil		
	Service System (4.13)		
119027	Amends § 23.44.041(A),		
	parking waiver process for accessory dwelling units		
	(23.44)		
119030	Amends § 4.26.005, family		
	and medical leave (4.26)		
119032	Amends § 3.116.030, The		
	City of Seattle Industrial Development Corporation		
	(3.116)		
119033	Establishes certain titles		
	and pay rates and amends §		
	4.20.300, payment for		
	performance of out-of-class and limited term		
	assignment duties (4.20)		
119056	Amends Ch. 25.02,		
I			

ference only. amending

For current SMC, contact the Office of the City Clerk

commute trip reduction

(25.02)

Seattle Municipal Code ference only. June 2009 co 119079 Amends 1997 Uniform Building Code; repeals and replaces § 22.100.010, Uniform Building Code (22.100)119080 Amends 1997 Uniform amending Mechanical Code; repeals and replaces § 22.400.010, Uniform Mechanical Code (22.400)See ordinan 119081 Adopts 1997 Washington State Energy Code and local amendments, and sections for amends § 22.700.010, energy efficiency and energy conservation and tables a (22.700)Retitles certain exempt positions and amends § 4.20.040(A), time periods this source for automatic salary increases (4.20) 119091 Amends §§ 3.18.160, 3.18.200, 3.24.040(E) and

> 3.24.300, Executive Services Department (3.18)

See ordinances comsections for comand tables and tables and this source file.

25.05.210, 25.05.250, 25.05.300(E), 25.05.355, 25.05.408(H), 25.05.535(H), 25.05.721, 25.05.751, 25.05.775 and 25.05.800(Z); amends §§ 10.52.040(E), 23.12.105 (Policy 1 and 3B), 23.53.020(E), 23.76.004(A), 23.76.006(C), 23.76.012(A) and (C), 23.76.020(B) and (C), 23.76.022(C), 23.76.036(B), 23.76.052(C) and (D), 23.84.010, 25.05.055(B), 25.05.060(E), 25.05.070(B), 25.05.300(C) and (D), 25.05.305(A), 25.05.310, 25.05.315, 25.05.330(A) and (C), 25.05.340(B), 25.05.390(B), 25.05.408(B), 25.05.502(C) and (H), 25.05.508, 25.05.600(C), 25.05.660(A), 25.05.675(C), 25.05.680, 25.05.702, 25.05.728, 25.05.747, 25.05.790, 25.05.800(A), (F) and (I),

25.05.890, 25.05.900(B), 25.05.908, 25.05.912 (title), 25.05.938, 25.05.948(A) and (B), 25.05.960 and 25.05.970; repeals § 25.05.748, State Environmental Policy Act Rules (10.52, 23.12, 23.53, 23.76, 23.84, 25.05)

Establishes certain titles and salaries, corrects certain titles and position numbers, and amends § 4.13.010(E)-(G) and (Q), exempt positions (4.13)

Amends § 18.28.010, Seattle Aquarium and Woodland Park Zoo fees

Amends § 4.20.225, compensation for appellants to Civil Service Commission (4.20)

Amends Ch. 25.32, table of historical landmarks

(18.28)

(25.32)

ference only.

amending

raphics,

curacy of

For current SMC, contact the Office of the City Clerk

119100

119108

119114

119120

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Shor	ci10
	a anda	te ille
	119121	Adds §§ 25.16.065 and
		25.24.055; amends §§
	1 50# M1	22.901T.010(A),
9	dian	23.66.030(C) and (E),
		25.12.080, 25.12.680(E),
		25.12.700, 25.12.740(B), 25.16.080, 25.16.110(A),
		25.20.010(D),
		25.20.080(B) and (D),
	- 0002	25.20.110(B),
	ces crea	25.22.020(C),
-		25.22.100(B) and (D),
		25.22.130(B), 25.24.060
	COMPA	and 25.24.080(B), land use
•		and historic preservation
	140 00	(23.66, 25.12, 25.16, 25.20,
	an tu c	25.22, 25.24)
٠	119124	Repeals and replaces Ch.
		22.600, Seattle Fire Code;
	sile.	repeals Ords. 118018,
	222	118019, 118436 and
	110120	118574 (22.600)
	119128	Amends § 3.33.180,
		supervision of Director of Probation Services (3.33)
	119145	Amends § 3.38.824,
	119143	surplus equipment (3.38)
	119151	Adds § 23.45.004(D) and
	117131	amends § 23.84.025, land
		use (23.45, 23.84)
	119159	Adopts restated deferred
		compensation plan and
		trust for officers and
		employees; adds §
		4.38.010, voluntary
		deferred compensation
		plan; repeals Ords. 12517,
		112516, 113136, 114095,
		114697, 115911, 117151,
		118327 and 118956, and §§ 4.38.010-4.38.1510
		(4.38)
	119167	Establishes certain titles,
	11,10,	salaries and positions,
		corrects certain position
		number, and amends §
		4.13.010(D) and (O),
		exempt positions (4.13)
	119170	Amends § 2.08.010(A),
		submission of initiative
		petitions (2.08)
	119171	Amends §§ 7.24.020,
		7.24.030, 7.24.050 and
		7.24.060, residential rental
	440450	agreements (7.24)
	119178	Amends § 23.47.040,
		general provisions for
		pedestrian-designated
	II.	Zonos (TZ /TI)

eference only.

l amending

graphics,

ccuracy of

For current SMC, contact the Office of the City Clerk

zones (23.47)

June 2009 co

Text provide

See ordinant sections for and tables a this source

Gihar	0210
a seeds	te tille
119180	Adds § 11.30.105 and
	amends §§ 11.30.040,
1 Case hi	11.30.120, 11.30.160,
d ior	11.30.290 and 11.30.320,
	impoundment of vehicles (11.30)
119189	Adds §§ 11.14.183,
117107	11.14.257, 11.14.403,
	11.20.230, 11.56.350 and
- oc crea	11.56.355; amends §§
Ce2 cr	11.34.020, 11.56.020 and
	16.20.110, persons under
comple	the influence of
Comple	intoxicating liquor or any drug (11.14, 11.20, 11.34,
	11.56, 16.20)
119192	Amends §§ 21.16.290,
THE CO	21.16.300, 21.16.310 and
	21.16.380, side sewer
Sile.	operation and maintenance
	(21.16)
119214	Amends § 4.50.010, administration of certain
	benefit programs (4.50)
119217	Adds Wallingford Urban
11)217	Village map to §
	23.47.004(E); amends
	Seattle Comprehensive
	Plan and §§ 23.34.020(B),
	23.34.022(B), 23.47.004
	and 23.47.006(B), zoning (23.34, 23.47)
119218	Amends Official Land Use
11,210	Map and § 23.47.040,
	general provisions for
	pedestrian-designated
	zones (23.47)
119220	Amends §§ 4.36.040,
	4.36.050, 4.36.070 and
	4.36.155, Seattle City Employees' Retirement
	System (4.36)
119230	Amends Seattle
	Comprehensive Plan and §
	23.47.004(E), single
	purpose residential
110225	development (23.47) Amends Official Land Use
119235	Amends Official Land Use Map and amends §
	23.47.040(C), general
	provisions for
	pedestrian-designated
	zones (23.47)
119237	Adds Ch. 5.72; amends §
	3.02.125; and provides for
	annual program review,
	multifamily housing (3.02,
	5.72)

d amending graphics, accuracy of

Seattle Municipal Code ference only. June 2009 co 119238 Adds §§ 23.45.082, 23.47.035 and 23.48.035; Text provided for hi amends §§ 23.45.004(A), 23.47.004(E) and Chart A, 23.49.026(C), 23.50.012 (Chart A), 23.54.015 amending (Chart A), 23.54.030(B), 23.54.035 (Table A) and 23.84.032; and provides for evaluation, assisted living See ordinan facilities (23.45, 23.47, 23.48, 23.49, 23.50, 23.54, 23.84) sections for 119239 Adds § 23.22.062; amends §§ 23.22.100(A), and tables and to 23.24.040(A), 23.24.045(A), 23.44.006(A) and (J), 23.44.010(B), this source file. 23.44.014(D), 23.44.018(E), 23.44.022(E), 23.44.034(B), 23.44.080(D), 23.45.005(A), 23.45.008(B), 23.45.014(F), 23.45.016(B), 23.47.008(C) and (G), 23.47.010 (Chart B), 23.47.016(D), 23.47.023(C), 23.48.002(B). 23.48.014(D), 23.49.054(A), 23.49.074(A), 23.49.104(A), 23.49.132(A), 23.49.160(A), Exhibit 23.53.015A, 23.54.015(B) and Chart A, 23.54.020, 23.55.002(C), 23.55.034(D), 23.55.042(B) and (E), Exhibit 23.55.042A, 23.71.028(B), 23.76.024(A), 23.76.028(A), 23.84.018, 23.84.024, 23.84.026, 23.84.032, 23.84.038, 23.84.048, 23.86.018(G) and 25.09.260(F); repeals § 23.44.072, land use and environmental protection (23.22, 23.24, 23.44, 23.45, 23.47, 23.48, 23.49, 23.53,

> For current SMC, contact the Office of the City Clerk

23.54, 23.55, 23.71, 23.76, 23.84, 23.86, 25.09)

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Clhar		
		te file	
	119240	Amends §§ 23.60.070,	
		23.60.072(A) and (B),	
	- 0 - 10 In I	23.60.076(A) and	
	4 FOR III	23.60.196(B) and (C), land	
=	CI ZO	use and shorelines (23.60)	
	119241	Establishes demonstration	
		program for innovative	
		housing design and adds §	
		23.40.050, demonstration	
	oos crea	program for innovative	
V(		housing design (23.40)	
	119242	Adds § 23.45.009(E);	
	comple	amends §§ 23.12.060	
	Comple	(Policies 2 and 3),	
		23.34.016(A) and (B), 23.44.080(D),	
	and to Co	23.45.006(F)-(K),	
4	III CO	23.45.008(C) and (E),	
		23.45.009(E), 23.45.014(B)	
	C310	and (C), 23.45.016,	
	1116.	23.45.182(C) and (E),	
		23.45.184(C)-(E) and	
		23.86.002(B), land use and	
		zoning (23.12, 23.34,	
		23.45, 23.86)	
	119247	Amends § 4.13.010,	
		exemptions from Civil	
		Service System (4.13)	
	119249	Amends § 18.28.010,	
		Department of Parks and	
	110050	Recreation fees (18.28)	
	119252	Amends § 17.19.010,	
		Seattle Center parking	
	119253	charges (17.19) Amends § 17.16.015,	
	117233	Seattle Center schedule of	
		license fees, terms and	
		conditions (17.16)	
	119255	Adds Chs.	
		22.900A-22.900G, fees and	
		charges for permits and	
		activities of the	
		Department of Design,	
		Construction and Land	
		Use; repeals Chs.22.901A-	
		22.901T (22.900A,	
		22.900B, 22.900C, 22.900D, 22.900E,	
		22.900F, 22.900E, 22.900F, 22.900G)	
	119256	Amends §§ 22.602.040 and	
	117230	22.602.050, Building and	
		Construction Codes	
		(22.602)	
	119257	Amends §§ 5.44.100,	
		5.48.150 and 5.52.060,	
		over and underpayment of	
		certain taxes (5.44, 5.48,	
		5.52)	
	119266	Amends §§ 21.40.080 and	
		21.40.120, solid waste	
	- CM	system (21.40)	
	TOMEST STATE		

reference only.

delanation
all amending
graphics,
accuracy of

For current system (21.40)

The Office of the City Clerk

Seattle Municipal Code

June 2009 co 119267 up date

Adds §
amends
21.04.4
and 21.
services
119268

See ordinant sections for and tables a this source

	te file
119267	Adds § 21.04.470(D) and
0,00	amends §§ 21.04.430,
100	21.04.440, 21.04.465(C)
A FOR III	and 21.76.040, water
C I	services (21.04, 21.76)
119268	Amends §§ 21.28.040(B),
	21.28.260, 21.33.030(D)
	and 21.76.040(A)(1) and
	(2); repeals § 21.33.030(F),
012	drainage and wastewater
ces cro	services (21.28, 21.33,
	21.76)
119269	Amends § 21.76.030,
COLLIBATO	qualifications for utility
	rate assistance (21.76)
119270	Changes name of
and to co	Department of
Pares .	Construction and Land Use
	to be Department of
C110.	Design, Construction and
2110.	Land Use; amends §§
	3.06.010-3.06.050 and
	3.06.080, Department of
	Design, Construction and
	Land Use (3.06)
119271	Amends § 22.210.130(A),
	relocation assistance
	payments (22.210)
119272	Amends § 3.40.010, city
	auditor; establishes Office
	of the City Auditor and
	creates certain exempt
	position (3.40)

eference only.

amending
graphics,
ccuracy of

Seattle Municipal Code 119273 Creates Office of Housing and changes name of Text provided for hi Department of Housing and Human Services to be **Human Services** Department, creates Housing Operating Fund and redesignates Housing and Human Services Operating Fund to be See ordinan Human Services Operating Fund; adds §§ sections for comp 3.14.700-3.14.750 and 5.78.190; amends Subchapter V title of Ch. 3.14, Ch. 3.20 title, and tables and 3.20.010, 3.20.030, 3.20.040, 3.20.080, 3.20.120, 3.20.320(E) this source file. 3.60.040, 3.68.050, 3.68.070, 3.118.010(B), 5.78.060, 21.04.280(B), 21.49.040(B), 21.50.020, 21.52.230(B) and (D), 21.52.250, 21.52.260(E), 21.76.010, 21.76.030(C), 21.76.040(B), 21.76.050, 21.76.070, 22.220.080(E), 22.220.090, 22.220.100(C), 22.220.130(B)-(D) and (F), 23.22.024, 23.49.033, 23.49.050(A) and (D), 23.49.052(D), 23.49.070(A) and (D), 23.49.072(D), 23.49.100(A), 23.49.102(D), 23.49.126(A), 23.49.128(D), 23.49.152(A), 23.49.154(D), 23.49.180(B), 23.49.212(B), 23.49.214(D), 23.49.240(B), 23.84.024 and 23.84.030, various provisions relating to organization of City government (3.14, 3.20,

ference only.

amending

raphics,

curacy of

For current SMC, contact the Office of the City Clerk

119274

3.60, 3.68, 3.118, 5.78, 21.04, 21.49, 21.50, 21.52, 21.76, 22.220, 23.22, 23.49, 23.84)

Amends §§ 3.58.090 and 22.900D.170, Seattle Design Commission (3.58,

22.900D)

June 2009 co

Text provide

See ordinant sections for and tables a this source

119275	Adds § 4.36.215 and
	amends §§ 4.36.030,
10	4.36.040(G), 4.36.050(D)
	and 4.36.155(I), Seattle
	City Employees'
	Retirement System (4.36)
119276	Amends § 4.08.070(C),
11/2/0	Public Safety Civil Service
110200	Commission (4.08)
119280	Amends § 3.24.040(S),
	Executive Services
	Director (Repealed by
	120181)
119283	Adds Ch. 5.22,
	convenience fee for
	electronic payments via
and to G	electronic commerce
	technology (5.22)
119291	Amends §§ 4.36.100 and
11,2,1	
<b>1116</b> .	4.36.110(D), City
880	Employees' Retirement
	System (4.36)
119299	Transfers maintenance and
	management of City's Sand
	Point properties from
	Strategic Planning Office
	to Superintendent of Parks
	and Recreation; adds §
	3.26.041; amends §§
	3.14.501, 3.14.521,
	3.26.040 and 4.13.010,
	City's Sand Point
	-
	properties (3.14, 3.26,
119322	4.13) Amends Official Land Use
119322	
	Map and §§ 23.34.020(B)
	and 23.34.022(B), Seattle
	Comprehensive Plan
	(23.34)
119326	Amends Ch. 22.900C.010,
	land use and zoning review
	fees (22.900C)
119329	Amends §§ 4.13.010(A)
	and (D), 4.20.030 and
	4.20.040, information
	technology professional
	compensation program and
	power marketing
	compensation program
110220	(4.13, 4.20)
119338	Amends § 3.33.150,
	judicial officers (3.33)
119351	Adds § 4.20.440;
	reallocates and establishes
	certain positions, power
	marketing compensation
	program (4.20)
119352	
117334	Adds Seattle Empire
	Laundry Building to table
	of historical landmarks (25.32)

eference only.

all amending graphics, accuracy of

For current Sh (25.32) constant of the Office of the City Clerk

Seattle Municipal Code 119360 Amends §§ 4.36.155 and 4.36.215, Seattle city employees' retirement Text provide system (4.36) Adds additional type of cottage housing developments to be tested, clarifies the height departure for detached accessory dwelling units, See ordinan and amends § 23.40.050, demonstration program for innovative housing design sections for (23.40)Adds §§ 23.50.012(D), 23.50.020(C), and tables and 23.50.022(C) and 23.50.027(D); amends §§ 23.41.012(B), this source file. 23.47.012(G), 23.49.008(C), 23.49.048(C), 23.49.068(C), 23.49.098(C), 23.49.124(C), 23.49.150(C), 23.49.210(C), 23.49.328(C), 23.49.342(C), 23.50.022(B), 23.50.027(A), 23.50.028(E), 23.66.140(C) and 23.66.332(C), land use and zoning amending height and rooftop features provisions in downtown, commercial and industrial zones (23.41, 23.47, 23.49, 23.50, 23.66) 119371 Amends § 5.72.040, project eligibility (5.72) 119373 Reallocates, reclassifies, redesignates and retitles certain positions, establishes certain new titles and salary rates, corrects Ords. 119167 and

119379

119385

ference only.

For current SMC, contact the Office of the City Clerk

119212, and amends § 4.13.010(G), exempt positions (4.13)

Amends § 21.04.440(B), base service charge for water services for certain customers of Seattle Public Utilities with twenty inch water meters (21.04)

Adds § 14.04.185, enforcement by private persons (14.04)

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibar	0210
	te file
119391	Amends §§ 23.55.036(D)
	and 23.84.036, land use
1 Cost hi	and zoning amending sign
d in	provisions in industrial
119394	zones (23.55, 23.84) Adds Ch. 23.70, station
119394	area interim overlay district
	(Expired)
119395	Amends § 15.32.300,
- oc crea	attachments to city-owned
CE2 CT	poles (15.32)
119399	Adopts Downtown Design
comple	Guidelines, 1999; and
COMP	amends §§ 23.41.004, 23.41.010, 23.41.012(B)
	and (C) and 23.41.014(C),
and to co	(E) and (F), design review
Willes on	program threshold
	downtown (23.41)
119402	Adds Subchapter II, cable
33300	customer bill of rights, to
	Ch. 21.60 and establishes
	certain positions in the Executive Services
	Department; amends 1999
	budget of the Executive
	Services Department and §
	21.60.050, office of cable
	communications (21.60)
119403	Amends Seattle
	Comprehensive Plan and §§ 23.32.016, 23.34.020(B)
	and 23.34.022(B), Lowrise
	3 and Lowrise 4 rezone
	criteria (23.32, 23.34)
119411	Adds § 4.20.430,
	information technology
	professional compensation program (4.20)
119418	Adds M.V. Thea Foss to
	table of historical
	landmarks (25.32)
119419	Adds M.V. Malibu to table
	of historical landmarks
110420	(25.32)
119428	Adds Space Needle to table of historical landmarks
	(25.32)
119439	Amends §§ 25.12.350,
	25.12.750 and
	25.12.850(A), landmarks
	preservation (25.12)
119441	Amends § 2.04.060(A),
	executive director of Seattle ethics and elections
	commission (2.04)
119442	Adds § 2.04.165 and
127112	amends §§ 2.04.170 and
	4.16.080(A), Seattle ethics
	and elections commission
	(2.04, 4.16)

eference only.

amending
graphics,
accuracy of

For current (2.04, 4.16)

The Office of the City Clerk

Seattle Municipal Code ference only. June 2009 co 119470 Adds Camlin Hotel to table of historical landmarks (25.32)Text provide 119473 Adds §§ 23.91.002-23.91.026; amends §§ 23.40.002 and amending 23.84.028, enforcement of the land use code (23.40. 23.84, 23.91) 119478 Adds § 500 to Seattle See ordinan Boiler and Pressure Vessel Code and amends §§ 100, 160, 170, 190, 220, 230, sections for 290, 310, 320, 360, 370, 380, 390, 400, 410 and 420; amends § 22.450.010, and tables a adoption of Seattle Boiler and Pressure Vessel Code (22.450)119481 Amends § 25.05.675(N), this source additional environmental review of proposed development within

shoreline zones (25.05)

Seattle Municipal Code eference only. 119484 Adds §§ 23.49.050(E), 23.49.070(E), Text provided for hi 23.49.100(E), 23.49.126(C)-(E) and 23.49.152(C)-(E); amends §§ 3.20.320(A), 23.32.016, amending 23.34.100, 23.34.114, 23.34.116, Ch. 23.49 (Maps 1B, VD, VIA, VID), 23.49.035(B), See ordinan 23.49.036(E), 23.49.046(G) and (H), sections for comp 23.49.048(A)-(C), 23.49.050(A), (C) and (D), 23.49.052(A)-(D), 23.49.066(G) and (H), and tables and 23.49.068(A)-(C), 23.49.070(A), (C) and (D), 23.49.072(A)-(D), this source file. 23.49.096(H) and (I), 23.49.098, 23.49.100(A), (C) and (D), 23.49.102(B) and (D), 23.49.122(G) and (H), 23.49.124, 23.49.126(A), 23.49.128(A)-(D), 23.49.148(F), 23.49.150(C), 23.49.152(A), 23.49.154(A), (B) and (D), 23.49.180(A), (B) and (D), 23.49.212(A), (B) and (D), 23.49.240(A), (B) and (D), 23.49.324(G), Ch. 23.66 (Maps A-D), 23.66.100(A)-(C), 23.66.122(A) and (B), 23.66.124(A), 23.66.130(A)-(E), 23.66.140(A) and (B), 23.66.150, 23.66.170(D) and 23.84.024; repeals § 23.12.080, land use and zoning (3.20, 23.32, 23.34,

## For current SMC, contact the Office of the City Clerk

119490

119491

119494

119501

23.49, 23.66, 23.84)

Amends § 18.28.010, Seattle Aquarium admission fees (18.28)

Amends § 3.14.920(C), terms of service for various commissions (3.14)

Add § 3.06.055, restricted

reserves (3.06)

Amends § 23.41.004(A), north neighborhoods neighborhood plan (23.41)

June 2009 co

Text provided for

See ordinant sections for and tables a this source

119504 Creates Department of Information Technology, Information Technology Fund and certain positions; adds Ch. 3.22, Department of Information Technology; repeals §§ 3.24.200, 3.24.210 and 3.24.220 (3.22) 119506 Adds Map 23.47.004(C); amends Seattle Comprehensive Plan and §§ 23.32.016 and 23.47.004(E); repeals West

119507

file.

119509

119510

119521

eference only.

amending

graphics,

ccuracy of

Seattle Junction Comprehensive Plan Study (23.32, 23.47)Adds § 22.300.015, adoption of the National Electrical Code; repeals § 22.300.010; makes various changes to 1999 edition of the National Electrical Code including addition of Chs. 1-3 and §§ 110-23, 110-26(e), 215-12, 230-5, 230-10, 230-33, 230-44, 230-62(c) and (d), 250-104(e), 300-1(c), 338-3(d), 380-13(e), 450-19, 450-20 and 555-12; amends National Electrical Code Art. 100 and §§ 110-13, 210-8(a), 220-3(a), 220-15, 220-17, 230-1, 230-28, 230-29, 230-43, 230-82, 230-90(a), 250-56, 300-23, 324-4, 336-4, 336-5(a), 336-6, 348-4-348-6, 370-1, 380-3(a), 450-10, 553-4, 620-5, 620-21, 620-22(a), 620-44, 620-51(b), 620-71, 700-4, 700-16 and 701-10; repeals National Electrical Code §§ 90-1, 90-2, 230-46, 230-52, 336-21, 338-2, Art. 342, 380-10(a) and 450-41-450-48 (22.300)Adds §§ 22.207.002-22.207.026; amends § 22.202.050, enforcement process for building and construction codes (22.202, 22.207) Amends § 11.16.340(P), compression brakes (11.16) Amends §§ 23.34.020(B) and 23.34.022(B), Queen Anne Neighborhood Plan

For cur<del>rent 314 (25.34)</del>
the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

		te file	
	119525	Amends Seattle	!
		Comprehensive Plan and	
	101	§§ 23.32.016 and	
	A FOR III	23.47.004(E), Roosevelt	
3	CI ZO-	Neighborhood Plan (23.47)	
	119550	Repeals § 22 of Charter	
		Art. IV (Passed by voters at	
		general election of	
		November 2, 1999)	8
	119551	Amends § 4 of Charter Art.	
	ces cross	XIX, officers, terms and	
		vacancies (Passed by	2
		voters at general election of	
	COMPA	November 2, 1999)	
	119552	Amends Charter to replace	
	140.00	all exclusively male gender	
	and to co	references with	
	The state of	gender-neutral references	
		(Passed by voters at	
	C110.	general election of	
	Ille.	November 2, 1999)	
	119554	Amends §§ 5 and 6 of	
		Charter Art. XVI,	
		personnel system and civil	
		service (Passed by voters at	
		general election of	
		November 2, 1999)	
	119555	Amends § 1(B) of Charter	
		Art. IV, Legislative	
		Department (Passed by	
		voters at general election of	
		November 2, 1999)	
	119556	Amends § 24 of Charter	
		Art. IV, Legislative	
		Department (Passed by	
		voters at general election of	
		November 2, 1999)	
	119557	Amends § 2 of Charter Art.	
		X, Fire Department (Passed	
		by voters at general	
		election of November 2,	
	110550	1999)	
	119558	Amends § 9 of Charter Art.	
		IV, Legislative Department (Passed by voters at	
		general election of	
		November 2, 1999)	
	119559	Amends § 2(A) of Charter	
	11/33/	Art. IV, Legislative	
		Department, and §§ 1 and 2	
		of Charter Art. XVIII,	
		elections (Passed by voters	
		at general election of	
		November 2, 1999)	
	119560	Amends § 12 of Charter	
		Art. IV, Legislative	
		Department (Passed by	
		voters at general election of	
		November 2, 1999)	
		,,	1

eference only.

at d amending

tt. graphics,
of ccuracy of

119563

Text provided for

See ordinan sections for and tables and this source

Adds new section to Charter Art. VIII, financial and clerical; amends §§ 1(B), 13 and 22 of Art. IV, Legislative Department, § 8 of Art. V, Executive Department, §§ 3, 10, 11 and 14 of Art. VIII, financial and clerical, § 5 of Art. XII, The Library Department, and § 1 of Art. XVIII, elections (Passed by voters at general election of November 2, 1999) Changes headings of Charter Arts. VII and VIII,

tax or fee (5.44)

and amends § 1 of Charter Art. XVII, salaries and bonds; repeals Charter Art. II, § 12 of Art. V and Art. IX (Passed by voters at general election of November 2, 1999) 119566 Amends § 5.44.100(G), under or overpayment of

119576 Amends § 3 of Charter Art. XVIII [XVII], salaries and bonds; repeals § 13 of Charter Art. VIII, financial and clerical (Passed by voters at general election of November 2, 1999) 119578 Amends §§ 4.14.100, 4.14.120, 4.72.010(A) and (B), 4.72.020(A), 4.72.040-4.72.070 and 4.72.080(A)-(D); repeals §§ 4.72.090, 4.72.180 and 4.72.190, travel related expenses (4.14, 4.72) 119590 Amends § 4.36.385, compliance with Internal Revenue Code (4.36) 119591 Amends § 6.288.020(B). exception for licenses for used vessel dealers (6.288) 119601 Fair contracting practices (14.10)119602 Small economically disadvantaged businesses (20.49)119611 Amends § 5.44.120, business and occupation taxes (5.44) 119614 Amends Seattle Comprehensive Plan and §§ 23.32.016 and

ference only.

Plan (23.47) the Office of the City Clerk

23.47.004(E), Rainier Beach 2014 Neighborhood

119619

119622

119628

119633

119635

June 2009 co

Text provide

See ordinant sections for and tables a this source

CIDOS	
الارد م	ote file
119616	Amends § 23.47.004(E),
1630 0.7	South Lake Union
10 - 10 to	Neighborhood Plan (23.47)
119617	Amends §§ 22.206.160,
	23.44.041, 23.84.028,
	23.90.019(A) and § 310.6
	of the Seattle Building
	Code, accessory dwelling
	units (22.206, 23.44, 23.84,
of offer	23.90)
119618	Adds § 23.53.035; amends
	§§ 23.22.062(A),
	23.24.045(A),
COMPA	23.44.016(B),
	23.44.080(D),
1 40 C	23.49.212(D), 23.76.004
and to	(Exhibit A), 23.76.006(B)
	and (C), and 23.84.024,
009	land use and zoning (23.22,
	23.24, 23.49, 23.53, 23.76,

23.84)

Amends § 3.58.030, Seattle Design Commission (3.58)

Amends Section 1 of Ordinance 119576 to correct the erroneous Article number for the heading "Salaries and Bonds" (Charter Art. XVII)

Amends §§ 3.14.910(G), 3.14.931(B), 3.110.260, 4.80.020, 6.02.270, 6.202.230(F), 12A.06.115, 14.04,020, 14.04.030, 14.04.040, 14.08.020, 14.08.040, 14.08.050, 14.08.055, 14.08.060, 14.08.070, 14.08.080, 14.08.090, 14.08.190, 18.12.280, 20.44.040, 20.44.030 and 20.44.050, various provisions relating to discrimination (3.14, 3.110, 4.80, 6.02, 6.202, 12A.06, 14.04, 14.08, 18.12, 20.44)

Amends Seattle

to the North Neighborhoods

Comprehensive Plan and § 23.47.004(E), North Neighborhoods Neighborhood Plan; repeals Lake City Gateway Plan, 1981 (23.47)

Amends §§ 23.34.020(B) and 23.34.022(B), Lowrise 3 and Lowrise 4 rezone evaluation criteria relating

eference only.

l amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Sihos	0210
a mada	te tille
119636	Adds § 11.56.020(N)(10)
	and amends §
a com hi	11.56.020(N)(7), persons
d tor	under the influence of
	intoxicating liquor or any
	drug (11.56)
119637	Amends §§ 23.34.020(B)
	and 23.34.022(B), Lowrise 3 and Lowrise 4 rezone
- 0003	evaluation criteria relating
res crea	to the Morgan Junction
	Neighborhood Plan (23.34)
119640	Amends § 5.44.028(4),
COMPA	Business and Occupation
	Tax Code (5.44)
119651	Amends §§ 3.114.030,
ma to co	3.114.040, 3.114.050,
,350	3.114.140 and 3.114.150;
000	repeals and replaces §
file.	3.114.160; repeals §§
8800	3.114.020, 3.114.145 and 3.114.170, Consultant
	Contracting Ordinance
	(3.114)
119672	Repeals and replaces §
11,0,2	23.70.004 Exhibit Maps
	1A-1J, Station Area
	Interim Overlay District
	(Expired)
119673	Amends Street Use Fee
	Permit Schedule and §§
	15.02.046, 15.04.035 and
	15.04.074, shoreline street
119678	ends (15.02, 15.04) Amends §§ 14.04.030 and
119078	14.04.185(A), employment
	discrimination (14.04)
119686	Adds § 12A.06.060,
	negligent control of an
	animal (12A.06)
119688	Amends §§ 21.04.060(B),
	21.04.280 and
	21.16.270(B), water and
	sewer infrastructure
	extensions that serve
	residences of very
	low-income households (21.04, 21.16)
119691	Amends §§ 23.34.020(B),
11/0/1	23.34.022(B) and
	23.47.004(E), Bitter Lake
	Village Hub Urban Village
	(23.34, 23.47)
119694	Amends Comprehensive
	Plan and § 23.32.016,
	Official Land Use Map
	(Special)

reference only.

Set amending

Set aphics,

Laccuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	cilo
a anda	te me
119698	Adopts Map 23.47.004(H)
	and amends Official Land
a com hi	Use Map and §
d tor	23.47.004(E), land use and
	zoning (23.47)
119702	Amends § 23.90.014,
	review by the Director
	(23.90)
119705	Adds § 4.20.390, Manager
- oc crea	and Strategic Advisor
ces es	Compensation Program;
	repeals §§ 4.20.400 and
	4.20.420 (4.20)
119706	Amends § 4.20.380,
	Accountability Pay for
1 40 00	Executives Program (4.20)
119707	Adds §§ 4.36.040(J) and
Page -	4.36.185; amends §§
	4.36.192(B), 4.36.200(B)
C210	and (C), 4.36.210(F) and
1116.	(H), 4.36.230(E),
	4.36.260(D) and (E),
	4.36.270 and 4.36.320.
	City Employees'
	Retirement System (4.36)
119713	Amends Seattle
119/13	
	Comprehensive Plan and §
	23.32.016, Official Land
110714	Use Map (Special)
119714	Amends Seattle
	Comprehensive Plan, §
	23.32.016, Official Land
	Use Map, and §§
	23.34.020(B) and
	23.34.022(B), Admiral
110=1=	Neighborhood Plan (23.34)
119715	Amends §§ 23.48.032 and
	23.54.015 (Chart A), South
	Lake Union Neighborhood
	Plan (23.48, 23.54)
119724	Amends § 23.34.010(B)
	and Official Land Use
	Map, North Beacon Hill
	1,
	Neighborhood Plan (23.34)
119728	_
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G),
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041,
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G),
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and 23.76.026(F); amends §§ 23.49.008, 23.49.026(B),
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and 23.76.026(F); amends §§ 23.49.008, 23.49.026(B), 23.49.058, 23.49.068(C),
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and 23.76.026(F); amends §§ 23.49.008, 23.49.026(B), 23.49.058, 23.49.068(C), 23.49.078, 23.49.136, Ch.
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and 23.76.026(F); amends §§ 23.49.008, 23.49.026(B), 23.49.058, 23.49.068(C), 23.49.078, 23.49.136, Ch. 23.49 Maps IB, IIID and
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and 23.76.026(F); amends §§ 23.49.008, 23.49.026(B), 23.49.058, 23.49.068(C), 23.49.078, 23.49.136, Ch. 23.49 Maps IB, IIID and VD, and 23.86.030(A),
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and 23.76.026(F); amends §§ 23.49.008, 23.49.026(B), 23.49.058, 23.49.068(C), 23.49.078, 23.49.136, Ch. 23.49 Maps IB, IIID and VD, and 23.86.030(A), Transfer of Development
119728	Neighborhood Plan (23.34) Adds §§ 23.49.041, 23.49.076(G), 23.49.134(G) and 23.76.026(F); amends §§ 23.49.008, 23.49.026(B), 23.49.058, 23.49.068(C), 23.49.078, 23.49.136, Ch. 23.49 Maps IB, IIID and VD, and 23.86.030(A),

eference only.

and

amending

graphics,

accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

-40	te file
110505	1 33 5 40 055
119737	Amends §§ 5.48.055,
	21.36.014, 21.40.050,
1 Cost h1	21.40.060, 21.40.080,
d ior	21.40.085 and 21.76.040,
	solid waste system of
	Seattle Public Utilities
	(5.48, 21.36, 21.40, 21.76)
119738	Rezone, Official Land Use
	Map, page 109 (Special)
119743	Amends Seattle
SG2 CI On	Comprehensive Plan and §
	23.32.016, Official Land
100	Use Map,
comple	Greenwood/Phinney Ridge
	Neighborhood Plan
	(Special)
119747	Amends Ch. 21.49, Seattle
117/4/	
	City Light Department
110740	(21.49)
119748	Adds Ch. 20.45, City
	contracts-non-discriminatio
	n in benefits (20.45)
119757	Amends § 18.28.010,
	Department of Parks and
	Recreation fee schedule
	(18.28)
119761	Amends §§ 5.06.030(B),
	5.80.020 and 5.80.030,
	Cumulative Reserve
	Subfund (5.06, 5.80)
119762	Creates Health Care
119702	Subfund and amends §
	4.50.010, administration of
	certain benefit programs
	(4.50)
119763	Amends § 4.13.010,
119703	exemptions from the civil
	•
110766	service (4.13)
119766	Adds § 22.900D.145;
	amends §§ 22.900.130(B)
	and (C), 22.900A.050,
	22.900B.010, 22.900B.050,
	22.900C.010,
	22.900D.010,
	,
	22.900D.010,
	22.900D.010, 22.900D.060(D),
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A),
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C),
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020,
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and charges for permits and
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and charges for permits and activities of the
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and charges for permits and activities of the Department of Design,
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and charges for permits and activities of the Department of Design, Construction and Land Use
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and charges for permits and activities of the Department of Design, Construction and Land Use (22.900, 22.900A,
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and charges for permits and activities of the Department of Design, Construction and Land Use (22.900, 22.900A, 22.900B, 22.900C,
	22.900D.010, 22.900D.060(D), 22.900D.090, 22.200D.110(A), 22.900D.130(B) and (C), 22.900D.150, 22.900E.050 and 22.900F.010; repeals §§ 22.900D.020, 22.900D.040 and 22.900D.050, fees and charges for permits and activities of the Department of Design, Construction and Land Use (22.900, 22.900A,

reference only.

sel amending
spaphics,
curacy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Amends §§ 21.28.040(B) and 21.76.040(A), wastewater services of Seattle Public Utilities (21.28, 21.76)		Ipai o	
and 21.76.040(A), wastewater services of Seattle Public Utilities (21.28, 21.76)  119769  Amends §\$ 21.04.430, 21.04.440(A)-(D) and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 17.16.015, Seattle Center facility license fee schedule for 1999 and 2000 (17.16)  Adds § 22.502.016; amends §§ 22.502.016; amends §§ 22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  119784  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40) Adds § 23.45.014(I); amends §§ 23.42.020, 23.24.040(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds § 23.44.008(I) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds § 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Reighborhood Plan		a anda	te file
wastewater services of Seattle Public Utilities (21.28, 21.76)		119768	
Seattle Public Utilities (21.28, 21.76)			
(21.28, 21.76)  Amends § 21.04.430, 21.04.440(A)-(D) and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 17.16.015, Seattle Center facility license fee schedule for 1999 and 2000 (17.16)  Adds § 22.502.016; amends § 22.504.010 and 22.504.020; repeals § 22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(1); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds § 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan		a sor hi	
Amends § \$ 21.04.430, 21.04.440(A)-(D) and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)   119773	8	dini	
21.04.440(A)-(D) and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 17.16.015, Seattle Center facility license fee schedule for 1999 and 2000 (17.16)  119774  Adds § 22.502.016; amends § \$ 22.504.010 and 22.504.020; repeals § 22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends § § 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781  Adds § 11.72.352; amends § 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791  Adds § 23.45.014(I); amends § 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.0156(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds § 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan	1	119769	
21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76) Amends § 17.16.015, Seattle Center facility license fee schedule for 1999 and 2000 (17.16) Adds § 22.502.016; amends § 22.504.010 and 22.504.020; repeals § 22.502.015, Seattle Plumbing Code (22.502) Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20) Designates and adjusts salaries for certain positions and amends § § 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20) Adds § 11.72.352; amends § 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73) Amends § 11.30.040, towing vehicles without prior notice in restricted parking zone (11.72, 11.73) Amends § 23.40.050(B), demonstration program for innovative housing design (23.40) Adds § 23.45.014(I); amends § 23.42.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45) Aca service of the preservation and planting requirements (23.44, 23.45) Rezone, Delridge Neighborhood Plan		119709	
Utilities (21.04, 21.76)			` ' ` '
Amends § 17.16.015, Seattle Center facility license fee schedule for 1999 and 2000 (17.16)  Adds § 22.502.016; amends §§ 22.504.010 and 22.504.020; repeals § 22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
Seattle Center facility license fee schedule for 1999 and 2000 (17.16)	L	es cred	
license fee schedule for 1999 and 2000 (17.16)  Adds § 22.502.016; amends § § 22.504.010 and 22.504.020; repeals § 22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends § § 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  Adds § 11.72.352; amends § § 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends § 23.45.014(I); amends § 23.40.02, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds § 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan	Х	119773	
1999 and 2000 (17.16)		10	
Adds § 22.502.016; amends §§ 22.504.010 and 22.504.020; repeals § 22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan		comple	
amends §§ 22.504.010 and 22.504.020; repeals § 22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zone (11.72, 11.73)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan		119774	
22.502.015, Seattle Plumbing Code (22.502)  Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  119776  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791  Adds § 23.45.014(I); amends § 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds § 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan		140 00	
Plumbing Code (22.502)     Adds § 4.20.375, parking of privately-owned motor vehicles by   Councilmembers (4.20)     119776   Designates and adjusts salaries for certain positions and amends §§   4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology   Compensation Program (4.20)     119781   Adds § 11.72.352; amends §§ 11.73.050 and   11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)     119782   Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)     119784   Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)     119791   Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)     119792   Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)     119794   Rezone, Delridge Neighborhood Plan		and to co	
Adds § 4.20.375, parking of privately-owned motor vehicles by Councilmembers (4.20)  119776  Designates and adjusts salaries for certain positions and amends §   4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781  Adds § 11.72.352; amends § 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends § 8 3.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds § 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan		,8800	,
of privately-owned motor vehicles by Councilmembers (4.20)  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781 Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan	L	110555	
vehicles by Councilmembers (4.20)  119776  Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds § § 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan		1197/5	
Councilmembers (4.20)			
Designates and adjusts salaries for certain positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)			3
positions and amends §§ 4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781 Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan		119776	
4.20.300(A), (C) and (D), and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781 Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  119782 Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  119784 Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			
and 4.20.430(A) and (B), Information Technology Compensation Program (4.20)  119781 Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  119782 Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  119784 Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			
Information Technology Compensation Program (4.20)  119781  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  119782  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
Compensation Program (4.20)  119781  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
(4.20)  119781  Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
Adds § 11.72.352; amends §§ 11.73.050 and 11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
11.73.060, Husky Stadium event restricted parking zone (11.72, 11.73)  119782 Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  119784 Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan		119781	
event restricted parking zone (11.72, 11.73)  119782  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  119784  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan			
zone (11.72, 11.73)  Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  119784 Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			
Amends § 11.30.040, towing vehicles without prior notice in restricted parking zones (11.30)  119784 Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			
towing vehicles without prior notice in restricted parking zones (11.30)  119784 Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan	F	119782	
parking zones (11.30)  Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
Amends § 23.40.050(B), demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.41.014(B) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			
demonstration program for innovative housing design (23.40)  119791 Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan	L		
innovative housing design (23.40)  119791  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan		119784	
(23.40)  Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
Adds § 23.45.014(I); amends §§ 23.22.020(A), 23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
23.22.054, 23.24.020, 23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan	f	119791	
23.24.040(A), 23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  Rezone, Delridge Neighborhood Plan			
23.41.014(B), 23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			
23.44.014(D) and 23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			· //
23.45.056(D), land use and zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			* * *
zoning (23.22, 23.24, 23.41, 23.44, 23.45)  119792 Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan			
23.41, 23.44, 23.45)  Adds §§ 23.44.008(I) and 23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan			
23.45.015(C), tree preservation and planting requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan	L		23.41, 23.44, 23.45)
preservation and planting requirements (23.44, 23.45)  119794 Rezone, Delridge Neighborhood Plan		119792	
requirements (23.44, 23.45)  119794  Rezone, Delridge Neighborhood Plan			
23.45) 119794 Rezone, Delridge Neighborhood Plan			
119794 Rezone, Delridge Neighborhood Plan			
Neighborhood Plan	f	119794	
(Special)			
	L	- CM	(Special)

d amending graphics, accuracy of

For current Six (Special) Contact the Office of the City Clerk

June 2009 co Text provide

See ordinantes sections for and tables at this source

	a mada	te file
	119795	Rezone, Plat 5E, page 43
	119796	(Special) Amends § 23.34.010(B),
	d'for m	designation of
G	U 300	single-family zones (23.34)
	119797	Adds Subchapter IV to Ch.
		10.28, Seattle Fire Department Medical
		Incident Reports (10.28)
	119804	Rezone, Plat 45E, page 119
M	CES CS	(Special)
	119816	Adds Subchapter VIII to
10	comple	Ch. 3.28, Office of Professional
5		Accountability (Repealed
	140 00	by 120728)
8	119825	Amends §§ 3.28.600,
		3.28.620, 3.28.630, 3.28.640, 3.28.650 and
	C110.	3.28.660, Police
	1110.	Department (Repealed by
		120728)
	119836	Insert §§ 23.45.006(K), 23.47.029, 23.48.031 and
		23.49.015, land use and
		zoning (23.45, 23.47,
		23.48, 23.49)
	119837	Amends §§ 23.44.041(A), 23.47.012(B),
		23.49.008(A) and (C), and
		23.91.002(A), land use and
		zoning (23.44, 23.47,
	119839	23.49, 23.91) Amends § 23.84.036,
	119039	on-premises signs (23.84)
	119860	Amends § 5.48.050(C),
		utility business tax (5.48)
	119863	Establishes certain title and salary, and corrects civil
		service exempt status of
		certain positions; amends §
		4.13.010, salaries (4.13)
	119871	Repeals § 23.60.604(A)(2)
	119872	(Repealer) Amends §§ 6.310.110,
	117072	6.310.130, 6.310.150,
		6.310.205, 6.310.230,
		6.310.300, 6.310.320,
		6.310.330, 6.310.350, 6.310.370, 6.310.400,
		6.321.415, 6.310.420,
		6.310.430, 6.310.450,
		6.310.455, 6.310.465,
		6.310.470, 6.310.500,
		6.310.530, 6.310.605 and 6.310.710, taxicabs and
		for-hire vehicles (6.310)

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Japan An	te file
119893	Amanda \$ 2.29 600 and
119893	Amends § 3.28.600 and repeals § 3.28.600(B),
1-3	Office of Professional
	Accountability Review
C In I	Board (Repealed by
	120728)
119895	Amends § 23.76.062, land
119093	use and zoning (23.76)
119896	Amends §§ 22.207.004,
117070	22.207.006, 22.207.010,
ces crea	22.207.010, 22.207.010, 22.207.012, 23.91.004,
	23.91.006, 23.91.010 and
10	23.91.012, enforcement of
COMPLE	Housing and Building
Cosses	Maintenance Code and
	Land Use Code (22.207,
ad to co	23.91)
119903	Adds Ch. 6.235, gas,
11,500	piping regulations and
C:10	licensure (6.235)
119904	Amends §§ 23.42.040,
119904	23.76.006(C),
	23.76.010(A) and
	23.84.024. land use and
	zoning (23.42, 23.76,
	23.84)
	23.04)
119929	Amends §§ 23.60.092,
11//2/	23.60.662 and 23.60.668,
	water-based airports
	(23.60)
119933	Extends to 6/30/01
11,7,00	moratorium on new adult
	cabarets set by Ord.
	119448 (Ch. 6.270
	footnote)
119951	Amends § 4.94.040,
	awards programs (4.94)
119953	Adds § 20.40.020, small
	works roster (20.40)
119965	Adds §§ 22.800.020(C),
	22.800.060(C), 22.801.090
	and 22.802.016; amends §§
	22.800.010, 22.800.030,
	22.800.070, 22.800.080,
	22.801.010-22.801.080,
	22.801.100-22.801.210,
	22.801.240, 22.802.010,
	22.802.013, 22.802.015,
	22.802.020,
	22.802.090(A),
	22.804.030-22.804.050,
	22.804.110, 22.804.160,
	22.804.110, 22.804.100, 22.804.180, 22.808.020
	and 22.808.120; repeals §
	22.808.150, grading and
	drainage control (22.800,
	22.801, 22.802, 22.804,
	22.001, 22.002, 22.004,
	22.808)
119969	22.808) Amends § 23.54.015, Chart

eference only.

amending
graphics,
accuracy of

For current Simo, the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a mada	te file
	119971	Amends §§ 23.60.844 and
		23.60.848, land use (23.60)
	119972	Adds Ch. 23.74; amends §§
	A FOR III	23.41.004, 23.41.006,
C		23.41.012(B),
		23.50.012(A).
		23.50.014(B),
		23.50.026(A), 23.50.027,
		23.54.015(A) and Chart A,
	on orea	and Downtown Design
	ces cross	Guidelines; repeals and
		replaces Exhibit
		23.41.006A, land use and
10	COLLIBBIO	zoning (23.41, 23.50,
5		23.54, 23.74)
	119974	Adds 23.80.004(C);
	and to co	amends Exhibit 23.76.004
O.	The same	A, §§ 23.76.006(C),
		23.84.010 and 23.84.038,
	C310	light rail transit facilities
9	2116.	(23.76, 23.80, 23.84)
	119986	Amends § 23.70.004;
		renumbers § 23.70.004
		Exhibit 1D-1J; repeals §
		23.70.004 Exhibit 1C,
		Station Area Interim
		Overlay District (Expired)
	119998	Adds § 9.25.049; amends
		§§ 9.25.020, 9.25.021,
		9.25.023,
		9.25.030-9.25.037,
		9.25.045-9.25.047,
		9.25.050-9.25.054,
		9.25.080-9.25.085,
		9.25.090,
		9.25.092-9.25.110 and
		9.26.020-9.26.050; repeals
		§§ 9.25.024 and 9.25.048,
		animal control and fees
	120000	(9.25, 9.26)
	120000	Amends § 25.05.675(M),
	120002	parking policies (25.05)
	120002	Amends § 3.33.140, judges
	120004	pro tempore (3.33)
	120004	Adds § 23.73.010; amends
		§§ 23.32.016,
		23.47.032(A),
		23.47.040(C), 23.54.015
		(Chart A), 23.73.004 and
		23.73.008(B) and (C);
		repeals and replaces
		Exhibit 23.73.004A;

ference only.

amending

raphics,

curacy of

For current SMC, contact the Office of the City Clerk

120005

repeals § 5 of Ord. 119413, Pike/Pine Urban Center Village (23.32, 23.47, 23.54, 23.73)

Amends § 11.30.105(B) and (C), impoundment of vehicles (11.30)

June 2009 co

Text provide

See ordinan sections for and tables 3 120023 this source file.

5	a anda	te file
	120006	Amends §§ 11.30.105(A),
		11.30.120(B), (C) and (D) and 11.30.160,
	d for m	impoundment of vehicles (11.30)
	120007	Amends § 11.30.120(B),
		impoundment of vehicles
		(11.30)
	120012	Amends §§ 5.16.030 and
	oc crea	5.16.050, incurring debt
		(5.16)
	120021	Adds Ch. 20.49; repeals
		and replaces §§ 1, 4, 5 and
)	COMPL	8 of Ord. 119602; repeals
		Ord. 119956 and § 11 of
	140 00	Ord. 119602, Boost

Ord. 119602, Boost Program (20.49) Adds §§ 16.04.045, 16.04.065, 16.04.075, 16.04.105, 16.04.123, 16.08.011-16.08.015, 16.20.031, 16.20.071, 16.20.072, 16.20.081, 16.20.115, 16.28.040, 16.32.060, 16.64.050; amends §§ 16.04.020, 16.04.040, 16.04.060, 16.04.070, 16.04.080, 16.04.090, 16.04.110, 16.04.130, 16.04.160, 16.04.180-16.04.210, 16.08.020-16.08.050, 16.12.010-16.12.030, 16.16.010-16.16.030, 16.20.010-16.20.030, 16.20.040, 16.20.060, 16.20.070, 16.20.080, 16.20.090, 16.20.100, 16.20.120-16.20.190, 16.24.010, 16.24.020, 16.28.010-16.28.025, 16.32.020-16.32.050, 16.36.010-16.36.050, 16.40.010-16.40.030, 16.44.010, 16.44.020, 16.48.010, 16.48.020, 16.52.010, 16.52.020, 16.52.060, 16.52.070, 16.52.090, 16.52.110, 16.56.010, 16.56.030, 16.60.010-16.60.030, 16.60.060, 16.64.010, 16.64.040; repeals §§ 16.20.050, 16.20.230, 16.20.240, 16.52.040, 16.52.050, 16.52.080, 16.56.020, Harbor Code (16.04, 16.08, 16.12, 16.16, 16.20, 16.24, 16.28, 16.32, 16.36, 16.40, 16.44, 16.48, 16.52, 16.56, 16.60, 16.64)

eference only. amending graphics, ccuracy of

For current SM the Office of the City Clerk

June 2009 co

Text provide

See ordinand sections for and tables a this source

Cihas	C:10
a and	te tille
120046	Adds § 3.06.015; amends §
	3.102.010, administration
	(3.06, 3.102)
120057	Amends § 11.56.020(Q),
	persons under the influence of intoxicating liquor or
	any drug (11.56)
120059	Adds §§ 12A.06.180(G)
	and 12A.18.290
Loc crea	[12A.18.090]; amends §§
ICES 62	12A.06.130,
10	12A.06.165(A) and (B),
comple	12A.06.170(A), 12A.06.180(A)-(D), title of
Compage	Ch. 12A.18 and
	12A.18.010; repeals §
and to c	12A.06.310, criminal code
B'II C	(12A.06, 12A.18)
120060	Adds § 11.31.050(E) and
sile.	renumbers prior §
	11.31.050(E) to be §
	11.31.050(F), traffic code (11.31)
120066	Amends § 22.600.020,
120000	Seattle Fire Code (22.600)
120081	Amends §§ 23.41.010 and
	23.41.012, land use (23.41)
120086	Amends § 9.25.100(D);
	temporarily suspends §
	9.26.020(D), animal
120087	control (9.25) Adds § 22.206.217;
120087	amends §§ 22.200.030(B),
	22.204.050, 22.206.130,
	22.206.200(F), 22.206.220,
	22.206.230, 22.206.250,
	22.206.280 and title of
	Subchapter VII, housing code (22.200, 22.204,
	22.206)
120096	Adds § 3.28.130, certain
	uniformed peace officers
	required to wear
	identification and orally
120007	identify themselves (3.28)
120097	Amends § 11.72.440, traffic code (11.72)
120098	Amends § 11.14.010 and
120070	renumbers it to be
	11.14.268, junk motor
	vehicle (11.14)
120101	Amends § 11.66.120,
	locomotive and train speed
120102	limit (11.66)
120102	Adds § 11.72.500; amends §§ 11.30.040(A) and
	11.30.060(B), motor
	vehicles (11.30, 11.72)
	. , , , , , ,

eference only.

d amending

graphics,
of ccuracy of

Seattle Municipal Code eference only. June 2009 co 120111 Adds §§ 21.49.059 and 21.49.082; amends §§ 21.49.020(A) and 21.49.055, Seattle City Light Department (21.49) 120114 Adds § 3.38.030; amends amending §§ 3.38.040, 3.30.110, 3.30.120, 3.106.010, 4.14.150(B), 4.20.110(A), 4.20.700, 4.36.130(H), See ordinan 4.36.295(F), 4.72.050, 5.20.050, 5.24.020, 5.24.030, 5.44.030(D), sections for 5.78.020-5.78.100, 5.78.120, 5.78.130, and tables and to 5.78.150-5.78.190, 10.02.030(B), 11.23.140, 11.30.280, 12A.62.020, 15.04.090, 20.04.150, this source file. 20.08.030, 20.12.080(C), 20.76.080(E) [20.76.080], 20.76.120, 20.76.130, 20.76.200, 21.24.090, 21.28.350, 22.220.100(C), payment of claims or other obligations by check or warrant (3.30, 3.38, 3.106, 4.14, 4.20, 4.36, 4.72, 5.20, 5.24, 5.44, 5.78, 10.02, 11.23, 11.30, 12A.62, 15.04, 20.04, 20.08, 20.12, 20.76, 21.24, 21.28,

22.220)

Seattle Municipal Code ference only. date 120117 Adds § 23.45.016(C); amends §§ 23.04.010(C), Text provided for hi 23.24.020(G), 23.32.016, 23.34.010(B), 23.44.022(K), 23.45.006(K), 23.45.009(D), 23.45.012(D), 23.45.014(F), and (I), 23.45.015(C), See ordinan 23.45.018(B), 23.45.050(D), 23.45.058(B), sections for 23.45.066(C), 23.45.096(D), 23.45.166, 23.47.004(F) and Chart A, and tables and 23.47.008(C), 23.47.009(A) and (C), 23.47.012(G), this source file. 23.47.029(A)-(C), 23.47.042(D), 23.48.010(C), 23.48.014(D), 23.48.031(A)-(C), 23.49.008 (C), 23.49.015(A)-(C), 23.49.016(B), 23.49.054(A), 23.49.074(A), 23.49.098(C), 23.49.104(A), 23.49.132(A), 23.49.160(A), 23.50.012, Chart A, 23.50.014(B), 23.50.020(A), 23.53.020(E), 23.60.454(D), 23.60.572(D), 23.60.632(D), 23.60.692(C), 23.60.812(D), 23.66.140(C), 23.66.332(C), 23.71.004, Map A, 23.84.002, 23.84.004, 23.84.016, 23.84.024, 23.84.025, 23.84.032, 23.86.002(B) and 23.86.018(H) and (I); repeals Ch. 23.68 and §§ 23.34.075, 23.34.098 and 23.55.032, land use and zoning (23.04, 23.24, 23.34, 23.44, 23.45, 23.47, 23.48, 23.49, 23.50, 23.53, 23.60, 23.66, 23.71, 23.84, 23.86) 120118 Amends § 4.16.100; renumbers § 4.16.100(B)

> For current SMC, contact the Office of the City Clerk

and (C) to be § 4.16.105, code of ethics (4.16)

June 2009 co

Text provide

See ordinant sections for and tables a this source

4	te file
120121	Adds Subchapter VIII and
	§§ 3.14.950-3.14.980;
a com hi	amends § 3.14.501(D),
d tor	Office of Sustainability and
	Environment (3.14)
120132	Repeals and replaces §
	12A.06.115, criminal code
	(12A.06)
120133	Amends §§
oc crea	6.230.030-6.230.070,
CES CS	6.230.100 and 6.230.180,
	steam engineers and boiler
	firemen (6.230)
120134	Amends §§ 6.82.020,
	6.82.030, 6.82.060,
140 0	6.82.100-6.82.120,
ma to c	6.82.190 and 6.82.320;
188	repeals §§ 6.82.050 and
	6.82.180, refrigeration
<b>C110</b>	systems (6.82)
120135	Adds § 5.72.065; amends
	§§ 5.72.030 and 5.72.040,
	multifamily housing
	property tax exemption
	(5.72)
120138	Amends §§ 21.60.080(A),
	21.60.090, 21.60.220,
	21.60.290, 21.60.380,
	21.60.390 and 21.60.700,
	cable television (21.60)
120140	Amends § 6.288.065, used
	goods dealers (6.288)
120142	Amends § 4.13.010(O);
120112	establishes new titles
	and/or salaries; designates
	exempt positions; corrects
	effective date in § 1 of Ord.
	120069, personnel (4.13)
120145	Adds §§ 2.04.010(31) and
120113	2.04.270(C); amends §§
	2.04.010(8) and (13),
	2.04.060(H), 2.04.155,
	2.04.165(A), 2.04.180(A),
	2.04.230(C)-(G), 2.04.265,
	2.04.275, 2.04.290(B),
	2.04.370 and 2.04.375,
	elections (2.04)
120149	Amends §§ 21.49.030,
12017)	21.49.040 and
	21.49.040 and 21.49.052-21.49.058;
	21.49.052-21.49.058; repeals § 21.49.081, City
120155	Light Department (21.49)
120155	Amends §§ 23.50.012(C)
	and 23.50.027(A) and (C),
	land use and zoning
100177	(23.50)
120156	Amends §§ 23.90.018 and
	23.90.022, land use code enforcement (23.90)

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

defere

120175

120176

120177

120178

120179

Text provided for

See ordinan sections for and tables and this source

120157 Amends §§ 22.600.025, 23.66.030(E), 23.66.115(C), 23.66.318(C), 23.76.005, 23.76.015(B), 23.76.016(B), 23.76.024(B), 23.76.052(B), 25.12.375, 25.12.740(D), 25.12.835(E), 25.12.845(C), 25.16.110(C), 25.16.115(C), 25.20.110(D), 25.20.115(C), 25.22.130(E), 25.22.135(C), 25.24.080(E), 25.24.085(C) and subsection 106.6.1 of file. Section 106.6 of 1997 Seattle Building Code, various provisions relating to time periods for decisions on project permit applications (22.600, 23.66, 23.76, 25.12, 25.16, 25.20, 25.22, 25.24) 120166 Amends § 4.13.010, personnel (4.13) 120168 Amends § 18.28.010, parks and recreation fee schedule (18.28)120171 Amends § 3.35.060, P-Patch plot fees (3.35)

> Amends § 17.16.015, Seattle Center facility license fee schedule for 2001 and 2002 (17.16)

Amends §§ 21.28.040(B) and 21.76.040(A)(1), wastewater services (21.28,

Amends §§ 21.33.010(D), 21.76.040(A)(2) and 21.76.050, drainage services (21.33, 21.76)

Amends § 21.04.430, 21.04.440 and 21.76.040(A)(3), water services (21.04, 21.76)

6.310.230(A)(10)(a), 6.310.230(A)(15), 6.310.300(C)(1) and (2), 6.310.300(D), 6.310.350, 6.310.465(G) and 6.310.710, taxicabs and

Amends §§ 6.310.300(C)(6) and 6.310.430(A); repeals and

replaces §§

21.76)

ference only.

for-hire vehicles (6.310) For current the Office of the City Clerk

Text provided for hi

See ordinan sections for comp and tables and this source file.

Creates Department of Finance, Personnel Department and Fleets and Facilities Department; abolishes Executive Services Department and transfers functions; adds Ch. 3.18, Subchapter I; amends Chs. 3.18 and 3.38 titles, and §§ 3.14.620(B), 3.18.060, 3.18.140(B)(2), 3.18.160-3.18.200, 3.18.260, 3.18.280, 3.18.300, 3.38.010, 3.38.100(A) and (B), 3.38.800-3.38.840, 3.38.900(D), 3.38.902(1), (7), (12) and (19),

3.38.904(A)(1) and (D)-(G), 3.38.906(A), (B), and (C), 3.38.908(A)(1), (C), (D) and (E)-(G), 3.38.910(A) and (B),

3.38.912, 3.38.914, 3.38.918(A), 3.38.920, 3.38.924, 3.35.040(B) and (D), 3.114.010(B), 3.114.040(A), 3.114.050,

3.114.060(A), 3.114.090-3.114.120,

3.114.140, 3.114.150, 3.114.160(A), (D)(1)(a) and (E), 4.04.225(B), (E) and (F), 4.04.255,

4.14.100, 4.20.140, 4.20.300(C), 4.20.375,

4.44.070, 4.68.020, 4.68.030(A), 4.68.100(C) and (E), 4.68.110-4.68.130,

4.72.010(A)(3), (B)(5) and (B)(6), 4.72.020(A), 4.72.050-4.72.080,

5.22.010, 5.32.030(4), 5.32.130(A), 5.44.022(6) and (7), 5.48.020(6) and

(7), 5.52.010(B) and (C) [5.52.010(C) and (D)], 5.78.130, 6.02.030(A) and

(C), 6.202.060(6) and (7), 6.222.010(B) and (C),

6.270.030(C) and (D), 6.294.040(A) and (B), 6.310.110(G) and (H),

6.315.020(C) and (D), 7.04.130, 7.04.645(A), (B)

and (E), 7.04.650(A), 9.25.020(H) and (I),

10.02.090, 10.08.165(A), 11.30.280, 11.30.290(D), 15.32.300(C)(3), (C)(3)(j),

(k) and (m), 15.38.030(A), 20.38.005, 20.38.010(B),

20.38.020, 20.38.030, 20.40.010, 20.44.010(F),

20.44.030, 20.44.040,

20.44.060, 20.44.090, 20.44.100(C) and (F),

20.45.010(A), (C), and (D), 20.45.040, 20.46.050(D),

ference only.

For current the Office of

Text provided for

See ordinan sections for comp this source

5.32.160(C), 6.02.030(G), 6.294.040(I) and 20.44.120, various provisions relating to the organization of City government (3.14, 3.18, 3.35, 3.38, 3.76, 3.114, 5.52, 5.78, 6.02, 6.202, 6.222, 6.270, 6.310, 6.315, 7.04, 9.25, 10.02, 10.08, 11.30, 15.32, 15.38, 20.38, 20.40, 20.44, 20.45, 20.46, 20.46A, 20.48, 20.84, 21.52, 21.60, 22.900D, 23.44) § 5.48.020; amends §§ 5.44.444 and

4.04, 4.14, 4.20, 4.44, 4.68, 4.72, 5.22, 5.32, 5.44, 5.48, Adds subsection (A)(11) to 5.48.020(A)(3) and (A)(21) [(A)(22)]; renumbers former § 5.48.020(A)(11)-(23) to be 5.48.020(A)(12)-(24), utility taxes (5.44, 5.48) 120183 Creates arts account; amends 5.40.120, admissions tax (5.40) 120187 Establishes spending limits and adjusts pay bands in various compensation programs; amends § 4.20.430(B), personnel (4.20)120188 Amends § 4.20.375, parking of privately-owned motor vehicles by city officials (4.20) 120202 Amends 12A.06.180(A). (C) and (D), criminal code (12A.06)120208 Amends § 23.32.016, Official Land Use Map and § 23.47.040(C), land use and zoning (23.47) 120209 Adopts Ballard Municipal Center master plan design guidelines; amends §§ 23.41.010 and 23.41.012(B), land use and zoning (23.41) 120220 Adds § 21.49.045; amends § 21.76.050, utility rate assistance programs (21.49, 21.76) 120247 Adds § 21.49.081; amends §§ 21.49.030, 21.49.052, 21.49.055, 21.49.057 and

ference only. amending

Light Department (21.49) the Office of the City Clerk

21.49.058, Seattle City

June 2009 co

Text provide

See ordinant sections for and tables a this source

	ilhar 1	to file	
	4 4440012	TE 332	
	120250	Adds § 21.40.070; Amends	
		§§ 21.36.010-21.36.016,	
		21.40.050 and 21.40.060;	
	4 504 111	repeals § 21.40.040, solid	
٦	C IO	waste (21.36, 21.40)	
	120261		
	120261	Establishes new titles	
		and/or salaries, adjusts a	
		pay band and designates	
		exempt positions; amends	
		§ 4.13.010(7), personnel	
V	ces el es	(4.13)	
7	120263	Adds new section to Ch.	
	120203	21.60, cable	
		communications (21.60)	
	120266	Amends §§ 23.44.017(B),	
	1 40 00	23.45.100(C),	
	and to	23.45.112(A) and	
	'They	23.47.022(D), land use and	
		zoning (23.44, 23.45,	
	0210		
	<del>1110.</del>	23.47)	
	120267	Amends § 23.47.012, land	
		use and zoning (23.47)	
	120275	Amends § 21.49.040,	
		Seattle City Light	
		Department (21.49)	
	120278	Adds § 4.04.075,	
	120276	alternative dispute	
		resolution program (4.04)	
	120289	Repeals § 7.16.010	
		(Repealer)	
	120293	Adds §§	
		23.42.100-23.42.130;	
		amends §§ 23.44.008(F),	
		23.44.032, 23.45.006(A)	
		and (G), 23.48.038,	
		23.54.020(C), 23.72.014	
		and 23.84.026; repeals §§	
		23.43.080, 23.43.082,	
		23.44.080, 23.44.082,	
		23.45.180-23.45.190,	
		23.47.036, 23.47.038,	
		23.48.036, 23.49.028,	
		23.49.030, 23.50.008,	
		23.50.010, 23.69.016 and	
		23.69.017, land use and	
		zoning (23.42, 23.44,	
		23.45, 23.48, 23.54, 23.72,	
		23.84)	
	120298	Amends § 4.20.065,	
		personnel (4.20)	
	120301		
	120301	Amends § 21.40.070, solid	
	10000	waste (21.40)	
	120302	Adds § 22.206.305;	
		amends §§ 22.202.010,	
		22.206.180, 22.206.280	
		and 22.206.290, housing	
		code (22.202, 22.206)	
	120308		
	120308	Amends § 21.04.440(E),	
		wholesale water services	
		(21.04)	

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

deference only.

defere

June 2009 co

Text provide

See ordinant sections for and tables a this source

C	Sihor	40 Sile
i	4 mda	IG 111
	120318	Adds additional submittal
		period; amends §
	1 Cor MI	23.40.050(B), land use and
P	W IOI	zoning (23.40)
	120321	Amends § 2.04.370,
		election campaign
	120225	contributions (2.04)
	120325	Adds Ch. 3.51 and §
		17.08.015; amends §§
	nos crea	3.14.920, 9.25.025 and
Ŋ		21.60.060(A) and Res.
	-10	29532 and first resolution
	COMPLE	subsection of Res. 25534,
5	Com-2	Get Engaged pilot program (3.14, 3.51, 9.25, 17.08,
		21.60)
	120335	Amends § 21.52.210,
4	120333	conservation measures
		(21.52)
	120339	Extends through 6/30/02
	120337	moratorium on new adult
		cabarets (Ch. 6.270
		footnote)
	120340	Amends § 21.49.130, City
	120340	Light Department (21.49)
	120354	Establishes new title and
	120334	salary; designates exempt
		position; amends §
		4.13.010, personnel (4.13)
	120358	Adds § 4.36.150; amends
	120000	§§ 4.36.050(B), 4.36.200,
		4.36.210 and 4.36.400;
		renames table for §
		4.36.210(F) to be table for
		§ 4.36.210(B), City
		employees' retirement
		system (4.36)
	120372	Adds § 15.64.026,
		skybridge permits (15.64)
	120374	Adds § 23.47.023(D);
		amends §§ 23.47.004(E),
		23.47.006(B),
		23.47.009(B) and
		23.47.023(A), land use and
		zoning (23.47)
	120378	Amends 2000 Washington
		State Energy Code and §
		22.700.010, energy code
		(22.700)
	120379	Amends Seattle Building
		Code (22.100.010 footnote)
	120380	Amends Seattle
		Mechanical Code
		(22.400.010 footnote)
	120381	Amends §
		16.28.020(A)(2), harbor
		code (16.28)

eference only.

I amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	0.10
and a	te file
120385	Amends §§ 21.49.030(A),
12000	21.49.040(A), 21.49.52(A),
103	21.49.055(A) and (B),
4 FOT III	21.49.057(A),
ECI TOP	21.49.058(A), 21.49.081,
	City Light Department
	(21.49)
120388	Adds §§ 23.55.015 and
	23.55.050; amends §§
- a orea	15.12.010(A), 23.55.001,
res cross	23.55.014, 23.55.020(G),
	23.55.022(I), 23.55.024,
	23.55.028(E),
COMPA	23.55.030(E),
	23.55.034(E),
140 00	23.55.036(E), 23.69.021(E)
and to co	and 23.84.036, land use
CI BECO	and zoning (15.12, 23.55,
	23.69, 23.84)
120408	Amends §§ 21.04.030(A)
33300	[21.04.430(A)] and
	21.04.440(A), water
	services (21.04)
120410	Adds Ch. 25.11; amends §§
	23.41.012(B),
	23.41.016(B),
	23.44.014(D) and
	23.45.056(D); repeals §
	23.45.014(I), land use,
	zoning and tree protection
	(23.41, 23.44, 23.45, 25.11)
120411	Authorizes sale of
120411	city-owned property;
	amends §§ 5.06.030(B),
	5.80.020 and 5.80.030, city
	investments and
	cumulative reserve subfund
	(5.06, 5.80)
120413	Amends § 21.49.130, City
	Light Department (21.49)
120422	Amends § 18.28.010, parks
	and recreation fee schedule
	(18.28)
120427	Amends § 1 of Ord.
	120408, water services
	(21.04)
120433	Amends § 2.04.180,
	election cash contributions
120111	(2.04)
120441	Amends §§ 5.44.022,
	5.44.026, 5.44.470,
	taxation of distribution
	cooperatives and affiliates
	(5.44)

eference only.

d amending
graphics,
ccuracy of

Seattle Municipal Code 120443 Text provided for hi

See ordinan sections for and tables and this source file. Adds Ch. 23.49 Exhibit Maps 1A, 1B, 1E, 1F, 1G, 1H, 1I, 1J, 1K, 1L, 1M, 1N and 10, §§ 23.49.012-013, 23.49.014, 23.49.025, 23.49.027, 23.49.039, 23.49.108; amends §§ 23.41.012, 23.49.008, 23.49.009, 23.49.016, 23.49.026(A), 23.49.032, 23.49.041(E), 23.49.045(A),

23.49.046(B), 23.49.056, 23.058(A) and (B), 23.49.064(A) and (B), 23.49.066(B),

23.49.076(A) and (B), 23.49.078(A) and (B), 23.49.090(C), 23.49.096(B), 23.49.106,

23.49.120, 23.49.122(B), 23.49.126(A), 23.49.134(A) and (B),

23.49.136(A), (B) and (C), 23.49.162(B), 23.49.166(B),

23.49.248(B),

23.49.322(A) and (B), 23.49.332, 23.49.336(A)

and (B), 23.49.338(A) and

(B), 23.84.006(C),

23.84.016(H), 23.84.024(L),

23.84.025(M),

23.84.030(P),

23.84.038(T); renumbers § 23.49.012 as § 23.49.019, §

23.49.014 as 23.49.017;

repeals and replaces §

23.49.011; repeals Ch.

23.49 Exhibit Maps IA, IB,

IE, IIA, IIB, IIC, IID, IIIA, IIIB, IIIC, IVA, IVB, IVC,

VA, VB, VC, VD, VE,

VIA, VIB, BIC, VID, IXA, XIA and XIIA, §§

23.49.048, 23.49.050,

23.49.052, 23.49.054,

23.49.068, 23.49.070,

23.49.072, 23.49.074,

23.49.098, 23.49.100,

23.49.102, 23.49.104,

23.49.106, 23.49.124,

23.49.128, 23.49.132,

23.49.150, 23.49.152, 23.49.154, 23.49.160,

23.49.180, 23.49.210,

23.49.212, 23.49.214,

23.49.238, 23.49.240,

23.49.304, 23.49.328,

23.49.330, 23.49.331,

23.49.342, 23.49.344, land

use and zoning (23.41,

ference only. amending

23.49, 23.84) the Office

Seattle Municipal Code June 2009 co

Soattle Munic	30	to file
Seacos	120447	Amends §§ 23.41.010,
7.009 CU		23.41.012, land use and
June 2009	acor hi	zoning (23.41)
- aride	120448	Amends §§ 22.900A.050, 22.900C.010,
Toyt provide		22.900C.010, 22.900D.010(A) and
Text provide		Tables C-1, D-2, D-3 and
		(J), 22.900D.080,
		22.900D.090 and Table
10 000	nos crea	D-8, 22.900D.110(B),
a ardinally		22.900D.145 and Table D-SR, 22.900D.150
See of asse	100	andTables D-15 and D-16,
- 501	COMPA	22.900D.160(D),
cections for		22.900E.010, 22.900E.030
	ad to co	and Table E-2, 22.900F.020; repeals §§
4 4 4 1 1 2 3	Ma co	22.900D.150(E),
and tables a		25.08.570, Department of
Oliver and the second s	file.	Design, Construction and
this source	11100	Land Use fees (22.900A,
this some		22.900C, 22.900D, 22.900E)
	120451	Amends § 16.20.130,
		Harbor Code speed
		regulations (16.20)
	120452	Adds Ch. 23.61, §§
		23.34.089, Station Area Overlay District; amends
		§§ 23.47.004(E),
		23.47.008(D) and
		23.47.009(B), transit, land
		use and zoning (23.34, 23.47)
	120453	Amends Ch.23.32, Official
		Land Use Map Plats 51E
		and 51W, § 23.47.040(C);
		repeals § 23.70.004
		(Exhibit 1F), transit, land use and zoning (23.32,
		23.47, 23.70)
	120454	Amends Ch. 23.32, Official
		Land Use Map Plats 14E,
		15W, 23E and 24W (Exhibits A and B), §
		23.47.040(C); repeals §
		23.70.004 (Exhibit 1J),
		transit, land use and zoning
	120455	(23.32, 23.47, 23.70)
	120455	Amends Ch. 23.32, Official Land Use Map Plats 52W,
		52E, 58W and 58E, §
		23.47.040(C); repeals §
		23.70.004 (Exhibit 1E),
		transit, land use and zoning
		(23.32, 23.47, 23.70)

ference only. d amending graphics, accuracy of

June 2009 co Text provide

See ordinant sections for and tables a this source

	0210
a mada	te file
120456	Amends Ch. 23.32, Official
	Land Use Map Plats 80W
- 0 - m hi	and 80E, § 23.47.040(C);
A FOR III	repeals § 23.70.004
	(Exhibit 1A), transit, land
	use and zoning (23.32,
	23.47, 23.70)
120457	Amends Ch. 23.32, Official
	Land Use Map Plats 73W
oc crea	and 73E, § 23.47.040(C);
ces es	repeals § 23.70.004
	(Exhibit 1B), transit, land
	use and zoning (23.32,
COLLIE	23.47, 23.70)
120458	Amends Ch. 23.32, Official
140 00	Land Use Map Plats 58E,
ma to	59W, 65E and 66W, §§
	23.47.040(C); repeals §
	23.70.004 (Exhibit 1D),
\$112.	transit, land use and zoning
73700	(23.32, 23.47, 23.70)
120459	Amends Ch. 23.32, Official
	Land Use Map Plats 36W,
	36E, 40W and 40E; repeals
	§ 23.70.004 (Exhibit 1I),
	transit, land use and zoning
120460	(23.32, 23.70)
120460	Amends Ch. 23.32, Official
	Land Use Map Plats 40W
	and 40E, § 23.47.040(C); repeals § 23.70.004
	(Exhibit 1H), transit, land
	use and zoning (23.32,
	23.47, 23.70)
120466	Adds § 23.55.005, video
120100	display methods; amends
	§§ 23.55.003(A),
	23.55.030(B),
	23.55.034(C),
	23.55.036(B), 23.55.040,
	23.66.160(A),
	23.66.338(E),
	23.69.021(A) and
	23.84.036, signs (23.55,
	23. 66, 23.84)
120467	Amends §§ 5.06.010,
	5.06.030, 5.06.040 and
	5.06.050, investment
	policies (5.06)
120479	Amends §§ 3.58.040 and
	1
	3.58.080, Seattle Design

eference only.

d

ial amending
igraphics,
d, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Adds § 11.20.060, intermediate license restrictions; amends § 11.20.230(B), 11.13.120(A), 11.56.20(N) and (P), 11.56.320(C) and (D), 11.56.350, 11.57.25, 08, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)	Cihar	4- 6110
intermediate license restrictions; amends §§ 11.20.230(B), 11.13.120(A), 11.56.020(N) and (P), 11.56.320(C) and (D), 11.56.320(C) and (D), 11.56.350, 11.57.280, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  120483	a - mada	te me
restrictions; amends § \$ 11.20.230(B), 11.13.120(A), 11.56.320(C) and (D), 11.56.320(C) and (D), 11.56.350, 11.57.280, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  Amends § 4.13.010, City salaries and titles (4.13)  120483 Amends § 4.40.020 and 4.44.060, General Fund and subfunds (4.40, 44.4)  12049 Amends § 3.20.320, City TDR bank (3.20)  Amends § \$ 3.20.320, City TDR bank (3.20)  Amends § \$ 5.24.005, City TDR bank (3.20)  Amends § \$ 5.24.005, claims investigation (5.24)  Adds to § 22.700.01  Washington State Energy Code § \$ 141.34, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-11J, 14-1K, 14-1IL, 14-1M; amends § 22.700.010 Washington State Energy Code § \$ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322., 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1413.1, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.3, 1413.1, 1413.1, 1416, 1421.1, 1423, 1431.3, 1431.2, 1432.2, 1433, 1440, 1452. 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120535 Adds § 23.54.020(J), parking spaces (23.54)  Amends § 3.06.055, 22.02.050 and 16.08.050; repeals § 12.81.00, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy	120481	
11.20.230(B), 11.13.120(A), 11.56.020(N) and (P), 11.56.320(C) and (D), 11.56.320(C) and (D), 11.56.350, 11.57.280, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  120483 Amends § 4.13.010, City salaries and titles (4.13)  Amends § 4.40.020 and 4.44.060, General Fund and subfunds (4.40, 4.44)  120494 Amends § 3.20.320, City TDR bank (3.20)  120521 Amends § 5.24.005(B) and (C), 5.24.020, 5.24.030, 5.24.040 and 5.24.060; repeals § 5.24.050, claims investigation (5.24)  120525 Adds to § 22.700.01 Washington State Energy Code § 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  Amends Sh. 20.28, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  Amends Ch. 21.52, low-income energy		
11.13.120(A), 11.56.020(N) and (P), 11.56.320(C) and (D), 11.56.320(C) and (D), 11.56.350, 11.57.280, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  120483	1 Cor hi	
11.56.020(N) and (P), 11.56.320(C) and (D), 11.56.320(C) and (D), 11.56.350, 11.57.280, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  120483 Amends § 4.13.010, City salaries and titles (4.13)  120489 Amends §§ 4.40.020 and 4.44.060, General Fund and subfunds (4.40, 4.44)  120494 Amends §§ 5.24.005(B) and (C), 5.24.005(B) and (C), 5.24.005(B) and (C), 5.24.005(B) and (C), 5.24.005(B) and (C), 5.24.005 (claims investigation (5.24)  120525 Adds to § 22.700.01 Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1I, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322.3, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  120537 Amends § 3.06.055, 22.02.050 and 16.08,050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy	ad in	` //
11.56.320(C) and (D), 11.56.350, 11.57.280, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  120483		` ''
11.56.350, 11.57.280, 25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  120483		` ' ` ' '
25.08.430 and 25.08.450, motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)  Amends § 4.13.010, City salaries and titles (4.13)  Amends § 4.40.020 and 4.44.060, General Fund and subfunds (4.40, 4.44)  120494 Amends § 3.20.320, City TDR bank (3.20)  Amends § 5.24.005(B) and (C), 5.24.005(B) and (C), 5.24.005, 5.24.005 (claims investigation (5.24)  Adds to § 22.700.01  Washington State Energy Code § 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code § 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120535 Adds § 23.54.020(J), parking spaces (23.54)  Amends § 3.06.055, 22.00.16.08  120538 Amends Ch. 21.52, low-income energy		
motor vehicles (11.13, 11.20, 11.56, 11.57, 25.08)		
11.20, 11.56, 11.57, 25.08)  Amends § 4.13.010, City salaries and titles (4.13)  120489  Amends §§ 4.40.020 and 4.44.060, General Fund and subfunds (4.40, 4.44)  120494  Amends § 3.20.320, City TDR bank (3.20)  120521  Amends §§ 5.24.005(B) and (C), 5.24.020, 5.24.030, 5.24.040 and 5.24.060; repeals § 5.24.050, claims investigation (5.24)  120525  Adds to § 22.700.01  Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1I, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(I), parking spaces (23.54)  Amends § 8.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy	- 02023	
120483	ces cres	
Salaries and titles (4.13)	120483	
Amends §§ 4.40.020 and 4.44.060, General Fund and subfunds (4.40, 4.44)		
and subfunds (4.40, 4.44) Amends § 3.20.320, City TDR bank (3.20)  120521  Amends § 5.24.005(B) and (C), 5.24.020, 5.24.030, 5.24.040 and 5.24.060; repeals § 5.24.050, claims investigation (5.24)  120525  Adds to § 22.700.01 Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  Amends § 8.306.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy	120489	Amends §§ 4.40.020 and
Amends § 3.20.320, City TDR bank (3.20)		4.44.060, General Fund
TDR bank (3.20)  Amends §§ 5.24.005(B) and (C), 5.24.020, 5.24.030, 5.24.040 and 5.24.060; repeals § 5.24.050, claims investigation (5.24)  Adds to § 22.700.01  Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120537  Amends § \$3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy	44000	and subfunds (4.40, 4.44)
Amends § § 5.24.005(B) and (C), 5.24.020, 5.24.030, 5.24.040 and 5.24.060; repeals § 5.24.050, claims investigation (5.24)  120525  Adds to § 22.700.01  Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § \$ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revoue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy	120494	
and (C), 5.24.020, 5.24.040 and 5.24.060; repeals \$ 5.24.050, claims investigation (5.24)  120525 Adds to \$ 22.700.01  Washington State Energy Code \$\$ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends \$ 22.700.010 Washington State Energy Code \$\$ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends \$ 5.40.028, human services agencies (5.40)  120537 Adds \$ 23.54.020(J), parking spaces (23.54)  120538 Amends \$ 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy	, 8800	
5.24.030, 5.24.040 and 5.24.060; repeals \$ 5.24.050, claims investigation (5.24)  120525 Adds to \$ 22.700.01  Washington State Energy Code \$\$ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1B, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends \$ 22.700.010 Washington State Energy Code \$\$ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends \$ 5.40.028, human services agencies (5.40)  120537 Amends \$ 3.06.055, 22.202.050 and 16.08.050; repals \$ 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy	120521	
5.24.060; repeals § 5.24.050, claims investigation (5.24)  120525  Adds to § 22.700.01  Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy	sile.	
5.24.050, claims investigation (5.24)  Adds to § 22.700.01 Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § 8.306.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy	3330	-
investigation (5.24)  Adds to § 22.700.01  Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120537  Amends § 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
Adds to § 22.700.01 Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
Washington State Energy Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy	120525	
Code §§ 1413.4, 1436.2, 1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  120535 Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy	120323	ŭ
1436.3, 1540 and Tables 14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		0,
14-1A, 14-1B, 14-1C, 14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy		
14-1D, 14-1E, 14-1F, 14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
14-1J, 14-1K, 14-1L, 14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
14-1M; amends § 22.700.010 Washington State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
State Energy Code §§ 1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1132.2, 1132.3, 1133, 1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		22.700.010 Washington
1301, 1310.2, 1312.2, 1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40) Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1322, 1323, 1323.3, 1331, 1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1333, 1401, 1411.1, 1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40) Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § \$ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1411.4, 1412.2, 1412.4, 1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § 8 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1412.6, 1412.8, 1413, 1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends § 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1413.2, 1413.3, 1414.1, 1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40) Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1416, 1421.1, 1423, 1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40) Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends § \$ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		4 4 4 9 9 4 4 4 9 9 4 4 4 4 4
1431.2, 1432.2, 1433, 1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends § \$ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1436, 1437, 1438, 1440, 1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § § 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  Amends Ch. 21.52, low-income energy		
1452, 1501, 1512, 1512.2, 1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § § 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy		
1513.1, 1513.3, 1513.5, 1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends § \$ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy		
1513.6, 1521, 1530, 1532 and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  120535 Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy		
and Tables 10-5B, 10-6, 13-1 and 15-1, energy (22.70)  120533  Amends § 5.40.028, human services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends § § 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy		
(22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  120535 Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy		
(22.70)  120533 Amends § 5.40.028, human services agencies (5.40)  120535 Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy		13-1 and 15-1, energy
services agencies (5.40)  120535  Adds § 23.54.020(J), parking spaces (23.54)  120537  Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy		
120535 Adds § 23.54.020(J), parking spaces (23.54)  120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy	120533	· · · · · · · · · · · · · · · · · · ·
parking spaces (23.54)  120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy		
120537 Amends §§ 3.06.055, 22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538 Amends Ch. 21.52, low-income energy	120535	· //
22.202.050 and 16.08.050; repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy	100707	
repeals § 18.28.100, special revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy	120537	,
revenue funds (3.06, 22.20, 16.08)  120538  Amends Ch. 21.52, low-income energy		· · · · · · · · · · · · · · · · · · ·
16.08) 120538 Amends Ch. 21.52, low-income energy		-
120538 Amends Ch. 21.52, low-income energy		·
low-income energy	120529	,
	120538	
Conscivation (21.32)	- CA CM	
	Press Sw	consciyation (21.32)

d amending graphics, accuracy of

For current Six conservation (21.52)

The Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

Cipus Cipus	te file
120541	Amends § 23.54.015(C)-(I)
120341	and Chart A, parking
103	requirements (23.54)
120561	Amends official Land Use
120301	Map (23.32)
120573	Amends § 11.72.440,
	parking enforcement
	(11.72)
120574	Amends § 11.70.200, street
- a crea	and alley parking (11.70)
120581	Establishes new titles
120001	and/or salaries, adjusts a
10	pay band and designates
COMPLE	exempt positions; amends
603558	§ 4.13.010(8), personnel
	(4.13)
120591	Amends § 21.36.016,
12031	definitions, and §
	21.40.070, commercial
0210	
TIIC.	collection rates (21.36,
120594	21.40)
120594	Amends § 18.12.080(B),
	dogs in Magnuson Park
120605	(18.12)
120605	Amends § 25.05.675,
	specific environmental
120101	policies (25.05)
120606	Adds § 10.02.025; amends
	§§ 10.02.010, 10.02.020,
	10.02.060, 10.02.070,
	10.02.100, 12A.26.040,
	responses to civil
	emergencies (10.02,
120607	12A.26)
120607	Amends § 15.62.090,
	compensation for vacation
120608	(15.62) Amends § 23.45.008, land
120006	use and zoning (23.45)
120609	Adds §§ 23.24.050(C),
120009	
	23.34.007(G); amends §§
	23.34.007(A),
	23.34.008(A),
	23.42,106(A), (B) and (D),
	23.44.012(C),
	23.45.009(D), 23.47.004 Chart A 23.47.012(G)
	Chart A, 23.47.012(G),
	23.47.014(F), 23.47.016(D) and (E), 23.47.042(D),
	` '/'
	23.50.026(C), 23.54.015(I),
	23.55.020(D), 23.61.008, 23.76.036(B), land use and
	zoning (23.24, 23.34,
	23.42, 23.44, 23.45, 23.47,
	23.50, 23.54, 23.55, 23.61,
	23.76)

eference only.

I amending graphics, ccuracy of

June 2009 co 120611 upus Text provide d for hi

See ordinances cresections for compand tables and to this source file.

120611 Amends §§ 15.16.130, sidewalk cafe notice, 23.32.016, 23.41.004(A) and (B), 23.41.012(B), 23.43.040(A), 23.45.018(B), 23.47.028(C), 23.48.032(B), 23.49.018(A), 23.49.332(A), (C) and (E), 23.50.002, 23.53.020(E), 23.55.036(D), 23.66.130(A), 23.66.170(D), 23.76.006(B), 23.84.016, 23.84.024, land use code and zoning, 25.05.675 Exhibit 1; repeals and replaces Ch. 23.49 Map 1N; repeals Ch. 7.16, obsolete provision, § 23.84.036, "single family attached structure" (7.16, 15.16, 23.32, 23.41, 23.43, 23.45, 23.47, 23.48, 23.49, 23.50, 23.53, 23.55, 23.66, 23.76, 23.84, 25.05) 120615 Amends §§ 21.28.040(B), 21.76.040(A)(1), wastewater services (21.28, 21.76) Adds § 21.49.083, 120623 voluntary green power program (21.49) 120626 Adds § 4.50.020; amends §§ 4.50.010(A) and (D), health care benefits (4.50) 120631 Adds § 15.52.005, definitions; amends §§ 15.52.010, 15.52.030, 15.52.040, 15.52.050, 15.52.060, special events 120638 Amends § 18.28.010, parks and recreation fee schedule (18.28)120639 Amends § 22.602.040, fire code fees (22.602) 120644 Amends § 5.40.120, admission tax receipts (5.40)120647 Amends § 5.48.050, water business tax (5.48) 120649 Amends § 4.13.010, civil service exemptions (4.13) 120653 Amends § 4.20.300,

eference only.

l amending

graphics,

For current SMC, Condense of the City Clerk

120654

payment for out-of-class

Amends § 21.04.440, water rates and charges (21.04)

duties (4.20)

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipar o	0210
a and	ate file
120655	Amends § 3.33.040, municipal court structure
120656	(3.33) Amends §§ 4.36.030,
120030	4.36.050, 4.36.115 and
	4.36.215, retirement
	benefit adjustments (4.36)
120658	Amends §§
- 0002	4.04.020-4.04.050, public safety civil service
ces cree	employees, 4.08.020,
	4.08.050, 4.08.070,
	4.08.090, 4.08.110, civil
COMPA	service commission (4.08)
120661	Adds § 23.47.004, Exhibits
ad to C	A and B; amends § 23.47.004(E), land use and
Her co	zoning (23.47)
120664	Amends §§ 20.49.010,
sile.	20.49.020, 20.49.040,
7770	20.49.050, 20.49.060, city
120667	contracting (20.49)
120667	Amends § 21.49.058, City Light Department rates
	(21.49)
120668	Adds Ch. 5.30, § 5.40.005,
	Ch. 5.45, §§ 5.48.015,
	5.52.005, Ch. 5.55; amends
	§§ 3.02.125(A), 3.14.750,
	3.38.806(H), 5.32.020,
	5.32.150-5.32.170, 5.32.200, 5.40.010,
	5.40.025, 5.40.060,
	5.40.070, 5.40.080,
	5.48.020, 5.48.050,
	5.48.055(C), 5.48.060,
	5.48.070, 5.48.260, 5.52.010, 6.20.030(A),
	6.20.040(B)(1) and (2),
	6.20.050(A) and (B),
	6.288.100(B); repeals §§
	5.32.010,
	5.32.030-5.32.140,
	5.32.180, 5.40.075, 5.40.100, 5.40.110,
	5.40.135, 5.40.140,
	5.40.150; Ch. 5.44, §§
	5.48.030, 5.48.040,
	5.48.080-5.48.220,
	5.48.270;
	5.52.040-5.52.160, revenue code, licenses and taxes
	(3.02, 3.14, 3.38, 5.30,
	5.32, 5.38, 5.40, 5.44, 5.48,
	5.52, 6.20, 6.288)
120684	Adds §§ 4.36.101,
	4.36.124, 4.36.191,
	4.36.192, 4.36.195(H); amends §§ 4.36.030,
	4.36.190(B), 4.36.400, City
	employees retirement
Loss t SI	system (4,36)
Leme	004-0

eference only.

d amending
graphics,
ccuracy of

For current SN system (4,36)
the Office of the City Clerk

Seattle Municipal Code ference only. date 120685 Amends § 4.36.215, cost of living adjustment (4.36) 120691 Adds § 23.71.001, Text provide Northgate comprehensive plan; amends §§ 23.20.008, 23.24.040(A), 23.34.072(C), 23.34.090(F). 23.34.124(A), (B) and (D), See ordinances 23.40.020(C), 23.44.036(D), 23.47.006(C), sections for comp 23.47.007(A), 23.49.036(E), 23.49.037(B), 23.50.015(A), and tables and 23.54.020(C). 23.54.030(F), 23.60.060(B), this source file. 23.60.220(C), 23.67.040(C), 23.69.002, 23.69.006(B), 23.69.024(B), 23.69.030(E), 23.69.032(D) and (E), 23.69.035(H), 23.69.036(B),

23.76.023(A), 23.76.036, 23.76.050(A), 23.79.008(C); repeals Ch. 23.12 and § 23.34.008(J), land use and zoning (23.24, 23.34, 23.40, 23.44, 23.47, 23.49, 23.50, 23.54, 23.60, 23.67, 23.69, 23.76, 23.79) 120692 Amends § 25.05.675, environmental policies (25.05)120694 Amends §§ 23.340.020 and 23.34.022, land use and zoning (23.34) 120705 Amends §§ 4.100.010 and 4.100.020, salary reduction agreements (4.100) 120718 Amends § 4.20.430, Information Technology Professional (ITP) Compensation Program (4.20)120725 Establishes new titles and/or salaries, designates exempt positions; amends § 4.13.010, city employment (4.13)

Seattle Municipal Code June 2009 co Text provide See ordinan

-4410 MUIII	218	a cale
eattle	a - manda	te me
	120728	Adds §§ 3.28.800,
2009		3.28.805, 3.28.810,
IIIIe Mo	a coss hi	3.28.815, 3.28.820,
ext provide	d tor "	3.28.825, 3.28.830,
arovius		3.28.850, 3.28.855,
ext by		3.28.860, 3.28.865,
		3.28.870, 3.28.900,
		3.28.905, 3.28.910,
	- 2000	3.28.915 and 3.28.920;
	age crea	repeals §§
- adinan		3.28.600-3.28.760, Office of Professional
See Orussia	-10	Accountability (3.28)
	120736	Adds § 3.42.050 and Ch.
cotions for	120/36	
coctions in		3.123, city records management program
266020	ad to co	(3.42, 3.123)
and tables a	120747	Amends § 4.24.035, paid
and tables	120747	sick leave (4.24)
	120773	Adds § 3.14.502; amends
	120//3	§§ 3.06.010, 3.06.030,
Abic source		3.12.030, 3.14.510,
		3.14.521, 3.14.530,
		3.20.300, 3.20.310,
		3.35.010, 3.35.030,
		3.38.010, 3.64.040,
		4.13.010, 5.44.200(B) and
		5.48.120(D); repeals §§
		3.14.501, 3.14.540,
		3.14.800, 3.14.810,
		3.14.820
	120775	Amends §§ 21.60.800,
		21.60.810, 21.60.820 and
		21.60.830, cable customer
		bill of rights (21.60)
	120782	Amends §§ 3.38.100 and
		20.48.010, small
		construction projects roster
		(3.38, 20.48) and 3.14.830,
		various provisions relating
		to the organization of the
		executive branch of city
		government (3.06, 3.12,
		3.14, 3.20, 3.35, 3.38, 3.64, 4.13, 5.44, 5.48)
	120785	4.13, 5.44, 5.48) Amends § 23.41.010, land
	120/63	
	120790	use and zoning (23.41) Amends § 4.13.010(15),
	120/90	Amends § 4.13.010(13), personnel (4.13)
	L	personner (4.13)

reference only. d amending graphics, ccuracy of

Seattle Municipal Code eference only. June 2009 co 120794 Adds §§ 3.04.010, 3.04.020, 3.04.030, Text provided for hi 3.04.040, 3.04.050, 3.04.101, 3.04.201, 3.114.010(E) and 5.40.010(E) and (F); amending amends §§ 2.04.060(K), 3.02.090(A), 3.06.055(A), 3.14.915(B), 3.18.180, See ordinances crea 3.28.010(D), 3.28.180, 3.30.110, 3.30.120, 3.33.060, 3.33.090, sections for comple 3.33.160(A), 3.33.170, 3.38.010, 3.38.060, 3.38.065, 3.42.010, and tables and to 3.56.060(A), 3.63.020, 3.70.100(H), 3.90.010, 3.90.050, 3.90.060, 3.106.010, 3.108.010, this source file. 3.114.030(B), 3.114.040(A), 3.114.050, 3.114.060(A), 3.114.090, 3.114.100, 3.114.110, 3.114.120, 3.114.130, 3.114.140, 3.114.150, 3.114.160, 3.121.020, 3.121.040(B), 3.124.040(B), 4.08.150, 4.14.100, 4.14.120, 4.14.150(B), 4.20.110, 4.20.160, 4.20.325(A), 4.20.370, 4.20.375, 4.24.080, 4.24.090, 4.28.030, 4.36.110(B), 4.36.125(D), 4.36.130(E) and (H), 4.36.135, 4.36.140(A), 4.40.040,

> 4.44.010, 4.44.070, 4.68.140, 4.72.010, 4.72.020(A), 4.72.050, 4.72.060, 4.72.070, 4.72.080, 4.88.010(B), 4.96.030(A), 5.04.030, 5.04.040, 5.06.010, 5.06.030, 5.06.040, 5.06.050, 5.12.010, 5.12.020, 5.12.030, 5.12.040, 5.12.050, 5.14.020, 5.14.030, 5.14.040, 5.14.050, 5.14.060, 5.16.010, 5.16.030, 5.16.040, 5.16.050, 5.20.010, 5.20.020, 5.20.030, 5.20.040, 5.22.010, 5.24.005, 5.24.020.

Seattle Municipal Code date June 2009 co 5.24.030, 5.24.040, 5.24.060, 5.30.025(10), Text provided for hi 5.30.035(8), 5.30.060(7), 5.40.028(F), 5.40.060, 5.40.070, 5.40.080, 5.40.085(A), 5.40.090, 5.45.100(P), (A)(21), 5.52.010(C) and (D), 5.55.120(E), 5.55.140,

sections for comple and tables and to this source file.

See ordinances crea

5.48.020(A)(6), (A)(7) and 5.55.150(A), 5.55.165, 5.55.200(C), 5.55.230(B), 5.55.260, 5.56.110, 5.56.130, 5.64.030, 5.78.010, 5.78.020, 5.78.030, 5.78.040, 5.78.060, 5.78.070, 5.78.080, 5.78.090, 5.78.100, 5.78.120, 5.78.130, 5.78.150, 5.78.160, 5.78.170, 5.78.180, 5.78.190, 6.02.030(A) and (C), 6.02.040(A), 6.02.060, 6.02.080(A), 6.02.090, 6.02.120, 6.02.150, 6.02.170, 6.02.190, 6.02.200, 6.02.290(A), 6.14.020, 6.14.090, 6.14.100, 6.36.020(A) and (C), 6.36.030, 6.36.040, 6.38.040, 6.42.010, 6.42.020, 6.48.040(E), 6.48.050, 6.48.080, 6.102.020, 6.102.070, 6.102.090, 6.202.060(A)(6) and (A)(7), 6.202.460, 6.222.010(B) and (C), 6.270.030(C) and (D), 6.270.040, 6.270.050, 6.270.070(A), 6.294.040(A) and (B), 6.294.140(B) and (C), 6.310.110(G) and (H), 6.315.020, 7.04.130, 7.04.570, 7.08.010, 7.20.080(B), 9.25.020(H) and (I), 9.25.025(D), 9.25.030(A)(13),

eference only. amending

Seattle Municipal Code date eference only. June 2009 co 9.25.130, 10.02.030(B), 10.02.070, 10.02.080, Text provided for hi 11.16.420, 11.16.480, 11.23.140, 11.30.220, 11.30.280, 11.30.290(D), 11.30.320, 12A.08.105, amending 12A.62.020, 15.04.040(A), 15.04.044(B), 15.04.050, 15.04.090, 15.38.030(A), See ordinances crea 15.62.110, 16.08.050(A), 18.12.080(D), 20.04.040, 20.04.050, 20.04.060, sections for comple 20.04.120, 20.04.130, 20.04.140, 20.04.150, 20.04.210(C) and (G), and tables and to 20.04.220, 20.04.230, 20.04.270(H), 20.04.280, 20.04.290, 20.08.030, 20.12.080, 20.38.005, this source file. 20.38.010(B), 20.38.020, 20.38.030, 20.40.010, 20.40.020, 20.44.010(F), 20.44.030, 20.44.040, 20.44.060, 20.44.090, 20.44.100(C) and (F), 20.45.010(C) and (D), 20.45.040, 20.46A.050, 20.46A.300, 20.48.030, 20.48.040, 20.48.060, 20.49.020, 20.49.040, 20.76.020, 20.76.040,

20.76.060, 20.76.070, 20.76.080, 20.76.090, 20.76.120, 20.76.130, 20.76.140, 20.76.200, 20.76.230, 20.76.240, 20.76.260, 20.76.260, 20.76.270, 20.76.280, 20.80.010,

Seattle Municipal Code eference only. date June 2009 co 20.80.020, 20.80.030, 20.84.030, 21.04.020, Text provided for hi 21.04.050, 21.04.260, 21.16.050(B), 21.16.060(A), 21.16.110, 21.16.160, 21.16.180, amending 21.16.280, 21.24.070, 21.24.090, 21.28.010, 21.28.020, 21.28.250, See ordinances crea 21.28.260, 21.28.350, 21.36.087(A), 21.40.030, 21.40.120(A), 21.72.010, sections for comple 21.76.060, 22.208.110(C) and (D), 22.220.100(C), 22.904.060, 22.904.070, and tables and to 23.22.066(C), 23.45.094(A)(2)(b)(9), 23.45.096(E)(1)(c), 25.05.914(D) and this source file. 25.28.350(B); amends and renumbers §§ 3.38.030, 3.38.040, 3.38.070, 3.38.080, 3.38.100, 3.38.800, 3.38.802, 3.38.806, 3.38.808, 3.38.810, 3.38.812, 3.38.814, 3.38.816, 3.38.818, 3.38.820, 3.38.822, 3.38.824, 3.38.826, 3.38.830, 3.38.832, 3.38.840, 3.38.900, 3.38.902, 3.38.904, 3.38.906, 3.38.907, 3.38.908, 3.38.910, 3.38.912, 3.38.914, 3.38.916, 3.38.918, 3.38.920, 3.38.922, 3.38.924 and 3.38.926 as 3.04.060, 3.04.070, 3.04.080,

3.04.090, 3.04.095, 3.04.100, 3.04.102, 3.04.106, 3.04.112, 3.04.114, 3.04.116, 3.04.112, 3.04.122, 3.04.124, 3.04.126, 3.04.132, 3.04.132, 3.04.132, 3.04.132, 3.04.132, 3.04.140, 3.04.200, 3.04.202, 3.04.204, 3.04.206, 3.04.207, 3.04.208, 3.04.210, 3.04.212, 3.04.214, 3.04.216,

3.04.218.

June 2009 co Text provide

See ordinan sections for and tables a this source

date file toric reference only. 3.04.220, 3.04.222, 3.04.224 and 3.04.226; repeals §§ 3.38.050, 21.101.010, 21.101.020, 21.101.030, 21.101.040, 21.101.050, 21.101.060, ing and amending 21.101.070, 21.101.080, 21.101.090 and 21.101.100; removes Ch. 20.46 from codification, executive branch of city government (2.04, 3.02, ccuracy of 3.04, 3.06, 3.14, 3.18, 3.28, 3.30, 3.33, 3.38, 3.42, 3.63, 3.70, 3.90, 3.106, 3.108, 3.114, 3.121, 3.124, 4.08, 4.14, 4.20, 4.24, 4.28, 4.36, 4.40, 4.44, 4.68, 4.72, 4.88, 4.96, 5.04, 5.06, 5.12, 5.14, 5.16, 5.20, 5.22, 5.24, 5.30, 5.40, 5.45, 5.48, 5.52, 5.55, 5.56, 5.64, 5.78, 6.02, 6.14, 6.36, 6.38, 6.42, 6.48, 6.102, 6.202, 6.222, 6.270, 6.310, 6.315, 7.04, 7.08, 7.20, 9.25, 10.02, 11.16, 11.23, 11.30, 12A.08, 12A.62, 15.04, 15.38, 15.62, 16.08, 18.12, 20.04, 20.08, 20.12, 20.38, 20.40, 20.44, 20.45, 20.46A, 20.48, 20.49, 20.76, 20.80, 20.84, 21.04, 21.16, 21.24, 21.28, 21.36, 21.40, 21.72, 21.76, 22.208, 22.220, 22.904, 23.22, 23.45, 25.05, 25.28) 120804 Adopts 2001 Washington State Energy Code and local amendments; amends § 22.700.010, energy efficiency and energy conservation (22.700) 120805 Amends § 6.288.065, pawnshops (6.288) 120811 Amends §§ 21.49.030 and 21.49.040, light residential electric rate schedules (21.49)Amends §§ 21.04.430 and 120817 21.04.440, water rates and charges (21.04)

Seattle Municipal Code reference only. 120818 Amends §§ 22.900A.060(B), Text provided for hi 22.900B.020(D), 22.900B.060(A), 22.900C.010(D), 22.900C.010(H), 22.900D.010, 22.900D.060(D), 22.900D.090, See ordinances 22.900D.100, 22.900D.110, 22.900D.140, sections for comp 22.900D.145, 22.900D.150. 22.900D.160(H) and (I), 22.900E.020, 22.900E.030 and tables and and 22.900F.020; repeals §§ 22.900A.030(D), 22.900C.010(G) and this source file. 22.900D.120, fees and charges for permits and activities of the Department of Design, Construction and Land Use (22.900A, 22.900B, 22.900C, 22.900D, 22.900E, 22.900F) 120819 Amends §§ 4.13.010, 4.20.380(A) and 4.20.390(A) and (B), First Quarter 2002 Salary Ordinance (4.13, 4.20) Adds §§ 15.17.080, 120822 15.48.900 and Ch. 15.91; amends §§ 15.04.074, 15.17.020, 15.17.050, 15.50.040, 15.50.070 and 15.50.080; repeals §§ 15.48.050, 15.50.050 and 15.50.060, street and sidewalk use (15.04, 15.17, 15.48, 15.50, 15.91) 120827 Amends § 18.12.080(B)(3) and Exhibit 18.12.080 J, Department of Parks and Recreation and animal control (18.12) 120831 Amends §§ 2.04.010(21), 2.04.160(A)(2), 2.04.180(B), 2.04.220, 2.04.250, 2.04.260, 2.04.265(A), 2.04.270(A)(2),

> For current SMC, contact the Office of the City Clerk

2.04.275(B), 2.04.290(B)(1) and 2.04.370(D)(1), election campaigns (2.04)

June 2009 co

Text provide

See ordinant sections for and tables a this source

Amends §§ 2.14.010,	
2.14.020, 2.14.040, 2.14.050, 2.14.060, 2.14.080, 2.14.090, 2.14.100 and 3.42.030; removes §§ 2.14.120 and 2.14.13 from codification elections (2.14, 3.42)  120837	
2.14.050, 2.14.060, 2.14.080, 2.14.090, 2.14.100 and 3.42.030; removes §§ 2.14.120 and 2.14.13 from codification elections (2.14, 3.42)  120837  Amends § 15.04.030, us and occupation permit applications (15.04)  120842  Amends Official Land Us Map (23.32)  120843  Amends § 15.17.050, str and sidewalk use (15.17)  Amends § 4.04.265, Civ Service Commission (4.4)  120857  Amends § 8.23.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  120860  Amends Official Land Us Map (23.32)  120861  Amends § 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 3.14.920(D), Human Rights Commission (23.60)  120871  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends § \$ 1.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
2.14.080, 2.14.090, 2.14.100 and 3.42.030; removes §§ 2.14.120 and 2.14.13 from codification elections (2.14, 3.42)  Amends § 15.04.030, us and occupation permit applications (15.04)  Amends § 15.17.050, str and sidewalk use (15.17)  Amends § 4.04.265, Civ Service Commission (4.0)  120845  Amends §§ 2.3.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 3.60.016, Shoreline District regulations (23.60)  120871  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  Repeals § 4.04.235,	
2.14.100 and 3.42.030; removes §§ 2.14.120 and 2.14.13 from codification elections (2.14, 3.42)  120837	
removes §§ 2.14.120 and 2.14.13 from codification elections (2.14, 3.42)  Amends § 15.04.030, us and occupation permit applications (15.04)  Amends Official Land U Map (23.32)  120843  Amends § 15.17.050, str and sidewalk use (15.17)  120845  Amends § 4.04.265, Civ Service Commission (4.4)  120857  Amends § 23.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  120860  Amends Official Land U Map (23.32)  120861  Amends § 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  120871  Amends § 3.14.920(D), Human Rights  Commission (3.14)  120874  Amends § 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  Repeals § 4.04.235,	
2.14.13 from codification elections (2.14, 3.42)  Amends § 15.04.030, us and occupation permit applications (15.04)  Amends Official Land Using (23.32)  120843  Amends § 15.17.050, str and sidewalk use (15.17)  120845  Amends § 4.04.265, Civ Service Commission (4.6)  120857  Amends § 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Comparison (23.76)  120860  Amends Official Land Using (23.32)  120861  Amends § 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  120871  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends § 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  Repeals § 4.04.235,	i
elections (2.14, 3.42)	
120837	Ĺ
and occupation permit applications (15.04)  120842	e (
applications (15.04) Amends Official Land U Map (23.32)  120843 Amends § 15.17.050, str and sidewalk use (15.17)  120845 Amends § 4.04.265, Civ Service Commission (4.4)  120857 Amends § 23.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  120860 Amends Official Land U Map (23.32)  120861 Amends § 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866 Amends § 23.60.016, Shoreline District regulations (23.60)  120871 Amends § 3.14.920(D), Human Rights Commission (3.14)  120874 Amends § 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76) Repeals § 4.04.235,	
120842   Amends Official Land U Map (23.32)     120843   Amends § 15.17.050, str and sidewalk use (15.17)     120845   Amends § 4.04.265, Civ Service Commission (4.0)     120857   Amends §§ 23.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)     120860   Amends Official Land U Map (23.32)     120861   Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)     120866   Amends § 23.60.016, Shoreline District regulations (23.60)     120871   Amends § 3.14.920(D), Human Rights Commission (3.14)     120874   Amends § 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)     120876   Amends § 21.76.030(A)(2), low-income utility credit (21.76)     120879   Repeals § 4.04.235,	
Amends § 15.17.050, str and sidewalk use (15.17)	se
Amends § 15.17.050, str and sidewalk use (15.17)	
and sidewalk use (15.17)  Amends § 4.04.265, Civ Service Commission (4.0)  Amends §§ 23.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  120860  Amends Official Land U Map (23.32)  Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  Repeals § 4.04.235,	eet
Amends § 4.04.265, Civ Service Commission (4.05)  Amends §§ 23.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Comparison (23.76)  120860  Amends Official Land Use Map (23.32)  Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
Service Commission (4.4)   120857	
Amends §§ 23.76.005, 23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  120860 Amends Official Land U Map (23.32)  120861 Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866 Amends § 23.60.016, Shoreline District regulations (23.60)  120871 Amends § 3.14.920(D), Human Rights Commission (3.14)  120874 Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	
23.76.010(D) and subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  120860 Amends Official Land U Map (23.32)  120861 Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866 Amends § 23.60.016, Shoreline District regulations (23.60)  120871 Amends § 3.14.920(D), Human Rights Commission (3.14)  120874 Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	/
subsection 106.6.1 of the 1997 Seattle Building Co (23.76)  120860  Amends Official Land U Map (23.32)  120861  Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  120871  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
1997 Seattle Building Co (23.76)  120860  Amends Official Land U Map (23.32)  120861  Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  120871  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	,
(23.76)  Amends Official Land U Map (23.32)  120861  Amends §§ 4.50.020, 4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
120860   Amends Official Land U Map (23.32)     120861   Amends §§ 4.50.020,	,
Map (23.32)   Amends §§ 4.50.020,   4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)   120866   Amends § 23.60.016,   Shoreline District regulations (23.60)   120871   Amends § 3.14.920(D),   Human Rights Commission (3.14)   120874   Amends §§ 21.04.430,   21.04.440 and   21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)   120876   Amends § 21.76.030(A)(2),   low-income utility credit (21.76)   120879   Repeals § 4.04.235,	se
Amends §§ 4.50.020,	50
4.100.020; repeals § 5 of Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866 Amends § 23.60.016, Shoreline District regulations (23.60)  120871 Amends § 3.14.920(D), Human Rights Commission (3.14)  120874 Amends § 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	
Ord. 120489, Health Car Subfund of the General Fund (4.50, 4.100)  120866 Amends § 23.60.016, Shoreline District regulations (23.60)  120871 Amends § 3.14.920(D), Human Rights Commission (3.14)  120874 Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	
Subfund of the General Fund (4.50, 4.100)  120866 Amends § 23.60.016, Shoreline District regulations (23.60)  120871 Amends § 3.14.920(D), Human Rights Commission (3.14)  120874 Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	
Fund (4.50, 4.100)  120866  Amends § 23.60.016, Shoreline District regulations (23.60)  120871  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends § 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	•
Amends § 23.60.016,     Shoreline District     regulations (23.60)     120871	
Shoreline District regulations (23.60)  120871	
regulations (23.60)  120871  Amends § 3.14.920(D), Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
120871 Amends § 3.14.920(D), Human Rights Commission (3.14)  120874 Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	
Human Rights Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
Commission (3.14)  120874  Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
120874 Amends §§ 21.04.430, 21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	
21.04.440 and 21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
21.76.040(A)(3), water services of Seattle Public Utilities (21.04, 21.76)  120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879 Repeals § 4.04.235,	
services of Seattle Public Utilities (21.04, 21.76)  120876  Amends § 21.76.030(A)(2), low-income utility credit (21.76)  120879  Repeals § 4.04.235,	
120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76) 120879 Repeals § 4.04.235,	2
120876 Amends § 21.76.030(A)(2), low-income utility credit (21.76) 120879 Repeals § 4.04.235,	
21.76.030(A)(2), low-income utility credit (21.76) 120879 Repeals § 4.04.235,	
low-income utility credit (21.76) 120879 Repeals § 4.04.235,	
(21.76) 120879 Repeals § 4.04.235,	S
corrective action process	
(Repealer)	
120887 Adds § 12A.02.085;	
amends §§ 12.02.080,	
12A.02.150, 12A.08.010	١,
12A.10.010(A)(4),	
12A.18.010(A),	
12A.18.090, 12A.24.025	
and 12A.24.150, crimina	
code (12A.02, 12A.08,	
12A.10, 12A.18, 12A.24	)
120889 Adds Ch. 6.295; repeals	
Ch. 6.294, all-ages dance	es
and dance venues (6.295	

reference only.

d amending
se graphics,
extractions of

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a - anda	te file
	120890	Amends § 3.14.521, Office
		of Policy and Management
	a contact	(3.14)
	120891	Adds § 4.36.120(D), City
3		Employees' Retirement
		System (4.36)
	120907	Amends § 12A.10.070 and
		Ord. 116666, illegal sex
		industry (12A.10)
	120910	Amends Official Land Use
	cea ca	Map (23.32)
	120914	Amends §§ 3.51.010,
	- 400 40 16	3.56.010, 3.58.030,
	COMPA	3.64.020, 23.41.008,
		23.66.110(A) and
	140 00	25.12.270, Get Engaged
	ma to co	board and commission
		positions (3.51, 3.58, 3.64,
		23.41, 23.66, 25.12)
	120915	Amends § 2.04.270(A)(2),
5	Pare	election campaign
		reporting requirements for
		independent expenditures
		(2.04)
	120916	Designates Bryant
		Elementary School as a
		landmark (25.32)
	120917	Designates Dunlap
		Elementary School as a
	100010	landmark (25.32)
	120918	Designates Concord
		Elementary School as a
	120010	landmark (25.32)
	120919	Designates Emerson
		Elementary School as a
	120021	landmark (25.32)
	120921	Adds § 4.36.199; amends §
		4.36.190(B)(3), City
		Employees' Retirement
	120022	System (4.36)
	120922	Amends § 4.36.120,
		employees of the
		Department of Parks and
		Recreation, Zoo Program (4.36)
	120024	` '
	120924	Amends § 23.55.015(E),
	120026	sign kiosks (23.55) Amends Official Land Use
	120926	Map (23.32)
ı		1v1ap (23.32)

eference only.

d amending
graphics,
accuracy of

Seattle Municipal Code ference only. 120927 Amends §§ 23.60.394(D), 23.60.420(A), Text provided for hi 23.60.484(G), 23.60.488 (name of section only), 23.60.544(B), 23.60.572(D), 23.60.600(E), 23.60.606(F), 23.60.632(D), 23.60.660, 23.60.660(H), See ordinances 23.60.662(A), 23.60.668(D), 23.60.692(C), sections for 23.60.720(C), 23.60.728(E), and tables and to 23.60.730(A), 23.60.752(D), 23.60.780(F), 23.60.786(D), this source file. 23.60.812(D), 23.60.840(E), 23.60.848(D), 23.60.872(C) and 23.60.906, telecommunication utilities in shoreline environments (23.60)

Seattle Municipal Code eference only. date 120928 Adds §§ 23.57.004 and 23.57.016; amends §§ Text provided for hi 23.41.004(B), 23.41.012(B), 23.44.002, 23.44.012(C), 23.45.002(B), 23.45.009(D), 23.45.016(B), and 23.45.050(D), See ordinances crea 23.45.058(B), 23.45.066(C), 23.45.074(B), sections for comp 23.47.002(E), 23.47.012, 23.47.024(B), 23.48.002(B), 23.48.010(C), and tables and 23.49.002(E), 23.49.008(C), 23.49.009(E), 23.49.208, this source file. 23.49.338(A), 23.50.002(C), 23.50.020(A), 23.57.001, 23.57.002, 23.57.003(B), 23.57.005, 23.57.008(A), 23.57.008(H), 23.57.009, 23.57.010, 23.57.011, 23.57.012, 23.57.013(B) and (C), 23.57.014, 23.57.015, 23.66.122(A), 23.66.140(C), 23.66.322(A), 23.66.332(C), 23.76.022(C), 23.76.056(D), 23.84.006 and 25.05.675(F), land use and zoning (23.41, 23.44, 23.45, 23.47, 23.48, 23.49, 23.50, 23.57, 23.66, 23.76, 23.84, 25.05) 120936 Amends § 4.04.240, intradepartmental grievance procedure (4.04) 120938 Adds Ch. 4.78, Workplace Violence Prevention Program (4.78) 120947 Amends §§ 21.36.012, 21.36.080 and 21.40.070(C), solid waste collection (21.36, 21.40) 120953 Amends Chart A of § 23.54.015, land use and

## For current SMC, contact the Office of the City Clerk

120966

zoning (23.54)

Amends § 5.55.030(A), business license fee surcharge (5.55)

June 2009 co

Text provide

See ordinant sections for and tables a this source

ilbar	0210
a made	ate file
120967	Amends §§ 23.49.008(A),
	23.49.009(B),
a com bi	23.49.011(A) and (B),
4 FOL IN	Chart 23.49.014A,
	23.49.014(C),
	23.49.025(A) and
	23.49.058(C)[(A)], land
	use and zoning (23.49)
120969	Amends §§ 21.33.030(D)
oc cred	and 21.76.040(A)(2),
ces es	drainage services of Seattle
	Public Utilities (21.33,
	21.76)
120970	Amends §§
	21.28.040(B)(1) and
140 0	21.76.040(A)(1),
ma to o	wastewater services of
133	Seattle Public Utilities
	(21.28, 21.76)
120971	Amends §§ 21.40.050,
Fire	21.40.060, 21.40.070,
	21.40.080, 21.40.085(A)
	and 21.76.040, solid waste
	system of Seattle Public
	Utilities (21.40, 21.76)
120975	Amends § 5.40.120,
	admissions tax (5.40)
120976	Adds § 7.04.038; amends
	§§ 7.04.145, 7.04.175,
	7.04.645 and 7.04.690(A),
	electronic pricing systems
	(7.04)
120978	Amends §§ 6.42.030,
	6.48.020 and 6.270.060,
	regulatory business and
	professional license fees
	(6.42, 6.48, 6.270)
120981	Adds § 5.08.025; amends §
	5.08.020; repeals §
	5.08.030, transfers of
	expenditure allowances in
	the budget (5.08)
120982	Amends §§ 22.602.040,
	22.602.050 and
	22.602.070, fire code
	permit and inspection fees
	(22.602)
120983	Amends § 22.504.010,
	plumbing permit fees
	(22.504)
120985	Amends § 3.35.060,
	Department of
	Neighborhoods (3.35)
120986	Amends § 18.28.010,
	Department of Parks and
	Recreation fee schedule
	(18.28)
120989	Amends § 17.19.010,
	G at G a D

eference only.

I amending graphics, ccuracy of

For current SMC, Constitution of the City Clerk the Office of the City Clerk

Seattle Center Department

(17.19)

June 2009 co Text provide

See ordinant sections for and tables a this source

da	te file
120990	Amends §§ 17.16.010 and
	17.16.015, Seattle Center
100	Department (17.16)
120994	Amends §§ 11.16.319 and
Gar. Pos	11.26.260, Traffic Code
	(11.16, 11.26)
120995	Amends § 4.13.010,
	exemptions from the Civil
	Service (4.13)
120996	Amends § 4.50.020, Health
res cross	Care Subfund (4.50)
120997	Amends §§
	22.900A.050(B),
6 COMPA	22.900B.010, 22.900B.020,
	22.900C.010,
1 40 00	22.900D.010,
and to co	22.900D.060(C) and (D),
0,1100	22.900D.080,
	22.900D.090,
9112	22.900D.100,
3 3 3 3 5 .	22.900D.110, 22.900D.130
	(B) and (C), 22.900D.140,
	22.900D.145,
	22.900D.150,
	22.900D.160(A) and (D),
	22.900E.010, 22.900E.020,
	22.900E.030, 22.900E.040,
	22.900E.050,
	22.900E.060(C) and
	22.900G.020 Department
	of Design, Construction
	and Land Use fees and
	charges (22.900A,
	22.900B, 22.900C,
	22.900D, 22.900E, 22.900G)
120998	Amends § 22.900G.030,
120990	gas piping permit fees
	(22.900G)
121000	Amends § 5.48.055, rates
121000	and terms of taxes on the
	collection or transfer of
	solid waste (5.48)
121004	Amends §§ 9.26.020,
121001	9.26.025, 9.26.040,
	9.26.050 and 9.26.060,
	animal fees (9.26)
121005	Adds § 11.31.121,
-21000	penalties for parking
	infractions (11.31)
<u> </u>	(-1.61)

eference only.

l amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinan sections for and tables a this source

		te file
121006	1111 mar	Adds §§ 3.14.800,
		3.14.810, 3.14.815,
		3.14.820, 3.14.825,
_4 \$		3.14.830, 3.14.840 and
		3.14.845; amends §§
		5.40.120, 5.78.030,
		6.295.190, 20.32.020,
		20.32.030, 20.32.040,
		20.32.050, 20.32.040, 20.32.050.
	- 4000	22.900A.040(B)(5) and
205	CIES	
1000		22.900G.040; repeals §§
		3.56.010, 3.56.020,
	see this	3.56.030, 3.56.040,
CU	1116	3.56.060 and 3.56.070,
		organization of city
	140 00	government (3.14, 3.56,
		5.40, 5.78, 6.295, 20.32,
		22.900A, 22.900G)
121009		Amends §§ 10.08.173 and
231		10.08.180, automatic alarm
		systems (10.08)
121010		Amends § 18.28.010,
		Department of Parks and
		Recreation fee schedule
		(18.28)
121018		Amends § 23.47.008(C),
		land use and zoning
		(23.47)
121026		Adds § 12A.10.120,
		voyeurism in public places
		(12A.10)
121028		Amends § 5.06.010,
121020		investment authority of the
		Director of Executive
		Administration (5.06)
121029		Amends §§ 4.24.005,
12102)		4.24.035 and 4.30.020, sick
		leave and eligibility criteria
		(4.24, 4.30)
121030		Adds § 4.36.285, Seattle
121030		City Employees'
101020		Retirement System (4.36)
121038		Adds § 15.48.105; amends
		§§ 15.12.010(A),
		15.48.110 and 15.48.120;
		repeals § 15.48.100, signs
		on public property (15.12,
100000		15.48)
121042		Designates Lincoln
		Park/Lincoln Reservoir and
		Bobby Morris Playfield as
		a landmark (25.32)
121043		Designates the Gas Works
		Dark as a landmark (25.32)

ference only. amending

For current SMC, contact the Office of the City Clerk

121047

121051

Park as a landmark (25.32)

Amends § 23.47.012, land

use and zoning (23.47)

Adds § 4.04.250(L) [§ 4.04.250(M)], Civil Service Commission (4.04)

June 2009 co Text provide

See ordinant sections for and tables a this source

Clhar	C:10
a anda	te tille
121063	Adds §§ 4.18.015 and
	4.18.035; amends §§
100 - 101	4.18.020 and 4.18.030,
4 50t III	inquiries by city officers
HO POS	and employees into
	immigration status (4.18)
121066	Amends § 18.12.080,
121000	
	Department of Parks and
101071	Recreation (18.12)
121071	Amends § 12A.10.120,
ICCO CO	voyeurism in public places
	(12A.10)
121074	Adds Ch. 4.79, city
o COMPLY	employee safe driving
	program (4.79)
121076	Adds §§ 22.202.060,
and to C	22.206.265 and
913362	22.206.295; amends §§
	22.200.020, 22.202.050,
C210	22.206.260 and
	22.206.280 and 22.206.280, relocation
	assistance for emergency
	orders (22.200, 22.202,
	22.206)
121085	Amends § 21.04.440(E)(2),
	water rates and charges
	(21.04)
121091	Amends official land use
	map (23.32)
121093	Adds § 23.47.004(I);
	amends §§ 23.40.002,
	23.47.004 and 23.47.006,
	land use and zoning (23.40,
	23.47)
121098	Amends §§ 21.49.020,
121070	21.49.030(A),
	21.49.040(A),
	21.49.052(A),
	21.49.055(A),
	21.49.057(A) and
	` '
	21.49.058(A), city light
	department rate schedules
121000	(21.49)
121099	Designates the North East
	Library as a landmark
	(25.32)
121100	Designates the Magnolia
	Library as a landmark
	(25.32)
121101	Designates the Queen
	Anne Library as a
	landmark (25.32)
121102	Designates the West
121102	Seattle Library as a
	landmark (25.32)
121102	
121103	Designates the Fremont
	Library as a landmark
	(25.32)
121104	Designates the University
	Library as a landmark
461	(25.32)
	7

reference only.

d amending
graphics,
accuracy of

For current State (25.32) Court to the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Designates the University Library as a landmark (25.32) Designates the Green Lake
L DESIGNATES THE CITEEN LAKE
Library as a landmark
(25.32)
Designates the
Douglass-Truth Library as
a landmark (25.32)
Adds § 23.76.028(D);
amends § 23.76.026;
repeals and replaces §
23.76.032, land use and
zoning (23.76)
Amends §§ 25.06.030,
25.06.050, 25.06.110,
25.06.120; repeals §
25.06.090(I), floodplain development (25.06)
Amends § 4.88.010;
repeals § 4.88.020, city
service award program
(4.88)
Amends § 5.55.030(A),
business license fee
surcharge (5.55)
Amends § 4.36.199, zoo
employees (4.36)
Amends § 6.310.530,
taxicab rates (6.310)
Amends §§ 23.45.110,
23.46.004(B), 23.47.004 Chart A, 23.47.006(B),
23.47.010 Chart B,
23.47.028(A), 23.47.042,
23.48.008, 23.50.014(B),
23.50.027 Chart B,
23.53.015(D),
23.53.020(E),
23.53.030(E), 23.54.015(E)
and (I), 23.54.015 Chart A,
23.66.324(A),
23.72.008(B), 23.84.004, 23.84.010, 23.84.012,
23.84.010, 23.84.012, 23.84.030 and 23.84.038,
land use and zoning (23.45,
23.46, 23.47, 23.50, 23.53,
23.54, 23.66, 23.72, 23.84)
Amends §§ 3.28.910 and
3.28.915, Office of
Professional
Accountability Review
Board (3.28)
Amends § 21.60.210A,
cable television (21.60) Amends official land use
map (23.32)
Amends §§ 4.18.020,
4.18.030 and 4.18.040,
federal immigration laws
(4.18)

reference only.

asil amending
praphics,
accuracy of

For current Sine, the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	Ther	4- 5112
	a - moda	Te min
	121162	Amends § 23.84.038, land
		use code (23.84)
	121163	Adds § 23.24.046; amends
9	dian	§ 23.24.040(A), land use
	121164	and zoning (23.24)
	121164	Amends official land use
	121172	map (23.32) Amends § 4.04.40, fire
	1211/2	lieutenant promotional
	- or orea	examination (4.04)
	121178	Amends §§ 9.25.020,
		9.25.023, 9.25.024,
	eamile	9.25.030, 9.25.035,
9	Comple	9.25.083, 9.25.084,
		9.25.090, 9.25.100,
	ad to co	12A.06.060, Animal
4	iller co	Control Code (9.25, 12A.06)
	121192	Adds §§ 25.08.225,
		25.08.505, 25.08.508 and
3	III.	25.08.805; amends §§
		25.08.660 and 25.08.820,
		noise (25.08)
	121196	Adds §§ 23.42.106(E),
		23.47.004(I)
		[23.47.004(J)], 23.47.036 and 23.48.016(C); amends
		§§ 23.46.004(A) and (D),
		23.46.006(C),
		23.46.012(A),
		23.47.004(E),
		23.47.004(A), 23.47.024,
		23.47.032(A),
		23.47.042(C),
		23.49.008(B), 23.49.011(B), 23.49.016,
		23.49.026(B),
		23.49.146(B), 23.50.012
		Chart A, 23.53.005(A),
		23.53.015(D),
		23.53.025(E),
		23.53.030(E), 23.54.015(I),
		23.54.015 Chart A,
		23.54.030(B), (D), (F) and (J), 23.55.028(D),
		23.71.038, 23.73.010(A)
		and (C), 23.84.004,
		23.84.024, 23.90.006(F),
		25.06.110(B) and
		25.06.130(B), live-work
		units (23.42, 23.46, 23.47,
		23.48, 23.49, 23.50, 23.53,
		23.54, 23.55, 23.71, 23.73, 23.84, 23.90, 25.06)
	121215	Designates the Rosen
	121213	House as a landmark
		(25.32)
	121216	Designates the New
		Richmond Laundry
		Building as a landmark
		(25.32)'
	MC TROM	

eference only.

d amending
graphics,
accuracy of

For current SMES; compared the Office of the City Clerk

June 2009 co

Text provide

See ordinantes sections for and tables at this source

Cihar	. 6:10
a mada	te ille
121217	Designates the Queen
	Anne Water Tank #1 as a
Sor hi	landmark (25.32)
121218	Designates the Van Vorst
	Building as a landmark (25.32)
121219	Designates the El Rio
121217	Apartments as a landmark
	(25.32)
121220	Designates the Belltown
CE2 CT	Cottages as a landmark
	(35.32)
121223	Amends §§ 2.04.060 and
COMP	2.04.250(B)(3), reporting
	of election campaign contributions (2.04)
121225	Amends § 4.04.030,
THE STATE OF	definition of seniority
	(4.04)
121226	Adds §§ 5.45.075,
Paren	5.45.082, 5.55.215,
	5.55.216, 5.55.225; amends
	§§ 5.30.020, 5.30.025,
	5.30.030, 5.30.035, 5.30.040, 5.30.050,
	53.30.060, 5.45.050,
	5.45.060, 5.45.070,
	5.45.080, 5.45.090,
	5.45.100, 5.48.050,
	5.48.072, 5.55.040,
	5.55.060, 5.55.090,
	5.55.095, 5.55.100,
	5.55.110, 5.55.140, 5.55.150, 5.55.170,
	5.55.200, 5.55.220; repeals
	§ 5.45.105, taxation (5.30,
	5.45, 5.48, 5.55)
121234	Amends § 4.04.230,
	progressive discipline
121240	(4.04)
121240	Designates the Seattle Monorail as a landmark
	(25.32)
121241	Adds Ch. 15.90; amends §
	15.48.900; repeals Ch.
	15.50, street and sidewalk
101010	use (15.48, 15.90)
121242	Amends §§ 15.17.005, 15.17.010, 15.17.050,
	15.17.010, 15.17.050, 15.17.080, vending permits
	in the stadium area (15.17)
121244	Amends § 23.71.008,
	Northgate Overlay District
	(23.71)
121245	Amends § 23.61.008,
	Station Area Overlay
101046	District (23.61)
121246	Amends § 23.47.040,
	principal pedestrian streets in regard to Station Area
	Overlay District (23.47)
mont SM	District (25.47)
TELL DO	

reference only.
d amending
graphics,
accuracy of

For current Single Court Clerk the Office of the City Clerk

Seattle Municipal Code ference only. June 2009 co 121247 Amends §§ 23.84.002, 23.84.010 (23.84) 121253 Amends §§ 5.40.010, Text provide 5.40.020, 5.40.025, 5.40.040, 5.40.060, 5.40.080, 5.40.085, amending 5.40.090; repeals §§ 5.40.026, 5.40.028, 5.40.050, taxes on admission fees (5.40) See ordinan 121261 Rezone, Official Land Use Map, pages 116 and 117 (Special) sections for 121270 Designates the Pier 59 as a landmark (25.32) 121274 Designates the Henry and tables Owen Shuey House as a landmark (25.32) 121275 Designates the Grover this source Cleveland High School as a landmark (25.32)

Seattle Municipal Code June 2009 co 121276 Amends Ch. 3.06, §§ 3.06.010, 3.06.015, Text provided for hi 3.06.020, 3.06.030, 3.06.040, 3.06.050, 3.06.055, 3.06.080, 5.72.040, 5.72.050, 6.82.020, 6.230.030, 7.20.040, 7.24.020, 10.52.010, 15.02.042, See ordinances crea 15.02.044, 15.04.010, 15.04.015, 15.04.020, 15.04.030, 15.18.010, sections for comple 15.32.300, 15.52.020, 15.64.070, 15.76.060, 21.04.020, 21.04.270, and tables and to 21.16.030, 21.16.040, 21.16.200, 21.16.260, 21.16.350, 22.204.050, 22.206.160, 22.210.030, this source file. 22.220.040, 22.450.010, 22.602.070, 22.800.070, 22.800.080, 22.800.090, 22.801.010, 22.801.020,

22.900A.020, 22.903.020, 22.904.010, 23.42.106, 23.44.008, 23.44.041, 23.45.015, 23.53.015, 23.53.030, 23.60.090, 23.60.908, 23.66.030, 23.71.018, 23.79.010, 23.84.006, 23.84.008, 23.84.030, 25.08.060, 25.09.060, 25.09.320, 25.09.345, 25.09.420, 25.11.020, 25.11.050, 25.12.040, 25.12.115, 25.12.240, 25.12.370, 25.12.400, 25.12.670, 25.12.680, 25.12.690, 25.12.730, 25.12.740, 25.12.760, 25.12.770, 25.12.835, 25.12.910, 25.20.010, 25.22.020, 25.22.100, 25.22.130, 25.24.070, 25.24.080, 25.24.090, 25.24.100, 25.28.230, changing name of Department of Design, Construction and Land Use to Department of Planning and Development (3.06, 5.72, 6.82, 6.230, 7.20, 7.24, 10.52, 15.02, 15.04, 15.32, 15.52, 15.64, 15.76, 21.04, 21.16, 22.204, 22.206, 22.210, 22.220, 22.450, 22.602, 22.800, 22.801, 22.900A, 22.903,

eference only.

l amending
graphics,
ccuracy of

22.904, 23.44, 23.45, 23.53, 23.60, 23.66, 23.71, 23.79, 23.84, 25.08, 25.09, 25.11, 25.12, 25.20, 25.22, 25.24, 25.28)

The Office of the City

June 2009 co 121277 U

Text provide

See ordinant sections for and tables a this source

a anda	te file
121277	Amends §§ 23.42.040,
	23.76.004, intermittent
121226	uses (23.42, 23.76)
121278	Adds Ch. 15.54; amends §§
	23.41.004, 23.47.012,
	23.49.011, 23.50.024, 23.50.026, 23.50.028,
	23.76.004, 23.76.006,
	23.80.004, 23.70.000, 23.80.004, 23.84.006,
- 0402	23.84.010, 23.84.025,
res crea	23.84.038, monorail transit
	facilities (15.54, 23.41,
	23.47, 23.49, 23.50, 23.76,
comple	23.80, 23.84)
121281	Amends 23.50.027, special
440.00	exception to size-of-use
and to Co	limits for retail sales and
11100	service uses in Duwamish
	Manufacturing/Industrial
C310.	Center (23.50)
121285	Amends § 23.44.051, bed
	and breakfasts (23.44)
121286	Adds § 22.300.016; repeals
	§ 22.300.015, National
	Electric Code (22.300)
121291	Amends §§ 1.20.010,
	23.22.020, vertical and
	horizontal reference datum
	(1.20, 23.22)
121303	Amends § 23.41.010,
	Northgate Urban Center
	and Overlay District
	Design Guidelines (23.41)
121305	Amends § 23.41.010,
	neighborhood design
	guidelines for South Lake
	Union Urban Village
121212	(23.41)
121313	Amends § 18.12.080,
101217	off-leash areas (18.12)
121317	Amends §§ 3.26.040, 11.23.120, 15.04.074,
	15.35.010, film permits
	(3.26, 11.23, 15.04, 15.35)
121320	Adds Ch. 7.25, third party
121320	billing regulation (7.25)
121327	Amends §§ 21.28.005,
121321	21.28.040, 21.76.040,
	wastewater services of
	Seattle Public Utilities
	(21.28, 21.76)
121328	Amends §§ 21.33.030,
	21.76.040, drainage
	services of Seattle Public
	Utilities (21.33, 21.76)
121330	Amends § 11.16.120; adds
	§ 11.16.121, parking rates

d amending graphics, ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cihar	C:10
a mada	te tille
121332	Adds Ch. 6.10; amends §§
	10.08.140, 10.08.165,
a sor hi	10.08.178, 10.08.180;
ad in	repeals §§ 10.08.080, 10.08.090, 10.08.100,
	10.08.110, 10.08.120,
	10.08.173, alarm systems
	(6.10, 10.08)
121334	Amends § 15.04.074, street
orea	and sidewalk use (15.04)
121338	Amends §§ 22.900A.020,
10	22.900A.060,
comple	22.900C.010, 22.900D.150,
Compage	22.900D.160, 22.900F.040,
1 40 00	22.900G.010, fees and
and to co	charges for permits and
d'Il ge	activities of the
	Department of Planning
sile.	and Development
8880	(22.900A, 22.900C,
	22.900D, 22.900F, 22.900G)
121339	Repeals § 9.25.025;
121337	amends § 9.25.030, Seattle
	Animal Control (9.25)
121341	Amends § 3.35.060,
	P-Patch fees (3.35)
121345	Amends § 4.13.010,
	exemptions from Civil
121347	Service (4.13) Amends § 18.28.010,
121347	Department of Parks and
	Recreation (18.28)
121355	Amends § 21.49.083,
	Seattle City Light's
	Voluntary Green Power
101075	Program (21.49)
121357	Adds §§ 6.310.225, 6.310.480; amends §
	6.310.465, taxicabs (6.310)
121359	Amends §§ 23.41.012,
121337	23.47.012, 23.48.010,
	23.50.020, 23.50.026,
	23.50.028, 23.54.015,
	23.54.035, 23.84.032,
	23.86.006, land use and
	zoning, biotech uses in South Lake Union (23.41,
	23.47, 23.48, 23.50, 23.54,
	23.84, 23.86)
121362	Amends §§ 23.71.004,
	23.71.008, 23.71.014,
	23.71.020, 23.71.038,
	23.71.044, 23.76.004,
	23.76.006; repeals §§ 23.71.024, 23.71.026,
	23.71.024, 23.71.026, 23.71.028, 23.71.029, land
	use and zoning, Northgate
	Overlay District (23.71,
	23.76)
MONT DIVI	

reference only.

amending

graphics,

curacy of

For current Sme, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cipar	
40	te file
121364	Amends Ch. 5.22 (title), §§
	5.22.010, 5.22.020, City's
a com hi	acceptance of credit cards,
ad for 111	debit cards, and other
	electronic commerce
	methods for payment of
	taxes, license fees, fines, other fees, and other City
	services (5.22)
121365	Amends §§ 4.36.100,
ces crea	4.36.103, 4.36.110,
	4.36.124, 4.36.190,
	4.36.191, 4.36.195,
COMPA	4.36.400; repeals §§
	4.36.191.5, 4.36.197,
and to Co	Seattle City Employees' Retirement Code (4.36)
121366	Amends §§ 4.36.040,
121300	4.36.050, 4.36.200,
C310.	4.36.210, 4.36.230,
1116.	4.36.260, 4.36.270,
	4.36.320, 4.36.330, Seattle
	City Employees'
101270	Retirement Code (4.36)
121372	Adds §§ 21.36.082, 21.36.083; amends §
	21.36.922, Seattle Public
	Utilities, recyclable
	materials (21.36)
121379	Amends §§ 3.02.110,
	3.92.010, Hearing
	Examiner (3.02, 3.92)
121381	Amends § 11.14.277,
121388	traffic regulations (11.14) Amends §§ 11.14.425,
121300	11.16.120, 11.16.300,
	11.16.480, 11.23.120,
	11.23.380, 11.26.060,
	11.26.080, 11.26.100,
	11.26.220, 11.31.121,
	11.72.065, 11.72.220, Ch.
	11.76 (title), 11.76.010, 11.76.015, 11.76.020,
	11.76.040, 12A.08.105,
	15.17.150, 15.35.010,
	15.35.020; adds §
	11.76.005; repeals §
	11.76.050, regulations for
	payment for vehicular
	parking at pay stations
	(11.14, 11.16, 11.23, 11.26, 11.31, 11.72, 11.76,
	12A.08, 15.17, 15.35)
121389	Amends § 11.16.125,
	authorizes Director of
	Transportation to close or
	authorize closure of
	shoreline street ends from
	dusk to dawn (11.16)

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Gihar	0210
and a	te tile
121407	Repeals § 7.20.115;
	amends §§ 7.20.125,
1 0 - m hi	7.20.150, rights of first
d for III	refusal for tenants of
	floating home moorages
	(7.20)
121408	Amends § 22.206.160,
	tenant's right of first refusal
	(22.206)
121409	Repeals §§ 22.902.070,
	22.902.090, 22.902.100,
-10	22.902.220; amends §§
COMPLE	22.902.080, 22.902.210,
Com-2	tenant's right of first refusal to purchase when rental
	units are converted to
and to co	cooperatives (22.902)
121415	Adds Ch. 5.73, 2004
==	Multifamily Housing
Cá10.	Property Tax Exemption
2330.	Program (5.73)
121417	Rezone, Official Land Use
	Map, page 111 (Special)
121420	Amends Ch. 3.12 (title), §§
	3.12.10, 3.12.030,
	3.12.120, 3.12.210,
	11.16.121, 11.16.319,
	11.24.100, 11.26.260,
	11.58.305, 11.66.200,
	15.32.300, 15.70.030,
	15.72.020, 15.72.040,
	21.100.010, 21.100.020, 21.100.050, 22.900G.020,
	23.47.022, 23.48.024,
	23.84.006, 25.05.675,
	25.09.340; repeals §
	3.12.040, changing name
	of Seattle Transportation to
	Seattle Department of
	Transportation (3.12,
	11.16, 11.24, 11.26, 11.58,
	11.66, 15.32, 15.70, 15.72,
	21.100, 22.900G, 23.47,
101404	23.48, 23.84, 25.05, 25.09)
121424	Adds Ch. 5.09,
	Opinion-Gathering
	Activities (5.09)
121426	Designates the Cibbs
121426	Designates the Gibbs
121426	House as a landmark (Ch.
	House as a landmark (Ch. 25.32)
121426 121429	House as a landmark (Ch. 25.32) Amends §§ 23.44.017,
	House as a landmark (Ch. 25.32)  Amends §§ 23.44.017, 23.55.020, 23.55.022,
	House as a landmark (Ch. 25.32)  Amends §§ 23.44.017, 23.55.020, 23.55.022, 23.78.002, 23.78.010,
	House as a landmark (Ch. 25.32)  Amends §§ 23.44.017, 23.55.020, 23.55.022, 23.78.002, 23.78.010, 23.79.004, 23.79.008,
	House as a landmark (Ch. 25.32)  Amends §§ 23.44.017, 23.55.020, 23.55.022, 23.78.002, 23.78.010,

reference only.

sal amending

graphics,

sal accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibos	4 - 6110
a anda	te m
121443	Adds §§ 21.04.105,
	24.04.115, 21.04.125;
1 Car hi	repeals §§ 21.04.110,
d for m	21.04.120, 21.04.130,
	21.04.340, 21.04.350;
	amends § 21.04.465,
	Seattle Public Utilities
121452	(21.04)
121453	Amends § 4.04.180,
DOS CITED	performance evaluation
121454	(4.04)
121434	Amends § 4.24.035, paid
121468	sick leaveuse (4.24) Amends §§ 7.20.080,
121408	7.20.130, 7.20.150,
	regulation of floating
and to co	homes and floating home
	moorages (7.20)
121474	Rezone, Official Land Use
C110.	Map, pages 188, 208
1110.	(Special)
121475	Amends § 23.41.008,
	Design Review Board
	(23.41)
121476	Amends §§ 23.24.046,
	23.24.050, 23.34.009,
	23.41.010, 23.44.010,
	23.44.012, 23.44.014,
	23.47.009, 23.48.032,
	23.49.064, 23.50.012,
	23.50.029, 23.50.032,
	23.54.015, 23.54.030,
	23.73.008, 23.76.006,
	23.76.010, 23.76.011,
	23.76.012, 23.76.040, 23.76.050, 23.84.010,
	23.86.006, 23.86.008,
	23.88.020, land use and
	zoning (23.24, 23.34,
	23.41, 23.44, 23.47, 23.48,
	23.49, 23.50, 23.54, 23.73,
	23.76, 23.84, 23.86, 23.88)
	23.70, 23.04, 23.00, 23.00)

eference only.

d amending
graphics,
accuracy of

Seattle Municipal Code June 2009 co 121477

See ordinances sections for comple and tables and this source file.

Amends §§ 23.22.100, 23.42.106, 23.44.016, Text provided for hi 23.44.036, 23.44.041, 23.45.015, 23.45.057, 23.45.073, 23.45.112, 23.46.018, 23.47.016, 23.47.020, 23.47.028, 23.49.009, 23.49.017, 23.49.020, 23.49.056, 23.49.076, 23.49.106, 23.49.134, 23.49.162, 23.49.332, 23.50.014, 23.50.016, 23.50.034, 23.50.044, 23.50.046, 23.53.020, 23.53.035, 23.54.015, 23.54.016, 23.54.030, 23.54.035, 23.55.005, 23.55.012, 23.55.014, 23.55.015, 23.60.070, 23.60.078, 23.60.196, 23.60.902, 23.66.020, 23.69.032, 23.71.012, 23.71.018, 23.76.012, 23.76.014, 23.76.020, 23.76.022, 23.76.024, 23.76.026, 23.76.036, 23.76.042, 23.76.052, 23.76.062, 23.78.006, 23.78.012, 23.79.010, 23.84.008, 23.84.014, 23.84.024, 23.84.030, 23.84.036, 23.88.010, 23.88.020, 23.91.010, 23.91.012, land use and zoning (23.22, 23.42, 23.44, 23.45, 23.46, 23.47, 23.49, 23.50, 23.53, 23.54, 23.55, 23.60, 23.66, 23.69, 23.71, 23.76, 23.78, 23.79, 23.84, 23.88, 23.91) 121483 Amends §§ 11.30.105, 11.30.120, impoundment of vehicles (11.30) 121509 Adds § 12A.20.060, marijuana law enforcement policy (12A.20) 121518 Adds §§ 11.14.184, 11.14.333, Ch. 11.46, electric personal assistive mobility devices and motorized foot scooters (11.14, 11.46)121519 Amends § 22.100.010. building code (22.100) 121520 Adds § 22.420.010, fuel gas code (22.420) 121521 Adds § 22.150.010, residential code (22.150) 121522 Amends § 22.700.000, energy code (22.700) 121523 Amends § 22.400.010,

eference only. amending

mechanical code (22.400) the Office of the City

June 2009 co

Text provide

See ordinant sections for and tables a this source

	There	. C:10
	a anda	te Ille
	121524	Amends § 22.600.020;
		repeals 22.600.040, fire
	- C 101	code (22.600)
•	121525	Amends §§ 11.20.100,
		11.20.230, 11.20.345,
		11.30.040, 11.30.120,
		11.52.100, 11.56.020,
		11.56.320, 11.56.350, 11.72.065, 11.84.290; adds
	- 0002	§§ 11.53.230, 11.56.025;
	ces crea	repeals §§ 11.31.110,
- '		11.58.350, motor vehicles
		(11.20, 11.30, 11.31, 11.38,
	COMPA	11.52, 11.53, 11.56, 11.58,
		11.72, 11.84)
	121527	Amends § 4.20.300,
	and to co	payment for performance
٠		of out-of-class duties
	121520	(4.20.300)
	121530	Amends § 3.70.150, Seattle
5	3850	Ethics and Elections
ŀ	101524	Commission (3.70)
	121534	Amends § 23.41.010, Belltown Neighborhood
		Design Guidelines (23.41)
ŀ	121543	Amends §§ 2.14.020,
	1213-13	2.14.060, election
		pamphlet provided to City
		voters (2.14)
	121549	Amends § 18.12.080,
		off-leash areas (18.12)
	121562	Amends § 3.02.120,
		powers of Hearing
	101560	Examiner (3.02)
	121563	Amends §§ 23.41.004,
		23.42.040, 23.47.042, 23.47.050, 23.80.004,
		permitting and approval
		process for light rail transit
		facilities (23.41, 23.42,
		23.47, 23.80)
	121568	Amends §§ 3.14.825,
		3.51.010, 3.51.020,
		3.51.030, 3.58.030,
		3.64.020, 23.66.110,
		25.12.270; renumbers § 3.51.040 as § 3.51.050,
		amends § 3.51.50; repeals
		§§ 3.51.040, 3.51.060, Get
		Engaged program (3.14,
		3.51. 3.58, 3.64, 23.66,
		25.12)
	121574	Amends §§ 20.12.020,
		20.12.080, eligibility
		criteria for deferral of
		assessment collections
ŀ	121505	(20.12)
	121585	Rezone, Official Land Use
Į		Map, page 110 (Special)

eference only.

l amending

graphics,

ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibar	to file
11110	
121588	Designates the Providence
	1910 Building as a
-4-Co# H1	landmark (Ch. 25.32)
121593	Adds Ch. 14.06; amends §§
	14.08.010, 14.08.020,
	14.08.030, 14.08.040,
	14.08.045, 14.08.060,
	14.08.070, 14.08.080,
	14.08.095, 14.08.100,
DOS CIES	14.08.110, 14.08.120,
	14.08.130, 14.08.135,
10	14.08.140, 14.08.150,
committe	14.08.160, 14.08.165,
Comple	14.08.170, 14.08.180,
	14.08.185, 14.08.187,
- 1 to C	14.08.190, 14.08.200,
and to o	14.08.210; repeals §§
	14.08.050, 14.08.055,
009 -	14.08.090, public
1511C.	accommodations (14.06,
121504	14.08)
121594	Amends § 3.02.125; adds
	Ch. 7.26, income tax
	refund anticipation loans,
	refund anticipation checks,
	and similar refund-related
121505	bank products (3.02, 7.26)
121595	Amends § 4.36.130, City
	Employees' Retirement
	System Board of
121712	Administration (4.36)
121613	Amends § 21.40.070, rates
	and charges of solid waste
121622	system (21.40)
121622	Amends §§ 2.08.030,
	2.08.040, 2.12.010,
	2.12.020, specifications and form of initiative
	petitions; general
	requirements and form of
	referendum petitions (2.08, 2.12)
121639	Amends §§ 21.33.030,
121037	21.76.040, rates for
	drainage services of Seattle
	Public Utilities (21.33,
	21.76)
121642	Amends §§ 5.80.020,
121072	5.80.030, Cumulative
	Subfund's Capital Projects
	Account (5.80)
121650	Amends § 9.26.060,
121030	Animal Fee Ordinance
	(9.26)
121654	Amends §§ 11.16.319,
121034	11.23.120, Seattle
	Department of
	Transportation permit fees
	(11.16, 11.23)
121655	Amends §§ 17.16.010,
121033	17.16.015, Seattle Center
- 104 GM	Department (17.16)
rent Sw	Department (17.10)

eference only.

d amending
graphics,
accuracy of

For current Styl Department (17.16)

the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

Sihar	. 6:10
a anda	te ille
121656	Amends § 3.35.060,
	P-Patch garden plot fees
40-4 h1	(3.35)
121657	Amends § 5.40.120,
	receipts to General
	Subfund and Arts account
121661	(5.40) Amends §§ 5.80.020,
121001	5.80.030, 15.62.090, fees
- 0402	for vacation of streets and
ces crea	other public rights-of-way
	(5.80, 15.62)
121662	Amends § 18.28.010,
COMPA	Department of Parks and
	Recreation fees (18.28)
121664	Amends §§ 22.900A.030,
ma to co	22.900A.040,
'HEOD	22.900B.010, 22.900B.020,
-00	22.900B.030, Ch. 22.900C,
sile.	§§ 22.900D.010,
7770	22.900D.060,
	22.900D.080,
	22.900D.090,
	22.900D.100,
	22.900D.110,
	22.900D.130, 22.900D.140,
	22.900D.140, 22.900D.145, 2.900D.150,
	22.900D.160, 22.900E.010,
	22.900E.020, 22.900E.030,
	2.900E.040, 22.900E.050,
	22.900E.060, 2.900F.010,
	22.900F.020, 22.900F.050,
	22.900G.010,
	22.900G.050, fees and
	charges for permits and
	activities of Department of
	Planning and Development
	(22.900A, 22.900B,
	22.900C, 22.900D,
	22.900E, 22.900F, 22.900G)
121665	Amends § 21.28.080,
121000	summer residential charges
	for wastewater services of
	Seattle Public Utilities
	(21.28)
121666	Amends § 21.24.021,
	sewer and drainage permit
	fees and charges of Seattle
10115	Public Utilities (21.24)
121667	Amends § 4.13.010,
	exemptions from Civil
101660	Service (4.13)
121668	Repeals § 22.602.040; adds
	§ 22.602.045; amends §§
	22 602 050 22 602 070
	22.602.050, 22.602.070,
	22.602.050, 22.602.070, 22.602.080, Seattle Fire Department fees (22.602)

eference only.

d amending
graphics,
accuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

cipai	to file
4 4 11 10 0	
121670	Amends § 5.48.055, fee or
4 Sor hi	tax on persons carrying on business of collection of
digital and	garbage, rubbish, trash, CDL Waste and other solid
121671	waste (5.48) Amends § 5.48.050, fee or
1210/1	tax on persons carrying on business of selling or
ces crea	furnishing water for hire to consumers (5.48)
121672	Amends § 5.48.050, fee or
comple	tax on persons carrying on business of selling or
140 0	furnishing water for hire to consumers (5.48)
121673	Amends § 5.48.050, rate
, H do	for tax levied upon total gross income from
C210	drainage and wastewater
1116.	charges (5.48)
121674	Amends §§ 21.40.050,
	21.40.060, rates and
	charges of solid waste
	system of Seattle Public
	Utilities (21.40)
121675	Amends §§ 21.28.040, 21.76.040, rates for
	wastewater services of
	Seattle Public Utilities
	(21.28, 21.76)
121676	Amends §§ 21.04.430,
	21.04.440, rates for water
	services of Seattle Public Utilities (21.04)
121677	Amends § 4.13.010,
	exemptions from Civil
101770	Service (4.13)
121678	Amends §§ 11.16.315, 11.23.380, 11.72.065,
	11.73.080, Seattle
	Department of
	Transportation to impose
	four-hour on-street parking
	time limits (11.16, 11.23,
121679	11.72, 11.73) Amends §§ 5.30.020,
1410/9	5.30.025, 5.30.030,
	5.30.035, 5.30.040,
	5.30.050, 5.30.060,
	5.45.050, 5.45.070,
	5.45.075, 5.45.076,
	5.45.090, 5.55.030,
	5.55.040, 6.20.050, business licensing (5.30)
	business licensing (5.30, 5.45, 5.55, 6.20)
	J.7J, J.JJ, U.2U <i>j</i>

eference only.

lid

or
lamending
to graphics,
or
couracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Sihar .	. c:10
a - mada	te file
121700	Amends §§ 5.72.020,
	5.73.020, 23.34.008,
- 0 - m hi	23.34.010, 23.34.020,
4 FOL 117	23.34.022, 23.84.030,
	23.84.036, 25.05.675;
	repeals § 23.71.001, land
	use and zoning (5.72, 5.73,
	23.34, 23.84, 25.05)
121716	Designates Golden
orea	Gardens Bath House as a
ces crea	landmark (Ch. 25.32)
121717	Amends § 3.33.200; adds
	Ch. 20.42; repeals Ch.
COMPLE	20.44 and Ch. 20.46A.
COSTO	affirmative efforts to
	ensure equality in City
and to Co	Contracting practices (3.33)
	20.42)
121718	Repeals § 3.04.095, review
Call	by Hearing Examiner of
	recommendation for
	contract awards (3.04)
121719	Amends § 20.40.010, small
121/1)	public works (20.40)
121720	Repeals § 3.04.201;
121720	renumbers Ch. 3.04,
	Subchs. II, III; adds Ch.
	20.60, Subchs. II, III;
	amends §§ 20.40.010,
	20.42.020, 20.60.200,
	20.42.020, 20.60.200, 20.60.204,
	20.60.206, 20.60.208,
	, , , , , , , , , , , , , , , , , , ,
	20.60.210, 20.60.216,
	20.60.222, 20.60.224,
	purchasing programs and procurement of reusable
	1
	products, recycled content
	products and recyclable
	products (3.04, 20.40,
121721	20.42, 20.60)
121/21	Amends Tit. 15, Apps. I, II,
	portion of Alaskan Way
	associated Olympic
	Sculpture Park (Tit. 15, Apps. I, II)

eference only.

amending
graphics,
accuracy of

June 2009 co 121722 tt

Text provide

See ordinant sections for and tables a this source

	ILE BASS
121722	Amends §§ 3.04.040,
	4.72.030, 20.42.020,
a cost hi	20.60.102, 20.60.140;
d tor	renumbers 3.114.010,
	3.114.030, 3.114.040,
	3.114.070, 3.114.080,
	3.114.090, 3.114.100,
	3.114.110, 3.114.140,
	3.114.150, 3.114.160;
oos crea	repeals 3.114.050,
CE2 G2	3.114.060, 3.114.120,
	3.114.130; adds 20.50.010,
	20.50.030, 20.50.040,
Comple	20.50.050, 20.50.060,
	20.50.070, 20.50.080,
1 40 00	20.50.090, 20.50.100,
ma to o	20.50.110, 20.50.120,
	20.50.130, consultant
-00	contracting (3.04, 4.72,
101700	20.42, 20.50, 20.60)
121723	Adds Ch. 20.70, debarment
	from contracting with City
121525	(20.70)
121735	Amends § 21.76.040, solid
	waste system of Seattle
101707	Public Utilities (21.76)
121737	Amends § 6.310.530,
121738	taximeter rates (6.310)
121/38	Amends §§ 6.310.110,
	6.310.150, 6.310.230, 6.310.300, 6.310.320,
	6.310.330, 6.310.340,
	6.310.350, 6.310.400, 6.310.415, 6.310.420,
	6.310.425, 6.310.450,
	6.310.425, 6.310.450, 6.310.455, 6.310.460,
	6.310.465, 6.310.470,
	6.310.475, 6.310.500,
	6.310.510, 6.310.605,
	6.310.635, 6.310.700,
	6.310.710; adds §
	6.310.125, taxicabs (6.310)
121747	Amends §§ 4.04.030,
121/7/	4.04.040, 4.08.020,
	4.08.070, 4.08.090,
	4.08.110, Civil Service
	systems of the City (4.08)
121759	Amends § 23.41.010,
141/37	Amenus § 25.41.010,

Wallingford Neighborhood Design Guidelines (23.41)

Wallingford Avenue North as Principal Pedestrian Streets (23.47, Special)

Amends § 23.47.040; rezone, Official Land Use Map, pages 58, 59, North

Amends § 23.42.112, nonconformity to development standards

45th Street and

eference only.

d amending
graphics,
accuracy of

For current SM (23.42) control
the Office of the City Clerk

121760

121762

June 2009 co

Text provide

See ordinantes sections for and tables at this source

	ilhes .	0:10
		te tille
	121765	Amends §§ 9.25.100,
		9.26.020, animal licensing
	hi	(9.25, 9.26)
	121768	Rezone, Official Land Use
G		Map, pages 58, 59
		(Special)
	121771	Amends §§ 4.04.030,
		4.04.075, 4.13.020,
		4.20.055, 4.20.190,
	ces crea	4.20.220, 4.24.010,
R		4.28.010, 4.30.010, 4.34.005, 4.34.045,
	10	4.34.065, 4.44.020,
10	COMPLE	temporary worker benefits
5		(4.04, 4.13, 4.20, 4.24,
	1 40 00	4.28, 4.30, 4.34, 4.44)
	121773	Amends § 22.600.020
C.		(editor's note), amendment
		to Seattle Fire Code
		(22.600)
5	121781	Amends § 23.41.010,
		design review guidelines (23.41)
	121782	Amends, §§ 23.30.010,
	121702	23.32.016, 23.34.126,
		23.34.128, 23.41.004,
		23.41.012, 23.42.106,
		23.42.126, 23.42.128,
		23.47.004, 23.47.012,
		23.48.002, 23.48.008,
		23.48.010, 23.48.012, 23.48.014, 23.48.016,
		23.48.014, 23.48.016, 23.48.018, 23.48.020,
		23.48.024, 23.48.031,
		23.48.032, 23.48.034,
		23.53.010, 23.53.015,
		23.53.030, 23.54.015,
		23.54.020, 23.54.030,
		23.55.030, 23.84.048,
		25.05.675; adds §
		23.48.019; rezone, Official Land Use Map, pages 101,
		102 (23.30, 23.32, 23.34,
		23.41, 23.42, 23.47, 23.48,
		23.53, 23.54, 23.55, 23.84,
		25.05, Special)
	121788	Amends § 18.12.080,
	121200	off-leash areas (18.12)
	121789	Amends § 23.47.040;
		rezone, Official Land Use Map, pages 23, 24, 37, 38,
		pedestrian designated
		zones (23.47, Special)
	121792	Amends § 23.54.015,
		parking requirements
		multifamily uses and SEPA
		mitigation authority within
	101707	certain areas (23.54)
	121795	Rezone, Official Land Use
		Map, page 153 (Special)

eference only.

l amending

graphics,

ccuracy of

Seattle Municipal Code

June 2009 co 121797 up date

Amend 5.40.02
5.40.02
5.40.02
admissi
121813

See ordinant sections for and tables at this source

		. 6310
	a - mada	te tille
M	121797	Amends §§ 5.40.010,
		5.40.020, 5.40.025,
	a sor hi	5.40.060; repeals §§
	d in in	5.40.027, 5.40.040,
1	121813	admissions tax (5.40) Amends § 15.17.010,
	121813	mobile vending (15.17)
ŀ	121821	Amends § 22.700.010,
	121021	adoption of Washington
	- a orea	State Energy Code and
d	ces crea	amendments (22.700)
4	121823	Rezone, Official Land Use
		Map, pages 100, 101
	COMPA	(Special)
Ī	121828	Amends §§ 23.46.004,
	140 00	23.47.004, 23.47.012,
	ma to co	23.47.042, 23.49.011,
		23.50.028, 23.53.015,
		23.53.030, 23.54.015,
	<b>2112</b>	23.71.038, 23.76.004,
	7770	23.76.006, 25.06.110,
		reenacting certain sections
		of Seattle Municipal Code
		(23.46, 23.47, 23.49, 23.50,
		23.53, 23.54, 23.71, 23.76,
ļ		25.06)
	121831	Amends §§ 23.34.079,
		23.47.012; rezone, Official
		Land Use Map, pages 103,
		111, Neighborhood
		Commercial 3/Residential
		designation (23.34, 23.47,
ŀ	101046	Special)
	121846	Amends § 23.47.012, floor area ratios (23.47)
ŀ	121847	Adds § 21.49.084,
	121847	voluntary Green Power,
		Green-Up Program (21.49)
ŀ	121851	Amends § 22.900D.010,
ı	121631	Table D-2, calculation of
-1		
		development fees
		development fees determined by value
		determined by value
	121855	determined by value (22.900D)
	121855	determined by value (22.900D) Amends § 15.64.090,
	121855	determined by value (22.900D)
	121855	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other
	121855 121856	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)
		determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other
		determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030,
		determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for
	121856	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for department heads (4.34)
	121856	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for department heads (4.34)  Amends §§ 6.310.110,
	121856	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for department heads (4.34)  Amends §§ 6.310.110, 6.310.320, 6.310.330,
	121856	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for department heads (4.34)  Amends § 6.310.110, 6.310.320, 6.310.330, 6.310.455, 6.310.500,
	121856 121857	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for department heads (4.34)  Amends § 6.310.110, 6.310.320, 6.310.330, 6.310.455, 6.310.500, 6.310.600, taxicabs (6.310)
-	121856 121857	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for department heads (4.34)  Amends § 6.310.110, 6.310.320, 6.310.330, 6.310.455, 6.310.500, 6.310.600, taxicabs (6.310)  Amends § 4.16.070; adds §
-	121856 121857	determined by value (22.900D)  Amends § 15.64.090, long-term permits for use of streets and other purposes (15.64)  Amends § 4.34.030, vacation allowance for department heads (4.34)  Amends §§ 6.310.110, 6.310.320, 6.310.330, 6.310.455, 6.310.500, 6.310.600, taxicabs (6.310)  Amends § 4.16.070; adds § 4.16.071, Application of

d amending graphics, accuracy of

Seattle Municipal Code

June 2009 co 121861 up date

Amend
7.12.02
7.12.04
price re

Text provide 121865 Amend

See ordinand sections for and tables a this source

	, ipus	te file
_ [	121861	Amends §§ 7.12.010,
	121601	
		7.12.020, 7.12.030,
	1 Cor HI	7.12.040, 7.12.050, unit
•	U IOI	price regulation (7.12)
	121865	Amends § 22.450.010,
		adoption of Seattle Boiler
		and Pressure Vessel Code
		(22.450)
	121866	Designates Cooper
	oc crea	Elementary School as a
M	CE2 CZ	landmark (Ch. 25.32)
	121870	Amended § 4.20.080.E
Ì	121874	Amends §§ 23.49.011,
	COMPA	23.49.014, transferable
		development rights (23.49)
Ì	121884	Amends §§ 4.24.005,
	TO C	4.24.035, sick leave (4.24)
	121897	Rezone, Official Land Use
		Map, page 111 (Special)
	121900	Amends §§ 5.40.020,
		5.40.025, exemption from
		admission tax for
		Children's Charity Bacon
		Bowl Football Game (5.40)
ŀ	121911	Amends § 4.13.010,
	121911	exemptions from the Civil
		Service and Public Safety
		,
		Civil Service systems (4.13)
	121915	Amends §§ 5.73.020,
	121913	5.73.040, 5.73.050,
		Multifamily Property Tax
		Exemption Program (5.73)
	121917	Adds §§ 11.14.567,
	121917	11.14.568, 11.23.034,
		11.72.357; amends §
		11.31.121, shuttle bus load
		zones (11.14, 11.23, 11.31, 11.72)
	121919	
	121919	Amends § 22.300.016,
		adoption of Electrical Code (22.300)
	121923	Amends §§ 11.62.020,
	121723	11.62.040, transport of
		hazardous materials
		(11.62)
	121925	Amends §§ 15.02.046,
	121/23	15.90.018, 15.91.002,
		15.91.012, 15.91.016,
		enforcement of SMC Title
		15, Street and Sidewalk
		Use Code (15.02, 15.90,
		15.91)
	121931	Rezone, Official Land Use
	121/31	Map, page 97 (Special)
Ĺ		map, page // (Special)

eference only.

amending
graphics,
ccuracy of

Seattle Municipal Code
June 2009 co 121932 up date
Amend
6.10.01
6.10.02
6.10.04
6.10.08
6.10.11

See ordinant sections for and tables a this source

a made	te file
121932	Amends §§ 6.10.005,
(3)	6.10.010, 6.10.015,
100	6.10.020, 6.10.030,
4 501° III	6.10.040, 6.10.050,
CI TO	6.10.080, 6.10.100,
	6.10.110, 6.10.205,
	6.10.210, 6.10.215,
	6.10.220, 6.10.230,
	6.68.010, 10.08.140,
- 0002	10.08.165, 10.08.168; adds
ces crea	§§ 6.10.025, 6.10.035,
	6.10.120, 6.10.130,
	6.10.235, alarm systems
COMPA	and fire alarm systems
6032-8	(6.10, 6.68, 10.08)
121942	Amends § 6.310.320,
and to C	taxicab and for-hire vehicle
High or	operating requirements
	(6.310)
121944	Amends §§ 11.31.020,
IIIc.	11.31.120; adds §
	11.31.090, automated
	traffic safety cameras
	(11.31)
121954	Amends §§ 11.31.121,
	11.76.005, 11.76.015,
	parking pay stations
	(11.31, 11.76)
121959	Rezone, Official Land Use
	Map, page 104 (Special)
121960	Rezone, Official Land Use
	Map, page 41 (Special)
121961	Rezone, Official Land Use
	Map, page 42 (Special)
121965	Rezone, Official Land Use
	Map, page 111 (Special)
121983	Amends § 17.19.010,
	Seattle Center parking
	charges (17.19)
121986	Amends § 18.28.010,
	Department of Parks and
	Recreation fee schedule
	(18.28)
121987	Amends §§ 5.45.100,
	5.48.050, 5.48.055,
	5.48.070, 5.48.072,
	deductions to businesses
	selling water and electricity
	for resale (5.45, 5.48)
121992	Amends § 4.13.010; adds §
121//2	4.20.401, electric utility
	compensation program
	(4.13, 4.20)
121997	Amends § 23.47.010,
121///	maximum size of
	nonresidential use (23.47)
	TRANSPORTED IN THE TRANSPORT IN THE TRAN

d amending graphics, accuracy of

Seattle Municipal Code
June 2009 co 121998 up date
20.84.0
20.84.0
20.84.0
20.84.0
20.84.0
20.84.0

See ordinantes sections for and tables at this source

	te file
121998	Amends §§ 20.84.010,
121998	
	20.84.020, 20.84.030,
1 Co# 111	20.84.040; repeals §§
u ini	20.84.050, 20.84.060,
	20.84.070, 20.84.080,
	20.84.090, 20.84.100,
	20.84.110, 20.84.120,
	20.84.130, 20.84.140,
	20.84.150, 20.84.160,
- a orea	20.84.170, 20.84.180,
ces er en	20.84.190, 20.84.200,
	20.84.210, 20.84.220,
	20.84.230; adds §§
and mille	20.84.055, 20.84.065,
Compact	
	20.84.075, 20.84.085,
140 (	20.84.095, 20.84.105,
	20.84.115, 20.84.205,
	20.84.215, 20.84.225,
	20.84.235, 20.84.300,
C310.	20.84.310, 20.84.320,
THE.	20.84.330, 20.84.400,
	20.84.410, 20.84.420,
	20.84.430, 20.84.500,
	20.84.510, 20.84.520,
	20.84.620, 20.84.630,
	20.84.640, relocation
	assistance (20.84)
122003	Amends §§ 11.31.010,
	12A.06.010, expands
	definition of the crime of
	assault (11.31, 12A.06)
122004	Adds § 12A.08.115,
	making or having burglar
	or auto theft tools (12A.08)
122007	Amends § 4.20.040,
122007	Assistant City Attorney
	compensation program
	(4.20)
122020	Amends § 21.28.040,
	wastewater services
	(21.28)
122025	Amends § 12A.20.060,
	enforcement
	prioritymarijuana
	(12A.20)
122027	Amends § 4.13.010, certain
122027	positions exempt from
122026	Civil Service status (4.13)
122036	Amends §§ 21.16.030,
	21.16.060, 21.16.065,
	21.16.070, 21.16.080,
	21.16.090, 21.16.280,
	21.16.380; adds §§
	21.16.055, 21.16.068, Side
	Sewer Code (21.16)
122043	
17.7043	Amends §§ 23.61.006,
1220.0	
122010	23.61.008, 23.61.014,
122010	Station Area Overlay District (23.61)

reference only.
d amending
graphics,
accuracy of

June 2009 co Text provide

See ordinan sections for and tables a this source

122045   Amends §§ 20.04.010, 20.04.070; adds § 20.04.235, public improvement districts (20.04)   Amends § 22.150.010, amendments to Seattle Residential Code (22.150)   Amends § 22.400.010, amendments to Seattle Mechanical Code (22.100)   Amends § 22.400.010, amendments to Seattle Building Code (22.100)   Amends § 22.100.010, amendments to Seattle Building Code (22.100)   Amends § 23.20.058, 23.28.03.23.44.014, 23.44.014, 23.44.024, 23.45.014, 23.45.056, 23.45.072, 23.55.30.02, 33.50.15, 23.55.30.02, 33.50.15, 23.55.30.02, 23.50.02, 23.50.010, 25.09.020, 25.09.040, 25.09.020, 25.09.040, 25.09.020, 25.09.040, 25.09.020, 25.09.040, 25.09.200, 25.09.120, 25.09.100, 25.09.120, 25.09.100, 25.09.25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.300, 25.09.300, 25.09.300, 25.09.352, 25.09.300, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.		gpar o	40 Sile	
20.04.235, public improvement districts (20.04)  122047  Amends § 22.150.010, amendments to Seattle Residential Code (22.150)  Amends § 22.400.010, amendments to Seattle Mechanical Code (22.400)  Amends § 22.100.010, amendments to Seattle Mechanical Code (22.400)  Amends § 23.100.010, amendments to Seattle Building Code (22.100)  Amends § 23.20.58, 23.28.030, 23.44.014, 23.44.024, 23.45.014, 23.44.014, 23.44.008, 23.44.010, 23.44.014, 23.45.056, 23.45.072, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 25.09.010, 25.09.000, 25.09.080, 25.09.000, 25.09.080, 25.09.120, 25.09.100, 25.09.120, 25.09.100, 25.09.120, 25.09.240, 25.09.250, 25.09.250, 25.09.260, 25.09.280, 25.09.360, 25.09.280, 25.09.360, 25.09.380, 25.09.360, 25.09.382, 25.09.360, 25.09.383, 25.09.400, adds § 25.09.015, 25.09.050, 25.09.352, 25.09.352, 25.09.353, 25.09.410, 25.09.430, 25.09.430, 25.09.430, 25.09.440, 25.09.450, 25.09.480; repeals § 25.09.140, 25.09.480; repeals § 25.09.140, 25.09.345, 25.09.352, 25.09.3		4 440012	TE 1110	
20.04.235, public improvement districts (20.04)  122047  Amends § 22.150.010, amendments to Seattle Residential Code (22.150)  Amends § 22.400.010, amendments to Seattle Mechanical Code (22.400)  Amends § 22.100.010, amendments to Seattle Mechanical Code (22.400)  Amends § 23.100.010, amendments to Seattle Building Code (22.100)  Amends § 23.20.58, 23.28.030, 23.44.014, 23.44.024, 23.45.014, 23.44.014, 23.44.008, 23.44.010, 23.44.014, 23.45.056, 23.45.072, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 23.53.002, 25.09.010, 25.09.000, 25.09.080, 25.09.000, 25.09.080, 25.09.120, 25.09.100, 25.09.120, 25.09.100, 25.09.120, 25.09.240, 25.09.250, 25.09.250, 25.09.260, 25.09.280, 25.09.360, 25.09.280, 25.09.360, 25.09.380, 25.09.360, 25.09.382, 25.09.360, 25.09.383, 25.09.400, adds § 25.09.015, 25.09.050, 25.09.352, 25.09.352, 25.09.353, 25.09.410, 25.09.430, 25.09.430, 25.09.430, 25.09.440, 25.09.450, 25.09.480; repeals § 25.09.140, 25.09.480; repeals § 25.09.140, 25.09.345, 25.09.352, 25.09.3		122045	Amends §§ 20.04.010,	
improvement districts (20.04)  Amends § 22.150.010, amendments to Seattle Residential Code (22.150)  122048  Amends § 22.400.010, amendments to Seattle Mechanical Code (22.400)  Amends § 22.100.010, amendments to Seattle Building Code (22.100)  122050  Amends § 23.22.058, 23.28.030,23.44.008, 23.44.010, 23.44.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.040, 25.09.000, 25.09.040, 25.09.000, 25.09.120, 25.09.100, 25.09.120, 25.09.100, 25.09.120, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.290, 25.09.300, 25.09.380, 25.09.400; above the company of the company o			20.01.070, adds 3	
122047   Amends § 22.150.010, amendments to Seattle   Residential Code (22.150)   Amends § 22.400.010, amendments to Seattle   Mechanical Code (22.400)   122049   Amends § 22.100.010, amendments to Seattle   Building Code (22.100)   23.28.030.23.44.004, 23.34.008, 23.28.030.23.44.008, 23.28.030.23.44.004, 23.45.014, 23.44.024, 23.45.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.020, 23.53.025, 23.53.030, 23.54.010, 25.09.002, 25.09.0010, 25.09.002, 25.09.0010, 25.09.002, 25.09.004, 25.09.002, 25.09.100, 25.09.120, 25.09.160, 25.09.180, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.300, 25.09.230, 25.09.300, 25.09.230, 25.09.300, 25.09.230, 25.09.300, 25.09.330, 25.09.352, 25.09.350, 25.09.352, 25.09.352, 25.09.350, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.440, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.352, 25.09.35		1 Cor Mi		
122047   Amends § 22.150.010, amendments to Seattle Residential Code (22.150)   Amends § 22.400.010, amendments to Seattle Mechanical Code (22.400)   Memods § 22.100.010, amendments to Seattle Building Code (22.100)   Memods § 23.22.058, 23.28.030.23.44.008, 23.44.010, 23.44.014, 23.45.06, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.040, 25.09.002, 25.09.040, 25.09.005, 25.09.080, 25.09.160, 25.09.120, 25.09.160, 25.09.180, 25.09.160, 25.09.220, 25.09.240, 25.09.240, 25.09.240, 25.09.240, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.410, 25.09.420, 25.09.45, 25.09.450, 25.09	e	d in		
amendments to Seattle Residential Code (22.150) Amends \$22.40.010, amendments to Seattle Mechanical Code (22.400) Amends \$22.100.010, amendments to Seattle Building Code (22.100)  Amends \$2.20.08, 23.28.030,23.44.008, 23.44.010, 23.44.014, 23.44.024, 23.45.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.53.025, 23.53.030, 23.53.025, 23.53.030, 23.53.025, 23.53.030, 23.53.025, 23.53.030, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.040, 25.09.002, 25.09.040, 25.09.003, 25.09.080, 25.09.100, 25.09.120, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.280, 25.09.300, 25.09.380, 25.09.400; adds \$\$\$25.09.30, 25.09.360, 25.09.380, 25.09.400; adds \$\$\$\$\$\$\$25.09.35, 25.09.050, 25.09.35, 25.09.300, 25.09.45, 25.09.300, 25.09.45, 25.09.300, 25.09.45, 25.09.300, 25.09.440, 25.09.450, 25.09.450, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.340, 25.09.345, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.356; renumbers \$\$\$\$25.09.120, 25.09.440 as \$\$\$\$25.09.352, 25.09.350, 25.09.356; renumbers \$\$\$\$\$\$25.09.352, 25.09.350, 25.09.356; renumbers \$\$\$\$\$\$25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.356; renumbers \$\$\$\$\$25.09.320, 25.09.440 as \$\$\$\$25.09.320, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.353, 23.90, 25.09.350, 23.90, 25.09.90		122047		
Residential Code (22.150) Amends § 22.400.010, amendments to Seattle Mechanical Code (22.400)  122049 Amends § 22.100.010, amendments to Seattle Building Code (22.100)  122050 Amends § 23.22.058, 23.28.030,23.44.014, 23.44.024, 23.45.014, 23.45.056, 23.45.012, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.010, 25.09.006, 25.09.080, 25.09.100, 25.09.120, 25.09.160, 25.09.180, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.280, 25.09.300, 25.09.280, 25.09.300, 25.09.380, 25.09.400; adds § 25.09.015, 25.09.030, 25.09.380, 25.09.400; adds § 25.09.015, 25.09.030, 25.09.380, 25.09.400; adds § 25.09.015, 25.09.030, 25.09.380, 25.09.400; adds § 25.09.015, 25.09.030, 25.09.45, 25.09.050, 25.09.45, 25.09.050, 25.09.45, 25.09.050, 25.09.45, 25.09.400; 25.09.45, 25.09.300, 25.09.45, 25.09.300, 25.09.350, 25.09.400; 25.09.350, 25.09.330, 25.09.450, 25.09.450, 25.09.460, 25.09.480; repeals § 25.09.140, 25.09.460, 25.09.480; repeals § 25.09.450, 25.09.355, 25.09.355, 25.09.355, 25.09.355, 25.09.355, 25.09.355, 25.09.355, 25.09.356; renumbers § 25.09.420, 25.09.440 as § 25.09.420, 25.09.441 areas (23.22, 23.28, 23.44, 23.45, 23.55, 23.84, 23.90, 25.09)		122047		
122048				
amendments to Seattle Mechanical Code (22,400) Amends § 22,100,010, amendments to Seattle Building Code (22,100)  Amends § 23,20,88, 23,28,030,23,44,008, 23,44,014, 23,44,024, 23,45,014, 23,44,024, 23,45,014, 23,45,056, 23,45,072, 23,53,020, 23,53,025, 23,53,030, 23,53,015, 23,53,030, 23,53,015, 23,53,030, 23,53,015, 23,53,030, 23,53,016, 25,09,000, 25,09,040, 25,09,000, 25,09,040, 25,09,160, 25,09,120, 25,09,160, 25,09,120, 25,09,240, 25,09,260, 25,09,230, 25,09,200, 25,09,240, 25,09,260, 25,09,280, 25,09,300, 25,09,380, 25,09,400, adds § 25,09,015, 25,09,030, 25,09,055, 25,09,030, 25,09,055, 25,09,030, 25,09,055, 25,09,030, 25,09,055, 25,09,030, 25,09,055, 25,09,030, 25,09,440, 25,09,450, 25,09,440, 25,09,450, 25,09,440, 25,09,450, 25,09,440, 25,09,450, 25,09,340, 25,09,345, 25,09,350, 25,09,355, repeals § 25,09,140, 25,09,340, 25,09,355, 25,09,355, 25,09,355, 25,09,355, 25,09,355, 25,09,355, 25,09,355, 25,09,355, 25,09,356, renumbers § 25,09,420, 25,09,440, 25,09,450, 25,09,450, 25,09,356, renumbers § 25,09,420, 25,09,430, 25,09,356, renumbers § 25,09,420, 25,09,430, 25,09,356, renumbers § 25,09,520, 25,09,530, environmentally critical areas (23,22, 23,28, 23,44, 23,45, 23,53, 23,84, 23,90, 25,09)		122048		
Mechanical Code (22.400) Amends § 22.100.010, amendments to Seattle Building Code (22.100)  122050 Amends § 23.22.058, 23.28.030,23.44.008, 23.34.010, 23.44.014, 23.44.024, 23.45.014, 23.44.024, 23.45.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.010, 25.09.002, 25.09.040, 25.09.160, 25.09.180, 25.09.160, 25.09.180, 25.09.200, 25.09.20, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.320, 25.09.300, 25.09.320, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.380, 25.09.400; adds §§ 25.09.150, 25.09.300, 25.09.335, 25.09.310, 25.09.340, 25.09.450, 25.09.450, 25.09.450, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.352, 25.09.355, 25.09.355, 25.09.355, 25.09.356; renumbers §§ 25.09.340, 25.09.340, 25.09.352, 25.09.350, 25.09.355, 25.09.350, 25.09.352, 25.09.350,		122048		
Amends § 22.100.010, amendments to Seattle Building Code (22.100)  122050  Amends §§ 23.22.058, 23.28.030,23.44.008, 23.44.014, 23.44.014, 23.44.014, 23.45.014, 23.45.014, 23.45.015, 23.53.020, 23.53.025, 23.53.020, 23.53.025, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.040, 25.09.000, 25.09.040, 25.09.100, 25.09.100, 25.09.100, 25.09.100, 25.09.100, 25.09.100, 25.09.160, 25.09.180, 25.09.200, 25.09.200, 25.09.200, 25.09.300, 25.0		ces crea		
amendments to Seattle Building Code (22.100)  Amends §§ 23.22.058, 23.28.030,23.44.008, 23.44.010, 23.44.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.53.025, 23.53.030, 23.53.4010, 23.90.002, 25.09.010, 25.09.002, 25.09.040, 25.09.002, 25.09.100, 25.09.100, 25.09.120, 25.09.100, 25.09.180, 25.09.200,25.09.220, 25.09.240, 25.09.360, 25.09.320, 25.09.360, 25.09.320, 25.09.360, 25.09.320, 25.09.360, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.300, 25.09.380, 25.09.400, 25.09.055, 25.09.330, 25.09.055, 25.09.330, 25.09.055, 25.09.330, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.340, 25.09.355, 25.09.340, 25.09.356, repeals §§ 25.09.140, 25.09.340, 25.09.356, renumbers §§ 25.09.420, 25.09.355, 25.09.356, renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.440, as §§ 25.09.520, 25.09.350, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)	B	122040		
Building Code (22.100)  Amends § 23.22.058, 23.28.030,23.44.008, 23.44.010, 23.44.014, 23.45.056, 23.45.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.010, 25.09.000, 25.09.080, 25.09.100, 25.09.120, 25.09.160, 25.09.180, 25.09.200, 25.09.200, 25.09.200, 25.09.300, 25.09.280, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.385, 25.09.300, 25.09.385, 25.09.300, 25.09.385, 25.09.350, 25.09.340, 25.09.480, repeals § 25.09.410, 25.09.340, 25.09.480; repeals § 25.09.140, 25.09.340, 25.09.356; renumbers § 25.09.420, 25.09.350, 25.09.356; renumbers § 25.09.420, 25.09.440 as § 25.09.20, 25.09.350, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)		122049		
Amends §§ 23.22.058, 23.28.030,23.44.008, 23.44.010, 23.44.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.002, 23.53.025, 23.53.002, 23.53.025, 23.53.000, 22.5.09.010, 25.09.020, 25.09.040, 25.09.060, 25.09.100, 25.09.100, 25.09.180, 25.09.160, 25.09.180, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.280, 25.09.300, 25.09.280, 25.09.300, 25.09.320, 25.09.360, 25.09.380, 25.09.360, 25.09.380, 25.09.300, 25.09.35, 25.09.360, 25.09.35, 25.09.360, 25.09.35, 25.09.360, 25.09.35, 25.09.360, 25.09.35, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.35, 25.09.400; adds §§ 25.09.016, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.480, 25.09.350, 25.09.350, 25.09.356; renumbers §§ 25.09.420, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.550, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)		COMPLE		5
23.28.030,23.44.008, 23.44.010, 23.44.014, 23.45.056, 23.45.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.090.010, 25.09.020, 25.09.040, 25.09.060, 25.09.120, 25.09.100, 25.09.120, 25.09.160, 25.09.180, 25.09.200,25.09.200, 25.09.240, 25.09.200, 25.09.240, 25.09.300, 25.09.320, 25.09.360, 25.09.380, 25.09.360, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.050, 25.09.380, 25.09.300, 25.09.381, 25.09.400; adds §§ 25.09.015, 25.09.050, 25.09.35, 25.09.310, 25.09.35, 25.09.310, 25.09.340, 25.09.450, 25.09.460, 25.09.450, 25.09.460, 25.09.450, 25.09.460, 25.09.450, 25.09.340, 25.09.356; repeals §§ 25.09.140, 25.09.340, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.450, as environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)	5	122050		
23.44.010, 23.44.014, 23.44.024, 23.45.014, 23.45.056, 23.45.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.040, 25.09.060, 25.09.060, 25.09.100, 25.09.100, 25.09.180, 25.09.100, 25.09.180, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.200, 25.09.350, 25.09.350,		122030		
23.44.024, 23.45.014, 23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.090.010, 25.09.000, 25.09.040, 25.09.000, 25.09.120, 25.09.160, 25.09.180, 25.09.200, 25.09.20, 25.09.240, 25.09.20, 25.09.240, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.35, 25.09.330, 25.09.35, 25.09.430, 25.09.400, 25.09.440, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.450, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.350, 25.09.352, 25.09.353, 25.09.352, 25.09.350, 25.09.		and to co		
23.45.056, 23.45.072, 23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.090.010, 25.09.060, 25.09.080, 25.09.160, 25.09.180, 25.09.160, 25.09.20, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.050, 25.09.045, 25.09.050, 25.09.35, 25.09.310, 25.09.35, 25.09.410, 25.09.40, 25.09.440, 25.09.40, 25.09.450, 25.09.35, 25.09.410, 25.09.35, 25.09.35, 25.09.35, 25.09.35, 25.09.40, 25.09.450, 25.09.40, 25.09.450, 25.09.400, 25.09.450, 25.09.400, 25.09.450, 25.09.440, 25.09.450, 25.09.450, 25.09.350, 25.09.350, 25.09.355, 25.09.350, 25.09.356; renumbers §§ 25.09.356; renumbers §§ 25.09.356; renumbers §§ 25.09.356, renumbers §§ 25.09.350, 25.09.440 as §§ 25.09.520, 25.09.450 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)	8			
23.53.010, 23.53.015, 23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.040, 25.09.020, 25.09.040, 25.09.060, 25.09.080, 25.09.160, 25.09.120, 25.09.160, 25.09.120, 25.09.200,25.09.20, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.280, 25.09.300, 25.09.320, 25.09.300, 25.09.380, 25.09.300, 25.09.35, 25.09.300, 25.09.35, 25.09.300, 25.09.440, 25.09.50, 25.09.45, 25.09.30, 25.09.45, 25.09.400, 25.09.45, 25.09.400, 25.09.35, 25.09.300, 25.09.35, 25.09.350, 25.09.440, 25.09.450, 25.09.450, 25.09.450, 25.09.340, 25.09.450, 25.09.340, 25.09.450, 25.09.340, 25.09.35, 25.09.350, 25.09.352, 25.09.350, 2				
23.53.020, 23.53.025, 23.53.030, 23.84.010, 23.90.002, 25.09.010, 25.09.010, 25.09.020, 25.09.040, 25.09.060, 25.09.180, 25.09.180, 25.09.180, 25.09.180, 25.09.280, 25.09.280, 25.09.280, 25.09.280, 25.09.280, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.45, 25.09.050, 25.09.050, 25.09.300, 25.09.45, 25.09.400, 25.09.45, 25.09.45, 25.09.45, 25.09.45, 25.09.45, 25.09.300, 25.09.35, 25.09.45, 25.09.45, 25.09.350, 25.09.45, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.450, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.355, 25.09.356; renumbers §\$ 25.09.420, 25.09.440 as §\$ 25.09.250, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)		C:10		
23.53.030, 23.84.010, 23.90.002, 25.090.010, 25.09.020, 25.09.040, 25.09.060, 25.09.080, 25.09.100, 25.09.120, 25.09.160, 25.09.180, 25.09.200, 25.09.20, 25.09.240, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.300, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.050, 25.09.045, 25.09.050, 25.09.045, 25.09.050, 25.09.405, 25.09.400, 25.09.406, 25.09.480, repeals §§ 25.09.410, 25.09.440, 25.09.480, repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.356; renumbers §§ 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  Amends § 23.41.010	$\supseteq$	IIIC.		
23.90.002, 25.09.010, 25.09.020, 25.09.040, 25.09.020, 25.09.080, 25.09.080, 25.09.100, 25.09.180, 25.09.200, 25.09.220, 25.09.240, 25.09.260, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.300, 25.09.380, 25.09.400; adds §\$ 25.09.015, 25.09.030, 25.09.050, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)				
25.09.020, 25.09.040, 25.09.060, 25.09.080, 25.09.100, 25.09.120, 25.09.160, 25.09.180, 25.09.200,25.09.220, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.380, 25.09.360, 25.09.380, 25.09.300, 25.09.045, 25.09.030, 25.09.045, 25.09.030, 25.09.055, 25.09.30, 25.09.045, 25.09.30, 25.09.045, 25.09.30, 25.09.45, 25.09.30, 25.09.45, 25.09.30, 25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals § 25.09.140, 25.09.340, 25.09.35, 25.09.350, 25.09.352, 25.09.350, 25.09.356; renumbers § 25.09.356; renumbers § 25.09.420, 25.09.440 as § 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)				
25.09.060, 25.09.080, 25.09.100, 25.09.120, 25.09.160, 25.09.180, 25.09.200,25.09.220, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.320, 25.09.360, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.345, 25.09.350, 25.09.350, 25.09.352, 25.09.350, 25.09.356; renumbers §§ 25.09.440, 25.09.350, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.430 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)				
25.09.100, 25.09.120, 25.09.160, 25.09.160, 25.09.180, 25.09.200, 25.09.200, 25.09.220, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.380, 25.09.360, 25.09.380, 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.030, 25.09.45, 25.09.050, 25.09.45, 25.09.45, 25.09.410, 25.09.420, 25.09.440, 25.09.440, 25.09.440, 25.09.440, 25.09.440, 25.09.440, 25.09.440, 25.09.345, 25.09.340, 25.09.345, 25.09.340, 25.09.356, repeals § 25.09.356, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.356, 25.09.350, 25.09.350, 25.09.352, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, 25.09.350, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010				
25.09.160, 25.09.180, 25.09.200,25.09.220, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.320, 25.09.360, 25.09.380, 25.09.400; adds §\$ 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.45, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.440, 25.09.450, 25.09.340, 25.09.35c, repeals §\$ 25.09.140, 25.09.340, 25.09.35c, renumbers §\$ 25.09.35c; renumbers §\$ 25.09.35c, 25.09.350, 25.09.35c, 25.09.440 as §\$ 25.09.520, 25.09.440 as §\$ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)			· · · · · · · · · · · · · · · · · · ·	
25.09.200,25.09.220, 25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.320, 25.09.360, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.35c, 25.09.350, 25.09.35c; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.440 as §§ 25.09.520, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)				
25.09.240, 25.09.260, 25.09.280, 25.09.300, 25.09.320, 25.09.360, 25.09.380, 25.09.400; adds §\$ 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.430, 25.09.440, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §\$ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.350, 25.09.356; renumbers §\$ 25.09.420, 25.09.440 as §\$ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010				
25.09.280, 25.09.300, 25.09.320, 25.09.360, 25.09.380, 25.09.400; adds §\$ 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §\$ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §\$ 25.09.420, 25.09.440 as §\$ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010				
25.09.320, 25.09.360, 25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010				
25.09.380, 25.09.400; adds §§ 25.09.015, 25.09.030, 25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010				
\$\ \{25.09.015, 25.09.030, \\ 25.09.045, 25.09.050, \\ 25.09.055, 25.09.330, \\ 25.09.335, 25.09.410, \\ 25.09.420, 25.09.430, \\ 25.09.460, 25.09.480; \\ repeals \{\}\{\}\{25.09.340, 25.09.345, \\ 25.09.340, 25.09.345, \\ 25.09.350, 25.09.352, \\ 25.09.350, 25.09.356; \\ renumbers \{\}\{\}\{25.09.420, \\ 25.09.440 as \{\}\{\}\{25.09.520, \\ 25.09.530, environmentally \\ critical areas (23.22, 23.28, \\ 23.44, 23.45, 23.53, 23.84, \\ 23.90, 25.09)  Amends \{\}\{23.41.010}				
25.09.045, 25.09.050, 25.09.055, 25.09.330, 25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §\$ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §\$ 25.09.420, 25.09.440 as §\$ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010				
25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09) 121891  Amends § 23.41.010			, ,	
25.09.335, 25.09.410, 25.09.420, 25.09.430, 25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09) 121891  Amends § 23.41.010			25.09.055, 25.09.330,	
25.09.440, 25.09.450, 25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010				
25.09.460, 25.09.480; repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010			25.09.420, 25.09.430,	
repeals §§ 25.09.140, 25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010			25.09.440, 25.09.450,	
25.09.340, 25.09.345, 25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §\$ 25.09.420, 25.09.440 as §\$ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09) 121891 Amends § 23.41.010			25.09.460, 25.09.480;	
25.09.350, 25.09.352, 25.09.355, 25.09.356; renumbers §\$ 25.09.420, 25.09.440 as §\$ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09) 121891 Amends § 23.41.010			repeals §§ 25.09.140,	
25.09.355, 25.09.356; renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891  Amends § 23.41.010			25.09.340, 25.09.345,	
renumbers §§ 25.09.420, 25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09) 121891 Amends § 23.41.010			25.09.350, 25.09.352,	
25.09.440 as §§ 25.09.520, 25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09) 121891 Amends § 23.41.010			25.09.355, 25.09.356;	
25.09.530, environmentally critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891 Amends § 23.41.010				
critical areas (23.22, 23.28, 23.44, 23.45, 23.53, 23.84, 23.90, 25.09)  121891 Amends § 23.41.010				
23.44, 23.45, 23.53, 23.84, 23.90, 25.09) 121891 Amends § 23.41.010				
23.90, 25.09) 121891 Amends § 23.41.010				
121891 Amends § 23.41.010				
122033 Amends § 23.41.010				
		122033	Amends § 23.41.010	

eference only. amending graphics,

Seattle Municipal Code

June 2009 Co 122054 Amends Zoning Map, pp. 100, 101, 102, 108, 109, 110, 115, 116; renumbers ch. 23.49, subch. I, §§ 23.49.001--23.49.041, subch. II, §§ 23.49.042--23.49.058, subch. III, §§ 23.49.090--23.49.108, subch. IV, §§

See ordinances comsections for comand tables and tables and this source file.

ces crea comple nd to co file.

23.49.140--23.49.166, subch. V, §§ 23.49.170--23.49.178, subch. VI, §§ 23.49.198--23.49.208, subch. VII, §§ 23.49.223--23.49.248, subch. VIII, §§ 23.49.300--23.49.306, subch. IX, 23.43.318--23.49.332, subch. X, 23.49.336--23.49.338; amends § 23.41.004; repeals § 23.41.012; adds § 23.41.012; amends §§ 23.42.124, 23.49.002, 23.49.008; adds § 23.49.009; repeals § 23.49.010; renumbered § 23.49.026; add § 23.49.010; amends §§ 23.49.011, 23.49.012, 23.49.013, 2349.014; repeals § 23.49.015; adds § 23.49.015; repeals § 23.49.016; renumbered and amended § 23.49.009 as 23.49.016; renumbered and amended § 23.49.027 as 23.49.017; repeals §§ 23.49.017, 23.49.018, 23.49.019, 23.49.020; adds §§ 23.49.018, 23.49.019, 23.49.020; repeals § 23.49.025; adds § 23.49.025; amends §§ 23.49.032, 23.49.034, 23.49.035, 23.49.036;

ference only.

amending

raphics,

curacy of

Seattle Municipal Code

June 2009 co de up da te 1

repeals § 8 2
23.49.039, 2
renumbered as exhibit 23
§ 23.49.041;
23.49.042, 2
23.49.045 co

See ordinances comsections for comand tables and tables and this source file.

23.49 23.49 map 2 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49 23.49

repeals §§ 23.49.037, 23.49.039, 23.49.041; renumbered map 23.4904A as exhibit 23.49.056F; add § 23.49.041; amends §§ 23.49.042, 23.49.044, 23.49.045, 23.49.046, 23.49.056; renumbered map 23.49041A as exhibit 23.49.056F; amends § 23.49.056F; amends § 23.49.060, 23.49.062, 23.49.064, 23.49.066, 23.49.076, 23.49.078; amends §§ 23.49.094,

23.49.064, 23.49.066, 23.49.076, 23.49.078; amends §§ 23.49.094, 23.49.096, 23.49.106, 23.49.116, 23.49.118, 23.49.120, 23.49.122, 23.49.122, 23.49.126, 23.49.130, 23.49.134,

23.49.130, 23.49.134, 23.49.136; amends §§ 23.49.146, 23.49.148, 23.49.162, 23.49.164, 23.49.306, 23.49.324, 23.49.332, 23.49.338, 23.54.015, 23.54.020, 23.54.030, 23.66.122,

23.57.013, 23.76.004, 23.76.006, 23.76.011, 23.76.032, 23.76.036, 23.76.040, 23.76.058, 23.76.060, 23.84.008, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016, 23.84.014, 23.84.016,

23.66.170, 23.74.010,

23.84.024, 23.84.025, 23.84.030, 23.84.032, 23.84.038, 23.84.042, 23.90.018, 23.90.020;

repeals ch. 23.49, maps 1F, 1N, 1M, 1O; renumbered ch. 23.49, maps 1H, 1I, 1J, 1K, 1L; adds ch. 23.49, maps 1G, 1H, 1I, 1J, 1K; amends ch. 23.49, maps 1A, 1G, 1I, 1J, 1K

122055 Amends § \$ 22.800.030, 22.802.012

122059 Amends § 22.600.020

122062 Repeals § 4.04.225

122063 Amends § \$ 4.04.030, 4.20.055, 4.20.190, 4.20.220, 4.24.010, 4.28.010, 4.34.005; adds § 4.04.280; repeals § § 4.04.145, 4.04.255, temporary workers (4.04)

 122064
 Amends § 4.13.020

 122072
 Amends § 23.60.482

 122091
 Amends § 21.04.430,

 21.04.440, 21.76.040

ference only.

amending

raphics,

For current Sing, 500 Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

Cibes -	to file
122108	Repeals ch. 6.235 (gas
122108	piping regulations and
	licensure); adds ch. 6.430
1 50° N	(gas piping regulations and
30 707	licenure)
122109	Repeals ch. 6.82
122109	(refrigeration systems; adds
	ch. 6.410 (refrigeration
	systems)
122115	Repeals ch. 6.230 (steam
ces cro	engineers and boiler
	firemen); adds ch. 6.420
- 400 50 6	(steam engineers and boiler
COMPA	firemen)
122125	Amends § 23.47.040,
140 0	zoning map, pp. 37, 38, 23,
and to c	24
122126	Amends § 3.28.920
122127	Amends § 3.28.905
122136	Amends Appendix I to
88800	Title 15
122142	Amends § 3.18.300
122152	Amends § 23.41.010
122163	Amends § 7.04.645.C
122173	23.69.030.C
122174	Amends §§ 21.76 (title),
	21.76.010 (title), 21.76.030
	(title), 21.76.040 (title),
	21.76.040.B; adds § 21.76.065
122176	Amends § 20.04.135
122170	Amends § 3.40.010
122181	Amends § 3.64.030
122182	Amends §§ 3.04.020,
122102	3.18.020, 3.20.020,
	3.22.020, 17.04.020
122184	Amends Zoning Map 173
122185	Amends Zoning Map 184
122188	Amends §§ 3.20.040, 3.20
	(title), 3.20.080; 3.20.100,
	3.20.120
122190	Amends §§ 23.22.062.A,
	B, 23.24.045.A, B,
	23.34.011, 23.44.008.F,
	23.44.012.A, 23.44.014.D,
	23.44.040.E, 23.44.041, 23.44.050.D, 23.84.004,
	23.84.008, 23.84.032, 23.90.018, 23.90.019
122191	Adds Ch. 5.37; amends §§
1441/1	3.02.125.A, 5.30.010,
	5.55.010, 5.55.040.A,
	5.55.060.A, 5.55.150.E,
	5.55.165, 5.55.220.A, B,
	5.55.230.A.

reference only.

desired amending
paphics,
ler
ccuracy of

June 2009 co

Text provide

See ordinant sections for and tables a this source

Sibar	0210
- Ag	te tile
122192	Adds Ch. 5.35; amends §§
<del>(30</del> 2	3.02.125.A, 5.30.010,
- 0 - m hi	5.45.090.BB, 5.55.010,
A for III	5.55.040.A, 5.55.060.A,
CI ZO	5.55.150.E, 5.55.165,
	5.55.220.A, B,
	5.55.230.A., 6.48.020
122198	Amends 23.42.040,
	23.60.022, 23.60.090,
- co orea	23.80.004; adds 23.60.023
122200	Amends §§ 10.02.030,
	10.02.060
122205	Amends §§ 3.06.040,
COMPA	3.12.020, 23.22.052.G,
	23.22.100, 23.24.035.E,
140 00	23.53.010, 23.53.015,
ma to o	23.53.020, 23.53.025,
,33	23.53.030.A, E,
000	23.71.008.E, 23.84.032,
<b>5112.</b>	23.84.036; adds §
2320.	23.53.004
122206	Amends Zoning Map 15
122208	Amends §§ 23.44.051,
	23.45.160, 23.54.015
122213	Amends § 7.25.040
122235	Amends §§ 23.41.012.B,
	C, 23.45.008.B, 23.49
	maps, 23.49.009.A,
	23.49.015.A, B,
	23.49.018.E, 23.49.019.B,
	H, 23.49.024.A,
	23.49.058.CE,
	23.49.248.B, 23.49.322,
	23.49.336, 23.84.025; rep.
100040	§§ 23.76.026.F,
122242	Amends §§ 4.16.030,
	4.16.070, 4.16.100,
122243	4.16.105; adds § 4.16.085
122243	Amends §§ 11.14.277,
	11.72.075, 11.72.260, 11.72.300, 11.72.357,
	11.72.435, 11.74.020,
	11.76.015, 11.76.020
122244	Rep. § 23.71.016.C; ren. §
122244	23.71.016.D; amends §
	23.71.010.D, amends § 23.71.018.A, B
122251	Amends § 4.24.210
122256	Amends § 5.48.055
122257	Amends § 14.06.010.A;
122237	adds § 14.08.015
122258	Adds ch. 6.250
122262	Amends §§ 21.40.050,
122202	21.40.060, 21.40.070,
	21.40.080, 21.40.085,
	21.40.040.B, 21.44.060.D, E
122267	
122267 122270	Amends § 12A.12.040 Adds § 5.30.065; amends
122210	
	§§ 5.55.090, 5.55.095, 5.55.100, 5.55.110,
	5.55.120, 6.10.080.C
<b>45</b>	J.JJ.120, 0.10.000.C

eference only.

d amending
graphics,
accuracy of

For current SN 5.55,120, 6.10.080.c the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

CIDai	
a anda	te file
122273	Amends §§ 23.47.016.D,
	23.54.015, 23.84.030,
1 0 - m hi	23.84.032; adds 23.47.039,
A FOR III	23.49.028, 23.50.049
122274	Amends §§ 11.16.121.B.2,
	11.16.300.D, 11.72.440.C;
	renumbers 11.76.015.E and
	F; adds 11.76.015.D
122281	Amends §§ 21.33.030.D,
on orea	21.76.040.A.2
122282	Amends §§ 21.49.020,
	21.49.030, 21.49.040,
1000016	21.49.052.A, 21.49.055.A,
COMPLE	21.49.057, 21.49.058,
	21.49.060.A, B, J,
1 4 - 01	21.49.065, 21.49.081,
and to co	21.49.090.D, 21.49.100.B,
HILLS OF	C, H, K, 2149.110.E;
	repeals §§ 21.49.021 and
C310.	21.49.058; and adds §§
1110.	21.49.085 and 21.49.110.X
122283	Amends § 18.28.010
122287	Amends § 17.19.010
122288	Amends § 17.16.015
122292	Amends §§ 21.28.040.B.1,
	21.76.040.A.1
122297	Amends §§
	22.900A.040.C,
	22.900A.050.B,
	22.900B.010,
	22.900B.020.D,
	22.900B.060.B,
	22.900C.010,
	22.900D.010,
	22.900.D.090,
	22.900D.100,
	22.900D.110,
	22.900D.140,
	22.900D.160,
	22.900D.160, 22.900E.020,
	22.900E.030, 22.900E.040,
	22.900E.050,
	22.900E.060.C,
	22.900F.010, 22.900F.020,
	22.900F.050, 22.900G.030

eference only.

l amending

graphics,

ccuracy of

Seattle Municipal Code eference only. June 2009 co 122310 U Amended 23.60.246, 23.60.306, 23.60.365, Text provided for hi 23.60.368, 23.60.424, 23.60.426, 23.60.430, 23.60.484, 23.60.486, 23.60.488, 23.60.544, amending 23.60.600, 23.60.604, 23.60.606, 23.60.608, 23.60.636, 23.60.660, See ordinances crea 23.60.664, 23.60.668, 23.60.670, 23.60.720, 23.60.724, 23.60.728, sections for comple 23.60.730, 23.60.780, 23.60.786, 23.60.788, 23.60.840, 23.60.844, and tables and to 23.60.848, 23.60.850, 23.60.900, 23.60.902, 23.60.906, 23.60.908, 23.60.918, 23.60.926, this source file. 23.60.924, 23.60.936, 23.60.938, 23.60.940; repealed ch. 23.84

Seattle Municipal Code 122311 Text provided for hi 23.30.010, 23.34.007, 23.34.010, 23.34.074, 23.34.076, 23.34.078, 23.34.080, 23.34.082, 23.42.042, 23.42.050,

See ordinan sections for and tables and to this source file.

Amends §§ 3.20.320.B, 10.08.140.M, 15.54.010, 23.34.086, 23.34.094.B.2, 23.40.002, 23.40.050.B, 23.41.010.A, 23.41.012, 23.42.108, 23.42.124, 23.42.126, 23.42.128.D, 23.42.130, 23.43.040.C, 23.44.006, 23.44.015, 23.44.022.A, 23.44.026.B, 23.44.048, 23.44.050, 23.45.004, 23.45.110.B, 23.45.124.B, 23.45.148,

23.45.152, 23.46.004.B, 23.48.006, 23.48.016.A, 23.48.019.A, 23.48.026, 23.48.028, 23.48.030, 23.48.032, 23.49.009.A, 23.49.019.A.1.a, 23.49.034.B, 23.49.044, 23.49.092, 23.49.144, 23.49.320, 23.49.338.A,

23.50.002.C, 23.50.012 Cht. A. 23.50.014.B. 23.50.015, 23.50.022.B,

23.50.024.B, 23.53.015.D, E, 23.53.030.E, 23.54.015, 23.54.016, 23.54.020, 23.54.030, 23.57.008.H, 23.57.012.C.1.c, d,

23.61.008, 23.61.012, 23.61.014, 23.66.122.A, 23.66.130.B, C, E,

23.66.322.A, 23.66.326.B, C, E, 23.66.342.B.1, 23.67.050, 23.69.008.C,

23.69.022, 23.69.024.B, 23.71.008.B, 23.71.016.A, 23.71.038, 23.71.044,

23.72.008, 23.73.008, 23.73.010, 23.74.008,

23.74.010.A.1, B.2, C.3, 23.76.026.B,

23.76.032.A.1.d,

23.91.002.A, 25.08.225, 25.08.500, and land use maps; repeals §§

23.34.077, 23.34.079, 23.34.088, Ch. 23.47, and § 23.61.010; adds 23.42.050. 23.42.052, 23.42.106.F,

Ch. 23.47A, 23.76.042.D, 23.84.001, Ch. 23.84A, 23.86.023

122316

Designates the Medical Dental Building as a landmark in ch. 25.32

eference only. amending

the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	40 File
122322	Amends § 5.45.100
122330	
122330	Amends, §§ 23.49.178.C, 23.66.122.A, 23.84A.038
122332	Amends § 3.26.010
122334	Amends 23.41.010
122343	
	Amends Zoning Map 8
122349	Amends § 4.20.190
122353	Designates the Treat House as a landmark in ch. 25.32
122356	Amends Zoning Map 139
122357	Adds §§
122337	20.47.01020.47.060
122364	Amends § 4.20.430.D
122370	Amends § 25.09.015.A,
	25.09.020.A, 25.09.030.A
122382	Amends § 22.504.010
122396	Adds §§
122370	10.52.03110.52.038,
	10.52.041, 10.52.042,
<b>S112</b> .	10.52.051; repeals §§
Prince	10.52.050, 10.52.080,
	10.52.090
122397	Amends §§ 22.202.050.F,
12207	22.206.160, 22.206.200.F,
	22.206.220, 22.206.230,
	22.206.280, 22.206.290,
	22.208.030, 22.208.050,
	22.208.150, 22.208.160;
	adds §§ 22.206.235,
	22.206.315; repeals
	22.206.300, 22.206.310,
	Ch. 22.207
122398	Amends zoning maps 152,
	153, 166, 167
122407	Amends 23.90.006,
	23.90.008, 23.90.014,
	23.90.01823.90.020,
	23.90.025, 23.91.006,
	23.91.012; added
	23.90.015; repealed
	23.90.022
122411	Amends §§ 23.47A.004,
	23.48.004, 23.50.012,
	23.54.015, 23.84A.002,
	23.84A.006, 23.84A.010;
	adds 23.47A.004.H;
	23.49.030, 23.50.012.E
122414	23.49.030, 23.50.012.E Amends § 4.36.193
122415	23.49.030, 23.50.012.E Amends § 4.36.193 Adds 3.125.0103.125.050
	23.49.030, 23.50.012.E Amends § 4.36.193 Adds 3.125.0103.125.050 Adds §§ 23.49.178.B,
122415 122435	23.49.030, 23.50.012.E Amends § 4.36.193 Adds 3.125.0103.125.050 Adds §§ 23.49.178.B, 23.66.140.C
122415	23.49.030, 23.50.012.E Amends § 4.36.193 Adds 3.125.0103.125.050 Adds §§ 23.49.178.B, 23.66.140.C Adds §§
122415 122435 122441	23.49.030, 23.50.012.E Amends § 4.36.193 Adds 3.125.0103.125.050 Adds §§ 23.49.178.B, 23.66.140.C Adds §§ 3.14.5403.14.570
122415 122435 122441 122443	23.49.030, 23.50.012.E Amends § 4.36.193 Adds 3.125.0103.125.050 Adds §§ 23.49.178.B, 23.66.140.C Adds §§ 3.14.5403.14.570 Amended 23.41.010
122415 122435 122441	23.49.030, 23.50.012.E  Amends § 4.36.193  Adds 3.125.0103.125.050  Adds §§ 23.49.178.B, 23.66.140.C  Adds §§ 3.14.5403.14.570  Amended 23.41.010  Amends §§ 21.36.010,
122415 122435 122441 122443	23.49.030, 23.50.012.E  Amends § 4.36.193  Adds 3.125.0103.125.050  Adds §§ 23.49.178.B, 23.66.140.C  Adds §§ 3.14.5403.14.570  Amended 23.41.010  Amends §§ 21.36.010, 21.36.012, 21.36.016,
122415 122435 122441 122443	23.49.030, 23.50.012.E  Amends § 4.36.193  Adds 3.125.0103.125.050  Adds §§ 23.49.178.B, 23.66.140.C  Adds §§ 3.14.5403.14.570  Amended 23.41.010  Amends §§ 21.36.010, 21.36.012, 21.36.016, 21.36.085.C, 21.44.010;
122415 122435 122441 122443	23.49.030, 23.50.012.E  Amends § 4.36.193  Adds 3.125.0103.125.050  Adds §§ 23.49.178.B, 23.66.140.C  Adds §§ 3.14.5403.14.570  Amended 23.41.010  Amends §§ 21.36.010, 21.36.012, 21.36.016, 21.36.085.C, 21.44.010; repeals §§ 21.36.96,
122415 122435 122441 122443	23.49.030, 23.50.012.E  Amends § 4.36.193  Adds 3.125.0103.125.050  Adds §§ 23.49.178.B, 23.66.140.C  Adds §§ 3.14.5403.14.570  Amended 23.41.010  Amends §§ 21.36.010, 21.36.012, 21.36.016, 21.36.085.C, 21.44.010; repeals §§ 21.36.96, 21.36.180, 21.36.185,
122415 122435 122441 122443	23.49.030, 23.50.012.E  Amends § 4.36.193  Adds 3.125.0103.125.050  Adds §§ 23.49.178.B, 23.66.140.C  Adds §§ 3.14.5403.14.570  Amended 23.41.010  Amends §§ 21.36.010, 21.36.012, 21.36.016, 21.36.085.C, 21.44.010; repeals §§ 21.36.96,

eference only.

l amending

graphics,

ccuracy of

For current 5141-217-20-21.44.090
the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
122448	Amends 23.60.022
122457	Amends § 2.04.250
122473	Amends § 10.09.040
122474	Adds §§
1621,300	10.11.01010.11.025
122479	Amended § 4.14.140
122489	Amends §§ 9.25.036.C.
122409	and 9.25.037.D
122491	Amends § 22.600.020
122491	
122496	Amends §§ 25.12.520,
	25.12.560, 25.12.570, 25.12.610, 25.12.620,
122407	25.12.630, 25.12.640
122497	Amends §§ 20.04.090.D.1,
1 1 - 01	23.34.004, 23.69.032.I,
and to co	23.76.004, 23.76.005.C,
	23.76.036, 23.76.040.B,
	23.76.050.D, 23.76.052,
C310	23.76.054, 23.76.056, 23.76.058, 23.76.060,
THE.	23.76.038, 23.76.060, 23.84A.030
122502	
122503	Amends Tit. 15, Apps. I, II
122508	Amends §§ 9.25.002,
	9.25.030.A, 9.25.050.A, C,
	9-25.052, 9.25.080.E,
	9.25.100.E, G, 9.26.020,
	23.42.052.F; adds §
122511	9.25.084.H
122511	Adds § 22.903.035
122513 122516	Adds § 3.28.812
122516	Amends §§ 21.40.050,
	21.40.060, 21.40.070,
	21.40.080, 21.40.085, 21.76.040.B
122517	Amends §§ 21.33.010,
122317	21.33.030, 21.33.050.B,
	21.33.030, 21.33.030.B, 21.33.070.A,
	21.76.040.A.2; adds
	21.33.120
122518	Amends §§ 21.28.040.B,
122316	21.76.040.A.1
122523	Amends §§ 21.49.020.A,
1-2-2-3	21.49.030, 21.49.040,
	21.49.052, 21.49.055.A,
	21.49.057.A, 21.49.110.X
122524	Amends §§ 23.49.011,
	23.49.013, 23.49.035
122528	Amends § 22.100.010
122529	Amends § 22.110.010
122530	Amends § 22.700.010
122531	Amends § 22.400.010
122532	Amends § 22.420.010
122533	Amends § 22.150.010  Amends § 22.150.010
122554	Amends § 22.130.010 Amends §§ 11.31.090,
122334	
122557	11.31.120 Amends § 5.80.020.B
122557	
122559	Amends Ch. 5.76
122561	Amends §§ 5.37.020,
	5.37.030, 5.37.050,
	5.37.070, 5.37.080

eference only.

l amending
graphics,
ccuracy of

For current SMC, 5.37.070, 5.37.080 the City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	te file
122563	Amends §§ 5.30.025,
122303	5.30.050, 5.30.060,
	5.45.075, 5.45.080,
	5.45.100, 5.48.050; adds §
U In .	
100564	5.45.081
122564	Amended §§ 3.02.125.A.,
	5.30.010, 5.30.060.B,
	5.55.010, 5.55.040.A,
	5.55.060.A, 5.55.150.E,
oc crea	5.55.165, 5.55.220,
CE2 C2	5.55.230.A.3; added §§
	5.46.0105.46.070
122567	Amends § 5.55.040.D
122575	Amends §§ 23.34.010,
	23.34.072.B
122577	Amends § 12A.06.115
122578	Amends § 25.05.675.P
122582	Amends §§ 23.49.008,
122382	23.49.015, 23.86.006
122597	· · · · · · · · · · · · · · · · · · ·
122587	Amends § 4.20.370
122588	Amends § 18.04.01;
	repealed §§
	18.04.02018.040;
	renumbered and amended §
	18.04.050 as § 18.04.020
122589	Amends § 11.30.220
122591	Amends § 4.13.010; adds §
	4.36.196
122594	Designates L'Amourita
	Apartment Building as
	landmark in ch. 25.32
122595	Designates Windham
122373	Apartments as landmark in
	ch. 25.32
122601	
122001	Amends §§ 23.50.027,
122707	23.50.028
122605	Amends § 23.66.140
122611	Amends §§ 23.48.016,
	23.49.014, 23.49.020.A,
	23.50.020,
	23.50.02623.50.028,
	23.76.026, 23.84A.006,
	23.84A. 038, 23.90.018,
	and 23.90.020; adds §§
	23.50.05123.50.053
122614	Adds §§ 25.08.068,
-	25.08.221, 25.08.501,
	25.08.89025.08.970
122615	Amends §§ 23.22.052,
14411.7	23.22.100, 23.24.020,
	23 24 035 22 24 040 4
	23.24.035, 23.24.040A,
	23.48.022, 23.53.015,
	23.48.022, 23.53.015, 23.53.020; adds §
	23.48.022, 23.53.015, 23.53.020; adds § 23.53.006
122636	23.48.022, 23.53.015, 23.53.020; adds § 23.53.006 Added § 4.34.035
	23.48.022, 23.53.015, 23.53.020; adds § 23.53.006 Added § 4.34.035 Amended §§ 2.04.060,
122636	23.48.022, 23.53.015, 23.53.020; adds § 23.53.006 Added § 4.34.035
122636	23.48.022, 23.53.015, 23.53.020; adds § 23.53.006 Added § 4.34.035 Amended §§ 2.04.060,
122636	23.48.022, 23.53.015, 23.53.020; adds § 23.53.006 Added § 4.34.035 Amended §§ 2.04.060, 2.04.070, 2.04.075,

d amending graphics, accuracy of

For current SMC, Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinan sections for and tables a this source

	STED	. C:10	
	a mada	te tille	
	122670	Amended §§ 23.41.004,	ference of
		23.54.015, 25.05.800,	Sign
	101	25.05.908; added	
	A FOR III	23.42.044	
	122679	Amended §§ 21.49.030,	
		21.49.040, 21.49.052,	40 -
		21.49.055, 21.49.057	
	122682	Amended §§ 21.33.010,	l amellusse
		21.33.030, 21.33.050,	6,335
	of orea	21.33.070.B, 21.33.090,	1-205
	ces cross	21.33.120.A; added	
<b>B</b>		21.33.040	Flab
	122702	Amended table 25.32	
	122705	Added §§	ccuracy of
<b>b</b>		18.14.01018.14.130	CCMI
	122713	Amended § 6.98.020	
	122714	Amended § 23.50.027.A.1	
	122715	Amended §§ 11.31.090,	
		11.31.120	
	122723	Amended § 15.22.020	
5	122728	Amended §§ 22.206.160.C,	
		22.903.020, 22.903.030,	
		22.903.035, 22.903.040,	
		22.903.050, 22.903.090;	
		added 22.903.065 and	
		22.903.120	
	122730	Amended §§ 5.73.010,	
		5.73.020, 5.73.030,	
		5.73.040, 5.73.050,	
		5.73.065, 5.73.070,	
		5.73.090, 5.73.100,	
		5.73.120	
	122734	Amended § 6.310.530	
	122738	Amended §§ 22.802.020,	
		23.47A.012, 23.47A.013,	
		25.09.015A, 25.09.020,	
		25.09.030, 25.09.055A,	
		25.09.520; added	
		23.54.020.k, 25.09.110	
	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	

eference only. amending graphics,

122752

122761

122763

122764

122771

122773

June 2009 co li upas Text provide d for hi

See ordinances crossections for compand tables and to this source file.

Amended §§ 11.14.257, 11.20.230, 11.22.025, 11.22.080, 11.30.340, 11.34.020, 11.40.040, 11.52.170, 11.53.020, 11.53.200, 11.53.205, 11.53.210, 11.53.220, 11.53.240, 11.53.290, 11.53.300, 11.56. 025; 11.56.320.C.4, 11.56.355, 11.57.160, 11.58.140, 11.58.195, 11.58.272, 11.58.295, 11.70.200, 11.72.065, 11.74.150, 11.74.160, 11.82.360, 11.82.400, 11.82.520, 11.84.070, 11.84.150, 11.84.200.A, 11.84.440, 25.08.480; added §§ 11.14.578; 11.84.370, 11.84.460, 11.84.480, 11.84.500; repealed 11.58.015 122744 Amended §§ 3.28.800, 3.28.805, 3.28.810, 3.28.812, 3.28.825, 3.28.850, 3.28.855, 3.28.860, 3.28.900, 3.28.905, 3.28.910; repealed § 3.28.915 122745 Amended § 21.49.020, 21.49.082 122750 Amended § 22.900C.010, table 25.32; added §§ 25.21.010--25.21.140 122751 Amended § 21.36.922; added §§ 21.36.084, 21.36.086

> Amended §§ 5.45.100, 21.36.922; added § 21.40.075

Amended §§ 11.31.121,

Amended §§ 6.310.110, 6.310.150, 6.310.300, 6.310.315, 6.310.320, 6.310.330, 6.310.340, 6.310.350, 6.310.450, 6.310.455, 6.310.460, 6.310.465, 6.310.4610; added 6.310.380

Amended §§ 5.55.040, 5.55.220, 5.55.225; repealed §§ 5.55.215,

Amended §§ 23.60.600, 23.60.606, 23.60.660, 23.60.720, 23.60.900

Amended § 22.100.010

11.72.230

5.55.216

eference only.

l amending

graphics,

For current State City Clerk the Office of the City Clerk

June 2009 co

Text provide

See ordinant sections for and tables a this source

	a canda	te file
	122778	Amended table 25.32
	122779	Amended § 11.16.280,
	- 0 - m 101	11.31.121; added
	A FOR III	11.14.087, 11.14.088,
-		11.14.089, 11.72.054
	122787	Amended Zoning Map 112
	122789	Amended §§ 12A.02.085,
	122789	12A.04.200.C,
		12A.06.035.F, 12A.06.180,
	- 0003	12A.08.010, 12A.08.130,
	ces crea	12A.10.040.D,
-		12A.16.040.D,
	10	12A.16.060, 12A.18.010,
	COMPLE	12A.18.090, 12A.24.150,
	Cosses	15.48.900, 16.20.220,
		18.12.255, 21.36.922;
	and to co	added §§ 12A.06.040.E,
4		
		12A.06.045, 12A.06.070,
	0210	21.36.415; repealed §§ 12A.24.050, 18.12.140
	122704	
	122794 122796	Added § 4.20.450
	122/90	Amended Zoning Maps 145 and 158
	122700	
ŀ	122798	Added § 6.310.740
	122800	Amended § 12A.10.070.B
	122802	Amended § 6.310.320
	122809	Added § 4.04.120.F., G.
	122816	Amended §§ 23.40.002,
		23.42.040, 23.42.104,
		23.47A.032.C,
		23.54.030.F, 23.76.004,
		23.76.006, 23.76.032
	122818	Amended table 25.32
	122821	Amended §§ 21.33.010,
		21.33.030
	122823	Amended §§ 23.43.008.B,
		23.44.010, 23.44.012,
		23.44.016, 23.44.017.B,
		23.44.022.K, 23.44.040,
		23.54.015.B, 23.84A.032,
	12221	23.86.006
	122824	Amended §§ 15.02.046,
		15.04.015, 15.16.010,
		15.16.04015.16.080,
		23.76.006; added §
		15.16.012; repealed
ļ		15.16.030
	122825	Amended §§ 3.02.125,
		25.02.030, 25.02.040,
		25.02.050,
		25.02.07025.02.100;
		added §§ 25.02.035,
		25.02.055, 25.02.065,
		25.02.110; repealed
Į		25.02.060
	122835	Amended §§ 23.48.002,
		23.48.010, zoning maps 72,
		73, 86, 87; added §§
		23.48.011, 23.48.016.C,
		23.54.020.L
	461	

eference only.

It amending graphics, accuracy of

June 2009 co

Text provide

See ordinant sections for and tables at this source

	te file
122841	Repealed § 21.04.420;
	amended §§ 21.04.430,
101	21.04.440, 21.76.040.A.3
122842	Amended §§ 5.30.020,
(B. 20)	5.30.025, 5.30.035,
	5.30.040, 5.30.050,
	5.45.090, 5.45.100,
	5.48.055, 21.36.016;
	repealed 5.48.072
122844	
122844	Amended §§ 21.36.012,
	21.36.085, 21.40.050,
-10	21.40.060, 21.40.070,
- ammile	21.40.080, 21.40.085,
COMBRE	21.76.040
122845	Amended §§ 7.04.038,
140 00	7.04.130, 7.04.145,
ma to o	7.04.175, 7.04.645,
1880	7.04.690
122846	Amended § 17.19.010.B
122847	Amended § 9.26.060
122849	Amended §§ 22.602.045,
12204)	22.602.050, 22.602.060,
	22.602.070
122950	
122850	Amended § 18.28.010
122852	Amended §§ 11.16.120,
	11.16.121
122853	Amended § 11.16.319
122855	Amended 22.206.280,
	22.900B.010, 22.900B.020,
	22.900C.010,
	22.900D.010,
	22.900D.09022.900D.110
	, 22.900D.140,
	22.900D.14522.900D.160
	, 22.900E.01022.900E.060
	, 22.900F.010,
	22.900F.020, 22.900F.040,
	22.900G.010, 23.90.018,
	23.91.016
122856	Amended § 11.23.120
1	(history note only)
122857	Amended §§ 17.16.010,
	17.16.015; added §
	17.16.020
122858	Added § 11.16.312
122868	Amended §§ 21.28.040,
122000	
i	21.33.030, 21.76.040
122976	A 11-1 9 F 70 000
122876	Added § 5.78.200
122880	Amended zoning map 114
	Amended zoning map 114 Amended §§ 23.48.011,
122880	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§
122880	Amended zoning map 114 Amended §§ 23.48.011,
122880	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§
122880 122882	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§ 23.58A.00223.58A.014
122880 122882	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§ 23.58A.00223.58A.014 Amended §§ 23.40.006, 23.90.018; added §
122880 122882 122901	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§ 23.58A.00223.58A.014 Amended §§ 23.40.006, 23.90.018; added § 23.40.007
122880 122882	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§ 23.58A.00223.58A.014 Amended §§ 23.40.006, 23.90.018; added § 23.40.007 Amended §§ 21.49.040,
122880 122882 122901 122903	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§ 23.58A.00223.58A.014 Amended §§ 23.40.006, 23.90.018; added § 23.40.007 Amended §§ 21.49.040, 21.76.030
122880 122882 122901	Amended zoning map 114 Amended §\$ 23.48.011, 23.48.016; added §\$ 23.58A.00223.58A.014 Amended §\$ 23.40.006, 23.90.018; added § 23.40.007 Amended §\$ 21.49.040, 21.76.030 Amended §\$ 21.40.030,
122880 122882 122901 122903	Amended zoning map 114 Amended §§ 23.48.011, 23.48.016; added §§ 23.58A.00223.58A.014 Amended §§ 23.40.006, 23.90.018; added § 23.40.007 Amended §§ 21.49.040, 21.76.030

eference only.
d amending
graphics,
accuracy of

For current Sitter, J. 1.40.07

The Office of the City Clerk

June 2009 co

Text provide

See ordinantes sections for and tables at this source

		te file
	122913	Added § 21.04.420
	122916	Amended § 5.48.050
İ	122918	Amended zoning maps
0	4 for III	117, 118
	122919	Amended §§ 25.11.020,
		25.11.030, 25.11.040,
ŀ	122022	25.11.100
	122923	Amended §§ 25.08.040, 25.08.050, 25.08.060,
	0102	25.08.100, 25.08.160,
	ces crea	25.08.180, 25.08.190,
		25.08.230, 25.08.270,
	- ammle	25.08.300, 25.08.320,
P	COIIIB	25.08.330, 25.08.340,
		25.08.380, 25.08.390,
	and to CO	25.08.400, 25.08.410,
2	iller co	25.08.420, 25.08.425, 25.08.430, 25.08.470,
		25.08.480, 25.08.485,
	Calle.	25.08.510, 25.08.530,
2	Ille.	25.08.540, 25.08.545,
		25.08.550, 25.08.560,
		25.08.580, 25.08.590,
		25.08.600, 25.08.610,
		25.08.630, 25.08.640,
		25.08.650, 25.08.670, 25.08.680, 25.08.700,
		25.08.760, 25.08.700, 25.08.710, 25.08.730,
		25.08.740, 25.08.760,
		25.08.800, 25.08.820,
		25.08.960; added §§
		25.08.069, 25.08.155,
		25.08.165, 25.08.168,
		25.08.615, 25.08.655, 25.08.720, 25.08.765;
		repealed §§ 25.08.130,
		25.08.240, 25.08.535,
		25.08.620, 25.08.750,
		25.08.770, 25.08.780,
		25.08.790, 25.08.810
	122924	Amended § 25.05.675.P
ŀ	122925 122927	Amended § 23.50.012
	122921	Amended § 6.250.030.B.; added § 6.250.065
	122935	Amended §§ 23.41.012,
		23.47A.004.G,
		23.47A.005, 23.47A.006,
		23.47A.008,
		23.47A.01223.47A.014,
		23.47A.016, 23.47A.027, 23.47A.032, 23.50.012,
		23.54.015, 23.54.020,
		23.54.030, 23.66.140,
		23.74.010, 23.84A.004,
		23.84A.014, 23.84A.032,
		23.84A.036, 23.84A.038,
		23.84A.042; added §
	122939	23.86.019
	122942	Amended § 12A.10.110  Amended zoning map 165
	122947	Amended table 25.32
ı	wont SM	0,000

eference only.

d amending graphics, accuracy of

For current Sivie,
the Office of the City Clerk

June 2009 co 122948 122949

 122948
 Amended table 25.32

 122949
 Amended table 25.32

 122950
 Amended table 25.32

 122951
 Amended table 25.32

 122952
 Amended table 25.32

eference only.

Text provide 122952 Amended table 25.32

See ordinances creating and amending sections for complete text, graphics, sections for complete text, graphics, and tables and to confirm accuracy of this source file.